

Delft University of Technology

### Martingales and stochastic calculus in Banach spaces

Yaroslavtsev, Ivan

DOI 10.4233/uuid:6c3937b1-aa5b-4860-a58c-f57e87518ce9

Publication date 2019 **Document Version** 

Final published version

Citation (APA) Yaroslavtsev, I. (2019). Martingales and stochastic calculus in Banach spaces. [Dissertation (TU Delft), Delft University of Technology]. https://doi.org/10.4233/uuid:6c3937b1-aa5b-4860-a58c-f57e87518ce9

### Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

#### Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

This work is downloaded from Delft University of Technology. For technical reasons the number of authors shown on this cover page is limited to a maximum of 10.

# MARTINGALES AND STOCHASTIC CALCULUS IN BANACH SPACES

### MARTINGALES AND STOCHASTIC CALCULUS IN BANACH SPACES

# Proefschrift

ter verkrijging van de graad van doctor aan de Technische Universiteit Delft, op gezag van de Rector Magnificus Prof. dr. ir. T.H.J.J. van der Hagen, voorzitter van het College voor Promoties, in het openbaar te verdedigen op woensdag 13 maart 2019 om 10:00 uur

door

### Ivan YAROSLAVTSEV

Specialist in Mathematics Lomonosov Moscow State University, Russia geboren te Neftekamsk, Sovjet-Unie Dit proefschrift is goedgekeurd door de

promotor: Prof. dr. J.M.A.M. van Neerven promotor: Prof. dr. ir. M.C. Veraar

Samenstelling promotiecommissie:

Rector Magnificus,	voorzitter
Prof. dr. J.M.A.M. van Neerven	Technische Universiteit Delft
Prof. dr. ir. M.C. Veraar	Technische Universiteit Delft

*Onafhankelijke leden:* Prof. dr. E.A. Cator Prof. dr. S. Dirksen Prof. dr. S. Geiss Prof. dr. ir. G. Jongbloed Prof. dr. A. Osękowski Prof. dr. B. de Pagter

Radboud Universiteit Nijmegen RWTH Aachen Universiteit, Duitsland Universiteit van Jyväskylä, Finland Technische Universiteit Delft Universiteit van Warschau, Polen Technische Universiteit Delft, reservelid



- *Keywords:* martingales, UMD Banach spaces, Fourier multipliers, martingale decompositions, weak differential subordination, Burkholder-Davis-Gundy inequalities, stochastic integration, random measures, Novikov inequalities, Burkholder-Rosenthal inequalities, Hilbert transform
- Printed by: Ipskamp Printing
- *Cover:* Designed by canva.com

ISBN 978-94-028-1398-2

Copyright © 2019 by I.S. Yaroslavtsev

# **CONTENTS**

Ι

Sı	ımma	ary		ix
Sa	men	vatting		xiii
In	trodu	iction		1
1	Intr	oductio	on	3
	1.1	Weak	differential subordination	. 3
		1.1.1	Discrete case	. 4
		1.1.2	Continuous-time case	. 5
	1.2	Marti	ngale decompositions	. 6
		1.2.1	Meyer-Yoeurp decomposition	. 6
		1.2.2	The canonical decomposition	. 7
	1.3	Burkh	nolder–Davis–Gundy inequalities. Stochastic integration	. 8
		1.3.1	General right-hand side	. 9
		1.3.2	Predictable right-hand side	. 10
	1.4	Misce	llanea	. 10
		1.4.1	Fourier multipliers	. 11
		1.4.2	Hilbert transform and orthogonal martingales	. 11
		1.4.3	Burkholder-Rosenthal inequalities.	. 12
		1.4.4	Random measures	. 13
		1.4.5	Bellman functions	. 13
	1.5	What	is not in the thesis	. 14
		1.5.1	Cylindrical continuous martingales and stochastic integra-	
			tion, paper [177]	. 14
		1.5.2	Brownian representations of cylindrical continuous local mar-	
			tingales, paper [186]	. 15
		1.5.3	Even Fourier multipliers and martingale transforms, paper	
			[188]	. 15
2	Prel	liminar	ies	17
	2.1	Basic	notions on stochastic processes	. 17
	2.2	Marti	ngales	. 18
		2.2.1	Quadratic variation.	. 20
		2.2.2	Continuous martingales	. 21
		2.2.3	Purely discontinuous martingales. Meyer-Yoeurp decompo-	
			sition	21

		2.3	UMD	Banach spaces	. 23
		2.4	Stopp	ing times	. 24
			2.4.1	Predictable and totally inaccessible stopping times	. 25
			2.4.2	Quasi-left continuous martingales and martingales with ac-	
				cessible jumps	. 27
			2.4.3	The canonical decomposition	. 27
			2.4.4	Time-change	. 31
		2.5	Stocha	astic integration	. 32
		2.6	Multio	limensional Wiener process	. 33
		2.7	Brown	vian representation	. 33
		2.8	Rando	om measures	. 33
		2.9	γ-rado	onifying operators	. 35
		2.10	Conve	ex, concave, biconcave, zigzag-concave functions	. 35
		2.11	Corres	sponding dual basis	. 36
		2.12	Itô's fe	ormula	. 37
Π	W	leak d	lifferer	ntial subordination and the canonical decomposition of mar-	
	tin	ngales	6		39
	3	Wea	k diffe	rential subordination of discrete and purely discontinuous	
		mar	tingale	S	41
		3.1	Introd	uction	. 42
		3.2	Prelin	linaries	. 43
		3.3	UMD	Banach spaces and weak differential subordination	. 45
			3.3.1	Discrete case	. 45
			3.3.2	Continuous time case.	. 53
		3.4	Fourie	er multipliers	. 55
			3.4.1	Basic definitions and the main theorem	. 56
			3.4.2	Examples of Theorem 3.4.1.	. 60
		3.5	Hilber	t transform and general conjecture	. 61
			3.5.1	Hilbert transform and Burkholder functions	. 61
			3.5.2	General conjecture	. 65
	4	I <sup>p</sup> -e	stimate	es for weak differential subordination and for martingale	
	•	deco	mposi	tions	69
		4.1	Introd	uction	. 70
		42	Prelim	ninaries	. 70
		4.3	UMD	Banach spaces and martingale decompositions	. 72
		1.0	4.3.1	Mever-Yoeurn decomposition in UMD case	. 73
			4.3.2	Yoeurp decomposition of purely discontinuous martingales	. 81
			433	Stochastic integration	
		44	Weak	differential subordination and general martingales	. 05 84
		1.1	i i cuit	anterentari Suborantation and general martingates	· 01

5	Exis	tence of the canonical decomposition and weak $L^1$ -estimates	91
	5.1	Introduction	92
	5.2	Preliminaries	96
		5.2.1 Martingales and càdlàg processes	96
		5.2.2 Compensator and variation	97
	5.3	Gundy's decomposition of continuous-time martingales	97
	5.4	The canonical decomposition of local martingales	101
		5.4.1 Weak differential subordination martingale transforms	102
		5.4.2 Sufficiency of the UMD property	108
		5.4.3 Necessity of the UMD property	116
6	Ortl	nogonal martingales and the Hilbert transform	123
	6.1	Introduction	124
	6.2	Preliminaries	126
		6.2.1 Periodic Hilbert transform	126
		6.2.2 Orthogonal martingales	126
		6.2.3 Subharmonic and plurisubharmonic functions	128
		6.2.4 Meyer-Yoeurp decomposition	130
	6.3	Main theorem	131
	6.4	Applications	143
		6.4.1 Hilbert transforms on $\mathbb{T}$ , $\mathbb{R}$ , and $\mathbb{Z}$	143
		6.4.2 Decoupling constants	151
		6.4.3 Necessity of the UMD property	152
		6.4.4 Weak differential subordination of martingales: sharper <i>L</i> <sup>1</sup>	<sup>0</sup> _
		inequalities	157
		6.4.5 Weak differential subordination of harmonic functions	161
		6.4.6 Inequalities for singular integral operators	165
		6.4.7 Hilbert operators	166
III S	Stoch	astic integration and Burkholder–Davis–Gundy inequalities	169
7	L <sup>q</sup> -v	valued Burkholder-Rosenthal inequalities and sharp estimates for	or
	stoc	hastic integrals	171
	7.1	Introduction	172
	7.2	Preliminaries	176
	7.3	$L^q$ -valued Burkholder-Rosenthal inequalities	178
	7.4	The dual of $H_p^{S_q}(X)$	183
	7.5	Sharp bounds for $L^q$ -valued stochastic integrals	189
		7.5.1 Decomposition of stochastic integrals	189
		7.5.2 Purely discontinuous martingales with accessible jumps .	190
		7.5.3 Quasi-left continuous purely discontinuous martingales	197
		7.5.4 Integrals with respect to random measures	199

		7.5.5 Integration with respect to continuous martingales 20			
		7.5.6 Integration with respect to general local martingales. $\dots$ 20			
	7.A	Duals of $\mathscr{S}_{q}^{p}$ , $\mathscr{D}_{q,q}^{p}$ , $\mathscr{D}_{p,q}^{p}$ , $\mathscr{S}_{q}^{p}$ , $\mathscr{D}_{q,q}^{p}$ , and $\mathscr{D}_{p,q}^{p}$			
		7.A.1 $\mathscr{D}_{q,q}^{p}$ and $\mathscr{D}_{p,q}^{p}$ spaces			
		7.A.2 $\mathscr{S}_q^p$ and $\widehat{\mathscr{S}}_q^p$ spaces			
8	Bur	xholder–Davis–Gundy inequalities in UMD Banach function spaces 22			
	8.1	Introduction			
	8.2	Preliminaries			
	8.3	Lattice Doob's maximal inequality			
	8.4	Main result			
9	Bur	cholder–Davis–Gundy inequalities and stochastic integration in gen-			
	eral	UMD Banach spaces 24			
	9.1	Introduction			
	9.2	Burkholder–Davis–Gundy inequalities: the discrete time case 24			
	9.3	Gaussian characteristics			
		9.3.1 Basic definitions			
		9.3.2 Dasic properties of $\gamma(\cdot)$			
	0.4	9.5.5 Finite dimensional case			
	9.4 9.5	Burkholder-Davis-Cundy inequalities: the continuous-time case 25			
	9.5	Ramifications of Theorem 9.5.1			
	2.0	961 Continuous and purely discontinuous martingales 26			
		9.6.2 Martingales with independent increments			
	9.7	Applications and miscellanea			
		9.7.1 Itô isomorphism: general martingales			
		9.7.2 Itô isomorphism: Poisson and general random measures 26			
		9.7.3 Necessity of the UMD property			
		9.7.4 Martingale domination			
		9.7.5 Martingale approximations			
		9.7.6 The canonical decomposition			
		9.7.7 Covariation bilinear forms for pairs of martingales			
	9.8	UMD Banach function spaces			
In	dex	27			
Re	eferer	ces 28			
A	cknov	vledgments 29			
Cı	Curriculum Vitæ 29				
Li	List of Publications 30				

# **SUMMARY**

In this thesis we study martingales and stochastic integration of processes with values in UMD Banach spaces. Recall that for a Banach space *X*, a stochastic process  $M : \mathbb{R}_+ \times \Omega \to X$  is called a *martingale* if

$$\mathbb{E}(M_t|\mathscr{F}_s) = M_s, \ 0 \le s \le t.$$

A Banach space *X* has the *UMD property* if and only if the Hilbert transform is bounded on  $L^p(\mathbb{R}; X)$  for all (equivalently, for some) 1 .

The thesis has three parts. Part I gives an introduction to the material covered in Part II and Part III. Part II is devoted to new properties and corresponding inequalities of martingales themselves. First in Chapter 3 and 4 we extend the notion of *differential subordination* to infinite dimensions. For two real-valued martingales M and N we say that N is *differentially subordinate* to M (we will denote this by  $N \ll M$ ) if a.s.  $|N_0| \le |M_0|$  and

$$t \mapsto [M]_t - [N_t]$$
 is nondecreasing in  $t \ge 0$ ,

where [*M*] and [*N*] are *quadratic variations* of *M* and *N*, respectively. Burkholder [33] and Wang [179] showed that the following  $L^p$  inequality holds true for any 1

$$\mathbb{E}|N_t|^p \le (p^* - 1)^p \mathbb{E}|M_t|^p, \ t \ge 0,$$
(S.1)

where  $p^* := \max\{p, p/(p-1)\}$ . These inequalities have been widely used in harmonic analysis (see e.g. [7, 9, 10, 14, 15, 79, 140] and references therein). Note that Wang [179] extended (S.1) to the Hilbertian setting. Unfortunately, due to Kwapień's result [101] one can not prove an analogue of (S.1) for more general Banach spaces. Surprisingly, in many applications one has differential subordination of its *weak form* (i.e. under actions of linear functionals). Therefore, we define *weak differential subordination*: for a given Banach space *X* an *X*-valued martingale *N* is *weakly differentially subordinate* to an *X*-valued martingale *M* (we will denote this by  $N \stackrel{w}{\ll} M$ ) if  $\langle N, x^* \rangle \ll \langle M, x^* \rangle$  for all  $x^* \in X^*$ . In Chapter 3 and 4 we show that for any  $1 , <math>L^p$ -estimates for weakly differentially subordinated martingales exist if and only if *X* has *the UMD property* and the constant  $c_{p,X}$  in the corresponding inequality

$$\mathbb{E}\|N_t\|^p \le c_{p,X}^p \mathbb{E}\|M_t\|^p, \ t \ge 0,$$
(S.2)

can be characterized in terms of the  $UMD_p$  constant  $\beta_{p,X}$  of X (recall that  $\beta_{p,X}$  expresses the norm of a certain martingale transform and it is finite if and only if X has the UMD property).

In Chapter 6 we show that weak differential subordination together with *orthogonality* of martingales is closely related with the Hilbert transform. More specifically, we show that for any Banach space *X*, for any *X*-valued orthogonal martingales *M* and *N* with  $N \stackrel{w}{\ll} M$ , and for any convex functions  $\Phi, \Psi : X \to \mathbb{R}_+$  with  $\Psi(0) = 0$  the following inequality holds true

$$\mathbb{E}\Psi(N_t) \le C_{\Phi,\Psi,X} \mathbb{E}\Phi(M_t), \quad t \ge 0, \tag{S.3}$$

where the sharp constant  $C_{\Phi,\Psi,X} \in [0,\infty]$  coincides with the  $\Phi, \Psi$ -norm of the periodic Hilbert transform  $\mathscr{H}^{\mathbb{T}}$ 

$$|\mathscr{H}^{\mathbb{T}}|_{\Phi,\Psi} := \sup_{f:\mathbb{T}\to X \text{ step}} \frac{\int_{\mathbb{T}} \Psi(\mathscr{H}^{\mathbb{T}}f(s)) \,\mathrm{d}s}{\int_{\mathbb{T}} \Phi(f(s)) \,\mathrm{d}s}.$$

Inequality (S.3) has several applications outlined in Section 6.4. In particular, it is shown that the optimal  $c_{p,X}$  in (S.2) is of the order max{ $\beta_{p,X}, \hbar_{p,X}$ }, where  $\hbar_{p,X}$  is the norm of  $\mathscr{H}^{\mathbb{T}}$  on  $L^{p}(\mathbb{T}; X)$ .

Another topic described in Part II is the *canonical decomposition* of local martingales. The canonical decomposition as a natural extension of Lévy-Itô decomposition first appeared in the paper [190] by Yoeurp, and it has the following form. A local martingale M is said to have a canonical decomposition if there exist a continuous local martingale  $M^c$  (a Wiener-like part), a purely discontinuous quasi-left continuous local martingale  $M^q$  (a Poisson-like part, which jumps at non-predictable stopping times), and a purely discontinuous local martingale  $M^a$ with accessible jumps (a discrete-like part, which jumps only at certain predictable stopping times) such that  $M_0^c = M_0^q = 0$  and  $M = M^c + M^q + M^a$ . In the same paper [190] Yoeurp showed existence and uniqueness of the canonical decomposition for any real-valued martingale. In Chapter 4 and 5 we show that for a Banach space X the following are equivalent

- *X* is UMD;
- any *X*-valued local martingale admits the canonical decomposition.

Moreover, if *X* is UMD, then the following estimates hold for any  $i \in \{c, q, a\}$ 

$$\mathbb{E} \|M_t^i\|^p \le \beta_{p,X}^p \mathbb{E} \|M_t\|^p, \quad t \ge 0, \quad 1 
$$\lambda \mathbb{P} \left( (M^i)_t^* > \lambda \right) \lesssim_X \mathbb{E} \|M_t\|, \quad t \ge 0, \quad \lambda > 0.$$$$

Note that the canonical decomposition is exceptionally important for stochastic integration (see Chapter 7).

Part III is devoted to sharp bounds for stochastic integrals and Burkholder– Davis–Gundy inequalities. Namely, we try to find an answer to the following question. Given a (UMD) Banach space *X*, a real-valued martingale *M*, an elementary predictable *X*-valued process  $\Phi$ , and p > 0. How do sharp bounds for  $\sup_{t>0} \| \int_0^t \Phi dM \|^p$  look like?

First the answer for this question was given by van Neerven, Veraar, and Weis in [126] in the case M = W is a standard Brownian motion. In this setting one has that

$$\sup_{t\geq 0} \left\| \int_0^t \Phi \, \mathrm{d}W \right\|^p \sim_{p,X} \mathbb{E} \|\Phi\|_{\gamma(L^2(\mathbb{R}_+),X)}^p, \tag{S.4}$$

where  $\|\Phi\|_{\gamma(L^2(\mathbb{R}_+),X)}$  is the  $\gamma$ -norm of  $\Phi$  which e.g. coincides with the Hilbert–Schmidt norm if X is a Hilbert space. Later in [175, 177] (S.4) was extended to stochastic integrals with respect to *continuous* martingales.

In Part III we extend (S.4) in two ways. First, in Chapter 7 in the case  $X = L^q(S)$ ,  $1 < q < \infty$ , for a *general* real-valued martingale M we find a *predictable* norm  $\|\cdot\|_{M,p,q}$  (i.e. the process  $t \mapsto \||\Phi \mathbf{1}_{[0,t]}\||_{M,p,q}$ ,  $t \ge 0$ , is predictable for any elementary predictable X-valued  $\Phi$ ) such that for any 1

$$\sup_{t\geq 0} \left\| \int_0^t \Phi \, \mathrm{d}M \right\|^p \sim_{p,q} \mathbb{E} \, \|\!|\!|\!| \Phi \, \|\!|\!|_{M,p,q}^p.$$

Though the norm  $\|\cdot\|_{M,p,q}$  has a complicated form (which depends on the mutual positions of p, q, and 2), the latter inequalities have two major features: they are sharp (since they are two-sided) and their right-hand side as a predictable process is locally bonded by any a priori given number (up to a stopping time), which is useful in SPDE's for a fixed point argument. It remains open how an analogue of  $\|\cdot\|_{M,p,q}$  for more general Banach spaces looks like.

If we omit the predictability assumption, then we end up with *Burkholder–Davis–Gundy* inequalities. Recall that Burkholder, Davis, and Gundy proved in [40] that for any real-valued martingale *N* and for any  $1 \le p < \infty$  one has that

$$\mathbb{E}\sup_{t\geq 0}|N_t|^p \approx_p \mathbb{E}[N]_{\infty}^{p/2}.$$
(S.5)

Thus for any real-valued martingale M and for any real-valued elementary predictable process  $\Phi$  one has the following two-sided inequalities

$$\mathbb{E}\sup_{t\geq 0} \left| \int_0^t \Phi(s) \, \mathrm{d}M_s \right|^p \approx_p \mathbb{E} \int_0^\infty \Phi(s)^2 \, \mathrm{d}[M]_s.$$
(S.6)

In order to extend (S.6) to general Banach spaces we extend (S.5) to general Banach spaces. First in Chapter 8 we show that if *X* is a UMD Banach *function* space over a measure space (S,  $\Sigma$ ,  $\mu$ ) (i.e. a Banach space consisting of measurable functions on *S*), then for any *X*-valued martingale *N* and for any 1

$$\mathbb{E}\sup_{t\geq 0} \|N_t\|^p \approx_{p,X} \mathbb{E}\left\| [N]_{\infty}^{1/2} \right\|^p, \tag{S.7}$$

where the quadratic variation  $[N]_{\infty}$  is taken pointwise on *S*. In Chapter 9 we present a more general, but a more complicated version of (S.7). We prove that for any UMD Banach space *X* and for any  $t \ge 0$ , any *X*-valued martingale *N* has a *covariation bilinear form*  $[[N]]_t$  satisfying a.s.

$$[[N]]_t(x^*, x^*) = [\langle N, x^* \rangle]_t, \quad x^* \in X^*$$

Moreover, a.e. in  $\Omega$  there exists an *X*-valued centred Gaussian random variable  $\xi_{[[N]]_t}$  having  $[[N]]_t$  as its covariance bilinear form:

$$[[N]]_t(x^*, x^*) = \mathbb{E}_{\xi} |\langle \xi_{[[N]]_t}, x^* \rangle|^2, \ x^* \in X^*,$$

and if one denotes  $(\mathbb{E}_{\xi} \|\xi_{[[N]]_t}\|^2)^{1/2}$  by  $\gamma([[N]]_t)$ , then the following holds true for any  $1 \le p < \infty$ 

$$\mathbb{E} \sup_{0 \le s \le t} \|N_s\|^p \approx_{p,X} \mathbb{E} \gamma([[N]]_t)^p.$$
(S.8)

In particular, if  $N = \int \Phi dM$  for some real-valued martingale *M* and for some elementary predictable *X*-valued  $\Phi$ , then (S.8) implies that for any  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{t\geq 0}\left\|\int_0^t \Phi \,\mathrm{d} M\right\|^p \approx_{p,X} \mathbb{E}\|\Phi\|_{\gamma(L^2(\mathbb{R}_+,[M]),X)}^p,$$

which fully extends (S.4).

# SAMENVATTING

In dit proefschrift bestuderen we martingalen en stochastische integralen van processen met waarden in UMD Banachruimten. Voor een Banachruimte *X* wordt een stochastisch proces  $M : \mathbb{R}_+ \times \Omega \to X$  een martingaal genoemd indien

$$\mathbb{E}(M_t|\mathscr{F}_s) = M_s, \ 0 \le s \le t.$$

Een Banachruimte *X* heeft de UMD eigenschap dan en slechts dan als de Hilberttransformatie begrensd is op  $L^p(\mathbb{R}; X)$  voor iedere (equivalent, voor een) 1 .

Het proefschrift heeft twee hoofddelen: Deel II en Deel III. Deel II gaat over nieuwe eigenschappen van martingalen en de bijbehorende ongelijkheden. Eerst in Hoofdstuk 3 en later in 4 breiden we het begrip differentiële subordinatie uit naar oneindige dimensies. Voor twee reëel-waardige martingalen *M* en *N* zeggen we dat *N differentieel gesubordineerd* wordt door *M* (dit noteren we met  $N \ll M$ ) als b.z.  $|N_0| \le |M_0|$  en

 $t \mapsto [M]_t - [N_t]$  is niet-dalend in  $t \ge 0$ ,

waarbij [*M*] en [*N*] de *kwadratische variatie* van *M* en *N* zijn. Burkholder [33] en Wang [179] hebben laten zien dat de volgende  $L^p$  ongelijkheden gelden voor iedere 1

$$\mathbb{E}|N_t|^p \le (p^* - 1)^p \mathbb{E}|M_t|^p, \ t \ge 0,$$
(S.1)

waarbij  $p^* := \max\{p, p/(p-1)\}$ . Deze ongelijkheden worden veel gebruikt in de harmonische analyse (zie bijv. [7, 9, 10, 14, 15, 79, 140] en de referenties daarin). Merk op dat Wang [179] (S.1) naar de Hilbertwaardige setting heeft uitgebreid. Helaas, volgt uit Kwapień's resultaat [101] dat het analagon van (S.1) niet geldt voor algemenre Banachruimten. Het is verrassend dat in veel toepassingen we differentiële subordinatie in *zwakke vorm* hebben (d.w.z. na toepassing van een lineaire functionaal). Daarom definiëren we *zwakke differentiële subordinatie*: voor een gegeven Banachruimte *X* noemen we een *X*-waardige martingaal *N* is *zwak differentieel gesubordineerd* ten aanzien van een *X*-waardige martingaal *M* (notatie  $N \stackrel{w}{\ll} M$ ) als  $\langle N, x^* \rangle \ll \langle M, x^* \rangle$  voor alle  $x^* \in X^*$ . In Hoofdstuk 3 en 4 laten we zien dat er voor elke  $1 , <math>L^p$ -afschattingen voor zwak differentieel gesubordineerde martingalen gelden dan en slechts dan als *X* voldoet aan de *UMD eigenschap* en de constanten  $c_{p,X}$  in de ongelijkheid

$$\mathbb{E}\|N_t\|^p \le c_{p,X}^p \mathbb{E}\|M_t\|^p, \ t \ge 0,$$
(S.2)

kunnen worden gekarakteriseerd in termen van de  $UMD_p$  constante  $\beta_{p,X}$  van X (herinner dat  $\beta_{p,X}$  is de norm van een bepaalde martingaaltransformatie en is eindig dan en slechts dan als X voldoet aan de UMD eigenschap).

In Hoofdstuk 6 laten we zien dat zwakke differentiële subordinatie en *orthogonaliteit* van martingalen sterk gerelateerd is aan de begrensdheid van de Hilberttransformatie. Preciezer laten we zien dat voor iedere Banachruimte *X*, voor alle *X*-waardige orthogonale martingalen *M* en *N* met  $N \stackrel{w}{\ll} M$ , en voor iedere convexe functie  $\Phi, \Psi : X \to \mathbb{R}_+$  met  $\Psi(0) = 0$  de volgende ongelijkheid geldt

$$\mathbb{E}\Psi(N_t) \le C_{\Phi,\Psi,X} \mathbb{E}\Phi(M_t), \quad t \ge 0, \tag{S.3}$$

waarbij de optimale constante  $C_{\Phi,\Psi,X} \in [0,\infty]$  overeenkomt met de  $\Phi,\Psi$ -norm van de periodieke Hilberttransformatie  $\mathscr{H}^{\mathbb{T}}$ 

$$|\mathscr{H}^{\mathbb{T}}|_{\Phi,\Psi} := \sup_{f:\mathbb{T}\to X \text{ stap}} \frac{\int_{\mathbb{T}} \Psi(\mathscr{H}^{\mathbb{T}}f(s)) \,\mathrm{d}s}{\int_{\mathbb{T}} \Phi(f(s)) \,\mathrm{d}s}.$$

Ongelijkheid (S.3) heeft verschillende toepassingen zoals uitgelegd in Sectie 6.4. In het bijzonder wordt daar bewezen dat de optimale constante  $c_{p,X}$  in (S.2) van de orde max{ $\beta_{p,X}, \hbar_{p,X}$ } is, waarbij  $\hbar_{p,X}$  de norm van  $\mathscr{H}^{\mathbb{T}}$  op  $L^{p}(\mathbb{T}; X)$  is.

Een ander onderwerp in Part II is de *canonieke decompositie* van lokale martingalen. De canonieke decompositie als uitbreiding van de Lévy-Itô decompositie verscheen voor het eerst in het artikel [190] van Yoeurp, en heeft de volgende vorm. Een lokale martingaal heeft een canonieke decompositie als er een continue lokale martingaal  $M^c$  bestaat (een Wiener-achtig deel), een puur discontinue quasi-links continue lokale martingaal  $M^q$  (een Poisson-achtig deel dat springt op niet-voorspelbare stoptijden), en een puur discontinue lokale martingaal  $M^a$  met toegankelijke sprongen (een discreet-achtig deel, met sprongen op voorspelbare stoptijden) zó dat  $M_0^c = M_0^q = 0$  en  $M = M^c + M^q + M^a$ . In hetzelfde artikel [190] heeft Yoeurp existentie en eenduidigheid van de canonieke decompositie voor een willekeurige reëel-waardige martingaal laten zien. In Hoofdstuk 4 en 5 laten we zien dat voor een Banachruimte *X* de volgende eigenschappen equivalent zijn:

- *X* is UMD;
- iedere *X*-waardige lokale martingaal heeft een canonieke decompositie.

Bovendien geldt dat als *X* UMD is en  $i \in \{c, q, a\}$ , de volgende afschattingen gelden:

$$\mathbb{E} \|M_t^i\|^p \le \beta_{p,X}^p \mathbb{E} \|M_t\|^p, \quad t \ge 0, \quad 1 
$$\lambda \mathbb{P} \left( (M^i)_t^* > \lambda \right) \lesssim_X \mathbb{E} \|M_t\|, \quad t \ge 0, \quad \lambda > 0.$$$$

De canonieke decompositie is extreem belangrijk voor stochastische integratie (zie Hoofdstuk 7).

Deel III is gewijd aan scherpe afschattingen voor stochastische integralen en Burkholder–Davis–Gundy ongelijkheden. We proberen namelijk om de volgende vraag te beantwoorden. Gegegeven een (UMD) Banachruimte *X*, een reëel-waardige martingaal *M*, een elementair voorspelbaar *X*-waardig proces  $\Phi$ , en *p* > 0. Hoe zien twee-zijdige afschattingen voor  $\sup_{t>0} \|\int_0^t \Phi dM\|^p$  er uit?

Allereerst was deze vraag beantwoord door Neerven, Veraar, en Weis in [126] in het geval M = W een standaard Brownse beweging is. In deze setting geldt dat

$$\sup_{t\geq 0} \left\| \int_0^t \Phi \, \mathrm{d}W \right\|^p \approx_{p,X} \mathbb{E} \|\Phi\|_{\gamma(L^2(\mathbb{R}_+),X)}^p, \tag{S.4}$$

waarbij  $\|\Phi\|_{\gamma(L^2(\mathbb{R}_+),X)}$  de  $\gamma$ -norm van  $\Phi$  is, welke bijv. overeenkomt met de Hilbert– Schmidt norm als X een Hilbertruimte is. Daarna is (S.4) in [175, 177] uitgebreid naar stochastische integralen ten aanzien van *continue* martingalen.

In Deel III breiden we (S.4) uit op twee manieren. Ten eerste in Hoofdstuk 7 in het geval  $X = L^q(S)$ ,  $1 < q < \infty$ , voor een *algemene* reëel-waardige martingaal M vinden we een *voorspelbare* norm  $\|\cdot\|_{M,p,q}$  (d.w.z. het proces  $t \mapsto \||\Phi \mathbf{1}_{[0,t]}\||_{M,p,q'}$  $t \ge 0$ , is voorspelbaar voor iedere elementaire voorspelbare X-waardige  $\Phi$ ) zó dat voor elke 1

$$\sup_{t\geq 0} \left\| \int_0^t \Phi \, \mathrm{d}M \right\|^p \approx_{p,q} \mathbb{E} \, \| \Phi \|_{M,p,q}^p.$$

Hoewel de norm  $\|\cdot\|_{M,p,q}$  een gecompliceerde vorm heeft (die afhangt van de wederzijde posities van p, q, en 2), hebben de genoemde ongelijkheden twee belangrijke kenmerken: ze zijn optimaal (want twee-zijdig) en de rechterzijde is als voorspelbaar proces lokaal begrensd door een willekeurig getal (tot en met een stoptijd), wat handig is in dekpuntargumenten voor SPDV's. Het blijft een open probleem hoe  $\|\cdot\|_{M,p,q}$  eruit ziet voor algemenere Banachruimten.

Indien we de voorspelbaarheidseis weglaten, dan kunnen we de *Burkholder–Davis–Gundy* ongelijkheden gebruiken. Herinner dat Burkholder, Davis, en Gundy in [40] hebben bewezen dat voor iedere reëel-waardige martingaal N en voor elke  $1 \le p < \infty$  geldt dat

$$\mathbb{E}\sup_{t>0}|N_t|^p \approx_p \mathbb{E}[N]_{\infty}^{p/2}.$$
(S.5)

Dus voor elke reëel-waardige martingaal M en voor elke reëel-waardig elementair voorpelbaar proces  $\Phi$  geldt de volgende twee-zijdige afschatting

$$\mathbb{E}\sup_{t\geq 0} \left| \int_0^t \Phi(s) \, \mathrm{d}M_s \right|^p \approx_p \mathbb{E} \int_0^\infty \Phi(s)^2 \, \mathrm{d}[M]_s.$$
(S.6)

Om (S.6) uit te breiden naar algemenere Banachruimten, breiden we (S.5) uit naar algemenere Banachruimten. Eerst laten we in Hoofdstuk 8 zien dat als *X* een UMD *Banachfunctieruimte* over een maatruimte ( $S, \Sigma, \mu$ ) is (d.w.z. een Banachruimte bestaande uit meetbare functies op *S*), dan geldt voor iedere *X*-waardige martingaal *N* en voor iedere 1 dat

$$\mathbb{E}\sup_{t\geq 0} \|N_t\|^p \approx_{p,X} \mathbb{E}\left\| [N]_{\infty}^{1/2} \right\|^p, \tag{S.7}$$

waarbij de kwadratische variatie  $[N]_{\infty}$  puntsgewijs op *S* genomen wordt. In Hoofdstuk 9 presenteren we een algemenere, maar ook ingewikkeldere versie van (S.7). We bewijzen voor elke UMD Banachruimte *X* en voor elke  $t \ge 0$  dat voor iedere *X*-waardige martingaal *N* een *covariatie bilineaire vorm*  $[[N]]_t$  bestaat zó dat b.z.

$$[[N]]_t(x^*, x^*) = [\langle N, x^* \rangle]_t, \quad x^* \in X^*$$

Bovendien geldt dat er b.o. in  $\Omega$  een *X*-waardige gecentreerde Gaussische stochast  $\xi_{[[N]]_t}$  bestaat zó dat de covariantie bilineare vorm  $[[N]]_t$  voldoet aan :

$$[[N]]_{t}(x^{*}, x^{*}) = \mathbb{E}_{\xi} |\langle \xi_{[[N]]_{t}}, x^{*} \rangle|^{2}, \quad x^{*} \in X^{*},$$

en als we  $(\mathbb{E}_{\xi} \| \xi_{[[N]]_t} \|^2)^{1/2}$  schrijven als  $\gamma([[N]]_t)$ , dan geldt het volgende voor iedere  $1 \le p < \infty$ 

$$\mathbb{E} \sup_{0 \le s \le t} \|N_s\|^p \approx_{p,X} \mathbb{E} \gamma([[N]]_t)^p.$$
(S.8)

In het bijzonder als  $N = \int \Phi dM$  waarbij M een reëel-waardige martingaal en  $\Phi$  een elementair voorspelbaar X-waardig proces, dan volgt uit (S.8) dat voor alle  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{t\geq 0}\left\|\int_{0}^{t}\Phi\,\mathrm{d}M\right\|^{p} \approx_{p,X} \mathbb{E}\|\Phi\|_{\gamma(L^{2}(\mathbb{R}_{+},[M]),X)}^{p},$$

wat (S.4) volledig generaliseerd.

# Ι

# INTRODUCTION

# **1** Introduction

Let *X* be a Banach space,  $(\Omega, \mathscr{F}, \mathbb{P})$  be a probability space with a filtration  $\mathbb{F} = (\mathscr{F}_t)_{t \ge 0}$ . A stochastic process  $M : \mathbb{R}_+ \times \Omega \to X$  is called a *martingale* if  $\mathbb{E}(M_t | \mathscr{F}_s) = M_s$  for all  $0 \le s \le t$  (see Section 2.2).

The notion of martingale was introduced by Paul Lévy in 1934, and nowadays it plays an important rôle in probability theory, stochastic analysis, functional analysis, harmonic analysis, complex analysis, and in such applied areas as physics and finance, where martingales are often used as a natural model of a noise. Even though real-valued martingales are of bigger interest, Banach space-valued martingales appear naturally and are of exceptional importance while one needs to extend a theoretical result involving martingales to an infinite-dimensional setting.

The present thesis is devoted to new properties of and new methods while working with Banach space-valued martingales, and it combines papers [54, 146, 178, 184, 185, 187, 189].

Let us outline the main results of the thesis. It is worth noticing that almost all the presented results assume the so-called  $UMD^1$  property. This property is very natural for Banach spaces when one works with martingales. In particular, due to Bourgain [23] and Burkholder [32] having the UMD property for a Banach space X is equivalent to the boundedness of the Hilbert transform on  $L^p(\mathbb{R}; X)$  for all (equivalently for some) 1 . We refer the reader to Section 2.3 for details onUMD Banach spaces.

### **1.1.** WEAK DIFFERENTIAL SUBORDINATION

Differential subordination of martingales was introduced by Burkholder in [33] as a natural way of martingale domination. It turned out that real-valued differentially subordinated martingales appear inherently in harmonic analysis (see e.g. [9, 10, 12, 13, 133, 139, 140, 145]). Due to the aforementioned references sharp  $L^{p}$ bounds for differentially subordinated martingales (also under different types of additional assumptions) are of great interest. Here we extend differential subordination to infinite dimensions (this extension is called *weak differential subordination*), and provide  $L^{p}$ -estimates for weakly differentially subordinated martingales. First

<sup>&</sup>lt;sup>1</sup>UMD stands for unconditional martingale differences

let us explain the discrete setting as a demonstration, and then we will turn to the continuous-time setting (note that the continuous-time case is more important for applications).

### 1.1.1. Discrete case

Let  $(d_n)_{n\geq 0}$ ,  $(e_n)_{n\geq 0}$  be two *X*-valued martingale difference sequences. Then  $(e_n)_{n\geq 0}$  is called to be *differentially subordinate* to  $(d_n)_{n\geq 0}$  if a.s.

$$\|e_n\| \le \|d_n\|, \quad n \ge 0. \tag{1.1.1}$$

As we already mentioned,  $L^p$ -bounds for differentially subordinated martingales are of importance. In [33] Burkholder showed the following theorem.

**Theorem 1.1.1.** Let  $(d_n)_{n\geq 0}$ ,  $(e_n)_{n\geq 0}$  be two  $\mathbb{R}$ -valued martingale difference sequences such that  $(e_n)_{n\geq 0}$  is called to be differentially subordinate to  $(d_n)_{n\geq 0}$ . Then for each  $p \in (1,\infty)$ ,

$$\mathbb{E}\Big|\sum_{n\geq 0}e_n\Big|^p\leq (p^*-1)^p\mathbb{E}\Big|\sum_{n\geq 0}d_n\Big|^p,$$

where  $p^* = \max\{p, p/(p-1)\}$ , and  $p^* - 1$  is sharp.

Unfortunately, if one wants to broaden the applications of Theorem 1.1.1 to infinite dimensions, one can not apply Theorem 1.1.1 anymore. Therefore we have the following natural question. *Can one extend Theorem 1.1.1 to the general Banach space-valued setting?* Unluckily, due to the following result by Osękowski (see [140, Theorem 3.24(i)]), which is heavily based on Kwapień's paper [101], one can not leave the Hilbertian setting.

**Theorem 1.1.2.** A Banach space X is isomorphic to a Hilbert space if and only if for some (equivalently, for all)  $1 there exists a constant <math>\alpha_{p,X} > 0$  such that for any pair of X-valued martingale difference sequences  $(d_n)_{n\geq 0}$  and  $(e_n)_{n\geq 0}$  with  $(e_n)_{n\geq 0}$  being differentially subordinate to  $(d_n)_{n\geq 0}$  one has that

$$\mathbb{E}\left\|\sum_{n\geq 0}e_n\right\|^p\leq \alpha_{p,X}^p\mathbb{E}\left\|\sum_{n\geq 0}d_n\right\|^p.$$

Thus in order to extend Theorem 1.1.1 to more general Banach spaces one needs to weaken the assumption (1.1.1). We will do this in the following way, which shortly can be explained as "differential subordination under action of any linear functional".

**Definition 1.1.3.** Let *X* be a Banach space. Then  $(e_n)_{n\geq 0}$  is called to be *weakly differentially subordinate* to  $(d_n)_{n\geq 0}$  if for any  $x^* \in X^*$  a.s.

$$|\langle e_n, x^* \rangle| \le |\langle d_n, x^* \rangle|, \quad n \ge 0.$$
(1.1.2)

Notice that  $L^p$ -bounds for weakly differentially subordinated martingale difference sequences imply *X* having the UMD property thanks to its definition (see Section 2.3). In Chapter 3 we show the converse, i.e. we prove that the UMD property yields the desired  $L^p$ -bounds, and that the UMD<sub>p</sub> constant  $\beta_{p,X}$ , the one characterizing the UMD property, is sharp for weak differential subordination.

**Theorem 1.1.4.** A Banach space X is a UMD space if and only if for some (equivalently, for all)  $1 there exists a constant <math>\beta > 0$  such that for all X-valued martingale different sequences  $(d_n)_{n\geq 0}$  and  $(e_n)_{n\geq 0}$  such that  $(e_n)_{n\geq 0}$  is weakly differentially subordinate to  $(d_n)_{n\geq 0}$  one has

$$\mathbb{E}\left\|\sum_{n\geq 0}e_n\right\|^p\leq\beta^p\mathbb{E}\left\|\sum_{n\geq 0}d_n\right\|^p.$$

If this is the case then the smallest admissible  $\beta$  is the UMD constant  $\beta_{p,X}$ .

### 1.1.2. Continuous-time case

The continuous-time case is a bit more complicated then the discrete case. The first question is how to define differential subordination for continuous-time martingales. To this end we will need the notion of *quadratic variation* (see Section 2.2.1). Recall that any martingale  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  has a quadratic variation

$$[M]_t := \mathbb{P} - \lim_{\text{mesh}\to 0} \sum_{n=1}^N |M(t_n) - M(t_{n-1})|^2, \ t \ge 0,$$

where the limit in probability is taken over partitions  $0 = t_0 < ... < t_N = t$ . Quadratic variation is remarkably important for the martingale theory at least because of Burkholder–Davis–Gundy inequalities (see (1.3.2)). Using quadratic variation one can define differential subordination of continuous-time martingales.

**Definition 1.1.5.** Let  $M, N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be martingales. Then N is *differentially sub-ordinate* to M (we will often write  $N \ll M$ ) if  $|N_0| \le |M_0|$  a.s. and for all  $0 \le s \le t$  a.s.  $[N]_t - [N]_s \le [M]_t - [M]_s$ .

This definition is a natural extension of the discrete one. Moreover, due to Wang [179] the following generalization of Theorem 1.1.1 holds.

**Theorem 1.1.6.** Let  $M, N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be martingales such that N is differentially subordinate to M. Then for any 1

$$\mathbb{E}|M_t|^p \le (p^* - 1)^p \mathbb{E}|N_t|^p, \ t \ge 0.$$

Note that Wang actually proved the Hilbert space-valued version of Theorem 1.1.6, where differential subordination is defined analogously Definition 1.1.5 with using quadratic variations of Hilbert space-valued martingales (see (2.2.4)). In order to extend Theorem 1.1.6 we need first to extend Definition 1.1.5. This extension is fully analogous to Definition 1.1.3.

**Definition 1.1.7.** Let *X* be a Banach space,  $M, N : \mathbb{R}_+ \times \Omega \to X$  be martingales. Then *N* is *weakly differentially subordinate* to *M* (we will often write  $N \stackrel{w}{\ll} M$ ) if  $\langle N, x^* \rangle$  is differentially subordinate to  $\langle M, x^* \rangle$  for all  $x^* \in X^*$ .

It turns out that  $L^p$ -estimates hold for weakly differentially subordinated martingales only in UMD Banach spaces and the following theorem holds true (see Chapter 3, 4, and 6). Recall that  $\beta_{p,X}$  is the UMD constant and its boundedness characterizes the UMD property (see Section 2.3).

**Theorem 1.1.8.** Let X be a Banach space, 1 . Then for any martingales <math>M, N:  $\mathbb{R}_+ \times \Omega \rightarrow X$  such that  $N \stackrel{w}{\ll} M$  one has that

$$\mathbb{E}\|N_t\|^p \le c_{p,X}^p \mathbb{E}\|M_t\|^p, \ t \ge 0,$$
(1.1.3)

where the sharp constant  $c_{p,X}$  is within the interval  $[\beta_{p,X}, \beta_{p,X} + \beta_{p,X}^2]$ .

Notice that sharp bounds of  $c_{p,X}$  in terms of  $\beta_{p,X}$  is of big interest due to the open problem concerning bounds of the norm of the Hilbert transform on  $L^p(\mathbb{R}; X)$  in terms of the UMD<sub>*p*</sub>-constant of *X* (see e.g. Subsection 1.4.2), even though one can provide such sharp bounds of  $c_{p,X}$  in terms of  $\beta_{p,X}$  and the Hilbert transform norm (see Subsection 1.4.2 and Chapter 6).

In addition to  $L^p$ -estimates one can show weak  $L^1$ -estimates for weakly differentially subordinated martingales, which we will not present here (see the forthcoming paper [183]).

### **1.2.** MARTINGALE DECOMPOSITIONS

A significant part of the present thesis is devoted to different types of *martingale decompositions*.

### 1.2.1. Meyer-Yoeurp decomposition

Throughout the history continuous martingales used to be much better understood than general martingales. This has several reasons: a continuous martingale is always locally uniformly bounded, its quadratic variation is continuous and hence locally uniformly bounded as well, and after a certain time-change procedure a continuous martingale can be represented as either a stopped Brownian motion (in the one-dimensional case) or as a stochastic integral with respect to a Brownian motion (in the multidimensional case). If one wants to move from continuous to general martingales, then the following reasonable question can be asked. *Is there a linear space of martingales "orthogonal" to continuous martingales?* The definitive answer to this question in the real-valued case was given by Meyer in [122] and Yoeurp in [190]. They proved that any local real-valued martingale Mhas a *unique* decomposition into a sum of a continuous local martingale  $M^c$  with  $M_0^c = 0$  and a *purely discontinuous* local martingale  $M^d$ , i.e. a local martingale  $M^d$  such that its quadratic variation  $[M^d]$  is pure jump.

In Chapter 4 and 5 we extend the result of Meyer and Yoeurp to general UMD Banach spaces. First notice that for any Banach space *X* a local martingale *M* :  $\mathbb{R}_+ \times \Omega \to X$  is called *purely discontinuous* if  $\langle M, x^* \rangle$  is purely discontinuous for all  $x^* \in X^*$ . Then the following theorem holds true (see Subsection 4.3.1 and Section 5.4).

**Theorem 1.2.1.** Let X be a Banach space. Then X has the UMD property if and only if any local martingale  $M : \mathbb{R}_+ \times \Omega \to X$  has the Meyer-Yoeurp decomposition, i.e. there exist an X-valued continuous local martingale  $M^c$  with  $M_0^c = 0$  and an X-valued purely discontinuous local martingale  $M^d$  such that  $M = M^c + M^d$ . Moreover, if this is the case, then for any 1

$$\mathbb{E}\|M_t^c\|^p, \mathbb{E}\|M_t^d\|^p \le c_{p,X}^p \mathbb{E}\|M_t\|^p, \ t \ge 0,$$
(1.2.1)

$$\lambda \mathbb{P}(M_t^{c*} > \lambda), \lambda \mathbb{P}(M_t^{d*} > \lambda) \lesssim_X \mathbb{E} ||M_t||, \quad t \ge 0, \ \lambda > 0,$$

where sharp  $c_{p,X}$  is within the interval  $\left[\frac{\beta_{p,X}-1}{2},\beta_{p,X}\right]$ .

Note that the sharp constant  $c_{p,X}$  in (1.2.1) is known and equals  $\text{UMD}_p^{\{0,1\}}$ constant of *X* (see Subsection 1.5.3 and Remark 4.4.6).

### 1.2.2. The canonical decomposition

Historically there were three main separate types of martingales: continuous martingales, discrete martingales, and integrals with respect to random measures. Continuous martingales enjoy such properties as local  $L^p$ -integrability for any  $1 \le p \le \infty$ , a rather simple time-change argument due to Kazamaki [94], Lévy's characterization of a Brownian motion (see [89, Theorem 18.3]), and Brownian representation (see [93, Theorem 3.4.2]). Discrete martingales are suitable to work with since the filtration is at most countable and in many applications even can be considered finite, so it is often easier to prove a statement in the discrete setting rather than in the general continuous-time one. The theory of quasi-left continuous random measures (or just *random measures*) was discovered by Novikov in [131] and is of particular interest from the practical point of view since this is a logical generalization of Poisson measures. Somehow all these three "martingale worlds" used to be separated and there were no direct connection between them (though discrete martingales have been heavily applied for proving assertions concerning continuous martingales and random measures).

Due to the work [190] of Yoeurp it turned out that all these "martingale worlds" comprise all the martingales. First we give a couple of useful definitions. A process is said to have *accessible jumps* if it jumps only at a certain countable set of

predictable stopping times (i.e. stopping times that can be *announced* by other stopping times, see Subsection 2.4.1). A process is called quasi-left continuous if it does not jump at any predictable stopping time. A classical example of a process with accessible jumps is a process that jumps only at natural points, i.e. at {1,2,3,...}, for instance a discrete martingale. A representative example of a quasi-left continuous process is a Poisson process (literally, one can not predict *when* it will jump). It turns out that any quasi-left continuous purely discontinuous martingale can be naturally represented as a stochastic integral with respect to a random measure, while any purely discontinuous martingale with accessible jumps after a proper approximation and a time-change argument can be represented as a discrete martingale with the same value of jumps. Moreover, thanks to Yoeurp [190] the following theorem holds.

**Theorem 1.2.2** (Canonical decomposition). Let  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a local martingale. Then there exist unique local martingales  $M^c$ ,  $M^q$ , and  $M^a$  such that  $M^c$  is continuous,  $M^q$  is purely discontinuous quasi-left continuous,  $M^a$  is purely discontinuous with accessible jumps,  $M_0^c = M_0^q = 0$  a.s., and  $M = M^c + M^q + M^a$ .

The decomposition in Theorem 1.2.2 is called *canonical* though it would be more correct to call it *Yoeurp*. But historically *Yoeurp decomposition* is a decomposition of a purely discontinuous local martingale into a quasi-left continuous part and a part with accessible jumps (see e.g. [89]).

In Chapter 4 and 5 we show that Theorem 1.2.2 can be extended to UMD Banach space-valued local martingales, and the UMD property here is not only sufficient but necessary. More precisely, a full analogue of Theorem 1.2.1 (with the same type of estimates) for the canonical decomposition holds.

# **1.3.** BURKHOLDER–DAVIS–GUNDY INEQUALITIES. STOCHASTIC INTEGRATION

Stochastic integration appears naturally while working with stochastic PDEs. In particular, Banach space-valued stochastic integration is of special interest and it has been widely developed during the past decades (see [18, 25, 27, 51, 76, 126, 129, 130, 132, 162]). The first sharp inequalities for Banach space-valued stochastic integrals have been obtained in the paper [126] by van Neerven, Veraar, and Weis. They showed that for any UMD Banach space *X*, for a Brownian motion *W*, for any elementary predictable process  $\Phi : \mathbb{R}_+ \times \Omega \to X$ , and for any 0

$$\mathbb{E}\sup_{t\geq 0}\left\|\int_{0}^{t}\Phi dW\right\|^{p} \approx_{p,X} \mathbb{E}\|\Phi\|_{\gamma(L^{2}(\mathbb{R}_{+}),X)}^{p}, \qquad (1.3.1)$$

where  $\|\cdot\|_{\gamma(L^2(\mathbb{R}_+),X)}$  is a  $\gamma$ -norm which e.g. coincides with the Hilbert-Schmidt norm if X is Hilbert (see Section 2.9). Later this inequality was extended to stochastic integrals with respect to a general continuous martingale by Veraar in [175], and

to stochastic integrals with respect to a cylindrical continuous martingale noise by Veraar and the author (see [177]).

Our goal is to find sharp bounds for vector-valued stochastic integrals with respect to *general martingales*. We will consider two cases depending on whether the right-hand side of the desired inequality is predictable or not, which both extend (1.3.1) since its right-hand side is already predictable.

### 1.3.1. General right-hand side

Stochastic integration is very closely related to *Burkholder–Davis–Gundy inequalities*. Those inequalities connect a martingale *M* with its quadratic variation [*M*] and classically due to Burkholder, Davis, and Gundy [40] have the following form: for any  $\mathbb{R}$ -valued martingale *M* and for any  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{t\geq 0}|M_t|^p \approx_p \mathbb{E}[M]_{\infty}^{p/2}.$$
(1.3.2)

This yields sharp bounds for real-valued stochastic integrals. Indeed, for any real-valued martingale *M*, for any elementary predictable  $\Phi : \mathbb{R}_+ \times \Omega \to \mathbb{R}$ , and for any  $1 \le p < \infty$  one has that

$$\mathbb{E}\sup_{t\geq 0}\left|\int_{0}^{t}\Phi\,\mathrm{d}M\right|^{p} = \mathbb{E}\left[\int_{0}^{\cdot}\Phi\,\mathrm{d}M\right]_{\infty}^{p/2} = \mathbb{E}\left(\int_{0}^{\infty}\Phi^{2}(t)\,\mathrm{d}[M]_{t}\right)^{p/2}$$

In Chapter 8 and 9 we extend (1.3.2) to Banach function spaces and to general Banach spaces. First in Chapter 8 we show that for any UMD Banach function space *X*, for any *X*-valued martingale *M*, and for any 1 one has that

$$\mathbb{E}\sup_{t\geq 0} \|M_t\|^p \approx_{p,X} \mathbb{E}\|[M]_{\infty}^{1/2}\|^p.$$
(1.3.3)

Further in Chapter 9 we present a more complicated, but much more general form of (1.3.3). More specifically, we show that for any UMD Banach space *X* and for any  $t \ge 0$  any *X*-valued martingale *M* has a *covariation bilinear form*  $[[M]]_t$  satisfying the following a.s.

$$[[M]]_t(x^*, x^*) = [\langle M, x^* \rangle]_t, \quad x^* \in X^*$$

Moreover, a.s. there exists an *X*-valued centered Gaussian random variable  $\xi_{[[M]]_t}$  having  $[[M]]_t$  as its covariance bilinear form:

$$[[M]]_{t}(x^{*}, x^{*}) = \mathbb{E}_{\xi} |\langle \xi_{[[M]]_{t}}, x^{*} \rangle|^{2}, \quad x^{*} \in X^{*},$$

and if one denotes  $(\mathbb{E} \|\xi_{[[M]]_t}\|^2)^{1/2}$  by  $\gamma([[M]]_t)$ , then the following holds true for any  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{0\leq s\leq t}\|M_s\|^p \approx_{p,X} \mathbb{E}\gamma([[M]]_t)^p.$$
(1.3.4)

(1.3.4) extends (1.3.3) to the case p = 1, and it is a natural extension of (1.3.2). Furthermore, both (1.3.3) and (1.3.4) characterize the UMD property.

The estimate (1.3.4) will allow us to extend (1.3.1) to full generality. Namely, we show that for any real-valued local martingale M, for any Banach space X and for any elementary predictable  $\Phi : \mathbb{R}_+ \times \Omega \to X$  we have that for any  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{t\geq 0}\left\|\int_{0}^{t}\Phi\,\mathrm{d}M\right\|^{p} \approx_{p,X} \mathbb{E}\|\Phi\|_{\gamma(L^{2}(\mathbb{R}_{+},[M]),X)}^{p}.$$
(1.3.5)

By assuming p = 1 and extending the definition of a stochastic integral to general predictable functions we show that general predictable  $\Phi : \mathbb{R}_+ \times \Omega \to X$  is stochastically integrable if it is locally in  $L^1(\Omega; \gamma(L^2(\mathbb{R}_+, [M]), X))$ , which is a natural generalization of the real-valued case [89, p. 526].

### 1.3.2. Predictable right-hand side

The sharp estimates (1.3.5) have one serious disadvantage: their right-hand side is not predictable in general. Since it is not predictable, one can not use a stopping time argument in order to bound it locally and therefore make it useful for solving SPDEs (where local boundedness of a stochastic integral plays a significant rôle for fixed point arguments) even with a Poisson noise. In Chapter 7 we find a predictable right-hand side in the case  $X = L^q(S)$  for any  $1 < q < \infty$ . These estimates for the Poisson case appeared first in the paper [51] by Dirksen. Even in this simple case the predictable right-hand side has six different possibilities depending on the order of p, q, and 2, and in each of this cases the right-hand side has a complicated structure. In Chapter 7 we extend this result to a general martingale noise with the same six cases involved. We will not present the main result of Chapter 7 – Theorem 7.5.30 – here, but just notice that it heavily exploits the following techniques

- Burkholder-Rosenthal inequalities (the discrete analogue of Burkholder-Davis-Gundy inequalities with the predictable right-hand side, see Subsection 1.4.3),
- the canonical decomposition,
- random measure theory (see Subsection 1.4.4),
- stochastic integration with respect to continuous martingales (see Subsection 1.5.1).

### **1.4. MISCELLANEA**

While proving the primary results of the thesis we needed some powerful tools, or we had some meaningful applications. We want to outline some of these topics here.

### 1.4.1. Fourier multipliers

The first motivation for considering weak differential subordination (at first it was considered only for discrete and purely discontinuous martingales) cones from *Fourier multipliers*, i.e. operators acting on  $L^2(\mathbb{R}^d)$  of the form

$$T_m f := \mathscr{F}^{-1}(m\mathscr{F}(f)), \quad f \in L^2(\mathbb{R}^d),$$

where  $m \in L^{\infty}(\mathbb{R}^d)$  is bounded by 1. Such operators appear naturally in Harmonic analysis (see e.g. [69, 79, 168, 169]). There is a natural question whether one can extend  $T_m$  to  $L^p(\mathbb{R}^d)$  for a general  $1 , or even to <math>L^p(\mathbb{R}^d; X)$  for a general Banach space X. In order to answer this question, theories as theory of Mihlin, Marcinkiewicz, even homogenous, and Lévy (also known as Bañuelos-Bogdan) multipliers have been created, and for many of them it has been shown that  $T_m$  is bounded not only on  $L^p(\mathbb{R}^d)$ , but even on  $L^p(\mathbb{R}^d; X)$  given X has UMD. In particular, in Chapter 3 we show that the so-called *Lévy multipliers* are bounded on  $L^p(\mathbb{R}^d; X)$ for any 1 and any UMD Banach space <math>X, and provide sharp upper bound for the norm of  $T_m$  in terms of the UMD constant. Recall that Bañuelos and Bogdan in [10] and Bañuelos, Bielaszewski, and Bogdan in [9] had shown that Lévy multipliers are bounded on  $L^p(\mathbb{R}^d)$  by using differential subordination. In Chapter 3 we extend their result to infinite dimensions using weak differential subordination.

### 1.4.2. Hilbert transform and orthogonal martingales

Let *X* be a Banach space,  $\mathbb{T} \simeq [-\pi, \pi)$  be a torus equipped with the Lebesque measure,  $f : \mathbb{T} \to X$  be a step function. We define *the periodic Hilbert transform* of *f* in the following way

$$\mathscr{H}_X^{\mathbb{T}} f(\theta) := \frac{1}{2\pi} \text{p.v.} \int_{-\pi}^{\pi} f(s) \cot \frac{\theta - s}{2} \mathrm{d}s, \qquad \theta \in [-\pi, \pi).$$

Recall that the periodic Hilbert transform is closely related to the UMD property since if we denote the  $L^p$ -norm of  $\mathscr{H}_X^{\mathbb{T}}$  by  $\hbar_{p,X}$ , then thanks to Bourgain [23] and Burkholder [32]  $\hbar_{p,X}$  is finite if and only if the UMD constant  $\beta_{p,X}$  is finite. Moreover, by Bourgain [23] and Garling [61] the following estimate holds

$$\sqrt{\beta_{p,X}} \le \hbar_{p,X} \le \beta_{p,X}^2$$

Due to a classical Doob's argument it is known that the periodic Hilbert transform has a representation in terms of stochastic integrals, which turn out to be weakly differentially subordinated orthogonal martingales. Remind that we call two *X*-valued martingales *M* and *N* orthogonal if  $[\langle M, x^* \rangle, \langle N, x^* \rangle] = 0$  and  $\langle M_0, x^* \rangle \cdot \langle N_0, x^* \rangle = 0$  for all  $x^* \in X^*$ .

Section 6 is devoted to showing the converse connection. Namely, we prove there that for any convex continuous functions  $\Phi, \Psi : X \to \mathbb{R}_+$  with  $\Psi(0) = 0$  and for any pair of *X*-valued orthogonal martingales *M* and *N* such that *N* is weakly differentially subordinate to *M* one has that

$$\mathbb{E}\Psi(N_t) \le C_{\Phi,\Psi,X} \mathbb{E}\Phi(M_t), \quad t \ge 0,$$

where  $C_{\Phi,\Psi,X}$  (finite or infinite) coincides with

$$|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} := \sup_{f:\mathbb{T}\to X \text{ step}} \frac{\int_{\mathbb{T}} \Psi(\mathscr{H}_X^{\mathbb{T}}f(s)) \,\mathrm{d}s}{\int_{\mathbb{T}} \Phi(f(s)) \,\mathrm{d}s}.$$
 (1.4.1)

This fact has a number of useful applications which we will shortly outline here and which can be found in Section 6.4.

- If  $\Phi$  is symmetric and  $\Phi(0) = 0$ , then  $\Phi, \Psi$ -norms of the periodic Hilbert transform, the discrete Hilbert transform, and the nonperiodic Hilbert transform (these norms are defined similarly to (1.4.1)) are the same.
- $\hbar_{p,X}$  dominates linearly the Wiener decoupling constants of the Banach space *X*.
- Finiteness of the  $\Phi, \Psi$ -norm  $|\mathscr{H}_X^T|_{\Phi,\Psi}$  of the periodic Hilbert transform together with some natural broad assumptions on  $\Phi$  and  $\Psi$  yields that *X* has the UMD property.
- Sharp  $L^p$ -bounds for weakly differentially subordinated martingales and  $L^p$ -bounds for weakly differentially subordinated harmonic functions. In particular, it is shown that sharp  $c_{p,X}$  in (1.1.3) satisfies

$$\max\{\beta_{p,X}, \hbar_{p,X}\} \le c_{p,X} \le \beta_{p,X} + \hbar_{p,X}.$$

### 1.4.3. Burkholder-Rosenthal inequalities

In [161] Rosenthal proved that for any sequence of independent mean-zero random variables  $(d_i)_{i>1}^n$  and of any  $p \ge 2$ 

$$\left(\mathbb{E}\Big|\sum_{i=1}^{n} d_{i}\Big|^{p}\right)^{\frac{1}{p}} \approx_{p} \max\left\{\left(\sum_{i=1}^{n} \mathbb{E}|d_{i}|^{p}\right)^{\frac{1}{p}}, \left(\mathbb{E}\Big(\sum_{i=1}^{n} \mathbb{E}_{i-1}|d_{i}|^{2}\Big)^{\frac{p}{2}}\right)^{\frac{1}{p}}\right\}.$$
(1.4.2)

Later in [29] Burkholder extended (1.4.2) to a general martingale difference sequence. Note that the right-hand side of (1.4.2) is predictable. Therefore it is natural to ask: let *X* be a Banach space and let  $1 . Is there a norm <math>||| \cdot ||_{p,X}$  on all *X*-valued martingale difference sequences depending only on *predictable moments* of the individual differences such that for any *X*-valued martingale difference sequence ( $d_i$ )<sub> $i \ge 1$ </sub>

$$c_{p,X} \| (d_i) \|_{p,X} \le \left( \mathbb{E} \left\| \sum_i d_i \right\|_X^p \right)^{\frac{1}{p}} \le C_{p,X} \| (d_i) \|_{p,X},$$
(1.4.3)

In Chapter 7 we present the explicit formula of  $||| \cdot |||_{p,X}$  for the case  $X = L^q(S)$ ,  $1 < q < \infty$ . We also show that Burkholder-Rosenthal inequalities lead to sharp estimates for integrals with respect to random measures and sharp predictable estimates for stochastic integrals with respect to general martingales, which in particular are presented in Theorem 7.5.30 in the  $L^q$ -valued case. Thus Burkholder-Rosenthal inequalities for more general Banach spaces are of exceptional interest since they might yield sharp estimates for corresponding stochastic integrals.

### 1.4.4. Random measures

Random measure theory appeared in 1970's in works of Grigelionis and Novikov as a natural extension of Poisson random measures. A random measure  $\mu$  is defined as a measure  $\mu(\omega)$  on  $\mathbb{R}_+ \times J$  for some measurable space  $(J, \mathscr{J})$  (which is called the *jump space*) that depends on  $\omega \in \Omega$  in an optional way. Any random measure  $\mu$  has a *compensator* random measure  $\nu$  which is predictable such that integral of an elementary predictable function with respect to  $\bar{\mu} := \mu - \nu$  is a local martingale. Thanks to Novikov [131] the following inequality holds for any  $p \ge 2$  and for any predictable  $f : \mathbb{R}_+ \times \Omega \times J \to \mathbb{R}$ 

$$\mathbb{E}\left|\int_{0}^{t} f \,\mathrm{d}\bar{\mu}\right|^{p} = p\left(\mathbb{E}\int_{0}^{t} |f|^{2} \,\mathrm{d}\nu\right)^{p/2} + \mathbb{E}\int_{0}^{t} |f|^{p} \,\mathrm{d}\nu, \quad t \geq 0.$$

Note that the process on the right-hand side of the latter inequality is predictable in  $t \ge 0$  since both f and v are predictable. In Subsection 7.5.4 we extend Novikov's inequality to  $L^q$ -valued integrals with respect to a random measure. Moreover, we prove that for any Banach space X, for any 1 , and for any elementary $predictable <math>f : \mathbb{R}_+ \times \Omega \times J \to X$ 

$$\mathbb{E}\left\|\int_{0}^{t}f\,\mathrm{d}\bar{\mu}\right\|^{p} = E\int_{0}^{t}\|f\|^{p}\,\mathrm{d}\nu, \quad t \geq 0,$$

if  $v(\mathbb{R}_+ \times J) \leq 1$  a.s.

### 1.4.5. Bellman functions

For a Banach space *X* and a function  $V : X \times X \to \mathbb{R}$  a function  $U : X \times X \to \mathbb{R}$  is called *Bellman* if

- U has nice properties,
- $U(x, y) \le V(x, y)$  for all  $x, y \in X$ , and
- *U*(*x*, *y*) ≥ 0 if *x*, *y* ∈ *X* are from a certain *good* subset *A* of *X* × *X* (e.g. *A* = {(0,0)} or *A* = {(*x*, 0), *x* ∈ *X*}).

Bellman functions are widely used in stochastic analysis (see numerous papers by Bañuelos, Burkholder, Nazarov, Osękowski, Volberg, etc.) and usually their application has the following form: in order to show that for a pair of *X*-valued martingales *M* and *N* under some natural assumptions  $\mathbb{E}V(M_t, N_t) \ge 0$  one proves the following

$$EV(M_t, N_t) \stackrel{(i)}{\geq} EU(M_t, N_t) \stackrel{(ii)}{\geq} EU(M_0, N_0) \stackrel{(iii)}{\geq} 0, \qquad (1.4.4)$$

where in (*i*) one uses the fact that  $V \ge U$  on  $X \times X$ , (*ii*) follows from Itô's formula and nice properties of *U*, and (*iii*) holds by the fact that  $(M_0, N_0) \in A$  a.s. Often in the literature *X* is taken to be  $\mathbb{R}^d$  for some  $d \ge 1$ , so in the overwhelming majority of all the papers concerning Bellman function approach to martingale inequalities the corresponding Bellman function has a precise expression. The only exceptions when the Bellman function is given in an abstract nonconstructive way known to the author can be found in [13, 31, 35]. Here in Chapter 3, 4, and 6, as well as in papers [183, 188] we apply and even invent Bellman functions for general UMD Banach spaces *X* with an abstract construction. It turned out that in order to work with a Bellman function one does not need to know what the function looks like, but just the necessary properties, which often could be figured out if one needs (*ii*) from (1.4.4) to hold.

### **1.5.** What is not in the thesis

Unfortunately, due to the lack of space not all results obtained during the PhD period are presented in the thesis. Let us sketch the content of the papers which are treated here.

### 1.5.1. Cylindrical continuous martingales and stochastic integration, paper [177]

In the paper [177] Veraar and the author have studied cylindrical continuous martingales and stochastic integration with respect to a cylindrical continuous martingale. Namely, a wider version of (1.3.1) was proved: let *X* be UMD, *M* be a cylindrical *continuous* martingale on a Hilbert space H,  $\Phi : \mathbb{R}_+ \times \Omega \rightarrow \mathcal{L}(H, X)$  be elementary predictable. Then

$$\mathbb{E}\sup_{t\geq 0} \left\| \int_0^t \Phi \, \mathrm{d}M \right\|^p \approx_{p,X} \mathbb{E} \| \Phi Q_M^{1/2} \|_{\gamma(L^2(\mathbb{R}_+, [[M]]), X)}^p, \quad 0$$

where  $[[M]] : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  is the quadratic variation of M and  $Q_M : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H)$  is a quadratic variation derivative (for the precise definitions of a cylindrical continuous martingale, [[M]], and  $Q_M$  please have a look at [177]).

Even though the inequality (1.5.1) follows directly from (1.3.4), at that time (1.5.1) was new and important e.g. for obtaining Theorem 7.5.30, the main result of Chapter 7. Also notice that this work was for the author an introduction to stochastic analysis in Banach spaces; in particular, it led to deeper understanding of the vector-valued stochastic integration phenomenon.

# **1.5.2.** Brownian representations of cylindrical continuous local martingales, paper [186]

The paper [186] is a spin-off of the paper [177]. Many questions concerning cylindrical continuous martingales remained open after [177]; in particular, does any cylindrical continuous martingale have a *Brownian representation*, i.e. can any cylindrical continuous martingale be represented as a stochastic integral with respect to a cylindrical Brownian motion after a certain time-change? The paper [186] contains the answer to as well as counterexamples concerning this question.

#### 1.5.3. Even Fourier multipliers and martingale transforms, paper [188]

It turns out (see Remark 4.4.6) that the sharp  $L^p$ -estimate for the canonical decomposition has the following form. For any UMD Banach space X, for any  $1 , and for any martingale <math>M : \mathbb{R}_+ \times \Omega \to X$  one has that for  $i \in \{c, q, a\}$ 

$$\mathbb{E} \|M_t^i\|^p \le c_{p,X}^p \mathbb{E} \|M_t\|^p, \ t \ge 0,$$

where  $M = M^c + M^q + M^a$  is the canonical decomposition and the sharp constant  $c_{p,X}$  equals the  $UMD^{\{0,1\}}$ -constant  $\beta_{p,X}^{\{0,1\}}$  of X, i.e. the least constant  $\beta \ge 0$  such that for any n > 0, for any X-valued martingale difference sequence  $(d_i)_{i=1}^n$ , and for any  $\{0,1\}$ -valued sequence  $(\varepsilon_i)_{i=1}^n$  one has that

$$\mathbb{E}\left\|\sum_{i=1}^{n}\varepsilon_{i}d_{i}\right\|^{p} \leq \beta^{p}\mathbb{E}\left\|\sum_{i=1}^{n}d_{i}\right\|^{p}.$$

Such type of martingale transforms and the corresponding sharp constants were discovered only in the real-valued case by Choi [42] and by Bañuelos and Osekowski [13]. In the paper [188] we consider the vector-valued case and extend many statements from [13] to Banach spaces including sharp bounds for even Fourier multipliers. In particular, it is shown that  $\beta_{p,X}^{[0,1]}$  equals the norm of the second order Riesz transform.

# 2

### PRELIMINARIES

Before presenting the results that will be used throughout this thesis, we introduce some basic notation. We denote the set of natural numbers by  $\mathbb{N} = \{1, 2, 3, ...\}$ . We denote the half-line  $\mathbb{R}_+ = [0, +\infty)$  and  $\overline{\mathbb{R}}_+ = [0, +\infty]$ . Throughout this thesis we assume the scalar field  $\mathbb{K}$  to be  $\mathbb{R}$  or  $\mathbb{C}$  unless otherwise is stated. We will use the *Kronecker symbol*  $\delta_{ij}$ , which is defined in the following way:  $\delta_{ij} = 1$  if i = j, and  $\delta_{ij} = 0$  if  $i \neq j$ . For any numbers  $a, b \in \mathbb{R}$  we will often denote min $\{a, b\}$  by  $a \wedge b$  and max $\{a, b\}$  by  $a \lor b$ .

For each  $p \in (1,\infty)$  we set  $p' \in (1,\infty)$  and  $p^* \in [2,\infty)$  to be such that  $\frac{1}{p} + \frac{1}{p'} = 1$  and  $p^* = \max\{p, p'\}$ .

We write  $a \leq_A b$  if there exists a constant *c* depending only on *A* such that  $a \leq cb$ .  $\geq_A$  is defined analogously. We write  $a \approx_A b$  if  $a \leq_A b$  and  $a \geq_A b$  simultaneously.

The letters *X* and *Y* are used to denote Banach spaces, and we write *X*<sup>\*</sup> for the dual of *X*. We denote by  $\mathscr{L}(X, Y)$  the space of all bounded linear operators, with norm  $\|\cdot\|_{\mathscr{L}(X,Y)}$ .

Let  $(S, \mu, \Sigma)$  be a measure space. A function  $f : S \to X$  is called *strongly measurable* if it is the a.e. limit of a sequence of simple functions. For any  $1 \le p \le \infty$  we denote by  $L^p(S; X)$  the Banach space of all strongly measurable functions  $f : S \to X$  such that

$$\|f\|_{L^p(S;X)} := \left| \int_S \|f\|^p \,\mathrm{d}\mu \right|^{1/p} < \infty, \text{ if } p < \infty,$$
$$\|f\|_{L^\infty(S;X)} := \text{ess. sup}_{s \in S} \|f(s)\| < \infty, \text{ if } p = \infty$$

Note that if  $X^*$  has the *Radon-Nikodym property* (e.g. X is reflexive, see [79, Section 1.3]), then for all  $1 \le p < \infty$ ,  $(L^p(S; X))^* = L^{p'}(S; X^*)$ . Let  $\mathscr{A}$  be a sub- $\sigma$ -algebra of  $\Sigma$ . Then for any  $f \in L^p(S; X)$  there exists a *conditional expectation with respect to*  $\mathscr{A}$ , which we will denote by  $\mathbb{E}(f|\mathscr{A})$ , such that  $\mathbb{E}(f|\mathscr{A})$  is  $\mathscr{A}$ -measurable, and

$$\langle \mathbb{E}(f|\mathscr{A}), x^* \rangle = \mathbb{E}(\langle f, x^* \rangle | \mathscr{A}), x^* \in X^*.$$

The reader can find more information in [79, Section 2.6].

### **2.1.** BASIC NOTIONS ON STOCHASTIC PROCESSES

Let  $I \subset \mathbb{R}$  be a closed interval (perhaps, infinite), *X* be a Banach space. A function *F* :  $I \rightarrow X$  is called *càdlàg* (from a French acronym "continue à droite, limite à gauche")

if *F* is right-continuous and has left limits. Definitions of a *càglàd*, *càd*, *càg*, *làd*, and *làg* function are analogous.

Let  $(\Omega, \mathscr{F}, \mathbb{P})$  be a probability space with a filtration  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$  which satisfies the *usual conditions*, i.e.  $\mathbb{F}$  is right-continuous and  $\mathscr{F}_0$  contains all sets of  $\mathbb{P}$ -measure zero (see e.g. [93, Definition 1.2.25] and [155]). A process  $F : \mathbb{R}_+ \times \Omega \to X$  is called *adapted* if  $F_t$  is  $\mathscr{F}_t$ -measurable for any  $t \ge 0$ . We denote by  $\mathscr{P}$  the *predictable*  $\sigma$ *algebra* on  $\mathbb{R}_+ \times \Omega$ , the  $\sigma$ -algebra generated by all càg adapted processes. We use  $\mathscr{O}$  to denote the *optional*  $\sigma$ -*algebra*  $\mathbb{R}_+ \times \Omega$ , the  $\sigma$ -algebra generated by all càdlàg adapted processes.

### **2.2.** MARTINGALES

Let *X* be a Banach space. A process  $M : \mathbb{R}_+ \times \Omega \to X$  is called a *martingale* if *M* is adapted,  $M_t \in L^1(\Omega; X)$  for all  $t \ge 0$ , and  $\mathbb{E}(M_t | \mathscr{F}_s) = M_s$  for all  $t \ge s \ge 0$ . *M* is called a *local martingale* if there exists a sequence  $(\tau_n)_{n\ge 1}$  of stopping times (see Section 2.4 for the definition) such that  $\tau_n \to \infty$  a.s. as  $n \to \infty$  and  $(M_t^{\tau_n})_{t\ge 0} := (M_{t \land \tau_n})_{t\ge 0}$  is a martingale for all  $n \ge 1$ .

Since  $\mathbb{F} = (\mathscr{F}_t)_{t \ge 0}$  satisfies the usual conditions,  $\mathbb{F}$  is right-continuous and the following proposition holds:

**Proposition 2.2.1.** Let X be a Banach space. Then any martingale  $M : \mathbb{R}_+ \times \Omega \to X$  admits a càdlàg version.

For proving the proposition we will need the following lemma. Recall that for a Banach space *X* and for a closed (perhaps, infinite) interval  $I \subset \mathbb{R}$  we define a *Skorohod space*  $\mathcal{D}(I; X)$  as a linear space consisting of all càdlàg functions  $f : I \to X$ . We denote the linear space of all bounded càdlàg functions  $f : I \to X$  by  $\mathcal{D}_b(I; X)$ .

**Lemma 2.2.2.**  $\mathcal{D}_b(I; X)$  equipped with the norm  $\|\cdot\|_{\infty}$  is a Banach space.

*Proof.* The proof is analogous to the proof of the same statement for continuous functions (see [154, Problem V.6.1] and [167]).  $\Box$ 

*Proof of Proposition 2.2.1.* One can find the proof in [174, Proposition 2.2.2], but we will repeat it here for the convenience of the reader. Without loss of generality suppose that  $M_{\infty} := \lim_{t \to \infty} M_t$  exists a.s. and is in  $L^1(\Omega; X)$ . Also we can assume that there exists t > 0 such that  $M_t = M_{\infty}$ . Let  $(\xi^n)_{n\geq 1}$  be a sequence of simple functions in  $L^1(\Omega; X)$  such that  $\xi^n \to M_t$  in  $L^1(\Omega; X)$  as  $n \to \infty$ . For each  $n \ge 1$  define a martingale  $M^n : \mathbb{R}_+ \times \Omega \to X$  such that  $M_s^n = \mathbb{E}(\xi^n | \mathscr{F}_s)$  for each  $s \ge 0$ . Fix  $n \ge 1$ . Since  $\xi^n$  takes its values in a finite dimensional subspace of X,  $M^n$  takes its values in the same finite dimensional subspace as well, and therefore by [49] (or [155, p.8]) it has a càdlàg version. But  $M_t^n = \xi^n \to M_t$  in  $L^1(\Omega; X)$  as  $n \to \infty$ , so by the (2.2.1),  $M^n \to M$  in the ucp topology (the topology of the uniform convergence on compacts in probability). By taking an appropriate subsequence we can assume

that  $M^n \to M$  a.s. uniformly on [0, *t*], and consequently, uniformly on  $\mathbb{R}_+$ . Therefore, by Lemma 2.2.2 *M* has a càdlàg version.

Thanks to Proposition 2.2.1 we can define  $\Delta M_t$  and  $M_{t-}$  for each  $t \ge 0$ ,

$$\begin{split} \Delta M_t &:= M_t - \lim_{\varepsilon \to 0} M_{(t-\varepsilon) \vee 0}, \\ M_{t-} &:= \lim_{\varepsilon \to 0} M_{t-\varepsilon}, \ M_{0-} &:= M_0 \end{split}$$

Let  $1 \le p \le \infty$ . A martingale  $M : \mathbb{R}_+ \times \Omega \to X$  is called an  $L^p$ -bounded martingale if  $M_t \in L^p(\Omega; X)$  for each  $t \ge 0$  and there exists a limit  $M_\infty := \lim_{t\to\infty} M_t \in L^p(\Omega; X)$  in  $L^p(\Omega; X)$ -sense. We will denote the space of all *X*-valued  $L^p$ -bounded martingales on  $\mathbb{F}$  by  $\mathcal{M}^p_X(\mathbb{F})$ . For brevity we will use  $\mathcal{M}^p_X$  instead. Notice that  $\mathcal{M}^p_X$  is a Banach space with the given norm:  $\|M\|_{\mathcal{M}^p_X} := \|M_\infty\|_{L^p(\Omega; X)}$  (see [84, 89] and [79, Chapter 1]). We also denote all the *X*-valued locally  $L^p$ -bounded martingales by  $\mathcal{M}^{p, \text{loc}}_X$ .

**Proposition 2.2.3.** Let X be a Banach space with X\* having the Radon-Nikodým property (e.g. reflexive),  $1 . Then <math>(\mathcal{M}_X^p)^* = \mathcal{M}_{X^*}^{p'}$ , and  $\|M\|_{(\mathcal{M}_X^p)^*} = \|M\|_{\mathcal{M}_{X^*}^{p'}}$  for each  $M \in \mathcal{M}_{X^*}^{p'}$ .

*Proof.* Since  $||M||_{\mathcal{M}_X^p} = ||M_{\infty}||_{L^p(\Omega;X)}$  for each  $M \in \mathcal{M}_X^p$ , and since for each  $\xi \in L^p(\Omega;X)$ we can construct a martingale  $M = (M_t)_{t\geq 0} = (\mathbb{E}(\xi|\mathcal{F}_t))_{t\geq 0}$  satisfying  $||M||_{\mathcal{M}_X^p} = ||\xi||_{L^p(\Omega;X)}$ ,  $\mathcal{M}_X^p$  is isometric to  $L^p(\Omega;X)$ , and therefore the proposition follows from [79, Proposition 1.3.3].

Since  $\|\cdot\|: X \to \mathbb{R}_+$  is a convex function, and *M* is a martingale,  $\|M\|$  is a submartingale by Jensen's inequality (see [89, Lemma 7.11]), and hence by Doob's inequality (see e.g. [93, Theorem 1.3.8(i)]) we have that for all 1

$$\mathbb{E} \sup_{0 \le s \le t} \|M_s\|^p \sim_p \mathbb{E} \|M_t\|^p, \ t \ge 0.$$
(2.2.1)

Moreover, by [93, Theorem 1.3.8(i)] we have that for each  $t \ge 0$ ,  $p \ge 1$  and  $\lambda > 0$ 

$$\mathbb{P}(M_t^* > \lambda) \le \frac{\mathbb{E} \|M_t\|^p}{\lambda^p},$$
(2.2.2)

where  $M_t^* := \sup_{0 \le s \le t} \|M_s\|$  for all  $t \ge 0$ .

In the sequel we will need a definition of a Paley-Walsh martingale.

**Definition 2.2.4** (Rademacher random variable). Let  $\xi : \Omega \to \mathbb{R}$  be a random variable. Then  $\xi$  has the *Rademacher distribution* (or simply  $\xi$  is *Rademacher*) if  $\mathbb{P}(\xi = 1) = \mathbb{P}(\xi = -1) = \frac{1}{2}$ .

**Definition 2.2.5** (Paley-Walsh martingale). Let *X* be a Banach space. A discrete *X*-valued martingale  $(f_n)_{n\geq 0}$  is called a *Paley-Walsh martingale* if there exist a sequence
of independent Rademacher variables  $(r_n)_{n\geq 1}$ , a function  $\phi_n : \{-1,1\}^{n-1} \to X$  for each  $n \geq 2$ , and  $\phi_1 \in X$  such that  $f_n - f_{n-1} = r_n \phi_n(r_1, \dots, r_{n-1})$  for each  $n \geq 2$ ,  $f_1 = r_1 \phi_1$ , and  $f_0 = 0$ .

For a discrete *X*-valued martingale  $(f_n)_{n\geq 0}$  we define  $df_n := f_n - f_{n-1}$  for  $n \geq 1$ and  $df_0 := f_0$ .

#### 2.2.1. Quadratic variation

Let  $(\Omega, \mathscr{F}, \mathbb{P})$  be a probability space with a filtration  $\mathbb{F} = (\mathscr{F}_t)_{t \ge 0}$  that satisfies the usual conditions. Let  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a local martingale. We define a *quadratic variation* of *M* in the following way:

$$[M]_t := \mathbb{P} - \lim_{\text{mesh} \to 0} \sum_{n=1}^N |M(t_n) - M(t_{n-1})|^2, \qquad (2.2.3)$$

where the limit in probability is taken over all partitions  $0 = t_0 < ... < t_N = t$ . Note that [*M*] exists and is nondecreasing a.s. For any martingales  $M, N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  we can define a *covariation*  $[M, N] : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  as  $[M, N] := \frac{1}{4}([M+N] - [M-N])$ . Since *M* and *N* have càdlàg versions, [M, N] has a càdlàg version as well (see [85, Theorem I.4.47] and [120]).

*Remark* 2.2.6 ([120]). The process  $\langle M, N \rangle - [M, N]$  is a local martingale.

Let *H* be a Hilbert space,  $M : \mathbb{R}_+ \times \Omega \to H$  be a local martingale. We define a *quadratic variation* of *M* in the following way:

$$[M]_t := \mathbb{P} - \lim_{\text{mesh} \to 0} \sum_{n=1}^N \|M(t_n) - M(t_{n-1})\|^2,$$
(2.2.4)

where the limit in probability is taken over partitions  $0 = t_0 < ... < t_N = t$ . Note that [*M*] exists and is nondecreasing a.s. and that for any orthogonal basis  $(h_n)_{n\geq 1}$  of *H*, for any  $t \ge 0$  a.s.

$$[M]_{t} = \sum_{n \ge 1} [\langle M, h_{n} \rangle]_{t}.$$
 (2.2.5)

The reader can find more on quadratic variations in [120, 121, 177] for the vector-valued setting, and in [49, 89, 121, 155] for the real-valued setting.

As it was shown in [123, Proposition 1] (see also [163, Theorem 2.13] and [177, Example 3.19] for the continuous case), for any *H*-valued martingale *M* there exists an adapted process  $q_M : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H)$  which we will call a *quadratic variation derivative*, such that the trace of  $q_M$  does not exceed 1 on  $\mathbb{R}_+ \times \Omega$ ,  $q_M$  is self-adjoint nonnegative on  $\mathbb{R}_+ \times \Omega$ , and for any  $h, g \in H$  a.s.

$$[\langle M, h \rangle, \langle M, g \rangle]_t = \int_0^t \langle q_M^{1/2}(s)h, q_M^{1/2}(s)g \rangle d[M]_s, \ t \ge 0.$$
(2.2.6)

For any martingales  $M, N : \mathbb{R}_+ \times \Omega \to H$  we can define a *covariation*  $[M, N] : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  as  $[M, N] := \frac{1}{4}([M+N] - [M-N])$ . Since M and N have càdlàg versions, [M, N] has a càdlàg version as well (see [85, Theorem I.4.47] and [120]). Moreover,  $\langle M, N \rangle - [M, N]$  is a local martingale.

We will frequently use the *Burkholder–Davis–Gundy inequality*: for any  $1 \le p < \infty$ , for any local martingale  $M : \mathbb{R}_+ \times \Omega \to H$  with  $M_0 = 0$ , and for any stopping time  $\tau$  one has that

$$\left(\mathbb{E}\sup_{0\le t\le \tau} \|M_t\|^p\right)^{1/p} =_p (\mathbb{E}[M]_{\tau}^{p/2})^{1/p}.$$
(2.2.7)

We refer to [115] for a self-contained proof.

# 2.2.2. Continuous martingales

Let *X* be a Banach space. A martingale  $M : \mathbb{R}_+ \times \Omega \to X$  is called *continuous* if *M* has continuous paths.

*Remark* 2.2.7 ([89, 121]). If *X* is a Hilbert space,  $M, N : \mathbb{R}_+ \times \Omega \to X$  are continuous martingales, then [M, N] has a continuous version.

Let  $1 \le p \le \infty$ . We will denote the linear space of all continuous *X*-valued  $L^{p}$ bounded martingales on  $\mathbb{F}$  which start at zero by  $\mathcal{M}_X^{p,c}(\mathbb{F})$ . For brevity we will write  $\mathcal{M}_X^{p,c}$  instead of  $\mathcal{M}_X^{p,c}(\mathbb{F})$  since  $\mathbb{F}$  is fixed. Analogously to [89, Lemma 17.4] by applying (2.2.1) one can show the following proposition.

**Proposition 2.2.8.** Let X be a Banach space,  $p \in (1,\infty)$ . Then  $\mathcal{M}_X^{p,c}$  equipped with the norm  $||M||_{\mathcal{M}_X^{p,c}} := ||M_{\infty}||_{L^p(\Omega;X)}$  is a Banach space.

# 2.2.3. Purely discontinuous martingales. Meyer-Yoeurp decomposition

An increasing càdlàg process  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  is called *pure jump* if a.s. for each  $t \ge 0$ ,  $A_t = A_0 + \sum_{s=0}^t \Delta A_s$ . A local martingale  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  is called *purely discontinuous* if [*M*] is a pure jump process. We leave the following evident lemma without proof.

**Lemma 2.2.9.** Let  $A: \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be an increasing adapted càdlàg process such that  $A_0 = 0$ . Then there exist unique up to indistinguishability increasing adapted càdlàg processes  $A^c, A^d: \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  such that  $A^c$  is continuous a.s.,  $A^d$  is pure jump a.s.,  $A_0^c = A_0^d = 0$  and  $A = A^c + A^d$ .

The following decomposition theorem is known due to Meyer and Yoeurp (see [122, 190] and [89, Theorem 26.14]).

**Theorem 2.2.10** (Meyer-Yoeurp decomposition). Let  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a local martingale. Then there exist a unique continuous local martingale  $M^c$  and a unique purely discontinuous local martingale  $M^d$  such that  $M_0^c = 0$  and  $M = M^c + M^d$ . Moreover, in this case  $[M]^c = [M^c]$  and  $[M]^d = [M^d]$ , where  $[M]^c$  and  $[M]^d$  are defined as in Lemma 2.2.9.

**Corollary 2.2.11.** Let  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a martingale which is both continuous and purely discontinuous. Then  $M = M_0$  a.s.

*Proof.* Let  $M = M^c + M^d$  be the Meyer-Yoeurp decomposition. Since M is continuous, then  $M^d = M_0$ , and since M is purely discontinuous, then  $M^c = 0$ , so the desired holds true.

Later we will need the following proposition.

**Proposition 2.2.12.** A martingale  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  is purely discontinuous if and only if *MN* is a martingale for any continuous bounded martingale  $N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  with  $N_0 = 0$ .

Note that some authors take this equivalent condition as the definition of a purely discontinuous martingale, see e.g. [85, Definition I.4.11] and [84, Chapter I].

*Proof of Proposition* 2.2.12. One direction follows from [89, Corollary 26.15]. Indeed, if *M* is purely discontinuous, then a.s. [M, N] = 0. Therefore by Remark 2.2.6, *MN* is a local martingale, and due to integrability it is a martingale.

For the other direction we apply Theorem 2.2.10. Let  $N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a continuous martingale such that  $N_0 = 0$  and M - N is purely discontinuous. Then there exists an increasing sequence of stopping times  $(\tau_n)_{n\geq 1}$  such that  $\tau_n \nearrow \infty$  as  $n \to \infty$  and  $N^{\tau_n}$  is a bounded continuous martingale for each  $n \ge 1$ . Therefore  $MN^{\tau_n}$  and  $(M-N)N^{\tau_n}$  are martingales for any  $n \ge 1$ , and hence  $(N^{\tau_n})^2 = (MN^{\tau_n} - (M-N)N^{\tau_n})^{\tau_n}$  is a martingale that starts at zero. On the other hand it is a nonnegative martingale, so it is the zero martingale. By letting *n* to infinity we prove that N = 0 a.s., so *M* is purely discontinuous.

Let us now move to the vector-valued case.

**Definition 2.2.13.** Let *X* be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a local martingale. Then *M* is called *purely discontinuous* if for each  $x^* \in X^*$  the local martingale  $\langle M, x^* \rangle$  is purely discontinuous.

*Remark* 2.2.14. Let *X* be finite dimensional. Then similarly to Theorem 2.2.10 any martingale  $M : \mathbb{R}_+ \times \Omega \to X$  can be uniquely decomposed into a sum of a purely discontinuous local martingale  $M^d$  and a continuous local martingale  $M^c$  such that  $M_0^c = 0$ .

*Remark* 2.2.15. Analogously to Proposition 2.2.12, a martingale  $M : \mathbb{R}_+ \times \Omega \to X$  is purely discontinuous if and only if  $\langle M, x^* \rangle N$  is a martingale for any  $x^* \in X^*$  and any continuous bounded martingale  $N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  such that  $N_0 = 0$ .

Let  $p \in [1,\infty]$ . We will denote the linear space of all purely discontinuous *X*-valued  $L^p$ -bounded martingales on  $\mathbb{F}$  by  $\mathcal{M}_X^{p,d}(\mathbb{F})$ . Since  $\mathbb{F}$  is fixed, we will use  $\mathcal{M}_X^{p,d}$  instead. The scalar case of the next result have been presented in [84, Lemme I.2.12].

**Proposition 2.2.16.** Let X be a Banach space,  $p \in (1,\infty)$ . Then  $\mathcal{M}_X^{p,d}$  equipped with the norm  $||M||_{\mathcal{M}^{p,d}} := ||M_{\infty}||_{L^p(\Omega;X)}$  is a Banach space.

*Proof.* Let  $(M^n)_{n\geq 1}$  be a sequence of purely discontinuous *X*-valued  $L^p$ -bounded martingales such that  $(M^n_{\infty})_{n\geq 1}$  is a Cauchy sequence in  $L^p(\Omega; X)$ . Let  $\xi \in L^p(\Omega; X)$  be such that  $\lim_{n\to\infty} M^n_{\infty} = \xi$ . Define a martingale  $M : \mathbb{R}_+ \times \Omega \to X$  as follows:  $M = (M_s)_{s\geq 0} = (\mathbb{E}(\xi|\mathscr{F}_s))_{s\geq 0}$ . Let us show that  $M \in \mathscr{M}^{p,d}_X$ . First notice that  $||M_{\infty}||_{L^p(\Omega;X)} = ||\xi||_{L^p(\Omega;X)} < \infty$ . Further for each  $x^* \in X^*$  by [84, Lemme I.2.12] we have that  $\langle M, x^* \rangle$  as a limit of real-valued purely discontinuous martingales  $(\langle M^n, x^* \rangle)_{n\geq 1}$  in  $\mathscr{M}^p_{\mathbb{R}}$  is purely discontinuous. Therefore *M* is purely discontinuous by the definition. □

In the sequel we will use the following lemma.

**Lemma 2.2.17.** Let X be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a martingale such that M is both continuous and purely discontinuous. Then  $M = M_0$  a.s.

Proof. Follows analogously to Corollary 2.2.11.

**Definition 2.2.18.** A local martingale  $M : \mathbb{R}_+ \times \Omega \to X$  is called to have the *Meyer-Yoeurp decomposition* if there exist local martingales  $M^c, M^d : \mathbb{R}_+ \times \Omega \to X$  such that  $M^c$  is continuous,  $M^d$  is purely discontinuous,  $M_0^c = 0$ , and  $M = M^c + M^d$ .

*Remark* 2.2.19. Note that if  $M = M^c + M^d$  is the Meyer-Yoeurp decomposition, then  $\langle M^c, x^* \rangle$  is continuous and  $\langle M^d, x^* \rangle$  is purely discontinuous for any  $x^* \in X^*$ ; therefore this decomposition is unique by the uniqueness of the Meyer-Yoeurp decomposition of a real-valued local martingale (see [89, Theorem 26.14] for details).

The reader can find more on purely discontinuous martingales in [84, 85, 89].

# **2.3. UMD BANACH SPACES**

Suppose that  $(\Omega, \mathscr{F}, \mathbb{P})$  is a nonatomic probability space. A Banach space *X* is called a *UMD Banach space* if for some (or equivalently, for all)  $p \in (1, \infty)$  there exists a finite constant  $\beta$  such that the following holds. If  $(d_n)_{n=1}^{\infty}$  is any *X*-valued martingale difference sequence (relative to some discrete-time filtration) contained in  $L^p(\Omega; X)$ and  $(\varepsilon_n)_{n=1}^{\infty}$  is any deterministic  $\{-1,1\}$ -valued sequence, then

$$\left(\mathbb{E}\left\|\sum_{n=1}^{N}\varepsilon_{n}d_{n}\right\|^{p}\right)^{\frac{1}{p}} \leq \beta\left(\mathbb{E}\left\|\sum_{n=1}^{N}d_{n}\right\|^{p}\right)^{\frac{1}{p}}.$$

The least admissible constant  $\beta$  above is denoted by  $\beta_{p,X}$  and is called the  $UMD_p$  *constant* of *X*, or, if the value of *p* is understood, the *UMD constant* of *X*. It is well-known that UMD spaces enjoy a large number of useful properties, such as being reflexive. Examples of UMD spaces include all finite dimensional spaces, Hilbert spaces (then  $\beta_{p,X} = p^* - 1$  with  $p^* = \max\{p, p/(p-1)\}$ ), the reflexive range of  $L^q$ -spaces, Sobolev spaces, Schatten class spaces, Orlicz, and Musielak–Orlicz spaces.

On the other hand, all nonreflexive Banach spaces, e.g.  $L^1(0, 1)$  and C([0, 1]), are not UMD. We refer the reader to [39, 79, 80, 153, 164] for further details.

The following proposition is a vector-valued version of [42, Theorem 4.1].

**Proposition 2.3.1.** Let X be a Banach space,  $p \in (1,\infty)$ . Then X has the UMD property *if and only if there exists* C > 0 *such that for each*  $n \ge 1$ , *for every martingale difference sequence*  $(d_j)_{j=1}^n$  *in*  $L^p(\Omega; X)$ , *and every sequence*  $(\varepsilon_j)_{j=1}^n$  *such that*  $\varepsilon_j \in \{0, 1\}$  *for each* j = 1, ..., n we have

$$\left(\mathbb{E}\left\|\sum_{j=1}^{n}\varepsilon_{j}d_{j}\right\|^{p}\right)^{\frac{1}{p}} \leq C\left(\mathbb{E}\left\|\sum_{j=1}^{n}d_{j}\right\|^{p}\right)^{\frac{1}{p}}.$$

If this is the case, then the least admissible C is in the interval  $\left[\frac{\beta_{p,X}-1}{2},\beta_{p,X}\right]$ 

*Remark* 2.3.2. UMD Banach spaces form a natural environment for the  $L^p$ -boundedness of the periodic Hilbert transform (see Subsection 6.2.1). It follows from [23, 32] that for every 1 we have

$$\sqrt{\beta_{p,X}} \le \|\mathscr{H}_X^{\mathbb{T}}\|_{L^p(\mathbb{T},X) \to L^p(\mathbb{T},X)} \le \beta_{p,X}^2.$$
(2.3.1)

It is not known whether the quadratic dependence can be improved on either of the sides (see e.g. [39, 66, 79]). Notice that if  $X = \mathbb{R}$ , then the dependence becomes linear: indeed,

$$\frac{2}{\pi}\beta_{p,\mathbb{R}} = \frac{2}{\pi}(p^*-1) \le \cot\left(\frac{\pi}{2p^*}\right) = \|\mathcal{H}_X^{\mathbb{T}}\|_{\mathcal{L}(L^p(\mathbb{T},X))} \le p^*-1 = \beta_{p,\mathbb{R}},$$

where, as above,  $p^* := \max\{p, p/(p-1)\}.$ 

# **2.4.** STOPPING TIMES

A random variable  $\tau : \Omega \to \mathbb{R}_+$  is called an *optional stopping time* (or just a *stopping time*) if  $\{\tau \le t\} \in \mathcal{F}_t$  for each  $t \ge 0$ . With an optional stopping time  $\tau$  we associate a  $\sigma$ -field  $\mathcal{F}_{\tau} := \{A \in \mathcal{F}_{\infty} : A \cap \{\tau \le t\} \in \mathcal{F}_t, t \in \mathbb{R}_+\}$ . Note that  $M_{\tau}$  is strongly  $\mathcal{F}_{\tau}$ -measurable for any local martingale M. For any stopping time  $\tau$  we define  $\sigma$ -field  $\mathcal{F}_{\tau-}$  in the following way

$$\mathscr{F}_{\tau-} := \sigma \left\{ \mathscr{F}_0 \cup (\mathscr{F}_t \cap \{t < \tau\}), t > 0 \right\}$$

$$(2.4.1)$$

(see [89, p. 491]). Note that for any stopping time  $\tau$  and  $\sigma$  both  $\tau \wedge \sigma$  and  $\tau \vee \sigma$  are stopping times as well. We refer the reader to [89, Chapter 7] for details on stopping times.

Due to the existence of a càdlàg version of a martingale  $M : \mathbb{R}_+ \times \Omega \to X$ , we can define an *X*-valued random variables  $M_{\tau-}$  and  $\Delta M_{\tau}$  for any stopping time  $\tau$  in the following way:  $M_{\tau-} = \lim_{\epsilon \to 0} M_{(\tau-\epsilon) \lor 0}$ ,  $\Delta M_{\tau} = M_{\tau} - M_{\tau-}$ .

#### 2.4.1. Predictable and totally inaccessible stopping times

**Definition 2.4.1.** Let  $\tau$  be a stopping time. Then  $\tau$  is called *predictable* if there exists a sequence of stopping times  $(\tau_n)_{n\geq 1}$  such that  $\tau_n < \tau$  a.s. on  $\{\tau > 0\}$  for each  $n \geq 1$  and  $\tau_n \nearrow \tau$  a.s.

For a predictable stopping time  $\tau$  we define  $\mathscr{F}_{\tau-}$  in the following way analogous to (2.4.1) (see [89, Chapter 25])

$$\mathscr{F}_{\tau-} = \sigma(\mathscr{F}_{\tau_n})_{n \ge 1}.$$

Due to the equivalent form (2.4.1),  $F_{\tau-}$  does not depend on the choice of the announcing sequence  $(\tau_n)_{n\geq 0}$  (see also [89, Lemma 25.2(iii)]).

**Definition 2.4.2.** Let  $\tau$  be a stopping time. Then  $\tau$  is called *totally inaccessible* if  $\mathbb{P}\{\tau = \sigma < \infty\} = 0$  for each predictable stopping time  $\sigma$ .

The reader can find more information on predictable and totally inaccessible stopping times in [85, Definition I.2.7] and [89, Chapter 25].

**Lemma 2.4.3.** Let X be a Banach space,  $V : \mathbb{R}_+ \times \Omega \to X$  be a predictable càdlàg process. Let  $\tau$  be a totally inaccessible stopping time. Then  $\Delta V_{\tau} = 0$  a.s.

*Proof.* It is sufficient to show that  $\langle \Delta V_{\tau}, x^* \rangle = 0$  a.s. for any  $x^* \in X^*$ . Then the statement follows from [85, Proposition I.2.24].

Let *X* be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a local martingale. Then *M* has a càdlàg version (see Proposition 2.2.1), and therefore we can define an adapted càdlàg process  $M^{\tau-} = (M_t^{\tau-})_{t\geq 0}$  in the following way

$$M_t^{\tau-} := \lim_{\varepsilon \to 0} M_{(\tau-\varepsilon) \wedge t}, \quad t \ge 0,$$
(2.4.2)

where we set  $M_t = 0$  for t < 0. Notice that  $M^{\tau-}$  is not necessarily a local martingale. For instance if  $X = \mathbb{R}$  and M is a compensated Poisson process,  $\tau := \inf_{t \ge 0} \{\Delta M_t > 0\}$ , then  $M_t^{\tau-} = -(t \land \tau)$  a.s. for each  $t \ge 0$ , so it is a supermartingale which is not even a local martingale. Nevertheless, if  $\tau$  is a predictable stopping time, then the following lemma holds.

**Lemma 2.4.4.** Let X be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a local martingale,  $\tau$  be a predictable stopping time. Then  $M^{\tau-}$  defined as in (2.4.2) is a local martingale. Moreover, if M is an L<sup>1</sup>-bounded martingale, then  $M^{\tau-}$  is an L<sup>1</sup>-bounded martingale as well.

*Proof.* Without loss of generality we can let  $M_0 = 0$  a.s. First assume that M is an  $L^{\infty}$ -bounded martingale. Let  $(\tau_n)_{n\geq 1}$  be an announcing to  $\tau$  sequence of stopping times, i.e.  $\tau_n < \tau$  a.s. on  $\{\tau > 0\}$  and  $\tau_n \nearrow \tau$  a.s. as  $n \to \infty$ . Then  $M^{\tau_n}$  is an  $L^1$ -bounded martingale for each  $n \ge 1$ . Moreover,  $M_t^{\tau_n} \to M_t^{\tau_-}$  a.s. as  $n \to \infty$  for each  $t \ge 0$ . On

the other hand,  $M_t^{\tau_n} = \mathbb{E}(M_t | \mathscr{F}_{\tau_n}) \to \mathbb{E}(M_t | \mathscr{F}_{\tau^-})$  a.s. as  $n \to \infty$  by [79, Theorem 3.3.8] and [89, Lemma 25.2(iii)], and hence in  $L^1$  by the uniform boundedness due to the boundedness of  $M_{\infty}$ . Therefore for each  $t \ge 0$  we have that  $M_t^{\tau^-} = \mathbb{E}(M_t | \mathscr{F}_{\tau^-})$  is integrable, hence for all  $0 \le s \le t$ 

$$\mathbb{E}(M_t^{\tau-}|\mathscr{F}_s) = \mathbb{E}\left(\lim_{n \to \infty} M_t^{\tau_n}|\mathscr{F}_s\right) = \lim_{n \to \infty} \mathbb{E}(M_t^{\tau_n}|\mathscr{F}_s) = \lim_{n \to \infty} M_s^{\tau_n} = M_s^{\tau-},$$

where all the limits are taken in  $L^1(\Omega; X)$ . Hence  $(M_t^{\tau-})_{t\geq 0}$  is a martingale. Moreover, by [79, Corollary 2.6.30]

$$\mathbb{E}\|M_t^{\tau-}\| = \mathbb{E}\|\mathbb{E}(M_t|\mathscr{F}_{\tau-})\| \le \mathbb{E}\|M_t\| \le \mathbb{E}\|M_{\infty}\|, \quad t \ge 0.$$

$$(2.4.3)$$

Now we treat the general case. Without loss of generality using a stopping time argument assume that M is an  $L^1$ -bounded martingale. Let  $(M^m)_{m\geq 1}$  be a sequence of X-valued  $L^{\infty}$ -bounded martingales such that  $M_{\infty}^m \to M_{\infty}$  in  $L^1(\Omega; X)$  as  $m \to \infty$ . Analogously the first part of the proof  $M_t^{\tau-} = \mathbb{E}(M_t | \mathscr{F}_{\tau-})$  for each  $t \ge 0$ ; moreover, by (2.4.3)  $((M^m)_t^{\tau-})_{m\geq 1}$  is a Cauchy sequence in  $L^1(\Omega; X)$ . Therefore by [79, Corollary 2.6.30],  $(M^m)_t^{\tau-} \to M_t^{\tau-}$  in  $L^1(\Omega; X)$  for each  $t \ge 0$ , hence for each  $t \ge s \ge 0$  by [79, Corollary 2.6.30]

$$\mathbb{E}(M_t^{\tau-}|\mathscr{F}_s) = \mathbb{E}\left(\lim_{m \to \infty} (M^m)_t^{\tau-}|\mathscr{F}_s\right) = \lim_{m \to \infty} \mathbb{E}((M^m)_t^{\tau-}|\mathscr{F}_s)$$
$$= \lim_{m \to \infty} (M^m)_s^{\tau-} = M_s^{\tau-},$$

where all the limits are again taken in  $L^1(\Omega; X)$ . Therefore  $(M_t^{\tau-})_{t\geq 0}$  is an  $L^1$ -martingale.

**Lemma 2.4.5.** Let X be a Banach space,  $1 \le p < \infty$ ,  $M : \mathbb{R}_+ \times \Omega \to X$  be an  $L^p$ -bounded martingale,  $\tau$  be a predictable stopping time. Then  $(\Delta M_\tau \mathbf{1}_{[0,t]}(\tau))_{t\ge 0}$  is an  $L^p$ -bounded martingale as well.

*Proof.* By the definition of a predictable stopping time there exists an increasing sequence of stopping times  $(\tau_n)_{n\geq 0}$  such that  $\tau_n < \tau$  a.s. for each  $n \geq 0$  on  $\{\tau > 0\}$  and  $\tau_n \nearrow \tau$  a.s. as  $n \to \infty$ . Then  $M^{\tau}, M^{\tau_1}, \dots, M^{\tau_n}, \dots$  are  $L^p$ -bounded martingales. Moreover,  $M_t^{\tau} - M_t^{\tau_n} \to \Delta M_{\tau} \mathbf{1}_{[0,t]}(\tau)$  is in  $L^p(\Omega; X)$  for each  $t \geq 0$  due to the fact that  $\Delta M_{\tau} = \mathbb{E}(M_{\infty}|\mathscr{F}_{\tau}) - \mathbb{E}(M_{\infty}|\mathscr{F}_{\tau-})$  and [79, Corollary 2.6.30]. Consequently,  $(\Delta M_{\tau} \mathbf{1}_{[0,t]}(\tau))_{t\geq 0}$  is an  $L^p$ -bounded martingale.

**Lemma 2.4.6.** Let  $F : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be a locally integrable càdlàg adapted process,  $\tau$  be a predictable stopping time. Let  $G, H : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be such that  $G_t = F_\tau \mathbf{1}_{[0,t]}(\tau)$ ,  $H_t = \mathbf{1}_{[0,t]}(\tau)\mathbb{E}_{\mathscr{F}_{\tau^-}}F_\tau$  for each  $t \ge 0$ . Then G - H is a local martingale.

*Proof.* Without loss of generality suppose that *F* is integrable. First of all notice that *H* is a predictable process thanks to [89, Lemma 25.3(ii)], and *G* is adapted due to

the fact that  $G_t = F_{\tau \wedge t} \mathbf{1}_{[0,t]}(\tau)$ . Fix  $t > s \ge 0$ . By [89, Lemma 25.2(i)],  $\mathscr{F}_s \cap \{s < \tau\} \subset \mathscr{F}_{\tau-}$ and  $\mathscr{F}_s \cap \{t < \tau\} \subset \mathscr{F}_t \cap \{t < \tau\} \subset \mathscr{F}_{\tau-}$ . Hence,

$$\mathscr{F}_{s} \cap \{s < \tau \leq t\} \subset \mathscr{F}_{\tau-}$$

and so

$$\begin{split} \mathbb{E}(G_t - H_t | \mathscr{F}_s) &= \mathbb{E}(F_\tau \mathbf{1}_{\{\tau \le t\}} - \mathbf{1}_{\{\tau \le t\}} \mathbb{E}_{\mathscr{F}_{\tau}-} F_\tau | \mathscr{F}_s) \\ &= \mathbb{E}(F_\tau \mathbf{1}_{\{\tau \le s\}} - \mathbf{1}_{\{\tau \le s\}} \mathbb{E}_{\mathscr{F}_{\tau}-} F_\tau | \mathscr{F}_s) \\ &+ \mathbb{E}(F_\tau \mathbf{1}_{\{s < \tau \le t\}} - \mathbf{1}_{\{s < \tau \le t\}} \mathbb{E}_{\mathscr{F}_{\tau}-} F_\tau | \mathscr{F}_s) \\ &= G_s - H_s + \mathbb{E}(\mathbb{E}(F_\tau - \mathbb{E}_{\mathscr{F}_{\tau}-} F_\tau | \mathscr{F}_{\tau}-) \mathbf{1}_{\{s < \tau \le t\}} | \mathscr{F}_s \cap \{s < \tau \le t\}) = G_s - H_s. \end{split}$$

**Corollary 2.4.7.** Let X be a Banach space,  $\tau$  be a predictable stopping time,  $\xi \in L^1(\Omega; X)$  be  $\mathscr{F}_{\tau}$ -measurable such that  $\mathbb{E}_{\mathscr{F}_{\tau}_{-}}\xi = 0$ . Let  $M : \mathbb{R}_{+} \times \Omega \to X$  be such that  $M_t = \xi \mathbf{1}_{[0,t]}(\tau)$ . Then M is a martingale.

*Proof.* The case  $X = \mathbb{R}$  follows from Lemma 2.4.6 and the fact that  $\xi \mathbf{1}_{\tau \le t}$  is  $\mathscr{F}_t$ -measurable for each  $t \ge 0$  by the definition of  $\mathscr{F}_t$ . For the general case we notice that  $\langle M, x^* \rangle$  is a martingale for each  $x^* \in X$  and since M is integrable it follows that M is a martingale.

#### 2.4.2. Quasi-left continuous martingales and martingales with accessible jumps

Let *X* be a Banach space. An *X*-valued local martingale is called *quasi-left continu*ous if  $\Delta M_{\tau} = 0$  a.s. on the set { $\tau < \infty$ } for each predictable stopping time  $\tau$  (see [85, Chapter I.2] for more information).

We call

$$[\tau] = \{((\omega, t) \in \Omega \times \mathbb{R}_+ : t = \tau(\omega)\}$$

the *graph* of  $\tau$  (although it is strictly speaking, the restriction of the graph of  $\tau$  to  $\Omega \times \mathbb{R}_+$ ). An *X*-valued local martingale is said to have *accessible jumps* if there exists a sequence of predictable stopping times  $(\tau_n)_{n\geq 0}$  with disjoint graphs such that a.s.

$$\{t \ge 0 : \Delta M_t \neq 0\} \subset \{\tau_1, \tau_2, \dots, \tau_n, \dots\}.$$
(2.4.4)

(see [89, p.499] and [89, Corollary 26.16]).

The reader can find more information on quasi-left continuous martingales and martingales with accessible jumps in [54, 85, 89, 184, 185]

# 2.4.3. The canonical decomposition

**Definition 2.4.8.** Let  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be an adapted càdlàg process. *A* has *accessible jumps* if  $\Delta A_{\tau} = 0$  a.s. for any totally inaccessible stopping time  $\tau$ . *A* is called *quasi-left continuous* if  $\Delta A_{\tau} = 0$  a.s. for any predictable stopping time  $\tau$ .

 $\square$ 

*Remark* 2.4.9. According to [89, Proposition 25.17] one can show that for any pure jump increasing adapted càdlàg process  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  there exist unique increasing adapted càdlàg processes  $A^a, A^q : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  such that  $A^a$  has accessible jumps,  $A^q$  is quasi-left continuous,  $A_0^q = 0$  and  $A = A^a + A^q$ .

Because of Remark 2.4.9 the following lemma makes sense.

**Lemma 2.4.10.** Let  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be an increasing adapted càdlàg process,  $A_0 = 0$ a.s. Then there exist unique increasing adapted càdlàg  $A^c, A^q, A^a : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  such that  $A_0^c = A_0^q = A_0^a = 0$ ,  $A^c$  is continuous a.s.,  $A^q$  and  $A^a$  are pure jump a.s.,  $A^q$  is quasi-left continuous,  $A^a$  has accessible jumps, and  $A = A^c + A^q + A^a$ .

*Proof.* The statement follows from [89, Proposition 25.17] and Lemma 2.2.9.

The following decomposition theorem was shown by Yoeurp in [190] and follows from [89, Theorem 26.14 and Corollary 26.16].

**Proposition 2.4.11** (Decomposition of local martingales, Yoeurp, Meyer). Let M:  $\mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a local martingale. Then there exists a unique decomposition  $M = M^c + M^q + M^a$ , where  $M^c : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  is a continuous local martingale,  $M^q, M^a : \mathbb{R}_+ \times \Omega \to \mathbb{R}$ are purely discontinuous local martingales,  $M^q$  is quasi-left continuous,  $M^a$  has accessible jumps,  $M_0^c = M_0^q = 0$ , and then  $[M^c] = [M]^c$ ,  $[M^q] = [M]^q$  and  $[M^a] = [M]^a$ , with  $[M]^c$ ,  $[M]^q$  and  $[M]^a$  are defined as in Lemma 2.4.10.

We will refer to the decomposition in Proposition 2.4.11 as the *canonical* decomposition of *M*.

**Corollary 2.4.12** (Yoeurp decomposition). Let  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a purely discontinuous martingale. Then there exist unique purely discontinuous martingales  $M^a, M^q : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  such that  $M^a$  is has accessible jumps,  $M^q$  is quasi-left continuous,  $M_0^q = 0$  and  $M = M^a + M^q$ . Moreover, then  $[M^a] = [M]^a$  and  $[M^q] = [M]^q$ .

**Corollary 2.4.13.** Let  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a purely discontinuous martingale which is both with accessible jumps and quasi-left continuous. Then  $M = M_0$  a.s.

*Proof.* Without loss of generality we can set  $M_0 = 0$ . Then M = M + 0 = 0 + M are decompositions of *M* into a sum of a martingale with accessible jumps and a quasileft continuous martingale. Since by Corollary 2.4.12 this decomposition is unique, M = 0 a.s.

In the sequel we will need the following proposition.

**Proposition 2.4.14.** Let  $1 , <math>M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a purely discontinuous  $L^p$ -martingale. Let  $(M^n)_{n\geq 1}$  be a sequence of purely discontinuous martingales such that  $M^n_{\infty} \to M_{\infty}$ in  $L^p(\Omega)$ . Then the following assertions hold

(a) if  $(M^n)_{n\geq 1}$  have accessible jumps, then M has accessible jumps as well;

(b) if (M<sup>n</sup>)<sub>n≥1</sub> are quasi-left continuous martingales, then M is quasi-left continuous as well.

*Proof.* We will only show (a), (b) can be proven in the same way. Without loss of generality suppose that  $M_0 = 0$  and  $M_0^n = 0$  for each  $n \ge 1$ . Let  $M^a, M^q : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be purely discontinuous martingales such that  $M^a$  has accessible jumps,  $M^q$  is quasileft continuous,  $M_0^a = M_0^q = 0$  and  $M = M^a + M^q$  (see Corollary 2.4.12). Then by Corollary 2.4.12, the Doob maximal inequality [93, Theorem 1.3.8(iv)] and the fact the a quadratic variation is a.s. nonnegative

$$\mathbb{E}|M_{\infty} - M_{\infty}^{n}|^{p} = \mathbb{E}[M - M^{n}]_{\infty}^{\frac{p}{2}} = \mathbb{E}\left([M^{a} - M^{n}]_{\infty} + [M^{q}]_{\infty}\right)^{\frac{p}{2}} \ge \mathbb{E}[M^{q}]_{\infty}^{\frac{p}{2}},$$

and since  $\mathbb{E}|M_{\infty} - M_{\infty}^{n}|^{p} \to 0$  as  $n \to \infty$ ,  $\mathbb{E}[M^{q}]_{\infty}^{\frac{p}{2}} = 0$ . Therefore  $M^{q} = 0$  a.s., so *M* has accessible jumps.

Let us turn to the infinite dimensional case.

**Definition 2.4.15.** Let *X* be a Banach space. A martingale  $M : \mathbb{R}_+ \times \Omega \to X$  has *accessible jumps* if  $\Delta M_{\tau} = 0$  a.s. for any totally inaccessible stopping time  $\tau$ . A martingale  $M : \mathbb{R}_+ \times \Omega \to X$  is called *quasi-left continuous* if  $\Delta M_{\tau} = 0$  a.s. for any predictable stopping time  $\tau$ .

**Lemma 2.4.16.** Let X be a reflexive Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontin*uous martingale.* 

- (i) *M* has accessible jumps if and only if for each  $x^* \in X^*$  the martingale  $\langle M, x^* \rangle$  has accessible jumps;
- (ii) *M* is quasi-left continuous if and only if for each  $x^* \in X^*$  the martingale  $\langle M, x^* \rangle$  is quasi-left continuous.

*Proof.* Without loss of generality we can assume that *X* is a separable Banach space. We will show only (*i*), while (*ii*) can be proven analogously.

(i): The "only if" part is obvious. For "if" part we fix a dense subset  $(x_m^*)_{m\geq 1}$  of  $X^*$ . Let  $\tau$  be a totally inaccessible stopping time. Then  $\Delta \langle M_\tau, x_m^* \rangle = \langle \Delta M_\tau, x_m^* \rangle = 0$  a.s. for each  $m \geq 1$ . Hence  $\Delta M_\tau = 0$  a.s., and the "if" part is proven.

**Definition 2.4.17.** Let *X* be a Banach space,  $p \in (1,\infty)$ . Then we define  $\mathcal{M}_X^{p,q} \subset \mathcal{M}_X^{p,d}$  as the linear space of all *X*-valued purely discontinuous quasi-left continuous  $L^p$ -bounded martingales which start at 0. We define  $\mathcal{M}_X^{p,a} \subset \mathcal{M}_X^{p,d}$  as the linear space of all *X*-valued purely discontinuous  $L^p$ -bounded martingales with accessible jumps.

**Proposition 2.4.18.** Let X be a Banach space,  $1 . Then <math>\mathcal{M}_X^{p,q}$  and  $\mathcal{M}_X^{p,a}$  are closed subspaces of  $\mathcal{M}_X^{p,d}$ .

*Proof.* We only will show the case of  $\mathcal{M}_X^{p,q}$ , the proof for  $\mathcal{M}_X^{p,a}$  is analogous. Let  $(M^n)_{n\geq 1} \in \mathcal{M}_X^{p,q}$  be such that  $(\mathcal{M}_\infty^n)_{n\geq 1}$  is a Cauchy sequence in  $L^p(\Omega; X)$ . Let  $\xi = \lim_{n\to\infty} \mathcal{M}_\infty^n$  in  $L^p(\Omega; X)$ . Define an *X*-valued martingale *M* as follows:  $M_t = \mathbb{E}(\xi|\mathscr{F}_t)$ ,  $t\geq 0$ . Then since conditional expectation is a contraction in  $L^p(\Omega; X)$ ,  $M_0 = \lim_{n\to\infty} \mathcal{M}_0^n = 0$ . Now let us show that *M* is quasi-left continuous. By Lemma 2.4.16 it is sufficient to show that  $\langle M, x^* \rangle$  is quasi-left continuous for each  $x^* \in X^*$ . Fix  $x^* \in X^*$ . Define  $N := \langle M, x^* \rangle$  and  $N^n := \langle M^n, x^* \rangle$  for each  $n \geq 1$ . Then

$$\begin{split} \mathbb{E} \| N_{\infty} - N_{\infty}^{n} \|^{p} &\approx_{p} \mathbb{E} [N - N^{n}]_{\infty}^{\frac{p}{2}} = \mathbb{E} \left( [N - N^{n}]_{\infty}^{c} + [N - N^{n}]_{\infty}^{q} + [N - N^{n}]_{\infty}^{a} \right)^{\frac{p}{2}} \\ &= \mathbb{E} \left( [N]_{\infty}^{c} + [N - N^{n}]_{\infty}^{q} + [N]_{\infty}^{a} \right)^{\frac{p}{2}} \ge \mathbb{E} \left( [N]_{\infty}^{c} + [N]_{\infty}^{a} \right)^{\frac{p}{2}}, \end{split}$$

and since the first expression vanishes as  $n \to \infty$ ,  $[N]_{\infty}^{c} = [N]_{\infty}^{a} = 0$  a.s., so *N* is quasi-left continuous. Since  $x^* \in X^*$  was arbitrary,  $M \in \mathcal{M}_X^{p,q}$ .

The following lemma follows from Corollary 2.4.13.

**Lemma 2.4.19.** Let X be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous martingale. Let M be both with accessible jumps and quasi-left continuous. Then  $M = M_0$  a.s. In other words,  $\mathcal{M}_X^{p,q} \cap \mathcal{M}_X^{p,a} = 0$ .

*Proof.* Without loss of generality set  $M_0 = 0$ . Suppose that  $\mathbb{P}(M \neq 0) > 0$ . Then there exists  $x^* \in X^*$  such that  $\mathbb{P}(\langle M, x^* \rangle \neq 0) > 0$ . Let  $N = \langle M, x^* \rangle$ . Then *N* is both with accessible jumps and quasi-left continuous. Hence by Corollary 2.4.13, N = 0 a.s.

**Definition 2.4.20.** A purely discontinuous local martingale  $M^d : \mathbb{R}_+ \times \Omega \to X$  is called to have the *Yoeurp decomposition* if there exist purely discontinuous local martingales  $M^q, M^a : \mathbb{R}_+ \times \Omega \to X$  such that  $M^q$  is quasi-left continuous,  $M^a$  has accessible jumps,  $M_0^q = 0$ , and  $M^d = M^q + M^a$ .

*Remark* 2.4.21. Analogously to Remark 2.2.19 it follows from [89, Corollary 26.16] that the Yoeurp decomposition is unique.

Composing Definition 2.2.18 and 2.4.20 we get the canonical decomposition.

**Definition 2.4.22.** A local martingale  $M : \mathbb{R}_+ \times \Omega \to X$  is called to have the *canonical decomposition* if there exist local martingales  $M^c, M^q, M^a : \mathbb{R}_+ \times \Omega \to X$  such that  $M^c$  is continuous,  $M^q$  and  $M^a$  are purely discontinuous,  $M^q$  is quasi-left continuous,  $M^a$  has accessible jumps,  $M_0^c = M_0^q = 0$ , and  $M = M^c + M^q + M^a$ .

*Remark* 2.4.23. Notice that if  $M = M^c + M^q + M^a$  is the canonical decomposition, then  $\Delta M_\tau^q = \Delta M_\tau$  for any totally inaccessible stopping time  $\tau$  since in this case  $\Delta M_\tau^c = \Delta M_\tau^a = 0$  by the definition of a continuous local martingale and a local martingale with accessible jumps. Analogously,  $\Delta M_\tau^a = \Delta M_\tau$  for any predictable stopping time  $\tau$ .

The reader can find further details on the martingale decomposition discussed above in [54, 85, 89, 122, 184, 190].

*Remark* 2.4.24. Note that if a local martingale *M* has some canonical decomposition, then this decomposition is unique (see Remark 2.2.19 and [89, 184, 185, 190]).

## 2.4.4. Time-change

A nondecreasing, right-continuous family of stopping times  $\tau = (\tau_s)_{s\geq 0}$  is called a *random time-change*. Since  $\mathbb{F}$  is right-continuous, according to [89, Lemma 7.3] the same holds true for the *induced filtration*  $\mathbb{G} = (\mathscr{G}_s)_{s\geq 0} = (\mathscr{F}_{\tau_s})_{s\geq 0}$  (see more in [89, Chapter 7]). Let *X* be a Banach space. A martingale  $M : \mathbb{R}_+ \times \Omega \to X$  is said to be  $\tau$ -continuous if *M* is an a.s. constant on every interval  $[\tau_{s-}, \tau_s]$ ,  $s \geq 0$ , where we let  $\tau_{0-} = 0$ . In the sequel we will frequently apply the following theorem.

**Theorem 2.4.25.** Let  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be a strictly increasing continuous predictable process such that  $A_0 = 0$  and  $A_t \to \infty$  as  $t \to \infty$  a.s. Let  $\tau = (\tau_s)_{s\geq 0}$  be a random timechange defined as  $\tau_s := \{t : A_t = s\}$ ,  $s \geq 0$ . Then  $(A \circ \tau)(t) = (\tau \circ A)(t) = t$  a.s. for each  $t \geq 0$ . Let  $\mathbb{G} = (\mathcal{G}_s)_{s\geq 0} = (\mathcal{F}_{\tau_s})_{s\geq 0}$  be the induced filtration. Then  $(A_t)_{t\geq 0}$  is a random time-change with respect to  $\mathbb{G}$  and for any  $\mathbb{F}$ -bounded martingale  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  the following holds

- (i)  $M \circ \tau$  is a continuous G-bounded martingale if and only if M is continuous, and
- (ii)  $M \circ \tau$  is a purely discontinuous G-bounded martingale if and only if M is purely discontinuous.

*Proof.* Let us first show that  $(A \circ \tau)(t) = (\tau \circ A)(t) = t$  a.s. for each  $t \ge 0$ . Fix  $t \ge 0$ . Then a.s.

$$(\tau \circ A)(t) = \tau_{A_t} = \{s : A_s = A_t\} = t.$$
(2.4.5)

Since *A* is strictly increasing continuous and starts at zero, there exists  $S_t : \Omega \to \mathbb{R}_+$  such that  $A_{S_t} = t$  a.s. Then by (2.4.5) and the definition of  $S_t$  a.s.

$$(A \circ \tau)(t) = (A \circ \tau)(A_{S_t}) = (A \circ (\tau \circ A))(S_t) = A_{S_t} = t.$$

Now we turn to the second part of the theorem. Notice that  $s \mapsto \tau_s$ ,  $s \ge 0$ , is a continuous strictly increasing G-predictable process which starts at zero. Then for each  $t \ge 0$  one has that  $A_t = \{s : \tau_s = t\}$ , so  $(A_t)_{t\ge 0}$  is a random time-change with respect to the filtration G. Since  $(A \circ \tau)(t) = (\tau \circ A)(t) = t$  a.s. for each  $t \ge 0$ , it is sufficient to show only "if" parts of both (i) and (ii).

(i) follows from the fact that  $\tau_{s-} = \tau_s$  (so M is  $\tau$ -continuous), and the Kazamaki theorem [89, Theorem 17.24]. Let us now show (ii). Thanks to [89, Theorem 7.12]  $M \circ \tau$  is a martingale. Let  $N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a continuous bounded G-bounded martingale such that  $N_0 = 0$ . Then by (i),  $N \circ A$  is a continuous bounded F-bounded martingale, and therefore by Proposition 2.2.12 the process  $M \cdot (N \circ A)$  is a martingale. Consequently due to [89, Theorem 7.12],  $(M \circ \tau)N = (M \cdot (N \circ A)) \circ \tau$  is a

martingale. Since *N* is taken arbitrary and due to Proposition 2.2.12,  $M \circ \tau$  is purely discontinuous.

## **2.5. STOCHASTIC INTEGRATION**

Let *X* be a Banach space, *H* be a Hilbert space. For each  $h \in H$ ,  $x \in X$  we denote a linear operator  $g \mapsto \langle g, h \rangle x$ ,  $g \in H$ , by  $h \otimes x$ . The process  $\Phi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(H, X)$  is called *elementary predictable* with respect to the filtration  $\mathbb{F} = (\mathscr{F}_t)_{t \ge 0}$  if it is of the form

$$\Phi(t,\omega) = \sum_{k=1}^{K} \sum_{m=1}^{M} \mathbf{1}_{(t_{k-1},t_k] \times B_{mk}}(t,\omega) \sum_{n=1}^{N} h_n \otimes x_{kmn}, \quad t \ge 0, \omega \in \Omega,$$
(2.5.1)

where  $0 \le t_0 < ... < t_K < \infty$ , for each k = 1,...,K the sets  $B_{1k},...,B_{Mk}$  are in  $F_{t_{k-1}}$ , and vectors  $h_1,...,h_N$  are orthogonal.

Let  $M : \mathbb{R}_+ \times \Omega \to H$  be a martingale. Then we define the *stochastic integral* of  $\Phi$  with respect to *M* in the following way:

$$\int_{0}^{t} \Phi(s) \, \mathrm{d}M(s) = \sum_{k=1}^{K} \sum_{m=1}^{M} \mathbf{1}_{B_{mk}} \sum_{n=1}^{N} \langle (M(t_{k} \wedge t) - M(t_{k-1} \wedge t)), h_{n} \rangle x_{kmn}, \ t \ge 0.$$
(2.5.2)

We will often write  $\Phi \cdot M$  for the process  $\int_0^{\cdot} \Phi(s) dM(s)$ . The reader can find more on stochastic integration in the finite dimensional case in [89].

Later we will need the following proposition on the canonical decomposition of a stochastic integral.

**Proposition 2.5.1.** *Let H* be a Hilbert space, *X* be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to H$  be a martingale,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X)$  be elementary progressive. Then

- (*i*) *if M is continuous, then*  $\Phi \cdot M$  *is continuous;*
- (ii) if *M* is purely discontinuous, then  $\Phi \cdot M$  is purely discontinuous;
- *(iii) if M has accessible jumps, then*  $\Phi \cdot M$  *has accessible jumps;*
- *(iv) if M is quasi-left continuous, then*  $\Phi \cdot M$  *is quasi-left continuous.*

*Proof.* (i): If *M* is continuous, then by the construction of a stochastic integral (2.5.2),  $\Phi \cdot M$  is a finite sum of continuous martingales, so it is continuous as well.

(ii): Notice that according to Remark 2.2.15 the space of purely discontinuous martingales is linear, so again as in (i) by Proposition 2.2.12 and (2.5.2),  $\Phi \cdot M$  is a finite sum of purely discontinuous martingales, so it is purely discontinuous as well.

(iii) and (iv): By (2.5.2) we have that for any stopping time  $\tau$  a.s.  $\Delta(\Phi \cdot M)_{\tau} \neq 0$  implies  $\Delta M_{\tau} \neq 0$ . Therefore by Definition 2.4.8 if *M* has accessible jumps, then  $\Phi \cdot M$  has them as well, and if *M* is quasi-left continuous, then  $\Phi \cdot M$  is quasi-left continuous as well.

# **2.6.** MULTIDIMENSIONAL WIENER PROCESS

Let *d* be a natural number.  $W : \mathbb{R}_+ \times \Omega \to \mathbb{R}^d$  is called a *standard d-dimensional Wiener process* if  $\langle W, h \rangle$  is a standard Wiener process for each  $h \in \mathbb{R}^d$  such that ||h|| = 1. The following lemma is a multidimensional variation of [93, (3.2.19)].

**Lemma 2.6.1.** Let  $X = \mathbb{R}$ ,  $d \ge 1$ , W be a standard d-dimensional Wiener process,  $\Phi, \Psi$ :  $\mathbb{R}_+ \times \Omega \rightarrow \mathscr{L}(\mathbb{R}^d, \mathbb{R})$  be elementary predictable. Then for all  $t \ge 0$  a.s.

$$[\Phi \cdot W, \Psi \cdot W]_t = \int_0^t \langle \Phi^*(s), \Psi^*(s) \rangle \, \mathrm{d}s$$

The reader can find more on stochastic integration with respect to a Wiener process in the Hilbert space case in [48], in the case of Banach spaces with a martingale type 2 in [25], and in the UMD case in [126]. Notice that the last mentioned work provides sharp  $L^p$ -estimates for stochastic integrals for the broadest till now known class of spaces.

# **2.7. BROWNIAN REPRESENTATION**

The following theorem can be found in [93, Theorem 3.4.2] (see also [170, 186]).

**Theorem 2.7.1.** Let  $d \ge 1$ ,  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}^d$  be a continuous martingale such that [M]is a.s. absolutely continuous with respect to the Lebesgue measure on  $\mathbb{R}_+$ . Then there exist an enlarged probability space  $(\widetilde{\Omega}, \widetilde{\mathscr{F}}, \widetilde{\mathbb{P}})$  with an enlarged filtration  $\widetilde{\mathbb{F}} = (\widetilde{F}_t)_{t\ge 0}$ , a *d*dimensional standard Wiener process  $W : \mathbb{R}_+ \times \widetilde{\Omega} \to \mathbb{R}^d$  which is defined on the filtration  $\widetilde{\mathbb{F}}$ , and predictable  $\Phi : \mathbb{R}_+ \times \widetilde{\Omega} \to \mathscr{L}(\mathbb{R}^d)$  such that  $M = \Phi \cdot W$ .

# **2.8.** RANDOM MEASURES

Throughout, *H* always denotes a Hilbert space. We let  $(\Omega, \mathscr{F}, \mathbb{P})$  be a complete probability space and let  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$  be a filtration that satisfies the usual conditions. Let  $(J, \mathscr{J})$  be a measurable space. We write  $\widetilde{\mathscr{P}} = \mathscr{P} \otimes \mathscr{J}$  and  $\widetilde{\mathscr{O}} := \mathscr{O} \otimes \mathscr{J}$  for the induced  $\sigma$ -algebras on  $\widetilde{\Omega} = \mathbb{R}_+ \times \Omega \times J$ .

A family  $\mu = \{\mu(\omega; dt, dx), \omega \in \Omega\}$  of nonnegative measures on  $(\mathbb{R}_+ \times J; \mathcal{B}(\mathbb{R}_+) \otimes \mathcal{J})$  is called a *random measure*. A random measure  $\mu$  is called *integer-valued* if it takes values in  $\mathbb{N} \cup \{\infty\}$ , i.e. for each  $A \in \mathcal{B}(\mathbb{R}_+) \otimes \mathcal{F} \otimes \mathcal{J}$  one has that  $\mu(A) \in \mathbb{N} \cup \{\infty\}$  a.s., and if  $\mu(\{t\} \times J) \in \{0, 1\}$  a.s. for all  $t \ge 0$ . We say that  $\mu$  is *non-atomic in time* if  $\mu(\{t\} \times J) = 0$  a.s. for all  $t \ge 0$ .

A process  $F : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  is called *optional* if it is  $\mathscr{O}$ -measurable. A random measure  $\mu$  is called *optional* (resp. *predictable*) if for any  $\widetilde{\mathscr{O}}$ -measurable (resp.  $\widetilde{\mathscr{P}}$ -measurable) nonnegative  $F : \mathbb{R}_+ \times \Omega \times J \to \mathbb{R}_+$  the stochastic integral

$$(F \star \mu)_t(\omega) := \int_{\mathbb{R}_+ \times J} \mathbf{1}_{[0,t]}(s) F(s,\omega,x) \mu(\omega; \mathrm{d} s, \mathrm{d} x), \quad t \ge 0, \ \omega \in \Omega,$$

as a function from  $\mathbb{R}_+ \times \Omega$  to  $\overline{\mathbb{R}}_+$  is optional (resp. predictable).

Let *X* be a Banach space. Then we can extend stochastic integration to *X*-valued processes in the following way. Let  $F : \mathbb{R}_+ \times \Omega \times J \to X$ ,  $\mu$  be a random measure. The integral

$$(F \star \mu)_t := \int_{\mathbb{R}_+ \times J} F(s, \cdot, x) \mathbf{1}_{[0,t]}(s) \mu(\cdot; \mathrm{d} s, \mathrm{d} x), \quad t \ge 0,$$

is well-defined and optional (resp. predictable) if  $\mu$  is optional (resp. predictable), F is  $\tilde{\mathcal{O}}$ -strongly-measurable (resp.  $\tilde{\mathcal{P}}$ -strongly-measurable), and  $\int_{\mathbb{R}_+\times J} ||F|| d\mu$  is a.s. bounded.

A random measure  $\mu$  is called  $\widetilde{\mathscr{P}}$ - $\sigma$ -finite if there exists an increasing sequence of sets  $(A_n)_{n\geq 1} \subset \widetilde{\mathscr{P}}$  such that  $\int_{\mathbb{R}_+\times J} \mathbf{1}_{A_n}(s, \omega, x)\mu(\omega; ds, dx)$  is finite a.s. and  $\bigcup_n A_n = \mathbb{R}_+ \times \Omega \times J$ . According to [85, Theorem II.1.8] every  $\widetilde{\mathscr{P}}$ - $\sigma$ -finite optional random measure  $\mu$  has a *compensator*: a unique  $\widetilde{\mathscr{P}}$ - $\sigma$ -finite predictable random measure  $\nu$ such that  $\mathbb{E} \int_{\mathbb{R}_+\times J} F d\mu = \mathbb{E} \int_{\mathbb{R}_+\times J} F d\nu$  for each  $\widetilde{\mathscr{P}}$ -measurable real-valued nonnegative F. We refer the reader to [85, Chapter II.1] for more details on random measures. For any optional  $\widetilde{\mathscr{P}}$ - $\sigma$ -finite measure  $\mu$  we define the associated compensated random measure by  $\overline{\mu} = \mu - \nu$ .

For each  $\widehat{\mathscr{P}}$ -strongly-measurable  $F : \mathbb{R}_+ \times \Omega \times J \to X$  such that  $\mathbb{E}(||F|| \star \mu)_{\infty} < \infty$ (or, equivalently,  $\mathbb{E}(||F|| \star \nu)_{\infty} < \infty$ , see the definition of a compensator above) we can define a process  $F \star \overline{\mu}$  by  $F \star \mu - F \star \nu$ . The reader should be warned that in the literature  $F \star \overline{\mu}$  is often used to denote the integral of F over the whole  $\mathbb{R}_+$ (i.e.  $(F \star \overline{\mu})_{\infty}$  in our notation). The following lemma is a vector-valued version of [85, Definition 1.27].

**Lemma 2.8.1.** Let X be a Banach space,  $\mu$  be a  $\widetilde{\mathcal{P}}$ - $\sigma$ -finite optional random measure, F:  $\mathbb{R}_+ \times \Omega \times J \to X$  be  $\widetilde{\mathcal{P}}$ -strongly-measurable such that  $\mathbb{E} \int_{\mathbb{R}_+ \times J} \|F\| \, d\mu < \infty$ . Then  $(\int_{[0,t] \times J} F \, d\bar{\mu})_{t \ge 0}$  is a purely discontinuous X-valued martingale.

*Proof.* It is sufficient to show that

$$t \mapsto \left\langle \int_{[0,t] \times J} F \, \mathrm{d}\bar{\mu}, x^* \right\rangle = \int_{[0,t] \times J} \langle F, x^* \rangle \, \mathrm{d}\bar{\mu}, \quad t \ge 0,$$

is a purely discontinuous martingale for each  $x^* \in X^*$ , which can be shown similarly the discussion right below [85, Definition 1.27].

We will also need the following lemma.

**Lemma 2.8.2.** Let  $A \in \widetilde{\mathcal{P}}$ ,  $\mu_1$  be a  $\widetilde{\mathcal{P}}$ - $\sigma$ -finite random measure with a compensator  $v_1$ . Then  $\mu_2 = \mu_1 \mathbf{1}_A$  is a  $\widetilde{\mathcal{P}}$ - $\sigma$ -finite random measure and  $v_2 = v_1 \mathbf{1}_A$  is a compensator for  $\mu_2$ .

*Proof.*  $\mu_2$  is  $\widetilde{\mathscr{P}}$ - $\sigma$ -finite since  $\mu_2 \leq \mu_1$  a.s. Moreover,  $\mu_2$  is optional. Indeed, let F:  $\mathbb{R}_+ \times \Omega \times J \to \mathbb{R}_+$  be  $\widetilde{\mathscr{O}}$ -measurable. Then

$$F \star \mu_2 = F \star (\mu_1 \mathbf{1}_A) = (F \mathbf{1}_A) \star \mu_1,$$

and the last process is obviously optional.

Now let us show that  $v_2 = v_1 \mathbf{1}_A$ . Let  $F : \mathbb{R}_+ \times \Omega \times J \to \mathbb{R}$  be simple  $\widetilde{\mathscr{P}}$ -measurable. Since  $\mu_1$  is  $\widetilde{\mathscr{P}}$ - $\sigma$ -finite, so are  $v_1, \mu_1, v_2$ . Hence, we can assume without loss of generality that  $F \star \mu_1$  exists and is integrable. Then  $F \star \mu_2 = F \star (\mu_1 \mathbf{1}_A) = (F \mathbf{1}_A) \star \mu_1$  exists and is integrable. Moreover,

$$\mathbb{E}(F \star \mu_2)_{\infty} = \mathbb{E}((F\mathbf{1}_A) \star \mu_1)_{\infty} = \mathbb{E}((F\mathbf{1}_A) \star v_1)_{\infty} = \mathbb{E}(F \star v_2)_{\infty},$$

so  $v_2$  is a compensator of  $\mu_2$ .

The reader can find more information on random measures in [85, 89, 110, 114, 131].

# **2.9.** $\gamma$ -radonifying operators

Let  $(\gamma'_n)_{n\geq 1}$  be a sequence of independent standard Gaussian random variables on a probability space  $(\Omega', \mathscr{F}', \mathbb{P}')$  (we reserve the notation  $(\Omega, \mathscr{F}, \mathbb{P})$  for the probability space on which our processes live) and let *H* be a separable Hilbert space. A bounded operator  $R \in \mathscr{L}(H, X)$  is said to be  $\gamma$ -radonifying if for some (and then for each) orthonormal basis  $(h_n)_{n\geq 1}$  of *H* the Gaussian series  $\sum_{n\geq 1} \gamma'_n Rh_n$  converges in  $L^2(\Omega'; X)$ . We then define

$$\|R\|_{\gamma(H,X)} := \left(\mathbb{E}' \left\| \sum_{n \ge 1} \gamma'_n Rh_n \right\|_X^2 \right)^{\frac{1}{2}}.$$
 (2.9.1)

Often we will call  $||R||_{\gamma(H,X)}$  the  $\gamma$ -norm of R. This number does not depend on the sequence  $(\gamma'_n)_{n\geq 1}$  and the basis  $(h_n)_{n\geq 1}$ , and defines a norm on the space  $\gamma(H,X)$  of all  $\gamma$ -radonifying operators from H into X. Endowed with this norm,  $\gamma(H,X)$  is a Banach space, which is separable if X is separable. Moreover, if  $X = L^q(S)$  for some separable measure space  $(S, \Sigma, \rho)$ , then thanks to the Trace Duality that is presented e.g. in [80] we have that

$$(\gamma(H, X))^* \simeq \gamma(H^*, X^*).$$
 (2.9.2)

We refer to [80, Section 9.2] and [125] for further details on  $\gamma$ -radonifying operators.

# 2.10. CONVEX, CONCAVE, BICONCAVE, ZIGZAG-CONCAVE FUNC-TIONS

**Definition 2.10.1.** Let *E* be a linear space over the scalar field  $\mathbb{K}$ .

- (i) A function  $f : E \to \mathbb{R}$  is called *convex* if for each  $x, y \in E$ ,  $\lambda \in [0, 1]$  one has that  $f(\lambda x + (1 \lambda)y) \le \lambda f(x) + (1 \lambda)f(y)$ .
- (ii) A function  $f : E \to \mathbb{R}$  is called *concave* if for each  $x, y \in E$ ,  $\lambda \in [0, 1]$  one has that  $f(\lambda x + (1 \lambda)y) \ge \lambda f(x) + (1 \lambda)f(y)$ .

- (iii) A function  $f : E \times E \to \mathbb{R}$  is called *biconcave* if for each  $x, y \in E$  one has that the mappings  $e \mapsto f(x, e)$  and  $e \mapsto f(e, y)$  are concave.
- (iv) A function  $f : E \times E \to \mathbb{R}$  is called *zigzag-concave* if for each  $x, y \in E$  and  $\varepsilon \in \mathbb{K}$ ,  $|\varepsilon| \le 1$  the function  $z \mapsto f(x + z, y + \varepsilon z)$  is concave.

Note that our definition of zigzag-concavity is a bit different from the classical one (e.g. as in [79]): usually one sets in the definition  $|\varepsilon| = 1$ . The reader should pay attention to this extension: thanks to this additional property Theorem 3.3.7 later will be more general than [79, Theorem 4.5.6].

# **2.11.** CORRESPONDING DUAL BASIS

**Definition 2.11.1.** Let *d* be a natural number, *E* be a *d*-dimensional linear space,  $(e_n)_{n=1}^d$  be a basis of *E*. Then  $(e_n^*)_{n=1}^d \subset E^*$  is called the *corresponding dual basis* of  $(e_n)_{n=1}^d$  if  $\langle e_n, e_m^* \rangle = \delta_{nm}$  for each m, n = 1, ..., d.

Note that the corresponding dual basis is uniquely determined. Moreover, if  $(e_n^*)_{n=1}^d$  is the corresponding dual basis of  $(e_n)_{n=1}^d$ , then, the other way around,  $(e_n)_{n=1}^d$  is the corresponding dual basis of  $(e_n^*)_{n=1}^d$  (here we identify  $E^{**}$  with E in the natural way). The following lemma shows that a trace of bilinear forms does not depend on the choice of basis.

**Lemma 2.11.2.** Let *d* be a natural number, *E* be a *d*-dimensional linear space. Let *V* :  $E \times E \rightarrow \mathbb{R}$  and  $W : E^* \times E^* \rightarrow \mathbb{R}$  be two bilinear functions. Then the expression

$$\sum_{n,m=1}^{d} V(e_n, e_m) W(e_n^*, e_m^*)$$
(2.11.1)

does not depend on the choice of basis  $(e_n)_{n=1}^d$  of E (here  $(e_n^*)_{n=1}^d$  is the corresponding dual basis of  $(e_n)_{n=1}^d$ ).

*Proof.* Let  $(e_n)_{n=1}^d$  be a basis of E,  $(e_n^*)_{n=1}^d$  be the corresponding dual basis. Fix another basis  $(\tilde{e}_n)_{n=1}^d$  of E. Let  $(\tilde{e}_n^*)_{n=1}^d$  be the corresponding dual basis of  $E^*$ . Let matrices  $A = (a_{ij})_{i,j=1}^d$  and  $B = (b_{ij})_{i,j=1}^d$  be such that  $\tilde{e}_n = \sum_{i=1}^d a_{ni}e_i$ ,  $\tilde{e}_n^* = \sum_{i=1}^d b_{ni}e_i^*$  for each n = 1, ..., d. Then for each n, m = 1, ..., d

$$\delta_{nm} = \langle \tilde{e}_n, \tilde{e}_m^* \rangle = \left\langle \sum_{i=1}^d a_{ni} e_i, \sum_{j=1}^d b_{mj} e_j^* \right\rangle = \sum_{i=1}^d a_{ni} b_{mi}.$$

Hence  $A^T B = I$ , and thus also  $AB^T = I$  is the identical matrix as well, and therefore  $\sum_{i=1}^{d} a_{in} b_{im} = \delta_{nm}$  for each n, m = 1, ..., d. Consequently, if we paste  $(\tilde{e}_n)_{n=1}^d$  and  $(\tilde{e}_n^*)_{n=1}^d$  in (2.11.1), due to the bilinearity of *V* and *W* 

$$\sum_{n,m=1}^{d} V(\tilde{e}_{n},\tilde{e}_{m})W(\tilde{e}_{n}^{*},\tilde{e}_{m}^{*}) = \sum_{i,j,k,l,n,m=1}^{d} V(a_{ni}e_{i},a_{mj}e_{j})W(b_{nk}e_{k}^{*},b_{ml}e_{l}^{*})$$

$$= \sum_{i,j,k,l=1}^{d} \sum_{n=1}^{d} a_{ni} b_{nk} \sum_{m=1}^{d} a_{mj} b_{ml} V(e_i, e_j) W(e_k^*, e_l^*)$$
  
$$= \sum_{i,j,k,l=1}^{d} \delta_{ik} \delta_{jl} V(e_i, e_j) W(e_k^*, e_l^*)$$
  
$$= \sum_{i,j=1}^{d} V(e_i, e_j) W(e_i^*, e_j^*).$$

**Corollary 2.11.3.** Let *d* be a natural number, *E* be a *d*-dimensional linear space. Let  $V : E \times E \to \mathbb{R}$  and  $W_1, W_2 : E^* \times E^* \to \mathbb{R}$  be bilinear functions. Assume additionally that *V* is symmetric nonnegative (i.e.  $V(x, x) \ge 0$  for all  $x \in E$ ) and that  $W_1(x^*, x^*) \le W_2(x^*, x^*)$  for all  $x^* \in X^*$ . Then

$$\sum_{n,m=1}^{d} V(e_n, e_m) W_1(e_n^*, e_m^*) \le \sum_{n,m=1}^{d} V(e_n, e_m) W_2(e_n^*, e_m^*)$$

for any basis  $(e_n)_{n=1}^d$  of E (here  $(e_n^*)_{n=1}^d$  is the corresponding dual basis of  $(e_n)_{n=1}^d$ ).

*Proof.* Since *V* is symmetric and nonnegative it defines an inner product on  $E \times E$ . Let  $(\tilde{e}_n)_{n=1}^d$  be an orthogonal basis of *E* under the inner product *V* (i.e.  $V(\tilde{e}_n, \tilde{e}_m) = 0$  for all  $n \neq m$ , and  $V(\tilde{e}_n, \tilde{e}_n) \ge 0$  for all n = 1, ..., d). Then we have that

$$\sum_{n,m=1}^{d} V(\tilde{e}_{n}, \tilde{e}_{m}) W_{1}(\tilde{e}_{n}^{*}, \tilde{e}_{m}^{*}) = \sum_{n=1}^{d} V(\tilde{e}_{n}, \tilde{e}_{n}) W_{1}(\tilde{e}_{n}^{*}, \tilde{e}_{n}^{*})$$

$$\leq \sum_{n=1}^{d} V(\tilde{e}_{n}, \tilde{e}_{n}) W_{2}(\tilde{e}_{n}^{*}, \tilde{e}_{n}^{*}) = \sum_{n,m=1}^{d} V(\tilde{e}_{n}, \tilde{e}_{m}) W_{2}(\tilde{e}_{n}^{*}, \tilde{e}_{m}^{*}),$$
(2.11.2)

where  $(\tilde{e}_n^*)_{n=1}^d$  is the corresponding dual basis of  $(\tilde{e}_n)_{n=1}^d$ . Consequently, the desired follows from (2.11.2) and Lemma 2.11.2.

# 2.12. ITÔ'S FORMULA

The following theorem is a variation of [89, Theorem 26.7] which does not use the Hilbert space structure of a finite dimensional space.

**Theorem 2.12.1** (Itô's formula). Let *d* be a natural number, *X* be a *d*-dimensional Banach space,  $f \in C^2(X)$ ,  $M : \mathbb{R}_+ \times \Omega \to X$  be a martingale. Let  $(x_n)_{n=1}^d$  be a basis of *X*,  $(x_n^*)_{n=1}^d$  be the corresponding dual basis. Then for each  $t \ge 0$ 

$$f(M_{t}) = f(M_{0}) + \int_{0}^{t} \langle \partial_{x} f(M_{s-}), dM_{s} \rangle + \frac{1}{2} \int_{0}^{t} \sum_{n,m=1}^{d} f_{x_{n},x_{m}}(M_{s-}) d[\langle M, x_{n}^{*} \rangle, \langle M, x_{m}^{*} \rangle]_{s}^{c}$$
(2.12.1)  
+ 
$$\sum_{s \leq t} (\Delta f(M_{s}) - \langle \partial_{x} f(M_{s-}), \Delta M_{s} \rangle).$$

Here  $\partial_x f(y) \in X^*$  is the Fréchet derivative of f in point  $y \in X$ .

*Proof.* To apply [89, Theorem 26.7] one needs only to endow *X* with a proper Euclidean norm  $||| \cdot |||$ . Define  $||| x ||| = (\sum_{n=1}^{d} |\langle x, x_n^* \rangle|^2)^{1/2}$  for each  $x \in X$ . Then  $(x_n)_{n=1}^{d}$  is an orthonormal basis of  $(X, ||| \cdot |||)$ ,  $M = \sum_{n=1}^{d} \langle M, x_n^* \rangle x_n$  is a decomposition of *M* in this orthonormal basis, and therefore (2.12.1) is equivalent to the formula in [89, Theorem 26.7].

# II

# WEAK DIFFERENTIAL SUBORDINATION AND THE CANONICAL DECOMPOSITION OF MARTINGALES

# WEAK DIFFERENTIAL SUBORDINATION OF PURELY DISCONTINUOUS MARTINGALES

This chapter is based on the paper *Fourier multipliers and weak differential subordination of martingales in UMD Banach spaces* by Ivan Yaroslavtsev, see [189].

In this chapter we introduce the notion of weak differential subordination for martingales and show that a Banach space X is a UMD Banach space if and only if for all  $p \in (1,\infty)$ and all purely discontinuous X-valued martingales M and N such that N is weakly differentially subordinate to M, one has the estimate  $\mathbb{E}||N_{\infty}||^{p} \leq C_{p}\mathbb{E}||M_{\infty}||^{p}$ . As a corollary we derive the sharp estimate for the norms of a broad class of even Fourier multipliers, which includes e.g. the second order Riesz transforms.

<sup>2010</sup> Mathematics Subject Classification. 42B15, 60G46 Secondary: 60B11, 60G42, 60G44, 60G51. *Key words and phrases.* Fourier multipliers, differential subordination, weak differential subordination, UMD Banach spaces, Burkholder function, sharp estimates, Hilbert transform, stochastic integration, Lévy process, purely discontinuous martingales.

# **3.1.** INTRODUCTION

Applying stochastic techniques to Fourier multiplier theory has a long history (see e.g. [9, 10, 15, 23, 32, 61, 66, 118]). It turns out that the boundedness of certain Fourier multipliers with values in a Banach space *X* is equivalent to this Banach space being in a special class, namely in the class of UMD Banach spaces. Burkholder in [32] and Bourgain in [23] showed that the Hilbert transform is bounded on  $L^p(\mathbb{R}; X)$  for  $p \in (1, \infty)$  if and only if *X* is UMD. The same type of assertion can be proven for the Beurling-Ahlfors transform, see the paper [66] by Geiss, Montgomery-Smith and Saksman. Examples of UMD spaces include the reflexive range of  $L^{q}$ -, Sobolev and Besov spaces.

A more general class of Fourier multiplier has been considered in recent works of Bañuelos and Bogdan [10] and Bañuelos, Bielaszewski and Bogdan [9]. They derive sharp estimates for the norm of a Fourier multiplier with symbol

$$m(\xi) = \frac{\int_{\mathbb{R}^d} (1 - \cos\xi \cdot z)\phi(z)V(\,\mathrm{d}z) + \frac{1}{2}\int_{S^{d-1}} (\xi \cdot \theta)^2 \psi(\theta)\mu(\,\mathrm{d}\theta)}{\int_{\mathbb{R}^d} (1 - \cos\xi \cdot z)V(\,\mathrm{d}z) + \frac{1}{2}\int_{S^{d-1}} (\xi \cdot \theta)^2 \mu(\,\mathrm{d}\theta)}, \ \xi \in \mathbb{R}^d,$$
(3.1.1)

on  $L^p(\mathbb{R}^d)$ . Here we will extend their result to  $L^p(\mathbb{R}^d; X)$  for UMD spaces X. More precisely, we will show that a Fourier multiplier  $T_m$  with a symbol of the form (3.1.1) is bounded on  $L^p(\mathbb{R}^d; X)$  if V is a Lévy measure,  $\mu$  is a Borel positive measure,  $|\phi|, |\psi| \leq 1$ , and that then the norm of  $T_m$  does not exceed the UMD<sub>p</sub> constant of X. In Subsection 3.4.2, several examples of symbols m of the form (3.1.1) are given, and we will see that for some particular symbols m the norm of  $T_m$  equals the UMD constant.

To prove the generalization of the results in [9, 10] we will need additional geometric properties of a UMD Banach space. In the fundamental paper [35], Burkholder showed that a Banach space *X* is UMD if and only if for some  $\beta > 0$  there exists a zigzag-concave function  $U : X \times X \to \mathbb{R}$  (i.e., a function *U* such that  $U(x+z, y+\varepsilon z)$  is concave in *z* for any sign  $\varepsilon$  and for any  $x, y \in X$ ) such that  $U(x, y) \ge ||y||^p - \beta^p ||x||^p$  for all  $x, y \in X$ . Such a function *U* is called a *Burkholder function*. In this situation, we can in fact take  $\beta$  equal to the UMD<sub>p</sub> constant of *X* (see Section 2.3 and Theorem 3.3.7). By exploiting appropriate Burkholder functions *U* one can prove a wide variety of interesting results (see [11, 14, 15, 16, 33, 34, 179] and the works [133, 134, 135, 138, 139, 140, 141, 142, 143, 144] by Osekowski). For our purposes the following result due to Burkholder [33] (for the scalar case) and Wang [179] (for the Hilbert space case) is of special importance:

**Theorem 3.1.1.** Let *H* be a Hilbert space,  $(d_n)_{n\geq 0}$ ,  $(e_n)_{n\geq 0}$  be two *H*-valued martingale difference sequences such that  $||e_n|| \leq ||d_n||$  a.s. for all  $n \geq 0$ . Then for each  $p \in (1, \infty)$ ,

$$\mathbb{E} \left\| \sum_{n \ge 0} e_n \right\|^p \le (p^* - 1)^p \mathbb{E} \left\| \sum_{n \ge 0} d_n \right\|^p.$$

Here and in the sequel  $p^* = \max(p, p')$ , where  $\frac{1}{p} + \frac{1}{p'} = 1$ . This result cannot be generalized beyond the Hilbertian setting; see [140, Theorem 3.24(i)] and [79, Example 4.5.17]. In the present chapter we will show the following UMD variant of Theorem 3.1.1:

**Theorem 3.1.2.** Let X be a UMD space,  $(d_n)_{n\geq 0}$ ,  $(e_n)_{n\geq 0}$  be two X-valued martingale difference sequences,  $(a_n)_{n\geq 0}$  be a scalar-valued adapted sequence such that  $|a_n| \leq 1$  and  $e_n = a_n d_n$  for all  $n \geq 0$ . Then for each  $p \in (1, \infty)$ 

$$\mathbb{E}\left\|\sum_{n\geq 0}e_n\right\|^p\leq \beta_{p,X}^p\mathbb{E}\left\|\sum_{n\geq 0}d_n\right\|^p,$$

where  $\beta_{p,X}$  is the UMD<sub>*p*</sub>-constant of *X* (notice that Burkholder proved the identity  $\beta_{p,H} = p^* - 1$  for a Hilbert space *H*, see [33]). Theorem 3.1.2 generalizes a famous Burkholder's result [30, Theorem 2.2] on martingale transforms, where  $(a_n)_{n\geq 0}$  was supposed to be predictable. The main tool for proving Theorem 3.1.2 is a Burkholder function with a stricter zigzag-concavity: now we also require  $U(x + z, y + \varepsilon z)$  to be concave in *z* for any  $\varepsilon$  such that  $|\varepsilon| \leq 1$ . In the finite dimensional case one gets it for free thanks to the existence of an explicit formula of *U* (see Remark 3.5.4 and [179]). Here we show the existence of such a Burkholder function.

For the applications of our abstract results to the theory of Fourier multipliers we extend Theorem 3.1.2 to the continuous time setting. Namely, we show an analogue of Theorem 3.1.2 for purely discontinuous martingales (i.e. martingales which quadratic variations are pure jump processes, see Subsection 3.3.2).

An extension of Theorem 3.1.2 to general continuous-time martingales is shown in the paper [184]. Nevertheless, the sharp estimate in this extension for the case of continuous martingales remains an open problem. This problem is in fact of interest in Harmonic Analysis. If true, this sharp estimate can be used to study a larger class of multipliers, including the Hilbert transform  $\mathcal{H}_X$ . Garling in [61] proved that

$$\|\mathscr{H}_X\|_{\mathscr{L}(L^p(\mathbb{R};X))} \leq \beta_{p,X}^2$$

and it is a long-standing open problem (see [79, pp.496–497]) to prove a linear estimate of the form

$$\|\mathcal{H}_X\|_{\mathcal{L}(L^p(\mathbb{R};X))} \leq C\beta_{p,X}$$

for some constant *C*. Here we will show that the latter estimate would indeed follow if one can show the existence of a Burkholder function with certain additional properties. At present, the existence of such Burkholder functions is known only in the Hilbert space case (see Remark 3.5.4).

# **3.2.** PRELIMINARIES

The following lemma is a multidimensional version of [89, Theorem 26.6(v)].

**Lemma 3.2.1.** Let d be a natural number, H be a d-dimensional Hilbert space,  $M : \mathbb{R}_+ \times \Omega \to H$  be a martingale,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, \mathbb{R})$  be elementary progressive. Then  $[\Phi \cdot M] \leq_d ||\Phi||^2 \cdot [M]$  a.s.

*Proof.* Let  $(h_n)_{n=1}^d$  be an orthogonal basis of H,  $\Phi_1, \ldots, \Phi_d : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be such that  $\Phi = \sum_{n=1}^d \Phi_n h_n$ , and  $M_1, \ldots, M_d : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be martingales such that  $M = \sum_{n=1}^d M_n h_n$ . Notice that thanks to the definition of a quadratic variation (2.2.4) one has that  $[M] = [M_1] + \cdots + [M_d]$ . Then since a quadratic variation is a positive-definite quadratic form (see [89, Theorem 26.6]), thanks to [89, Theorem 26.6(v)] one has for each  $t \ge 0$  a.s.,

$$\begin{split} [\Phi \cdot M]_t &= [\Phi_1 \cdot M_1 + \dots + \Phi_d \cdot M_d]_t \lesssim_d [\Phi_1 \cdot M_1]_t + \dots + [\Phi_d \cdot M_d]_t \\ &= (\|\Phi_1\|^2 \cdot [M_1])_t + \dots + (\|\Phi_d\|^2 \cdot [M_d])_t \\ &\lesssim_d (\|\Phi\|^2 \cdot [M])_t. \end{split}$$

Using Lemma 3.2.1 one can extend stochastic integral to the case of general  $\Phi$ . In particular, the following lemma on stochastic integration can be shown.

**Lemma 3.2.2.** Let *d* be a natural number, *H* be a *d*-dimensional Hilbert space,  $p \in (1,\infty)$ ,  $M, N : \mathbb{R}_+ \times \Omega \to H$  be  $L^p$ -martingales,  $F : H \to H$  be a measurable function such that  $\|F(h)\| \le C \|h\|^{p-1}$  for each  $h \in H$  and some C > 0. Let  $N_- : \mathbb{R}_+ \times \Omega \to H$  be such that  $(N_-)_t = N_{t-}$  for each  $t \ge 0$ . Then  $F(N_-) \cdot M$  is a martingale and for each  $t \ge 0$ ,

$$\mathbb{E}|(F(N_{-})\cdot M)_{t}| \lesssim_{p,d} C(\mathbb{E}||N_{t}||^{p})^{\frac{p-1}{p}} (\mathbb{E}||M_{t}||^{p})^{\frac{1}{p}}.$$
(3.2.1)

*Proof.* First notice that  $F(N_{-})$  is predictable. Therefore, thanks to Lemma 3.2.1 and [89, Theorem 26.12], in order to prove that  $F(N_{-})$  is stochastically integrable with respect to M and that  $F(N_{-}) \cdot M$  is a martingale it is sufficient to show that  $\mathbb{E}(||F(N_{-})||^{2} \cdot [M])_{t}^{\frac{1}{2}} < \infty$ . Without loss of generality suppose that  $M_{0} = N_{0} = 0$  a.s. and C = 1. Then

$$\mathbb{E}(\|F(N_{-})\|^{2} \cdot [M])_{t}^{\frac{1}{2}} \leq \mathbb{E}(\|N_{t-}\|^{2(p-1)} \cdot [M]_{t})^{\frac{1}{2}} \leq \mathbb{E}\left(\sup_{0 \leq s \leq t} \|N_{s}\|^{p-1} [M]_{t}^{\frac{1}{2}}\right)$$

$$\stackrel{(i)}{\leq} (\mathbb{E}\sup_{0 \leq s \leq t} \|N_{s}\|^{p})^{\frac{p-1}{p}} (\mathbb{E}[M]_{t}^{\frac{p}{2}})^{\frac{1}{p}}$$

$$\stackrel{(ii)}{\lesssim}_{p} (\mathbb{E}\|N_{t}\|^{p})^{\frac{p-1}{p}} (\mathbb{E}\|M_{t}\|^{p})^{\frac{1}{p}} < \infty,$$
(3.2.2)

where (*i*) follows from the Hölder inequality, and (*ii*) holds thanks to [89, Theorem 26.12] and [93, Theorem 1.3.8(iv)].

Now let us show (3.2.1):

$$\mathbb{E}|(F(N_{-})\cdot M)_{t}| \lesssim_{p} \mathbb{E}[F(N_{-})\cdot M]_{t}^{\frac{1}{2}} \lesssim_{d} \mathbb{E}(||F(N_{-})||^{2}\cdot [M])_{t}^{\frac{1}{2}}$$

 $\overset{(iii)}{\lesssim}_{p} \left( \mathbb{E} \| N_{t} \|^{p} \right)^{\frac{p-1}{p}} \left( \mathbb{E} \| M_{t} \|^{p} \right)^{\frac{1}{p}}.$ 

Here (*i*) follows from [89, Theorem 26.12], (*ii*) holds thanks to Lemma 3.2.1, and (*iii*) follows from (3.2.2).  $\Box$ 

# **3.3. UMD BANACH SPACES AND WEAK DIFFERENTIAL SUBORDI-**NATION

From now on the scalar field  $\mathbb{K}$  can be either  $\mathbb{R}$  or  $\mathbb{C}$ .

# 3.3.1. Discrete case

In this subsection we assume that *X* is a Banach space over the scalar field  $\mathbb{K}$  and with a separable dual *X*<sup>\*</sup>. Let  $(\Omega, \mathscr{F}, \mathbb{P})$  be a complete probability space with filtration  $\mathbb{F} := (\mathscr{F}_n)_{n \ge 0}, \mathscr{F}_0 = \{\emptyset, \Omega\}.$ 

**Definition 3.3.1.** Let  $(f_n)_{n\geq 0}$ ,  $(g_n)_{n\geq 0}$  be *X*-valued local martingales. For each  $n \geq 1$  we define  $df_n := f_n - f_{n-1}$ ,  $dg_n := g_n - g_{n-1}$ .

- (i) g is *differentially subordinate* to f (we will often write  $g \ll f$ ) if one has that  $||dg_n|| \le ||df_n||$  a.s. for all  $n \ge 1$  and  $||g_0|| \le ||f_0||$  a.s.
- (ii) *g* is *weakly differentially subordinate* to *f* (we will often write  $g \ll^{w} f$ ) if for each  $x^* \in X^*$  one has that  $|\langle dg_n, x^* \rangle| \le |\langle df_n, x^* \rangle|$  a.s. for all  $n \ge 1$  and  $|\langle g_0, x^* \rangle| \le |\langle f_0, x^* \rangle|$  a.s.

The following characterization of Hilbert spaces can be found in [140, Theorem 3.24(i)]:

**Theorem 3.3.2.** A Banach space X is isomorphic to a Hilbert space if and only if for some (equivalently, for all)  $1 there exists a constant <math>\alpha_{p,X} > 0$  such that for any pair of X-valued local martingales  $(f_n)_{n\geq 0}$ ,  $(g_n)_{n\geq 0}$  such that g is differentially subordinate to f one has that

$$\mathbb{E}\|g_n\|^p \le \alpha_{p,X}^p \mathbb{E}\|f_n\|^p \tag{3.3.1}$$

for each  $n \ge 1$ .

By the Pettis measurability theorem [79, Theorem 1.1.20], we may assume that *X* is separable. Then weak differential subordination implies differential subordination. Indeed, let  $(x_k)_{k\geq 1}$  be a dense subset of *X*,  $(x_k^*)_{k\geq 1}$  be a sequence of linear functionals on *X* such that  $\langle x_k, x_k^* \rangle = ||x_k||$  and  $||x_k^*|| = 1$  for each  $k \geq 1$  (such a sequence exists by the Hahn-Banach theorem). Let  $(g_n)_{n\geq 0}$  be weakly differentially subordinate to  $(f_n)_{n\geq 0}$ . Then for each  $n \geq 1$  a.s.

$$\|dg_n\| = \sup_{k\geq 1} |\langle dg_n, x_k^*\rangle| \leq \sup_{k\geq 1} |\langle df_n, x_k^*\rangle| = \|df_n\|.$$

By the same reasoning  $||g_0|| \le ||f_0||$  a.s. This means that the weak differential subordination property is more restrictive than the differential subordination property. Therefore, under the weak differential subordination, one could expect that the assertions of the type (3.3.1) characterize a broader class of Banach spaces *X*. Actually we will prove the following theorem, which extends [34, Theorem 2] to the UMD case.

**Theorem 3.3.3.** A Banach space X is a UMD space if and only if for some (equivalently, for all)  $1 there exists a constant <math>\beta > 0$  such that for all X-valued local martingales  $(f_n)_{n\geq 0}$  and  $(g_n)_{n\geq 0}$  such that g is weakly differentially subordinate to f one has

$$\mathbb{E} \|g_n\|^p \le \beta^p \mathbb{E} \|f_n\|^p, \ n \ge 1.$$
(3.3.2)

If this is the case then the smallest admissible  $\beta$  is the UMD constant  $\beta_{p,X}$ .

Theorem 3.1.2 is contained in this result as a special case. The proof of Theorem 3.3.3 consists of several steps.

**Proposition 3.3.4.** Let X be a Banach space. Let  $(f_n)_{n\geq 0}$ ,  $(g_n)_{n\geq 0}$  be two X-valued local martingales. Then g is weakly differentially subordinate to f if and only if there exists an adapted scalar-valued process  $(a_n)_{n\geq 0}$  such that  $|a_n| \leq 1$  a.s. for all  $n \geq 1$ ,  $dg_n = a_n df_n$  a.s. and  $g_0 = a_0 f_0$  a.s.

For the proof we will need two lemmas.

**Lemma 3.3.5.** Let X be a Banach space,  $\ell_1, \ell_2 \in X^*$  be such that  $ker(\ell_1) \subset ker(\ell_2)$ . Then there exists  $a \in \mathbb{K}$  such that  $\ell_2 = a\ell_1$ .

*Proof.* If  $\ell_2 = 0$ , then the assertion is obvious and one can take a = 0. Suppose that  $\ell_2 \neq 0$ . Then  $\operatorname{codim}(\ker(\ell_2)) = 1$  (see [96, p.80]), and there exists  $x_0 \in X \setminus \ker(\ell_2)$  such that  $x_0 \oplus \ker(\ell_2) = X$ . Notice that since  $\operatorname{codim}(\ker(\ell_1)) \leq 1$  and  $\ker(\ell_1) \subset \ker(\ell_2)$ , one can easily conclude that  $\ker(\ell_1) = \ker(\ell_2)$ . Let  $a = \ell_2(x_0)/\ell_1(x_0)$ . Fix  $y \in X$ . Then there exists  $\lambda \in \mathbb{K}$  such that  $y - \lambda x_0 \in \ker(\ell_1) = \ker(\ell_2)$ . Therefore

$$\ell_2(y) = \ell_2(\lambda x_0) + \ell_2(y - \lambda x_0) = a\ell_1(\lambda x_0) + a\ell_1(y - \lambda x_0) = a\ell_1(y),$$

hence  $\ell_2 = a\ell_1$ .

**Lemma 3.3.6.** Let X be a Banach space,  $(S, \Sigma, \mu)$  be a measure space. Let  $f, g : S \to X$  be strongly measurable such that  $|\langle g, x^* \rangle| \le |\langle f, x^* \rangle| \mu$ -a.s. for each  $x^* \in X^*$ . Then there exists a measurable function  $a : S \to \mathbb{K}$  such that  $||a||_{\infty} \le 1$  and g = af.

*Proof.* By the Pettis measurability theorem [79, Theorem 1.1.20] we can assume *X* to be separable. Let  $(x_m)_{m\geq 1}$  be a dense subset of *X*. By the Hahn-Banach theorem we can find a sequence  $(x_m^*)_{m\geq 1}$  of linear functionals on *X* such that  $\langle x_m, x_m^* \rangle = ||x_m||$  and  $||x_m^*|| = 1$  for each  $m \ge 1$ . Let  $Y_0 = \mathbb{Q} - \text{span}(x_1^*, x_2^*, ...)$ , and let  $Y = \overline{\text{span}(x_1^*, x_2^*, ...)}$ 

be a separable closed subspace of  $X^*$ . Then  $X \hookrightarrow Y^*$  isometrically. Fix a set of full measure  $S_0$  such that for all  $x^* \in Y_0$ ,  $|\langle g, x^* \rangle| \le |\langle f, x^* \rangle|$  on  $S_0$ . Fix  $x^* \in Y$ . Let  $(y_k)_{k\ge 1}$  be a sequence in  $Y_0$  such that  $y_k \to x^*$  in Y as  $k \to \infty$ . Then on  $S_0$  we have that  $|\langle g, y_k^* \rangle| \to |\langle g, x^* \rangle|$  and  $|\langle f, y_k \rangle| \to |\langle f, x^* \rangle|$ . Consequently for each  $s \in S_0$ ,

$$|\langle g(s), x^* \rangle| \le |\langle f(s), x^* \rangle|, \quad x^* \in Y.$$
(3.3.3)

Therefore the linear functionals f(s),  $g(s) \in X \hookrightarrow Y^*$  are such that ker  $g(s) \subset \ker f(s)$ , and hence by Lemma 3.3.5 there exist a(s) defined for each fixed  $s \in S_0$  such that g(s) = a(s)f(s). By (3.3.3) one has that  $|a(s)| \le 1$ .

Let us construct a measurable version of *a*.  $Y_0$  is countable since it is a  $\mathbb{Q}$  – span of a countable set. Let  $Y_0 = (y_m)_{m \ge 1}$ . For each m > 1 construct  $A_m \in \Sigma$  as follows:

$$A_m = \{s \in S : \langle g(s), y_m \rangle \neq 0, \langle g(s), y_{m-1} \rangle = 0, \dots, \langle g(s), y_1 \rangle = 0\}$$

and put  $A_1 = \{s \in S : \langle g(s), y_1 \rangle \neq 0\}$ . Obviously on the set  $S \setminus \bigcup_{m=1}^{\infty} A_m$  one has that g = 0, so one can redefine a := 0 on  $S \setminus \bigcup_{m=1}^{\infty} A_m$ . For each  $m \ge 1$  we redefine  $a := \frac{\langle g, y_m \rangle}{\langle f, y_m \rangle}$  on  $A_m$ . Then *a* constructed in such a way is  $\Sigma$ -measurable.

*Proof of Proposition* 3.3.4. The proposition follows from Lemma 3.3.6: the assumption of this lemma holds for  $df_n$  and  $dg_n$  for any  $n \ge 1$ , and for  $f_0$  and  $g_0$ . So according to Lemma 3.3.6 there exists a sequence  $(a_n)_{n\ge 0}$  which is a.s. bounded by 1, such that  $dg_n = a_n df_n$  for each  $n \ge 1$  and  $g_0 = a_0 f_0$  a.s. Moreover, again thanks to Lemma 3.3.6,  $a_n$  is  $\mathcal{F}_n$ -measurable, so  $(a_n)_{n\ge 0}$  is adapted.

In [35] Burkholder showed that the UMD property is equivalent to the existence of a certain biconcave function  $V : X \times X \rightarrow \mathbb{R}$ . With a slight variation of his argument (see Remark 3.3.10) one can also show the equivalence with the existence of a certain zigzag-concave function with a better structure.

Theorem 3.3.7 (Burkholder). For a Banach space X the following are equivalent

- 1. X is a UMD Banach space;
- 2. for each  $p \in (1,\infty)$  there exists a constant  $\beta > 0$  and a zigzag-concave function U:  $X \times X \rightarrow \mathbb{R}$  such that

$$U(x, y) \ge \|y\|^p - \beta^p \|x\|^p, \ x, y \in X.$$
(3.3.4)

*The smallest admissible*  $\beta$  *for which such U exists is*  $\beta_{p,X}$ *.* 

*Proof.* The proof is essentially the same as the one given in [79, Theorem 4.5.6], but the construction of *U* is a bit different. The only difference is allowing  $|\varepsilon| \le 1$  instead of  $|\varepsilon| = 1$  for the appropriate scalars  $\varepsilon$ .

For each  $x, y \in X$  we define S(x, y) as a set of all pairs (f, g) of discrete martingales such that

- 1.  $f_0 \equiv x, g_0 \equiv y;$
- 2. there exists  $N \ge 0$  such that  $df_n \equiv 0$ ,  $dg_n \equiv 0$  for  $n \ge N$ ;
- 3.  $(dg_n)_{n\geq 1} = (\varepsilon_n df_n)_{n\geq 1}$  for some sequence of scalars  $(\varepsilon_n)_{n\geq 1}$  such that  $|\varepsilon_n|\leq 1$  for each  $n\geq 1$ .

Then we define  $U: X \times X \rightarrow \mathbb{R} \cup \{\infty\}$  as follows:

$$U(x, y) := \sup \{ \mathbb{E}(\|g_{\infty}\|^{p} - \beta^{p} \|f_{\infty}\|^{p}) : (f, g) \in \mathbb{S}(x, y) \}.$$
(3.3.5)

The rest of the proof repeats the one given in [79, Theorem 4.5.6].

*Remark* 3.3.8. We will call the function *U* constructed above *the Burkholder function*. Notice that this function coincides with the one in the proof of [79, Theorem 4.5.6]. This is due to the fact that the function

$$(\varepsilon_n)_{n=1}^N \mapsto \left(\mathbb{E} \left\| g_0 + \sum_{n=1}^N \varepsilon_n df_n \right\|^p \right)^{\frac{1}{p}}$$

is convex on the  $\mathbb{K}$ -cube { $(\varepsilon_n)_{n=1}^N : |\varepsilon_1|, ..., |\varepsilon_N| \le 1$ } because of the triangle inequality, therefore it takes its supremum on the set of the domain endpoints, namely on the set { $(\varepsilon_n)_{n=1}^N : |\varepsilon_1|, ..., |\varepsilon_N| = 1$ }.

*Remark* 3.3.9. Analogously to [79, (4.31)] by (3.3.5) we have that  $U(\alpha x, \alpha y) = |\alpha|^p U(x, y)$  for each  $x, y \in X$ ,  $\alpha \in \mathbb{K}$ . Therefore U(0,0) = 0, and hence for each  $x \in X$  and each scalar  $\varepsilon$  such that  $|\varepsilon| \le 1$ , by the zigzag-concavity of U in the point (0,0)

$$U(x,\varepsilon x) = \frac{1}{2}U(0+x,0+\varepsilon x) + \frac{1}{2}U(0-x,0-\varepsilon x) \le U(0,0) = 0.$$
(3.3.6)

Let  $\xi, \eta \in L^0(\Omega; X)$  be such that  $|\langle \eta, x^* \rangle| \le |\langle \xi, x^* \rangle|$  for each  $x^* \in X^*$  a.s. Then thanks to Lemma 3.3.6 and (3.3.6),  $U(\xi, \eta) \le 0$  a.s.

*Remark* 3.3.10. For each zigzag-concave function  $U : X \times X \rightarrow \mathbb{R}$  one can construct a biconcave function  $V : X \times X \rightarrow \mathbb{R}$  as follows:

$$V(x, y) = U\left(\frac{x-y}{2}, \frac{x+y}{2}\right), \ x, y \in X.$$
(3.3.7)

Indeed, by the definition of *U*, for each  $x, y \in X$  the functions

$$z \mapsto V(x+z, y) = U\left(\frac{x-y}{2} + \frac{z}{2}, \frac{x+y}{2} + \frac{z}{2}\right),$$
$$z \mapsto V(x, y+z) = U\left(\frac{x-y}{2} - \frac{z}{2}, \frac{x+y}{2} + \frac{z}{2}\right)$$

are concave. Moreover, for each  $x, y \in X$  and  $a, b \in \mathbb{K}$  such that  $|a + b| \le |a - b|$  one has that the function

$$z \mapsto V(x+az, y+bz) = U\left(\frac{x-y}{2} + \frac{(a-b)z}{2}, \frac{x+y}{2} + \frac{(a+b)z}{2}\right)$$

is concave since  $\left|\frac{a+b}{a-b}\right| \le 1$ .

*Remark* 3.3.11. Due to the explicit representation (3.3.5) of *U* we can show that for each  $x_1, x_2, y_1, y_2 \in X$ ,

$$|U(x_1, y_1) - U(x_2, y_2)| \le ||x_1 - x_2||^p + \beta_{p, X}^p ||y_1 - y_2||^p.$$

Therefore *U* is continuous, and consequently *V* is continuous as well.

*Remark* 3.3.12. Notice that if *X* is finite dimensional then by Theorem 2.20 and Proposition 2.21 in [59] there exists a unique translation-invariant measure  $\lambda_X$  on *X* such that  $\lambda_X(\mathbb{B}_X) = 1$  for the unit ball  $\mathbb{B}_X$  of *X*. We will call  $\lambda_X$  a *Lebesgue measure*. Thanks to the Alexandrov theorem [57, Theorem 6.4.1]  $x \mapsto V(x, y)$  and  $y \mapsto V(x, y)$  are a.s. Fréchet differentiable with respect to  $\lambda_X$ , and by [86, Proposition 3.1] and Remark 3.3.11 for a.a.  $(x, y) \in X \times X$  for each  $u, v \in X$  there exists the directional derivative  $\frac{\partial V(x+tu, y+tv)}{\partial t}$ . Moreover,

$$\frac{\partial V(x+tu, y+tv)}{\partial t} = \langle \partial_x V(x, y), u \rangle + \langle \partial_y V(x, y), v \rangle, \qquad (3.3.8)$$

where  $\partial_x V$  and  $\partial_y V$  are the corresponding Fréchet derivatives with respect to the first and the second variable. Thanks to (3.3.8) and Remark 3.3.10 one obtains that for a.e.  $(x, y) \in X \times X$ , for all  $z \in X$  and  $a, b \in \mathbb{K}$  such that  $|a + b| \le |a - b|$ ,

$$V(x + az, y + bz) \le V(x, y) + \frac{\partial V(x + atz, y + btz)}{\partial t}$$
  
=  $V(x, y) + a \langle \partial_x V(x, y), z \rangle + b \langle \partial_y V(x, y), z \rangle.$  (3.3.9)

**Lemma 3.3.13.** Let X be a finite dimensional Banach space,  $V : X \times X \rightarrow \mathbb{R}$  be as defined in (3.3.7). Then there exists C > 0 which depends only on V such that for a.e. pair  $x, y \in X$ ,

$$\|\partial_x V(x, y)\|, \|\partial_y V(x, y)\| \le C(\|x\|^{p-1} + \|y\|^{p-1}).$$

*Proof.* We show the inequality only for  $\partial_x V$ , the proof for  $\partial_y V$  being analogous. First we prove that there exists C > 0 such that  $||\partial_x V(x, y)|| \le C$  for a.e.  $x, y \in X$  such that  $||x||, ||y|| \le 1$ . Let us show this by contradiction. Suppose that such *C* does not exist. Since *V* is continuous by Remark 3.3.11, and since a unit ball in *X* is a compact set, there exists K > 0 such that ||Y(x, y)| < K for all  $x, y \in X$  such that  $||x||, ||y|| \le 2$ . Let  $x_0, y_0 \in X$  be such that  $||x_0||, ||y_0|| \le 1$  and  $||\partial_x V(x_0, y_0)|| > 3K$ . Therefore there exists  $z \in X$  such that ||z|| = 1 and  $\langle \partial_x V(x_0, y_0), z \rangle < -3K$ . Hence we have that  $||x_0 + z|| \le 2$  and because of the concavity of *V* in the first variable

$$V(x_0 + z, y_0) \le V(x_0, y_0) + \langle \partial_x V(x_0, y_0), z \rangle \le K - 3K \le -2K.$$

Consequently,  $|V(x_0 + z, y_0)| > K$ , which contradicts with our suggestion.

Now fix C > 0 such that  $|\partial_x V(x, y)| \le C$  for all  $x, y \in X$  such that  $||x||, ||y|| \le 1$ . Fix  $x, y \in X$ . Without loss of generality assume that  $||x|| \ge ||y||$ . Let L = ||x||. Then  $||\partial_x V(\frac{x}{L}, \frac{y}{L})|| \le C$ . Let  $z \in X$  be such that ||z|| = 1. Then by Remark 3.3.9,

$$|\langle \partial_x V(x,y),z\rangle| = \left|\lim_{t\to 0} \frac{V(x+tz,y) - V(x,y)}{t}\right| = \left|\lim_{t\to 0} \frac{L^p V(\frac{x}{L} + \frac{t}{L}z,\frac{y}{L}) - L^p V(\frac{x}{L},\frac{y}{L})}{L\frac{t}{L}}\right|$$

$$= L^{p-1} \left| \lim_{t \to 0} \frac{V(\frac{x}{L} + tz, \frac{y}{L}) - V(\frac{x}{L}, \frac{y}{L})}{t} \right| = L^{p-1} \left| \left\langle \partial_x V\left(\frac{x}{L}, \frac{y}{L}\right), z \right\rangle \right| \\ \le L^{p-1} C \le C(\|x\|^{p-1} + \|y\|^{p-1}).$$

Therefore since *z* was arbitrary,  $\|\partial_x V(x, y)\| \le C(\|x\|^{p-1} + \|y\|^{p-1})$ . The case  $\|x\| < \|y\|$  can be done in the same way.

**Lemma 3.3.14.** Let X be a finite dimensional Banach space,  $1 , <math>(f_n)_{n \ge 0}$ ,  $(g_n)_{n \ge 0}$ be X-valued martingales on a probability space  $(\Omega, \mathscr{F}, \mathbb{P})$  with a filtration  $\mathbb{F} = (\mathscr{F}_n)_{n \ge 0}$  and assume that  $(g_n)_{n \ge 0}$  is weakly differentially subordinate to  $(f_n)_{n \ge 0}$ . Let  $Y = X \oplus \mathbb{R}$  be the Banach space with the norm as follows:

$$||(x,r)||_Y := (||x||_X^p + |r|^p)^{\frac{1}{p}}, x \in X, r \in \mathbb{R}.$$

Then there exist two sequences  $(f^m)_{m\geq 1}$  and  $(g^m)_{m\geq 1}$  of *Y*-valued martingales on an enlarged probability space  $(\overline{\Omega}, \overline{\mathscr{F}}, \overline{\mathbb{P}})$  with an enlarged filtration  $\overline{\mathbb{F}} = (\overline{\mathscr{F}}_n)_{n\geq 0}$  such that

- 1.  $f_n^m$ ,  $g_n^m$  have absolutely continuous distributions with respect to the Lebesgue measure on Y for each  $m \ge 1$  and  $n \ge 0$ ;
- 2.  $f_n^m \to (f_n, 0), g_n^m \to (g_n, 0)$  pointwise as  $m \to \infty$  for each  $n \ge 0$ ;
- 3. *if for some*  $n \ge 0 \mathbb{E} ||f_n||^p < \infty$ , then for each  $m \ge 1$  one has that  $\mathbb{E} ||f_n^m||^p < \infty$  and  $\mathbb{E} ||f_n^m (f_n, 0)||^p \to 0$  as  $m \to \infty$ ;
- 4. *if for some*  $n \ge 0 \mathbb{E} \|g_n\|^p < \infty$ , then for each  $m \ge 1$  one has that  $\mathbb{E} \|g_n^m\|^p < \infty$  and  $\mathbb{E} \|g_n^m (g_n, 0)\|^p \to 0$  as  $m \to \infty$ ;
- 5. for each  $m \ge 1$  we have that  $(g_n^m)_{n\ge 0}$  is weakly differentially subordinate to  $(f_n^m)_{n\ge 0}$ .

*Proof.* First of all let us show that we may assume that  $f_0$  and  $g_0$  are nonzero a.s. For this purpose we can modify  $f_0$  and  $g_0$  as follows:

$$\begin{split} f_0^{\varepsilon} &= f_0 + \varepsilon \mathbf{x} \mathbf{1}_{f_0 = 0}, \\ g_0^{\varepsilon} &= g_0 + \varepsilon \mathbf{x} \mathbf{1}_{f_0 = 0} + \varepsilon f_0 \mathbf{1}_{g_0 = 0, f_0 \neq 0}, \end{split}$$

where  $\varepsilon > 0$  is arbitrary and  $x \in X$  is fixed. This small perturbation does not destroy the weak differential subordination property. Moreover, if we let  $f_n^{\varepsilon} := f_0^{\varepsilon} + \sum_{k=1}^n df_k$ ,  $g_n^{\varepsilon} := g_0^{\varepsilon} + \sum_{k=1}^n dg_k$  for any  $n \ge 1$ , then  $f_n^{\varepsilon} \to f_n$  and  $g_n^{\varepsilon} \to g_n$  a.s., and  $f_n^{\varepsilon} - f_n \to 0$  and  $g_n^{\varepsilon} - g_n \to 0$  in  $L^p(\Omega; X)$  as  $\varepsilon \to 0$ .

From now we assume that  $f_0$  and  $g_0$  are nonzero a.s. This in fact means that random variable  $a_0$  from Proposition 3.3.4 is nonzero a.s. as well. Let  $\mathbb{B}_Y$  be the unit ball of Y,  $(\mathbb{B}_Y, \mathscr{B}(\mathbb{B}_Y), \hat{\mathbb{P}})$  be a probability space such that  $\hat{\mathbb{P}} := \lambda_Y|_{\mathbb{B}_Y}$  has the uniform Lebesgue distribution on  $\mathbb{B}_Y$  (see Remark 3.3.12). Fix some scalar product  $\langle \cdot, \cdot \rangle : Y \times Y \to \mathbb{R}$  in *Y*. We will construct a random operator  $T : \mathbb{B}_Y \to \mathcal{L}(Y)$  as follows:

$$T(b, y) := \langle b, y \rangle b \quad b \in \mathbb{B}_Y, y \in Y.$$

Note that for each fixed  $b \in \mathbb{B}_Y$  the mapping  $y \mapsto \langle b, y \rangle b$  is a linear operator on *Y*. Moreover,

$$\sup_{b \in \mathbb{B}_Y} \|T(b, \cdot)\|_{\mathscr{L}(Y)} < \infty.$$
(3.3.10)

Now let  $(\overline{\Omega}, \overline{\mathscr{F}}, \overline{\mathbb{P}}) := (\Omega \times \mathbb{B}_Y, \mathscr{F} \otimes \mathscr{B}(\mathbb{B}_Y), \mathbb{P} \otimes \hat{\mathbb{P}})$ . For each  $m \ge 1$  define an operatorvalued function  $Q_m : \overline{\Omega} \to \mathscr{L}(Y)$  as follows:  $Q_m := I + \frac{1}{m}T$ .

Fix  $\varepsilon > 0$ . For each  $n \ge 0$  define  $\tilde{f}_n^{\varepsilon} := (f_n, \varepsilon)$ ,  $\tilde{g}_n^{\varepsilon} := (g_n, \varepsilon a_0)$ . Then  $(\tilde{f}_n^{\varepsilon})_{n\ge 0}$  and  $(\tilde{g}_n^{\varepsilon})_{n\ge 0}$  are *Y*-valued martingales which are nonzero a.s. for each  $n \ge 0$  and are such that  $(\tilde{g}_n^{\varepsilon})_{n\ge 0}$  is weakly differentially subordinate to  $(\tilde{f}_n^{\varepsilon})_{n\ge 0}$ . For each  $m \ge 1$  define *Y*-valued martingales  $f^m$  and  $g^m$  in the following way:

$$\begin{split} f_n^m &:= Q_m \tilde{f}_n^\varepsilon, \ m \ge 1, n \ge 0, \\ g_n^m &:= Q_m \tilde{g}_n^\varepsilon, \ m \ge 1, n \ge 0. \end{split}$$

Let us illustrate that for each  $m \ge 1$ ,  $f^m$  and  $g^m$  are martingales with respect to the filtration  $\overline{\mathbb{F}} = (\overline{\mathscr{F}}_n)_{n\ge 0} := (\mathscr{F}_n \otimes \mathscr{B}(\mathbb{B}_Y))_{t\ge 0}$ : for each  $n \ge 1$  we have

$$\mathbb{E}(f_n^m | \overline{\mathscr{F}}_{n-1}) = \mathbb{E}(Q_m \tilde{f}_n^{\varepsilon} | \mathscr{F}_{n-1} \otimes \mathscr{B}(\mathbb{B}_Y)) \stackrel{(i)}{=} Q_m \mathbb{E}(\tilde{f}_n^{\varepsilon} | \mathscr{F}_{n-1} \otimes \mathscr{B}(\mathbb{B}_Y))$$
$$\stackrel{(ii)}{=} Q_m \tilde{f}_{n-1}^{\varepsilon} = f_{n-1}^m,$$

where (*i*) holds since  $Q_m$  is  $\mathscr{B}(\mathbb{B}_Y)$ -measurable, and (*ii*) holds since  $\tilde{f}_n^{\varepsilon}$  is independent of  $\mathscr{B}(\mathbb{B}_Y)$ . The same can be proven for  $g^m$ . Thanks to (3.3.10) one has that  $\lim_{m\to\infty} \sup_{b\in\mathbb{B}_Y} \|Q_m - I\|_{\mathscr{L}(Y)} = 0$  and hence (2), (3) and (4) hold for  $\tilde{f}^{\varepsilon}$  and  $\tilde{g}^{\varepsilon}$ .

Let us prove (5). For each  $m \ge 1$  and  $n \ge 1$  one has:

$$dg_n^m = dQ_m \tilde{g}_n^\varepsilon = dQ_m a_n \tilde{f}_n^\varepsilon = a_n dQ_m \tilde{f}_n^\varepsilon = a_n df_n^m.$$

The same holds for  $g_0^m$  and  $f_0^m$ .

Now we will show (1). Let us fix a set  $A \subset Y$  of Lebesgue measure zero. Then for each fixed  $n \ge 0$  and  $m \ge 1$ ,

$$\mathbb{E} \mathbf{1}_{f_n^m \in A} = \int_{\Omega} \int_{\mathbb{B}_Y} \mathbf{1}_{\tilde{f}_n^{\varepsilon} + \frac{1}{m} \langle b, \tilde{f}_n^{\varepsilon} \rangle b \in A} d\hat{\mathbb{P}}(b) d\mathbb{P}$$
  
$$= \int_{\Omega} \int_{\mathbb{B}_Y} \mathbf{1}_{\frac{1}{m} \langle b, \tilde{f}_n^{\varepsilon} \rangle b \in A - \tilde{f}_n^{\varepsilon}} d\hat{\mathbb{P}}(b) d\mathbb{P},$$
(3.3.11)

where F - y is a translation of a set  $F \subset Y$  by a vector  $y \in Y$ . For each fixed  $y \in Y \setminus \{0\}$  the distribution of a *Y*-valued random variable  $b \mapsto \langle b, y \rangle b$  is absolutely continuous with respect to  $\lambda_Y$ . Since  $\hat{\mathbb{P}}(A - y) = 0$  for each  $y \in Y \setminus \{0\}$ , one has

$$\int_{\mathbb{B}_Y} \mathbf{1}_{\frac{1}{m}\langle b, y \rangle b \in A - y} \, \mathrm{d}\hat{\mathbb{P}}(b) = 0.$$
(3.3.12)

Recall that  $\mathbb{P}{\{\tilde{f}_n^{\varepsilon} = 0\}} = 0$ , therefore due to (3.3.12) a.s.

$$\int_{\mathbb{B}_Y} \mathbf{1}_{\frac{1}{m} \langle b, \tilde{f}_n^{\varepsilon} \rangle b \in A - \tilde{f}_n^{\varepsilon}} d\hat{\mathbb{P}}(b) = 0$$

Consequently the last double integral in (3.3.11) vanishes. The same works for  $g^m$ .

Now to construct such a sequence for  $((f_n, 0))_{n \ge 0}$  and  $((g_n, 0))_{n \ge 0}$  one needs to construct it for different  $\varepsilon$  and take an appropriate subsequence.

*Proof of Theorem* 3.3.3. The "if" part is obvious thanks to the definition of a UMD Banach space. Let us prove the "only if" part. As in the proof of the lemma above, without loss of generality suppose that *X* is separable and that the set  $\bigcup_n (\{f_n = 0\} \cup \{g_n = 0\})$  is of  $\mathbb{P}$ -measure 0. If it does not hold, we consider  $Y := X \oplus \mathbb{R}$  instead of *X* with the norm of  $(x, r) \in Y$  given by  $||(x, r)||_Y = (||x||_X^p + |r|^p)^{1/p}$ . Notice that then  $\beta_{p,Y} = \beta_{p,X}$ . We can suppose that  $a_0$  is nonzero a.s., so we consider  $(f_n^{\varepsilon})_{n\geq 0} := (f_n \oplus \varepsilon)_{n\geq 0}$  and  $(g_n^{\varepsilon})_{n\geq 0} := (g_n \oplus \varepsilon a_0)_{n\geq 0}$  with  $\varepsilon > 0$ , and let  $\varepsilon$  go to zero.

One can also restrict to a finite dimensional case. Indeed, since *X* is a separable reflexive space,  $X^*$  is separable as well. Let  $(Y_m)_{m\geq 1}$  be an increasing sequence of finite-dimensional subspaces of  $X^*$  such that  $\bigcup_m Y_m = X^*$  and  $\|\cdot\|_{Y_m} = \|\cdot\|_{X^*}$  for each  $m \geq 1$ . Then for each fixed  $m \geq 1$  there exists a linear operator  $P_m : X \to Y_m^*$  of norm 1 defined as follows:  $\langle P_m x, y \rangle = \langle x, y \rangle$  for each  $x \in X, y \in Y_m$ . Then since  $Y_m$  is a closed subspace of  $X^*$ , [79, Proposition 4.33] yields  $\beta_{p',Y_m} \leq \beta_{p',X^*}$ , consequently again by [79, Proposition 4.33]  $\beta_{p,Y_m^*} \leq \beta_{p,X^{**}} = \beta_{p,X}$ . So if we prove the finite dimensional version, then

$$\mathbb{E}\|P_m g_n\|^p \le \beta_{n,X}^p \mathbb{E}\|P_m f_n\|^p, \quad n \ge 0,$$

for each  $m \ge 1$ , and due to the fact that  $||P_m x||_{Y_m^*} \nearrow ||x||_X$  for each  $x \in X$  as  $m \to \infty$ , we would obtain (3.3.2) in the general case.

Let  $\beta$  be the UMD constant of X, and let  $U, V : X \times X \to \mathbb{R}$  be as defined in Theorem 3.3.7 and in (3.3.7) respectively,  $(a_n)_{n\geq 0}$  be as defined in Proposition 3.3.4. By Lemma 3.3.14 we can suppose that  $f_n$  and  $g_n$  have distributions which are absolutely continuous with respect to the Lebesgue measure. Then

$$\mathbb{E}(\|g_{n}\|^{p} - \beta\|f_{n}\|^{p}) \stackrel{(i)}{\leq} \mathbb{E}U(f_{n}, g_{n}) = \mathbb{E}U(f_{n-1} + df_{n}, g_{n-1} + a_{n}df_{n})$$

$$\stackrel{(ii)}{=} \mathbb{E}V(g_{n-1} + f_{n-1} + (a_{n}+1)df_{n}, g_{n-1} - f_{n-1} + (a_{n}-1)df_{n})$$

$$\stackrel{(iii)}{\leq} \mathbb{E}V(g_{n-1} + f_{n-1}, g_{n-1} - f_{n-1})$$

$$+ \mathbb{E}\langle \partial_{x}V(g_{n-1} + f_{n-1}, g_{n-1} - f_{n-1}), (a_{n}+1)df_{n}\rangle \qquad (3.3.13)$$

$$+ \mathbb{E}\langle \partial_{y}V(g_{n-1} + f_{n-1}, g_{n-1} - f_{n-1}), (a_{n}-1)df_{n}\rangle$$

$$\stackrel{(iv)}{=} \mathbb{E}V(g_{n-1} + f_{n-1}, g_{n-1} - f_{n-1})$$

$$\stackrel{(iv)}{=} \mathbb{E}U(f_{n-1}, g_{n-1}).$$

Here (*i*) and (*iii*) hold by Theorem 3.3.7 and (3.3.9) respectively, (*ii*) and (v) follow from the definition of *V*. Let us prove (*iv*). We will show that

$$\mathbb{E}\langle \partial_x V(g_{n-1} + f_{n-1}, g_{n-1} - f_{n-1}), (a_n + 1)df_n \rangle = 0.$$
(3.3.14)

Since both  $f_n$  and  $a_n f_n$  are martingale differences,  $(a_n + 1)df_n$  is a martingale difference as well. Therefore  $\mathbb{E}((a_n - 1)df_n | \mathscr{F}_{n-1}) = 0$ . Note that according to Lemma 3.3.13 a.s.

$$\|\partial_x V(g_{n-1}+f_{n-1},g_{n-1}-f_{n-1})\| \lesssim_V \|f_n\|^{p-1} + \|g_n\|^{p-1}.$$

Therefore by the Hölder inequality  $\langle \partial_x V(g_{n-1}+f_{n-1},g_{n-1}-f_{n-1}), (a_n+1)df_n \rangle$  is integrable. Since  $\partial_x V(g_{n-1}+f_{n-1},g_{n-1}-f_{n-1})$  is  $\mathscr{F}_{n-1}$ -measurable,

$$\mathbb{E}\Big(\big\langle\partial_{x}V\big(g_{n-1}+f_{n-1},g_{n-1}-f_{n-1}\big),(a_{n}+1)df_{n}\big\rangle\big|\mathscr{F}_{n-1}\Big) \\ = \Big\langle\partial_{x}V\big(g_{n-1}+f_{n-1},g_{n-1}-f_{n-1}\big),\mathbb{E}\big((a_{n}+1)df_{n}\big|\mathscr{F}_{n-1}\big)\Big\rangle \\ = \big\langle\partial_{x}V\big(g_{n-1}+f_{n-1},g_{n-1}-f_{n-1}\big),0\big\rangle = 0,$$

so (3.3.14) holds. By the same reason

$$\mathbb{E}\langle \partial_{\gamma} V(g_{n-1}+f_{n-1},g_{n-1}-f_{n-1}),(a_n-1)df_n\rangle = 0,$$

and (*iv*) follows.

Notice that thanks to Remark 3.3.9  $\mathbb{E}(f_0, g_0) \le 0$ . Therefore from the inequality (3.3.13) by an induction argument we get

$$\mathbb{E}(\|g_n\|^p - \beta^p \|f_n\|^p) \le \mathbb{E}U(f_n, g_n) \le \mathbb{E}U(f_{n-1}, g_{n-1}) \le \dots \le \mathbb{E}U(f_0, g_0) \le 0.$$

This terminates the proof.

#### 3.3.2. Continuous time case

Now we turn to continuous time martingales. Let  $(\Omega, \mathscr{F}, \mathbb{P})$  be a probability space with a filtration  $\mathbb{F} = (\mathscr{F}_t)_{t \ge 0}$  that satisfies the usual conditions.

**Definition 3.3.15.** Let  $M, N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be local martingales. Then we say that N is *differentially subordinate* to M (we will often write  $N \ll M$ ) if for each  $x^* \in X^*$  one has that [M] - [N] is an a.s. nondecreasing function and  $|N_0| \le |M_0|$  a.s.

**Definition 3.3.16.** Let  $M, N : \mathbb{R}_+ \times \Omega \to X$  be local martingales. Then we say that N is *weakly differentially subordinate* to M (we will often write  $N \stackrel{w}{\ll} M$ ) if  $\langle N, x^* \rangle \ll \langle M, x^* \rangle$  for each  $x^* \in X^*$ .

The following theorem is a natural extension of Proposition 3.3.4.

**Theorem 3.3.17.** Let X be a Banach space. Then X is a UMD space if and only if for some (equivalently, for all)  $1 there exists <math>\beta > 0$  such that for each purely discontinuous X-valued local martingales  $M, N : \mathbb{R}_+ \times \Omega \to X$  such that N is weakly differentially subordinate to M one has

$$\mathbb{E}\|N_t\|^p \le \beta^p \mathbb{E}\|M_t\|^p. \tag{3.3.15}$$

If this is the case then the smallest admissible  $\beta$  equals the UMD constant  $\beta_{p,X}$ .

**Lemma 3.3.18.** Let X be a finite dimensional Banach space,  $1 , <math>M, N : \mathbb{R}_+ \times \Omega \to X$ be local martingales on a probability space  $(\Omega, \mathscr{F}, \mathbb{P})$  with a filtration  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$  such that N is weakly differentially subordinate to M. Let  $Y = X \oplus \mathbb{R}$  be a Banach space such that  $\|(x, r)\|_Y = (\|x\|_X^p + |r|^p)^{\frac{1}{p}}$  for each  $x \in X$ ,  $r \in \mathbb{R}$ . Then there exist two sequences  $(M^m)_{m\geq 1}$ and  $(N^m)_{m\geq 1}$  of Y-valued martingales on an enlarged probability space  $(\overline{\Omega}, \overline{\mathscr{F}}, \overline{\mathbb{P}})$  with an enlarged filtration  $\overline{\mathbb{F}} = (\overline{\mathscr{F}}_t)_{t\geq 0}$  such that

- 1.  $M_t^m$ ,  $N_t^m$  have absolutely continuous distributions with respect to the Lebesgue measure on Y for each  $m \ge 1$  and  $t \ge 0$ ;
- 2.  $M_t^m \to (M_t, 0), N_t^m \to (N_t, 0)$  pointwise as  $m \to \infty$  for each  $t \ge 0$ ;
- 3. *if for some*  $t \ge 0 \mathbb{E} ||M_t||^p < \infty$ , then for each  $m \ge 1$  one has that  $\mathbb{E} ||M_t^m||^p < \infty$  and  $\mathbb{E} ||M_t^m (M_t, 0)||^p \to 0$  as  $m \to \infty$ ;
- 4. *if for some*  $t \ge 0 \mathbb{E} \|N_t\|^p < \infty$ , then for each  $m \ge 1$  one has that  $\mathbb{E} \|N_t^m\|^p < \infty$  and  $\mathbb{E} \|N_t^m (N_t, 0)\|^p \to 0$  as  $m \to \infty$ ;
- 5. for each  $m \ge 1$  we have that  $N^m$  is weakly differentially subordinate to  $M^m$ .

*Proof.* The proof is essentially the same as one of Lemma 3.3.14.

*Proof of Theorem* 3.3.17. We use a modification of the argument in [179, Theorem 1], where the Hilbert space case was considered. Thanks to the same methods as were applied in the beginning of the proof of Theorem 3.3.3 and using Lemma 3.3.18 instead of Lemma 3.3.14, one can suppose that *X* is finite-dimensional and  $M_t$  and  $N_t$  are nonzero a.s. for each  $t \ge 0$ . We know that  $\mathbb{E}U(M_t, N_t) \ge \mathbb{E}(||N_t||^p - \beta^p ||M_t||^p)$  for each  $t \ge 0$ . On the other hand, thanks to the fact that  $[\langle M, x^* \rangle]$  and  $[\langle N, x^* \rangle]$  are pure jump for each  $x^* \in X^*$  and the finite-dimensional version of Itô formula [89, Theorem 26.7], one has

$$\mathbb{E}U(M_t, N_t) = \mathbb{E}U(M_0, N_0) + \mathbb{E}\int_0^t \langle \partial_x U(M_{s-}, N_{s-}), dM_s \rangle + \mathbb{E}\int_0^t \langle \partial_y U(M_{s-}, N_{s-}), dN_s \rangle + \mathbb{E}I,$$
(3.3.16)

where

$$I = \sum_{0 < s \le t} [\Delta U(M_s, N_s) - \langle \partial_x U(M_{s-}, N_{s-}), \Delta M_s \rangle - \langle \partial_y U(M_{s-}, N_{s-}), \Delta N_s \rangle].$$

Note that since a.s.

$$\Delta|\langle N, x^*\rangle|^2 = \Delta[\langle N, x^*\rangle] \le \Delta[\langle M, x^*\rangle] = \Delta|\langle M, x^*\rangle|^2$$

for each  $x^* \in X^*$ , one has that thanks to Lemma 3.3.6 for each  $s \ge 0$ , for a.e.  $\omega \in \Omega$  there exists  $a_s(\omega)$  such that  $|a_s(\omega)| \le 1$  and  $\Delta N_s(\omega) = a_s(\omega)\Delta M_s(\omega)$ . Therefore for each  $s \ge 0$  by (3.3.9)  $\mathbb{P}$ -a.s.

$$\begin{split} \Delta U(M_{s},N_{s}) &- \langle \partial_{x}U(M_{s-},N_{s-}), \Delta M_{s} \rangle - \langle \partial_{y}U(M_{s-},N_{s-}), \Delta N_{s} \rangle \\ &= V(M_{s-}+N_{s-}+(a_{s}+1)\Delta M_{s},N_{s-}-M_{s-}+(a_{s}-1)\Delta M_{s}) \\ &- V(M_{s-}+N_{s-},N_{s-}-M_{s-}) \\ &- \langle \partial_{x}V(M_{s-}+N_{s-},N_{s-}-M_{s-}), (a_{s}+1)\Delta M_{s} \rangle \\ &- \langle \partial_{y}V(M_{s-}+N_{s-},N_{s-}-M_{s-}), (a_{s}-1)\Delta M_{s} \rangle \leq 0, \end{split}$$

so  $I \le 0$  a.s., and  $\mathbb{E}I \le 0$ . Also

$$\begin{split} \int_0^t \langle \partial_x U(M_{s-}, N_{s-}), \, \mathrm{d}M_s \rangle + \int_0^t \langle \partial_y U(M_{s-}, N_{s-}), \, \mathrm{d}N_s \rangle \\ &= \int_0^t \langle \partial_x V(M_{s-} + N_{s-}, N_{s-} - M_{s-}), \, \mathrm{d}(M_s + N_s) \rangle \\ &+ \int_0^t \langle \partial_y V(M_{s-} + N_{s-}, N_{s-} - M_{s-}), \, \mathrm{d}(N_s - M_s) \rangle, \end{split}$$

so by Lemma 3.2.2 and Lemma 3.3.13 it is a martingale that starts at zero, and therefore its expectation is zero as well. Consequently, thanks to (3.3.4), (3.3.16) and Remark 3.3.9,

$$\mathbb{E}\|N_t\|^p - \beta_{p,X}^p \mathbb{E}\|M_t\|^p \le \mathbb{E}U(M_t, N_t) \le \mathbb{E}U(M_0, N_0) \le 0,$$

and therefore (3.3.15) holds.

As one can see, in our proof we did not need the second order terms of the Itô formula thanks to the nature of the quadratic variation of a purely discontinuous process. Nevertheless, Theorem 3.3.17 holds for arbitrary martingales M and N, but with worse estimates (see Chapter 4). The connection of Theorem 3.3.17 for continuous martingales with the Hilbert transform will be discussed in Section 3.5.

# **3.4.** FOURIER MULTIPLIERS

In [10] and [9] the authors exploited the differential subordination property to show boundedness of certain Fourier multipliers in  $\mathscr{L}(L^p(\mathbb{R}^d))$ . It turned out that it is sufficient to use the weak differential subordination property to obtain the same assertions, but in the vector-valued situation.
### 3.4.1. Basic definitions and the main theorem

Let  $d \ge 1$  be a natural number. Recall that  $\mathscr{S}(\mathbb{R}^d)$  is a space of Schwartz functions on  $\mathbb{R}^d$ . For a Banach space X with a scalar field  $\mathbb{C}$  we define  $\mathscr{S}(\mathbb{R}^d) \otimes X$  as the space of all functions  $f : \mathbb{R}^d \to X$  of the form  $f = \sum_{k=1}^K f_k \otimes x_k$ , where  $K \ge 1, f_1, \dots, f_K \in \mathscr{S}(\mathbb{R}^d)$ , and  $x_1, \dots, x_K \in X$ . Notice that for each  $1 \le p < \infty$  the space  $\mathscr{S}(\mathbb{R}^d) \otimes X$  is dense in  $L^p(\mathbb{R}^d; X)$ .

We define the *Fourier transform*  $\mathscr{F}$  and the *inverse Fourier transform*  $\mathscr{F}^{-1}$  on  $\mathscr{S}(\mathbb{R}^d)$  as follows:

$$\mathscr{F}(f)(t) = \frac{1}{(2\pi)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{-i\langle t, u \rangle} f(u) \, \mathrm{d}u, \quad f \in \mathscr{S}(\mathbb{R}^d), t \in \mathbb{R}^d,$$
$$\mathscr{F}^{-1}(f)(t) = \frac{1}{(2\pi)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{i\langle t, u \rangle} f(u) \, \mathrm{d}u, \quad f \in \mathscr{S}(\mathbb{R}^d), t \in \mathbb{R}^d.$$

It is well-known that for any  $f \in \mathscr{S}(\mathbb{R}^d)$  we have  $\mathscr{F}(f), \mathscr{F}^{-1}(f) \in \mathscr{S}(\mathbb{R}^d)$ , and  $\mathscr{F}^{-1}(\mathscr{F}(f)) = f$ . The reader can find more details on the Fourier transform in [69].

Let  $m : \mathbb{R}^d \to \mathbb{C}$  be measurable and bounded. Then we can define a linear operator  $T_m$  on  $\mathscr{S}(\mathbb{R}^d) \otimes X$  as follows:

$$T_m(f \otimes x) = \mathscr{F}^{-1}(m\mathscr{F}(f)) \cdot x, \quad f \in \mathscr{S}(\mathbb{R}^d), x \in X.$$
(3.4.1)

The operator  $T_m$  is called a *Fourier multiplier*, while the function m is called the *symbol* of  $T_m$ . If X is finite-dimensional then  $T_m$  can be extended to a bounded linear operator on  $L^2(\mathbb{R}^d; X)$ . The question is usually whether one can extend  $T_m$  to a bounded operator on  $L^p(\mathbb{R}^d; X)$  for a general 1 and a given <math>X. Here the answer will be given for m of quite a special form and X with the UMD property.

Let *V* be a Lévy measure on  $\mathbb{R}^d$ , that is  $V(\{0\}) = 0$ ,  $V \neq 0$  and

$$\int_{\mathbb{R}^d} (|x|^2 \wedge 1) V(\mathrm{d} x) < \infty.$$

Let  $\phi \in L^{\infty}(\mathbb{R}^d;\mathbb{C})$  be such that  $\|\phi\|_{L^{\infty}(\mathbb{R}^d;\mathbb{C})} \leq 1$ . Also let  $\mu \geq 0$  be a finite Borel measure on the unit sphere  $S^{d-1} \subset \mathbb{R}^d$ , and  $\psi \in L^{\infty}(S^{d-1};\mathbb{C})$  satisfies  $\|\psi\|_{L^{\infty}(S^{d-1};\mathbb{C})} \leq 1$ .

In the sequel we set  $\frac{a}{0} = 0$  for each  $a \in \mathbb{C}$ . The following result extends [9, Theorem 1.1] to the UMD Banach space setting.

**Theorem 3.4.1.** Let X be a UMD Banach space. Then the Fourier multiplier  $T_m$  with a symbol

$$m(\xi) = \frac{\int_{\mathbb{R}^d} (1 - \cos\xi \cdot z)\phi(z)V(\,\mathrm{d}z) + \frac{1}{2}\int_{S^{d-1}} (\xi \cdot \theta)^2 \psi(\theta)\mu(\,\mathrm{d}\theta)}{\int_{\mathbb{R}^d} (1 - \cos\xi \cdot z)V(\,\mathrm{d}z) + \frac{1}{2}\int_{S^{d-1}} (\xi \cdot \theta)^2 \mu(\,\mathrm{d}\theta)}, \quad \xi \in \mathbb{R}^d, \tag{3.4.2}$$

has a bounded extension on  $L^p(\mathbb{R}^d; X)$  for  $1 . Moreover, then for each <math>f \in L^p(\mathbb{R}^d; X)$ 

$$\|T_m f\|_p \le \beta_{p,X} \|f\|_p.$$
(3.4.3)

*Remark* 3.4.2. The coefficient  $\frac{1}{2}$  in both numerator and denominator of (3.4.1), even though it looks wired and useless (because one can always transform  $\mu$  to  $2\mu$ ), exists because of the strong connection with the Lévy–Khintchin representation of Lévy processes (see e.g. [8, Part 4.1]).

The proof is a modification of the arguments given in [9] and [10], but instead of real-valued process we will work with processes that take their values in a finite dimensional space. For the convenience of the reader the proof will be given in the same form and with the same notations as the original one. However, we will need to justify here some steps, so we cannot just skip the proof. First of all as that was done in [9], we reduce to the case of symmetric *V* and  $\mu = 0$ , and proceed as in the proof of [10, Theorem 1].

In the rest of the section we may assume that *X* is finite dimensional, since it is sufficient to show (3.4.3) for all *f* with values in  $X_0$  for each finite dimensional subspace  $X_0$  of *X*.

Let *v* be a positive finite symmetric measure on  $\mathbb{R}^d$ ,  $\tilde{v} = v/|v|$ . Let  $T_i$  and  $Z_i$ ,  $i = \pm 1, \pm 2, \pm 3, ...$ , be a family of independent random variables, such that each  $T_i$ is exponentially distributed with parameter |v| (i.e.  $\mathbb{E}T_i = 1/|v|$ ), and each  $Z_i$  has  $\tilde{v}$  as a distribution. Let  $S_i = T_1 + \cdots + T_i$  for a positive *i* and  $S_i = -(T_{-1} + \cdots + T_i)$ for a negative *i*. For each  $-\infty < s < t < \infty$  we define  $X_{s,t} := \sum_{s < S_i \le t} Z_i$  and  $X_{s,t-} := \sum_{s < S_i < t} Z_i$ . Note that  $\mathcal{N}(B) = \#\{i : (S_i, Z_i) \in B\}$  defines a Poisson measure on  $\mathbb{R} \times \mathbb{R}^d$ with the intensity measure  $\lambda \otimes v$ , and  $X_{s,t} = \int_{s < v \le t} x \mathcal{N}(dv, dv)$  (see e.g. [165]). Let  $N(s, t) = \mathcal{N}((s, t] \times \mathbb{R}^d)$  be the number of *signals*  $S_i$  such that  $s < S_i \le t$ . The following Lemmas 3.4.3-3.4.6 are multidimensional versions of [10, Lemma 1–5], which can be proven in the same way as in the scalar case.

**Lemma 3.4.3.** Let  $f : \mathbb{R} \times \mathbb{R}^d \times \mathbb{R}^d \to X$  be Borel measurable and be either nonnegative or bounded, and let  $s \le t$ . Then

$$\mathbb{E}\sum_{s< S_i\leq t}F(S_i, X_{s,S_i-,X_{s,S_i}}) = \mathbb{E}\int_s^t \int_{\mathbb{R}^d}F(\nu, X_{s,\nu-,X_{s,\nu-}+z})\nu(\,\mathrm{d} z)\,\mathrm{d} \nu.$$

We will consider the following filtration:

$$\mathbb{F} = \{\mathscr{F}_t\}_{t \in \mathbb{R}} = \{\sigma\{X_{s,t} : s \le t\}\}_{t \in \mathbb{R}}.$$

Recall that for measures  $v_1$  and  $v_2$  on  $\mathbb{R}^d$  the expression  $v_1 * v_2$  means the *convolution* of these measures (we refer the reader [22, Chapter 3.9] for the details). Also for each  $n \ge 1$  we define  $v_1^{*n} := v_1 * \cdots * v_1$ . For each  $t \in \mathbb{R}$  define

*n* times

$$p_t = e^{*t(v-|v|\delta_0)} = \sum_{n=0}^{\infty} \frac{t^n}{n!} (v-|v|\delta_0)^{*n} = e^{-t|v|} \sum_{n=0}^{\infty} \frac{t^n}{n!} v^{*n}.$$

The series converges in the norm of absolute variation of measures. As in [10, (18)] and [9, (3.9)]  $p_t$  is symmetric, and

$$\frac{\partial}{\partial t}p_t = (v - |v|\delta_0) * p_t, \ t \in \mathbb{R}.$$

Also  $p_{t_1+t_2} = p_{t_1} * p_{t_2}$  for each  $t_1, t_2 \in \mathbb{R}$ . In fact for all  $t \le u$  the measure  $p_{u-t}$  is the distribution of  $X_{t,u}$  and  $X_{t,u-}$ . Put

$$\Psi(\xi) = \int_{\mathbb{R}^d} (e^{i\xi \cdot z} - 1) \nu(\,\mathrm{d} z), \ \xi \in \mathbb{R}^d.$$

Thanks to the symmetry of *v* one has as well that

$$\Psi(\xi) = \int_{\mathbb{R}^d} (\cos \xi \cdot z - 1) \nu(\,\mathrm{d} z) = \Psi(-\xi) \le 0.$$

Therefore  $\Psi$  is bounded on  $\mathbb{R}^d$ , and due to [9, (3.12)] we have that the characteristic function of  $p_t$  is of the following form:

$$\hat{p}_t(\xi) = e^{t\Psi(\xi)}, \ \xi \in \mathbb{R}^d.$$

(The reader can find more on characteristic functions in [22, Chapter 3.8].)

Let  $g \in L^{\infty}(\mathbb{R}^d; X)$ . Then for  $x \in \mathbb{R}^d$ ,  $t \le u$ , we define the *parabolic extension* of g by

$$P_{t,u}g(x) := \int_{\mathbb{R}^d} g(x+y) p_{u-t}(dy) = g * p_{u-t}(x) = \mathbb{E}g(x+X_{t,u}).$$

For  $s \le t \le u$  we define the *parabolic martingale* by

$$G_t = G_t(x; s, u; g) := P_{t,u}g(x + X_{s,t}).$$

**Lemma 3.4.4.** We have that  $G_t$  is a bounded  $\mathbb{F}$ -martingale.

Let  $\phi \in L^{\infty}(\mathbb{R}^d; \mathbb{C})$  be symmetric. For each  $x \in \mathbb{R}^d$ ,  $s \le t \le u$ , and  $f \in C_c(\mathbb{R}^d; X)$  we define  $F_t$  as follows:

$$F_{t} = F_{t}(x; s, u; f, \phi) :=$$

$$\sum_{s < S_{i} \le t} [P_{S_{i}, u}f(x + X_{s, S_{i}}) - P_{S_{i}, u}f(x + X_{s, S_{i}})]\phi(X_{s, S_{i}} - X_{s, S_{i}})$$

$$- \int_{s}^{t} \int_{\mathbb{R}^{d}} [P_{v, u}f(x + X_{s, v-} + z) - P_{u, v}f(x + X_{s, v-})]\phi(z)v(dz)dv$$

**Lemma 3.4.5.** We have that  $F_t = F_t(x; s, u; f, \phi)$  is an  $\mathbb{F}$ -martingale for  $t \in [s, u]$ . Moreover,  $\mathbb{E} ||F_t||^p < \infty$  for each p > 0.

**Lemma 3.4.6.**  $G_t(x; s, u; g) = F_t(x; s, u; g, 1) + P_{s,u}g(x)$ .

Analogously to [10, (21)-(22)] one has that for each  $x^* \in X^*$  the quadratic variations of  $\langle F_t(x; s, u; f, \phi), x^* \rangle$  and  $\langle G_t(x; s, u; g), x^* \rangle$  satisfy the following a.s. identities,

$$\begin{split} [\langle F, x^* \rangle]_t &= \sum_{s < S_i \le t} \left( \langle P_{S_i, u} f(x + X_{s, S_i}) - P_{S_i, u} f(x + X_{s, S_i -}), x^* \rangle \right)^2 \phi^2(\Delta X_{s, S_i}), \\ [\langle G, x^* \rangle]_t &= |\langle P_{s, u} g(x), x^* \rangle|^2 + \sum_{s < S_i \le t} \left( \langle P_{S_i, u} g(x + X_{s, S_i}) - P_{S_i, u} g(x + X_{s, S_i -}), x^* \rangle \right)^2. \end{split}$$

It follows that for each  $f \in C_c(\mathbb{R}^d; X)$ ,  $(F_t(x; s, u; f, \phi))_{t \in [s, u]}$  is weakly differentially subordinate to  $(G_t(x; s, u; f))_{t \in [s, u]}$  and by Theorem 3.3.17 one has for each  $t \in [s, u]$ 

$$\mathbb{E} \|F_t(x;s,u;f,\phi)\|^p \le \beta_{p,X}^p \mathbb{E} \|G_t(x;s,u;f)\|^p.$$

Note that  $G_u(x; s, u; f) = f(x + X_{s,u})$ , so

$$\int_{\mathbb{R}^{d}} \mathbb{E} \|F_{u}(x; s, u; f, \phi)\|^{p} dx \leq \beta_{p, X}^{p} \int_{\mathbb{R}^{d}} \mathbb{E} \|f(x + X_{s, u})\|^{p} dx$$
  
=  $\beta_{p, X}^{p} \|f\|_{L^{p}(\mathbb{R}^{d}; X)}^{p}.$  (3.4.4)

Let p' be such that  $\frac{1}{p} + \frac{1}{p'} = 1$ . Consider the linear functional on  $L^{p'}(\mathbb{R}^d; X^*)$ :

$$L^{p'}(\mathbb{R}^d; X^*) \ni g \mapsto \int_{\mathbb{R}^d} \mathbb{E} \langle F_u(x; s, u; f, \phi), g(x + X_{s, u}) \rangle \, \mathrm{d}x.$$

Then by Hölder's inequality and (3.4.4) one has

$$\int_{\mathbb{R}^d} \mathbb{E} |\langle F_u(x; s, u; f, \phi), g(x + X_{s,u}) \rangle| \, \mathrm{d}x \le \beta_{p,X} ||f||_{L^p(\mathbb{R}^d; X)} ||g||_{L^{p'}(\mathbb{R}^d; X^*)}.$$

By Theorem 1.3.10 and Theorem 1.3.21 in [79],  $(L^{p'}(\mathbb{R}^d; X^*))^* = L^p(\mathbb{R}^d; X)$ , so there exists  $h \in L^p(\mathbb{R}^d; X)$  such that for each  $g \in L^{p'}(\mathbb{R}^d; X^*)$ 

$$\int_{\mathbb{R}^d} \mathbb{E} \langle F_u(x; s, u; f, \phi), g(x + X_{s,u}) \rangle \, \mathrm{d}x = \int_{\mathbb{R}^d} \langle h(x), g(x) \rangle \, \mathrm{d}x,$$

and

$$\|h\|_{L^{p}(\mathbb{R}^{d};X)} \le \beta_{p,X} \|f\|_{L^{p}(\mathbb{R}^{d};X)}.$$
(3.4.5)

In particular, since *X* is finite dimensional

$$\int_{\mathbb{R}^d} \mathbb{E}F_u(x; s, u; f, \phi) g(x + X_{s, u}) \, \mathrm{d}x = \int_{\mathbb{R}^d} h(x) g(x) \, \mathrm{d}x, \quad g \in L^{p'}(\mathbb{R}^d).$$
(3.4.6)

For each s < 0 define  $m_s : \mathbb{R}^d \to \mathbb{C}$  as follows

$$m_{s}(\xi) = \begin{cases} \left(1 - e^{2|s|\Psi(\xi)}\right) \frac{1}{\Psi(\xi)} \int_{\mathbb{R}^{d}} (e^{i\xi \cdot z} - 1)\phi(z)\nu(dz), & \Psi(\xi) \neq 0, \\ 0, & \Psi(\xi) = 0. \end{cases}$$

Let u = 0. Then analogously to [10, (32)], by (3.4.6) one obtains

$$\mathscr{F}(h)(\xi) = m_s(\xi)\mathscr{F}(f)(\xi), \ \xi \in \mathbb{R}^d.$$

Let  $T_{m_s}$  be the Fourier multiplier on  $L^2(\mathbb{R}^d; X)$  with symbol  $m_s$  (that is bounded by 1). By (3.4.5) one obtains that  $T_{m_s}$  extends uniquely to a bounded operator on  $L^p(\mathbb{R}^d; X)$  with  $||T_{m_s}||_{\mathcal{L}(L^p(\mathbb{R}^d; X))} \leq \beta_{p,X}$ . Let  $T_m$  be the multiplier on  $L^2(\mathbb{R}^d; X)$  with the symbol m given by

$$m(\xi) = \begin{cases} \frac{1}{\Psi(\xi)} \int_{\mathbb{R}^d} (e^{i\xi \cdot z} - 1)\phi(z)v(\,\mathrm{d}z), & \Psi(\xi) \neq 0, \\ 0, & \Psi(\xi) = 0. \end{cases}$$

Note that *m* is a pointwise limit of  $m_s$  as  $s \to -\infty$ . Also note that  $T_{m_s}f \to T_mf$  in  $L^2(\mathbb{R}^d; X)$  as  $s \to -\infty$  for each  $f \in C_c(\mathbb{R}^d; X)$  by Plancherel's theorem. Therefore by Fatou's lemma one has that for each  $f \in C_c(\mathbb{R}^d; X)$  the following holds:

$$\|T_m f\|_{L^p(\mathbb{R}^d;X)} \le \lim_{s \to -\infty} \|T_{m_s} f\|_{L^p(\mathbb{R}^d;X)} \le \beta_{p,X} \|f\|_{L^p(\mathbb{R}^d;X)},$$

hence  $T_m$  uniquely extends to a bounded operator on  $L^p(\mathbb{R}^d; X)$  with

$$\|T_m\|_{\mathscr{L}(L^p(\mathbb{R}^d;X))} \leq \beta_{p,X}.$$

### 3.4.2. Examples of Theorem 3.4.1

In this subsection *X* is a UMD Banach space,  $p \in (1,\infty)$ . The examples will be mainly the same as were given in [9, Chapter 4] with some author's remarks. Recall that we set  $\frac{a}{0} = 0$  for any  $a \in \mathbb{C}$ .

**Example 3.4.7.** Let  $V_1, V_2$  be two nonnegative Lévy measures on  $\mathbb{R}^d$  such that  $V_1 \leq V_2$ . Let

$$m(\xi) = \frac{\int_{\mathbb{R}^d} (1 - \cos(\xi \cdot z)) V_1(\,\mathrm{d}z)}{\int_{\mathbb{R}^d} (1 - \cos(\xi \cdot z)) V_2(\,\mathrm{d}z)}, \quad \xi \in \mathbb{R}^d.$$

Then  $||T_m||_{\mathcal{L}(L^p(\mathbb{R}^d;X))} \leq \beta_{p,X}$ .

**Example 3.4.8.** Let  $\mu_1, \mu_2$  be two nonnegative measures on  $S^{d-1}$  such that  $\mu_1 \le \mu_2$ . Let

$$m(\xi) = \frac{\int_{S^{d-1}} (\xi \cdot \theta)^2 \mu_1(\mathrm{d}\theta)}{\int_{S^{d-1}} (\xi \cdot \theta)^2 \mu_2(\mathrm{d}\theta)}, \quad \xi \in \mathbb{R}^d.$$

Then  $||T_m||_{\mathscr{L}(L^p(\mathbb{R}^d;X))} \leq \beta_{p,X}$ .

**Example 3.4.9** (Beurling-Ahlfors transform). Let d = 2. Put  $\mathbb{R}^2 = \mathbb{C}$ . Then the Fourier multiplier  $T_m$  with a symbol  $m(z) = \frac{\overline{z}^2}{|z|^2}$ ,  $z \in \mathbb{C}$ , has the norm at most  $2\beta_{p,X}$  on  $L^p(\mathbb{R}^d; X)$ . This multiplier is also known as the *Beurling-Ahlfors transform*. It is well-known that  $||T_m||_{\mathscr{L}(L^p(\mathbb{R}^2;X))} \ge \beta_{p,X}$ . There is quite an old problem whether  $||T_m||_{\mathscr{L}(L^p(\mathbb{R}^2;X))} = \beta_{p,X}$ . This question was firstly posed by Iwaniec in [82] in  $\mathbb{C}$ . Nevertheless it was neither proved nor disproved even in the scalar-valued case. We refer the reader to [8] and [79] for further details.

**Example 3.4.10.** Let  $\alpha \in (0,2)$ ,  $\mu$  be a finite positive measure on  $S^{d-1}$ ,  $\psi$  be a measurable function on  $S^{d-1}$  such that  $|\psi| \le 1$ . Let

$$m(\xi) = \frac{\int_{S^{d-1}} |(\xi \cdot \theta)|^{\alpha} \psi(\theta) \mu(\mathrm{d}\theta)}{\int_{S^{d-1}} |(\xi \cdot \theta)|^{\alpha} \mu(\mathrm{d}\theta)}, \ \xi \in \mathbb{R}^{d}.$$

Then analogously to [9, (4.7)],  $||T_m||_{\mathscr{L}(L^p(\mathbb{R}^d;X))} \leq \beta_{p,X}$ .

**Example 3.4.11** (Double Riesz transform). Let  $\alpha \in (0, 2]$ . Let

$$m(\xi) = \frac{|\xi_1|^{\alpha}}{|\xi_1|^{\alpha} + \dots + |\xi_d|^{\alpha}}, \quad \xi = (\xi_1, \dots, \xi_d) \in \mathbb{R}^d,$$

Then according to Example 3.4.10,  $||T_m||_{\mathcal{L}(L^p(\mathbb{R}^d;X))} \leq \beta_{p,X}$ . Note that if  $\alpha = 2$ , then  $T_m$  is a double Riesz transform. (In the paper [188] it is shown that the norm  $||T_m||_{\mathcal{L}(L^p(\mathbb{R}^d;X))}$  does not depend on  $\alpha$  and equals the UMD<sup>{[0,1]</sup></sup> constant of X).

**Example 3.4.12.** Let  $\alpha \in [0, 2]$ ,  $d \ge 2$ . Let

$$m(\xi) = \frac{|\xi_1|^{\alpha} - |\xi_2|^{\alpha}}{|\xi_1|^{\alpha} + \dots + |\xi_d|^{\alpha}}, \quad \xi = (\xi_1, \dots, \xi_d) \in \mathbb{R}^d$$

Then by Example 3.4.10,  $||T_m||_{\mathscr{L}(L^p(\mathbb{R}^d;X))} \leq \beta_{p,X}$ . Moreover, if  $d = 2, \alpha \in [1,2]$ , then  $\max_{\xi \in \mathbb{R}^2} m(\xi) = 1$ ,  $\min_{\xi \in \mathbb{R}^2} m(\xi) = -1$  and  $m|_{S^1} \in W^{2,1}(S^1)$ . Therefore due to Proposition 3.4, Proposition 3.8 and Remark 3.9 in [66] one has  $||T_m||_{\mathscr{L}(L^p(\mathbb{R}^2;X))} \geq \beta_{p,X}$ . This together with Theorem 3.4.1 implies  $||T_m||_{\mathscr{L}(L^p(\mathbb{R}^2;X))} = \beta_{p,X}$ , which extends [66, Theorem 1.1], where the same assertion was proven for  $\alpha = 2$ .

**Example 3.4.13.** Let  $\mu$  be a nonnegative Borel measure on  $S^{d-1}$ ,  $\psi \in L^{\infty}(S^{d-1}, \mu)$ ,  $\|\psi\|_{\infty} \leq 1$ . Let

$$m(\xi) = \frac{\int_{S^{d-1}} \ln(1 + (\xi \cdot \theta)^{-2}) \psi(\theta) \mu(\mathrm{d}\theta)}{\int_{S^{d-1}} \ln(1 + (\xi \cdot \theta)^{-2}) \mu(\mathrm{d}\theta)}, \quad \xi \in \mathbb{R}^d$$

Then  $||T_m||_{\mathscr{L}(L^p(\mathbb{R}^d;X))} \leq \beta_{p,X}$ .

# **3.5.** HILBERT TRANSFORM AND GENERAL CONJECTURE

In this section we assume that *X* is a finite dimensional Banach space to avoid difficulties with stochastic integration. Many of the assertions below can be extended to the general UMD Banach space case by using the same techniques as in the proof of Theorem 3.3.3.

### 3.5.1. Hilbert transform and Burkholder functions

It turns out that the generalization of Theorem 3.3.17 to the case of continuous martingales is connected with the boundedness of the Hilbert transform. The Fourier multiplier  $\mathcal{H} \in \mathcal{L}(L^2(\mathbb{R}))$  with the symbol  $m \in L^{\infty}(\mathbb{R})$  such that  $m(t) = -i \operatorname{sign}(t)$ ,  $t \in \mathbb{R}$ , is called the *Hilbert transform*. This operator can be extended to a bounded operator on  $L^p(\mathbb{R})$ , 1 (see [158] and [79, Chapter 5.1] for the details).

Let *X* be a Banach space. Then one can extend the Hilbert transform  $\mathscr{H}$  to  $\mathscr{S}(\mathbb{R}) \otimes X$  in the same way as it was done in (3.4.1). Denote this extension by  $\mathscr{H}_X$ . By [23, Lemma 2] and [61, Theorem 3] the following holds true:

**Theorem 3.5.1** (Bourgain, Burkholder). Let *X* be a Banach space. Then *X* is a UMD Banach space if and only if  $\mathcal{H}_X$  can be extended to a bounded operator on  $L^p(\mathbb{R}; X)$  for each 1 . Moreover, then

$$\sqrt{\beta_{p,X}} \le \|\mathcal{H}_X\|_{\mathscr{L}(L^p(\mathbb{R};X))} \le \beta_{p,X}^2.$$
(3.5.1)

The proof of the right-hand side of (3.5.1) is based on the following result.

**Proposition 3.5.2.** Let X be a finite dimensional Banach space,  $B_1$ ,  $B_2$  be two real-valued Wiener processes,  $f_1, f_2 : \mathbb{R}_+ \times \Omega \to X$  be two stochastically integrable functions. Let us define  $M := f_1 \cdot B_1 + f_2 \cdot B_2$ ,  $N := f_2 \cdot B_1 - f_1 \cdot B_2$ . Then for each  $T \ge 0$ 

$$(\mathbb{E} \|N_T\|^p)^{\frac{1}{p}} \le \beta_{p,X}^2 (\mathbb{E} \|M_T\|^p)^{\frac{1}{p}}.$$

*Proof.* The theorem follows from Theorem 4.4.2. Nevertheless we wish to illustrate an easier and more specific proof. Let  $\tilde{B}_1, \tilde{B}_2 : \tilde{\Omega} \times \mathbb{R}_+ \to \mathbb{R}$  be two Wiener process defined on an enlarged probability space  $(\tilde{\Omega}, \widetilde{\mathscr{F}}, \mathbb{P})$  with an enlarged filtration  $\mathbb{F} = (\widetilde{\mathscr{F}}_t)_{t\geq 0}$  such that  $\tilde{B}_1$  and  $\tilde{B}_2$  are independent of  $\mathscr{F}$ . Then by applying the decoupling theorem [79, Theorem 4.4.1] twice (see also [119]) and the fact that  $-\tilde{B}_1$  is a Wiener process

$$\begin{split} \mathbb{E} \|N_{T}\|^{p} &= \mathbb{E} \|(f_{2} \cdot B_{1})_{T} - (f_{1} \cdot B_{2})_{T}\|^{p} \leq \beta_{p,X}^{p} \mathbb{E} \|(f_{2} \cdot \widetilde{B}_{1})_{T} - (f_{1} \cdot \widetilde{B}_{2})_{T}\|^{p} \\ &= \beta_{p,X}^{p} \mathbb{E} \|(f_{1} \cdot (-\widetilde{B}_{2}))_{T} + (f_{2} \cdot \widetilde{B}_{1})_{T}\|^{p} \\ &\leq \beta_{p,X}^{2p} \mathbb{E} \|(f_{1} \cdot B_{1})_{T} + (f_{2} \cdot B_{2})_{T}\|^{p} \\ &= \beta_{p,X}^{2p} \mathbb{E} \|M_{T}\|^{p}. \end{split}$$

Let  $p \in (1,\infty)$ . A natural question is whether there exists a constant  $C_p > 0$  such that

$$\|\mathscr{H}_X\|_{\mathscr{L}(L^p(\mathbb{R};X))} \le C_p \beta_{p,X}.$$
(3.5.2)

Then the following theorem is applicable.

**Theorem 3.5.3.** Let X be a Banach space,  $p \in (1,\infty)$ . Then there exists  $C_p \ge 1$  such that (3.5.2) holds if there exists **some** Burkholder function  $U: X \times X \to \mathbb{R}$  such that U is continuous and a.s. twice Fréchet differentiable,  $U(x, y) \ge ||y||^p - (C_p \beta_{p,X})^p ||x||^p$  for any  $x, y \in X$ ,  $U(\alpha x, \alpha y) = |\alpha|^p U(x, y)$  for any  $\alpha \in \mathbb{R}$  and  $x, y \in X$ , and the function

$$t \mapsto U(x + tz_1, y + tz_2) + U(x + tz_2, y - tz_1), t \in \mathbb{R},$$

or, equivalently,

$$t \mapsto U(x + tz_1, y + tz_2) + U(x - tz_2, y + tz_1), \ t \in \mathbb{R},$$

*is concave for each*  $x, y, z_1, z_2 \in X$  *at* t = 0.

*Proof.* Let M and N be as in Proposition 3.5.2. By the approximation argument we can suppose that M and N have absolutely continuous distributions. Let d be the dimension of X. Then by the Itô formula in Theorem 2.12.1

$$\mathbb{E} \|N_t\|_X^p - (C_p \beta_{p,X})^p \mathbb{E} \|M_t\|_X^p \le \mathbb{E} U(M_t, N_t) = \mathbb{E} U(M_0, N_0) + \mathbb{E} \int_0^t \langle \partial_X U(M_s, N_s), dM_s \rangle + \mathbb{E} \int_0^t \langle \partial_Y U(M_s, N_s), dN_s \rangle + \frac{1}{2} \mathbb{E} I,$$
(3.5.3)

where

$$I = \int_{0}^{t} \sum_{i,j=1}^{d} (U_{x_{i},x_{j}}(M_{s},N_{s}) d[\langle x_{i}^{*},M_{s} \rangle, \langle x_{j}^{*},M_{s} \rangle] + 2U_{x_{i},y_{j}}(M_{s},N_{s}) d[\langle x_{i}^{*},M_{s} \rangle, \langle y_{j}^{*},N_{s} \rangle] + U_{y_{i},y_{i}}(M_{s},N_{s}) d[\langle y_{i}^{*},N_{s} \rangle, \langle y_{j}^{*},N_{s} \rangle]), \quad (3.5.4)$$

where  $(x_i)_{i=1}^d = (y_i)_{i=1}^d \subset X$  is the same basis of *X*, and  $(x_i^*)_{i=1}^d = (y_i^*)_{i=1}^d \subset X^*$  are the same corresponding dual bases of  $X^*$ .

Notice that by Remark 3.3.9  $\mathbb{E}U(M_0, N_0) \le 0$  since  $||N_0|| \le ||M_0||$  a.s. and  $C_p, \beta_{p,X} \ge 1$ , and that

$$\mathbb{E}\left(\int_0^t \langle \partial_x U(M_s, N_s), \, \mathrm{d}M_s \rangle + \int_0^t \langle \partial_y U(M_s, N_s), \, \mathrm{d}N_s \rangle\right) = 0,$$

since due to the same type of discussion as was done in the proof of Theorem 3.3.17,  $\int_0^{\cdot} \langle \partial_x U(M_s, N_s), dM_s \rangle + \int_0^{\cdot} \langle \partial_y U(M_s, N_s), dN_s \rangle$  is a martingale which starts at zero.

Let us now prove that  $I \leq 0$ . For each i = 1, 2, ..., d we define  $f_i^1 := \langle x_i^*, f_1 \rangle$  and  $f_i^2 := \langle x_i^*, f_2 \rangle$ . Then for each i, j = 1, 2, ..., d one has that

$$d[\langle x_i^*, M_s \rangle, \langle x_j^*, M_s \rangle] = d[\langle y_i^*, N_s \rangle, \langle y_j^*, N_s \rangle] = (f_i^1 f_j^1 + f_i^2 f_j^2) dt, \qquad (3.5.5)$$

and

$$d[\langle x_i^*, M_s \rangle, \langle y_j^*, N_s \rangle] = (f_i^1 f_j^2 - f_i^2 f_j^1) dt.$$
(3.5.6)

Notice also that for each  $x, y \in X$ 

$$\begin{aligned} \frac{\partial^2}{\partial u^2} U(x+uf_1,y+uf_2)|_{u=0} &= \sum_{i,j=1}^d \left( (U_{x_i^*,x_j^*}(x,y)f_i^1f_j^1 + 2U_{x_i^*,y_j^*}(x,y)f_i^1f_j^2 + U_{y_i^*,y_j^*}(x,y)f_i^2f_j^2 \right), \\ &+ U_{y_i^*,y_j^*}(x,y)f_i^2f_j^2), \\ \frac{\partial^2}{\partial u^2} U(x+uf_2,y-uf_1)|_{u=0} &= \frac{\partial^2}{\partial u^2} U(x-uf_2,y+uf_1)|_{u=0} \\ &= \sum_{i,j=1}^d \left( (U_{x_i^*,x_j^*}(x,y)f_i^2f_j^2 - 2U_{x_i^*,y_j^*}(x,y)f_i^2f_j^1 + U_{y_i^*,y_j^*}(x,y)f_i^1f_j^1 \right). \end{aligned}$$
(3.5.7)

Therefore by (3.5.4), (3.5.5), (3.5.6), and (3.5.7) we have that

$$\begin{split} I &= \int_{0}^{t} \sum_{i,j=1}^{d} ((U_{x_{i}^{*},x_{j}^{*}}(M_{s-},N_{s-})(f_{i}^{1}f_{j}^{1}+f_{i}^{2}f_{j}^{2})+2U_{x_{i}^{*},y_{j}^{*}}(M_{s-},N_{s-})(f_{i}^{1}f_{j}^{2}-f_{i}^{2}f_{j}^{1}) \\ &+ U_{y_{i}^{*},y_{j}^{*}}(M_{s-},N_{s-})(f_{i}^{1}f_{j}^{1}+f_{i}^{2}f_{j}^{2})) \,\mathrm{d}t = \int_{0}^{t} \frac{\partial^{2}}{\partial u^{2}} U(M_{s-}+uf_{1},N_{s-}+uf_{2})|_{u=0} \\ &+ \frac{\partial^{2}}{\partial u^{2}} U(M_{s-}+uf_{2},N_{s-}-uf_{1})|_{u=0} \,\mathrm{d}s \\ &= \int_{0}^{t} \frac{\partial^{2}}{\partial u^{2}} \Big( U(M_{s-}+uf_{1},N_{s-}+uf_{2}) + U(M_{s-}+uf_{2},N_{s-}-uf_{1}) \Big) \Big|_{u=0} \,\mathrm{d}s, \end{split}$$

and thanks to the concavity of  $U(x + uf_1, y + uf_2) + U(x + uf_2, y - uf_1)$  in point u = 0 for each  $x, y \in X$  one deduces that a.s.  $I \le 0$ . Then thanks to (3.5.3) one has that

$$\mathbb{E}\|N_t\|_X^p - (C_p\beta_{p,X})^p \mathbb{E}\|M_t\|_X^p \le \mathbb{E}U(M_t, N_t) \le 0.$$
(3.5.8)

Now one can prove that (3.5.8) implies (3.5.2) in the same way as it was done for instance in [61, Theorem 3], [15, p.592] or [39, Chapter 3].

*Remark* 3.5.4. Note that if *X* is a finite dimensional Hilbert space, then one gets condition (iii) in Theorem 3.5.3 for free from [179]. Indeed, let  $U: X \times X \rightarrow \mathbb{R}$  be as in [179, p. 527], namely

$$U(x, y) = p(1 - 1/p^*)^{p-1} (||y|| - (p^* - 1)||x||) (||x|| + ||y||)^{p-1}, x, y \in X.$$

Then *U* is a.s. twice Fréchet differentiable, and thanks to the property (c) of *U*, which is given on [179, p.527], for all nonzero  $x, y \in X$  there exists a constant  $c(x, y) \ge 0$  such that

$$\begin{aligned} \langle \partial_{xx} U(x, y), (h, h) \rangle + 2 \langle \partial_{xy} U(x, y), (h, k) \rangle + \langle \partial_{yy} U(x, y), (k, k) \rangle \\ \leq -c(x, y) (\|h\|^2 - \|k\|^2), \quad h, k \in X. \end{aligned}$$

Therefore for any  $z_1, z_2 \in X$ 

$$\frac{\partial^2}{\partial t^2} \left[ U(x+tz_1, y+tz_2) + U(x+tz_2, y-tz_1) \right] \Big|_{t=0}$$

$$= \langle \partial_{xx} U(x, y), (z_1, z_1) \rangle + 2 \langle \partial_{xy} U(x, y), (z_1, z_2) \rangle + \langle \partial_{yy} U(x, y), (z_2, z_2) \rangle + \langle \partial_{xx} U(x, y), (z_2, z_2) \rangle - 2 \langle \partial_{xy} U(x, y), (z_2, z_1) \rangle + \langle \partial_{yy} U(x, y), (z_1, z_1) \rangle \le -c(x, y) (\|z_1\|^2 - \|z_2\|^2) - c(x, y) (\|z_2\|^2 - \|z_1\|^2) = 0.$$

#### 3.5.2. General conjecture

By Theorem 3.5.3 the estimate (3.5.2) is a direct corollary of the following conjecture.

**Conjecture 3.5.5.** Let X be a finite dimensional Banach space,  $p \in (1, \infty)$ . Then there exists  $C_p \ge 1$  such that for each pair of continuous martingales  $M, N : \mathbb{R}_+ \times \Omega \to X$  such that N is weakly differentially subordinate to M one has that for each  $t \ge 0$ 

$$(\mathbb{E}\|N_t\|^p)^{\frac{1}{p}} \le C_p \beta_{p,X} (\mathbb{E}\|M_t\|^p)^{\frac{1}{p}}.$$
(3.5.9)

*Remark* 3.5.6. Notice that the estimate (3.5.9) follows from Theorem 4.4.2 with the constant  $\beta_{p,X}^2$  instead of  $C_p\beta_{p,X}$ . Moreover, it is shown in Theorem 4.4.2 that  $C_p$  can not be less then 1.

We wish to finish by pointing out some particular cases in which Conjecture 3.5.5 holds. These results are about stochastic integration with respect to a Wiener process. Recall that we assume that *X* is a finite dimensional space. Later we will need a couple of definitions.

Let  $W^H : \mathbb{R}_+ \times H \to L^2(\Omega)$  be an *H*-cylindrical Brownian motion, i.e.

- $(W^H h_1, ..., W^H h_d) : \mathbb{R}_+ \times \Omega \to \mathbb{R}^d$  is a *d*-dimensional Wiener process for all  $d \ge 1$ and  $h_1, ..., h_d \in H$ ,
- $\mathbb{E}W^H(t)hW^H(s)g = \langle h, g \rangle \min\{t, s\} \forall h, g \in H, t, s \ge 0.$

(We refer the reader to [48, Chapter 4.1] for further details). Let *X* be a Banach space,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X)$  be elementary progressive of the form (2.5.1). Then we define a *stochastic integral*  $\Phi \cdot W^H : \mathbb{R}_+ \times \Omega \to X$  of  $\Phi$  with respect to  $W^H$  in the following way:

$$(\Phi \cdot W^{H})_{t} = \sum_{k=1}^{K} \sum_{m=1}^{M} \mathbf{1}_{B_{mk}} \sum_{n=1}^{N} (W^{H}(t_{k} \wedge t)h_{n} - W^{H}(t_{k-1} \wedge t)h_{n})x_{kmn}, \ t \ge 0.$$

The following lemma is a multidimensional variant of [93, (3.2.19)] and it is closely connected with Lemma 3.2.1.

**Lemma 3.5.7.** Let  $X = \mathbb{R}$ ,  $\Phi, \Psi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(H, \mathbb{R})$  be elementary progressive. Then for all  $t \ge 0$  a.s.

$$[\Phi \cdot W^H, \Psi \cdot W^H]_t = \int_0^t \langle \Phi^*(s), \Psi^*(s) \rangle \, \mathrm{d}s.$$

The reader can find more on stochastic integration with respect to an *H*-cylindrical Brownian motion in the UMD case in [126]. The following theorem follows from (6.4.26).

**Theorem 3.5.8.** Let X be a finite dimensional Banach space,  $W^H$  be an H-cylindrical Brownian motion,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X)$  be stochastically integrable with respect to  $W^H$  function. Let  $A \in \mathcal{L}(H)$  be self-adjoint. Then

$$(\mathbb{E}\|((\Phi A) \cdot W^{H})_{\infty}\|_{X}^{p})^{\frac{1}{p}} \le \beta_{p,X} \|A\| (\mathbb{E}\|(\Phi \cdot W^{H})_{\infty}\|_{X}^{p})^{\frac{1}{p}}.$$
(3.5.10)

Notice that by Lemma 3.5.7 for each  $x^* \in X^*$  and  $0 \le s < t < \infty$  a.s.

$$\begin{split} [\langle (\Phi A) \cdot W^{H}, x^{*} \rangle]_{t} &- [\langle (\Phi A) \cdot W^{H}, x^{*} \rangle]_{s} = \int_{s}^{t} \|A\Phi^{*}(r)x^{*}\|^{2} dr \\ &\leq \|A\|^{2} \int_{s}^{t} \|\Phi^{*}(r)x^{*}\|^{2} dr \\ &= \|A\|^{2} \big( [\langle \Phi \cdot W^{H}, x^{*} \rangle]_{t} - [\langle \Phi \cdot W^{H}, x^{*} \rangle]_{s} \big). \end{split}$$

Hence if  $||A|| \le 1$ , then  $(\Phi A) \cdot W^H$  is weakly differentially subordinate to  $\Phi \cdot W^H$ , and therefore Theorem 3.5.8 provides us with a special case of Conjecture 3.5.5.

Remark 3.5.9. Theorem 3.5.8 in fact can be shown using [66, Proposition 3.7.(i)].

*Remark* 3.5.10. An analogue of Theorem 3.5.8 for antisymmetric *A* (i.e. *A* such that  $A^* = -A$ ) remains open. It is important for instance for the possible estimate (3.5.2). Indeed, in Proposition 3.5.2 the Hilbert space *H* can be taken 2-dimensional,  $A = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$ , and  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X)$  is such that  $\Phi\begin{pmatrix} a \\ b \end{pmatrix} = af_1 + bf_2$  for each  $a, b \in \mathbb{R}$ . Then  $M = \Phi \cdot W^H$ ,  $N = (\Phi A) \cdot W^H$ , and if one shows (3.5.10) for an antisymmetric operator *A*, then one automatically gains (3.5.2).

The next theorem shows that Conjecture 3.5.5 holds for stochastic integrals with respect to a one-dimensional Wiener process.

**Theorem 3.5.11.** Let X be a finite dimensional Banach space,  $W : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a onedimensional Wiener process,  $\Phi, \Psi : \mathbb{R}_+ \times \Omega \to X$  be stochastically integrable with respect to  $W, M = \Phi \cdot W, N = \Psi \cdot W$ . Let N be weakly differentially subordinate to M. Then for each  $p \in (1, \infty)$ ,

$$\mathbb{E}\|N_{\infty}\|^{p} \le \beta_{p,X}^{p} \mathbb{E}\|M_{\infty}\|^{p}.$$
(3.5.11)

*Proof.* Without loss of generality suppose that there exists  $T \ge 0$  such that  $\Phi \mathbf{1}_{[T,\infty]} = \Psi \mathbf{1}_{[T,\infty]} = 0$ . Since *N* is weakly differentially subordinate to *M*, by the Itô isomorphism for each  $x^* \in X^*$ ,  $0 \le s < t < \infty$  we have a.s.

$$\begin{split} [\langle x^*, N \rangle]_t - [\langle x^*, N \rangle]_s &= \int_s^t |\langle x^*, \Psi(r) \rangle|^2 \, \mathrm{d}r \\ &\leq \int_s^t |\langle x^*, \Phi(r) \rangle|^2 \, \mathrm{d}r = [\langle x^*, M \rangle]_t - [\langle x^*, M \rangle]_s. \end{split}$$

Therefore we can deduce that  $|\langle x^*, \Psi \rangle| \le |\langle x^*, \Phi \rangle|$  a.s. on  $\mathbb{R}_+ \times \Omega$ . By Lemma 3.3.6 there exists progressively measurable  $a : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  such that  $|a| \le 1$  on  $\mathbb{R}_+ \times \Omega$  and  $\Psi = a\Phi$  a.s. on  $\mathbb{R}_+ \times \Omega$ . Now for each  $n \ge 1$  set  $a_n : \mathbb{R}_+ \times \Omega \to \mathbb{R}$ ,  $\Phi_n : \mathbb{R}_+ \times \Omega \to X$  be elementary progressively measurable such that  $|a_n| \le 1$ ,  $a_n \to a$  a.s. on  $\mathbb{R}_+ \times \Omega$  and  $\mathbb{E} \int_0^T ||\Phi(t) - \Phi_n(t)||^2 dt \to 0$  as  $n \to \infty$ . Then by the triangle inequality

$$\left( \mathbb{E} \int_{0}^{T} \|\Psi(t) - a_{n}(t)\Phi_{n}(t)\|^{2} dt \right)^{\frac{1}{2}} \leq \left( \mathbb{E} \int_{0}^{T} \|\Phi(t)\|^{2} (a(t) - a_{n}(t))^{2} dt \right)^{\frac{1}{2}} + \left( \mathbb{E} \int_{0}^{T} \|\Phi(t) - \Phi_{n}(t)\|^{2} a_{n}^{2} dt \right)^{\frac{1}{2}},$$

$$(3.5.12)$$

which vanishes as  $n \to \infty$  by the dominated convergence theorem. For each  $n \ge 1$  the inequality

$$(\mathbb{E}\|((a_n\Phi_n)\cdot W)_{\infty}\|^p)^{\frac{1}{p}} \leq \beta_{p,X}(\mathbb{E}\|(\Phi_n\cdot W)_{\infty}\|^p)^{\frac{1}{p}}$$

holds thanks to the martingale transform theorem [79, Theorem 4.2.25]. Then (3.5.11) follows from the previous estimate and (3.5.12) when one lets n go to infinity.

*Remark* 3.5.12. Let *W* be a one-dimensional Wiener process,  $\mathbb{F}$  be a filtration which is generated by *W*. Let  $M, N : \mathbb{R}_+ \times \Omega \to X$  be  $\mathbb{F}$ -martingales such that  $M_0 = N_0 = 0$  and *N* is weakly differentially subordinate to *M*. Then thanks to the Itô isomorphism [126, Theorem 3.5] there exist progressively measurable  $\Phi, \Psi : \mathbb{R}_+ \times \Omega \to X$  such that  $M = \Phi \cdot W$  and  $N = \Psi \cdot W$ , and thanks to Theorem 3.5.11

$$\mathbb{E} \|N_{\infty}\|^{p} \leq \beta_{p,X}^{p} \mathbb{E} \|M_{\infty}\|^{p}, \quad p \in (1,\infty).$$

This shows that on certain probability spaces the estimate (3.5.9) automatically holds with a constant  $C_p = 1$ .

# 4

# *L<sup>p</sup>*-ESTIMATES FOR WEAK DIFFERENTIAL SUB-ORDINATION AND FOR MARTINGALE DECOM-POSITIONS

This chapter is based on the paper *Martingale decompositions and weak differential subordination in UMD Banach spaces* by Ivan Yaroslavtsev, see [184].

In this chapter we consider Meyer-Yoeurp decompositions for UMD Banach space-valued martingales. Namely, we prove that X is a UMD Banach space if and only if for any fixed  $p \in (1,\infty)$ , any X-valued  $L^p$ -bounded martingale M has a unique decomposition  $M = M^d + M^c$  such that  $M^d$  is a purely discontinuous martingale,  $M^c$  is a continuous martingale,  $M^c_0 = 0$  and

 $\mathbb{E} \|M_{\infty}^{d}\|^{p} + \mathbb{E} \|M_{\infty}^{c}\|^{p} \leq c_{p,X} \mathbb{E} \|M_{\infty}\|^{p}.$ 

An analogous assertion is shown for the Yoeurp decomposition of a purely discontinuous martingales into a sum of a quasi-left continuous martingale and a martingale with accessible jumps.

As an application we show that X is a UMD Banach space if and only if for any fixed  $p \in (1,\infty)$  and for all X-valued martingales M and N such that N is weakly differentially subordinate to M, one has the estimate

 $\mathbb{E}\|N_{\infty}\|^{p} \leq C_{p,X}\mathbb{E}\|M_{\infty}\|^{p}.$ 

<sup>2010</sup> Mathematics Subject Classification. 60G44 Secondary: 60B11, 60G46.

*Key words and phrases.* Differential subordination, weak differential subordination, UMD Banach spaces, Burkholder function, stochastic integration, Brownian representation, Meyer-Yoeurp decomposition, Yoeurp decomposition, purely discontinuous martingales, continuous martingales, quasi-left continuous, accessible jumps, canonical decomposition of martingales.

# **4.1.** INTRODUCTION

It is well-known from the fundamental paper of Itô [81] on the real-valued case, and several works [1, 5, 17, 50, 157] on the vector-valued case, that for any Banach space *X*, any centered *X*-valued Lévy process has a unique decomposition  $L = W + \tilde{N}$ , where *W* is an *X*-valued Wiener process, and  $\tilde{N}$  is an *X*-valued weak integral with respect to a certain compensated Poisson random measure. Moreover, *W* and  $\tilde{N}$  are independent, and therefore since *W* is symmetric, for each  $1 and <math>t \ge 0$ ,

$$\mathbb{E}\|\widetilde{N}_t\|^p \le \mathbb{E}\|L_t\|^p. \tag{4.1.1}$$

The natural generalization of this result to general martingales in the realvalued setting was provided by Meyer in [122] and Yoeurp in [190]. Namely, it was shown that any real-valued martingale M can be uniquely decomposed into a sum of two martingales  $M^d$  and  $M^c$  such that  $M^d$  is purely discontinuous (i.e. the quadratic variation  $[M^d]$  has a pure jump version), and  $M^c$  is continuous with  $M_0^c = 0$ . The reason why they needed such a decomposition is a further decomposition of a semimartingale, and finding an exponent of a semimartingale (we refer the reader to [89] and [190] for the details on this approach). In the present article we extend Meyer-Yoeurp theorem to the vector-valued setting, and provide extension of (4.1.1) for a general martingale (see Subsection 4.3.1). Namely, we prove that for any UMD Banach space X and any 1 , an <math>X-valued  $L^p$ -bounded martingale M can be uniquely decomposed into a sum of two martingales  $M^d$  and  $M^c$  such that  $M^d$  is purely discontinuous (i.e.  $\langle M^d, x^* \rangle$  is purely discontinuous for each  $x^* \in X^*$ ), and  $M^c$  is continuous with  $M_0^c = 0$ . Moreover, then for each  $t \ge 0$ ,

$$(\mathbb{E}\|M_t^d\|^p)^{\frac{1}{p}} \le \beta_{p,X}(\mathbb{E}\|M_t\|^p)^{\frac{1}{p}}, \quad (\mathbb{E}\|M_t^c\|^p)^{\frac{1}{p}} \le \beta_{p,X}(\mathbb{E}\|M_t\|^p)^{\frac{1}{p}}, \quad (4.1.2)$$

where  $\beta_{p,X}$  is the UMD<sub>*p*</sub> constant of *X* (see Section 2.3). Theorem 4.3.13 shows that such a decomposition together with  $L^p$ -estimates of type (4.1.2) is possible if and only if *X* has the UMD property.

The purely discontinuous part can be further decomposed: in [190] Yoeurp proved that any real-valued purely discontinuous  $M^d$  can be uniquely decomposed into a sum of a purely discontinuous quasi-left continuous martingale  $M^q$  (analogous to the "compensated Poisson part", which does not jump at predictable stopping times), and a purely discontinuous martingale with accessible jumps  $M^a$  (analogous to the "discrete part", which jumps only at certain predictable stopping times). In Subsection 4.3.2 we extend this result to a UMD space-valued setting with appropriate estimates. Namely, we prove that for each 1 the same type of decomposition is possible and unique for an*X* $-valued purely discontinuous <math>L^p$ -bounded martingale  $M^d$ , and then for each  $t \ge 0$ ,

$$(\mathbb{E}\|M_t^q\|^p)^{\frac{1}{p}} \le \beta_{p,X}(\mathbb{E}\|M_t^d\|^p)^{\frac{1}{p}}, \quad (\mathbb{E}\|M_t^a\|^p)^{\frac{1}{p}} \le \beta_{p,X}(\mathbb{E}\|M_t^d\|^p)^{\frac{1}{p}}.$$
(4.1.3)

Again as Theorem 4.3.13 shows, the (4.1.3)-type estimates are a possible only in UMD Banach spaces.

Even though the Meyer-Yoeurp and Yoeurp decompositions can be easily extended from the real-valued case to a Hilbert space case, the author could not find the corresponding estimates of type (4.1.2)-(4.1.3) in the literature, so we wish to present this special issue here. If *H* is a Hilbert space,  $M : \mathbb{R}_+ \times \Omega \rightarrow H$  is a martingale, then there exists a unique decomposition of *M* into a continuous part  $M^c$ , a purely discontinuous quasi-left continuous part  $M^q$ , and a purely discontinuous part  $M^a$  with accessible jumps. Moreover, then for each 1 , and for<math>i = c, q, a,

$$(\mathbb{E}\|M_t^i\|^p)^{\frac{1}{p}} \le (p^* - 1)(\mathbb{E}\|M_t\|^p)^{\frac{1}{p}},$$
(4.1.4)

where  $p^* = \max\{p, \frac{p}{p-1}\}$ . Notice that though (4.1.4) follows from (4.1.2)-(4.1.3) since  $\beta_{p,H} = p^* - 1$ , it can be easily derived from the differential subordination estimates for Hilbert space-valued martingales obtained by Wang in [179].

Both the Meyer-Yoeurp and Yoeurp decompositions play a significant rôle in stochastic integration: if  $M = M^c + M^q + M^a$  is a decomposition of an *H*-valued martingale *M* into continuous, purely discontinuous quasi-left continuous and purely discontinuous with accessible jumps parts, and if  $\Phi : \mathbb{R}_+ \times \Omega \rightarrow \mathcal{L}(H, X)$  is elementary predictable for some UMD Banach space *X*, then the decomposition  $\Phi \cdot M = \Phi \cdot M^c + \Phi \cdot M^q + \Phi \cdot M^a$  of a stochastic integral  $\Phi \cdot M$  is a decomposition of the martingale  $\Phi \cdot M$  into continuous, purely discontinuous quasi-left continuous and purely discontinuous with accessible jumps parts, and for any 1 we have that

$$\mathbb{E}\|(\Phi \cdot M)_{\infty}\|^{p} = \sum_{p,X} \mathbb{E}\|(\Phi \cdot M^{c})_{\infty}\|^{p} + \mathbb{E}\|(\Phi \cdot M^{q})_{\infty}\|^{p} + \mathbb{E}\|(\Phi \cdot M^{a})_{\infty}\|^{p}.$$

The corresponding Itô isomorphism for  $\Phi \cdot M^c$  for a general UMD Banach space *X* was derived by Veraar and the author in [177], while Itô isomorphisms for  $\Phi \cdot M^q$  and  $\Phi \cdot M^a$  are shown in Chapter 7 for the case  $X = L^r(S)$ ,  $1 < r < \infty$ .

The major underlying techniques involved in the proofs of (4.1.2) and (4.1.3) are rather different from the original methods of Meyer in [122] and Yoeurp in [190]. They include the results on the differentiability of the Burkholder function of any finite dimensional Banach space, which have been proven recently in [189] and which allow us to use Itô formula in order to show the desired inequalities in the same way as it was demonstrated by Wang in [179].

The main application of the Meyer-Yoeurp decomposition are  $L^p$ -estimates for weakly differentially subordinated martingales. The weak differential subordination property was introduced in Chapter 3, and can be described in the following way: an *X*-valued martingale *N* is weakly differentially subordinate to an *X*-valued martingale *M* if for each  $x^* \in X^*$  a.s.  $|\langle N_0, x^* \rangle| \le |\langle M_0, x^* \rangle|$  and for each  $t \ge s \ge 0$ 

$$[\langle N, x^* \rangle]_t - [\langle N, x^* \rangle]_s \le [\langle M, x^* \rangle]_t - [\langle M, x^* \rangle]_s.$$

If both *M* and *N* are purely discontinuous, and if *X* is a UMD Banach space, then by [189], for each  $1 we have that <math>\mathbb{E}||N_{\infty}||^p \leq \beta_{p,X}^p \mathbb{E}||M_{\infty}||^p$ . Section 4.4 is devoted to the generalization of this result to continuous and general martingales. There we show that if both *M* and *N* are continuous, then  $\mathbb{E}||N_{\infty}||^p \leq c_{p,X}^p \mathbb{E}||M_{\infty}||^p$ , where the least admissible  $c_{p,X}$  is within the interval  $[\beta_{p,X}, \beta_{p,X}^2]$ . Furthermore, using the Meyer-Yoeurp decomposition and estimates (4.1.2) we show that for general *X*-valued martingales *M* and *N* such that *N* is weakly differentially subordinate to *M* the following holds

$$(\mathbb{E}\|N_{\infty}\|^{p})^{\frac{1}{p}} \leq \beta_{p,X}^{2}(\beta_{p,X}+1)(\mathbb{E}\|M_{\infty}\|^{p})^{\frac{1}{p}}.$$

The weak differential subordination as a stronger version of the differential subordination is of interest in Harmonic Analysis. For instance, it was shown in [189] that sharp  $L^p$ -estimates for weakly differentially subordinated purely discontinuous martingales imply sharp estimates for the norms of a broad class of Fourier multipliers on  $L^p(\mathbb{R}^d; X)$ . Also there is a strong connection between the weak differential subordination of continuous martingales and the norm of the Hilbert transform on  $L^p(\mathbb{R}; X)$  (see [189] and Remark 4.4.4).

Alternative approaches to Fourier multipliers for functions with values in UMD spaces have been constructed from the differential subordination for purely discontinuous martingales (see Bañuelos and Bogdan [10], Bañuelos, Bogdan and Bielaszewski [9], and recent work [189]), and for continuous martingales (see Mc-Connell [118] and Geiss, Montgomery-Smith and Saksman [66]). It remains open whether one can combine these two approaches using the general weak differential subordination theory.

# **4.2. PRELIMINARIES**

We set the scalar field to be  $\mathbb{R}$ .

Let *X* be a finite dimensional Banach space. Then according to Theorem 2.20 and Proposition 2.21 in [59] there exists a unique translation-invariant measure  $\lambda_X$  on *X* such that  $\lambda_X(\mathbb{B}_X) = 1$  for the unit ball  $\mathbb{B}_X$  of *X*. We will call  $\lambda_X$  the *Lebesgue measure*.

## **4.3. UMD BANACH SPACES AND MARTINGALE DECOMPOSITIONS**

Let *X* be a Banach space, 1 . In this section we will show that the Meyer-Yoeurp and Yoeurp decompositions for*X* $-valued <math>L^p$ -bounded martingales take place if and only if *X* has the UMD property.

### 4.3.1. Meyer-Yoeurp decomposition in UMD case

This subsection is devoted to the generalization of Meyer-Yoeurp decomposition (see Subsection 2.2.3) to the UMD Banach space case:

**Theorem 4.3.1** (Meyer-Yoeurp decomposition). Let X be a UMD Banach space,  $p \in (1,\infty)$ ,  $M : \mathbb{R}_+ \times \Omega \to X$  be an  $L^p$ -bounded martingale. Then there exist unique martingales  $M^d, M^c : \mathbb{R}_+ \times \Omega \to X$  such that  $M^d$  is purely discontinuous,  $M^c$  is continuous,  $M_0^c = 0$  and  $M = M^d + M^c$ . Moreover, then for all  $t \ge 0$ 

$$(\mathbb{E}\|M_t^d\|^p)^{\frac{1}{p}} \le \beta_{p,X}(\mathbb{E}\|M_t\|^p)^{\frac{1}{p}}, \quad (\mathbb{E}\|M_t^c\|^p)^{\frac{1}{p}} \le \beta_{p,X}(\mathbb{E}\|M_t\|^p)^{\frac{1}{p}}.$$
(4.3.1)

The following proposition follows from Section 3.3.

**Proposition 4.3.2.** Let X be a finite dimensional Banach space,  $p \in (1,\infty)$ . Let  $Y = X \oplus \mathbb{R}$ be a Banach space such that  $\|(x,r)\|_Y = (\|x\|_X^p + |r|^p)^{\frac{1}{p}}$ . Then  $\beta_{p,Y} = \beta_{p,X}$ . Moreover, if  $M : \mathbb{R}_+ \times \Omega \to X$  is a martingale on a probability space  $(\Omega, \mathscr{F}, \mathbb{P})$  with a filtration  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$ , then there exists a sequence  $(M^m)_{m\geq 1}$  of Y-valued martingales on an enlarged probability space  $(\overline{\Omega}, \overline{\mathscr{F}}, \overline{\mathbb{P}})$  with an enlarged filtration  $\overline{\mathbb{F}} = (\overline{\mathscr{F}}_t)_{t\geq 0}$  such that

- 1.  $M_t^m$  has absolutely continuous distributions with respect to the Lebesgue measure on Y for each  $m \ge 1$  and  $t \ge 0$ ;
- 2.  $M_t^m \rightarrow (M_t, 0)$  pointwise as  $m \rightarrow \infty$  for each  $t \ge 0$ ;
- 3. *if for some*  $t \ge 0 \mathbb{E} ||M_t||^p < \infty$ , *then for each*  $m \ge 1$  *one has that*  $\mathbb{E} ||M_t^m||^p < \infty$  *and*  $\mathbb{E} ||M_t^m (M_t, 0)||^p \to 0$  *as*  $m \to \infty$ ;
- 4. if M is continuous, then  $(M^m)_{m\geq 1}$  are continuous as well,
- 5. *if M* is purely discontinuous, then  $(M^m)_{m\geq 1}$  are purely discontinuous as well.

*Proof.* The proof of (1)-(3) follows from Lemma 3.3.18, while (4) and (5) follow from the construction of  $M^m$  given in 3.3.18.

*Remark* 4.3.3. Notice that the construction in Section 3.3 also allows us to sum these approximations for different martingales. Namely, if *M* and *N* are two *X*-valued martingales, then we can construct the corresponding *Y*-valued martingales  $(M^m)_{m\geq 1}$  and  $(N^m)_{m\geq 1}$  as in Proposition 4.3.2 in such a way that  $M_t^m + N_t^m$  has an absolutely continuous distribution for each  $t \ge 0$  and  $m \ge 1$ .

*Proof of Theorem* 4.3.1. *Step* 1: *finite dimensional case.* Let *X* be finite dimensional. Then  $M^d$  and  $M^c$  exist due to Remark 2.2.14. Without loss of generality  $\mathscr{F}_t = \mathscr{F}_{\infty}$ ,  $M^d_t = M^d_{\infty}$  and  $M^c_t = M^c_{\infty}$ . Let *d* be the dimension of *X*.

Let  $\|\|\cdot\|\|$  be a Euclidean norm on *X*. Then  $(X, \|\|\cdot\|\|)$  is a Hilbert space, and by Remark 2.2.7 the quadratic variation  $[M^c]$  exists and has a continuous version. Let

us show that without loss of generality we can suppose that  $[M^c]$  is a.s. absolutely continuous with respect to the Lebesgue measure on  $\mathbb{R}_+$ . Let  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be as follows:  $A_t = [M^c]_t + t$ . Then A is strictly increasing continuous,  $A_0 = 0$  and  $A_{\infty} = \infty$  a.s. Let the time-change  $\tau = (\tau_s)_{s\geq 1}$  be defined as in Theorem 2.4.25. Then by Theorem 2.4.25,  $M^c \circ \tau$  is a continuous martingale,  $M^d \circ \tau$  is a purely discontinuous martingale,  $(M^c \circ \tau)_0 = 0$ ,  $(M^d \circ \tau)_0 = M_0^d$  and due to the Kazamaki theorem [89, Theorem 17.24],  $[M^c \circ \tau] = [M^c] \circ \tau$ . Therefore for all  $t > s \geq 0$  by Theorem 2.4.25 and the fact that  $\tau_t \geq \tau_s$  a.s.

$$[M^{c} \circ \tau]_{t} - [M^{c} \circ \tau]_{s} = [M^{c}]_{\tau_{t}} - [M^{c}]_{\tau_{s}} \le [M^{c}]_{\tau_{t}} - [M^{c}]_{\tau_{s}} + (\tau_{t} - \tau_{s})$$
$$= ([M^{c}]_{\tau_{t}} + \tau_{t}) - ([M^{c}]_{\tau_{s}} + \tau_{s})$$
$$= A_{\tau_{t}} - A_{\tau_{s}} = t - s.$$

Hence  $[M^c \circ \tau]$  is a.s. absolutely continuous with respect to the Lebesgue measure on  $\mathbb{R}_+$ . Moreover,  $(M^i \circ \tau)_{\infty} = M^i_{\infty}$ ,  $i \in \{c, d\}$ , so this time-change argument does not affect (4.3.1). Hence we can redefine  $M^c := M^c \circ \tau$ ,  $M^d := M^d \circ \tau$ ,  $\mathbb{F} = (\mathscr{F}_s)_{s \ge 0} := \mathbb{G} = (\mathscr{F}_{\tau_s})_{s \ge 0}$ .

Since  $[M^c]$  is a.s. absolutely continuous with respect to the Lebesgue measure on  $\mathbb{R}_+$  and thanks to Theorem 2.7.1, we can extend  $\Omega$  and find a *d*-dimensional Wiener process  $W : \mathbb{R}_+ \times \Omega \to \mathbb{R}^d$  and a stochastically integrable progressively measurable function  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(\mathbb{R}^d, X)$  such that  $M^c = \Phi \cdot W$ .

Let  $U: X \times X \to \mathbb{R}$  be the Burkholder function that was discussed in Section 3.3. Let us show that  $\mathbb{E}U(M_t, M_t^d) \leq 0$ .

Due to Proposition 4.3.2 and Remark 4.3.3 we can assume that  $M_s^c$ ,  $M_s^d$  and  $M_s = M_s^d + M_s^c$  have absolutely continuous distributions with respect to the Lebesgue measure  $\lambda_X$  on X for each  $s \ge 0$ . Let  $(x_n)_{n=1}^d$  be a basis of X,  $(x_n^*)_{n=1}^d$  be the corresponding dual basis of  $X^*$  (see Definition 2.11.1). By the Itô formula (2.12.1),

$$\mathbb{E}U(M_t, M_t^d) = \mathbb{E}U(M_0, M_0^d) + \mathbb{E}\int_0^t \langle \partial_x U(M_{s-}, M_{s-}^d), dM_s \rangle + \mathbb{E}\int_0^t \langle \partial_y U(M_{s-}, M_{s-}^d), dM_s^d \rangle + \mathbb{E}I_1 + \mathbb{E}I_2,$$
(4.3.2)

where

$$I_{1} = \sum_{0 < s \le t} [\Delta U(M_{s}, M_{s}^{d}) - \langle \partial_{x} U(M_{s-}, M_{s-}^{d}), \Delta M_{s} \rangle - \langle \partial_{y} U(M_{s-}, M_{s-}^{d}), \Delta M_{s}^{d} \rangle],$$

$$I_{2} = \frac{1}{2} \int_{0}^{t} \sum_{i,j=1}^{d} U_{x_{i},x_{j}}(M_{s-}, M_{s-}^{d}) d[\langle M, x_{i}^{*} \rangle, \langle M, x_{j}^{*} \rangle]_{s}^{c}$$

$$= \frac{1}{2} \int_{0}^{t} \sum_{i,j=1}^{d} U_{x_{i},x_{j}}(M_{s-}, M_{s-}^{d}) \langle \Phi^{*}(s) x_{i}^{*}, \Phi^{*}(s) x_{j}^{*} \rangle ds.$$

(Recall that by (3.3.7) and (3.3.8), *U* is Fréchet-differentiable a.s. on  $X \times X$ , hence  $\partial_x U$  and  $\partial_y U$  are well-defined. Moreover, *U* is zigzag-concave, so *U* is concave

in the first variable, and therefore the second-order derivatives  $U_{x_i,x_j}$  in the first variable are well-defined and exist a.s. on  $X \times X$  by the Alexandrov theorem [57, Theorem 6.4.1].) The last equality holds due to Theorem 2.12.1 and the fact that by Lemma 2.6.1 for all  $s \ge 0$  a.s.

$$\begin{split} [\langle M, x_i^* \rangle, \langle M, x_j^* \rangle]_s^c &= [\langle \Phi \cdot W, x_i^* \rangle, \langle \Phi \cdot W, x_j^* \rangle]_s = [(\Phi^* x_i^*) \cdot W, (\Phi^* x_j^*) \cdot W]_s \\ &= \int_0^s \langle \Phi^*(r) x_i^*, \Phi^*(r) x_j^* \rangle \, \mathrm{d}r. \end{split}$$

Let us first show that  $I_1 \le 0$  a.s. Indeed, since  $M^d$  is a purely discontinuous part of M, then by Definition 2.2.13  $\langle M^d, x^* \rangle$  is a purely discontinuous part of  $\langle M, x^* \rangle$ , and due to Theorem 2.2.10 a.s. for each  $t \ge 0$ 

$$\Delta|\langle M^d, x^*\rangle|_t^2 = \Delta[\langle M^d, x^*\rangle]_t = \Delta[\langle M, x^*\rangle]_t = \Delta|\langle M, x^*\rangle|_t^2$$

for each  $x^* \in X^*$ . Thus for each  $s \ge 0$  by (3.3.8) and (3.3.9)  $\mathbb{P}$ -a.s.

$$\begin{split} \Delta U(M_{s}, M_{s}^{d}) &- \langle \partial_{x} U(M_{s-}, M_{s-}^{d}), \Delta M_{s} \rangle - \langle \partial_{y} U(M_{s-}, M_{s-}^{d}), \Delta M_{s}^{d} \rangle \\ &= V(M_{s-} + M_{s-}^{d} + 2\Delta M_{s}, M_{s-}^{d} - M_{s-}) - V(M_{s-} + M_{s-}^{d}, M_{s-}^{d} - M_{s-}) \\ &- \langle \partial_{x} V(M_{s-} + M_{s-}^{d}, M_{s-}^{d} - M_{s-}), 2\Delta M_{s} \rangle \leq 0, \end{split}$$

so  $I_1 \leq 0$  a.s., and  $\mathbb{E}I_1 \leq 0$ . Now we show that

$$\mathbb{E}\left(\int_0^t \langle \partial_x U(M_{s-}, M_{s-}^d), \, \mathrm{d}M_s \rangle + \int_0^t \langle \partial_y U(M_{s-}, M_{s-}^d), \, \mathrm{d}M_s^d \rangle\right) = 0.$$

Indeed,

$$\begin{split} \int_{0}^{t} \langle \partial_{x} U(M_{s-}, M_{s-}^{d}), dM_{s} \rangle &+ \int_{0}^{t} \langle \partial_{y} U(M_{s-}, M_{s-}^{d}), dM_{s}^{d} \rangle \\ &= \int_{0}^{t} \langle \partial_{x} V(M_{s-} + M_{s-}^{d}, M_{s-}^{d} - M_{s-}), d(M_{s} + M_{s}^{d}) \rangle \\ &+ \int_{0}^{t} \langle \partial_{y} V(M_{s-} + M_{s-}^{d}, M_{s-}^{d} - M_{s-}), d(M_{s}^{d} - M_{s}) \rangle \end{split}$$

so by Lemma 3.2.2 and 3.3.13 it is a martingale which starts at zero, hence its expectation is zero.

Finally let us show that  $I_2 \le 0$  a.s. Fix  $s \in [0, t]$  and  $\omega \in \Omega$ . Then  $x^* \mapsto ||\Phi^*(s, \omega)x^*||^2$  defines a nonnegative definite quadratic form on  $X^*$ , and since any nonnegative quadratic form defines a Euclidean seminorm, there exists a basis  $(\tilde{x}_n^*)_{n=1}^d$  of  $X^*$  and a  $\{0, 1\}$ -valued sequence  $(a_n)_{n=1}^d$  such that

$$\langle \Phi^*(s,\omega)\tilde{x}_n^*, \Phi^*(s,\omega)\tilde{x}_m^* \rangle = a_n \delta_{mn}, \ m, n = 1, \dots, d.$$

Let  $(\tilde{x}_n)_{n=1}^d$  be the corresponding dual basis of *X* as it is defined in Definition 2.11.1. Then due to Lemma 2.11.2 and the linearity of  $\Phi$  and directional derivatives of *U*  (we skip *s* and  $\omega$  for the simplicity of the expressions)

$$\sum_{i,j=1}^{d} U_{x_{i},x_{j}}(M_{s-}, M_{s-}^{d}) \langle \Phi^{*} x_{i}^{*}, \Phi^{*} x_{j}^{*} \rangle = \sum_{i,j=1}^{d} U_{\tilde{x}_{i},\tilde{x}_{j}}(M_{s-}, M_{s-}^{d}) \langle \Phi^{*} \tilde{x}_{i}^{*}, \Phi^{*} \tilde{x}_{j}^{*} \rangle$$
$$= \sum_{i=1}^{d} U_{\tilde{x}_{i},\tilde{x}_{i}}(M_{s-}, M_{s-}^{d}) \| \Phi^{*} \tilde{x}_{i}^{*} \|^{2}.$$

Recall that *U* is zigzag-concave, so  $t \mapsto U(x + t\tilde{x}_i, y)$  is concave for each  $x, y \in X$ , i = 1, ..., d. Therefore  $U_{\tilde{x}_i, \tilde{x}_i}(M_{s-}, M_{s-}^d) \le 0$  a.s., and a.s.

$$\sum_{i=1}^{d} U_{\tilde{x}_{i},\tilde{x}_{i}}(M_{s-}(\omega), M_{s-}^{d}(\omega)) \|\Phi^{*}(s, \omega)\tilde{x}_{i}^{*}\|^{2} \leq 0.$$

Consequently,  $I_2 \leq 0$  a.s., and by (4.3.2), Remark 3.3.9 and the fact that  $M_0^d = M_0$ 

$$\mathbb{E}U(M_t, M_t^d) \le \mathbb{E}U(M_0, M_0) \le 0.$$

By (3.3.4),  $\mathbb{E} \| M_t^d \|^p - \beta_{n,x}^p \mathbb{E} \| M_t \|^p \le \mathbb{E} U(M_t, M_t^d) \le 0$ , so the first part of (4.3.1) holds.

The second part of (4.3.1) follows from the same machinery applied for *V*. Namely, one can analogously show that

$$\mathbb{E} \|M_t^c\|^p - \beta_{p,X}^p \mathbb{E} \|M_t\|^p \le \mathbb{E} U(M_t, M_t^c) = \mathbb{E} V(M^d + 2M^c, -M^d) \le 0$$

by using a *V*-version of (4.3.2), inequality (3.3.9), and the fact that *V* is concave in the first variable a.s. on  $X \times X$ .

Step 2: general case. Without loss of generality we set  $\mathscr{F}_{\infty} = \mathscr{F}_t$ . Let  $M_t = \xi$ . If  $\xi$  is a simple function, then it takes its values in a finite dimensional subspace  $X_0$  of X, and therefore  $(M_s)_{s\geq 0} = (\mathbb{E}(\xi|\mathscr{F}_s))_{s\geq 0}$  takes its values in  $X_0$  as well, so the theorem and (4.3.1) follow from Step 1.

Now let  $\xi$  be general. Let  $(\xi_n)_{n\geq 1}$  be a sequence of simple  $\mathscr{F}_t$ -measurable functions in  $L^p(\Omega; X)$  such that  $\xi_n \to \xi$  as  $n \to \infty$  in  $L^p(\Omega; X)$ . For each  $n \geq 1$  define  $\mathscr{F}_t$ -measurable  $\xi_n^d$  and  $\xi_n^c$  such that

$$M^{d,n} = (M^{d,n}_s)_{s\geq 0} = (\mathbb{E}(\xi^d_n | \mathscr{F}_s))_{s\geq 0},$$
  

$$M^{c,n} = (M^{c,n}_s)_{s\geq 0} = (\mathbb{E}(\xi^c_n | \mathscr{F}_s))_{s\geq 0}$$
(4.3.3)

are the respectively purely discontinuous and continuous parts of martingale  $M^n = (\mathbb{E}(\xi_n | \mathscr{F}_s))_{s \ge 0}$  as in Remark 2.2.14. Then due to Step 1 and (4.3.1),  $(\xi_n^d)_{n \ge 1}$  and  $(\xi_n^c)_{n \ge 1}$  are Cauchy sequences in  $L^p(\Omega; X)$ . Let  $\xi^c := L^p - \lim_{n \to \infty} \xi_n^c$  and  $\xi^d := L^p - \lim_{n \to \infty} \xi_n^d$ . Define the *X*-valued  $L^p$ -bounded martingales  $M^d$  and  $M^c$  by

$$M^{d} = (M_{s}^{d})_{s \ge 0} := (\mathbb{E}(\xi^{d} | \mathscr{F}_{s}))_{s \ge 0}, \ M^{c} = (M_{s}^{c})_{s \ge 0} := (\mathbb{E}(\xi^{c} | \mathscr{F}_{s}))_{s \ge 0}$$

Thanks to Proposition 2.2.16,  $M^d$  is purely discontinuous, and due to Proposition 2.2.8  $M^c$  is continuous and  $M_0^c = 0$ , so  $M = M^d + M^c$  is the desired decomposition.

The uniqueness of the decomposition follows from Lemma 2.2.17. For estimates (4.3.1) we note that by Step 1, (4.3.1) applied for Step 1, and [79, Proposition 4.2.17] for each  $n \ge 1$ 

$$(\mathbb{E}\|\xi_{n}^{d}\|^{p})^{\frac{1}{p}} \leq \beta_{p,X}(\mathbb{E}\|\xi_{n}\|^{p})^{\frac{1}{p}}, \quad (\mathbb{E}\|\xi_{n}^{c}\|^{p})^{\frac{1}{p}} \leq \beta_{p,X}(\mathbb{E}\|\xi_{n}\|^{p})^{\frac{1}{p}},$$

and it remains to let  $n \to \infty$ .

*Remark* 4.3.4. Let *X* be a UMD Banach space,  $1 , <math>M : \mathbb{R}_+ \times \Omega \to X$  be continuous (resp. purely discontinuous)  $L^p$ -bounded martingale. Then there exists a sequence  $(M^n)_{n\geq 1}$  of continuous (resp. purely discontinuous) *X*-valued  $L^p$ -bounded martingales such that  $M^n$  takes its values is a finite dimensional subspace of *X* for each  $n \ge 1$  and  $M_{\infty}^n \to M_{\infty}$  in  $L^p(\Omega; X)$  as  $n \to \infty$ . Such a sequence can be provided e.g. by (4.3.3).

We have proven the Meyer-Yoeurp decomposition in the UMD setting. Next we prove a converse result which shows the necessity of the UMD property.

**Theorem 4.3.5.** Let X be a finite dimensional Banach space,  $p \in (1,\infty)$ ,  $\delta \in (0, (\beta_{p,X}-1) \land 1)$ . 1). Then there exist a purely discontinuous martingale  $M^d : \mathbb{R}_+ \times \Omega \to X$ , a continuous martingale  $M^c : \mathbb{R}_+ \times \Omega \to X$  such that  $\mathbb{E} \|M^d_{\infty}\|^p, \mathbb{E} \|M^c_{\infty}\|^p < \infty$ ,  $M^d_0 = M^c_0 = 0$ , and for  $M = M^d + M^c$  and  $i \in \{c, d\}$  the following hold

$$(\mathbb{E} \| M_{\infty}^{i} \|^{p})^{\frac{1}{p}} \ge \left(\frac{\beta_{p,X} - 1}{2} - \delta\right) (\mathbb{E} \| M_{\infty} \|^{p})^{\frac{1}{p}}.$$
(4.3.4)

Recall that by [79, Proposition 4.2.17]  $\beta_{p,X} \ge \beta_{p,\mathbb{R}} = p^* - 1 \ge 1$  for any UMD Banach space *X* and 1 .

**Definition 4.3.6.** A random variable  $r : \Omega \to \{-1, 1\}$  is called a *Rademacher variable* if  $\mathbb{P}(r = 1) = \mathbb{P}(r = -1) = \frac{1}{2}$ .

**Lemma 4.3.7.** Let  $\varepsilon > 0$ ,  $p \in (1, \infty)$ . Then there exists a continuous martingale  $M : [0, 1] \times \Omega \rightarrow [-1, 1]$  with a symmetric distribution such that sign $M_1$  is a Rademacher random variable and

$$\|M_1 - \operatorname{sign} M_1\|_{L^p(\Omega)} < \varepsilon.$$
(4.3.5)

*Proof.* Let  $W : [0,1] \times \Omega \to \mathbb{R}$  be a standard Wiener process. For each  $n \ge 1$  we define a stopping time  $\tau_n := \inf\{t : |W_t| > \frac{1}{n}\} \land 1$ . Then  $\tau_n \to 0$  a.s. as  $n \to \infty$ , and hence there exists  $N \ge 1$  such that  $\mathbb{P}(NW_1^{\tau_N} = \operatorname{sign} W_1^{\tau_N}) > 1 - \frac{\varepsilon^p}{2p}$ . Let  $M = NW^{\tau_N}$ . Then

$$\|M_1 - \operatorname{sign} M_1\|_{L^p(\Omega)} \le \left(\mathbb{E}\left[(|M_1| + 1)^p \mathbf{1}_{M_1 \neq \operatorname{sign} M_1}\right]\right)^{\frac{1}{p}} < \left(2^p \cdot \frac{\varepsilon^p}{2^p}\right)^{\frac{1}{p}} \le \varepsilon,$$

and (4.3.5) follows.

Notice that since W is a Wiener process,  $W_1$  has a standard Gaussian distribution. Consequently,

$$\mathbb{P}(M_1 = 0) = \mathbb{P}(NW_1^{\tau_N} = 0) \le \mathbb{P}(NW_1 = 0) = 0,$$

and since  $W^{\tau_N}$  has a symmetric distribution, sign $M_1$  is Rademacher.

*Remark* 4.3.8. Let *X* be a UMD space,  $1 , <math>\delta > 0$ . Then using Proposition 2.3.1 one can construct a martingale difference sequence  $(d_j)_{j=1}^n \in L^p(\Omega; X)$  and a  $\{-1,1\}$ -valued sequence  $(\varepsilon_j)_{j=1}^n$  such that

$$\left(\mathbb{E}\left\|\sum_{j=1}^{n}\frac{\varepsilon_{j}\pm 1}{2}d_{j}\right\|^{p}\right)^{\frac{1}{p}}\geq\frac{\beta_{p,X}-\delta-1}{2}\left(\mathbb{E}\left\|\sum_{j=1}^{n}d_{j}\right\|^{p}\right)^{\frac{1}{p}}.$$

*Proof of Theorem* 4.3.5. Denote  $\frac{\beta_{p,X}-\delta-1}{2}$  by  $\gamma_{p,X}^{\delta}$ . By Proposition 2.3.1 there exists a natural number  $N \ge 1$ , a discrete *X*-valued martingale  $(f_n)_{n=0}^N$  such that  $f_0 = 0$ , and a sequence of scalars  $(\varepsilon_n)_{n=1}^N$  such that  $\varepsilon_n \in \{0,1\}$  for each n = 1, ..., N, such that

$$\left(\mathbb{E}\left\|\sum_{n=1}^{N}\varepsilon_{n}df_{n}\right\|^{p}\right)^{\frac{1}{p}} \geq \gamma_{p,X}^{\delta}(\mathbb{E}\|f_{N}\|^{p})^{\frac{1}{p}}.$$
(4.3.6)

According to [79, Theorem 3.6.1] we can assume that  $(f_n)_{n=0}^N$  is a Paley-Walsh martingale. Let  $(r_n)_{n=1}^N$  be a sequence of Rademacher variables and  $(\phi_n)_{n=1}^N$  be a sequence of functions as in Definition 2.2.5, i.e. be such that

$$f_n = \sum_{k=2}^n r_k \phi_k(r_1, \dots, r_{k-1}) + r_1 \phi_1, \quad n = 1, \dots, N$$

Without loss of generality we assume that

$$(\mathbb{E}\|f_N\|^p)^{\frac{1}{p}} \ge 2. \tag{4.3.7}$$

For each n = 1,...,N define a continuous martingale  $M^n : [0,1] \times \Omega \rightarrow [-1,1]$  as in Lemma 4.3.7, i.e. a martingale  $M^n$  with a symmetric distribution such that sign $M_1^n$  is a Rademacher variable and

$$\|M_{1}^{n} - \operatorname{sign} M_{1}^{n}\|_{L^{p}(\Omega)} < \frac{\delta}{KL},$$
(4.3.8)

where  $K = \beta_{p,X} N \max\{\|\phi_1\|, \|\phi_2\|_{\infty}, ..., \|\phi_N\|_{\infty}\}$ , and  $L = 2\beta_{p,X}$ . Without loss of generality suppose that  $(M^n)_{n=1}^N$  are independent. For each n = 1, ..., N set  $\sigma_n = \operatorname{sign} M_1^n$ . Define a martingale  $M : [0, N+1] \times \Omega \to X$  in the following way:

$$M_t = \begin{cases} 0, & \text{if } 0 \le t < 1; \\ M_{n-} + M_{t-n}^n \phi_n(\sigma_1, \dots, \sigma_{n-1}), & \text{if } t \in [n, n+1) \text{ and } \varepsilon_n = 0; \\ M_{n-} + \sigma_n \phi_n(\sigma_1, \dots, \sigma_{n-1}), & \text{if } t \in [n, n+1) \text{ and } \varepsilon_n = 1. \end{cases}$$

Let  $M = M^d + M^c$  be the decomposition of Theorem 4.3.1. Then

$$M_{N+1}^{c} = \sum_{n=1}^{N} M_{1}^{n} \phi_{n}(\sigma_{1},\ldots,\sigma_{n-1}) \mathbf{1}_{\varepsilon_{n}=0},$$

$$M_{N+1}^d = \sum_{n=1}^N \sigma_n \phi_n(\sigma_1, \dots, \sigma_{n-1}) \mathbf{1}_{\varepsilon_n = 1} = \sum_{n=1}^N \varepsilon_n \sigma_n \phi_n(\sigma_1, \dots, \sigma_{n-1}).$$

Notice that  $(\sigma_n)_{n=1}^N$  is a sequence of independent Rademacher variables, so by (4.3.6) and the discussion thereafter

$$\left(\mathbb{E}\left\|\sum_{n=1}^{N}\varepsilon_{n}\sigma_{n}\phi_{n}(\sigma_{1},\ldots,\sigma_{n-1})\right\|^{p}\right)^{\frac{1}{p}} \geq \gamma_{p,X}^{\delta}\left(\mathbb{E}\left\|\sum_{n=1}^{N}\sigma_{n}\phi_{n}(\sigma_{1},\ldots,\sigma_{n-1})\right\|^{p}\right)^{\frac{1}{p}}.$$
(4.3.9)

Let us first show (4.3.4) with i = d. Note that by the triangle inequality, (4.3.7) and (4.3.8)

$$(\mathbb{E} \| M_{N+1} \|^{p})^{\frac{1}{p}} \ge (\mathbb{E} \| f_{N} \|^{p})^{\frac{1}{p}} - \sum_{n=1}^{N} \left( \mathbb{E} \| (M_{1}^{n} - \sigma_{n})\phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \|^{p} \right)^{\frac{1}{p}}$$

$$\ge 2 - \frac{\delta}{KL} \cdot N \cdot \max\{ \| \phi_{1} \|, \| \phi_{2} \|_{\infty}, \dots, \| \phi_{N} \|_{\infty} \} > 1.$$

$$(4.3.10)$$

Therefore,

$$\begin{split} (\mathbb{E} \| M_{N+1}^{d} \|^{p})^{\frac{1}{p}} &= \left( \mathbb{E} \left\| \sum_{n=1}^{N} \varepsilon_{n} \sigma_{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \right\|^{p} \right)^{\frac{1}{p}} \stackrel{(i)}{\geq} \gamma_{p,X}^{\delta} \Big( \mathbb{E} \left\| \sum_{n=1}^{N} \sigma_{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \right\|^{p} \Big)^{\frac{1}{p}} \\ &\stackrel{(ii)}{\geq} \gamma_{p,X}^{\delta} \Big( \mathbb{E} \left\| \sum_{n=1}^{N} \mathbf{1}_{\varepsilon_{n}=1} \sigma_{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) + \sum_{n=1}^{N} \mathbf{1}_{\varepsilon_{n}=0} M_{1}^{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \right\|^{p} \Big)^{\frac{1}{p}} \\ &- \gamma_{p,X}^{\delta} \sum_{n=1}^{N} \Big( \mathbb{E} \left\| (M_{1}^{n} - \sigma_{n}) \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \right\|^{p} \Big)^{\frac{1}{p}} \\ &\stackrel{(iii)}{\geq} \gamma_{p,X}^{\delta} (\mathbb{E} \| M_{N+1} \|^{p})^{\frac{1}{p}} - \frac{\delta}{L} \stackrel{(iv)}{\geq} \Big( \frac{\beta_{p,X} - 1}{2} - \delta \Big) (\mathbb{E} \| M_{N+1} \|^{p})^{\frac{1}{p}}, \end{split}$$

where (*i*) follows from (4.3.9), (*ii*) holds by the triangle inequality, (*iii*) holds by (4.3.8), and (*iv*) follows from (4.3.10). By the same reason and Remark 4.3.8, (4.3.4) holds for i = c.

Let  $p \in (1,\infty)$ . Recall that  $\mathcal{M}_X^p$  is a space of all *X*-valued  $L^p$ -bounded martingales,  $\mathcal{M}_X^{p,d}, \mathcal{M}_X^{p,c} \subset \mathcal{M}_X^p$  are its subspaces of purely discontinuous martingales and continuous martingales that start at zero respectively (see Section 2.2).

**Theorem 4.3.9.** Let X be a Banach space. Then X is UMD if and only if for some (or, equivalently, for all)  $p \in (1,\infty)$ , for any probability space  $(\Omega, \mathscr{F}, \mathbb{P})$  with any filtration  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$  that satisfies the usual conditions,  $\mathscr{M}_X^p = \mathscr{M}_X^{p,d} \oplus \mathscr{M}_X^{p,c}$ , and there exist projections  $A^d, A^c \in \mathscr{L}(\mathscr{M}_X^p)$  such that ran  $A^d = \mathscr{M}_X^{p,d}$ , ran  $A^c = \mathscr{M}_X^{p,c}$ , and for any  $M \in \mathscr{M}_X^p$  the decomposition  $M = A^d M + A^c M$  is the Meyer-Yoeurp decomposition from Theorem 4.3.1. If this is the case, then

$$\|A^{a}\| \le \beta_{p,X} \text{ and } \|A^{c}\| \le \beta_{p,X}.$$
(4.3.11)

*Moreover, there exist*  $(\Omega, \mathscr{F}, \mathbb{P})$  *and*  $\mathbb{F} = (\mathscr{F}_t)_{t \ge 0}$  *such that* 

$$\|A^d\|, \|A^c\| \ge \frac{\beta_{p,X} - 1}{2} \lor 1.$$
(4.3.12)

*Proof.* The "if" part follows from (4.3.11), and the "only if" part follows from (4.3.12), so it is sufficient to show (4.3.11) and (4.3.12). (4.3.11) is equivalent to (4.3.1). The bound  $\geq \frac{\beta_{p,X}-1}{2}$  in (4.3.12) follows from Theorem 4.3.5, while the bound  $\geq 1$  follows from the fact that both  $A^d$  and  $A^c$  are projections onto nonzero spaces  $\mathcal{M}_X^{p,d}$  and  $\mathcal{M}_X^{p,c}$  respectively.

**Corollary 4.3.10.** Let X be a UMD Banach space,  $p \in (1,\infty)$ . Let  $i \in \{c,d\}$ . Then  $(\mathcal{M}_X^{p,i})^* \simeq \mathcal{M}_{X^*}^{p',i}$ , and for each  $M \in \mathcal{M}_{X^*}^{p',i}$  and  $N \in \mathcal{M}_X^{p,i}$ .

$$\langle M, N \rangle := \mathbb{E} \langle M_{\infty}, N_{\infty} \rangle, \quad \|M\|_{(\mathcal{M}_{X}^{p,i})^{*}} \eqsim_{p,X} \|M\|_{\mathcal{M}_{Y^{*}}^{p',i}}.$$

To prove the corollary above we will need the following lemma.

**Lemma 4.3.11.** Let X be a UMD Banach space,  $p \in (1,\infty)$ ,  $M \in \mathcal{M}_X^{p,d}$ ,  $N \in \mathcal{M}_{X^*}^{p',c}$ . Then  $\mathbb{E}\langle M_{\infty}, N_{\infty} \rangle = 0$ .

*Proof.* First suppose that  $N_{\infty}$  takes it values in a finite dimensional subspace *Y* of *X*<sup>\*</sup>. Let  $d \ge 1$  be the dimension of *Y*,  $(y_k)_{k=1}^d$  be the basis of *Y*. Then there exist  $N^1, \ldots, N^d \in \mathcal{M}_{\mathbb{R}}^{p',c}$  such that  $N = \sum_{k=1}^d N^k y_k$ . Hence

$$E\langle M_{\infty}, N_{\infty}\rangle = E\left\langle M_{\infty}, \sum_{k=1}^{d} N_{\infty}^{k} y_{k} \right\rangle = \sum_{k=1}^{d} \mathbb{E}\langle M_{\infty}, y_{k} \rangle N_{\infty}^{k} \stackrel{(*)}{=} 0, \qquad (4.3.13)$$

where (\*) holds due to Proposition 2.2.12.

Now turn to the general case. By Remark 4.3.4 for each  $N \in \mathcal{M}_{X^*}^{p',c}$  there exists a sequence  $(N^n)_{n\geq 1}$  of continuous martingales such that each of  $N^n$  is in  $\mathcal{M}_{X^*}^{p',c}$  and takes its valued in a finite dimensional subspace of  $X^*$ , and  $N_{\infty}^n \to N_{\infty}$  in  $L^{p'}(\Omega; X^*)$ as  $n \to \infty$ . Then due to (4.3.13),  $E\langle M_{\infty}, N_{\infty} \rangle = \lim_{n\to\infty} E\langle M_{\infty}, N_{\infty}^n \rangle = 0$ , so the lemma holds.

*Proof of Corollary* 4.3.10. We will show only the case i = d, the case i = c can be shown analogously.

 $\mathcal{M}_{X^*}^{p',d} \subset (\mathcal{M}_X^{p,d})^*$  and  $\|M\|_{(\mathcal{M}_X^{p,d})^*} \leq \|M\|_{\mathcal{M}_{X^*}^{p',d}}$  for each  $M \in \mathcal{M}_{X^*}^{p',d}$  thanks to the Hölder inequality. Now let us show the inverse. Let  $f \in (\mathcal{M}_X^{p,d})^*$ . Since due to Proposition 2.2.16  $\mathcal{M}_X^{p,d}$  is a closed subspace of  $\mathcal{M}_X^p$ , by the Hahn-Banach theorem and Proposition 2.2.3 there exists  $L \in \mathcal{M}_{X^*}^{p'}$  such that  $\mathbb{E}\langle L_{\infty}, N_{\infty} \rangle = f(N)$  for any  $N \in \mathcal{M}_X^{p,d}$ , and  $\|L\|_{\mathcal{M}_{X^*}^{p',d}} = \|f\|_{(\mathcal{M}_X^{p,d})^*}$ . Let  $L = L^d + L^c$  be the Meyer-Yoeurp decomposition of *L* as in Theorem 4.3.1. Then by (4.3.1)

$$\|L^{d}\|_{\mathcal{M}^{p',d}_{X^{*}}} \lesssim_{p,X} \|L\|_{\mathcal{M}^{p'}_{X^{*}}} = \|f\|_{(\mathcal{M}^{p,d}_{X})^{*}}$$

and  $\mathbb{E}\langle L_{\infty}^{d}, N_{\infty} \rangle = \mathbb{E}\langle L_{\infty}, N_{\infty} \rangle$ , so the theorem holds.

#### 4.3.2. Yoeurp decomposition of purely discontinuous martingales

As Yoeurp shown in [190] (see Subsection 2.4.3 and [89]), one can provide further decomposition of a purely discontinuous martingale into two parts: a martingale with accessible jumps and a quasi-left continuous martingale. This subsection is devoted to the generalization of this result to a UMD case.

**Theorem 4.3.12.** Let X be a UMD Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous  $L^p$ -bounded martingale. Then there exist unique purely discontinuous martingales  $M^a, M^q : \mathbb{R}_+ \times \Omega \to X$  such that  $M^a$  has accessible jumps,  $M^q$  is quasi-left continuous,  $M_0^q = 0$  and  $M = M^a + M^q$ . Moreover, if this is the case, then for  $i \in \{a, q\}$ 

$$(\mathbb{E} \| M_{\infty}^{i} \|^{p})^{\frac{1}{p}} \le \beta_{p,X} (\mathbb{E} \| M_{\infty} \|^{p})^{\frac{1}{p}}.$$
(4.3.14)

*Proof.* Step 1: finite dimensional case. First assume that X is finite dimensional. Then  $M^a$  and  $M^q$  exist and unique due to coordinate-wise applying of Corollary 2.4.12. Let  $M = M^a + M^q$ ,  $N = M^a$ . Then for any  $x^* \in X^*$ ,  $t \ge 0$  by Corollary 2.4.12 and Lemma 2.4.16 a.s.

$$[\langle M, x^* \rangle]_t = [\langle M, x^* \rangle]_t^a + [\langle M, x^* \rangle]_t^q = [\langle M^a, x^* \rangle]_t + [\langle M^q, x^* \rangle]_t,$$

and

$$[\langle N, x^* \rangle]_t = [\langle N, x^* \rangle]_t^a + [\langle N, x^* \rangle]_t^q = [\langle M^a, x^* \rangle]_t.$$

Therefore a.s.

$$[\langle N, x^* \rangle]_t - [\langle N, x^* \rangle]_s \le [\langle M, x^* \rangle]_t - [\langle M, x^* \rangle]_s, \quad 0 \le s < t.$$

Moreover  $M_0 = N_0$ . Hence *N* is weakly differentially subordinate to *M* (see Section 4.4), and (4.3.14) for i = a follows from Theorem 3.3.17. By the same reason and since  $M_0^q = 0$ , (4.3.14) holds true for i = q.

Step 2: general case. Now let X be general. Let  $\xi = M_{\infty}$ . Without loss of generality we set  $\mathscr{F}_{\infty} = \mathscr{F}_t$ . Let  $(\xi_n)_{n\geq 1}$  be a sequence of simple  $\mathscr{F}_t$ -measurable functions in  $L^p(\Omega; X)$  such that  $\xi_n \to \xi$  as  $n \to \infty$  in  $L^p(\Omega; X)$ . For each  $n \ge 1$  define  $\mathscr{F}_t$ -measurable  $\xi_n^d$  and  $\xi_n^c$  such that  $M^{d,n} = (\mathbb{E}(\xi_n^d | \mathscr{F}_s))_{s\geq 0}$  and  $M^{c,n} = (\mathbb{E}(\xi_n^c | \mathscr{F}_s))_{s\geq 0}$  are respectively purely discontinuous and continuous parts of a martingale  $(\mathbb{E}(\xi_n | \mathscr{F}_s))_{s\geq 0}$  as in Remark 2.2.14. Then thanks to Theorem 4.3.1,  $\xi_n^d \to \xi$  and  $\xi_n^c \to 0$  in  $L^p(\Omega; X)$  as  $n \to \infty$ since M is purely discontinuous.

Since for each  $n \ge 1$  the random variable  $\xi_n^d$  takes its values in a finite dimensional space, by Corollary 2.4.12 there exist  $\mathscr{F}_t$ -measurable  $\xi^a, \xi^q \in L^p(\Omega; X)$  such that purely discontinuous martingales  $M^{a,n} = (\mathbb{E}(\xi_n^a | \mathscr{F}_s))_{s \ge 0}$  and  $M^{q,n} = (\mathbb{E}(\xi_n^q | \mathscr{F}_s))_{s \ge 0}$  are respectively with accessible jumps and quasi-left continuous,  $\mathbb{E}(\xi_n^q | \mathscr{F}_0) = 0$ , and the decomposition  $M^{d,n} = M^{a,n} + M^{q,n}$  is as in Corollary 2.4.12. Since  $(\xi_n^d)_{n \ge 1}$  is a Cauchy sequence in  $L^p(\Omega; X)$ , by Step 1 both  $(\xi_n^a)_{n \ge 1}$  and  $(\xi_n^q)_{n \ge 1}$  are Cauchy in

 $L^{p}(\Omega; X)$  as well. Let  $\xi^{a}$  and  $\xi^{q}$  be their limits. Define martingales  $M^{a}, M^{q}: \mathbb{R}_{+} \times \Omega \rightarrow X$  in the following way:

$$M_s^a := \mathbb{E}(\xi^a | \mathscr{F}_s), \ M_s^q := \mathbb{E}(\xi^q | \mathscr{F}_s), \ s \ge 0.$$

By Proposition 2.4.18  $M^a$  is a martingale with accessible jumps,  $M^q$  is quasi-left continuous,  $M_0^q = 0$  a.s., and therefore  $M = M^a + M^q$  is the desired decomposition. Moreover, by Step 1 for each  $n \ge 1$  and  $i \in \{a, q\}$ ,  $(\mathbb{E} \|\xi_n^i\|^p)^{\frac{1}{p}} \le \beta_{p,X} (\mathbb{E} \|\xi_n^d\|^p)^{\frac{1}{p}}$ , and hence the estimate (4.3.14) follows by letting *n* to infinity.

The uniqueness of the decomposition follows from Lemma 2.4.19.  $\Box$ 

The following theorem, as Theorem 4.3.5, illustrates that the decomposition in Theorem 4.3.12 takes place only in the UMD space case.

**Theorem 4.3.13.** Let X be a finite dimensional Banach space,  $p \in (1,\infty)$ ,  $\delta \in \left(0, \frac{\beta_{p,X}-1}{2}\right)$ . Then there exist purely discontinuous martingales  $M^a, M^q : \mathbb{R}_+ \times \Omega \to X$  such that  $M^a$  has accessible jumps,  $M^q$  is quasi-left continuous,  $\mathbb{E} \|M^a_{\infty}\|^p$ ,  $\mathbb{E} \|M^q_{\infty}\|^p < \infty$ ,  $M^a_0 = M^q_0 = 0$ , and for  $M = M^a + M^q$  and  $i \in \{a, q\}$  the following holds

$$(\mathbb{E} \| M_{\infty}^{i} \|^{p})^{\frac{1}{p}} \ge \left(\frac{\beta_{p,X} - 1}{2} - \delta\right) (\mathbb{E} \| M_{\infty} \|^{p})^{\frac{1}{p}}.$$
(4.3.15)

For the proof we will need the following lemma.

**Lemma 4.3.14.** Let  $\varepsilon \in (0, \frac{1}{2})$ ,  $p \in (1, \infty)$ . Then there exist martingales  $M, M^a, M^q$ :  $[0,1] \times \Omega \rightarrow [-1-\varepsilon, 1+\varepsilon]$  with symmetric distributions such that  $M^a$  is a martingale with accessible jumps,  $\|M_1^a\|_{L^p(\Omega)} < \varepsilon$ ,  $M^q$  is a quasi-left continuous martingale,  $M_0^q = 0$  a.s.,  $M = M^a + M^q$ , sign  $M_1$  is a Rademacher random variable and

$$\|M_1 - \operatorname{sign} M_1\|_{L^p(\Omega)} < \varepsilon.$$
(4.3.16)

*Proof.* Let  $N^+$ ,  $N^-$  :  $[0, 1] \times \Omega \to \mathbb{R}$  be independent Poisson processes with the same intensity  $\lambda_{\varepsilon}$  such that  $\mathbb{P}(N_1^+ = 0) = \mathbb{P}(N_1^- = 0) < \frac{\varepsilon^p}{2^p}$  (such  $\lambda_{\varepsilon}$  exists since  $N_1^+$  and  $N_1^-$  have Poisson distributions, see [95]). Define a stopping time  $\tau$  in the following way:

$$\tau = \inf\{t : N_t^+ \ge 1\} \land \inf\{t : N_t^- \ge 1\} \land 1.$$

Let  $M_t^q := N_{t\wedge\tau}^+ - N_{t\wedge\tau}^-$ ,  $t \in [0,1]$ . Then  $M^q$  is quasi-left continuous with a symmetric distribution. Let r be an independent Rademacher variable,  $M_t^a = \frac{\varepsilon}{2}r$  for each  $t \in [0,1]$ . Then  $M^a$  is a martingale with accessible jumps and symmetric distribution, and  $\|M_1^a\|_{L^p(\Omega)} = \frac{\varepsilon}{2} < \varepsilon$ . Let  $M = M^a + M^q$ . Then a.s.

$$M_1 \in \left\{-1 - \frac{\varepsilon}{2}, -1 + \frac{\varepsilon}{2}, -\frac{\varepsilon}{2}, \frac{\varepsilon}{2}, 1 - \frac{\varepsilon}{2}, 1 + \frac{\varepsilon}{2}\right\},$$
(4.3.17)

so  $\mathbb{P}(M_1 = 0) = 0$ , and therefore sign $M_1$  is a Rademacher random variable. Let us prove (4.3.16). Notice that due to (4.3.17) if  $|M_1^q| = 1$ , then  $|M_1 - \text{sign}M_1| < \frac{\varepsilon}{2}$ , and if  $|M_1^q| = 0$ , then  $|M_1 - \text{sign}M_1| < 1$ . Therefore

$$\mathbb{E}|M_{1} - \operatorname{sign} M_{1}|^{p} = \mathbb{E}|M_{1} - \operatorname{sign} M_{1}|^{p} \mathbf{1}_{|M_{1}^{q}|=1} + \mathbb{E}|M_{1} - \operatorname{sign} M_{1}|^{p} \mathbf{1}_{|M_{1}^{q}|=0}$$
$$< \frac{\varepsilon^{p}}{2^{p}} + \frac{\varepsilon^{p}}{2^{p}} < \varepsilon^{p},$$

so (4.3.16) holds.

*Proof of Theorem* 4.3.13. The proof is analogous to the proof of Theorem 4.3.5, while one has to use Lemma 4.3.14 instead of Lemma 4.3.7.  $\Box$ 

Theorem 4.3.13 yields the following characterization of the UMD property.

**Theorem 4.3.15.** Let X be a Banach space. Then X is a UMD Banach space if and only if for some (equivalently, for all)  $p \in (1, \infty)$  there exists  $c_{p,X} > 0$  such that for any  $L^p$ -bounded martingale  $M := \mathbb{R}_+ \times \Omega \to X$  there exist unique martingales  $M^c, M^q, M^a : \mathbb{R}_+ \times \Omega \to X$  such that  $M_0^c = M_0^q = 0$ ,  $M^c$  is continuous,  $M^q$  is purely discontinuous quasi-left continuous,  $M^a$  is purely discontinuous with accessible jumps,  $M = M^c + M^q + M^a$ , and

$$(\mathbb{E}\|M_{\infty}^{c}\|^{p})^{\frac{1}{p}} + (\mathbb{E}\|M_{\infty}^{q}\|^{p})^{\frac{1}{p}} + (\mathbb{E}\|M_{\infty}^{a}\|^{p})^{\frac{1}{p}} \le c_{p,X}(\mathbb{E}\|M_{\infty}\|^{p})^{\frac{1}{p}}.$$
(4.3.18)

If this is the case, then the least admissible  $c_{p,X}$  is in the interval  $\left[\frac{3\beta_{p,X}-3}{2} \vee 1, 3\beta_{p,X}\right]$ .

The decomposition  $M = M^c + M^q + M^a$  is called the *canonical decomposition* of the martingale *M* (see Subsection 2.4.3).

*Proof.* The "if and only if" part follows from Theorem 4.3.9, Theorem 4.3.12 and Theorem 4.3.13. The estimate  $c_{p,X} \leq 3\beta_{p,X}$  follows from (4.3.1) and (4.3.14). The estimate  $c_{p,X} \geq \frac{3\beta_{p,X}-3}{2} \vee 1$  follows from (4.3.4) and (4.3.15).

**Corollary 4.3.16.** Let X be a Banach space. Then X is a UMD Banach space if and only if  $\mathcal{M}_X^{p,d} = \mathcal{M}_X^{p,a} \oplus \mathcal{M}_X^{p,q}$  and  $\mathcal{M}_X^p = \mathcal{M}_X^{p,c} \oplus \mathcal{M}_X^{p,q} \oplus \mathcal{M}_X^{p,a}$  for any filtration that satisfies the usual conditions.

*Proof.* The corollary follows from Theorem 4.3.12, Theorem 4.3.13 and Theorem 4.3.15.  $\Box$ 

#### 4.3.3. Stochastic integration

The current subsection is devoted to application of Theorem 4.3.15 to stochastic integration with respect to a general martingale.

**Theorem 4.3.17.** Let *H* be a Hilbert space, *X* be a UMD Banach space,  $p \in (1,\infty)$ , *M* :  $\mathbb{R}_+ \times \Omega \to H$  be a local martingale,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H,X)$  be elementary progressive. Let  $M = M^c + M^q + M^a$  be the canonical decomposition from Theorem 5.1.1. Then

$$\mathbb{E}\|(\Phi \cdot M)_{\infty}\|^{p} = \sum_{p,X} \mathbb{E}\|(\Phi \cdot M^{c})_{\infty}\|^{p} + \mathbb{E}\|(\Phi \cdot M^{q})_{\infty}\|^{p} + \mathbb{E}\|(\Phi \cdot M^{a})_{\infty}\|^{p}.$$
(4.3.19)

and if  $(\Phi \cdot M)_{\infty} \in L^{p}(\Omega; X)$ , then  $\Phi \cdot M = \Phi \cdot M^{c} + \Phi \cdot M^{q} + \Phi \cdot M^{a}$  is the canonical decomposition from Theorem 4.3.15.

*Proof.* The statement that  $\Phi \cdot M = \Phi \cdot M^c + \Phi \cdot M^q + \Phi \cdot M^a$  is the canonical decomposition follows from Proposition 2.5.1, Theorem 4.3.15 and the fact that a.s.  $(\Phi \cdot M)_0 = (\Phi \cdot M^c)_0 = (\Phi \cdot M^q)_0 = 0$ . (4.3.19) follows then from (4.3.18) and the triangle inequality.

*Remark* 4.3.18. Notice that the Itô isomorphism for the term  $\Phi \cdot M^c$  from (4.3.19) was explored in [177]. It remains open what to do with the other two terms, but positive results in this direction were obtained in the case of  $X = L^q(S)$  in Chapter 7.

# 4.4. WEAK DIFFERENTIAL SUBORDINATION AND GENERAL MAR-TINGALES

This section is devoted to the generalization of Theorem 3.3.17. Namely, here we show the  $L^p$ -estimates for general *X*-valued weakly differentially subordinated martingales.

**Theorem 4.4.1.** Let X be a UMD Banach space,  $M, N : \mathbb{R}_+ \times \Omega \to X$  be two martingales such that N is weakly differentially subordinate to M. Then for each  $p \in (1,\infty)$ ,  $t \ge 0$ ,

$$(\mathbb{E}\|N_t\|^p)^{\frac{1}{p}} \le \beta_{p,X}^2 (\beta_{p,X} + 1) (\mathbb{E}\|M_t\|^p)^{\frac{1}{p}}.$$
(4.4.1)

The proof will be done in several steps. First we show an analogue of Theorem 3.3.17 for continuous martingales.

**Theorem 4.4.2.** Let X be a Banach space. Then X is a UMD Banach space if and only if for some (equivalently, for all)  $p \in (1,\infty)$  there exists c > 0 such that for any continuous martingales  $M, N : \mathbb{R}_+ \times \Omega \to X$  such that N is weakly differentially subordinate to M,  $M_0 = N_0 = 0$ , one has that

$$(\mathbb{E} \| N_{\infty} \|^{p})^{\frac{1}{p}} \le c_{p,X} (\mathbb{E} \| M_{\infty} \|^{p})^{\frac{1}{p}}.$$
(4.4.2)

If this is the case, then the least admissible  $c_{p,X}$  is in the segment  $[\beta_{p,X}, \beta_{p,X}^2]$ .

For the proof we will need the following proposition, which demonstrates that one needs a slightly weaker assumption rather then in Theorem 4.4.2 so that the estimate (4.4.2) holds in a UMD Banach space. **Proposition 4.4.3.** Let X be a UMD Banach space,  $1 , <math>M, N : \mathbb{R}_+ \times \Omega \to X$  be continuous  $L^p$ -bounded martingales s.t.  $M_0 = N_0 = 0$  and for each  $x^* \in X^*$  a.s. for each  $t \ge 0$ 

$$[\langle N, x^* \rangle]_t \le [\langle M, x^* \rangle]_t. \tag{4.4.3}$$

*Then for each*  $t \ge 0$ 

$$(\mathbb{E}\|N_t\|^p)^{\frac{1}{p}} \le \beta_{p,X}^2 (\mathbb{E}\|M_t\|^p)^{\frac{1}{p}}.$$
(4.4.4)

*Proof.* Without loss of generality by a stopping time argument we assume that M and N are bounded and that  $M_{\infty} = M_t$  and  $N_{\infty} = N_t$ .

One can also restrict to a finite dimensional case. Indeed, since *X* is a separable reflexive space,  $X^*$  is separable as well. Let  $(Y_m)_{m\geq 1}$  be an increasing sequence of finite-dimensional subspaces of  $X^*$  such that  $\bigcup_m Y_m = X^*$  and  $\|\cdot\|_{Y_m} = \|\cdot\|_{X^*|_{Y_m}}$  for each  $m \geq 1$ . Then for each fixed  $m \geq 1$  there exists a linear operator  $P_m : X \to Y_m^*$  of norm 1 defined as follows:  $\langle P_m x, y \rangle = \langle x, y \rangle$  for each  $x \in X, y \in Y_m$ . Therefore  $P_m M$  and  $P_m N$  are  $Y_m^*$ -valued martingales. Moreover, (4.4.3) holds for  $P_m M$  and  $P_m N$  since there exists  $P_m^* : Y_m \to X^*$ , and for each  $y \in Y_m$  we have that  $\langle P_m M, y \rangle = \langle M, P_m y \rangle$  and  $\langle P_m N, y \rangle = \langle N, P_m y \rangle$ . Since  $Y_m$  is a closed subspace of  $X^*$ , [79, Proposition 4.2.17] yields  $\beta_{p',Y_m} \leq \beta_{p,X^*}$ , consequently again by [79, Proposition 4.2.17]  $\beta_{p,Y_m^*} \leq \beta_{p,X^*} = \beta_{p,X^*}$ . So if we prove the finite dimensional version, then

$$(\mathbb{E} \|P_m N_t\|^p)^{\frac{1}{p}} \le \beta_{p,Y_m^*}^2 (\mathbb{E} \|P_m M_t\|^p)^{\frac{1}{p}} \le \beta_{p,X}^2 (\mathbb{E} \|P_m M_t\|^p)^{\frac{1}{p}},$$

and (4.4.4) with  $c_{p,X} = \beta_{p,X}^2$  will follow by letting  $m \to \infty$ .

Let *d* be the dimension of *X*,  $\|\|\cdot\|\|$  be a Euclidean norm on  $X \times X$ . Let L = (M, N):  $\mathbb{R}_+ \times \Omega \to X \times X$  be a continuous martingale. Since  $(X \times X, \|\|\cdot\|\|)$  is a Hilbert space, *L* has a continuous quadratic variation  $[L] : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  (see Remark 2.2.7). Let  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be such that  $A_s = [L]_s + s$  for each  $s \ge 0$ . Then *A* is continuous strictly increasing predictable. Define a random time-change  $(\tau_s)_{s\ge 0}$  as in Theorem 2.4.25. Let  $\mathbb{G} = (\mathscr{G}_s)_{s\ge 0} = (\mathscr{F}_{\tau_s})_{s\ge 0}$  be the induced filtration. Then thanks to the Kazamaki theorem [89, Theorem 17.24]  $\widetilde{L} = L \circ \tau$  is a *G*-martingale, and  $[\widetilde{L}] = [L] \circ \tau$ . Notice that  $\widetilde{L} = (\widetilde{M}, \widetilde{N})$  with  $\widetilde{M} = M \circ \tau$ ,  $\widetilde{N} = N \circ \tau$ , and since by Kazamaki theorem [89, Theorem 17.24]  $[M \circ \tau] = [M] \circ \tau$ ,  $[N \circ \tau] = [N] \circ \tau$ , and  $(M \circ \tau)_0 = (N \circ \tau)_0 = 0$ , we have that by (4.4.3) for each  $x^* \in X^*$  a.s. for each  $s \ge 0$ 

$$[\langle \widetilde{N}, x^* \rangle]_s = [\langle N, x^* \rangle]_{\tau_s} \le [\langle M, x^* \rangle]_{\tau_s} = [\langle \widetilde{M}, x^* \rangle]_s \tag{4.4.5}$$

Moreover, for all  $0 \le u < s$  we have that a.s.

$$\begin{split} [\tilde{L}]_{s} - [\tilde{L}]_{u} &= ([L] \circ \tau)_{s} - ([L] \circ \tau)_{u} \le ([L] \circ \tau)_{s} + \tau_{s} - ([L] \circ \tau)_{u} - \tau_{u} \\ &= ([L]_{\tau_{s}} + \tau_{s}) - ([L]_{\tau_{u}} + \tau_{u}) = s - u. \end{split}$$

Therefore  $[\tilde{L}]$  is a.s. absolutely continuous with respect to the Lebesgue measure on  $\mathbb{R}_+$ . Consequently, due to Theorem 2.7.1, there exists an enlarged probability space  $(\tilde{\Omega}, \widetilde{\mathscr{F}}, \widetilde{\mathbb{P}})$  with an enlarged filtration  $\widetilde{\mathbb{G}} = (\widetilde{\mathscr{G}}_s)_{s \ge 0}$ , a 2*d*-dimensional standard Wiener process *W*, which is defined on  $\widetilde{\mathbb{G}}$ , and a stochastically integrable progressively measurable function  $f : \mathbb{R}_+ \times \widetilde{\Omega} \to \mathscr{L}(\mathbb{R}^{2d}, X \times X)$  such that  $\widetilde{L} = f \cdot W$ . Let  $f^M, f^N : \mathbb{R}_+ \times \Omega \to \mathscr{L}(\mathbb{R}^{2d}, X)$  be such that  $f = (f^M, f^N)$ . Then  $\widetilde{M} = f^M \cdot W$  and  $\widetilde{N} = f^N \cdot W$ . Let  $(\overline{\Omega}, \overline{\mathscr{F}}, \overline{\mathbb{P}})$  be an independent probability space with a filtration  $\overline{\mathbb{G}}$  and a 2*d*-dimensional Wiener process  $\overline{W}$  on it. Denote by  $\overline{\mathbb{E}}$  the expectation on  $(\overline{\Omega}, \overline{\mathscr{F}}, \overline{\mathbb{P}})$ . Then because of the decoupling theorem [79, Theorem 4.4.1], for each  $s \ge 0$ 

$$(\mathbb{E}\|\widetilde{N}_{s}\|^{p})^{\frac{1}{p}} = (\mathbb{E}\|(f^{N}\cdot W)_{s}\|^{p})^{\frac{1}{p}} \leq \beta_{p,X}(\mathbb{E}\overline{\mathbb{E}}\|(f^{N}\cdot \overline{W})_{s}\|^{p})^{\frac{1}{p}},$$

$$\frac{1}{\beta_{p,X}}(\mathbb{E}\overline{\mathbb{E}}\|(f^{M}\cdot \overline{W})_{s}\|^{p})^{\frac{1}{p}} \leq (\mathbb{E}\|(f^{M}\cdot W)_{s}\|^{p})^{\frac{1}{p}} = (\mathbb{E}\|\widetilde{M}_{s}\|^{p})^{\frac{1}{p}}.$$

$$(4.4.6)$$

Due to the multidimensional version of [89, Theorem 17.11] and (4.4.5) for each  $x^* \in X^*$  we have that

$$s \mapsto [\langle \widetilde{M}, x^* \rangle]_s - [\langle \widetilde{N}, x^* \rangle]_s = \int_0^s (|\langle x^*, f^M(r) \rangle|^2 - |\langle x^*, f^N(r) \rangle|^2) \,\mathrm{d}r \tag{4.4.7}$$

is nonnegative and absolutely continuous a.s. Since *X* is separable, we can fix a set  $\tilde{\Omega}_0 \subset \tilde{\Omega}$  of full measure on which the function (4.4.7) is nonnegative for each  $s \ge 0$ .

Now fix  $\omega \in \widetilde{\Omega}_0$  and  $s \ge 0$ . Let us prove that

$$\overline{\mathbb{E}} \| (f^N(\omega) \cdot \overline{W})_s \|^p \le \overline{\mathbb{E}} \| (f^M(\omega) \cdot \overline{W})_s \|^p.$$

Since  $f^M(\omega)$  and  $f^N(\omega)$  are deterministic on  $\overline{\Omega}$ , and since due to (4.4.7) for each  $x^* \in X^*$ 

$$\begin{split} \overline{\mathbb{E}}|\langle (f^{N}(\omega) \cdot \overline{W})_{s}, x^{*} \rangle|^{2} &= \int_{0}^{s} |\langle x^{*}, f^{N}(r, \omega) \rangle|^{2} \, \mathrm{d}r \\ &\leq \int_{0}^{s} |\langle x^{*}, f^{M}(r, \omega) \rangle|^{2} \, \mathrm{d}r = \overline{\mathbb{E}}|\langle (f^{M}(\omega) \cdot \overline{W})_{s}, x^{*} \rangle|^{2}, \end{split}$$

by [129, Corollary 4.4] we have that  $\overline{\mathbb{E}} \| (f^N(\omega) \cdot \overline{W})_s \|^p \leq \overline{\mathbb{E}} \| (f^M(\omega) \cdot \overline{W})_s \|^p$ . Consequently, due to (4.4.6) and the fact that  $\widetilde{\mathbb{P}}(\Omega_0) = 1$ 

$$(\mathbb{E}\|\widetilde{N}_{s}\|^{p})^{\frac{1}{p}} \leq \beta_{p,X}(\mathbb{E}\overline{\mathbb{E}}\|(f^{N} \cdot \overline{W})_{s}\|^{p})^{\frac{1}{p}} \leq \beta_{p,X}(\mathbb{E}\overline{\mathbb{E}}\|(f^{M} \cdot \overline{W})_{s}\|^{p})^{\frac{1}{p}} \leq \beta_{p,X}^{2}(\mathbb{E}\|\widetilde{M}_{s}\|^{p})^{\frac{1}{p}}$$

Recall that  $\widetilde{M}$  and  $\widetilde{N}$  are bounded, so thanks to the dominated convergence theorem one gets (4.4.4) with  $c_{p,X} = \beta_{p,X}^2$  by letting *s* to infinity.

*Proof of Theorem* 4.4.2. *The "only if" part & the upper bound of*  $c_{p,X}$ : The "only if" part and the estimate  $c_{p,X} \leq \beta_{p,X}^2$  follows from Proposition 4.4.3 since (4.4.3) holds for *M* and *N* because *N* is weakly differentially subordinate to *M*.

The "if" part & the lower bound of  $c_{p,X}$ : Let  $\beta_{p,X}$  be the UMD constant of X ( $\beta_{p,X} = \infty$  if X is not a UMD space). Fix  $K \ge 1$ . Then by [79, Theorem 4.2.5] there exists  $N \ge 1$ , a Paley-Walsh martingale difference sequence  $(d_n)_{n=1}^N$ , and a  $\{-1,1\}$ -valued sequence  $(\varepsilon_n)_{n=1}^N$  such that

$$\left(\mathbb{E}\left\|\sum_{n=1}^{N}\varepsilon_{n}d_{n}\right\|^{p}\right)^{\frac{1}{p}} \geq \left(\beta_{p,X} \wedge 2K - \frac{1}{2K}\right)\left(\mathbb{E}\left\|\sum_{n=1}^{N}d_{n}\right\|^{p}\right)^{\frac{1}{p}}$$

Without loss of generality we can assume that

$$\left(\mathbb{E}\left\|\sum_{n=1}^{N}\varepsilon_{n}d_{n}\right\|^{p}\right)^{\frac{1}{p}}, \left(\mathbb{E}\left\|\sum_{n=1}^{N}d_{n}\right\|^{p}\right)^{\frac{1}{p}} \leq 1.$$

Let  $(r_n)_{n=1}^N$  be a sequence of Rademacher variables and  $(\phi_n)_{n=1}^N$  be a sequence of functions as in Definition 2.2.5, i.e. be such that  $d_n = r_n \phi_n(r_1, ..., r_{n-1})$  for each n = 1, ..., N.

By the same techniques as were used in the proof of Theorem 4.3.5 we can find a sequence of independent continuous real-valued symmetric martingales  $(M^n)_{n=1}^N$  on [0, 1] such that for each n = 1, ..., N

$$\|(M^{n} - \operatorname{sign} M^{n})\phi_{n}(\operatorname{sign} M^{1}, \dots, \operatorname{sign} M^{n-1})\|_{L^{p}(\Omega; X)} \le \frac{1}{8NK^{2}}.$$
 (4.4.8)

Let  $\sigma_n = \operatorname{sign} M^n$  for each n = 1, ..., N. Then we define continuous martingales  $M, N : \mathbb{R}_+ \times \Omega \to X$  in the following way:

$$M_{t} = \begin{cases} 0, & \text{if } 0 \le t \le 1; \\ M_{n} + M_{t-n}^{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}), & \text{if } t \in (n, n+1], n \in \{1 \dots, N\}, \\ M_{N+1}, & \text{if } t > N+1, \end{cases}$$
$$N_{t} = \begin{cases} 0, & \text{if } 0 \le t \le 1; \\ M_{n} + \varepsilon_{n} M_{t-n}^{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}), & \text{if } t \in (n, n+1], n \in \{1 \dots, N\}, \\ N_{N+1}, & \text{if } t > N+1. \end{cases}$$

Then *N* is weakly differentially subordinate to *M*. Indeed, for each  $x^* \in X^*$ ,  $n \in \{1, ..., N\}$  and  $t \in [n, n+1]$  a.s.

$$\begin{split} [\langle M, x^* \rangle]_t - [\langle M, x^* \rangle]_n &= [M^n]_{t-n} |\langle \phi_n(\sigma_1, \dots, \sigma_{n-1}), x^* \rangle|^2 \\ &= [M^n]_{t-n} |\langle \varepsilon_n \phi_n(\sigma_1, \dots, \sigma_{n-1}), x^* \rangle|^2 \\ &= [\langle N, x^* \rangle]_t - [\langle N, x^* \rangle]_n, \end{split}$$

therefore, since  $M_1 = N_1 = 0$  a.s., we have that for each  $x^* \in X^*$  and  $t \ge 0$  a.s.  $[\langle M, x^* \rangle]_t = [\langle N, x^* \rangle]_t$ , so *N* is weakly differentially subordinate to *M*. Then

$$\begin{split} (\mathbb{E} \| N_{\infty} \|^{p})^{\frac{1}{p}} &= \left( \mathbb{E} \left\| \sum_{n=1}^{N} \varepsilon_{n} M_{1}^{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \right\| \right)^{\frac{1}{p}} \\ &\stackrel{(i)}{\geq} \left( \mathbb{E} \left\| \sum_{n=1}^{N} \varepsilon_{n} \sigma_{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \right\| \right)^{\frac{1}{p}} \\ &- \sum_{n=1}^{N} \| (M^{n} - \sigma_{n}) \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \|_{L^{p}(\Omega; X)} \\ &\stackrel{(ii)}{\geq} \left( \beta_{p, X} \wedge 2K - \frac{1}{2K} \right) \left( \mathbb{E} \left\| \sum_{n=1}^{N} \sigma_{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}) \right\| \right)^{\frac{1}{p}} - \frac{1}{8K^{2}} \end{split}$$

$$\stackrel{(iii)}{\geq} \left(\beta_{p,X} \wedge 2K - \frac{1}{2K}\right) \left(\mathbb{E} \left\|\sum_{n=1}^{N} M_{1}^{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1})\right\|\right)^{\frac{1}{p}} - 2K \sum_{n=1}^{N} \|(M^{n} - \sigma_{n})\phi_{n}(\sigma_{1}, \dots, \sigma_{n-1})\|_{L^{p}(\Omega; X)} - \frac{1}{8K^{2}} \right)$$

$$\stackrel{(iv)}{\geq} \left(\beta_{p,X} \wedge K - \frac{1}{K}\right) \left(\mathbb{E} \left\|\sum_{n=1}^{N} M_{1}^{n} \phi_{n}(\sigma_{1}, \dots, \sigma_{n-1})\right\|\right)^{\frac{1}{p}} = \left(\beta_{p,X} \wedge K - \frac{1}{K}\right) \left(\mathbb{E} \|M_{\infty}\|^{p}\right)^{\frac{1}{p}},$$

where (*i*) and (*iii*) follow from the triangle inequality, and (*ii*) and (*iv*) follow from (4.4.8). Hence if *X* is not UMD, then such  $c_{p,X}$  from (4.4.2) does not exist since  $(\beta_{p,X} \wedge K - \frac{1}{K}) \rightarrow \infty$  as  $K \rightarrow \infty$ . If *X* is UMD, then such  $c_{p,X}$  could exist, and if this is the case, then

$$c_{p,X} \ge \lim_{K \to \infty} \left( \beta_{p,X} \wedge K - \frac{1}{K} \right) = \beta_{p,X}.$$

*Remark* 4.4.4. Let *X* be a Banach space. Then according to [23, 32, 61] the Hilbert transform  $\mathscr{H}_X$  can be extended to  $L^p(\mathbb{R}; X)$  for each 1 if and only if*X*is a UMD Banach space. Moreover, if this is the case, then

$$\sqrt{\beta_{p,X}} \le \|\mathcal{H}_X\|_{\mathcal{L}(L^p(\mathbb{R};X))} \le \beta_{p,X}^2.$$

As it was shown in Section 3.5, the upper bound  $\beta_{p,X}^2$  can be also directly derived from the upper bound for  $c_{p,X}$  in Theorem 4.4.2. The sharp upper bound for  $\|\mathscr{H}_X\|_{\mathscr{L}(L^p(\mathbb{R};X))}$  remains an open question (see [79, pp. 496-497]), so the sharp upper bound for  $c_{p,X}$  is of interest.

**Lemma 4.4.5.** Let X be a Banach space,  $M^c$ ,  $N^c : \mathbb{R}_+ \times \Omega \to X$  be continuous martingales,  $M^d$ ,  $N^d : \mathbb{R}_+ \times \Omega \to X$  be purely discontinuous martingales,  $M_0^c = N_0^c = 0$ . Let  $M := M^c + M^d$ ,  $N := N^c + N^d$ . Suppose that N is weakly differentially subordinate to M. Then  $N^c$  is weakly differentially subordinate to  $M^c$ , and  $N^d$  is weakly differentially subordinate to  $M^d$ .

*Proof.* First notice that a.s.

$$\|N_0^c\| = 0 \le 0 = \|M_0^c\|,$$
$$\|N_0^d\| = \|N_0\| \le \|M_0\| = \|M_0^d\|.$$

Now fix  $x^* \in X^*$ . It is enough now to prove that  $\langle N^c, x^* \rangle$  is differentially subordinate to  $\langle M^c, x^* \rangle$ , and that  $\langle N^d, x^* \rangle$  is differentially subordinate to  $\langle M^d, x^* \rangle$ . But this follows from [179, Lemma 1], Theorem 2.2.10 and the fact that  $\langle M^d, x^* \rangle$  and  $\langle N^d, x^* \rangle$  are purely discontinuous processes, and  $\langle M^c, x^* \rangle$  and  $\langle N^c, x^* \rangle$  are continuous processes.

*Proof of Theorem* 4.4.1. By Theorem 4.3.1 there exist martingales  $M^d, M^c, N^d, N^c$ :  $\mathbb{R}_+ \times \Omega \to X$  such that  $M^d$  and  $N^d$  are purely discontinuous,  $M^c$  and  $N^c$  are continuous,  $M_0^c = N_0^c = 0$ , and  $M = M^d + M^c$  and  $N = N^d + N^c$ . By Lemma 4.4.5,  $N^d$  is weakly differentially subordinate to  $M^d$  and  $N^c$  is weakly differentially subordinate to  $M^c$ . Therefore for each  $t \ge 0$ 

$$(\mathbb{E} \|N_t\|^p)^{\frac{1}{p}} \stackrel{(i)}{\leq} (\mathbb{E} \|N_t^d\|^p)^{\frac{1}{p}} + (\mathbb{E} \|N_t^c\|^p)^{\frac{1}{p}} \stackrel{(ii)}{\leq} \beta_{p,X}^2 (\mathbb{E} \|M_t^d\|^p)^{\frac{1}{p}} + \beta_{p,X} (\mathbb{E} \|M_t^c\|^p)^{\frac{1}{p}}$$

$$\stackrel{(iii)}{\leq} \beta_{p,X}^2 (\beta_{p,X} + 1) (\mathbb{E} \|M_t\|^p)^{\frac{1}{p}},$$

where (*i*) holds thanks to the triangle inequality, (*ii*) follows from Theorem 3.3.17 and Theorem 4.4.2, and (*iii*) follows from (4.3.1).  $\Box$ 

*Remark* 4.4.6. It is worth noticing that in a view of recent results the sharp constant in (4.3.1) and (4.3.14) can be derived and equals the  $UMD_p^{\{0,1\}}$ -constant  $\beta_{p,X}^{\{0,1\}}$ . In order to show that this is the right upper bound one needs to use a  $\{0,1\}$ -Burkholder function instead of the Burkholder function, while the sharpness follows analogously Theorem 4.3.5 and 4.3.13. See [188] for details.

*Remark* 4.4.7. In Chapter 5 the existence of the canonical decomposition of a general local martingale together with the corresponding weak  $L^1$ -estimates were shown. Again existence of the canonical decomposition of any *X*-valued martingale is equivalent to *X* having the UMD property.

# EXISTENCE OF THE CANONICAL DECOMPOSI-TION AND WEAK $L^1$ -ESTIMATES

This chapter is based on the paper *On the martingale decompositions of Gundy, Meyer, and Yoeurp in infinite dimensions* by Ivan Yaroslavtsev, see [185].

In this chapter we show that the canonical decomposition (comprising both the Meyer-Yoeurp and the Yoeurp decompositions) of a general X-valued local martingale is possible if and only if X has the UMD property. More precisely, X is a UMD Banach space if and only if for any X-valued local martingale M there exist a continuous local martingale  $M^c$ , a purely discontinuous quasi-left continuous local martingale  $M^q$ , and a purely discontinuous local martingale  $M^a$  with accessible jumps such that  $M = M^c + M^q + M^a$ . The corresponding weak  $L^1$ -estimates are provided. Important tools used in the proof are a new version of Gundy's decomposition of continuous-time martingales and weak  $L^1$ -bounds for a certain class of vector-valued continuous-time martingale transforms.

<sup>2010</sup> Mathematics Subject Classification. 60G44 Secondary: 60G07, 60G57, 60H99, 46N30. *Key words and phrases.* Gundy's decomposition, continuous-time martingales, UMD spaces, canonical decomposition, Meyer-Yoeurp decomposition, Yoeurp decomposition, weak estimates, weak differential subordination.
# **5.1.** INTRODUCTION

It is well-known thanks to the scalar-valued stochastic integration theory that a stochastic integral  $\int \Phi dN$  of a general bounded predictable real-valued process  $\Phi$ with respect to a general real-valued local martingale *N* exists and is well defined (see e.g. Chapter 26 in [89]). Moreover,  $\int \Phi dN$  is a local martingale, so by the Burkholder-Davis-Gundy inequalities one can show the corresponding  $L^p$ estimates for  $p \in (1,\infty)$ :

$$\mathbb{E}\sup_{0\le s\le t}\left|\int_{0}^{s}\Phi\,\mathrm{d}N\right|^{p} \approx_{p} \mathbb{E}\left(\int_{0}^{t}\Phi^{2}\,\mathrm{d}[N]\right)^{\frac{p}{2}}, \quad t\ge 0$$
(5.1.1)

(here  $[N] : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  is a quadratic variation of *N*, see (2.2.3) for the definition). The inequality (5.1.1) together with a Banach fixed point argument play an important rôle in providing solutions to SPDE's with a general martingale noise (see e.g. [48, 72, 73, 89, 126, 177] and references therein). For this reason (5.1.1)-type inequalities for a broader class of *N* and  $\Phi$  are of interest. In particular, one can consider *H*-valued *N* and  $\mathcal{L}(H, X)$ -valued  $\Phi$  for some Hilbert space *H* and Banach space *X*. Building on ideas of Garling [61] and McConnell [119], van Neerven, Veraar, and Weis have shown in [126] that for a special choice of *N* (namely, *N* being a Brownian motion) and a general process  $\Phi$  it is necessary and sufficient that *X* is in the class of so-called *UMD Banach spaces* (see Section 2.3 for the definition) in order to obtain estimates of the form (5.1.1) with the right-hand side replaces by a generalized square function. Later in the paper [175] by Veraar and in the paper [177] by Veraar and the author, inequalities of the form (5.1.1) have been extended to a general continuous martingale *N*, again given that *X* has the UMD property.

Extending (5.1.1) to a general martingale *N* is an open problem, which was solved only for  $X = L^q(S)$  with  $q \in (1,\infty)$  in the recent work [54] by Dirksen and the author. One of the key tools applied therein was the so-called *canonical decomposition* of martingales. The canonical decomposition first appeared in the work [190] by Yoeurp, and partly in the paper [122] by Meyer, and has the following form: an *X*-valued local martingale *M* is said to admit the canonical decomposition if there exists a continuous local martingale  $M^c$ , a purely discontinuous quasileft continuous local martingale  $M^q$  (a "Poisson-like" martingale which does not jump at predictable stopping times), and a purely discontinuous local martingale  $M^a$  with accessible jumps (a "discrete-like" martingale which jumps only at a certain countable set of predictable stopping times) such that  $M_0^c = M_0^q = 0$  a.s. and  $M = M^c + M^q + M^a$ . The canonical decomposition (if it exists) is unique due to the uniqueness in the case  $X = \mathbb{R}$  (see Remark 2.2.19 and 2.4.21). Moreover, when *X* is UMD one has by [184] that for all  $p \in (1,\infty)$ ,

$$\mathbb{E}\|M_t\|^p \approx_{p,X} \mathbb{E}\|M_t^c\|^p + \mathbb{E}\|M_t^q\|^p + \mathbb{E}\|M_t^a\|^p, \ t \ge 0.$$
(5.1.2)

In particular, if *N* is *H*-valued and  $\Phi$  is  $\mathcal{L}(H, X)$ -valued, then

$$\int \Phi \,\mathrm{d}N = \int \Phi \,\mathrm{d}N^c + \int \Phi \,\mathrm{d}N^q + \int \Phi \,\mathrm{d}N^a$$

is the canonical decomposition given that  $N = N^c + N^q + N^a$  is the canonical decomposition, so

$$\mathbb{E}\left\|\int_{0}^{t} \Phi \,\mathrm{d}N\right\|^{p} \approx_{p,X} \mathbb{E}\left\|\int_{0}^{t} \Phi \,\mathrm{d}N^{c}\right\|^{p} + \mathbb{E}\left\|\int_{0}^{t} \Phi \,\mathrm{d}N^{q}\right\|^{p} + \mathbb{E}\left\|\int_{0}^{t} \Phi \,\mathrm{d}N^{a}\right\|^{p}, \quad t \ge 0,$$

which together with Doob's maximal inequality reduces the problem of extending (5.1.1) to the separate cases of  $N^c$ ,  $N^q$  and  $N^a$ . Possible approaches of how to work with  $\int \Phi dN^c$ ,  $\int \Phi dN^q$ , and  $\int \Phi dN^a$  have been provided by [54]: sharp estimates for the first were already obtained in [175, 177] and follow from the similar estimates for a Brownian motion from [126]; the second can be treated by using random measure theory (see Section 2.8), which is an extension of Poisson random measure integration theory (see [51] and [52]); finally, the latter one can be transformed to a discrete martingale by an approximation argument, so the desired  $L^p$ -estimates are nothing more but the *Burkholder-Rosenthal inequalities* (see [29, 54, 161] for details).

The canonical decomposition also plays a significant rôle in obtaining  $L^p$ -estimates for *weakly differentially subordinated* martingales. The weak differential subordination property as a vector-valued generalization of Burkholder's differential subordination property (see [33, 79, 102, 140]) was introduced by the author in [189], and can be described in the following way: an *X*-valued local martingale  $\widetilde{M}$  is weakly differentially subordinate to an *X*-valued local martingale *M* if for each  $x^* \in X^*$  and for each  $t \ge s \ge 0$  a.s.

$$\begin{split} |\langle \widetilde{M}_0, x^* \rangle| &\leq |\langle M_0, x^* \rangle|, \\ [\langle \widetilde{M}, x^* \rangle]_t - [\langle \widetilde{M}, x^* \rangle]_s &\leq [\langle M, x^* \rangle]_t - [\langle M, x^* \rangle]_s. \end{split}$$

If *X* is a UMD Banach space and  $p \in (1,\infty)$ , then applying  $L^p$ -bounds (5.1.2) for the terms of the canonical decomposition together with  $L^p$ -bounds for purely discontinuous (see [189]) and continuous (see [184]) weakly differentially subordinated martingales yields

$$\left(\mathbb{E}\|\widetilde{M}_{\infty}\|^{p}\right)^{\frac{1}{p}} \le c_{p,X}\left(\mathbb{E}\|M_{\infty}\|^{p}\right)^{\frac{1}{p}},\tag{5.1.3}$$

where the best known constant  $c_{p,X}$  equals  $\beta_{p,X}^2(\beta_{p,X} + 1)$  (here  $\beta_{p,X}$  is the  $UMD_p$  *constant of X*, see Section 2.3 for the definition). Sharp estimates for  $c_{p,X}$  in (5.1.3) remain unknown. Moreover, it is an open problem whether one can prove weak  $L^1$ -estimates of the form

$$\lambda \mathbb{P}(\widetilde{M}_{\infty}^* > \lambda) \lesssim_{p, X} \mathbb{E} \| M_{\infty} \|, \ \lambda > 0.$$
(5.1.4)

Here this question is partly solved: we show that (5.1.4) holds for  $\widehat{M}$  being one of the terms of the canonical decomposition of M (see (5.1.5) and (5.4.1)).

The discussion above demonstrates that the canonical decomposition is useful for vector-valued stochastic integration and weak differential subordination, so the following natural question arises: *for which Banach spaces X does every X-valued local martingale have the canonical decomposition?* The paper [184] together with the estimates (5.1.2) provides the answer for  $L^p$ -bounded martingales given  $p \in (1,\infty)$ . Then *X* being a UMD Banach space guarantees such a decomposition.

The present chapter is devoted to providing the definitive answer to this question (see Section 5.4):

**Theorem 5.1.1.** Let X be a Banach space. Then the following are equivalent:

- (*i*) X is a UMD Banach space;
- (ii) every local martingale  $M : \mathbb{R}_+ \times \Omega \to X$  admits the canonical decomposition  $M = M^c + M^q + M^a$ .

*Moreover, if this is the case, then for all*  $t \ge 0$  *and*  $\lambda > 0$ 

$$\begin{split} \lambda \mathbb{P}((M^c)_t^* > \lambda) &\lesssim_X \mathbb{E} \|M_t\|, \\ \lambda \mathbb{P}((M^q)_t^* > \lambda) &\lesssim_X \mathbb{E} \|M_t\|, \\ \lambda \mathbb{P}((M^a)_t^* > \lambda) &\lesssim_X \mathbb{E} \|M_t\|. \end{split}$$
(5.1.5)

Notice that the inequalities (5.1.5) are new even in the real-valued case, even though in that case they are direct consequences of the sharp weak (1,1)-estimates for differentially subordinated martingales proven by Burkholder in [36, 37] (see also [133, 140] for details), from which one can show the following estimates

$$\begin{split} \lambda \mathbb{P}((M^{c})_{t}^{*} > \lambda) &\leq 2\mathbb{E}|M_{t}|, \\ \lambda \mathbb{P}((M^{q})_{t}^{*} > \lambda) &\leq 2\mathbb{E}|M_{t}|, \\ \lambda \mathbb{P}((M^{a})_{t}^{*} > \lambda) &\leq 2\mathbb{E}|M_{t}|. \end{split}$$

The main instrument for proving  $(ii) \Rightarrow (i)$  in Theorem 5.1.1 is Burkholder's characterization of UMD Banach spaces from [30]: *X* is a UMD Banach space if and only if there exists a constant *C* > 0 such that for any *X*-valued discrete martingale  $(f_n)_{n\geq 0}$ , for any sequence  $(a_n)_{n\geq 0}$  with values in  $\{-1, 1\}$  one has that

$$g_{\infty}^* > 1 \text{ a.s.} \implies \mathbb{E} \| f_{\infty} \| > C,$$

where  $(g_n)_{n\geq 0}$  is an X-valued discrete martingale such that

$$g_n - g_{n-1} = a_n (f_n - f_{n-1}), \quad n \ge 1,$$
  

$$g_0 = a_0 f_0,$$
(5.1.6)

and where  $g_{\infty}^* := \sup_{n \ge 0} ||g_n||$ . Using this characterization for a given non-UMD Banach space *X* we construct a martingale  $M : \mathbb{R}_+ \times \Omega \to X$  which does not have the canonical decomposition (see Subsection 5.4.3).

In order to obtain weak  $L^1$ -estimates of the form (5.1.5) together with  $(i) \Rightarrow (ii)$ in Theorem 5.1.1 one needs to use two techniques. The first is the so-called *Gundy decomposition* of martingales. This decomposition was first obtained by Gundy in [71] for discrete real-valued martingales. Later in [41, 79, 116, 147] a more general version of this decomposition for vector-valued discrete martingales was obtained. In Section 5.3 we will present a continuous-time analogue of Gundy's decomposition, which has the following form: an X-valued martingale M can be decomposed into a sum of three martingales  $M^1$ ,  $M^2$ , and  $M^3$ , depending on  $\lambda > 0$ , such that for each  $t \ge 0$ 

- (i)  $||M_t^1||_{L^{\infty}(\Omega;X)} \le 2\lambda, \mathbb{E}||M_t^1|| \le 5\mathbb{E}||M_t||,$
- (ii)  $\lambda \mathbb{P}((M^2)_t^* > 0) \le 4\mathbb{E} ||M_t||,$
- (iii)  $\mathbb{E}(\operatorname{Var} M^3)_t \leq 7\mathbb{E} \|M_t\|,$

where Var *M* is a variation of the path of *M*.

The second important tool is *weak differential subordination martingale transforms*. Discrete martingale transforms were pioneered by Burkholder in [28], where he considered a transform  $(f_n)_{n\geq 0} \mapsto (g_n)_{n\geq 0}$  of a real-valued martingale  $(f_n)_{n\geq 0}$  such that

$$g_n - g_{n-1} = a_n(f_n - f_{n-1}), \ n \ge 1,$$
  
 $g_0 = a_0 f_0$ 

for some  $\{0,1\}$ -valued deterministic sequence  $(a_n)_{n\geq 0}$ . Later in [30, 41, 67, 75, 79, 116] several approaches and generalizations to the vector-valued setting and operatorvalued predictable sequence  $(a_n)_{n\geq 0}$  have been discovered, while the martingale  $(f_n)_{n\geq 0}$  remained discrete. In particular for a very broad class of discrete martingale transforms it was shown that  $L^p$ -boundedness of the transform implies weak  $L^1$ -bounds. In Subsection 5.4.1 (see Theorem 5.4.2) we prove the same assertion for a weak differential subordination martingale transform, i.e. for an operator T acting on continuous-time X-valued local martingales such that TM is weakly differentially subordinate to *M* and  $\{M_{\infty}^* = 0\} \subset \{(TM)_{\infty}^* = 0\}$  for any *X*-valued local martingale M. A particular example of such a martingale transform T is  $M \mapsto TM = M^c$ , where  $M^c$  is the continuous part of M in the canonical decomposition. Due to (5.1.2) this operator is bounded as an operator acting on  $L^p$ -bounded martingales if X is UMD, so by Theorem 5.4.2 the first inequality of (5.1.5) follows. Even though in the case of a discrete filtration such an operator has a classical Burkholder's form (5.1.6) from [30] (with  $(a_n)_{n\geq 0}$  being predictable instead of deterministic, see Proposition 5.4.6 and the remark thereafter), such transforms are of interest since they act on continuous-time martingales, which was not considered before.

### **5.2. PRELIMINARIES**

In the sequel the scalar field is assumed to be  $\mathbb{R}$ , unless stated otherwise.

#### 5.2.1. Martingales and càdlàg processes

We will denote by  $\mathcal{M}^{1,\infty}_X(\mathbb{F})$  the set of all *X*-valued local F-bounded martingales *M* such that

$$\sup_{\lambda>0} \lambda \mathbb{P}(M_{\infty}^* > \lambda) < \infty.$$

In the sequel we will omit  $\mathbb{F}$  from the notations  $\mathscr{M}_X^p(\mathbb{F})$ ,  $\mathscr{M}_X^{p,\mathrm{loc}}(\mathbb{F})$ , and  $\mathscr{M}_X^{1,\infty}(\mathbb{F})$ .

Let  $\tau$  be a stopping time,  $V : \mathbb{R}_+ \times \Omega \to X$  be càdlàg. Throughout this chapter we define  $\Delta V_\tau : \Omega \to X$  in the following way:

$$\Delta V_{\tau} = \begin{cases} V_0, & \tau = 0, \\ V_{\tau} - \lim_{\varepsilon \to 0} V_{0 \lor (\tau - \varepsilon)}, & 0 < \tau < \infty, \\ 0, & \tau = \infty, \end{cases}$$

where  $\lim_{\epsilon \to 0} V_{0 \vee (\tau - \epsilon)}$  exists since *V* has paths with left-hand limits.

One can define the so-called *ucp topology* (uniform convergence on compact sets in probability) on the linear space of all càdlàg adapted *X*-valued processes; convergence in this topology can be characterized in the following way: a sequence  $(V^n)_{n\geq 1}$  of càdlàg adapted *X*-valued processes converges to  $V : \mathbb{R}_+ \times \Omega \to X$  in the ucp topology if for any  $t \geq 0$  and K > 0 we have that

$$\mathbb{P}\left(\sup_{0\le s\le t} \|V_s - V_s^n\| > K\right) \to 0 \quad n \to \infty.$$
(5.2.1)

Then the following proposition holds.

**Proposition 5.2.1.** *The linear space of all càdlàg adapted X-valued processes endowed with the ucp topology is complete.* 

*Proof.* This is just the vector-valued analogue of [155, Theorem 62], for which one needs to apply the vector-valued variation of [154, Problem V.1].  $\Box$ 

We state without proof the following elementary but useful statement.

**Lemma 5.2.2.** Let X be a Banach space,  $(f_n)_{n\geq 1}$ , f be continuous X-valued functions on [0,1] such that  $f_n \to f$  in C([0,1]; X) as  $n \to \infty$ . Then the function  $F : [0,1] \to \mathbb{R}_+$  defined as follows

$$F(t) = \sup_{n} \|f_n(t)\|, \ t \in [0, 1],$$

is continuous.

#### 5.2.2. Compensator and variation

Let *X* be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be an adapted càdlàg process. Then a predictable process  $V : \mathbb{R}_+ \times \Omega$  is called a *predictable compensator* of *M* (or just a *compensator* of *M*) if  $V_0 = 0$  a.s. and if M - V is a local martingale.

The *variation*  $\operatorname{Var} M : \mathbb{R}_+ \times \Omega \to \overline{\mathbb{R}}_+$  of a càdlàg process  $M : \mathbb{R}_+ \times \Omega \to X$  is defined in the following way:

$$(\operatorname{Var} M)_t := \|M_0\| + \limsup_{\text{mesh}\to 0} \sum_{n=1}^N \|M(t_n) - M(t_{n-1})\|,$$
(5.2.2)

where the limit superior is taken over all the partitions  $0 = t_0 < ... < t_N = t$ .

Let  $V : \mathbb{R}_+ \times \Omega \to X$  be a càdlàg adapted process. Analogously to the scalarvalued situation we can define a càdlàg adapted process  $V^* : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  of the following form

$$V_t^* := \sup_{s \in [0,t]} \|V_s\|, t \ge 0.$$

# 5.3. GUNDY'S DECOMPOSITION OF CONTINUOUS-TIME MARTIN-GALES

For the proof of our main results, Theorem 5.4.1 and Theorem 5.4.2, we will need Gundy's decomposition of continuous-time martingales, which is a generalization of Gundy's decomposition of discrete martingales (see [71] and [79, Theorem 3.4.1] for the details).

**Theorem 5.3.1** (Gundy's decomposition). Let *X* be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a martingale. Then for each  $\lambda > 0$  there exist martingales  $M^1$ ,  $M^2 M^3 : \mathbb{R}_+ \times \Omega \to X$  such that  $M = M^1 + M^2 + M^3$  and

- (i)  $\|M_t^1\|_{L^{\infty}(\Omega;X)} \le 2\lambda$ ,  $\mathbb{E}\|M_t^1\| \le 5\mathbb{E}\|M_t\|$  for each  $t \ge 0$ ,
- (*ii*)  $\lambda \mathbb{P}((M^2)_t^* > 0) \le 4\mathbb{E} ||M_t||$  for each  $t \ge 0$ ,
- (*iii*)  $\mathbb{E}(\operatorname{Var} M^3)_t \leq 7\mathbb{E} \|M_t\|$  for each  $t \geq 0$ .

*Remark* 5.3.2. Notice that if *M* is a discrete martingale (i.e.  $M_t = M_{[t]}$  for any  $t \ge 0$ ), then the decomposition in Theorem 5.3.1 turns to the classical discrete one from [79, Theorem 3.4.1].

For the proof we will need the following intermediate steps.

**Lemma 5.3.3.** Let X be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a càdlàg adapted process such that  $\mathbb{E}(\operatorname{Var} M)_t < \infty$  for each  $t \ge 0$  and a.s.

$$M_t = \sum_{0 \le s \le t} \Delta M_s, \quad t \ge 0.$$

*Then M* has a càdlàg predictable compensator  $V : \mathbb{R}_+ \times \Omega \to X$  such that for each  $t \ge 0$ 

$$\mathbb{E}\|V_t\| \le \mathbb{E}(\operatorname{Var} V)_t \le \mathbb{E}(\operatorname{Var} M)_t.$$
(5.3.1)

In particular, if M has a.s. at most one jump, then

$$\mathbb{E}\|V_t\| \le \mathbb{E}(\operatorname{Var} V)_t \le \mathbb{E}(\operatorname{Var} M)_t = \mathbb{E}\|M_t\|.$$
(5.3.2)

*Proof.* Let  $\mu^M$  be a random measure defined on  $\mathbb{R}_+ \times X$  pointwise in  $\omega \in \Omega$  in the following way:

$$\mu^{M}(\omega; B \times A) := \sum_{u \in B} \mathbf{1}_{A \setminus \{0\}}(\Delta M_{u}(\omega)), \quad \omega \in \Omega, B \in \mathscr{B}(\mathbb{R}_{+}), A \in \mathscr{B}(X).$$
(5.3.3)

Notice that  $(\operatorname{Var} M)_t = \sum_{0 \le s \le t} \|\Delta M_s\|$  a.s. for each  $t \ge 0$ , so in particular a.s.

$$(\operatorname{Var} M)_t = \int_{[0,t]\times X} \|x\| \,\mathrm{d}\mu^M(x,s), \quad t \ge 0.$$
 (5.3.4)

Also note that  $\mu^M$  is  $\mathscr{P}$ - $\sigma$ -finite: for each  $0 \le u \le v$  and  $t \ge 0$  one has that

$$\mathbb{E}\int_{[0,t]\times X} \mathbf{1}_{\|x\|\in[u,\nu]} \,\mathrm{d}\mu^M \approx_{u,\nu} \mathbb{E}\int_{[0,t]\times X} \|x\|\mathbf{1}_{\|x\|\in[u,\nu]} \,\mathrm{d}\mu^M$$
$$\leq \mathbb{E}\int_{[0,t]\times X} \|x\| \,\mathrm{d}\mu^M$$
$$= \mathbb{E}(\operatorname{Var} M)_t < \infty.$$

Since  $\mu^M$  is an integer-valued optional  $\mathcal{P}$ - $\sigma$ -finite random measure, it has a predictable compensator  $v^M$  (see Section 2.8 and [85, Theorem II.1.8]), and therefore since by (5.3.4)

$$\mathbb{E}\int_{[0,t]\times X} \|x\| \,\mathrm{d}\mu^M(x,s) = \mathbb{E}(\operatorname{Var} M)_t < \infty,$$

we have that

$$t \mapsto V_t := \int_{[0,t] \times X} x \, \mathrm{d} v^M(x,s), \quad t \ge 0,$$

is integrable and càdlàg in time due to the fact that it is an integral with respect to the measure  $v^M$  a.s. Moreover, by the definition of variation (5.2.2) we have that  $||V_t|| \leq (\text{Var } V)_t$  a.s. for each  $t \geq 0$ , and hence

$$\mathbb{E}\|V_t\| \le \mathbb{E}(\operatorname{Var} V)_t \le \mathbb{E}\int_{[0,t]\times X} \|x\| \,\mathrm{d}\nu^M(x,s) \stackrel{(*)}{=} \mathbb{E}\int_{[0,t]\times X} \|x\| \,\mathrm{d}\mu^M(x,s)$$
$$\stackrel{(**)}{\stackrel{(**)}{=}} \mathbb{E}(\operatorname{Var} M)_t,$$

where (\*) holds due to the definition of a compensator, and (\*\*) follows from (5.3.4). To show (5.3.2) it is sufficient to notice that if *M* has at most one jump then  $(\operatorname{Var} M)_t = ||M_t||$  a.s. for each  $t \ge 0$ .

The following lemma is folklore, but the author could not find an appropriate reference, so we present it with the proof here.

**Lemma 5.3.4.** Let X be a Banach space,  $V : \mathbb{R}_+ \times \Omega \rightarrow X$  be a right-continuous predictable process,  $V_0 = 0$  a.s. Then V is locally bounded.

*Proof.* For each  $n \ge 0$  define a stopping time  $\tau_n := \inf\{t \ge 0 : ||V_t|| \ge n\}$ . Then a sequence  $(\tau_n)_{n\ge 1}$  of stopping times is increasing a.s. and tends to infinity as  $n \to \infty$ . Moreover,  $(\tau_n)_{n\ge 1}$  are predictable by [89, Theorem 25.14] and the fact that for each  $n \ge 1$ 

$$\{\tau \le t\} = \{\sup_{0 \le s \le t} \|V_s\| \ge n\} \in \mathscr{P}.$$
(5.3.5)

Therefore for each  $n \ge 1$  there exists an announcing sequence  $(\tau_{m,n})_{m\ge 1}$  of stopping times. Choose  $m_n$  so that  $\mathbb{P}(\tau_n - \tau_{m_n,n} > \frac{1}{2^n}) < \frac{1}{2^n}$ . Then  $(\tau_{m_n,n})_{n\ge 1}$  is such that  $\tau_{m_n,n} \to \infty$  a.s. as  $n \to \infty$ , and for each  $n \ge 0$  we have that a.s.  $\sup_{0 \le s \le \tau_{m_n,n}} ||V_s|| \le \sup_{0 \le s \le \tau_n} ||V_s|| \le n$ .

Let  $\tau$  and  $\sigma$  be stopping times. Then we can set

$$\tau - \wedge \sigma - := (\tau \wedge \sigma) - . \tag{5.3.6}$$

Notice that if  $M : \mathbb{R}_+ \times \Omega \to X$  is a càdlàg process, then  $(M^{\tau})^{\sigma} = M^{\tau - \wedge \sigma}$ .

*Proof of Theorem* 5.3.1. By a stopping time argument we can assume that *M* is an  $L^1$ -martingale. Define a stopping time  $\tau$  is the following way:

$$\tau = \inf\left\{t \ge 0 : \|M_t\| \ge \frac{\lambda}{2}\right\}.$$

Let  $M^{2,1} := M - M^{\tau}$  and let  $M^{3,1}(\cdot) = \Delta M_{\tau} \mathbf{1}_{[0,\cdot]}(\tau) + M_0^{\tau-}$ , where by (2.4.2) we can conclude that a.s.

$$M_0^{\tau-} := \begin{cases} M_0, & \tau > 0, \\ 0, & \tau = 0. \end{cases}$$
(5.3.7)

Let  $N : \mathbb{R}_+ \times \Omega \to X$  be such that  $N_t = \Delta M_\tau \mathbf{1}_{[0,t]}(\tau), t \ge 0$ . Then due to the fact that  $M_\tau = \mathbb{E}(M_\infty | \mathscr{F}_\tau)$  by [89, Theorem 7.29], [79, Corollary 2.6.30], and the fact that  $||M_{\tau-}|| \le \frac{\lambda}{2}$  a.s., we get

$$\mathbb{E}(\operatorname{Var} N)_{\infty} = \mathbb{E}\|\Delta M_{\tau}\| = \mathbb{E}\|M_{\tau} - M_{\tau-}\| \le \mathbb{E}\|M_{\tau}\| + \mathbb{E}(\|M_{\tau-}\|\mathbf{1}_{\tau<\infty})$$
  
$$\le \mathbb{E}\|M_{\infty}\| + \frac{\lambda}{2} < \infty.$$
 (5.3.8)

Therefore by Lemma 5.3.3, *N* has a compensator *V*. Let

$$\sigma := \inf\{t \ge 0 : \|V_t\| \ge \lambda\}$$

be a stopping time. Then by (5.3.5)  $\sigma$  is a predictable stopping time. Define now  $M^1 = M^{\sigma-\Lambda\tau-} + V^{\sigma-} - M_0^{\tau-}$ ,  $M^{2,2} = (M^{\tau-} + V) - (M^{\sigma-\Lambda\tau-} + V^{\sigma-})$ ,  $M^{3,2} = N - V$  where

 $\sigma - \wedge \tau -$ is defined as in (5.3.6). Define  $M^2 := M^{2,1} + M^{2,2}$  and  $M^3 := M^{3,1} + M^{3,2}$ . Then  $M = M^1 + M^2 + M^3$ . Now let us describe why this is the right choice.

Step 1:  $M^1$ . First show that  $M^1$  is a martingale. Indeed, for each  $t \ge 0$ 

$$M_{t}^{1} = M_{t}^{\sigma - \wedge \tau -} + V_{t}^{\sigma -} - M_{0}^{\tau -} = (M_{t}^{\tau -} + V_{t} - M_{0}^{\tau -})^{\sigma -}$$
  
=  $(M_{t}^{\tau} - \mathbf{1}_{\tau \in [0, t]} \Delta M_{\tau} + V_{t} - M_{0}^{\tau_{2} -})^{\sigma -}$   
=  $((M_{t}^{\tau} - M_{0}^{\tau -}) - (N_{t} - V_{t}))^{\sigma -}$ , (5.3.9)

and the last expression is a martingale due to the fact that  $M^{\tau}$  is a martingale by [89, Theorem 7.12], the fact that N-V is a martingale by the definition of a compensator, Lemma 2.4.4, and the fact that by (5.3.8)

$$\mathbb{E}\|N_{\infty}\| \le \mathbb{E}(\operatorname{Var} N)_{\infty} \le \mathbb{E}\|M_{\infty}\| + \frac{\lambda}{2} < \infty.$$

Now let us check (i):  $||M_{\infty}^{\sigma-\wedge\tau-}||$ ,  $||M_{0}^{\tau-}|| \le \frac{\lambda}{2}$  a.s. by the definition of  $\tau$ , and  $||V_{\infty}^{\sigma-}|| \le \lambda$  by the definition of  $\sigma$ , so  $||M_{\infty}^{1}|| \le 2\lambda$  a.s.

Further, to prove the second part of (i) we will use the representation of  $M^1$  from the last line of (5.3.9). Notice that by [89, Theorem 7.12] and [79, Corollary 2.6.30] for each fixed  $t \ge 0$ 

$$\mathbb{E}\|M_t^{\mathsf{T}}\| \le \mathbb{E}\|M_t\|. \tag{5.3.10}$$

Moreover,

$$\begin{split} \mathbb{E}\|N_t\| &= \mathbb{E}\|M_t^{\mathsf{T}} - M_t^{\mathsf{T}-}\| \leq \mathbb{E}\|M_t^{\mathsf{T}}\| + \mathbb{E}(\|M_t^{\mathsf{T}-}\|\mathbf{1}_{\tau<\infty}) \\ &\leq \mathbb{E}\|M_t^{\mathsf{T}}\| + \mathbb{E}\Big(\frac{\lambda}{2}\mathbf{1}_{\tau<\infty}\Big) \leq 2\mathbb{E}\|M_t^{\mathsf{T}}\| \stackrel{(*)}{\leq} 2\mathbb{E}\|M_t\|, \end{split}$$

where  $||M_t^{\tau-}|| \le \frac{\lambda}{2} \le ||M_t^{\tau}||$  on  $\{\tau < \infty\}$  by the definition of  $\tau$ , and (\*) follows from [89, Theorem 7.12] and [79, Corollary 2.6.30]. Therefore by (5.3.2)

$$\mathbb{E}\|V_t\| \le \mathbb{E}\|N_t\| \le 2\mathbb{E}\|M_t\| \tag{5.3.11}$$

as well. Finally,  $\mathbb{E}||M_0^{\tau-}|| \le \mathbb{E}||M_0|| \le \mathbb{E}||M_t||$  by (5.3.7) and [79, Corollary 2.6.30]. Consequently, the second part of (i) holds by the estimates above and by the triangle inequality.

*Step 2:*  $M^2$ . First note that

$$M^{2} = M - M^{\tau} + (M^{\tau -} + V) - (M^{\tau -} + V)^{\sigma -}.$$
(5.3.12)

Let us check that  $M^2$  is a martingale.  $M - M^{\tau}$  is a martingale by [89, Theorem 7.12]. Furthermore,

$$M^{\tau-} + V = M^{\tau} - (N - V)$$

is a martingale as well due to [89, Theorem 7.12] and the fact that *V* is a compensator of *N*. Finally,  $(M^{\tau} + V)^{\sigma}$  is a martingale by Lemma 2.4.4.

Let us now prove (ii). Notice that by (5.3.12)

$$\mathbb{P}((M^2)_t^* > 0) \le \mathbb{P}((M - M^{\tau})_t^* > 0) + \mathbb{P}(((M^{\tau -} + V) - (M^{\tau -} + V)^{\sigma -})_t^* > 0).$$

First estimate  $\mathbb{P}((M - M^{\tau})_t^* > 0)$ :

$$\mathbb{P}((M - M^{\tau})_t^* > 0) \le \mathbb{P}(\tau \le t) \le \mathbb{P}\left(M_t^* \ge \frac{\lambda}{2}\right) \le \frac{2\mathbb{E}\|M_t\|}{\lambda},$$

where the latter inequality holds by (2.2.2). Using the same machinery we get

$$\begin{split} \mathbb{P}(((M^{\tau-}+V)-(M^{\tau-}+V)^{\sigma-})_t^* > 0) &\leq \mathbb{P}(\sigma \leq t) \\ &= \mathbb{P}(\|V_t\| \geq \lambda) \stackrel{(i)}{\leq} \frac{\mathbb{E}\|V_t\|}{\lambda} \stackrel{(ii)}{\leq} \frac{2\mathbb{E}\|M_t\|}{\lambda}, \end{split}$$

where (*i*) follows from the Chebyshev inequality, and (*ii*) follows from (5.3.11). This terminates the proof of (ii).

*Step 3: M*<sup>3</sup>. Recall that

$$M^3 = M_0^{\tau -} + N - V.$$

Therefore by the triangle inequality a.s. for each  $t \ge 0$ 

$$\mathbb{E}(\operatorname{Var} M^{3})_{t} \leq \mathbb{E} \|M_{0}^{\tau-}\| + \mathbb{E}(\operatorname{Var} N)_{t} + \mathbb{E}(\operatorname{Var} V)_{t}$$
  
$$\leq \mathbb{E} \|M_{t}\| + 2\mathbb{E} \|N_{t}\| \leq 5\mathbb{E} \|M_{t}\|, \qquad (5.3.13)$$

where the latter inequality holds by (5.3.11), while the rest follows from (5.3.1) and the fact that  $\mathbb{E}||M_0^{\tau-}|| \le \mathbb{E}||M_0|| \le \mathbb{E}||M_t||$ .

*Remark* 5.3.5. Let  $p \in (1,\infty)$ , M be an  $L^p$ -bounded martingale,  $\lambda > 0$ ,  $M = M^1 + M^2 + M^3$  be Gundy's decomposition (see the theorem above). Then  $M^1$  is an  $L^p$  martingale since  $||M_t^1||_{L^{\infty}(\Omega;X)} \le 2\lambda$  for all  $t \ge 0$ ;  $M^3$  is a local  $L^p$ -bounded martingale since  $M^3 = M_0^{\tau^-} + N - V$ , where both  $M_0^{\tau^-}$  and  $N_{\infty} = \Delta M_{\tau}$  are  $L^p$ -bounded (the latter is  $L^p$ -bounded by the argument similar to (5.3.8)), and V is locally  $L^p$ -bounded by Lemma 5.3.4; finally,  $M^2$  is a local  $L^p$ -bounded martingale since  $M^2 = M - M^1 - M^3$ . Therefore all the martingales in Gundy's decomposition are locally  $L^p$ -bounded given M is an  $L^p$ -bounded martingale.

#### **5.4.** THE CANONICAL DECOMPOSITION OF LOCAL MARTINGALES

The current section is devoted to the proof of the fact that the canonical decomposition (as well as the Meyer-Yoeurp and the Yoeurp decompositions) of any *X*-valued local martingale exists if and only if *X* has the UMD Banach property. Recall that the Meyer-Yoeurp decomposition split a local martingale *M* into a sum  $M = M^c + M^d$  of a continuous local martingale  $M^c$  and a purely discontinuous local martingale  $M^d$ , while the Yoeurp decomposition split a purely discontinuous local martingale  $M^d$  into a sum  $M^d = M^q + M^a$  of a quasi-left continuous local martingale  $M^q$  and a local martingale  $M^a$  with accessible jumps (see Chapter 2 and 4).

Due to Theorem 4.3.15 any UMD space-valued  $L^p$ -martingale enjoys the canonical decomposition given p > 1. It is a natural question whether the canonical decomposition is possible and whether one can extend (4.3.18) in the case p = 1. It turns out that the UMD property is necessary and sufficient for the canonical decomposition of a general local martingale, while instead of (4.3.18) one gets weaktype estimates:

**Theorem 5.4.1** (Canonical decomposition of local martingales). Let *X* be a Banach space. Then *X* has the UMD property if and only if any local martingale  $M : \mathbb{R}_+ \times \Omega \to X$  has the canonical decomposition  $M = M^c + M^q + M^a$ . If this is the case, then for any  $\lambda > 0$  and  $t \ge 0$ 

$$\begin{split} \lambda \mathbb{P}((M^c)_t^* > \lambda) &\lesssim_X \mathbb{E} \|M_t\|, \\ \lambda \mathbb{P}((M^q)_t^* > \lambda) &\lesssim_X \mathbb{E} \|M_t\|, \\ \lambda \mathbb{P}((M^a)_t^* > \lambda) &\lesssim_X \mathbb{E} \|M_t\|. \end{split}$$
(5.4.1)

For the proof of the main theorem we will need a considerable amount of machinery, which will be provided in Subsection 5.4.1-5.4.3.

#### 5.4.1. Weak differential subordination martingale transforms

The current subsection is devoted to the proof of the fact that boundedness of a continuous-time martingale transform from a certain specific class acting on  $L^{p}$ -bounded martingales implies the corresponding weak  $L^{1}$ -estimates. Such type of assertions for special discrete martingale transforms was first obtained by Burkholder in [28]. Later the Burkholder's original statement was widely generalized in different directions (see [30, 41, 67, 75, 79, 116]), even though the martingale transforms were remaining acting on discrete martingales. The propose of the current subsection is to provide new results for martingale transforms of the same spirit by considering continuous-time martingales. This will allow us to consider linear operators that map a local martingale to the continuous part of the canonical decomposition, or the part of the canonical decomposition which is purely discontinuous with accessible jumps, so weak  $L^{1}$ -estimates (5.4.1) will follow from  $L^{p}$ -estimates (4.3.18) and Theorem 5.4.2.

The following theorem will be an important tool to show Theorem 5.4.1 and it is connected with [79, Proposition 3.5.4]. Recall that  $\mathcal{M}_X^p$  is a space of all  $L^p$ -bounded *X*-valued martingales, and  $\mathcal{M}_X^{p,\text{loc}}$  is a space of all locally  $L^p$ -bounded *X*-valued martingales (see Section 2.2).

**Theorem 5.4.2.** Let X be a Banach space,  $p \in (1,\infty)$ ,  $T : \mathcal{M}_X^{p,\text{loc}} \to \mathcal{M}_X^{p,\text{loc}}$  be a linear operator such that  $TM \stackrel{w}{\ll} M$  and

$$M_{\infty}^* = 0 \Longrightarrow (TM)_{\infty}^* = 0 \quad a.s. \tag{5.4.2}$$

for each  $M \in \mathcal{M}_X^p$ . Assume that  $T \in \mathcal{L}(\mathcal{M}_X^p)$ . Then for any  $M \in \mathcal{M}_X^p$ 

$$\lambda \mathbb{P}(\|(TM)_{\infty}^{*}\| > \lambda) \le C_{p,T,X} \mathbb{E}\|M_{\infty}\|, \ \lambda > 0,$$
(5.4.3)

where  $C_{p,T,X} = 26 \|T\|_{\mathscr{L}(\mathscr{M}^p_X)} \frac{p}{p-1} + 28.$ 

*Remark* 5.4.3. Notice that if *X* is a UMD Banach space, then *T* is automatically bounded on  $\mathcal{M}_X^p$  and  $||T||_{\mathscr{L}(\mathcal{M}_X^p)} \leq \beta_{p,X}^2(\beta_{p,X}+1)$  by (5.1.3) and Theorem 4.4.1 since  $TM \stackrel{w}{\ll} M$  for any  $M \in \mathcal{M}_X^p$ .

For the proof we will need several lemmas.

**Lemma 5.4.4.** Let X be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous martingale with  $M_0 = 0$  a.s. Let  $\mu^M$  be the corresponding random measure defined as in (5.3.3). Assume that

$$\mathbb{E}\sum_{s\geq 0} \|\Delta M_s\| = \mathbb{E}\int_{\mathbb{R}_+\times X} \|x\| \,\mathrm{d}\mu^M < \infty.$$
(5.4.4)

Then  $M_t = \int_{[0,t]\times X} x \,\mathrm{d}\bar{\mu}^M$  for each  $t \ge 0$  a.s.

*Proof.* By (5.4.4) there exists  $N : \mathbb{R}_+ \times \Omega \to X$  such that  $N_t = \sum_{0 \le s \le t} \Delta M_s$  for each  $t \ge 0$ . Let V = N - M. Then both  $t \mapsto N_t - V_t = M_t$ ,  $t \ge 0$ , and

$$t \mapsto N_t - \int_{[0,t] \times X} x \, \mathrm{d} v^M = \int_{[0,t] \times X} x \, \mathrm{d} \mu^M - \int_{[0,t] \times X} x \, \mathrm{d} v^M = \int_{[0,t] \times X} x \, \mathrm{d} \bar{\mu}^M, \ t \ge 0,$$

are martingales. Therefore

$$t \mapsto V_t - \int_{[0,t] \times X} x \,\mathrm{d} v^M = M_t - \int_{[0,t] \times X} x \,\mathrm{d} \bar{\mu}^M, \quad t \ge 0,$$

is a predictable martingale, which is purely discontinuous as a difference of two purely discontinuous martingales (see Lemma 2.8.1). On the other hand it is continuous by the predictability (see e.g. [99, Theorem 4] and [92, Corollary 2.1.42]). Hence by Lemma 2.2.17 this martingale equals zero since it starts at zero, so  $M = N - V = \int_{10.1 \times X} x \, d\bar{\mu}^M$ .

**Lemma 5.4.5.** Let X be a Banach space,  $M, N : \mathbb{R}_+ \times \Omega \to X$  be purely discontinuous martingales such that  $N \stackrel{w}{\ll} M$ . Then  $\mathbb{E}(\operatorname{Var} N)_t \leq 2\mathbb{E}(\operatorname{Var} M)_t$  for each  $t \geq 0$ .

*Proof.* Without loss of generality  $\mathbb{E}(\operatorname{Var} M)_{\infty} < \infty$ . Notice that since  $N \stackrel{w}{\ll} M$ , for a.e.  $(t, \omega) \in \mathbb{R}_+ \times \Omega$  there exists  $a(t, \omega) \in [-1, 1]$  such that  $\Delta N_t(\omega) = a(t, \omega) \Delta M_t(\omega)$  (see Subsection 3.3.2). Therefore a.s. for each  $t \ge 0$ 

$$\int_{[0,t]\times X} \|x\| \, \mathrm{d}\mu^N(x,s) = \sum_{0 \le s \le t} \|\Delta N_s\| = \sum_{0 \le s \le t} |a(s,\cdot)| \|\Delta M_s\|$$
  
$$\leq \sum_{0 \le s \le t} \|\Delta M_s\| \le (\operatorname{Var} M)_t.$$
 (5.4.5)

So by Lemma 5.4.4  $N = \int_{[0,\cdot] \times X} x d\bar{\mu}^N$ , hence

$$(\operatorname{Var} N)_{t} = \left(\operatorname{Var} \int_{[0,\cdot]\times X} x \, \mathrm{d}\mu^{N}(x,s)\right)_{t}$$
  
=  $\left(\operatorname{Var} \left(\int_{[0,\cdot]\times X} x \, \mathrm{d}\mu^{N}(x,s) - \int_{[0,\cdot]\times X} x \, \mathrm{d}\nu^{N}(x,s)\right)\right)_{t}$   
 $\leq \left(\operatorname{Var} \int_{[0,\cdot]\times X} x \, \mathrm{d}\mu^{N}(x,s)\right)_{t} + \left(\operatorname{Var} \int_{[0,\cdot]\times X} x \, \mathrm{d}\nu^{N}(x,s)\right)_{t}$   
 $\leq \int_{[0,t]\times X} \|x\| \, \mathrm{d}\mu^{N}(x,s) + \int_{[0,t]\times X} \|x\| \, \mathrm{d}\nu^{N}(x,s)$   
=  $2\int_{[0,t]\times X} \|x\| \, \mathrm{d}\mu^{N}(x,s) \stackrel{(*)}{\leq} 2(\operatorname{Var} M)_{t},$ 

where (\*) holds by (5.4.5).

*Proof of Theorem* 5.4.2. The proof has the same structure as the proof of [79, Proposition 3.5.16]. Fix  $M \in \mathcal{M}_X^p$  and  $\lambda > 0$ . Let  $K := ||T||_{\mathscr{L}(\mathcal{M}_X^p)}$ ,  $M = M^1 + M^2 + M^3$  be Gundy's decomposition of M from Theorem 5.3.1 at the level  $\alpha\lambda$  for some  $\alpha > 0$  which we will fix later. Notice that all  $M^1$ ,  $M^2$  and  $M^3$  are local  $L^p$ -bounded martingales by Remark 5.3.5. Then

$$\begin{aligned} \mathbb{P}(\|(TM)^*_{\infty}\| > \lambda) \\ &\leq \mathbb{P}(\|(TM^1)^*_{\infty}\| > \frac{\lambda}{2}) + \mathbb{P}(\|(TM^2)^*_{\infty}\| > 0) + \mathbb{P}(\|(TM^3)^*_{\infty}\| > \frac{\lambda}{2}). \end{aligned} (5.4.6) \end{aligned}$$

Let us estimate each of these three terms separately. First,

$$\begin{split} \mathbb{P}(\|(TM^{1})_{\infty}^{*}\| > \frac{\lambda}{2}) \stackrel{(i)}{\leq} \left(\frac{2}{\lambda}\right)^{p} \mathbb{E}\|(TM^{1})_{\infty}^{*}\|^{p} \stackrel{(*)}{\leq} \left(\frac{2}{\lambda}\frac{p}{p-1}\right)^{p} \mathbb{E}\|(TM^{1})_{\infty}\|^{p} \\ \stackrel{(ii)}{\leq} \left(\frac{2K}{\lambda}\frac{p}{p-1}\right)^{p} \mathbb{E}\|M_{\infty}^{1}\|^{p} \leq \left(\frac{2K}{\lambda}\frac{p}{p-1}\right)^{p}\|M_{\infty}^{1}\|_{\infty}^{p-1} \mathbb{E}\|M_{\infty}^{1}\| \\ \stackrel{(iii)}{\leq} \left(\frac{2K}{\lambda}\frac{p}{p-1}\right)^{p} (2\alpha\lambda)^{p-1} 5\mathbb{E}\|M_{\infty}\| = \frac{5\left(4\alpha K\frac{p}{p-1}\right)^{p}}{2\alpha\lambda} \mathbb{E}\|M_{\infty}\|, \end{split}$$

where (*i*) follows from (2.2.2), (\*) follows from Doob's maximal inequality [93, Theorem 1.3.8(iv)], (*ii*) holds by the definition of *K*, and (*iii*) follows from Gundy's decomposition.

Now turn to  $M^2$ . By (5.4.2)

$$\mathbb{P}((TM^{2})_{\infty}^{*} > 0) \le \mathbb{P}((M^{2})_{\infty}^{*} > 0) \le \frac{4}{\alpha\lambda} \mathbb{E} \|M_{\infty}\|.$$
(5.4.7)

Finally, by Lemma 5.4.5 and the fact that  $TM^3 \stackrel{w}{\ll} M^3$  we have that

$$\mathbb{E}(\operatorname{Var} TM^3)_{\infty} \le 2\mathbb{E}(\operatorname{Var} M^3)_{\infty}$$

hence

$$\mathbb{P}(\|(TM^3)^*_{\infty}\| > \frac{\lambda}{2}) \stackrel{(i)}{\leq} \frac{2}{\lambda} \mathbb{E}\|(TM^3)^*_{\infty}\| \le \frac{2}{\lambda} \mathbb{E}(\operatorname{Var} TM^3)_{\infty}$$
$$\stackrel{(ii)}{\leq} \frac{4}{\lambda} \mathbb{E}(\operatorname{Var} M^3)_{\infty} \stackrel{(*)}{\leq} \frac{28}{\lambda} \mathbb{E}\|M_{\infty}\|,$$

where (*i*) follows from (2.2.2), (*ii*) holds by (5.4.7), and (\*) holds by Theorem 5.3.1(iii). Therefore by (5.4.6)

$$\begin{split} \lambda \mathbb{P}(\|(TM)_{\infty}^{*}\| > \lambda) &\leq \lambda \Big( \frac{5 \Big(4\alpha K \frac{p}{p-1}\Big)^{p}}{2\alpha \lambda} + \frac{4}{\alpha \lambda} + \frac{28}{\lambda} \Big) \mathbb{E}\|M_{\infty}\| \\ &= \Big( \frac{5 \Big(4\alpha K \frac{p}{p-1}\Big)^{p}}{2\alpha} + \frac{4}{\alpha} + 28 \Big) \mathbb{E}\|M_{\infty}\|, \end{split}$$

and by choosing  $\alpha = \frac{p-1}{4Kp}$  we get

$$\begin{split} \lambda \mathbb{P}(\|(TM)_{\infty}^{*}\| > \lambda) &\leq \left(10K\frac{p}{p-1} + 16K\frac{p}{p-1} + 28\right) \mathbb{E}\|M_{\infty}\| \\ &= \left(26K\frac{p}{p-1} + 28\right) \mathbb{E}\|M_{\infty}\|, \end{split}$$

which is exactly (5.4.3).

The following proposition shows that the operator *T* from Theorem 5.4.2 has a special structure given the filtration  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$  is generated by  $(\mathscr{F}_n)_{n\geq 0}$ : such martingale transforms are the same as those considered in [79, Proposition 3.5.4] and [30].

**Proposition 5.4.6.** Let X be a separable Banach space. Let the filtration  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$  be of the following form:  $\mathscr{F}_t = \mathscr{F}_{\lfloor t \rfloor}$  for each  $t \geq 0$ , T be as in Theorem 5.4.2. Then there exists an  $(\mathscr{F}_n)_{n\geq 0}$ -predictable sequence  $(a_n)_{n\geq 0}$  with values in [-1,1] such that  $\Delta(TM)_n = a_n \Delta M_n$  a.s. for each  $n \geq 0$  for any  $M \in \mathscr{M}_X^p$ .

*Proof.* Let  $\mathbb{G} = (\mathcal{G}_n)_{n \ge 0} := (\mathcal{F}_n)_{n \ge 0}$  be a discrete filtration. Due to the construction of  $\mathbb{F}$  and the fact that  $\mathbb{G}$  is discrete we have that any  $\mathbb{F}$ -bounded martingale M is in fact discrete (i.e.  $M_t = M_{\lfloor t \rfloor}$  a.s. for each  $t \ge 0$ ), hence any martingale has accessible jumps, so by Lemma 5.4.13 it is sufficient to use the fact that  $TM \stackrel{w}{\ll} M$  for any

 $M \in \mathcal{M}_X^p$  in order to apply Theorem 5.4.2. Let us show that there exists a G-adapted [-1, 1]-valued sequence  $(a_n)_{n\geq 1}$  such that  $\Delta(TM)_n = a_n \Delta M_n$  a.s. for each  $n \geq 0$ . Since X is separable,  $L^p(\Omega; X)$  is separable by [79, Proposition 1.2.29]. Let  $(\xi^m)_{m\geq 1}$  be a dense subset of  $L^p(\Omega; X)$ . For each  $m \geq 1$  we construct a martingale  $M^m$  in the following way:  $M_t^m := \mathbb{E}(\xi^m | \mathscr{F}_t), t \geq 0$ . Then we have that  $((TM)_n^m)_{n\geq 0} \overset{w}{\ll} (M_n^m)_{n\geq 0}$  for each  $m \geq 1$ , so by Subsection 3.3.2 for each  $m \geq 1$  there exists a G-adapted [-1, 1]-valued sequence  $(a_n^m)_{n\geq 0}$  such that  $\Delta(TM^m)_n = a_n^m \Delta M_n^m$  for each  $n \geq 0$ . Let us show that for each  $m_1 \neq m_2$  and  $n \geq 0$  we have that

$$a_n^{m_1} = a_n^{m_2}$$
 a.s. on  $A_n^{m_1, m_2}$ , (5.4.8)

where  $A_n^{m_1,m_2} := \{\Delta M_n^{m_1} \neq 0\} \cap \{\Delta M_n^{m_2} \neq 0\}$ . Let  $((c_1^k, c_2^k))_{k\geq 1}$  be a dense subset of  $\mathbb{R}^2$  such that for each  $k \geq 1$ 

$$c_1^k \Delta M_n^{m_1} + c_2^k \Delta M_n^{m_2} \neq 0$$
 a.s. on  $A_n^{m_1, m_2}$ .

Then  $T(c_1^k M^{m_1} + c_2^k M^{m_2}) \stackrel{w}{\ll} c_1^k M^{m_1} + c_2^k M^{m_2}$  for each  $k \ge 1$ , and hence by the linearity of T we have that for each  $k \ge 1$  a.s.  $c_1^k a_n^{m_1} \Delta M_n^{m_1} + c_2^k a_n^{m_2} \Delta M_n^{m_2}$  and  $c_1^k \Delta M_n^{m_1} + c_2^k \Delta M_n^{m_2}$  are collinear vectors in X, and

$$\left|\frac{c_1^k a_n^{m_1} \Delta M_n^{m_1} + c_2^k a_n^{m_2} \Delta M_n^{m_2}}{c_1^k \Delta M_n^{m_1} + c_2^k \Delta M_n^{m_2}}\right| \le 1 \text{ a.s. on } A_n^{m_1, m_2},$$

by the weak differential subordination. Therefore we can redefine  $A_n^{m_1,m_2}$  up to a negligible set in the following way:

$$\begin{split} A_n^{m_1,m_2} &:= A_n^{m_1,m_2} \bigcap_{k \ge 1} \{ c_1^k \Delta M_n^{m_1} + c_2^k \Delta M_n^{m_2} \neq 0 \} \\ & \bigcap_{k \ge 1} \Big\{ \Big| \frac{c_1^k a_n^{m_1} \Delta M_n^{m_1} + c_2^k a_n^{m_2} \Delta M_n^{m_2}}{c_1^k \Delta M_n^{m_1} + c_2^k \Delta M_n^{m_2}} \Big| \le 1 \Big\}. \end{split}$$

Let us now fix any  $\omega \in A_n^{m_1,m_2}$  and  $\varepsilon > 0$ . Let  $x^* \in X^*$  be such that  $\langle \Delta M_n^{m_1}(\omega), x^* \rangle \neq 0$  and  $\langle \Delta M_n^{m_2}(\omega), x^* \rangle \neq 0$  (such  $x^*$  exists by the Hahn-Banach theorem and the definition of  $A_n^{m_1,m_2}$ ). Then we can find  $k \ge 1$  such that

$$0 < \frac{\langle c_1^k \Delta M_n^{m_1}(\omega) + c_2^k \Delta M_n^{m_2}(\omega), x^* \rangle}{|c_1^k| + |c_2^k|} < \varepsilon$$
(5.4.9)

since  $((c_1^k, c_2^k))_{k\geq 1}$  is dense in  $\mathbb{R}^2$  (i.e.  $k \geq 0$  such that  $(c_1^k, c_2^k)$  is almost orthogonal to  $(\langle \Delta M_n^{m_1}(\omega) x^* \rangle, \langle \Delta M_n^{m_2}(\omega), x^* \rangle))$ . But on the other hand (we will omit  $\omega$  for the convenience of the reader)

$$1 \ge \left| \frac{c_1^k a_n^{m_1} \Delta M_n^{m_1} + c_2^k a_n^{m_2} \Delta M_n^{m_2}}{c_1^k \Delta M_n^{m_1} + c_2^k \Delta M_n^{m_2}} \right| = \frac{|\langle c_1^k a_n^{m_1} \Delta M_n^{m_1} + c_2^k a_n^{m_2} \Delta M_n^{m_2}, x^* \rangle|}{\langle c_1^k \Delta M_n^{m_1} + c_2^k \Delta M_n^{m_2}, x^* \rangle}$$

$$=\frac{|\langle c_{2}^{k}(a_{n}^{m_{2}}-a_{n}^{m_{1}})\Delta M_{n}^{m_{2}},x^{*}\rangle|-|\langle c_{1}^{k}a_{n}^{m_{1}}\Delta M_{n}^{m_{1}}+c_{2}^{k}a_{n}^{m_{1}}\Delta M_{n}^{m_{2}},x^{*}\rangle|}{\langle c_{1}^{k}\Delta M_{n}^{m_{1}}+c_{2}^{k}\Delta M_{n}^{m_{2}},x^{*}\rangle}$$
(5.4.10)  
$$\stackrel{(*)}{\geq}|a_{n}^{m_{2}}-a_{n}^{m_{1}}||\langle \Delta M_{n}^{m_{2}},x^{*}\rangle|\frac{1}{\varepsilon}-1,$$

where (\*) holds by the triangle inequality, (5.4.9), and the fact that  $|a_n^{m_1}| \le 1$ . Since  $\varepsilon$  was arbitrary, (5.4.10) holds true if and only if  $a_n^{m_2}(\omega) - a_n^{m_1}(\omega) = 0$ . Now since  $\omega \in A_n^{m_1,m_2}$  was arbitrary,  $a_n^{m_1} = a_n^{m_2}$  on  $A_n^{m_1,m_2}$ .

Now we define for each  $n \ge 0$  and  $m \ge 1$ :

$$\begin{split} B_n^1 &= \{\Delta M_n^1 \neq 0\},\\ B_n^m &= \{\Delta M_n^m \neq 0\} \setminus B_n^{m-1}, \quad m \ge 2,\\ B_n^0 &= \Omega \setminus \bigcup_{m \ge 1} B_n^m, \end{split}$$

and define  $a_n$  in the following way:

$$a_n(\omega) := a_n^m, \ \omega \in B_n^m, \ m \ge 1,$$
  
 $a_n(\omega) := 0, \ \omega \in B_0^m.$ 
(5.4.11)

Then by (5.4.8)  $a_n = a_n^m$  a.s. on  $\{\Delta M_n^m \neq 0\}$  for all  $m \ge 1$ . Therefore  $\Delta (TM^m)_n = a_n \Delta M_n^m$  a.s. for all  $m \ge 1$ . Now let M be a general  $L^p$ -bounded martingale. Let  $(M^{m_k})_{k\ge 1}$  be a sequence which converges to M in  $\mathcal{M}_X^p$ . Fix  $n \ge 0$ . Then by [79, Corollary 2.6.30]  $\Delta M_n^{m_k}$  converges to  $\Delta M_n$  in  $L^p(\Omega; X)$  as  $k \to \infty$ , so by boundedness of  $a_n$  we have that  $a_n \Delta M_n^{m_k} \to a_n \Delta M_n$  in  $L^p(\Omega; X)$ . On the other hand by boundedness of T and by [79, Corollary 2.6.30]

$$\lim_{k \to \infty} a_n \Delta M_m^{n_k} = \lim_{k \to \infty} \Delta (TM_n^{m_k})_n = \Delta (TM)_n,$$

where the limit is taken in  $L^{p}(\Omega; X)$ . Hence  $\Delta(TM)_{n} = a_{n}\Delta M_{n}$  a.s.

It follows from (5.4.11) and Subsection 3.3.2 that  $(a_n)_{n\geq 0}$  is G-adapted and bounded by 1. Now let us show that  $(a_n)_{n\geq 0}$  is G-predictable. Assume the opposite. Then there exists  $N \geq 0$  such that  $a_N$  is  $\mathscr{F}_N$ -measurable, but not  $\mathscr{F}_{N-1}$ -measurable (here we set  $\mathscr{F}_{-1}$  to be the  $\sigma$ -algebra generated by all negligible sets). Fix  $x \in X \setminus \{0\}$ . Then we can construct the following  $L^p$ -bounded martingale  $M : \mathbb{R}_+ \times \Omega \to X$ :  $\Delta M_n = 0$  if  $n \neq N$  and  $\Delta M_N = (a_N - \mathbb{E}(a_N | \mathscr{F}_{N-1}))x$ . This is an  $L^p$ -bounded martingale since by the triangle inequality and [19, Theorem 34.2]

$$\begin{aligned} \|a_N - \mathbb{E}(a_N | \mathscr{F}_{N-1}) \|_{\infty} &\leq \|a_N\|_{\infty} + \|\mathbb{E}(a_N | \mathscr{F}_{N-1}) \|_{\infty} \leq 1 + \|\mathbb{E}(|a_N| | \mathscr{F}_{N-1}) \|_{\infty} \\ &\leq 1 + \|\mathbb{E}(1 | \mathscr{F}_{N-1}) \|_{\infty} \leq 2. \end{aligned}$$

Then we have that  $\Delta(TM)_N = a_N(a_N - \mathbb{E}(a_N | \mathscr{F}_{N-1}))x$ , and since *TM* is a martingale,

$$0 = \mathbb{E}(\Delta(TM)_N | \mathscr{F}_{N-1}) = \mathbb{E}(a_N (a_N - \mathbb{E}(a_N | \mathscr{F}_{N-1})) x | \mathscr{F}_{N-1})$$

$$\begin{split} &= x \mathbb{E}(a_N^2 - a_N \mathbb{E}(a_N | \mathscr{F}_{N-1}) | \mathscr{F}_{N-1}) \\ &= x \Big( \mathbb{E}(a_N^2 | \mathscr{F}_{N-1}) - \big( \mathbb{E}(a_N | \mathscr{F}_{N-1}) \big)^2 \Big) \\ &= x \mathbb{E}\Big( \Big(a_N - \mathbb{E}(a_N | \mathscr{F}_{N-1}) \Big)^2 \Big| \mathscr{F}_{N-1} \Big), \end{split}$$

so since  $x \neq 0$  and the fact that  $(a_N - \mathbb{E}(a_N | \mathscr{F}_{N-1}))^2$  is nonnegative we get that  $a_N - \mathbb{E}(a_N | \mathscr{F}_{N-1}) = 0$  a.s., hence  $a_N$  is  $\mathscr{F}_{N-1}$ -measurable.

*Remark* 5.4.7. One can extend Proposition 5.4.6 to the case of a Banach space *X* being over the scalar field  $\mathbb{C}$ . The point is that because of the structure of the filtration  $\mathbb{F}$  any  $\mathbb{F}$ -bounded martingale is purely discontinuous, so one can extend the definition of weak differential subordination in the way presented in [188]; namely,  $N \stackrel{w}{\ll} M$  if  $|\langle \Delta N_t, x^* \rangle| \leq |\langle \Delta M_t, x^* \rangle|$  a.s. for all  $t \geq 0$  and  $x^* \in X^*$ . Then by applying the same proof one can show that the sequence  $(a_n)_{n\geq 0}$  from Proposition 5.4.6 exists and is still  $(\mathscr{F}_n)_{n\geq 0}$ -predictable, but it takes values in the unit disk  $\mathbb{D} := \{\lambda \in \mathbb{C} : |\lambda| \leq 1\}.$ 

#### 5.4.2. Sufficiency of the UMD property

Now we will consider two examples of an operator T from Theorem 5.4.2, which will provide us with the Meyer-Yoeurp and the Yoeurp decompositions of any UMD space-valued local martingale.

**Theorem 5.4.8** (Meyer-Yoeurp decomposition of local martingales). Let *X* be a UMD Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a local martingale. Then there exist unique local martingales  $M^c, M^d : \mathbb{R}_+ \times \Omega \to X$  such that  $M^c$  is continuous,  $M^d$  is purely discontinuous,  $M_0^c = 0$ , and  $M = M^c + M^d$ . Moreover, for any  $\lambda > 0$  and  $t \ge 0$ 

$$\begin{split} \lambda \mathbb{P}((M^c)_t^* > \lambda) \lesssim_X \mathbb{E} \|M_t\|, \\ \lambda \mathbb{P}((M^d)_t^* > \lambda) \lesssim_X \mathbb{E} \|M_t\|. \end{split} \tag{5.4.12}$$

For the proof we will need the following lemma.

**Lemma 5.4.9.** Let  $M : \mathbb{R}_+ \times \Omega \to X$  be an  $L^1$ -bounded martingale,  $(M^n)_{n \ge 1}$  be a sequence of purely discontinuous X-valued  $L^1$ -bounded martingales such that  $M^n_{\infty} \to M_{\infty}$  in  $L^1(\Omega; X)$ . Then M is purely discontinuous.

*Proof.* Without loss of generality  $M_0 = 0$  and  $M_0^n = 0$  a.s. for each  $n \ge 1$ . By Proposition 2.2.12 it is sufficient to check that MN is a martingale for any bounded continuous real-valued martingale N with  $N_0 = 0$  a.s. Fix such N. Then due to Proposition 2.2.12  $M^n N$  is a martingale for each  $n \ge 0$ . Moreover, since  $N_t$  is bounded for each  $t \ge 0$ ,  $(M^n N)_t \to (MN)_t$  in  $L^1(\Omega; X)$ . Therefore by the boundedness of a conditional expectation operator (see [79, Corollary 2.6.30]) for each  $0 \le s \le t$ 

$$\mathbb{E}((MN)_t|\mathscr{F}_s) = \mathbb{E}\left(\lim_{n \to \infty} (M^n N)_t | \mathscr{F}_s\right) = \lim_{n \to \infty} \mathbb{E}((M^n N)_t | \mathscr{F}_s)$$

$$= \lim_{n \to \infty} (M^n N)_s = (MN)_s$$

Hence, MN is a martingale. Since N was arbitrary, M is a purely discontinuous martingale.

*Proof of Theorem* 5.4.8. By a stopping time argument we can assume that M is an  $L^1$ -bounded martingale. Fix  $p \in (1,\infty)$ . Let  $(M^n)_{n\geq 1}$  be a sequence of X-valued  $L^p$ -bounded martingales such that  $M_{\infty}^n \to M_{\infty}$  in  $L^1(\Omega; X)$ . Without loss of generality assume that  $\mathbb{E}||M_{\infty} - M_{\infty}^n|| < \frac{1}{2^{n+1}}$  for each  $n \geq 1$ . Let  $T \in \mathcal{L}(\mathcal{M}_X^p)$  be such that T maps an  $L^p$ -bounded martingale  $N : \mathbb{R}_+ \times \Omega \to X$  to its continuous part  $N^c$  (such an operator exists and bounded by Theorem 4.3.15). For each  $n \geq 1$  we denote  $TM^n$  by  $M^{n,c}$ . Then we know that by Theorem 5.4.2 for each  $m \geq n \geq 1$  and any K > 0

$$\mathbb{P}((M^{n,c} - M^{m,c})_{\infty}^* > K) \lesssim_{p,X} \frac{1}{K} \mathbb{E} \|M_{\infty}^{n,c} - M_{\infty}^{m,c}\| \le \frac{1}{2^n K},$$
(5.4.13)

hence  $(M^{n,c})_{n\geq 1}$  is a Cauchy sequence in the ucp topology by (5.2.1). Notice that all the  $M^{n,c'}$ 's are continuous local martingales, which are complete in the ucp topology (see [177, pp. 7–8] and Lemma 5.2.2). Hence there exists a local martingale  $M^c: \mathbb{R}_+ \times \Omega \to X$  which is the limit of  $(M^{n,c})_{n\geq 1}$  in the ucp topology. Now it is sufficient to prove that  $M_0^c = 0$  and that  $\langle M - M^c, x^* \rangle$  is a purely discontinuous local martingale for any  $x^* \in X^*$  in order to show that  $M^c$  is the desired continuous local martingale. Firstly,  $M_0^c = \mathbb{P} - \lim_{n\to\infty} M_0^{n,c} = 0$  since  $M^c$  is the limit of  $(M^{n,c})_{n\geq 1}$ in the ucp topology and since  $M_0^{n,c} = 0$  a.s. for each  $n \ge 1$ . Secondly, since  $M^{n,c} \to$  $M^c$  in the ucp topology for each fixed  $x^* \in X^*$ . Without loss of generality set that  $\mathbb{E}\|M_\infty\|, \mathbb{E}\|M_\infty^n\| \le 1$  for each  $n \ge 1$ . Also by choosing a subsequence we can assume that  $M^{c,n} \to M^c$  a.s. uniformly on compacts. Therefore by Lemma 5.2.2 the process  $t \mapsto \sup_{0 \le s \le t} \sup_n \|M^{c,n}\|$  exists and continuous, and for each  $m \ge 1$  we can define a stopping time  $\tau_m$  in the following way

$$\tau_m := \inf \{ t \ge 0 : \sup_{0 \le s \le t} \sup_n \| M^{c,n} \| \ge m \}.$$

Notice that a.s.  $\tau_m \to \infty$  as  $m \to \infty$ . First show that  $\langle (M-M^c)^{\tau_m}, x^* \rangle$  is purely discontinuous for each  $m \ge 1$ . Note that  $(M^{c,n})^{\tau_m}_{\infty} \to (M^c)^{\tau_m}_{\infty}$  and  $(M^n)^{\tau_m}_{\infty} \to M^{\tau_m}_{\infty}$  in  $L^1(\Omega; X)$  as  $n \to \infty$ . Therefore

$$\langle (M^n - M^{c,n})^{\tau_m}, x^* \rangle \rightarrow \langle (M - M^c)^{\tau_m}, x^* \rangle$$

in  $L^1(\Omega)$ , so by Lemma 5.4.9  $\langle (M - M^c)^{\tau_m}, x^* \rangle$  is purely discontinuous. Notice that by letting *m* to infinity we get that  $\langle M - M^c, x^* \rangle$  is a purely discontinuous local martingale for any  $x^* \in X^*$ , hence  $M - M^c$  is a purely discontinuous local martingale.

The uniqueness of the decomposition follows from Remark 2.2.19, while (5.4.12) holds due to the limiting argument, (5.4.13), and the completeness of  $L^{1,\infty}$ -spaces provided by (1.1.11) and Theorem 1.4.11 in [69].

Let us turn to the Yoeurp decomposition.

**Theorem 5.4.10** (Yoeurp decomposition of local martingales). Let X be a UMD Banach space,  $M^d : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous local martingale. Then there exist unique purely discontinuous local martingales  $M^q, M^a : \mathbb{R}_+ \times \Omega \to X$  such that  $M^q$  is quasi-left continuous,  $M^a$  has accessible jumps,  $M_0^q = 0$ , and  $M^d = M^q + M^a$ . Moreover, for any  $\lambda > 0$  and  $t \ge 0$ 

$$\begin{split} \lambda \mathbb{P}((M^{q})_{t}^{*} > \lambda) &\lesssim_{X} \mathbb{E} \|M_{t}^{d}\|, \\ \lambda \mathbb{P}((M^{a})_{t}^{*} > \lambda) &\lesssim_{X} \mathbb{E} \|M_{t}^{d}\|. \end{split}$$
(5.4.14)

For the proof we will need the following lemmas.

**Lemma 5.4.11.** Let  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a local martingale with accessible jumps,  $M_0 = 0$ a.s. Then  $\{M_{\infty}^* = 0\} = \{[M]_{\infty} = 0\}$  up to a negligible set.

*Proof.* Let  $M = M^c + M^q + M^a$  be the canonical decomposition of M (see Subsection 2.4.3). Then  $M^q = 0$  since M has accessible jumps. By [89, Exercise 17.3]  $\{(M^c)_{\infty}^* = 0\} = \{[M^c]_{\infty} = 0\}$  up to a negligible set. Let us show the same for  $M^a$ . Let  $\tau := \inf\{t \ge 0 : \Delta M_t^a \neq 0\}$  be a stopping time. Notice that a.s.

$$\begin{split} \{\tau < \infty\} &\subset \left\{\sum_{t \ge 0} |\Delta M^a_t| > 0\right\} \subset \{(M^a)^*_\infty > 0\},\\ \{\tau < \infty\} &\subset \left\{\sum_{t \ge 0} |\Delta M^a_t|^2 > 0\right\} = \{[M^a]_\infty > 0\}, \end{split}$$

so we can redefine  $M^a := (M^a)^{\tau}$ . By the definition of  $\tau$  we have that for each  $t \ge 0$ a.s.  $\sum_{0 \le s \le t} |\Delta M_s^a| = |\Delta M_\tau^a| \mathbf{1}_{\tau \le t}$ , hence by [190, Theoreme (1-6).3] a.s.

$$M_t^a = \Delta M_\tau^a \mathbf{1}_{\tau \le t}, \ t \ge 0.$$
 (5.4.15)

Therefore since  $[M^a]_t = |\Delta M^a_\tau|^2 \mathbf{1}_{\tau \le t}$  we have that  $\{(M^a)^*_\infty = 0\} = \{[M^a]_\infty = 0\}$  up to a negligible set.

Let us now show the desired. First notice that by [89, Corollary 26.16] a.s.

$$\{[M]_{\infty} = 0\} = \{[M^{c}]_{\infty} + [M^{a}]_{\infty} = 0\} = \{[M^{c}]_{\infty} = 0\} \cap \{[M^{a}]_{\infty} = 0\}.$$
 (5.4.16)

....

On the other hand a.s.

$$\begin{split} \{M_{\infty}^{*} = 0\} &= \{M_{\infty}^{*} = 0\} \cap \{\Delta M_{t} = 0 \;\forall t \geq 0\} \stackrel{(i)}{=} \{M_{\infty}^{*} = 0\} \cap \{(M^{a})_{\infty}^{*} = 0\} \\ \stackrel{(ii)}{=} \{(M^{c})_{\infty}^{*} = 0\} \cap \{(M^{a})_{\infty}^{*} = 0\} \stackrel{(iii)}{=} \{[M^{c}]_{\infty} = 0\} \cap \{[M^{a}]_{\infty} = 0\} \\ \stackrel{(i\nu)}{=} \{[M]_{\infty} = 0\}, \end{split}$$

where (*i*) holds by (5.4.15), (*ii*) follows from the fact that  $M^c = M - M^a$ , (*iii*) follows from the first half of the proof, and finally (*iv*) follows from (5.4.16).

**Lemma 5.4.12.** Let  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a local martingale,  $M = M^c + M^q + M^a$  be the canonical decomposition. Then up to a negligible set

$$\{M_{\infty}^* = 0\} = \{(M^c)_{\infty}^* = 0\} \cap \{(M^q)_{\infty}^* = 0\} \cap \{(M^a)_{\infty}^* = 0\}.$$
 (5.4.17)

*Proof.* Let  $N := M^c + M^a$ . First notice that by Lemma 5.4.11 and [89, Corollary 26.16] a.s.

$$\{N_{\infty}^{*} = 0\} = \{[N]_{\infty} = 0\} = \{[M^{c}]_{\infty} + [M^{a}]_{\infty} = 0\}$$
$$= \{[M^{c}]_{\infty} = 0\} \cap \{[M^{a}]_{\infty} = 0\}$$
$$= \{(M^{c})_{\infty}^{*} = 0\} \cap \{(M^{a})_{\infty}^{*} = 0\}.$$
(5.4.18)

Let  $\tau := \inf\{t \ge 0 : \Delta M_t \neq 0\}$  be a stopping time. Then a.s.

$$\{\tau < \infty\} \subset \{M_{\infty}^* > 0\} \subset \{N_{\infty}^* > 0\} \cup \{(M^q)_{\infty}^* > 0\}$$

since  $M = N + M^d$ . Let  $A = \{M_{\infty}^* = 0\} \subset \Omega$ . Then  $[M]_{\infty} = [N + M^q]_{\infty} = 0$  a.s. on *A*, and consequently  $[N]_{\infty} = 0$  a.s. on *A* by [89, Corollary 26.16]. Therefore by Lemma 5.4.11  $N_{\infty}^* = 0$  a.s. on *A*, so  $(M^q)_{\infty}^* = 0$  a.s. on *A*, and therefore by (5.4.18)

$$\begin{split} \{M_{\infty}^* = 0\} &= A \subset \{N_{\infty}^* = 0\} \cap \{(M^q)_{\infty}^* = 0\} \\ &= \{(M^c)_{\infty}^* = 0\} \cap \{(M^q)_{\infty}^* = 0\} \cap \{(M^a)_{\infty}^* = 0\}. \end{split}$$

The converse inclusion follows from the fact that  $M = N + M^q$  and (5.4.18).

**Lemma 5.4.13.** Let X be a Banach space,  $M, N : \mathbb{R}_+ \times \Omega \to X$  be local martingales such that N has accessible jumps and  $N \stackrel{w}{\ll} M$ . Then

$$\mathbb{P}(N_t^* > 0) \le \mathbb{P}(M_t^* > 0), \quad t \ge 0.$$
(5.4.19)

*Proof.* (5.4.19) follows from the fact that  $\{M_t^* = 0\} \subset \{N_t^* = 0\}$ . Let  $(x_n^*)_{n \ge 0} \subset X^*$  be a separating set. Then up to a negligible set

$$\begin{split} \{M_t^* = 0\} &= \bigcap_{n \geq 0} \{(\langle M, x_n^* \rangle)_t^* = 0\}, \\ \{N_t^* = 0\} &= \bigcap_{n \geq 0} \{(\langle N, x_n^* \rangle)_t^* = 0\}, \end{split}$$

therefore it is sufficient to consider  $X = \mathbb{R}$ . Let  $M = M^c + M^d + M^a$  be the canonical decomposition of M (see Subsection 2.4.3). By Lemma 5.4.12 and (5.4.18)

$$\{M_t^* = 0\} \subset \{(M^c + M^a)_t^* = 0\}$$

Moreover, by Lemma 5.4.11

$$\{ (M^{c} + M^{a})_{t}^{*} = 0 \} = \{ [M^{c} + M^{a}]_{t} = 0 \} \subset \{ [M]_{t} = 0 \},\$$

$$\{N_t^* = 0\} = \{[N]_t = 0\},\$$

and hence since  $N \ll M$ ,

$$\{M_t^* = 0\} \subset \{[M]_t = 0\} \subset \{[N]_t = 0\} = \{N_t^* = 0\}.$$

*Proof of Theorem* 5.4.10. Without loss of generality assume that  $M^d$  is an  $L^1$ -martingale and  $M_0^d = 0$  a.s. We will divide the proof into two steps.

Step 1. Define a stopping time  $\tau = \{t \ge 0 : ||M_t^d|| > \frac{1}{2}\}$ . In this step we assume that  $M^d = (M^d)^{\tau}$  (i.e. the martingale stops moving after reaching  $\frac{1}{2}$ , in particular after the first jump of absolute value bigger than 1). Let  $\mu^M$  be the random measure defined by (5.3.3),  $v^M$  be the corresponding compensator (see Section 2.8). For each  $n \ge 1$  define a stopping time

$$\tau_n = \inf \left\{ t \ge 0 : \int_{[0,t] \times X} \|x\| \mathbf{1}_{\|x\| > n} \, \mathrm{d} v^{M^d} > 1 \right\},\tag{5.4.20}$$

and a process  $M^{d,n}$ :  $\mathbb{R}_+ \times \Omega \to X$  in the following way

$$M_t^{d,n} = \left( (M^d)_t^{\tau-} + \Delta M_\tau^d \mathbf{1}_{\|\Delta M_\tau^d\| \le n} \mathbf{1}_{\tau \le t} + \int_{[0,t] \times X} x \mathbf{1}_{\|x\| > n} \, \mathrm{d} v^{M^d} \right)^{\tau_n}, \ t \ge 0,$$
(5.4.21)

where we define  $M^{\sigma^-}$  for a stopping time  $\sigma$  in the same way as in (2.4.2). First of all show that  $\tau_n \to \infty$  a.s. as  $n \to \infty$ . Notice that by due to Section 2.8

$$\mathbb{E}\int_{\mathbb{R}_{+}\times X} \|x\|\mathbf{1}_{\|x\|>1} dv^{M^{d}} = \mathbb{E}\int_{\mathbb{R}_{+}\times X} \|x\|\mathbf{1}_{\|x\|>1} d\mu^{M^{d}} \leq \mathbb{E}\|\Delta M_{\tau}^{d}\|$$

$$\leq \mathbb{E}\|M_{\tau}^{d}\| + \mathbb{E}\|M_{\tau-}^{d}\| \stackrel{(*)}{\leq} \mathbb{E}\|M_{\infty}^{d}\| + \frac{1}{2} \stackrel{(**)}{\leq} \infty,$$
(5.4.22)

where (\*) follows from the fact that  $M_{\tau} = M_{\infty}$  and the fact that  $||M_{\tau-}|| \le \frac{1}{2}$  a.s., and (\*\*) holds due to the fact that *M* is an *L*<sup>1</sup>-bounded martingale. Therefore

$$\int_{\mathbb{R}_+\times X} \|x\| \mathbf{1}_{\|x\|>1} \,\mathrm{d} v^{M^d} < \infty \quad \text{a.s.,}$$

so by the monotone convergence theorem a.s.

$$\int_{\mathbb{R}_+\times X} \|x\| \mathbf{1}_{\|x\|>n} \,\mathrm{d} v^{M^d} \to 0, \quad n \to \infty,$$

and hence  $\tau_n \to \infty$  as  $n \to \infty$ .

We need to show that  $M^{d,n}$  is an  $L^{\infty}$ -bounded martingale for each  $n \ge 1$ . Clearly  $M^{d,n}$  is adapted and càdlàg. It is also a local martingale since it can be rewritten in the following form:

$$M_t^{d,n} = (M^d)_t^{\tau_n -} - \int_{[0,t] \times X} x \mathbf{1}_{\|x\| > n} \mathbf{1}_{s < \tau_n} \, \mathrm{d}\bar{\mu}^{M^d}, \quad t \ge 0,$$

where the first term is a martingale by Lemma 2.4.4, and the second term is a local martingale by Lemma 2.8.1 and the fact that the process  $s \mapsto \mathbf{1}_{s < \tau_n}$  is predictable by [89, Theorem 25.14] and the predictability of  $\tau_n$  (the latter follows from (5.4.20) and the predictability of  $v^{M^d}$ , see Section 2.8). Moreover, for each fixed  $t \ge 0$  we have that a.s.

$$\begin{split} \|M_t^{d,n}\| &\leq \|(M^d)_t^{\tau-\wedge\tau_n-}\| + \|\Delta M_\tau^d \mathbf{1}_{\|\Delta M_\tau^d\| \leq n}\| + \int_{[0,\tau_n)\times X} \|x\| \mathbf{1}_{\|x\| > n} v^{M^d} \\ &\leq 1+n+1=n+2. \end{split}$$

(Recall that  $\tau - \wedge \tau_n - := (\tau \wedge \tau_n) -$ , see (5.3.6)). Therefore  $(M^{d,n})_{n \ge 1}$  are bounded martingales.

Now let us now show that  $M^{d,n}_{\infty} \to M^d_{\infty}$  in  $L^1(\Omega; X)$ . First,  $M^{d,n}_{\infty} = M^{d,n}_{\tau_n}$  a.s., so by the triangle inequality

$$\mathbb{E}\|M^d_\infty-M^{d,n}_\infty\|\leq \mathbb{E}\|M^d_\infty-M^d_{\tau_n-}\|+\mathbb{E}\|M^d_{\tau_n-}-M^{d,n}_{\tau_n-}\|$$

Notice that the first term vanishes as  $n \to \infty$  by the fact that  $||M_{\infty}^d - M_{\tau_n}^d|| \le 1 + ||\Delta M_{\tau}||$  a.s., the fact that  $\tau_n \to \infty$  a.s., and the dominated convergence theorem. Let us consider the second term:

$$\begin{split} & \mathbb{E} \| M_{\tau_{n-}}^{d} - M_{\tau_{n-}}^{d,n} \| \\ & = \mathbb{E} \left\| M_{\tau_{n-}}^{d} - (M^{d})_{\tau_{n-}}^{\tau} - \Delta M_{\tau}^{d} \mathbf{1}_{\|\Delta M_{\tau}^{d}\| \leq n} \mathbf{1}_{\tau < \tau_{n}} - \int_{[0,\tau_{n}) \times X} x \mathbf{1}_{\|x\| > n} \, \mathrm{d} v^{M^{d}} \right\| \\ & = \mathbb{E} \left\| \Delta M_{\tau}^{d} \mathbf{1}_{\tau < \tau_{n}} - \Delta M_{\tau}^{d} \mathbf{1}_{\|\Delta M_{\tau}^{d}\| \leq n} \mathbf{1}_{\tau < \tau_{n}} - \int_{[0,\tau_{n}) \times X} x \mathbf{1}_{\|x\| > n} \, \mathrm{d} v^{M^{d}} \right\| \\ & = \mathbb{E} \left\| \Delta M_{\tau}^{d} \mathbf{1}_{\|\Delta M_{\tau}^{d}\| > n} \mathbf{1}_{\tau < \tau_{n}} - \int_{[0,\tau_{n}) \times X} x \mathbf{1}_{\|x\| > n} \, \mathrm{d} v^{M^{d}} \right\| \\ & = \mathbb{E} \left\| \int_{[0,\tau_{n}) \times X} x \mathbf{1}_{\|x\| > n} \, \mathrm{d} \mu^{M^{d}} - \int_{[0,\tau_{n}) \times X} x \mathbf{1}_{\|x\| > n} \, \mathrm{d} v^{M^{d}} \right\| \\ & = \mathbb{E} \left\| \int_{[0,\tau_{n}) \times X} \| x \| \mathbf{1}_{\|x\| > n} \, \mathrm{d} \mu^{M^{d}} + \mathbb{E} \int_{[0,\tau_{n}) \times X} \| x \| \mathbf{1}_{\|x\| > n} \, \mathrm{d} v^{M^{d}} \right\| \\ & \leq \mathbb{E} \int_{[0,\tau_{n}) \times X} \| x \| \mathbf{1}_{\|x\| > n} \, \mathrm{d} \mu^{M^{d}} + \mathbb{E} \int_{[0,\tau_{n}) \times X} \| x \| \mathbf{1}_{\|x\| > n} \, \mathrm{d} v^{M^{d}} \end{split}$$

and the last expression vanishes as  $n \to \infty$  by the monotone convergence theorem. (Notice that (\*) follows from the definition of a compensator and from (5.4.22), while (\*\*) follows from the fact that  $||\Delta M_t|| \ge 1$  only if  $t = \tau$  by the assumptions on M.)

Since each of  $M^{d,n'}$ s is an  $L^p$ -bounded martingale for each  $p \in (1,\infty)$ , by Theorem 4.3.15 for each  $n \ge 1$  there exists the Yoeurp decomposition  $M^{d,n} = M^{q,n} + M^{a,n}$  of a martingale  $M^{d,n}$  into a sum of two purely discontinuous martingales  $M^{q,n}, M^{a,n} : \mathbb{R}_+ \times \Omega \to X$  such that  $M^{q,n}$  is quasi-left continuous,  $M^{a,n}$  has accessible jumps, and  $M_0^{q,n} = M_0^{a,n} = 0$  a.s. (recall that  $M_0^{d,n} = 0$  a.s.). Fix some  $p \in (1,\infty)$ .

Since an operator  $T^q$  that maps an  $L^p$ -bounded martingale  $M : \mathbb{R}_+ \times \Omega \to X$  to its purely discontinuous quasi-left continuous part  $M^q$  of the canonical decomposition is continuous on  $\mathscr{L}(\mathscr{M}^p_X)$  by Theorem 4.3.15, Theorem 5.4.2 together with Lemma 5.4.13 yields that for each  $m, n \ge 1$  and K > 0

$$\mathbb{P}\left((M^{q,n} - M^{q,m})_{\infty}^{*} > K\right) \lesssim_{p} \frac{1}{K} \mathbb{E} \|M_{\infty}^{d,n} - M_{\infty}^{d,m}\|$$
$$\leq \frac{1}{K} (\mathbb{E} \|M_{\infty}^{d,n} - M_{\infty}^{d}\| + \mathbb{E} \|M_{\infty}^{d,m} - M_{\infty}^{d}\|).$$

so  $(M^{q,n})_{n\geq 1}$  is a Cauchy sequence in the ucp topology. By Proposition 5.2.1 it has a càdlàg adapted limit. Denote this limit by  $M^q$ . Let us show that  $M^q$  is a purely discontinuous quasi-left continuous local martingale. Let  $\sigma$  be a predictable time. Then  $\Delta M_{\sigma}^{q,n} = 0$  a.s., and for any  $t \geq 0$  a.s.

$$\begin{split} \sup_{0 \le s \le t} \|M_{s}^{q,n} - M_{s}^{q}\| &\geq \mathbf{1}_{\sigma \le t} \sup_{0 \le s \le \sigma} \|M_{s}^{q,n} - M_{s}^{q}\| \\ &\geq \mathbf{1}_{\sigma \le t} \Big( \sup_{m \ge 1} \|M_{0 \lor \sigma - \frac{1}{m}}^{q,n} - M_{0 \lor \sigma - \frac{1}{m}}^{q} \| \lor \|M_{\sigma}^{q,n} - M_{\sigma}^{q}\| \Big) \\ &\geq \frac{1}{2} \mathbf{1}_{\sigma \le t} \Big( \limsup_{m \ge 1} \|M_{0 \lor \sigma - \frac{1}{m}}^{q,n} - M_{0 \lor \sigma - \frac{1}{m}}^{q} \| + \|M_{\sigma}^{q,n} - M_{\sigma}^{q}\| \Big) \\ &= \frac{1}{2} \mathbf{1}_{\sigma \le t} \Big( \|M_{\sigma^{-}}^{q,n} - M_{\sigma^{-}}^{q}\| + \|M_{\sigma}^{q,n} - M_{\sigma}^{q}\| \Big) \\ &\stackrel{(*)}{\geq} \frac{1}{2} \mathbf{1}_{\sigma \le t} \|M_{\sigma^{-}}^{q,n} - M_{\sigma^{-}}^{q} - M_{\sigma}^{q,n} + M_{\sigma}^{q}\| \\ &\geq \frac{1}{2} \mathbf{1}_{\sigma \le t} \|\Delta M_{\sigma^{-}}^{q} - \Delta M_{\sigma}^{q,n}\| = \frac{1}{2} \mathbf{1}_{\sigma \le t} \|\Delta M_{\sigma^{-}}^{q}\|, \end{split}$$

where (\*) follows from the triangle inequality. Since

$$\mathbb{P} - \lim_{n \to \infty} \sup_{0 \le s \le t} \|M_s^{q,n} - M_s^q\| = 0,$$

we have that for each  $t \ge 0$ 

$$\mathbb{P} - \lim_{n \to \infty} \mathbf{1}_{\sigma \le t} \|\Delta M_{\sigma^-}^q\| = 0.$$

But the expression under the limit in probability does not depend on *n*. Hence  $\mathbf{1}_{\sigma \leq t} \|\Delta M_{\sigma}^{q}\| = 0$  a.s. By letting  $t \to \infty$  we get that a.s.  $\|\Delta M_{\sigma}^{q}\| = 0$ , and since  $\sigma$  was arbitrary predictable,  $M^{q}$  is quasi-left continuous.

Let now  $\sigma$  be a totally inaccessible stopping time. Let us show that a.s.

$$\Delta M^q_\sigma = \Delta M^d_\sigma. \tag{5.4.24}$$

First notice that for each fixed  $m \ge n \ge 1$ 

$$\Delta M_{\sigma}^{q,m} \mathbf{1}_{\sigma < \tau \wedge \tau_{n}} \stackrel{(*)}{=} \Delta M_{\sigma}^{d,m} \mathbf{1}_{\sigma < \tau \wedge \tau_{n}} \stackrel{(**)}{=} \Delta M_{\sigma}^{d} \mathbf{1}_{\sigma < \tau \wedge \tau_{n}},$$
  
$$\Delta M_{\sigma}^{q,m} \mathbf{1}_{\sigma = \tau < \tau_{n}} \mathbf{1}_{\|\Delta M_{\tau}^{d}\| \le n} \stackrel{(*)}{=} \Delta M_{\sigma}^{d,m} \mathbf{1}_{\sigma = \tau < \tau_{n}} \mathbf{1}_{\|\Delta M_{\tau}^{d}\| \le n} \stackrel{(**)}{=} \Delta M_{\sigma}^{d} \mathbf{1}_{\sigma = \tau < \tau_{n}} \mathbf{1}_{\|\Delta M_{\tau}^{d}\| \le n},$$
  
(5.4.25)

where (\*) follows from Remark 2.4.23, and (\*\*) follows from the definition (5.4.21) of  $M^{d,m}$  and Lemma 2.4.3. Therefore by (5.4.23) applied for our  $\sigma$  a.s. for each  $n \ge 1$ 

$$\Delta M_{\sigma}^{d} \mathbf{1}_{\sigma < \tau \wedge \tau_{n}} = \Delta M_{\sigma}^{q} \mathbf{1}_{\sigma < \tau \wedge \tau_{n}},$$

$$\Delta M_{\sigma}^{d} \mathbf{1}_{\sigma = \tau < \tau_{n}} \mathbf{1}_{\|\Delta M_{\tau}^{d}\| \le n} = \Delta M_{\sigma}^{q} \mathbf{1}_{\sigma = \tau < \tau_{n}} \mathbf{1}_{\|\Delta M_{\tau}^{d}\| \le n}.$$
(5.4.26)

By letting  $n \to \infty$  we get (5.4.24).

Let us show that  $M^q$  is locally integrable. For each  $l \ge 1$  set  $\rho_l := \inf\{t \ge 0 : ||M_t^q|| \ge l\}$ . Then a.s. for each  $t \ge 0$ 

$$\begin{split} \|(M^{q})_{t}^{\rho_{l}}\| &\leq \|(M^{q})_{t-}^{\rho_{l}}\| + \|\Delta(M^{q})_{t}^{\rho_{l}}\| \leq l + \|\Delta(M^{q})_{t}^{\rho_{l}}\|\mathbf{1}_{t=\tau} + \|\Delta(M^{q})_{t}^{\rho_{l}}\|\mathbf{1}_{t<\tau} \\ &\leq l + \|\Delta M_{\tau}^{d}\| + 1. \end{split}$$

Therefore

$$\mathbb{E}\|(M^q)_t^{\rho_l}\| \le l+1 + \mathbb{E}\|\Delta M^d_\tau\| < \infty,$$

where  $\mathbb{E}||\Delta M_{\tau}|| < \infty$  by (5.4.22). Since  $M^q$  is càdlàg, by [154, Problem V.1] we have that  $\rho_l \to \infty$  as  $l \to \infty$ , so  $M^q$  is locally integrable.

Now let us show that  $M^q$  is a local martingale. Let  $(M^{q,n_k})_{k\geq 1}$  be a subsequence of  $(M^{q,n})_{n\geq 1}$  such that  $M^{q,n_k} \to M^q$  uniformly on compacts a.s. (existence of such a subsequence can be shown e.g. as in the proof of [155, Theorem 62]). It is sufficient to show that  $M^{\rho_l \wedge \tau_{n_k}-}$  is a local martingale for each  $l, k \geq 1$  since  $\rho_l \to \infty$  and  $\tau_{n_k} \to \infty$ a.s. as  $l, k \to \infty$ . Fix K > 0. Then by (5.4.25) and (5.4.26) for each  $k \geq K$  we have that a.s. for each  $t \geq 0$ 

$$\Delta(M^{q,n_k})_t^{\tau_{n_K}-\wedge\tau-} = \Delta(M^q)_t^{\tau_{n_K}-\wedge\tau-}.$$

Therefore by Lemma 5.2.2 there exists a continuous adapted process  $N : \mathbb{R}_+ \times \Omega \rightarrow \mathbb{R}_+$  such that a.s.

$$N_t = \sup_{k \ge K} \left\| (M^{q, n_k})_t^{\tau_{n_k} - \wedge \tau -} - (M^q)_t^{\tau_{n_k} - \wedge \tau -} \right\|, \quad t \ge 0.$$

Now for each  $j \ge 1$  define a stopping time  $\sigma_j = \inf\{t \ge 0 : N_t \ge j\}$ . Fix  $j \ge 1$ . Then for each  $t \ge 0$  we have that for any  $k \ge K$  a.s.

$$\left\| \left( M^{q,n_k} \right)_t^{\rho_l \wedge \tau_{n_K} - \wedge \sigma_j} - \left( M^q \right)_t^{\rho_l \wedge \tau_{n_K} - \wedge \sigma_j} \right\| \leq j + l + 2 \| \Delta M^d_\tau \|$$

and that  $(M^{q,n_k})_t^{\rho_l \wedge \tau_{n_k} - \wedge \sigma_j} - (M^q)_t^{\rho_l \wedge \tau_{n_k} - \wedge \sigma_j} \to 0$  a.s. as  $k \to \infty$ . Hence by the dominated convergence theory

$$(M^{q,n_k})_t^{\rho_l \wedge \tau_{n_k} - \wedge \sigma_j} \to (M^q)_t^{\rho_l \wedge \tau_{n_k} - \wedge \sigma_j} \text{ in } L^1(\Omega; X) \text{ as } k \to \infty.$$

Consequently,  $((M^q)_t^{\rho_l \wedge \tau_{n_K} - \wedge \sigma_j})_{t \ge 0}$  is an  $L^1$ -bounded martingale, which is moreover purely discontinuous by Lemma 5.4.9. By letting  $l, K, j \to \infty$  we get that  $M^q$  is a purely discontinuous quasi-left continuous local martingale.  $M^a$  can be constructed in the same way. The identity  $M^d = M^q + M^a$  follows from the following limiting argument:

$$\begin{split} M^{d} &= ucp - \lim_{n \to \infty} M^{d,n}, \\ M^{q} &= ucp - \lim_{n \to \infty} M^{q,n}, \\ M^{a} &= ucp - \lim_{n \to \infty} M^{a,n}, \end{split}$$

and the fact that  $M^{d,n} = M^{q,n} + M^{a,n}$  for each  $n \ge 1$ .

Step 2. For a general martingale  $M^d$  we construct a sequence of stopping times  $\tau_n = \inf\{t \ge 0 : \|M_t^d\| \ge \frac{n}{2}\}$ . For each  $M^{d,n} := (M^d)^{\tau_n}$  we construct the corresponding  $M^{q,n}$  by Step 1. Then for each  $m \ge n \ge 1$  we get that  $(M^{q,n})^{\tau_m} = M^{q,m}$  since for any  $x^* \in X^*$  a.s.

$$\langle (M^{q,n})^{\tau_m}, x^* \rangle = \langle M^{q,m}, x^* \rangle$$

due to the uniqueness of the Yoeurp decomposition in the real-valued case. Then we just set  $M_0^q := 0$  and

$$M_t^q := \sum_{n \ge 1} M_t^{q,n} \mathbf{1}_{t \in (\tau_{n-1}, \tau_n]}, \quad t \ge 0,$$

where  $\tau_0 \equiv 0$ . The obtained  $M^q$  will be the desired purely discontinuous quasi-left continuous local martingale.

We can construct  $M^a$  in the same way and show that then  $M^d = M^q + M^a$  similarly to how it was shown in step 1.

The uniqueness of the decomposition follows from Remark 2.4.21, while (5.4.14) follows analogously (5.4.12).  $\hfill \Box$ 

*Proof of Theorem* 5.4.1 (*sufficiency of UMD and* (5.4.1)). Sufficiency of the UMD property follows from Theorem 5.4.8 and Theorem 5.4.10, while (5.4.1) follows in the same way as (5.4.12) and (5.4.14).  $\Box$ 

#### 5.4.3. Necessity of the UMD property

In the current subsection we show that the UMD property is necessary in Theorem 5.4.8 and Theorem 5.4.10, and hence it is necessary for the canonical decomposition of a local martingale.

**Theorem 5.4.14.** Let X be a Banach space that does not have the UMD property. Then there exists a filtration  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$  and an  $\mathbb{F}$ -bounded martingale  $M : \mathbb{R}_+ \times \Omega \to X$  such that M provides neither the Meyer-Yoeurp nor the canonical decomposition.

For the proof we will need the following lemma which is a modification of the statements from p. 1001 and p. 1004 of [30]. Recall that if  $(f_n)_{n\geq 0}$  is an *X*-valued martingale, the we define  $df_n := f_n - f_{n-1}$  for  $n \geq 1$  and  $df_0 := f_0$ .

**Lemma 5.4.15.** Let X be a Banach space. Then X is a UMD Banach space if and only if there exists a constant C > 0 such that for any X-valued discrete martingale  $(f_n)_{n\geq 0}$ , for any  $\{0,1\}$ -valued sequence  $(a_n)_{n\geq 0}$  one has that

$$g_{\infty}^* > 1 \text{ a.s.} \implies \mathbb{E} \| f_{\infty} \| > C,$$

where  $(g_n)_{n\geq 0}$  is an X-valued discrete martingale such that  $dg_n = a_n df_n$  for each  $n \geq 0$ ,  $g_{\infty}^* := \sup_{n\geq 0} \|g_n\|.$ 

*Proof.* One needs to modify [30, Theorem 2.1] in such a way that  $dg_n = a_n df_n$  for some  $a_n \in \{0, 1\}$  for each  $n \ge 0$ . Then the proof is the same, and the desired statement follows from the equivalence of [30, (2.3)] and [30, (2.4)].

**Corollary 5.4.16.** Let X be a Banach space that does not have the UMD property. Then there exists an X-valued Paley-Walsh  $L^1$ -bounded martingale  $(f_n)_{n\geq 0}$  and a  $\{0,1\}$ -valued sequence  $(a_n)_{n\geq 0}$  such that  $\mathbb{P}(g_{\infty}^* = \infty) = 1$ , where  $(g_n)_{n\geq 0}$  is an X-valued martingale such that  $dg_n = a_n df_n$  for each  $n \geq 0$ .

*Proof.* Without loss of generality all the martingales used below are Paley-Walsh (see [79, Theorem 3.6.1]), so the resulting martingale will be Paley-Walsh as well. By Lemma 5.4.15 we can find  $N_1 > 0$ , an *X*-valued martingale  $f^1 = (f_n^1)_{n=0}^{N_1}$  and a {0,1}-valued sequence  $(a_n^1)_{n=0}^{N_1}$  such that  $\mathbb{E} \| f_{N_1}^1 \| < \frac{1}{2}$  and

$$\mathbb{P}((g^1)_{N_1}^*>1)>\frac{1}{2},$$

where  $g^1 = (g_n^1)_{n=0}^{N_1}$  is such that  $dg_n^1 = a_n^1 df_n^1$  for each  $n = 0, ..., N_1$ . Now inductively for each k > 1 we find  $N_k > 0$  and an *X*-valued Paley-Walsh martingale  $f^k = (f_n^k)_{n=0}^{N_k}$  independent of  $f^1, ..., f^{k-1}$  such that  $\mathbb{E} \| f_{N_k}^k \| < \frac{1}{2^k}$  and

$$\mathbb{P}((g^k)_{N_k}^* > 2C_k) > 1 - \frac{1}{2^k},$$

where  $g^k = (g_n^k)_{n=0}^{N_1}$  is such that  $dg_n^k = a_n^k df_n^k$  for each  $n = 0, ..., N_k$ , and  $C_k > 2^k$  is such that

$$\mathbb{P}((g^1)_{N_1}^* + \ldots + (g^{k-1})_{N_{k-1}}^* > C_k) < \frac{1}{2^k}.$$

Without loss of generality assume that  $f_0^k = 0$  a.s. for each  $k \ge 1$ . Now construct a martingale  $(f_n)_{n\ge 0}$  and a {0,1}-valued sequence  $(a_n)_{n\ge 0}$  in the following way:  $f_0 = a_0 = 0$  a.s.,  $df_n = df_m^k$  and  $a_n = a_m^k$  if  $n = N_1 + \dots + N_{k-1} + m$  for some  $k \ge 1$  and  $1 \le m \le N_k$ . Then  $(f_n)_{n\ge 0}$  is well-defined,

$$\lim_{n \to \infty} \mathbb{E} \|f_n\| = \mathbb{E} \|f_\infty\| \le \sum_{k \ge 1} \mathbb{E} \|f_{N_k}^k\| \le 1$$

by the triangle inequality, and for an *X*-valued martingale  $(g_n)_{n\geq 0}$  with  $dg_n = a_n df_n$  for each  $n \geq 0$ , for each  $k \geq 2$ 

$$\mathbb{P}(g_{N_1+\dots+N_k}^* > C_k) \ge \mathbb{P}((g^k)_{N_k}^* > 2C_k, (g^1)_{N_1}^* + \dots + (g^{k-1})_{N_{k-1}}^* \le C_k) > 1 - \frac{1}{2^{k-1}},$$

hence  $g_{\infty}^* = \infty$  a.s.

*Proof of Theorem* 5.4.14. By Corollary 5.4.16 we can construct a discrete filtration  $\mathbb{G} = (\mathcal{G}_n)_{n \ge 0}$  and an *X*-valued *L*<sup>1</sup>-bounded Paley-Walsh G-bounded martingale  $(f_n)_{n \ge 0}$  such that

$$\mathbb{E}\|f_{\infty}\| = \lim_{n \to \infty} \mathbb{E}\|f_n\| \le 1, \tag{5.4.27}$$

and such that there exists  $\{0, 1\}$ -valued sequence  $(a_n)_{n \ge 0}$  so that

$$\mathbb{P}(g_{\infty}^*=\infty)=1,$$

where  $(g_n)_{n\geq 0}$  is an *X*-valued martingale with  $dg_n = a_n df_n$  for each  $n \geq 0$ .

Since  $(f_n)_{n\geq 0}$  is Paley-Walsh, there exist a sequence  $(r_n)_{n\geq 0}$  of independent Rademacher variables, a sequence of functions  $(\phi_n)_{n\geq 1}$  with  $\phi_1 \in X$  and  $\phi_n : \{-1, 1\}^{n-1} \to X$  for each  $n \geq 2$ , so that  $df_n = r_n \phi_n(r_1, ..., r_{n-1})$  a.s. for each  $n \geq 1$ .

Now our goal is to construct a continuous-time *X*-valued martingale *M* which does not have the Meyer-Yoeurp decomposition (and hence the canonical decomposition) using  $(f_n)_{n\geq 0}$ . Let us first construct a filtration  $\mathbb{F} = (\mathscr{F}_t)_{t\geq 0}$  on  $\mathbb{R}_+$  in the following way. By Lemma 4.3.7 for each  $n \geq 0$  we can find a continuous martingale  $M^n : [0, \frac{1}{2^{n+1}}] \times \Omega \to \mathbb{R}$  with a symmetric distribution such that  $M_0^n = 0$  a.s.,  $\left| M_{\frac{1}{2^{n+1}}}^n \right| \leq 1$  a.s.,

$$\mathbb{P}\left(M^{n}_{\frac{1}{2^{n+1}}}=0\right)=0,$$
(5.4.28)

and

$$\mathbb{P}\left(M_{\frac{1}{2^{n+1}}}^{n} \neq \operatorname{sign} M_{\frac{1}{2^{n+1}}}^{n}\right) < \frac{1}{2^{n}(\|\phi_{n}\|_{\infty}+1)}.$$
(5.4.29)

Let  $(\tilde{r}_n)_{n\geq 0}$  be a sequence of independent Rademacher random variables. Without loss of generality assume that all  $(\tilde{r}_n)_{n\geq 0}$  and  $(M^n)_{n\geq 0}$  are independent. Then set  $\mathscr{F}_0$  to be the  $\sigma$ -algebra generated by all negligible sets, and set

$$\mathcal{F}_{t} := \begin{cases} \mathcal{F}_{1-\frac{1}{2^{n}}}, & t \in (1-\frac{1}{2^{n}}, 1-\frac{1}{2^{n+1}}), a_{n} = 0, n \ge 0, \\ \sigma(\mathcal{F}_{1-\frac{1}{2^{n}}}, \tilde{r}_{n}), & t = 1-\frac{1}{2^{n+1}}, a_{n} = 0, n \ge 0, \\ \sigma(\mathcal{F}_{1-\frac{1}{2^{n}}}, (M_{s}^{n})_{s \in [0, t-1-\frac{1}{2^{n}}]}), & t \in (1-\frac{1}{2^{n}}, 1-\frac{1}{2^{n+1}}], a_{n} = 1, n \ge 0, \\ \sigma(\mathcal{F}_{s}: s \in [0, 1)), & t \ge 1. \end{cases}$$

Let  $(\sigma_n)_{n\geq 0}$  be a sequence of independent Rademacher variables such that  $\sigma_n = \tilde{r}_n$  if  $a_n = 0$  and  $\sigma_n = \text{sign} M_{\frac{1}{2^{n+1}}}^n$  if  $a_n = 1$  (in the latter case  $\sigma_n$  has the Rademacher distribution by (5.4.28) and the fact that  $M_{\frac{1}{2^{n+1}}}^n$  is symmetric). Now construct M:

 $\mathbb{R}_+ \times \Omega \to X$  in the following way:

$$M_{t} = \begin{cases} 0, & t = 0, \\ M_{1-\frac{1}{2^{n}}}, & t \in (1 - \frac{1}{2^{n}}, 1 - \frac{1}{2^{n+1}}), a_{n} = 0, n \ge 0, \\ M_{1-\frac{1}{2^{n}}} + \sigma_{n}\phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}), & t = 1 - \frac{1}{2^{n+1}}, a_{n} = 0, n \ge 0, \\ M_{1-\frac{1}{2^{n}}} + M_{t-1-\frac{1}{2^{n}}}^{n}\phi_{n}(\sigma_{1}, \dots, \sigma_{n-1}), & t \in (1 - \frac{1}{2^{n}}, 1 - \frac{1}{2^{n+1}}], a_{n} = 1, n \ge 0, \\ \lim_{n \to \infty} M_{1-\frac{1}{2^{n}}}, & t \ge 1. \end{cases}$$

$$(5.4.30)$$

First we show that  $\lim_{n\to\infty} M_{1-\frac{1}{2^n}}$  exists a.s., hence M is well-defined. By [79, Theorem 3.3.8] it is sufficient to show that there exists  $\xi \in L^1(\Omega; X)$  such that  $M_{1-\frac{1}{2^n}} = \mathbb{E}(\xi|\mathscr{F}_{1-\frac{1}{2^n}})$  for all  $n \ge 1$ . Notice that  $(M_{1-\frac{1}{2^n}})_{n\ge 0}$  is a martingale since  $M_{1-\frac{1}{2^{n+1}}} - M_{1-\frac{1}{2^n}}$  equals either  $\sigma_n \phi_n(\sigma_1, \dots, \sigma_{n-1})$  (if  $a_n = 0$ ) or  $M_{\frac{1}{2^{n+1}}}^n \phi_n(\sigma_1, \dots, \sigma_{n-1})$  (if  $a_n = 1$ ). Both random variables are bounded, and in both cases the conditional expectation with respect to  $\mathscr{F}_{1-\frac{1}{2^n}}$  gives zero. Now let us show integrability. Let  $(\tilde{f}_n)_{n\ge 0}$  be an X-valued martingale such that  $\tilde{f}_0 = 0$  a.s. and

$$d\tilde{f}_n = \sigma_n \phi_n(\sigma_1, \dots, \sigma_{n-1}), \ n \ge 1.$$
 (5.4.31)

Then  $(\tilde{f}_n)_{n\geq 0}$  has the same distribution as  $(f_n)_{n\geq 0}$ , so it is  $L^1$ -bounded. Now fix  $n\geq 1$  and let us estimate  $\mathbb{E}\|\tilde{f}_n - M_{1-\frac{1}{2n}}\|$ :

where (*i*) follows from (5.4.31) and the definition of *M* from (5.4.30), (*ii*) holds by the triangle inequality, (*iii*) follows from the fact that a.s. for each  $n \ge 1$ 

$$\left|\sigma_{n}-M_{\frac{1}{2^{n+1}}}^{n}\right| \leq |\sigma_{n}|+\left|M_{\frac{1}{2^{n+1}}}^{n}\right| \leq 2;$$

finally,  $(i\nu)$  follows from (5.4.29). Let us show that there exists  $\mathscr{F}_1$ -measurable  $\xi \in L^1(\Omega; X)$  such that  $M_{1-\frac{1}{2n}} = \mathbb{E}(\xi | \mathscr{F}_{1-\frac{1}{2n}})$  for each  $n \ge 1$ . First notice that  $\mathbb{E}(\tilde{f}_{\infty} | \mathscr{F}_{1-\frac{1}{2n}}) =$ 

 $\tilde{f}_n$  for each  $n \ge 1$ . Moreover, by (5.4.32) the series

$$\eta := \sum_{k=1}^{\infty} \left( \sigma_k - M_{\frac{1}{2^{k+1}}}^k \right) \phi_k(\sigma_1, \dots, \sigma_{k-1}) \mathbf{1}_{a_k=1}$$

converges in  $L^1(\Omega; X)$ . Therefore, if we define  $\xi := \tilde{f}_{\infty} - \eta$ , then

$$\begin{split} \mathbb{E}(\xi|\mathscr{F}_{1-\frac{1}{2^n}}) &= \mathbb{E}(\tilde{f}_{\infty} - \eta|\mathscr{F}_{1-\frac{1}{2^n}}) \\ &= \tilde{f}_n - \mathbb{E}\Big(\sum_{k=1}^{\infty} \big(\sigma_k - M_{\frac{1}{2^{k+1}}}^k\big)\phi_k(\sigma_1, \dots, \sigma_{k-1})\mathbf{1}_{a_k=1}|\mathscr{F}_{1-\frac{1}{2^n}}\Big) \\ &= \tilde{f}_n - \sum_{k=1}^{\infty} \mathbb{E}\Big(\big(\sigma_k - M_{\frac{1}{2^{k+1}}}^k\big)\phi_k(\sigma_1, \dots, \sigma_{k-1})\mathbf{1}_{a_k=1}|\mathscr{F}_{1-\frac{1}{2^n}}\Big) \\ &= \tilde{f}_n - \sum_{k=1}^n \big(\sigma_k - M_{\frac{1}{2^{k+1}}}^k\big)\phi_k(\sigma_1, \dots, \sigma_{k-1})\mathbf{1}_{a_k=1} = M_{1-\frac{1}{2^n}}, \end{split}$$

so one has an a.s. convergence by the martingale convergence theorem [79, Theorem 3.3.8].

Now let us show that *M* is a martingale that does not have the Meyer-Yoeurp decomposition. Assume the contrary: let  $M = M^d + M^c$  be the Meyer-Yoeurp decomposition. Then one can show that for each  $n \ge 1$ 

$$M_{1-\frac{1}{2^{n}}}^{d} = \sum_{k=1}^{n} \sigma_{k} \phi_{k}(\sigma_{1}, \dots, \sigma_{k-1}) \mathbf{1}_{a_{k}=0},$$
$$M_{1-\frac{1}{2^{n}}}^{c} = \sum_{k=1}^{n} M_{\frac{1}{2^{k+1}}}^{k} \phi_{k}(\sigma_{1}, \dots, \sigma_{k-1}) \mathbf{1}_{a_{k}=1},$$

by applying  $x^* \in X^*$  and showing that the corresponding processes  $\langle M_{1-\frac{1}{2^n}}^d, x^* \rangle$ and  $\langle M_{1-\frac{1}{2^n}}^c, x^* \rangle$  are purely discontinuous and continuous local martingales respectively (see Remark 2.2.19). Now let us show that  $M^c$  is not an *X*-valued local martingale. If it is a local martingale, then

$$\mathbb{P}((M^c)^*_{\infty} = \infty) = \mathbb{P}((M^c)^*_1 = \infty) = 0.$$

since  $M^c$  as a local martingale should have càdlàg paths (even continuous since  $M^c$  assume to be continuous). But for each fixed  $n \ge 1$ 

$$\mathbb{P}((M^{c})_{1}^{*} = \infty) = \mathbb{P}\left(\left(M^{c} - M_{\frac{1}{2^{n}}}^{c}\right)_{1}^{*} = \infty\right) \ge \mathbb{P}\left(\left(\tilde{g} - \tilde{g}_{n}\right)_{\infty}^{*} = \left(M^{c} - M_{\frac{1}{2^{n}}}^{c}\right)_{1}^{*}\right)$$

where  $(\tilde{g}_n)_{n\geq 0}$  is an *X*-valued martingale such that  $d\tilde{g}_n = a_n d\tilde{f}_n$  a.s. for each  $n \geq 0$ , and hence by the construction in Lemma 5.4.15  $\tilde{g}_{\infty}^* = \infty$  a.s. Further,

$$\mathbb{P}\left((\tilde{g} - \tilde{g}_n)_{\infty}^* = \left(M^c - M_{\frac{1}{2^n}}^c\right)_1^*\right) = 1 - \mathbb{P}\left((\tilde{g} - \tilde{g}_n)_{\infty}^* \neq \left(M^c - M_{\frac{1}{2^n}}^c\right)_1^*\right)$$

$$\begin{split} &\geq 1 - \sum_{k=n}^{\infty} \mathbf{1}_{a_k=1} \mathbb{P} \Big( M_{\frac{1}{2^{k+1}}}^k \neq \sigma_k \Big) \\ &\geq 1 - \sum_{k=n}^{\infty} \mathbb{P} \Big( M_{\frac{1}{2^{k+1}}}^k \neq \mathrm{sign} M_{\frac{1}{2^{k+1}}}^k \\ &\stackrel{(*)}{\geq} 1 - \frac{1}{2^{n-1}}, \end{split}$$

where (\*) follows from (5.4.29). Since *n* was arbitrary,  $(M^c)_1^* = (M^c)_{\infty}^* = \infty$  a.s., so  $M^c$  can not be a local martingale.

*Proof of Theorem* 5.4.1 (*necessity of UMD*). Necessity of the UMD property follows from Theorem 5.4.14.

*Remark* 5.4.17. One can also show that existence of the Yoeurp decomposition of an arbitrary *X*-valued purely discontinuous local martingale is equivalent to the UMD property. We will not repeat the argument here, but just notice that one needs to modify the proof of Theorem 5.4.14 in a way which was demonstrated in Subsection 4.3.2.

*Remark* 5.4.18. The reader might assume that one can weaken the Meyer-Yoeurp decomposition and consider a decomposition of an *X*-valued local martingale *M* into a sum of a continuous *X*-valued *semi*martingale  $N^c$  and a purely discontinuous *X*-valued *semi*martingale  $N^d$ , which perhaps may happen in a broader (rather than UMD) class of Banach spaces. Then for any reasonable definition of an *X*-valued semimartingale we get that  $N^c = M^c + A$  for some continuous local martingale  $M^c$  and an adapted process of (weakly) bounded variation *A*. Hence  $M = N^c + N^d = M^c + (N^d + A)$ , where  $N^d + A = M - M^c$  is a local martingale, which is purely discontinuous, so *M* should have the Meyer-Yoeurp decomposition as well in this setting, which means that the UMD property is crucial.

# 6

# ORTHOGONAL MARTINGALES AND THE HILBERT TRANSFORM

This chapter is based on the paper *The Hilbert transform and orthogonal martingales in Banach spaces* by Adam Osękowski and Ivan Yaroslavtsev, see [146].

Let X be a given Banach space and let M, N be two orthogonal X-valued local martingales such that N is weakly differentially subordinate to M. This chapter contains the proof of the estimate

 $\mathbb{E}\Psi(N_t) \leq C_{\Phi,\Psi,X} \mathbb{E}\Phi(M_t), \ t \geq 0,$ 

where  $\Phi, \Psi : X \to \mathbb{R}_+$  are convex continuous functions and the least admissible constant  $C_{\Phi,\Psi,X}$  coincides with the  $\Phi, \Psi$ -norm of the periodic Hilbert transform. As a corollary, it is shown that the  $\Phi, \Psi$ -norms of the periodic Hilbert transform, the Hilbert transform on the real line, and the discrete Hilbert transform are the same if  $\Phi$  is symmetric. We also prove that under certain natural assumptions on  $\Phi$  and  $\Psi$ , the condition  $C_{\Phi,\Psi,X} < \infty$  yields the UMD property of the space X. As an application, we provide comparison of  $L^p$ -norms of the periodic Hilbert transform to Wiener and Paley-Walsh decoupling constants. We also study the norms of the periodic, nonperiodic and discrete Hilbert transforms, present the corresponding estimates in the context of differentially subordinate harmonic functions and more general singular integral operators.

<sup>2010</sup> Mathematics Subject Classification. 44A15, 60G44 Secondary: 60B11, 31C10, 31B05, 46B09 Key words and phrases. Weak differential subordination, orthogonal martingales, periodic Hilbert transform, UMD spaces, plurisubharmonic functions, decoupling constants, discrete Hilbert transform, harmonic functions, Riesz transform.

# **6.1.** INTRODUCTION

The purpose of this chapter is to study a certain class of estimates for singular integral operators acting on Banach-space-valued functions. Let us start with a related classical problem which has served as a motivation for many mathematicians for almost a century. The question is: how does the size of a periodic function control the size of its conjugate? Formally, assume that f is a trigonometric polynomial of the form

$$f(\theta) = \frac{a_0}{2} + \sum_{k=1}^{N} \left( a_k \cos(k\theta) + b_k \sin(k\theta) \right), \qquad \theta \in \mathbb{T} \simeq [-\pi, \pi),$$

with real coefficients  $a_0$ ,  $a_1$ ,  $a_2$ , ...,  $a_N$ ,  $b_1$ ,  $b_2$ , ...,  $b_N$ , and define the conjugate to f as

$$g(\theta) = \sum_{k=1}^{N} (a_k \sin(k\theta) - b_k \cos(k\theta)), \qquad \theta \in [-\pi, \pi).$$

Alternatively, the conjugate function can be defined as  $g = \mathscr{H}_{\mathbb{R}}^{T} f$ , where  $\mathscr{H}_{\mathbb{R}}^{T}$  is the periodic Hilbert transform given by

$$\mathscr{H}_{\mathbb{R}}^{\mathbb{T}}f(\theta) = \frac{1}{2\pi} \text{p.v.} \int_{-\pi}^{\pi} f(s) \cot \frac{\theta - s}{2} \mathrm{d}s, \qquad \theta \in [-\pi, \pi), \tag{6.1.1}$$

and the symbol  $\mathbb{R}$  in the lower index of  $\mathscr{H}^{\mathbb{T}}$  indicates that the operator acts on realvalued functions. We can state the problem as follows. For a given  $1 \le p \le \infty$ , does there exist a universal constant  $C_p$  (that is, not depending on the coefficients or the number *N*) such that

$$\left(\int_{[-\pi,\pi)} |g(\theta)|^p \mathrm{d}\theta\right)^{1/p} \le C_p \left(\int_{[-\pi,\pi)} |f(\theta)|^p \mathrm{d}\theta\right)^{1/p} ?$$

Furthermore, if the answer is yes, what is the optimal value of  $C_p$  (i.e., what is the  $L^p$ -norm of  $\mathscr{H}_{\mathbb{R}}^{\mathbb{T}}$ )? The first question was answered by M. Riesz in [158]: the inequality does hold if and only if  $1 . The best value of <math>C_p$  was determined by Pichorides [149] and Cole (unpublished): the constant  $\cot(\pi/(2p^*))$  is the best possible, where  $p^* = \max\{p, p/(p-1)\}$ . There is a natural further question concerning the version of the above result for Banach-space-valued functions (it is not difficult to see that the formula (6.1.1) makes perfect sense in the vector setting, at least for some special f, see Section 6.2 below). Few years after the results of Riesz, it was realized that not all spaces are well-behaved: Bochner and Taylor [20] showed that  $||\mathscr{H}_{\ell_1}^{\mathbb{T}}||_{L_p \to L_p} = \infty$  for all p. The problem of characterizing the 'good' Banach spaces was solved over forty years later: Burkholder [30] and Bourgain [23] showed that the so-called UMD spaces form a natural environment to the study of the  $L^p$ -boundedness ( $1 ) of the periodic Hilbert transform, and more generally, for the <math>L^p$ -boundedness of a wider class of singular integral operators.

UMD stands for "unconditional martingale differences"

The above problems, though expressed in an analytic language, have a very strong connection with probability theory, especially with the theory of martingales (see e.g. [12, 15, 16, 23, 32, 61, 66, 79, 144, 145]). Let us provide some necessary definitions. Suppose that  $(\Omega, \mathscr{F}, \mathbb{P})$  is a complete probability space, equipped with a continuous-time filtration  $(\mathscr{F}_t)_{t\geq 0}$ . Let  $M = (M_t)_{t\geq 0}$ ,  $N = (N_t)_{t\geq 0}$  be two adapted real-valued local martingales, whose trajectories are right-continuous and have limits from the left. Let [M], [N] stand for the associated quadratic variation (square brackets) of M and N, see (2.2.3). Furthermore,  $M^* = \sup_{t\geq 0} |M_t|$ ,  $N^* = \sup_{t\geq 0} |N_t|$  denote the corresponding maximal functions. We say that M and N are *orthogonal*, if  $[M, N] := \frac{[M+N]-[M-N]}{2} = 0$  and  $M_0N_0 = 0$  almost surely.

One of the remarkable examples of the aforementioned connection between the theory of singular integral operators and martingale theory was provided by Bañuelos and Wang in [15]. They have shown that the  $L^p$ -norm of  $\mathscr{H}^{\mathbb{T}}$  acting on real-valued functions is equal to the sharp constant in the corresponding  $L^p$ inequality

$$(\mathbb{E}|N_t|^p)^{\frac{1}{p}} \le C_p(\mathbb{E}|M_t|^p)^{\frac{1}{p}}, \quad t \ge 0,$$
(6.1.2)

where *N* is assumed to be differentially subordinate and orthogonal to *M*. The goal of the current article is to show that this interplay between the norm of  $\mathscr{H}^{\mathbb{T}}$  and the martingale inequality (6.1.2) can be extended to i) more general  $\Phi, \Psi$ -norms (see the beginning of Section 6.3 for the definition) and ii) more general Banach spaces in which the functions and processes take values.

Let us say a few words about the structure of the chapter. The next section is devoted to the introduction of the background which is needed for our further study. In particular, we define an appropriate analogue of Banach-space-valued orthogonality of martingales and provide some basic information about plurisubharmonic functions, fundamental objects in the complex analysis of several variables. Section 6.3 contains the main result of the chapter, connecting the best constants in certain  $\Phi$ ,  $\Psi$ -estimates for the periodic Hilbert transform and their counterparts in martingale theory. Though the rough idea of the proof can be tracked back to the classical works [15, 36, 77, 149] (the validity of a given estimate for the Hilbert transform / orthogonal differentially subordinate martingales is equivalent to the existence of a certain special plurisubharmonic function), there are several serious technical problems to be overcome, due to the fact that we work in the Banachspace-valued setting. Section 6.4 is devoted to some applications. The first and the most notable one connects together the  $\Phi$ ,  $\Psi$ -norms of the periodic Hilbert transform  $\mathscr{H}_X^{\mathbb{T}}$ , the Hilbert transform  $\mathscr{H}_X^{\mathbb{R}}$  defined on a real line, and the discrete Hilbert transform  $\mathcal{H}_{X}^{\text{dis}}$  (for the definition of the latter object, consult Definition 6.4.1 and 6.4.2 below). It turns out that all these norms coincide for quite general class of  $\Phi$ and  $\Psi$ . This in particular generalizes the recent result of Bañuelos and Kwaśnicki [12] on the discrete Hilbert transform  $\mathscr{H}^{dis}_{\mathbb{R}}$  , which asserts that

$$\|\mathscr{H}_{\mathbb{R}}^{\mathrm{dis}}\|_{L^{p}(\mathbb{Z}) \to L^{p}(\mathbb{Z})} = \|\mathscr{H}_{\mathbb{R}}^{\mathbb{T}}\|_{L^{p}(\mathbb{T}) \to L^{p}(\mathbb{T})} = \mathrm{cot}\left(\frac{\pi}{2p^{*}}\right), \qquad 1$$

This used to be an open problem for 90 years (see [12, 103, 172]). Subsection 6.4.2 is devoted to the comparison of  $L^p$ -norms of the periodic Hilbert transform to Wiener and Paley-Walsh decoupling constants. Application in Subsection 6.4.3 is concerned with UMD Banach spaces and can be regarded as an extension of Bourgain's result [23]: we show that under some mild assumption on  $\Phi$  and  $\Psi$ , the validity of the corresponding  $\Phi$ ,  $\Psi$ -estimate (with some finite constant) implies the UMD property of *X*. In Subsection 6.4.4 we prove that the results obtained in this chapter can be applied to obtain sharper estimates for weakly differentially subordinate martingales (not necessarily satisfying the orthogonality assumption). Subsection 6.4.5 contains the study of related estimates in the context of harmonic functions on Euclidean domains. In Subsection 6.4.6 we present the possibility of extending the estimates to the more general class of singular integral operators. Our final application, described in Subsection 6.4.7, discusses the vector-valued extension of the classical results of Hardy concerning Hilbert operators.

# **6.2. PRELIMINARIES**

This section contains the definitions of some basic notions and facts used later. Here and below, the scalar field is assumed to be  $\mathbb{R}$ , unless stated otherwise.

#### 6.2.1. Periodic Hilbert transform

In what follows, the symbol  $\mathbb{T}$  will stand for the torus ({ $z \in \mathbb{C} : |z| = 1$ }, ·) equipped with the natural multiplication. Sometimes, passing to the argument of a complex number, we will identify  $\mathbb{T}$  with the interval  $[-\pi,\pi)$ . Let *X* be a Banach space. A function  $f : \mathbb{T} \to X$  is called a *step function*, if it is of the form

$$f = \sum_{k=1}^{N} x_k \mathbf{1}_{A_k}(s), \quad -\pi \le s < \pi,$$

where *N* is finite,  $x_k \in X$  and  $A_k$  are intervals in  $\mathbb{T}$ . The *periodic Hilbert transform*  $\mathscr{H}_X^{\mathbb{T}}$  of a step function  $f : \mathbb{T} \to X$  is given by the singular integral

$$\mathscr{H}_{X}^{\mathbb{T}}f(t) = \frac{1}{2\pi} \text{p.v.} \int_{-\pi}^{\pi} f(s) \cot \frac{t-s}{2} \mathrm{d}s, \qquad -\pi \le t < \pi.$$
(6.2.1)

#### 6.2.2. Orthogonal martingales

We have defined the notion of orthogonality of real-valued local martingales in the introductory section. We turn our attention to its vector-valued analogue.

**Definition 6.2.1.** Let *M*, *N* be local martingales taking values in a given Banach space *X*. Then *M*, *N* are said to be *orthogonal*, if  $\langle M_0, x^* \rangle \cdot \langle N_0, x^* \rangle = 0$  and  $[\langle M, x^* \rangle, \langle N, x^* \rangle] = 0$  almost surely for all functionals  $x^* \in X^*$ .

*Remark* 6.2.2. Assume that *M*, *N* are local martingales taking values in some Banach space *X*. If *M*, *N* are orthogonal and *N* is weakly differentially subordinate to *M*, then  $N_0 = 0$  almost surely (which follows immediately from the above definitions). Moreover, under these assumptions, *N* must have continuous trajectories with probability 1. Indeed, in such a case for any fixed  $x^* \in X^*$  the real-valued local martingales  $\langle M, x^* \rangle$ ,  $\langle N, x^* \rangle$  are orthogonal and we have  $\langle N, x^* \rangle \ll \langle M, x^* \rangle$ . Therefore,  $\langle N, x^* \rangle$  has a continuous version for each  $x^* \in X^*$  by [136, Lemma 3.1] (see also [16, Lemma 1]), which in turn implies that *N* is continuous: any *X*-valued local martingale has a càdlàg version (see Proposition 2.2.1).

*Remark* 6.2.3. The requirement  $\langle M_0, x^* \rangle \cdot \langle N_0, x^* \rangle = 0$  for all  $x^* \in X^*$  in Definition 6.2.1 is usually omitted (see e.g. [15, 16, 85]). Nevertheless we need this requirement in order to simplify all the statements in the sequel concerning orthogonal martingales.

Weakly differentially subordinate orthogonal martingales appear naturally while working with the periodic Hilbert transform, which can be seen by exploiting the classical argument of Doob (the composition of a harmonic function with a Brownian motion is a martingale). Indeed, suppose that *X* is a given Banach space. Suppose that *f* is a simple function and put  $g = \mathscr{H}_X^{\mathbb{T}} f$ . Let  $u_f$ ,  $u_g$  denote the harmonic extensions of *f* and *g* to the unit disc, obtained by the convolution with the Poisson kernel. In particular, the equality  $g = \mathscr{H}_X^{\mathbb{T}} f$  implies that  $u_g(0,0) = 0$  and for any functional  $x^* \in X^*$ , the function  $\langle u_f, x^* \rangle + i \langle u_g, x^* \rangle$  is holomorphic on the disc.

Next, suppose that  $W = (W^1, W^2)$  is a planar Brownian motion started from (0,0) and stopped upon leaving the unit disc. Then the processes  $M = (M_t)_{t\geq 0} = (u_f(W_t))_{t\geq 0}$ ,  $N = (N_t)_{t\geq 0} = (u_g(W_t))_{t\geq 0}$  are *X*-valued martingales such that  $N_0 = 0$ . For any functional  $x^* \in X^*$ , we apply the standard, one-dimensional Itô's formula to obtain, for any  $t \ge 0$ ,

$$\langle M_t, x^* \rangle = \langle M_0, x^* \rangle + \int_0^t \nabla \langle u_f(W_s), x^* \rangle \mathrm{d} W_s$$

and

$$\langle N_t, x^* \rangle = \langle N_0, x^* \rangle + \int_0^t \nabla \langle u_g(W_s), x^* \rangle \mathrm{d} W_s.$$

By the aforementioned connection to analytic functions, the gradients  $\nabla \langle u_f, x^* \rangle$ ,  $\nabla \langle u_g, x^* \rangle$  are orthogonal and of equal length, so

$$[\langle M, x^* \rangle, \langle N, x^* \rangle]_t = \int_0^t \nabla \langle u_f(W_s), x^* \rangle \cdot \nabla \langle u_g(W_s), x^* \rangle \, \mathrm{d}s = 0,$$
and

$$[\langle M, x^* \rangle]_t - [\langle N, x^* \rangle]_t = \int_0^t \nabla \langle u_f(W_s), x^* \rangle^2 - \nabla \langle u_g(W_s), x^* \rangle^2 \,\mathrm{d}s = 0.$$

Hence *M*, *N* are orthogonal and satisfy the weak differential subordination  $N \stackrel{w}{\ll} M$ . Since the distribution of  $W_{\infty}$  is uniform on the unit circle  $\mathbb{T}$ , essentially any estimate of the form

$$\mathbb{E}V(M_t, N_t) \le 0, \qquad t \ge 0,$$

for weakly differentially subordinate orthogonal martingales leads to the analogous bound

$$\int_{\mathbb{T}} V(f, \mathcal{H}_X^{\mathbb{T}} f) \mathrm{d} x \leq 0$$

for the periodic Hilbert transform, at least when restricted to the class of simple functions. (Later in Theorem 6.3.1 we will show that the reverse holds true).

For more information and examples concerning orthogonal martingales, we refer the reader to [15, 16, 33, 179].

### 6.2.3. Subharmonic and plurisubharmonic functions

A function  $f : \mathbb{R}^d \to \mathbb{R} \cup \{-\infty\}$  is called *subharmonic* if for any ball  $B \subset \mathbb{R}^d$  and any harmonic function  $g : B \to \mathbb{R}$  such that  $f \leq g$  on  $\partial B$  one has the inequality  $f \leq g$  on the whole *B*. The following lemma follows from [105, Proposition I.9].

**Lemma 6.2.4.** Let  $d \ge 1$  and let  $f : \mathbb{R}^d \to \mathbb{R} \cup \{-\infty\}$  be a subharmonic function. Then either  $f \equiv -\infty$ , or f is locally integrable.

Let *X* be a Banach space. The function  $F : X + iX \to \mathbb{R} \cup \{-\infty\}$  is called *plurisub-harmonic* if for any  $x, y \in X + iX$  the restriction  $z \mapsto F(x + yz)$  is subharmonic in  $z \in \mathbb{C}$ . *Remark* 6.2.5. Notice that X + iX is a Banach space equipped with the norm

$$\|x + iy\|_{X+iX} := \sup_{x^* \in X^*, \|x^*\| \le 1} (|\langle x, x^* \rangle|^2 + |\langle y, x^* \rangle|^2)^{\frac{1}{2}}, \ x, y \in X$$

(see [79, Subsection B.4]).

*Remark* 6.2.6. Let *X* be finite-dimensional. Then any plurisubharmonic function defined on X + iX is subharmonic (see [105, Proposition I.9] and [64, Theorem 39]). Therefore, by Lemma 6.2.4, a plurisubharmonic function either identically equals  $-\infty$ , or is locally integrable.

Let  $F: X + iX \to \mathbb{R}$  be *k*-times differentiable,  $u_1, \ldots, u_k \in X + iX$ . Then we denote

$$\frac{\partial^k F(v)}{\partial u_1 \cdots \partial u_k} := \frac{\partial^k}{\partial t_1 \cdots \partial t_k} F(v + t_1 u_1 + \dots + t_k u_k) \Big|_{t_1, \dots, t_k = 0}, \quad v \in X + iX.$$

In particular, for any  $u \in X + iX$ ,

$$\frac{\partial^k F(v)}{\partial u^k} := \frac{\partial^k}{\partial t^k} F(v + tu) \Big|_{t=0}, \quad v \in X + iX.$$

*Remark* 6.2.7. Note that if *X* is finite-dimensional, *F* is plurisubharmonic and twice differentiable, then for all  $z_0 \in X + iX$  and  $x \in X$  we have

$$\frac{\partial^2 F(z_0)}{\partial x^2} + \frac{\partial^2 F(z_0)}{\partial i x^2} = \left(\frac{\partial^2 F(z_0 + zx)}{\partial \operatorname{Re} z^2} + \frac{\partial^2 F(z_0 + zx)}{\partial \operatorname{Im} z^2}\right)\Big|_{z=0}$$
$$= \Delta_z F(z_0 + zx)|_{z=0} \ge 0.$$

Later on we will need the following result.

**Proposition 6.2.8.** Let X be a Banach space and let  $F : X + iX \to \mathbb{R} \cup \{-\infty\}$  be plurisubharmonic. Assume further that  $y \mapsto F(x+iy)$  is concave in  $y \in X$  for any fixed  $x \in X$ . Then  $x \mapsto F(x+iy)$  is convex in  $x \in X$  for any  $y \in X$ , and F is continuous.

For the proof we will need the following lemma.

**Lemma 6.2.9.** Let X be a finite-dimensional Banach space and let  $V : X + iX \rightarrow \mathbb{R}$  be a continuous twice differentiable plurisubharmonic function. Let  $y \mapsto V(x + iy)$  be concave in  $y \in X$  for all  $x \in X$ . Then  $t \mapsto V(tx + z)$  is convex in  $t \in \mathbb{R}$  for all  $x \in X$  and  $z \in X + iX$ . In particular,  $t \mapsto V(tx + z)$  is differentiable, so

$$V(tx+z) \ge V(sx+z) + \partial_s V(sx+z)(t-s), \quad t,s \in \mathbb{R}.$$
(6.2.2)

*Proof.* The first part follows from the fact that *V* is plurisubharmonic and twice differentiable. Indeed, we have

$$\frac{\partial^2 V(tx+z)}{\partial t^2} = \left(\frac{\partial^2 V(tx+z+isx)}{\partial t^2} + \frac{\partial^2 V(tx+z+isx)}{\partial s^2}\right)\Big|_{s=0} - \frac{\partial^2 V(tx+z+isx)}{\partial s^2}\Big|_{s=0} \ge 0$$

since

$$\Big(\frac{\partial^2 V(tx+z+isx)}{\partial t^2} + \frac{\partial^2 V(tx+z+isx)}{\partial s^2}\Big)\Big|_{s=0} \ge 0$$

by plurisubharmonicity and  $\frac{\partial^2 V(tx+z+isx)}{\partial s^2} \le 0$  by concavity of  $y \mapsto V(x+z+iy)$ . The inequality (6.2.2) follows immediately from the convexity of  $t \mapsto V(tx+iy)$  and twice differentiability of *V*.

For the proof we will need the following observation which will allow us to integrate over a Banach space.

*Remark* 6.2.10. Let *X* be a finite dimensional Banach space. Then due to [59, Theorem 2.20 and Proposition 2.21] there exists a unique translation-invariant measure  $\lambda_X$  on *X* such that  $\lambda_X(\mathbb{B}_X) = 1$  for the unit ball  $\mathbb{B}_X$  of *X*. We will call  $\lambda_X$  the *Lebesgue measure*. In the sequel we will omit the Lebesgue measure notation while integrating over *X* (i.e. we will write  $\int_X F(s) ds$  instead of  $\int_X F(s) \lambda_X(ds)$ ).

*Proof of Proposition 6.2.8.* Without loss of generality we can assume that *X* is finitedimensional and that  $f \neq -\infty$ . Let  $\phi : X + iX \rightarrow \mathbb{R}_+$  be a  $C^{\infty}$  function with bounded support such that

$$\int_{X+iX} \phi(s) \, \mathrm{d}s = 1.$$

(This integral is well-defined due to Remark 6.2.5 and 6.2.10). For each  $\varepsilon > 0$  we define  $F_{\varepsilon} : X + iX \rightarrow \mathbb{R}$  in the following way:

$$F_{\varepsilon}(s) = \int_{X+iX} F(s-\varepsilon t)\phi(t) \,\mathrm{d}t, \quad s \in X+iX.$$
(6.2.3)

Then  $F_{\varepsilon}$  is plurisubharmonic due to [78, Theorem 4.1.4]. Moreover, again by [78, Theorem 4.1.4], we have  $F_{\varepsilon} \searrow F$  as  $\varepsilon \searrow 0$ . On the other hand,  $F_{\varepsilon}$  is well-defined and of class  $C^{\infty}$ . Furthermore, the function  $y \mapsto F_{\varepsilon}(x + iy)$  is concave in  $y \in X$  for any  $x \in X$  by (6.2.3): here we use the fact that F is locally integrable (see Remark 6.2.6) and the concavity of  $y \mapsto F(x + iy)$  for any fixed  $x \in X$ . Therefore by Lemma 6.2.9, the function  $x \mapsto F_{\varepsilon}(x + iy)$  is convex for any fixed  $y \in X$ ; hence so is F, being the pointwise limit of  $(F_{\varepsilon})_{\varepsilon>0}$  as  $\varepsilon \to 0$ .

Let us now show that  $F > -\infty$ . Assume that there exists  $x_0, y_0 \in X$  such that  $F(x_0 + iy_0) = -\infty$ . Since the function  $y \mapsto F(x_0 + iy)$  is concave, the set  $A = \{y \in X : F(x_0 + iy) > \infty\} \subset X$  is convex and open; moreover,  $y_0 \notin A$ , so  $X \setminus A$  is of positive measure. Now fix  $(x, y) \in X \times (X \setminus A)$ . Notice that  $F(x_0 + iy) = -\infty$ . On the other hand  $x \mapsto F(x+iy)$  is convex, so  $F(x+iy) = -\infty$  as well (if a convex function equals  $-\infty$  in one point, it equals  $-\infty$  on the whole X). Therefore  $F = -\infty$  in the set  $X \times (X \setminus A)$  of positive measure; hence  $F \equiv -\infty$  by Remark 6.2.6, which leads to a contradiction.

Finally, note that  $F < \infty$ : we have  $F \le F_1$  with  $F_1$  defined in (6.2.3). Therefore *F* is continuous as a finite concave-convex function (see [171, Proposition 3.3] and [86, Corollary 4.5]).

For further material on subharmonic and plurisubharmonic functions, we recommend the works [64, 78, 105, 159, 160].

### 6.2.4. Meyer-Yoeurp decomposition

The following result shows the connection between the Meyer-Yoeurp decomposition and the weak differential subordination.

**Proposition 6.2.11.** Let X be a Banach space and let M, N be local X-valued martingales possessing the Meyer-Yoeurp decompositions  $M = M^c + M^d$ ,  $N = N^c + N^d$ . Then  $N \stackrel{w}{\ll} M$  if and only if  $N^c \stackrel{w}{\ll} M^c$  and  $N^d \stackrel{w}{\ll} M^d$ . Moreover, if M and N are orthogonal, then  $M^c$  and  $N^c$ ,  $M^d$  and  $N^d$  are pairwise orthogonal.

*Proof.* The first part follows from Lemma 4.4.5 (see also [179, Lemma 1]). Due to Remark 6.2.2 we know that  $N^d = 0$ , so it is sufficient to show that  $M^c$  and  $N^c$ 

are orthogonal. The latter is equivalent to the fact that  $\langle M^c, x^* \rangle$  and  $\langle N^c, x^* \rangle$  are orthogonal for any  $x^* \in X^*$ , which holds true by [16, Lemma 1].

## **6.3.** MAIN THEOREM

Having introduced all the necessary notions, we turn to the study of our new results. For given two nonnegative and continuous functions  $\Phi, \Psi : X \to \mathbb{R}_+$ , we define the associated ' $\Phi, \Psi$ -norm' of  $\mathscr{H}_X^{\mathbb{T}}$  by the formula

$$|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} := \inf \bigg\{ c \in [0,\infty] : \int_{\mathbb{T}} \Psi(\mathscr{H}_X^{\mathbb{T}} f(s)) \, \mathrm{d}s \le c \int_{\mathbb{T}} \Phi(f(s)) \, \mathrm{d}s$$
 for all step functions  $f : \mathbb{T} \to X \bigg\}.$ 

Notice that if  $\Psi \equiv 0$ , then  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} = 0$ , and if  $\Phi \equiv 0$ , then  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} \in \{0, +\infty\}$ . Throughout the chapter we exclude these trivial cases: we will assume that both  $\Phi$  and  $\Psi$  are not identically zero. Furthermore, for any  $1 , we will denote the <math>L^p$ -norm of  $\mathscr{H}_X^{\mathbb{T}}$  by  $\hbar_{p,X}$  (in the language of  $\Phi, \Psi$ -norms, we have  $\hbar_{p,X}^p = |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$  with  $\Phi(x) = \Psi(x) = ||x||^p$ ).

The following theorem is the main result of this section.

**Theorem 6.3.1.** Let X be a separable Banach space and let  $\Phi, \Psi : X \to \mathbb{R}_+$  be continuous convex functions such that  $\Psi(0) = 0$  and  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} < \infty$ . Let M, N be two orthogonal X-valued local martingales such that  $N \stackrel{w}{\ll} M$ . Then

$$\mathbb{E}\Psi(N_t) \le C_{\Phi,\Psi,X} \mathbb{E}\Phi(M_t), \qquad t \ge 0, \tag{6.3.1}$$

and the least admissible  $C_{\Phi,\Psi,X}$  equals  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$ .

The idea behind the proof of (6.3.1) can be roughly described as follows. First, we will show that the condition  $|\mathscr{H}_X^T|_{\Phi,\Psi} < \infty$  (i.e., the validity of a  $\Phi, \Psi$ -estimate for the periodic Hilbert transform) implies the existence of a certain special function on X + iX, enjoying appropriate size conditions and concavity. Next, we will compose this function with M + iN and prove, using the concavity and Itô's formula from the previous section, that the resulting process has nonnegative expectation. This in turn will give the desired bound, in the light of the size condition of the special function. Though this reasoning is typical for this kind of martingale inequalities, there are two essential differences. First, we will see that the special function will not have any explicit form: in particular, this makes the exploitation of its properties much harder, as one can get them only from some abstract (and restricted) reasoning. The second difference is related to the fact that we work will Banach-space-valued processes: this enforces us to study some additional, structural properties of the local martingales involved. Moreover, since we will work in infinite-dimensional Banach spaces, the approximation to finite dimensions exploited in the proof should be especially delicate because we do not want to ruin weak differential subordination and orthogonality of the corresponding martingales.

Having described our plan, we turn to its realization. We will need several intermediate facts. The following theorem links the quantity  $|\mathscr{H}_X^T|_{\Phi,\Psi}$  with a certain special plurisubharmonic function.

**Theorem 6.3.2.** Let X be a separable Banach space and let  $\Phi, \Psi : X \to \mathbb{R}_+$  be continuous functions such that  $\Psi(0) = 0$  and  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} < \infty$ . Then there exists a plurisubharmonic function  $U_{\Phi,\Psi} : X + iX \to \mathbb{R}$  such that  $U_{\Phi,\Psi}(x) \ge 0$  for all  $x \in X$  and

$$U_{\Phi,\Psi}(x+iy) \le |\mathcal{H}_X^{\mathbb{T}}|_{\Phi,\Psi} \Phi(x) - \Psi(y), \ x, y \in X.$$

*Moreover, if*  $\Psi$  *is convex, then*  $y \mapsto U_{\Phi,\Psi}(x + iy)$  *is concave in*  $y \in X$  *for all*  $x \in X$ .

*Proof (sketch).* We repeat the reasoning presented in [77, Theorem 2.3] (the separability of *X* is a key part of the construction  $U_{\Phi,\Psi}$ ). The last property follows from the construction of  $U_{\Phi,\Psi}$ , the fact that  $y \mapsto |\mathscr{H}_X^T|_{\Phi,\Psi} \Phi(x) - \Psi(y)$  is a concave function in  $y \in X$ , and the fact that a minimum of concave functions is a concave function as well.

**Corollary 6.3.3.** Let X be a Banach space,  $1 . Then X is a UMD Banach space if and only if there exists a plurisubharmonic function <math>U_{p,X}: X + iX \rightarrow \mathbb{R}$  such that  $U_{p,X}(x) \ge 0$  for all  $x \in X$  and

$$U_{p,X}(x+iy) \le \hbar_{p,X}^p ||x||^p - ||y||^p, \ x, y \in X.$$

*Moreover, if this is the case, then*  $y \mapsto U_{p,X}(x+iy)$  *is concave in*  $y \in X$  *for all*  $x \in X$ .

*Proof.* It is sufficient to take  $\Phi(x) = \Psi(x) = ||x||^p$ ,  $x \in X$ , and apply Theorem 6.3.2 and the fact that  $\hbar_{p,X} < \infty$  if and only if *X* is a UMD Banach space (see [23, 30]).

**Lemma 6.3.4.** Let X be a Banach space, let M be an X-valued local martingale and let  $(\tau_n)_{n\geq 1}$  be a sequence of stopping times increasing to infinity almost surely. Let  $\Phi: X \to \mathbb{R}_+$  be a convex function such that  $\mathbb{E}\Phi(M_t) < \infty$  for some  $t \geq 0$ . Then  $\mathbb{E}\Phi(M_{t\wedge\tau_n}) \nearrow \mathbb{E}\Phi(M_t)$  as  $n \to \infty$ .

*Proof.* Notice that  $(\mathbb{E}\Phi(M_{t\wedge\tau_n}))_{n\geq 1}$  is an increasing sequence which is less then  $\mathbb{E}\Phi(M_t)$  by the conditional Jensen's inequality, [89, Theorem 7.12], and [89, Lemma 7.1(iii)]. On the other other hand  $\Phi(M_{t\wedge\tau_n}) \rightarrow \Phi(M_t)$  a.s. since  $\tau_n \rightarrow \infty$  as  $n \rightarrow \infty$ . It suffices to apply Fatou's lemma to get the assertion.

The next statement contains the proof of a structural property of orthogonal martingales. We need an additional notion. A linear operator *T* acting on a Hilbert space *H* is called *skew-symmetric* (or *antisymmetric*) if  $\langle Th, h \rangle = 0$  for all  $h \in H$ .

**Proposition 6.3.5.** Let  $d \ge 1$ , W be a d-dimensional standard Brownian motion, let X be a finite-dimensional Banach space and let  $\phi, \psi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(\mathbb{R}^d, X)$  be progressively measurable processes such that  $M := \phi \cdot W$  and  $N := \psi \cdot W$  are well-defined orthogonal martingales. Assume further that  $N \stackrel{w}{\ll} M$ . Then there exists a operator-valued progressivelymeasurable process  $A : \mathbb{R}_+ \times \Omega \to \mathscr{L}(\mathbb{R}^d)$  such that  $||A|| \le 1$ ,  $\psi^* = A\phi^*$  a.s. on  $\mathbb{R}_+ \times \Omega$ , and  $P_{\operatorname{Ran}(\phi^*)}(s, \omega) A(s, \omega)$  is skew-symmetric for all  $s \ge 0$  and  $\omega \in \Omega$ , where  $P_{\operatorname{Ran}(\phi^*)} \in \mathscr{L}(\mathbb{R}^d)$  is the orthoprojection on  $\operatorname{Ran}(\phi^*)$ .

*Proof.* Let  $(x_n^*)_{n\geq 1}$  be a dense sequence in  $X^*$ . Then by the orthogonality of M, N and the condition  $N \stackrel{w}{\ll} M$ , we have

$$\begin{split} \|\psi^*(t,\omega)x_n^*\| &\leq \|\phi^*(t,\omega)x_n^*\|, \\ \langle\psi^*(t,\omega)x_n^*,\phi^*(t,\omega)x_n^*\rangle &= 0 \end{split}$$

for almost all  $\omega \in \Omega$ , all  $t \in \mathbb{R}_+$  and all  $n \ge 1$ . Hence by the density argument, for any  $x^* \in X^*$ , almost all  $\omega \in \Omega$  and all  $t \in \mathbb{R}_+$ ,

$$\|\psi^*(t,\omega)x^*\| \le \|\phi^*(t,\omega)x^*\|, \tag{6.3.2}$$

$$\langle \psi^*(t,\omega)x^*, \phi^*(t,\omega)x^* \rangle = 0.$$
 (6.3.3)

Fix  $t \in \mathbb{R}_+$  and  $\omega \in \Omega$  such that (6.3.2) and (6.3.3) hold for any  $x^* \in X^*$ . Define  $A(t, \omega) : H \to H$  in the following way (we omit  $(t, \omega)$  for the convenience of the reader):

$$Ah := \begin{cases} \psi^* x^*, & \text{if } \exists x^* \in X^* \text{ such that } h = \phi^* x^*; \\ 0, & \text{if } h \perp \operatorname{Ran}(\phi^*). \end{cases}$$
(6.3.4)

Then *A* is well-defined since if  $h = \phi^*(t, \omega)x_1^* = \phi^*(t, \omega)x_2^*$  for some different  $x_1^*, x_2^* \in X^*$ , then by (6.3.2),

$$\begin{split} \|\psi^*(t,\omega)x_1^* - \psi^*(t,\omega)x_2^*\| &= \|\psi^*(t,\omega)(x_1^* - x_2^*)\| \\ &\leq \|\phi^*(t,\omega)(x_1^* - x_2^*)\| \\ &= \|\phi^*(t,\omega)x_1^* - \phi^*(t,\omega)x_2^*\| = \|h - h\| = 0. \end{split}$$

Moreover, *A* is linear on both  $\operatorname{Ran}(\phi^*)$  and  $(\operatorname{Ran}(\phi^*))^{\perp}$ , so it can be extended to a linear operator  $A \in \mathscr{L}(H)$ . Notice that then we have  $\psi^* = A\phi^*$ . Furthermore, the conditions (6.3.2) and (6.3.4) imply that  $||A|| \leq 1$ , while (6.3.3) and (6.3.4) give that  $P_{\operatorname{Ran}(\phi^*)}A$  is skew-symmetric ( $P_{\operatorname{Ran}(\phi^*)}$  being the orthoprojection on  $\operatorname{Ran}(\phi^*)$ ).

In our later considerations, we will also need the following technical result.

**Proposition 6.3.6.** Let X be a finite-dimensional Banach space and let  $\Phi, \Psi : X \to \mathbb{R}_+$ be continuous functions such that  $\Psi$  is convex,  $\Psi(0) = 0$  and  $|\mathscr{H}_X^T|_{\Phi,\Psi} < \infty$ . Let  $U_{\Phi,\Psi}$ :  $X + iX \rightarrow \mathbb{R}$  be the special function from Theorem 6.3.2. Assume additionally that  $U_{\Phi,\Psi}$  is twice differentiable. Then for any  $x, y \in X$ ,  $z_0 \in X + iX$  and any  $\lambda \in [-1,1]$  we have

$$\frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial x^{2}} + \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial y^{2}} + 2\lambda \Big( \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial x \partial i y} - \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial y \partial i x} \Big) + \lambda^{2} \Big( \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial i x^{2}} + \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial i y^{2}} \Big) \ge 0.$$
(6.3.5)

*Proof.* Notice that the function

$$\begin{split} \lambda \mapsto \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial x^2} + \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial y^2} + 2\lambda \Big( \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial x \partial i y} - \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial y \partial i x} \Big) \\ + \lambda^2 \Big( \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial i x^2} + \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial i y^2} \Big) \end{split}$$

is concave due to the fact that  $\frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial i x^2}$ ,  $\frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial i y^2} \le 0$  by the last part of Theorem 6.3.2. Therefore it is sufficient to show (6.3.5) for  $\lambda = 1$  and  $\lambda = -1$ . We will consider the first possibility only, the second can be handled analogously. We have

$$\begin{aligned} \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial x^2} + \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial y^2} + 2 \Big( \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial x \partial i y} - \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial y \partial i x} \Big) \\ &+ \Big( \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial i x^2} + \frac{\partial^2 U_{\Phi,\Psi}(z_0)}{\partial i y^2} \Big) \\ &= \frac{\partial^2 U_{\Phi,\Psi}(z_0 + t(x + iy))}{\partial t^2} + \frac{\partial^2 U_{\Phi,\Psi}(z_0 + t(y - ix))}{\partial t^2} \\ &= \Delta_z U_{\Phi,\Psi}(z_0 + z(y - ix)) \ge 0, \end{aligned}$$

since  $U_{\Phi,\Psi}$  is plurisubharmonic (here  $\Delta_z$  is the Laplace operator acting with respect to the *z*-variable).

**Corollary 6.3.7.** Under the assumptions of the previous Proposition, for any  $x, y \in X$ ,  $z_0 \in X + iX$ ,  $\lambda \in [-1, 1]$  and any  $\mu \in [-|\lambda|, |\lambda|]$  we have

$$\frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial x^{2}} + \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial y^{2}} + 2\mu \Big( \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial x \partial i y} - \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial y \partial i x} \Big) + \lambda^{2} \Big( \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial i x^{2}} + \frac{\partial^{2} U_{\Phi,\Psi}(z_{0})}{\partial i y^{2}} \Big) \ge 0.$$
(6.3.6)

*Proof.* The left-hand side of (6.3.6) is linear in  $\mu$ , so it is sufficient to check the estimate for  $\mu = \pm \lambda$ .

The next few statements aim at establishing an appropriate "localization" procedure: we will prove how to deduce the general, possibly infinite-dimensional context from its finite-dimensional counterpart. We need some additional notation. Let *X* be a Banach space with a dual  $X^*$ ,  $Y \subset X^*$  be a linear subspace. Let

 $P: Y \hookrightarrow X^*$  be the continuous embedding operator. Then  $P^*$  is a well-defined bounded linear operator from  $X^{**}$  to  $X_Y := Y^*$  such that  $\operatorname{Ran}(P^*) = X_Y$ . Moreover, if *Y* is finite-dimensional, then  $\operatorname{Ran}(P^*|_X) = X_Y$ , where  $P^*|_X : X \to X_Y$  is a welldefined restriction of  $P^*$  on *X* due to the natural embedding  $X \hookrightarrow X^{**}$ . For any function  $\phi: X \to \mathbb{R}_+$ , we can define  $\phi_Y: X_Y \to \mathbb{R}_+$  by the formula

$$\phi_Y(\tilde{x}) = \inf\{\phi(x) : x \in X, P^* x = \tilde{x}\}, \quad \tilde{x} \in X_Y.$$
(6.3.7)

**Lemma 6.3.8.** Let X be a Banach space with a dual  $X^*$  and let  $Y \subset X^*$  be a linear subspace. Let  $\phi: X \to \mathbb{R}_+$  be a convex function. Then  $\phi_Y: X_Y \to \mathbb{R}_+$  defined by (6.3.7) is convex and we have  $\phi_Y(P^*x) \le \phi(x)$  for all  $x \in X$ .

*Proof.* Fix  $\tilde{x}_1, \tilde{x}_2 \in X_Y$ ,  $\lambda \in [0, 1]$  and set  $\tilde{x} = \lambda \tilde{x}_1 + (1 - \lambda) \tilde{x}_2$ . Then

$$\begin{split} \phi_{Y}(\tilde{x}) &= \inf_{\substack{x \in X \\ P^{*}x = \tilde{x}}} \phi(x) = \inf_{\substack{x_{1} \in X, P^{*}x_{1} = \tilde{x}_{1} \\ x_{2} \in X, P^{*}x_{2} = \tilde{x}_{2}}} \phi(\lambda x_{1} + (1 - \lambda)x_{2}) \\ &\leq \inf_{\substack{x_{1} \in X, P^{*}x_{1} = \tilde{x}_{1} \\ x_{2} \in X, P^{*}x_{2} = \tilde{x}_{2}}} \lambda \phi(x_{1}) + (1 - \lambda)\phi(x_{2}) \\ &= \lambda \inf_{x_{1} \in X, P^{*}x_{1} = \tilde{x}_{1}} \phi(x_{1}) + (1 - \lambda) \inf_{x_{2} \in X, P^{*}x_{2} = \tilde{x}_{2}} \phi(x_{2}) \\ &= \lambda \phi_{Y}(\tilde{x}_{1}) + (1 - \lambda)\phi_{Y}(\tilde{x}_{2}), \end{split}$$

so  $\phi_Y$  is convex. The last part of the lemma follows from the definition of  $\phi_Y$ .  $\Box$ 

**Lemma 6.3.9.** Let X be a separable Banach space,  $\phi : X \to \mathbb{R}_+$  be convex lower semicontinuous. Then there exists an increasing sequence of finite-dimensional subspaces  $(Y_n)_{n\geq 1}$  of  $X^*$  such that the following holds. If  $P_n : Y_n \to X^*$  is the corresponding embedding for each  $n \geq 1$  and  $\phi_n : Y_n^* \to \mathbb{R}_+$  satisfies

$$\phi_n(\tilde{x}) = \inf\{\phi(x) : x \in X, P_n^* x = \tilde{x}\}, \quad \tilde{x} \in Y_n^*,$$
(6.3.8)

then for each  $x \in X$  the sequence  $(\phi_n(P_n^*x))_{n\geq 1}$  increases to  $\phi(x)$  as  $n \to \infty$ .

*Proof.* By [79, Lemma 1.2.10] there exist a sequence  $(x_n^*)_{n\geq 1}$  in  $X^*$  and a sequence  $(a_n)_{n\geq 1}$  of real numbers such that

$$\phi(x) = \sup_{n} \langle x, x_n^* \rangle + a_n, \ x \in X.$$
(6.3.9)

Let  $Y_n := \operatorname{span}(x_1^*, \dots, x_n^*)$  for each  $n \ge 1$ . Fix  $x \in X$ . First notice that  $\phi_n(P_n^*x) \le \phi(x)$  by Lemma 6.3.8. Moreover,  $\phi_n(P_n^*x) \le \phi_{n+1}(P_{n+1}^*x)$  for each  $n \ge 1$  since  $Y_n \subset Y_{n+1}$  (see (6.3.8)). Fix  $n \ge 1$ . Then for any  $y \in X$  such that  $P_n^*x = P_n^*y$  we have  $\langle x, x_k^* \rangle = \langle y, x_k^* \rangle$ for any  $k = 1, \dots, n$ , so by (6.3.9),

$$\phi_n(P_n^*x) = \inf\{\phi(y) : y \in X, P_n^*y = P_n^*x\}$$
  
$$\geq \inf\{\sup_{1 \le k \le n} \langle y, x_k^* \rangle + a_k : y \in X, P_n^*y = P_n^*x\}$$

 $= \inf\{\sup_{1 \le k \le n} \langle x, x_k^* \rangle + a_k : y \in X, P_n^* y = P_n^* x\}$  $= \sup_{1 \le k \le n} \langle x, x_k^* \rangle + a_k.$ 

Since the latter expression tends to  $\phi(x)$  as  $n \to \infty$ , we obtain the desired monotone convergence  $\phi_n(P_n^*x) \nearrow \phi(x)$ .

**Proposition 6.3.10.** Let X be a Banach space with a dual  $X^*$  and let  $Y \subset X^*$  be a finitedimensional linear subspace. Assume further that  $\Phi, \Psi : X \to \mathbb{R}_+$  are convex continuous functions and let  $\Phi_Y, \Psi_Y : X_Y \to \mathbb{R}_+$  be defined by (6.3.7). Then

$$|\mathscr{H}_{X_{Y}}^{\mathbb{I}}|_{\Phi_{Y},\Psi_{Y}} \leq |\mathscr{H}_{X}^{\mathbb{I}}|_{\Phi,\Psi}.$$

Proof. Recall that

$$|\mathscr{H}_{X_{Y}}^{\mathbb{T}}|_{\Phi_{Y},\Psi_{Y}} = \sup_{f \in F_{X_{Y}}^{\text{step}}} \frac{\int_{\mathbb{T}} \Psi_{Y}(\mathscr{H}_{X_{Y}}^{\mathbb{T}}f) \, d\mu}{\int_{\mathbb{T}} \Phi_{Y}(f) \, d\mu},$$

where  $\mu$  is the Lebesgue measure on  $\mathbb{T}$ . Fix  $f \in F_{X_Y}^{\text{step}}$  and  $\varepsilon > 0$ . Let  $(\tilde{x}_n)_{n=1}^N \subset X_Y$  be the range of f. For each n = 1, ..., N we define  $x_n \in X$  to be such that  $P^* x_n = \tilde{x}_n$  and  $\Phi(x_n) \leq (1+\varepsilon)\Phi_Y(\tilde{x}_n)$  (existence of such  $x_n$  follows from the fact that  $\text{Ran}(P^*) = X_Y$ ); we define  $g : \mathbb{T} \to X$  to be such that  $f(s) = \tilde{x}_n$  if and only if  $g(s) = x_n$ ,  $s \in \mathbb{T}$ . Then  $\Phi_Y(f) = \Phi_Y(P^*g)$  and  $\Psi_Y(\mathscr{H}_{X_Y}^{\mathbb{T}}f) = \Psi_Y(\mathscr{H}_{X_Y}^{\mathbb{T}}P^*g) = \Psi_Y(P^*\mathscr{H}_X^{\mathbb{T}}g)$  for any  $s \in \mathbb{T}$  by the definition of the Hilbert transform on the torus. Therefore

$$\frac{\int_{\mathbb{T}} \Psi_{Y}(\mathscr{H}_{X_{Y}}^{\mathbb{T}}f) \, \mathrm{d}\mu}{\int_{\mathbb{T}} \Phi_{Y}(f) \, \mathrm{d}\mu} = \frac{\int_{\mathbb{T}} \Psi_{Y}(P^{*}\mathscr{H}_{X}^{\mathbb{T}}g) \, \mathrm{d}\mu}{\int_{\mathbb{T}} \Phi_{Y}(P^{*}g) \, \mathrm{d}\mu} \stackrel{(*)}{\leq} (1+\varepsilon) \frac{\int_{\mathbb{T}} \Psi(\mathscr{H}_{X}^{\mathbb{T}}g) \, \mathrm{d}\mu}{\int_{\mathbb{T}} \Phi(g) \, \mathrm{d}\mu}$$

$$\stackrel{(**)}{\leq} (1+\varepsilon)|\mathscr{H}_{X}^{\mathbb{T}}|_{\Phi,\Psi},$$

where (\*) follows from the fact that  $\Phi(g(s)) \leq (1 + \varepsilon)\Phi_Y(f(s))$  for any  $s \in \mathbb{T}$  and from the fact that  $\Psi_Y(P^* \cdot) \leq \Psi(\cdot)$  on *X*, while (\*\*) follows from the definition of  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$ . Since  $f \in F_{X_Y}^{\text{step}}$  and  $\varepsilon > 0$  were arbitrary, the claim follows.

The final ingredient is the following well-known statement from the theory of stochastic integration.

**Lemma 6.3.11.** Let  $d \ge 1$  and let M be a martingale with values in  $\mathbb{R}^d$  satisfying the condition  $\mathbb{E}M^*_{\infty} < \infty$ . Let  $V : \mathbb{R}_+ \times \Omega \to \mathbb{R}^d$  be a predictable and bounded process. Then  $V \cdot M := \int \langle V, dM \rangle$  is a well-defined martingale and  $\mathbb{E}(V \cdot M)^*_{\infty} < \infty$ .

Equipped with the above statements, we are ready for the study of our main result. We should point out that the main difficulty lies in proving the inequality (6.3.1) for finite-dimensional Banach spaces. The novelty in comparison to other results from the literature is that we work under slightly different condition of weak differential subordination and orthogonality; therefore, though at some places the arguments might look similar to, for instance, those appearing in [15], there is no apparent connection between them.

*Proof of* (6.3.1) *for finite-dimensional X*. We split the reasoning into several intermediate parts.

Step 1. Some reductions. First assume that the function  $U_{\Phi,\Psi}$  (defined in Theorem 6.3.2) is continuous and twice differentiable. Since N has continuous paths almost surely, we may assume that N is a bounded martingale: this is due to a simple stopping time argument combined with Lemma 6.3.4. Moreover, we may assume that  $\mathbb{E}\Phi(M_t) < \infty$ , since otherwise there is nothing to prove. Let d be the dimension of X. Then analogously to the proof of Proposition 4.4.3 we can find a continuous time-change  $\tau = (\tau_s)_{s\geq 0}$  and redefine  $M := M \circ \tau$  and  $N := N \circ \tau$ , so that the following holds. For some 2d-dimensional standard Brownian motion W on an extended probability space  $(\widetilde{\Omega}, \widetilde{\mathscr{F}}, \widetilde{\mathbb{P}})$  equipped with an extended filtration  $\widetilde{\mathbb{F}} = (\widetilde{\mathscr{F}}_t)_{t\geq 0}$ , there exist progressively measurable processes  $\phi, \psi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(\mathbb{R}^{2d}, X)$  such that  $M^c = \phi \cdot W$  and  $N = \psi \cdot W$ , where  $M = M^c + M^d$  is the Meyer-Yoeurp decomposition of M (see Section 2.2.3). In addition, the arguments in the proof of Proposition 4.4.3 also yield the identities  $[M \circ \tau] = [M] \circ \tau$ ,  $[N \circ \tau] = [N] \circ \tau$  and  $[M \circ \tau, N \circ \tau] = [M, N] \circ \tau$ , so the weak differential subordination and orthogonality are not ruined under the time-change.

Now, for each  $n \ge 1$ , introduce the stopping time

$$\sigma_n := \inf\{t \ge 0 : M_t > n\}.$$
(6.3.10)

By Lemma 6.3.4 it is sufficient to show that

$$\mathbb{E}\Psi(N_{t\wedge\sigma_n}) \le |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} \mathbb{E}\Phi(M_{t\wedge\sigma_n})$$
(6.3.11)

for any  $n \ge 1$ . Actually, passing to M/n, N/n, we see that it is enough to show the above estimate for n = 1. For the sake of notational convenience, we redefine  $M := M^{\sigma_1}$  and  $N := N^{\sigma_1}$  and observe that it suffices to show  $\mathbb{E}U_{\Phi,\Psi}(M_t + iN_t) \ge 0$ , since then (6.3.11) follows at once from the majorization property of  $U_{\Phi,\Psi}$ .

Step 2. Application of Itô's formula. Let  $(e_n)_{n=1}^d$  be a basis of *X*, and  $(e_n^*)_{n=1}^d$  be the corresponding dual basis. Then by the Itô formula (2.12.1), we get

$$\mathbb{E}U_{\Phi,\Psi}(M_t + iN_t) = \mathbb{E}U_{\Phi,\Psi}(M_0 + iN_0) + \mathbb{E}\int_0^t \langle \partial_x U_{\Phi,\Psi}(M_{s-} + iN_{s-}), dM_s \rangle + \mathbb{E}\int_0^t \langle \partial_{ix} U_{\Phi,\Psi}(M_{s-} + iN_{s-}), dN_s \rangle$$
(6.3.12)  
+  $\mathbb{E}I_1 + \mathbb{E}I_2,$ 

where  $\partial_x U_{\Phi,\Psi}(\cdot), \partial_{ix} U_{\Phi,\Psi}(\cdot) \in X^*$  are the corresponding Fréchet derivatives of  $U_{\Phi,\Psi}$  in the real and the imaginary subspaces of X + iX respectively,

$$I_1 = \sum_{0 \le s \le t} (\Delta U_{\Phi,\Psi}(M_s + iN_s) - \langle \partial_x U_{\Phi,\Psi}(M_{s-} + iN_{s-}), \Delta M_s \rangle),$$

and

$$\begin{split} I_{2} &= \frac{1}{2} \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial e_{i} \partial e_{j}} \, \mathrm{d}[\langle M^{c}, e_{i}^{*} \rangle, \langle M^{c}, e_{j}^{*} \rangle]_{s} \\ &+ \frac{1}{2} \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial i e_{i} \partial i e_{j}} \, \mathrm{d}[\langle N, e_{i}^{*} \rangle, \langle N, e_{j}^{*} \rangle]_{s} \\ &+ \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial e_{i} \partial i e_{j}} \, \mathrm{d}[\langle M^{c}, e_{i}^{*} \rangle, \langle N, e_{j}^{*} \rangle]_{s} \\ &= \frac{1}{2} \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial e_{i} \partial e_{j}} \langle \phi^{*}(s) e_{i}^{*}, \phi^{*}(s) e_{j}^{*} \rangle \, \mathrm{d}s \\ &+ \frac{1}{2} \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial i e_{i} \partial i e_{j}} \langle \psi^{*}(s) e_{i}^{*}, \psi^{*}(s) e_{j}^{*} \rangle \, \mathrm{d}s \\ &+ \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial e_{i} \partial i e_{j}} \langle \phi^{*}(s) e_{i}^{*}, \psi^{*}(s) e_{j}^{*} \rangle \, \mathrm{d}s. \end{split}$$

Step 3. Analysis of the terms on the right of (6.3.12). Let us first show that

$$\mathbb{E}\int_{0}^{t} \langle \partial_{x} U_{\Phi,\Psi}(M_{s-}+iN_{s-}), \, \mathrm{d}M_{s} \rangle + \mathbb{E}\int_{0}^{t} \langle \partial_{ix} U_{\Phi,\Psi}(M_{s-}+iN_{s-}), \, \mathrm{d}N_{s} \rangle$$

exists and equals zero. First notice that since  $M = M^{\sigma_1}$ , the variable  $M_{s-}$  is bounded by 1 for any  $0 \le s \le \sigma_1$ . Furthermore, as we have assumed above, the process N is also bounded. Since  $U_{\Phi,\Psi}$  is twice differentiable, both  $\partial_x U_{\Phi,\Psi}(\cdot)$  and  $\partial_{ix} U_{\Phi,\Psi}(\cdot)$  are continuous functions, so  $s \mapsto \partial_x U_{\Phi,\Psi}(M_{s-}+iN_{s-})$  and  $s \mapsto \partial_{ix} U_{\Phi,\Psi}(M_{s-}+iN_{s-})$  define bounded processes on  $0 \le s \le \sigma_1$ . Furthermore, it is easy to see that

$$\mathbb{E}M_t^* = \mathbb{E}M_{t\wedge\sigma_1}^* \le \mathbb{E}\|M_{t\wedge\sigma_1}\| + 1 \le \mathbb{E}\|M_t\| + 1 < \infty,$$

and hence by Lemma 6.3.11,

$$t \mapsto \int_{0}^{t} \langle \partial_{x} U_{\Phi,\Psi}(M_{s-} + iN_{s-}) \mathbf{1}_{s \in [0,\sigma_{1}]}, dM_{s} \rangle, \quad t \ge 0,$$
  
$$t \mapsto \int_{0}^{t} \langle \partial_{ix} U_{\Phi,\Psi}(M_{s-} + iN_{s-}) \mathbf{1}_{s \in [0,\sigma_{1}]}, dN_{s} \rangle, \quad t \ge 0,$$
  
(6.3.13)

define martingales. Moreover, with probability 1,

$$\int_0^t \langle \partial_x U_{\Phi,\Psi}(M_{s-} + iN_{s-}) \mathbf{1}_{s \in [0,\sigma_1]}, dM_s \rangle = \int_0^t \langle \partial_x U_{\Phi,\Psi}(M_{s-} + iN_{s-}), dM_s \rangle,$$
$$\int_0^t \langle \partial_{ix} U_{\Phi,\Psi}(M_{s-} + iN_{s-}) \mathbf{1}_{s \in [0,\sigma_1]}, dN_s \rangle = \int_0^t \langle \partial_{ix} U_{\Phi,\Psi}(M_{s-} + iN_{s-}), dN_s \rangle,$$

since  $M = M^{\sigma_1}$  and  $N = N^{\sigma_1}$ , and consequently the expectations of the above integrals vanish. Let us now show that  $I_1, I_2 \ge 0$  almost surely. For the first term, the

argument is simple: by (6.2.2), each summand in  $I_1$  is nonnegative. The analysis of  $I_2$  is slightly more complex. By Proposition 6.2.11, we get that  $N \stackrel{w}{\ll} M^c$  and  $M^c$ , N are orthogonal, so Proposition 6.3.5 implies the existence of a progressively measurable operator-valued process  $A : \mathbb{R}_+ \times \Omega \to \mathscr{L}(\mathbb{R}^d)$  such that  $||A|| \leq 1$ ,  $\psi^* = A\phi^*$ , and  $P_{\operatorname{Ran}(\phi^*)}A$  is skew-symmetric on  $\mathbb{R}_+ \times \Omega$  (here  $P_{\operatorname{Ran}(\phi^*)}$  is an orthoprojection on  $\operatorname{Ran}(\phi^*)$ ). Thus it is enough to show that

$$\sum_{i,j=1}^{d} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial e_j} \langle \phi^*(s) e_i^*, \phi^*(s) e_j^* \rangle ds$$

$$+ \sum_{i,j=1}^{d} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_j} \langle \psi^*(s) e_i^*, \psi^*(s) e_j^* \rangle ds \qquad (6.3.14)$$

$$+ 2 \sum_{i,j=1}^{d} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial i e_j} \langle \phi^*(s) e_i^*, P_{\operatorname{Ran}(\phi^*)} A \phi^*(s) e_j^* \rangle ds \ge 0.$$

By the spectral theory of skew-symmetric matrices (see e.g. [191, Corollary 2]) there exist  $L \ge 0$ , positive numbers  $(\lambda_n)_{n=1}^L$  and an orthonormal basis  $(h_n)_{n=1}^{2d}$  of  $\mathbb{R}^{2d}$  such that  $P_{\text{Ran}(\phi^*)}Ah_{2n-1} = \lambda_nh_{2n}$  and  $P_{\text{Ran}(\phi^*)}Ah_{2n} = -\lambda_nh_{2n-1}$  for all n = 1, ..., L, and  $P_{\text{Ran}(\phi^*)}Ah_n = 0$  for all  $2L < n \le d$ . Moreover, the condition  $||A|| \le 1$  implies that  $|\lambda_1|, ..., |\lambda_L| \le 1$ , and since  $(\text{Ran}(\phi^*))^{\perp}$  is a zero eigenspace of  $P_{\text{Ran}(\phi^*)}A$  (see the construction of A in the proof of Proposition 6.3.5), we conclude that  $h_n \in \text{Ran}(\phi^*)$  for n = 1, 2, ..., 2L. By a usual orthogonalization procedure, we may assume that there exists  $K \ge 2L$  such that  $h_n \in \text{Ran}(\phi^*)$  for  $2L < n \le K$  and  $h_n \perp \text{Ran}(\phi^*)$  for  $K < n \le 2d$  (then K is the dimension of  $\text{Ran}(\phi^*)$ ). Notice that  $X^*$  is d-dimensional, so  $\text{Ran}(\phi^*)$  is at most d-dimensional and hence obviously  $K \le d$ . Due to Lemma 2.11.2, the expression (6.3.14) does not depend on the basis  $(e_n)_{n=1}^d$  (and the corresponding dual basis  $(e_n^*)_{n=1}^d$ ), so we can choose a basis  $(e_n)_{n=1}^d$  such that  $\phi^* e_n^* = h_n$  for all n = 1, ..., K and  $\phi^* e_n^* = 0$  for all  $K < n \le d$  (such a basis exists since span $\{h_1, ..., h_K\}$  =  $\text{Ran}(\phi^*)$ ). Then (6.3.14) becomes

$$\sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial e_i \partial e_j} \langle h_i, h_j \rangle$$

$$+ \sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial i e_i \partial i e_j} \langle \psi^* e_i^*, \psi^* e_j^* \rangle \qquad (6.3.15)$$

$$+ 2 \sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-} + iN_{s-})}{\partial e_i \partial i e_j} \langle h_i, P_{\operatorname{Ran}(\phi^*)} A h_j \rangle \ge 0$$

(The second sum is up to *K* due to the fact that  $\phi^* x^* = 0$  implies  $\psi^* x^* = 0$  for any  $x^* \in X^*$ , see (6.3.2)). Notice that the bilinear form  $V : X \times X \to \mathbb{R}$  defined by

$$V(x, y) := -\frac{\partial^2 U_{\Phi, \Psi}(M_{s-} + iN_{s-})}{\partial i x \partial i y}, \quad x, y \in X,$$

is nonnegative by Theorem 6.3.2 and symmetric by the definition. Moreover, by (6.3.2),

$$\langle \psi^* x^*, \psi^* x^* \rangle = \|\psi^* x^*\|^2 \le \|\phi^* x^*\|^2 = \langle \phi^* x^*, \phi^* x^* \rangle, \text{ for } x^* \in X^*.$$

Therefore Corollary 2.11.3 yields

$$\sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_j} \langle \psi^* e_i^*, \psi^* e_j^* \rangle \geq \sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_j} \langle \phi^* e_i^*, \phi^* e_j^* \rangle$$
$$= \sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_j} \langle h_i, h_j \rangle,$$

so (6.3.15) is not less than

$$\begin{split} \sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial e_j} \langle h_i, h_j \rangle + \sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_j} \langle h_i, h_j \rangle \\ &+ 2 \sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial i e_j} \langle h_i, P_{\text{Ran}(\phi^*)} A h_j \rangle \\ &= \sum_{i=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial e_i} \langle h_i, h_i \rangle + \sum_{i=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_i} \langle h_i, h_i \rangle \\ &+ 2 \sum_{i,j=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial e_i} \langle h_i, h_j \rangle + \sum_{i=1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_i} \langle h_i, h_j \rangle \end{split}$$

The latter expression consists of two parts:

$$\sum_{i=1}^{2L} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial e_i} + \sum_{i=1}^{2L} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_i} + 2\sum_{n=1}^{L} \lambda_n \Big( \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_{2n-1} \partial i e_{2n}} - \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_{2n} \partial i e_{2n-1}} \Big)$$

$$= \sum_{n=1}^{L} \Big\{ \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_{2n-1} \partial e_{2n-1}} + \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_{2n} \partial e_{2n}} + 2\lambda_n \Big( \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_{2n-1} \partial i e_{2n}} - \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_{2n} \partial i e_{2n-1}} \Big)$$

$$+ \Big( \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_{2n-1} \partial i e_{2n-1}} + \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_{2n} \partial i e_{2n}} \Big) \Big\}$$
(6.3.16)

and

$$\sum_{i=2L+1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial e_i} + \sum_{i=2L+1}^{K} \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_i}$$

$$= \sum_{i=2L+1}^{K} \left( \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial e_i \partial e_i} + \frac{\partial^2 U_{\Phi,\Psi}(M_{s-}+iN_{s-})}{\partial i e_i \partial i e_i} \right).$$
(6.3.17)

Now, the expression (6.3.16) is nonnegative by Corollary 6.3.7 and (6.3.17) is nonnegative by Remark 6.2.7. This gives  $I_2 \ge 0$ . Putting all the above facts together, we obtain

$$\mathbb{E}U_{\Phi,\Psi}(M_t + iN_t) \ge \mathbb{E}U_{\Phi,\Psi}(M_0 + iN_0).$$

However, by Remark 6.2.2, we have  $N_0 = 0$  almost surely, so Theorem 6.3.2 implies

$$\mathbb{E}U_{\Phi,\Psi}(M_0 + iN_0) = \mathbb{E}U_{\Phi,\Psi}(M_0) \ge 0,$$

which completes the proof.

Step 4. Now we assume that  $U_{\Phi,\Psi}$  is general (i.e., not necessarily twice integrable). We will use a standard mollification argument. Let  $\phi : X + iX \to \mathbb{R}_+$  be a  $C^{\infty}$  radial function with compact support such that  $\int_{X+iX} \phi(s) \, ds = 1$ . For each  $\varepsilon > 0$ , define  $U_{\Phi,\Psi}^{\varepsilon} : X + iX \to \mathbb{R}$  via the convolution

$$U_{\Phi,\Psi}^{\varepsilon}(x+iy) := \int_{X+iX} U_{\Phi,\Psi}(x+iy-\varepsilon s)\phi(s) \,\mathrm{d} s, \ x,y \in X.$$

Then  $U_{\Phi,\Psi}^{\varepsilon}$  is of class  $C^{\infty}$  and for any  $x \in X$  we have

$$U_{\Phi,\Psi}^{\varepsilon}(x) = \int_{X+iX} U_{\Phi,\Psi}(x-\varepsilon s)\phi(s) \,\mathrm{d}s \ge U_{\Phi,\Psi}(x) \ge 0, \tag{6.3.18}$$

since  $U_{\Phi,\Psi}$  is subharmonic (see Remark 6.2.6). Therefore, repeating the arguments from the above steps, we get

$$\mathbb{E} \int_{X+iX} \left[ |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} \Phi(M_t - \varepsilon r) - \Psi(N_t - \varepsilon u) \right] \phi(r + iu) \, \mathrm{d}s$$
  
$$\geq \mathbb{E} U_{\Phi,\Psi}^{\varepsilon}(M_t + iN_t) \geq \mathbb{E} U_{\Phi,\Psi}^{\varepsilon}(M_0) \geq 0, \tag{6.3.19}$$

where the latter bound follows from (6.3.18). Note that  $\Psi(N_t + \varepsilon u)$  is uniformly bounded (when r + iu runs over the support of  $\phi$ ) and notice that for any  $x, \varepsilon \mapsto \frac{\Phi(x-\varepsilon)+\Phi(x+\varepsilon)}{2}$  is an increasing function of  $\varepsilon > 0$ . Furthermore, we have  $\phi(r + iu) = \phi(-r + iu) \ge 0$  and hence

$$\varepsilon \mapsto \int_{X+iX} \Phi(M_t - \varepsilon r)\phi(r + iu) \,\mathrm{d}(r + iu) = \int_{X+iX} \frac{\Phi(M_t - \varepsilon r) + \Phi(M_t + \varepsilon r)}{2} \phi(r + iu) \,\mathrm{d}(r + iu),$$
(6.3.20)

decreases as  $\varepsilon \downarrow 0$ . Combining these observations with standard limiting theorems, we deduce the desired claim.

Now we prove our main result in full generality. Of course, we will exploit an appropriate limiting procedure, which enables us to deduce the claim from its finite-dimensional version just established above. *Proof of* (6.3.1) *for infinite-dimensional X*. We may assume that  $\mathbb{E}\Phi(M_t) < \infty$ , since otherwise the claim is obvious. Suppose that  $(Y_n)_{n\geq 1}$  is a sequence of finite-dimensional subspaces of  $X^*$  such that  $Y_n \subset Y_{n+1}$  for any  $n \geq 1$  and  $\overline{\bigcup_{n\geq 1} Y_n} = X^*$ . For each  $n \geq 1$  define  $X_n := Y_n^*$ , let  $P_n : Y_n \hookrightarrow X^*$  be the corresponding embedding operator and let  $P_n^* : X \to X_n$  be its adjoint (recall that *X* is reflexive). Finally, define  $\Phi_n, \Psi_n : X_n \to \mathbb{R}_+$  by the formulae

$$\Phi_n(\widetilde{x}) = \inf\{\Phi(x) : x \in X, P_n^* x = \widetilde{x}\}, \quad \Psi_n(\widetilde{x}) = \inf\{\Psi(x) : x \in X, P_n^* x = \widetilde{x}\},$$

for  $\tilde{x} \in X_n$ . In the light of Lemma 6.3.8, both  $\Phi_n$  and  $\Psi_n$  are convex functions. Moreover, by Proposition 6.3.10,

$$|\mathscr{H}_{X_n}^{\mathbb{T}}|_{\Phi_n,\Psi_n} \le |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}.$$
(6.3.21)

Let us show that the processes  $P_n^*M$  and  $P_n^*N$  are orthogonal for each  $n \ge 1$ . By the very definition, we must prove that for a fixed functional  $x^* \in X_n^*$ , the local martingales  $\langle P_n^*M, x^* \rangle$  and  $\langle P_n^*N, x^* \rangle$  are orthogonal. This follows at once from orthogonality of M, N and the identities

$$\langle P_n^* M, x^* \rangle = \langle M, P_n x^* \rangle, \qquad \langle P_n^* N, x^* \rangle = \langle N, P_n x^* \rangle. \tag{6.3.22}$$

These identities also immediately give the weak differential subordination  $P_n^* N \ll P_n^* M$ , since M, N enjoy this condition. Finally, observe that by Lemma 6.3.8, we have  $\mathbb{E}\Phi_n(P_n^* M_t) \leq \mathbb{E}\Phi(M_t) < \infty$ . Therefore, applying the finite-dimensional version of (6.3.1), we see that for each  $n \geq 1$ ,

$$\mathbb{E}\Psi_n(P_n^*N_t) \le |\mathscr{H}_{X_n}^{\mathbb{I}}|_{\Phi_n,\Psi_n} \mathbb{E}\Phi_n(P_n^*M_t) \le |\mathscr{H}_X^{\mathbb{I}}|_{\Phi,\Psi} \mathbb{E}\Phi_n(P_n^*M_t), \tag{6.3.23}$$

where the second passage is due to (6.3.21). Note that with probability 1 we have  $\Phi_n(P_n^*M_t) \nearrow \Phi(M_t)$  and  $\Psi_n(P_n^*N_t) \nearrow \Psi(N_t)$  monotonically as  $n \to \infty$  by Lemma 6.3.9. This establishes the desired estimate, by Lebesgue's monotone convergence theorem.

It remains to handle the sharpness of (6.3.1).

*Proof of the estimate*  $|\mathscr{H}_X^T|_{\Phi,\Psi} \leq C_{\Phi,\Psi,X}$ . This follows immediately from the reasoning presented in Section 6.2.2: indeed, (6.3.1) implies the corresponding bound

$$\int_{\mathbb{T}} \Psi \big( \mathscr{H}_X^{\mathbb{T}} f \big) \mathrm{d}x \le C_{\Phi, \Psi, X} \int_{\mathbb{T}} \Phi(f) \mathrm{d}x$$

for any step function  $f : \mathbb{T} \to X$ .

*Remark* 6.3.12. It is easy to see that if *X* is finite dimensional, then there is no need for  $\Phi$  to be convex. The limiting argument presented in the above proof does not need this requirement. (The only place where the convexity of  $\Phi$  is used is (6.3.20); we leave to the reader the question how to avoid this issue).

# **6.4.** APPLICATIONS

#### **6.4.1.** *Hilbert transforms on* $\mathbb{T}$ *,* $\mathbb{R}$ *, and* $\mathbb{Z}$

Let *X* be a Banach space and let  $\Phi, \Psi : X \to \mathbb{R}_+$  be continuous functions. Let  $(S, \Sigma, \mu)$  be a measure space, with *S* equal to  $\mathbb{T}, \mathbb{R}$ , or  $\mathbb{Z}$ . A function  $f : S \to X$  is called a *step function*, if it is of the form

$$f(t) = \sum_{k=1}^{N} x_k \mathbf{1}_{A_k}(t), \quad t \in S,$$

where *N* is finite,  $x_k \in X$  and  $A_k$  are intervals in *S* of a finite measure.

**Definition 6.4.1.** The *Hilbert transform*  $\mathscr{H}_X^{\mathbb{R}}$  is a linear operator that maps a step function  $f : \mathbb{R} \to X$  to the function

$$(\mathscr{H}_X^{\mathbb{R}}f)(t) := \frac{1}{\pi} \text{p.v.} \int_{\mathbb{R}} \frac{f(s)}{t-s} \, \mathrm{d}s, \quad t \in \mathbb{R}.$$
(6.4.1)

The associated  $\Phi, \Psi$ -norms  $|\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi}$  are given by a formula similar to that used previously:

$$|\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi} := \inf \bigg\{ c \in [0,\infty] : \int_{\mathbb{R}} \Psi(\mathscr{H}_X^{\mathbb{R}}f(s)) \, \mathrm{d}s \le c \int_{\mathbb{R}} \Phi(f(s)) \, \mathrm{d}s$$
 for all step functions  $f : \mathbb{R} \to X \bigg\}.$ 

**Definition 6.4.2.** The *discrete Hilbert transform*  $\mathcal{H}_X^{\text{dis}}$  is a linear operator that maps a step function  $f : \mathbb{Z} \to X$  to the function

$$(\mathcal{H}_X^{\mathrm{dis}}f)(t) := \frac{1}{\pi} \sum_{s \in \mathbb{Z} \setminus \{t\}} \frac{f(s)}{t-s}, \ t \in \mathbb{Z}.$$

The associated  $\Phi$ ,  $\Psi$ -norms  $|\mathscr{H}_X^{\text{dis}}|_{\Phi,\Psi}$  are given by

$$|\mathcal{H}_X^{\mathrm{dis}}|_{\Phi,\Psi} := \inf \left\{ c \in [0,\infty] : \sum_{s \in \mathbb{Z}} \Psi(\mathcal{H}_X^{\mathrm{dis}} f(s)) \le c \sum_{s \in \mathbb{Z}} \Phi(f(s)) \right.$$

for all step functions  $f : \mathbb{Z} \to X$ .

We will also need a certain variant of  $\Phi$ ,  $\Psi$ -norm in the periodic setting. Namely, define  $|\mathscr{H}_{X}^{\mathbb{T},0}|_{\Phi,\Psi}$  by

$$|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} := \inf \left\{ c \in [0,\infty] : \int_{\mathbb{T}} \Psi(\mathscr{H}_X^{\mathbb{T}} f(s)) ds \le c \int_{\mathbb{T}} \Phi(f(s)) ds \right.$$
 for all step functions  $f : \mathbb{T} \to X$  with  $\int_{\mathbb{T}} f(s) ds = 0 \left. \right\}$ .

The following theorem demonstrates that the norm of the Hilbert transform does not depend whether it is defined on  $\mathbb{T}$ ,  $\mathbb{R}$ , or  $\mathbb{Z}$ .

**Theorem 6.4.3.** Let X be a Banach space and let  $\Phi, \Psi : X \to \mathbb{R}$  be continuous convex functions such that  $\Phi(0) = 0$ . Then

$$|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} = |\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi} \le |\mathscr{H}_X^{\mathrm{dis}}|_{\Phi,\Psi} \le |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}.$$

Moreover, if  $\Phi$  is symmetric, then

$$|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} = |\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi} = |\mathscr{H}_X^{\mathrm{dis}}|_{\Phi,\Psi} = |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}.$$

The proof will consist of several steps.

**Proposition 6.4.4.** Let X be a Banach space and let  $\Phi, \Psi : X \to \mathbb{R}_+$  be convex functions. *Then we have* 

$$|\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi} \le |\mathscr{H}_X^{\mathrm{dis}}|_{\Phi,\Psi} \le |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$$

*Proof.* Introduce yet another Hilbert-type operator acting on step functions  $f : \mathbb{R} \rightarrow \mathbb{R}$  by

$$(\mathcal{H}_X^{\mathbb{R},\mathrm{dis}}f)(t) := \frac{1}{\pi} \sum_{s \in \mathbb{Z} \setminus \{0\}} \frac{f(t-s)}{s}, \ t \in \mathbb{R},$$

and define its  $\Phi$ ,  $\Psi$ -norm analogously. We will first prove that  $|\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi} \leq |\mathscr{H}_X^{\mathbb{R},dis}|_{\Phi,\Psi}$ . To this end, fix a step function f on  $\mathbb{R}$  and define its  $\varepsilon$ -dilation by  $f_{\varepsilon}(\cdot) := f(\varepsilon \cdot)$ . Then similarly to [103, Theorem 4.3], we have

$$\frac{\int_{\mathbb{R}} \Psi((\mathscr{H}_X^{\mathbb{R},\mathrm{dis}}f_{\varepsilon})(s))\,\mathrm{d}s}{\int_{\mathbb{R}} \Phi(f_{\varepsilon}(s))\,\mathrm{d}s} = \frac{\int_{\mathbb{R}} \Psi(\pi^{-1}\sum_{k\in\mathbb{Z}\setminus\{0\}}f_{\varepsilon}(s-k)/k)\,\mathrm{d}s}{\int_{\mathbb{R}} \Phi(f_{\varepsilon}(s))\,\mathrm{d}s}$$
$$= \frac{\int_{\mathbb{R}} \Psi(\pi^{-1}\sum_{k\in\mathbb{Z}\setminus\{0\}}\varepsilon f(\varepsilon s-\varepsilon k)/(\varepsilon k))\,\mathrm{d}(\varepsilon s)}{\int_{\mathbb{R}} \Phi(f(\varepsilon s))\,\mathrm{d}(\varepsilon s)}$$
$$= \frac{\int_{\mathbb{R}} \Psi(\pi^{-1}\sum_{k\in\mathbb{Z}\setminus\{0\}}\varepsilon f(s-\varepsilon k)/(\varepsilon k))\,\mathrm{d}s}{\int_{\mathbb{R}} \Phi(f(s))\,\mathrm{d}s}.$$

Since  $\frac{1}{\pi} \sum_{k \in \mathbb{Z} \setminus \{0\}} \frac{f(s - \varepsilon k)}{\varepsilon k} \varepsilon \to \mathscr{H}_X^{\mathbb{R}} f(s)$  for a.e.  $s \in \mathbb{R}$ , Fatou's lemma yields

$$\begin{aligned} |\mathcal{H}_X^{\mathbb{R}}|_{\Phi,\Psi} &= \sup_{f \in F_X^{\text{step}}} \frac{\int_{\mathbb{R}} \Psi(\mathcal{H}_X^{\mathbb{R}} f(s)) \, \mathrm{d}s}{\int_{\mathbb{R}} \Phi(f(s)) \, \mathrm{d}s} \leq \sup_{f \in F_X^{\text{step}}} \liminf_{\varepsilon \to 0} \frac{\int_{\mathbb{R}} \Psi((\mathcal{H}_X^{\mathbb{R}, \text{dis}} f_{\varepsilon})(s)) \, \mathrm{d}s}{\int_{\mathbb{R}} \Phi(f_{\varepsilon}(s)) \, \mathrm{d}s} \\ &\leq |\mathcal{H}_X^{\mathbb{R}, \text{dis}}|_{\Phi,\Psi} = |\mathcal{H}_X^{\text{dis}}|_{\Phi,\Psi}. \end{aligned}$$

where the latter equality follows from the direct repetition of the arguments from [103, Theorem 4.2]. This gives us the first inequality of the assertion. The proof of the fact that  $|\mathscr{H}_X^{\text{dis}}|_{\Phi,\Psi} \leq |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$  follows word-by-word from the infinite-dimensional analogue of the recent approach of Bañuelos and Kwaśnicki [12] combined with the estimate (6.3.1).

**Theorem 6.4.5.** Let X be a Banach space and let  $\Phi, \Psi : X \to \mathbb{R}_+$  be continuous functions. Then  $|\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi} \leq |\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi}$ . *Proof.* Fix a step function  $f : \mathbb{R} \to X$ . It takes only a finite number of values, so we may assume that *X* is finite dimensional (which will guarantee the validity of the reasoning below). For any  $n \ge 1$ , introduce the function  $g_n : \mathbb{R} \to X$  by

$$g_n(x) = \frac{1}{2\pi n} \int_{-\pi n}^{\pi n} f(t) \cot \frac{x-t}{2n} \mathrm{d}t, \ x \in \mathbb{R}.$$

It follows from the observation of Zygmund [194, p. 256] that  $g_n \to \mathcal{H}_X^{\mathbb{R}} f$  a.e. as  $n \to \infty$ . On the other hand, the function  $x \mapsto g_n(nx)$ ,  $|x| \le \pi$ , is precisely the periodic Hilbert transform of the function  $x \mapsto f(nx)$ ,  $|x| \le \pi$  (see (6.2.1)). Therefore, it is also the periodic Hilbert transform of the *centered* function

$$x \mapsto f(nx) - \frac{1}{2\pi} \int_{-\pi}^{\pi} f(ns) \mathrm{d}s, \qquad |x| \le \pi.$$

Clearly, the latter is a step function. Consequently, by Fatou's lemma and the definition of  $|\mathscr{H}_{X}^{\mathbb{T},0}|_{\Phi,\Psi}$ ,

$$\begin{split} \int_{\mathbb{R}} \Psi(\mathscr{H}_{X}^{\mathbb{R}}f) \mathrm{d}x &\leq \liminf_{n \to \infty} \int_{-\pi n}^{\pi n} \Psi(g_{n}) \mathrm{d}x \\ &= \liminf_{n \to \infty} \int_{-\pi}^{\pi} \Psi(g_{n}(nx)) n \mathrm{d}x \\ &\leq |\mathscr{H}_{X}^{\mathbb{T},0}|_{\Phi,\Psi} \liminf_{n \to \infty} \int_{-\pi}^{\pi} \Phi\left(f(nx) - \frac{1}{2\pi} \int_{-\pi}^{\pi} f(ns) \mathrm{d}s\right) n \mathrm{d}x \\ &= |\mathscr{H}_{X}^{\mathbb{T},0}|_{\Phi,\Psi} \liminf_{n \to \infty} \int_{-\pi n}^{\pi n} \Phi\left(f(x) - \frac{1}{2\pi n} \int_{-\pi n}^{\pi n} f(s) \mathrm{d}s\right) \mathrm{d}x. \end{split}$$

However,  $\frac{1}{2\pi n} \int_{-\pi n}^{\pi n} f(s) ds \to 0$  by the fact that f is a step function. Therefore, again using this property of f and the continuity of  $\Phi$ , the last expression of the above chain equals

$$|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \int_{\mathbb{R}} \Phi(f) \mathrm{d}x$$

Since *f* was arbitrary, the result follows.

Now we turn our attention to the estimate in the reverse direction. We start from the observation that it does not hold true if  $\Phi(0) > 0$  and  $\Psi \neq 0$ . Indeed, if  $\Phi(0) > 0$ , then  $\int_{\mathbb{R}} \Phi(f) dx = \infty$  for any step function and hence  $|\mathscr{H}^{\mathbb{R}}|_{\Phi,\Psi} = 0$ . On the other hand, the condition  $\Psi \neq 0$  implies that  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} > 0$ : it is easy to construct a step function  $f : \mathbb{T} \to X$  of mean zero for which  $\int_{\mathbb{R}} \Psi(\mathscr{H}^{\mathbb{T}}f) dx > 0$ .

In other words, the inequality  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \leq |\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi}$  fails, because of obvious reasons, if  $\Phi(0) > 0$  and  $\Psi \neq 0$ . If  $\Psi$  is identically 0, then the estimate holds true: the reason is even more trivial – both sides are zero. It remains to study the key possibility when  $\Phi(0) = 0$  and  $\Psi \neq 0$ .

**Theorem 6.4.6.** Let X be a Banach space and let  $\Phi, \Psi : X \to \mathbb{R}_+$  be arbitrary continuous functions such that  $\Phi(0) = 0$  and  $\Psi \neq 0$ . Then  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \leq |\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi}$ .

*Proof.* As was mentioned above, the assumption  $\Psi \neq 0$  implies  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} > 0$ . For the sake of clarity, we split the reasoning into a few separate parts.

Step 1. Auxiliary analytic maps. Let *D* denote the open unit disc of  $\mathbb{C}$  and let  $H = R \times (0, \infty)$  be the upper halfplane. Define  $K : D \cap H \to H$  by the formula  $K(z) = -(1-z)^2/(4z)$ . It is not difficult to verify that *K* is conformal and hence so is its inverse *L*. Let us extend *L* to the continuous function on  $\overline{H}$ . It is easy to see that  $L(z) \to 0$  as  $z \to \infty$ . Furthermore, *L* maps the interval [0,1] onto  $\{e^{i\theta} : 0 \le \theta \le \pi\}$ . More precisely, we have the following formula: if  $x \in [0,1]$ , then

$$L(x) = e^{i\theta}$$
, where  $\theta \in [0, \pi]$  is uniquely determined by  $x = \sin^2(\theta/2)$ . (6.4.2)

In addition, *L* maps the set  $\mathbb{R} \setminus [0, 1]$  onto the open interval (-1, 1); precisely, we have the identity

$$L(x) = \begin{cases} 1 - 2x - 2\sqrt{x^2 - x} & \text{if } x < 0, \\ 1 - 2x + 2\sqrt{x^2 - x} & \text{if } x > 1. \end{cases}$$
(6.4.3)

In particular, we easily check that for any  $\delta > 0$ , the function *L* is bounded away from 1 outside any interval of the form  $[-\delta, 1+\delta]$  and  $|L(x)| = O(|x|^{-1})$  as  $x \to \pm \infty$ .

*Step 2.* A *function on*  $\mathbb{T}$  *and its extension to a disc.* Fix a positive number  $\varepsilon$  and pick a step function  $f : \mathbb{T} \to X$  of integral 0 such that

$$\int_{\mathbb{T}} \Psi(\mathscr{H}_X^{\mathbb{T}} f) \mathrm{d}x > (|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} - \varepsilon) \cdot \int_{\mathbb{T}} \Phi(f) \mathrm{d}x.$$

We may assume that *X* is finite-dimensional, restricting to the range of *f* if necessary. Given a big number R > 0, consider a continuous function  $\kappa^R : X \to [0, 1]$  equal to 1 on B(0, R) and equal to 0 outside B(0, 2R). Set  $\Psi^R(x) = \Psi(x) \cdot \kappa^R(x)$  for  $x \in X$ . Note that  $\Psi^R$  is uniformly continuous, since it is continuous and supported on the compact ball B(0, 2R) (recall that *X* is finite dimensional). By Lebesgue's monotone convergence theorem, if *R* is sufficiently big, we also have

$$\int_{\mathbb{T}} \Psi^{R}(\mathscr{H}_{X}^{\mathbb{T}}f) \mathrm{d}x > (|\mathscr{H}_{X}^{\mathbb{T},0}|_{\Phi,\Psi} - \varepsilon) \cdot \int_{\mathbb{T}} \Phi(f) \mathrm{d}x.$$
(6.4.4)

There is an analytic function  $F: D \to X + iX$  with the property that the radial limit  $\lim_{r\to 1^-} F(re^{i\theta})$  is equal to  $f(e^{i\theta}) + i\mathcal{H}_X^{\mathbb{T}}f(e^{i\theta})$  for almost all  $|\theta| \le \pi$ . Note that we have

$$F(0) = \frac{1}{2\pi} \int_{\mathbb{T}} f \, \mathrm{d}x + i \cdot 0 = 0 \tag{6.4.5}$$

and that the "real part" of *F* is bounded (by the supremum norm of *f*). Consider the analytic function  $M_n: H \rightarrow X + iX$  given by the composition

$$M_n(z) = F(L^{2n}(z))$$

and decompose it as  $M_n(z) = \operatorname{Re} M_n(z) + i \operatorname{Im} M_n(z)$ , with  $\operatorname{Re} M_n$  and  $\operatorname{Im} M_n$  taking values in *X*. Observe that for each *n* the function  $\operatorname{Re} M_n$  is bounded by the supremum norm of *f* (which is directly inherited from the "real part" of the function

*F*). In addition, the function  $h = \mathbf{1}_{[0,1]} \operatorname{Re} M_n$  is a step function (with the number of steps depending on *n* and going to infinity). Since  $\lim_{z\to\infty} L(z) = 0$ , we have  $\lim_{z\to\infty} M_n(z) = 0$  and therefore  $\mathscr{H}_x^{\mathbb{T}} \operatorname{Re} M_n(x) = \operatorname{Im} M_n(x)$  for  $x \in \mathbb{R}$ .

Step 3. Calculations. We compute that

$$\begin{split} \int_{\mathbb{R}} \Phi(h(x)) \, \mathrm{d}x &= \int_{0}^{1} \Phi\left(\operatorname{Re} M_{n}(x)\right) \right) \mathrm{d}x \\ &= \int_{0}^{1} \Psi\left(f(L^{2n}(x))\right) \mathrm{d}x \\ &= \frac{1}{2} \int_{0}^{\pi} \Phi\left(f(e^{2in\theta})\right) \sin\theta \mathrm{d}\theta \\ &= \frac{1}{2} \int_{0}^{2n\pi} \Phi\left(f(e^{i\theta})\right) \sin\left(\frac{\theta}{2n}\right) \frac{\mathrm{d}\theta}{2n} \\ &= \frac{1}{2} \int_{0}^{2\pi} \Phi\left(f(e^{i\theta})\right) \sum_{k=0}^{n-1} \sin\left(\frac{k\pi}{n} + \frac{\theta}{2n}\right) \frac{\mathrm{d}\theta}{2n} \\ &= \frac{1}{2} \int_{0}^{2\pi} \Phi\left(f(e^{i\theta})\right) \frac{\cos\left(\frac{\theta-\pi}{n}\right)}{2n\sin\left(\frac{\pi}{2n}\right)} \mathrm{d}\theta \\ &= \frac{n - \infty}{2\pi} \int_{0}^{2\pi} \Phi\left(f(e^{i\theta})\right) \mathrm{d}\theta. \end{split}$$
(6.4.6)

Now, let us similarly handle the integral  $\int_{\mathbb{R}} \Psi^{K}(\mathcal{H}^{\mathbb{R}}h) dx$ . We have

$$\int_{\mathbb{R}} \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{R}} h(x) \right) dx$$

$$\geq \int_{0}^{1} \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{R}} h(x) \right) dx$$

$$= \int_{0}^{1} \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{R}} \operatorname{Re} M_{n} - \mathscr{H}_{X}^{\mathbb{R}} (\mathbf{1}_{\mathbb{R} \setminus [0,1]} \operatorname{Re} M_{n}) \right) dx \qquad (6.4.7)$$

$$= \int_{0}^{1} \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{R}} \operatorname{Re} M_{n} \right) dx$$

$$+ \int_{0}^{1} \left[ \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{R}} \operatorname{Re} M_{n} - \mathscr{H}_{X}^{\mathbb{R}} (\mathbf{1}_{\mathbb{R} \setminus [0,1]} \operatorname{Re} M_{n}) \right) - \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{R}} \operatorname{Re} M_{n} \right) \right] dx.$$

Now, we have  $\mathscr{H}_X^{\mathbb{R}} \operatorname{Re} M_n(x) = \operatorname{Im} M_n(x) = \mathscr{H}_X^{\mathbb{T}} f(L^{2n}(x))$ , so a calculation similar to that in (6.4.6) gives

$$\int_0^1 \Psi^R(\mathscr{H}_X^{\mathbb{R}} \operatorname{Re} M_n) \mathrm{d} x \xrightarrow{n \to \infty} \frac{1}{2\pi} \int_0^{2\pi} \Psi^R \Big( \mathscr{H}_X^{\mathbb{T}} f(e^{i\theta}) \Big) \mathrm{d} \theta.$$

To deal with the last integral in (6.4.7) we will first show that  $\mathscr{H}_X^{\mathbb{R}}(\mathbf{1}_{\mathbb{R}\setminus[0,1]}\operatorname{Re} M_n)$  converges to 0 in  $L^2$ , as  $n \to \infty$ . To this end, recall that *X* is finite-dimensional and hence it has the UMD property. Consequently, by [79, Corollary 5.2.11]

$$\int_{\mathbb{R}} |\mathscr{H}_X^{\mathbb{R}}(\mathbf{1}_{\mathbb{R}\setminus[0,1]}\operatorname{Re} M_n)|^2 \mathrm{d}x \le C_X \int_{\mathbb{R}\setminus[0,1]} |\operatorname{Re} M_n|^2 \mathrm{d}x$$
(6.4.8)

for some constant  $C_X$  depending only on X. Fix an arbitrary  $\eta > 0$ . As we have already noted above,  $\operatorname{Re} M_n$  is bounded by the supremum norm of f. Setting  $\delta = \eta/(C_X \sup_X ||f||^2)$ , we see that

$$\int_{(-\delta,0)} |\operatorname{Re} M_n(x)|^2 \mathrm{d}x + \int_{(1,1+\delta)} |\operatorname{Re} M_n(x)|^2 \mathrm{d}x \le 2\eta C_X^{-1}.$$
(6.4.9)

Furthermore, recall that L maps  $\mathbb{R} \setminus [0,1]$  onto (-1,1), it is bounded away from 1 outside  $[-\delta, 1+\delta]$  and  $|L(x)| = O(|x|^{-1})$  as  $x \to \pm \infty$ . Since F is analytic and vanishes at 0, we conclude that  $M_n(x) = F(L^{2n}(x)) = O(|x|^{-2n})$  and hence

$$\lim_{n \to \infty} \int_{\mathbb{R} \setminus [-\delta, 1+\delta]} |\operatorname{Re} M_n(x)|^2 \mathrm{d}x = 0.$$
(6.4.10)

Putting (6.4.8), (6.4.9) and (6.4.10) together, we see that if *n* is sufficiently large, then  $\int_{\mathbb{R}} |\mathscr{H}_X^{\mathbb{R}}(\mathbf{1}_{\mathbb{R}\setminus[0,1]} \operatorname{Re} M_n)|^2 dx \leq 3\eta$  and the aforementioned convergence in  $L^2$  holds. In particular, passing to a subsequence if necessary, we see that  $\mathscr{H}_X^{\mathbb{R}}(\mathbf{1}_{\mathbb{R}\setminus[0,1]} \operatorname{Re} M_n) \to 0$  almost everywhere. However, as we have already mentioned above, the function  $\Psi^R$  is uniformly continuous, so the expression in the square brackets in the last term in (6.4.7) converges to zero almost everywhere. In addition, this expression is bounded in absolute value by  $\sup \Psi^R$ . Consequently, by Lebesgue's dominated convergence theorem, the last integral in (6.4.7) converges to 0 as  $n \to \infty$ . Putting all the above facts together, we see that if *n* is sufficiently large, then

$$\int_{\mathbb{R}} \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{R}} h(x) \right) \mathrm{d}x \geq (1-\varepsilon) \cdot \frac{1}{2\pi} \int_{0}^{2\pi} \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{T}} f(e^{i\theta}) \right) \mathrm{d}\theta.$$

Combining this with (6.4.4) and (6.4.6), we obtain that for *n* large enough we have

$$\int_{\mathbb{R}} \Psi(\mathscr{H}_{X}^{\mathbb{R}}h(x)) dx \geq \int_{\mathbb{R}} \Psi^{R} \left( \mathscr{H}_{X}^{\mathbb{R}}h(x) \right) dx \geq (1-\varepsilon) (|\mathscr{H}_{X}^{\mathbb{T},0}|_{\Phi,\Psi} - \varepsilon) \int_{\mathbb{R}} \Phi(h) dx.$$

Since *h* is a step function and  $\varepsilon$  was arbitrary, the claim follows.

*Remark* 6.4.7. Note that if  $\Psi(0) \neq 0$  then Theorem 6.4.5 and 6.4.6 do not make any sense. Indeed, if this is the case, then there exists  $\varepsilon > 0$  and R such that  $\Psi(x) \ge \varepsilon$  for any  $x \in X$  with  $||x|| \le R$ . Since for any step function  $f : \mathbb{R} \to X$  the function  $\mathscr{H}_X^{\mathbb{R}} f$  is in  $L^2(\mathbb{R}; X)$ , the set  $\{||\mathscr{H}_X^{\mathbb{R}} f|| \le R\} \subset \mathbb{R}$  is of infinite measure, so

$$\int_{\mathbb{R}} \Psi(\mathscr{H}_X^{\mathbb{R}} f(s)) \, \mathrm{d}s \ge \int_{\mathbb{R}} \mathbf{1}_{\|\mathscr{H}_X^{\mathbb{R}} f\| \le R}(s) \varepsilon \, \mathrm{d}s = \infty,$$

so  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} \ge |\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} = |\mathscr{H}_X^{\mathbb{R}}|_{\Phi,\Psi} = \infty.$ 

*Remark* 6.4.8. The finiteness of  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi}$  implies the existence of a plurisubharmonic function  $U_{\Phi,\Psi}: X + iX \to \mathbb{R}$  such that  $U_{\Phi,\Psi}(0) \ge 0$ . Hence, modifying the proof of Theorem 6.3.1, we see that the inequality (6.3.1) holds, with  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$  replaced with  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi}$ , if the dominating martingale *M* is additionally assumed to start from 0.

**Theorem 6.4.9.** Let  $\Phi, \Psi : X \to \mathbb{R}_+$  be continuous such that  $\Phi$  is symmetric (i.e.,  $\Phi(x) = \Phi(-x)$  for all  $x \in X$ ) and  $\Psi$  is convex. Then  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} = |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$ .

*Proof.* It suffices to show the estimate  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \ge |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$ . Fix  $\varepsilon > 0$ . By the definition of  $|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$ , there is a step function  $f : \mathbb{T} \to X$  such that

$$\int_{\mathbb{T}} \Psi(\mathscr{H}_X^{\mathbb{T}} f) \mathrm{d}x > (|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} - \varepsilon) \int_{\mathbb{T}} \Phi(f) \mathrm{d}x.$$
(6.4.11)

Let  $F = F_1 + iF_2$  be the analytic extension of  $f + i\mathscr{H}_X^{\mathbb{T}} f : \mathbb{T} \to X + iX$  to the unit disc and suppose that  $B = (B^1, B^2)$  is the planar Brownian motion started at 0 and stopped upon hitting  $\mathbb{T}$ . Let  $\tau = \inf\{t \ge 0 : |B_t| = 1\}$  be the lifetime of B. The processes  $M_t = F_1(B_t)$ ,  $N_t = F_2(B_t)$  are orthogonal martingales such that N is weakly differentially subordinate to M. By Fatou's lemma and Lebesgue's monotone convergence theorem (observe that f, being a step function, is bounded) we see that if t is sufficiently large, then

$$\mathbb{E}\Psi(N_t) > (|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} - \varepsilon)\mathbb{E}\Phi(M_t).$$

If the expectation of *M* is zero, then by Remark 4.20 we know that

$$\mathbb{E}\Psi(N_t) \le |\mathcal{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \mathbb{E}\Phi(M_t)$$

and hence we obtain that

$$|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \ge |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} - \varepsilon.$$
(6.4.12)

We will show that this is also true if the expectation  $\mathbf{x} = \mathbb{E}M_t$  does not vanish. To this end, consider another Brownian motion  $W = (W^1, W^2)$  in  $\mathbb{R}^2$  started at 0 and stopped upon reaching the boundary of the strip  $S = \{(x, y) : |x| \le 1\}$ . Let  $\sigma = \inf\{t : |W_t^1| = 1\}$  denote its lifetime. We may assume that W is constructed on the same probability space as B and that both processes are independent. We splice these processes as follows: set

$$\widetilde{M}_{s} = \begin{cases} xW_{s}^{1} & \text{if } s \leq \sigma \\ \text{sgn}(W_{\sigma}^{1})M_{s-\sigma} & \text{if } s > \sigma \end{cases}$$

and

$$\widetilde{N}_{s} = \begin{cases} xW_{s}^{2} & \text{if } s \leq \sigma, \\ xW_{\sigma}^{2} + N_{s-\sigma} & \text{if } s > \sigma. \end{cases}$$

In other words, the pair  $(\widetilde{M}, \widetilde{N})$  behaves like a Brownian motion evolving in the strip Sx until its first coordinate reaches x or -x, and then it starts behaving like the pair  $(M, \widetilde{N}_{\sigma} + N)$  or  $(-M, \widetilde{N}_{\sigma} + N)$ , depending on which the side of the boundary of Sx the process  $\widetilde{M}$  reaches. Note that  $\widetilde{M}$ ,  $\widetilde{N}$  are orthogonal martingales such that

 $\tilde{N}$  is weakly differentially subordinate to  $\tilde{M}$  and  $\tilde{M}_0 = 0$ . Consequently, by Remark 6.4.8 for any t,

$$\mathbb{E}\Psi(\widetilde{N}_t) \le |\mathcal{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \mathbb{E}\Phi(\widetilde{M}_t).$$
(6.4.13)

Now,

$$\mathbb{E}\Psi(\widetilde{N}_t) \ge \mathbb{E}\Psi(\widetilde{N}_t) \mathbf{1}_{\{t \ge \sigma\}} = \mathbb{E}\Psi(\mathbf{x}W_{\sigma}^2 + N_{t-\sigma}) \mathbf{1}_{\{t \ge \sigma\}}.$$

However, *W* and *B* are independent, and the random variable  $xW_{\sigma}^2$  is symmetric. Therefore, using the fact that  $\Psi$  is convex, we see that

$$\mathbb{E}\Psi(\widetilde{N}_t) \ge \mathbb{E}\Psi(N_{t-\sigma})\mathbf{1}_{\{t \ge \sigma\}}.$$

Furthermore, using the symmetry of  $\Phi$ , we have

$$\mathbb{E}\Phi(\widetilde{M}_t)1_{\{t\geq\sigma\}} = \mathbb{E}\Phi(\operatorname{sgn}(W^1_{\sigma})M_{t-\sigma})1_{\{t\geq\sigma\}} = \mathbb{E}\Phi(M_{t-\sigma})1_{\{t\geq\sigma\}}.$$

As previously, combining (6.4.11) with Fatou's lemma and Lebesgue's dominated convergence theorem, if *t* is sufficiently large, then

$$\mathbb{E}\Psi(N_{t-\sigma})\mathbf{1}_{\{t\geq\sigma\}} > (|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} - \varepsilon)\mathbb{E}\Phi(M_{t-\sigma})\mathbf{1}_{\{t\geq\sigma\}}$$

and hence also

$$\mathbb{E}\Psi(\widetilde{N}_t) > (|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} - \varepsilon) \mathbb{E}\Phi(\widetilde{M}_t) \mathbf{1}_{\{t \ge \sigma\}}.$$

But  $\lim_{t\to\infty} \mathbb{E}\Phi(\widetilde{M}_t) \mathbb{1}_{\{t<\sigma\}} = 0$ , by Lebesgue's dominated convergence theorem (we have  $\mathbb{1}_{\{t<\sigma\}} \to 0$  and the norm of  $\widetilde{M}_t$  is bounded by ||x|| for  $t \in [0, \sigma]$ ). Therefore, the preceding estimate gives

$$\mathbb{E}\Psi(\widetilde{N}_t) > (|\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi} - \varepsilon)\mathbb{E}\Phi(\widetilde{M}_t)$$

if *t* is sufficiently big. By (6.4.13), this gives (6.4.12) and completes the proof of the theorem, since  $\varepsilon$  was arbitrary.

*Remark* 6.4.10. Assume that  $|\mathscr{H}_X^T|_{\Phi,\Psi} = |\mathscr{H}_X^{T,0}|_{\Phi,\Psi}$  (this holds true under some additional assumptions on  $\Phi$  and  $\Psi$ , see Theorem 6.4.9). Then the plurisubharmonic function  $U_{\Phi,\Psi}$  considered in Remark 6.4.8 coincides with the one considered in Theorem 6.3.2, and hence we automatically have that  $U_{\Phi,\Psi}(x) \ge 0$  for all  $x \in X$ .

*Proof of Theorem 6.4.3.* The theorem follows from Proposition 6.4.4, Theorem 6.4.5, 6.4.6, 6.4.9, and the fact that  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \leq |\mathscr{H}_X^{\mathbb{T}}|_{\Phi,\Psi}$ .

*Remark* 6.4.11. Notice that Theorem 6.4.3 can not be applied to more general norms. For example, if *X* is a UMD Banach space,  $1 < q < p < \infty$ , then

$$\|\mathscr{H}_X^{\mathbb{T}}\|_{\mathscr{L}(L^p(\mathbb{T};X),L^q(\mathbb{T};X))} < \infty,$$

and

$$\|\mathscr{H}_X^{\mathbb{R}}\|_{\mathscr{L}(L^p(\mathbb{R};X),L^q(\mathbb{R};X))} = \infty$$

### 6.4.2. Decoupling constants

We turn our attention to the next important application.

**Definition 6.4.12.** Let *X* be a Banach space and let  $1 be a fixed parameter. Then we define <math>\beta_{p,X}^{\Delta,+}$  and  $\beta_{p,X}^{\Delta,-}$  to be the smallest  $\beta^+$  and  $\beta^-$  such that

$$\frac{1}{(\beta^-)^p} \mathbb{E} \Big\| \sum_{n=0}^\infty df_n \Big\|^p \le \mathbb{E} \Big\| \sum_{n=0}^\infty r_n df_n \Big\|^p \le (\beta^+)^p \mathbb{E} \Big\| \sum_{n=0}^\infty df_n \Big\|^p$$

for any finite Paley-Walsh martingale  $(f_n)_{n\geq 0}$  and any independent Rademacher sequence  $(r_n)_{n\geq 0}$ . Furthermore, we define  $\beta_{p,X}^{\gamma,+}$  and  $\beta_{p,X}^{\gamma,-}$  to be the least possible values of  $\beta^+$  and  $\beta^-$  for which

$$\frac{1}{(\beta^{-})^{p}} \mathbb{E} \left\| \int_{0}^{\infty} \phi \, \mathrm{d}W \right\|^{p} \leq \mathbb{E} \left\| \int_{0}^{\infty} \phi \, \mathrm{d}\widetilde{W} \right\|^{p} \leq (\beta^{+})^{p} \mathbb{E} \left\| \int_{0}^{\infty} \phi \, \mathrm{d}W \right\|^{p},$$

where *W* is a standard Brownian motion,  $\phi : \mathbb{R}_+ \times \Omega \to X$  is an elementary progressive process, and  $\widetilde{W}$  is another Brownian motion independent of  $\phi$  and *W*.

Decoupling constants appear naturally while working with UMD Banach spaces (see e.g. [44, 45, 61, 65, 79, 119, 176]). The following result, a natural corollary of Theorem 6.3.1 for  $\Phi(x) = \Psi(x) = ||x||^p$ , exhibits the direct connection between decoupling constants and  $\hbar_{p,X} := ||\mathcal{H}_X^T||_{\mathscr{L}(L^p(\mathbb{T};X))}$  (see Corollary 6.3.3).

**Corollary 6.4.13.** Let X be a Banach space and let 1 be a fixed parameter. Then we have

$$\hbar_{p,X} \ge \max\{\beta_{p,X}^{\gamma,+}, \beta_{p,X}^{\gamma,-}\}$$
(6.4.14)

and hence

$$\hbar_{p,X} \ge C \max\{\beta_{p,X}^{\Delta,+}, \beta_{p,X}^{\Delta,-}\}.$$
(6.4.15)

*Here*  $C = \mathbb{E}|\gamma|\mathbb{E}\sqrt{\tau}$ , where  $\gamma$  is a standard normal random variable and  $\tau = \inf\{t \ge 0 : |W_t| = 1\}$  for a standard Brownian motion W.

Note that  $\mathbb{E}\tau \leq (\mathbb{E}\sqrt{\tau})^{\frac{2}{3}}(\mathbb{E}\tau^2)^{\frac{1}{3}}$  by Hölder's inequality, so *C* in (6.4.15) is bounded from below by  $\frac{(\mathbb{E}\tau)^{\frac{3}{2}}}{(\mathbb{E}\tau^2)^{\frac{1}{2}}}\mathbb{E}|\gamma| = \frac{\sqrt{6}}{\sqrt{5\pi}} \approx 0.618$  (since  $\mathbb{E}\tau = 1$  and  $\mathbb{E}\tau^2 = \frac{5}{3}$ ).

*Proof.* The inequality (6.4.14) follows directly from the definition of  $\beta_{p,X}^{\gamma,+}$  and  $\beta_{p,X}^{\gamma,-}$ . Indeed, for any Brownian motion W, elementary progressive process  $\phi$ , and a Brownian motion  $\widetilde{W}$  independent of  $\phi$  and W we have, for any  $x^* \in X^*$ ,

$$\left[\left\langle \int_{0}^{t} \phi \, \mathrm{d}W, x^{*} \right\rangle\right]_{t} = \left[\int_{0}^{t} \left\langle \phi, x^{*} \right\rangle \, \mathrm{d}W\right]_{t} = \int_{0}^{t} |\langle \phi(s), x^{*} \rangle|^{2} \, \mathrm{d}s,$$
$$\left[\left\langle \int_{0}^{t} \phi \, \mathrm{d}\widetilde{W}, x^{*} \right\rangle\right]_{t} = \left[\int_{0}^{t} \left\langle \phi, x^{*} \right\rangle \, \mathrm{d}\widetilde{W}\right]_{t} = \int_{0}^{t} |\langle \phi(s), x^{*} \rangle|^{2} \, \mathrm{d}s,$$

so  $\int \phi dW \stackrel{w}{\ll} \int \phi d\widetilde{W} \stackrel{w}{\ll} \int \phi dW$ . Moreover, by [89, Lemma 17.10],

$$\begin{split} \left[ \left\langle \int_0^t \phi \, \mathrm{d}W, x^* \right\rangle, \left\langle \int_0^t \phi \, \mathrm{d}\widetilde{W}, x^* \right\rangle \right]_t &= \left[ \int_0^t \left\langle \phi, x^* \right\rangle \, \mathrm{d}W, \int_0^t \left\langle \phi, x^* \right\rangle \, \mathrm{d}\widetilde{W} \right]_t \\ &= \int_0^t |\left\langle \phi(s), x^* \right\rangle|^2 \, \mathrm{d}[W, \widetilde{W}]_s = 0, \end{split}$$

where the latter holds since W and  $\widetilde{W}$  are independent. Therefore  $\int \phi dW$  and  $\int \phi d\widetilde{W}$  are orthogonal local martingales satisfying the differential subordination ("in both directions"), so by Theorem 6.3.1,

$$\frac{1}{(\hbar_{p,X})^p} \mathbb{E} \left\| \int_{\mathbb{R}_+} \phi \, \mathrm{d}W \right\|^p \le \mathbb{E} \left\| \int_{\mathbb{R}_+} \phi \, \mathrm{d}\widetilde{W} \right\|^p \le (\hbar_{p,X})^p \mathbb{E} \left\| \int_{\mathbb{R}_+} \phi \, \mathrm{d}W \right\|^p.$$

Let us now turn to the second part. First notice that  $\beta_{p,X}^{\gamma,+} \ge C\beta_{p,X}^{\Delta,+}$  (see [176, (2.5)] and the discussion thereafter), so  $\hbar_{p,X} \ge \beta_{p,X}^{\gamma,+} \ge C\beta_{p,X}^{\Delta,+}$ . On the other hand, *X* can be assumed UMD (and hence reflexive), so by the discussion above we have  $\hbar_{p',X^*} \ge C\beta_{p',X^*}^{\Delta,+}$ . But  $\hbar_{p',X^*} = \hbar_{p,X}$  (since  $(\mathscr{H}_X^{\mathbb{T}})^* = \mathscr{H}_{X^*}^{\mathbb{T}}$ ), and  $\beta_{p',X^*}^{\Delta,+} \ge \beta_{p,X}^{\Delta,-}$  analogously to [62, Theorem 1], so  $\hbar_{p,X} \ge C\beta_{p,X}^{\Delta,-}$ .

*Remark* 6.4.14. Notice that (6.4.14) together with [61, Theorem 3] yields the related estimate  $\max\{\beta_{p,X}^{\gamma,+}, \beta_{p,X}^{\gamma,-}\} \le \hbar_{p,X} \le \beta_{p,X}^{\gamma,+} \beta_{p,X}^{\gamma,-}$ .

*Remark* 6.4.15. Let *X* be a UMD Banach function space. Then inequality (6.4.15) together with [91] provide the lower bound for  $\hbar_{p,X}$  in terms of  $\beta_{p,X}$  of the same order as (2.3.1). Indeed, by [91] thanks to Banach function space techniques one can show that

$$\beta_{p,X} \lesssim_p q(c_{q,X}\beta_{p,X}^{\Delta,+})^2,$$

where *q* is the cotype of *X* and  $c_{q,X}$  is the corresponding cotype constant. Therefore by applying (6.4.15) we get the following square root dependence:

$$\sqrt{eta_{p,X}} \lesssim_p \sqrt{q} c_{q,X} \hbar_{p,X}.$$

## 6.4.3. Necessity of the UMD property

Our next result answers a very natural question about the link of the number  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi}$  to the UMD property.

**Theorem 6.4.16.** Let  $\Phi$ ,  $\Psi : X \to \mathbb{R}_+$  be continuous convex functions such that  $\Psi(0) = 0$ . Assume in addition that there is a positive number *C* such that the sets  $\{x \in X : \Psi(x) < C\}$  and  $\Phi(B(0, C))$  are bounded. If  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} < \infty$ , then *X* is UMD.

*Remark* 6.4.17. It is easy to see that the assumption  $\Psi(0) = 0$  combined with the boundedness of { $\Psi < C$ } enforces the function  $\Psi$  to explode "uniformly" in the whole space. That is, if B(0, R) is the ball containing { $\Psi < C$ }, then the convexity of

Ψ implies Ψ(x) ≥ C||x||/R for all  $x \notin B(0, R)$ . Some condition of this type is necessary, as the following simple example indicates. Take  $X = ℓ_∞$  and set  $Φ(x) = |x_1|^2 = Ψ(x)$ . Then  $|\mathscr{H}_X^{\mathbb{T},0}|_{Φ,\Psi} = 1 < ∞$ , while *X* is not UMD. The reason is that the function Ψ controls only the subspace generated by the first coordinate.

*Remark* 6.4.18. Note that *X* being UMD does not imply  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} < \infty$ . Indeed, if  $\Phi$  and  $\Psi$  are of different homogeneity (i.e.  $\Phi(ax) = a^{\alpha} \Phi(x)$ ,  $\Phi(ax) = a^{\beta} \Phi(x)$  for any  $x \in X$ ,  $a \ge 0$ , and for some fixed positive  $\alpha \neq \beta$ ), then for any nonzero step function  $f : \mathbb{T} \to X$  such that  $\int_{\mathbb{T}} f(s) ds = 0$  and for any  $a \ge 0$  we have that

$$\begin{split} \int_{\mathbb{T}} \Psi(\mathscr{H}_{X}^{\mathbb{T}}f(s)) \, \mathrm{d}s &= \frac{1}{a^{\beta}} \int_{\mathbb{T}} \Psi(\mathscr{H}_{X}^{\mathbb{T}}(af)(s)) \, \mathrm{d}s \leq \frac{1}{a^{\beta}} |\mathscr{H}_{X}^{\mathbb{T},0}|_{\Phi,\Psi} \int_{\mathbb{T}} \Phi(af(s)) \, \mathrm{d}s \\ &= a^{\alpha-\beta} |\mathscr{H}_{X}^{\mathbb{T},0}|_{\Phi,\Psi} \int_{\mathbb{T}} \Phi(f(s)) \, \mathrm{d}s, \end{split}$$

so  $a^{\alpha-\beta}|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} \leq |\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi}$  for any a > 0, and since  $\alpha \neq \beta$ ,  $|\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi} = \infty$ . The classical examples of such  $\Phi$  and  $\Psi$  are  $\Phi(x) = ||x||^p$ ,  $\Psi(x) = ||x||^q$ ,  $x \in X$  for different p and q.

The proof of Theorem 6.4.16 will exploit the following four lemmas. In what follows,  $N^* = \sup_{t>0} ||N_t||$  is the maximal function of *N*.

**Lemma 6.4.19.** Under the assumptions of Theorem 6.4.16, there exists a constant  $c_1$  depending on  $\Phi$ ,  $\Psi$  and X, such that if M, N are orthogonal martingales such that N is weakly differentially subordinate to M,  $M_0 = 0$  and  $||M||_{\infty} \le c_1$ , then  $\mathbb{P}(N^* \ge 1) < 1$ .

*Proof.* Let *R* be as in Remark 6.4.17 and suppose that  $\Phi(B(0, C)) \subseteq [-R', R']$ . Then for any  $\lambda \ge 1$  we have, in the light of Remark 6.4.8,

$$\mathbb{P}(\|N_t\| \ge 1) = \mathbb{P}(R\lambda \|N_t\| \ge R\lambda) \le \frac{\mathbb{E}\Psi(R\lambda N_t)}{C\lambda} \le \frac{|\mathcal{H}_X^{\mathbb{I},0}|_{\Phi,\Psi} \mathbb{E}\Phi(R\lambda M_t)}{C\lambda}$$

It suffices to take  $\lambda = \frac{2R' |\mathscr{H}_X^{\mathbb{T},0}|_{\Phi,\Psi}}{C}$  and  $c_1 = C/(R\lambda)$ .

**Lemma 6.4.20.** Suppose that the assumptions of Theorem 6.4.16 are satisfied. Let M and N be continuous-path orthogonal martingales such that N is weakly differentially subordinate to M,  $M_0 = 0$  and  $\mathbb{P}(N^* > 1) = 1$ . Then there exist continuous-path martingales  $\widetilde{M}$ ,  $\widetilde{N}$  such that  $\widetilde{N}$  is weakly differentially subordinate to  $\widetilde{M}$ ,  $\widetilde{M}_0 = 0$ ,  $\mathbb{P}(\widetilde{N}^* > 1) \ge 1/2$  and  $\|\widetilde{M}\|_{\infty} \le 2\|M\|_1$ .

*Proof.* Define  $\tau = \inf\{t \ge 0 : ||M_t|| \ge 2||M||_1\}$  (as usual,  $\inf \emptyset = +\infty$ ) and put  $\widetilde{M} = M^{\tau}$ ,  $\widetilde{N} = N^{\tau}$ . Since *M* has continuous paths and starts from 0, we have  $\|\widetilde{M}\|_{\infty} \le 2||M||_1$ . Furthermore,  $\mathbb{P}(\widetilde{N}^* > 1) \ge \mathbb{P}(\widetilde{N} = N) \ge 1/2$ , since

$$\mathbb{P}(N \neq N) = \mathbb{P}(\tau < \infty) = \mathbb{P}(M^* \ge 2 \|M\|_1) \le 1/2$$

by [93, Theorem 1.3.8(i)].

**Lemma 6.4.21.** Suppose that the assumptions of Theorem 6.4.16 are satisfied. Then there exists a constant c > 0 such that if M, N are continuous-path orthogonal martingales such that N is weakly differentially subordinated to M,  $M_0 = 0$  and  $N^* > 1$  almost surely, then  $||M||_1 \ge c$ .

*Proof.* Let  $c_1$  be the number guaranteed by Lemma 6.4.19. Suppose that such a c does not exist. Then for any positive integer j there exist a pair  $(M^j, N^j)$  of orthogonal martingales such that  $N^j$  is weakly differentially subordinate to  $M^j$ ,  $M_0^j = 0$ ,  $\mathbb{P}((N^j)^* > 2) = 1$  and  $||M^j||_1 \le 2^{-j-1}c_1$ . By Lemma 6.4.20, for each j there is a pair  $(\widetilde{M}^j, \widetilde{N}^j)$  of orthogonal, weakly differentially subordinate martingales satisfying  $\widetilde{M}_0^j = 0$ ,  $\mathbb{P}((\widetilde{N}^j)^* > 2) \ge 1/2$  and  $||\widetilde{M}^j||_{\infty} \le 2^{-j}c_1$ . We may assume that the underlying probability space is the same for all pairs and that all the pairs are independent. For each j there is a positive number  $t_j$  such that the event

$$A_i = \{ \| \widetilde{N}_t^j \| > 2 \text{ for some } t \le t_i \}$$

has probability greater than 1/3. Set  $t_0 = 0$  and consider the martingale pair (*M*, *N*) defined as follows: if  $t \in [t_0 + t_1 + ... + t_n, t_0 + t_1 + ... + t_{n+1})$  for some *n*, then

$$M_{t} = \widetilde{M}_{t_{1}}^{1} + \widetilde{M}_{t_{2}}^{2} + \ldots + \widetilde{M}_{t_{n}}^{n} + \widetilde{M}_{t-t_{1}-t_{2}-\ldots-t_{n}}^{n+1},$$
(6.4.16)

and analogously for *N*. Then *M* and *N* are orthogonal, *N* is weakly differentially subordinate to M,  $M_0 = 0$  and

$$||M||_{\infty} \leq \sum_{j=1}^{\infty} ||\widetilde{M}^{j}||_{\infty} \leq \sum_{j=1}^{\infty} 2^{-j} c_{1} = c_{1}.$$

Furthermore, by Borel-Cantelli lemma,

$$\mathbb{P}(N^* \ge 1) \ge \mathbb{P}\left(\limsup_{j \to \infty} A_j\right) = 1,$$

since the events  $A_j$  are independent and  $\sum_{j=1}^{\infty} \mathbb{P}(A_j) = \infty$ . Therefore we have that  $\|M\|_{\infty} \leq c_1$ ,  $\mathbb{P}(N^* \geq 1) = 1$ ,  $N \ll^w M$ , and M and N are orthogonal, which contradicts the assertion of Lemma 6.4.19.

**Lemma 6.4.22.** Suppose that the assumptions of Theorem 6.4.16 are satisfied. Then there exists a positive constant C such that if M, N are continuous-path orthogonal martingales such that N is weakly differentially subordinate to M and  $M_0 = 0$ , then

$$\mathbb{P}(N^* > 1) \le C \|M\|_1. \tag{6.4.17}$$

*Proof.* Let *c* be the constant guaranteed by the previous lemma. Suppose that the assertion is not true. Then for any positive integer *j* there is a martingale pair  $(M^j, N^j)$  satisfying the usual structural properties such that

$$\mathbb{P}((N^{j})^{*} > 2) > 2^{j+1}c^{-1} \|M^{j}\|_{1}.$$
(6.4.18)

We splice these martingale pairs into one pair (M, N) as previously, however, this time we allow pairs to appear several times. More precisely, denote  $a_j = \mathbb{P}((N^*)^j > 2)$ . Consider  $\lceil 1/a_1 \rceil$  copies of  $(M^1, N^1)$ ,  $\lceil 1/a_2 \rceil$  copies of  $(M^2, N^2)$ , and so on (all the pairs are assumed to be independent). Let  $t_j$  be positive numbers such that the events  $A_j = \{ \|N_t^j\| > 2 \text{ for some } t \le t_j \}$  have probability greater than  $a_j/2$ . Splice the aforementioned independent martingale pairs (with multiplicities) into one pair (M, N) using a formula analogous to (6.4.16). Then, by (6.4.18),

$$\|M\|_{1} \leq \sum \|M^{j}\|_{1} \leq \sum_{j=1}^{\infty} \left\lceil \frac{1}{a_{j}} \right\rceil \|M^{j}\|_{1} \leq \sum_{j=1}^{\infty} \frac{2}{a_{j}} \cdot a_{j} c 2^{-j-1} = c$$

and, again by Borel-Cantelli lemma,  $\mathbb{P}(N^* > 1) = 1$ . Here we use the independence of the events  $A_i$  and

$$\sum \mathbb{P}(A_j) \ge \sum_{j=1}^{\infty} \frac{1}{a_j} \cdot \frac{a_j}{2} = \infty.$$

This contradicts Lemma 6.4.21.

*Proof of Theorem* 6.4.16. We will prove that theorem using the well-known extrapolation technique (good- $\lambda$  inequalities) of Burkholder [29].

Step 1. First we show that for any fixed  $0 < \delta < 1$  and  $\beta > 1$  there exists  $\varepsilon > 0$  depending only on  $\delta$ ,  $\beta$ , and X such that for any orthogonal *continuous-path* martingales  $M, N : \mathbb{R}_+ \times \Omega \to X$  with  $M_0 = N_0 = 0$  and  $N \stackrel{w}{\ll} M$ ,

$$\mathbb{P}(N^* > \beta\lambda, M^* \le \delta\lambda) \le \varepsilon \mathbb{P}(N^* > \lambda) \tag{6.4.19}$$

for any  $\lambda > 0$ . Without loss of generality assume that both martingales take their values in a finite-dimensional subspace of *X*. Define three stopping times

$$\mu := \inf\{t \ge 0 : \|N_t\| > \lambda\},$$
  

$$\nu := \inf\{t \ge 0 : \|M_t\| > \delta\lambda\},$$
  

$$\sigma := \inf\{t \ge 0 : \|N_t\| > \beta\lambda\}.$$
  
(6.4.20)

All the stopping times are predictable since *M* and *N* are continuous. Therefore, the equation  $U(t) = \mathbf{1}_{[\mu,\nu\wedge\sigma]}(t)$  defines a predictable process, which in turn gives rise to the martingales

$$\widetilde{M} := \int U \, \mathrm{d}M = M^{\mu} - M^{\nu \wedge \delta},$$
  

$$\widetilde{N} := \int U \, \mathrm{d}N = N^{\mu} - N^{\nu \wedge \delta}.$$
(6.4.21)

Notice that by (6.4.20) and (6.4.21),  $\widetilde{M}^* \leq 2\delta\lambda$  on  $\{\mu < \infty\}$  and  $\widetilde{M}^* = 0$  on  $\{\mu = \infty\}$ , so

$$\|\widetilde{M}\|_1 \le 2\delta\lambda \mathbb{P}(N^* > \lambda). \tag{6.4.22}$$

 $\square$ 

Since  $\widetilde{N} \stackrel{w}{\ll} \widetilde{M}$ ,  $\widetilde{M}_0 = \widetilde{N}_0 = 0$  and  $\widetilde{M}$  and  $\widetilde{N}$  are orthogonal,

$$\mathbb{P}(N^* > \beta\lambda, M^* \le \delta\lambda) \le \mathbb{P}(\widetilde{N}^* > (\beta - 1)\lambda) \stackrel{(i)}{\le} \frac{C}{(\beta - 1)\lambda} \|\widetilde{M}\|_1 \stackrel{(ii)}{\le} \frac{2\delta C}{(\beta - 1)} \mathbb{P}(N^* > \lambda),$$

where (*i*) follows from (6.4.17) with the same constant *C* depending only on *X*, and (*ii*) follows from (6.4.22). Therefore (6.4.19) holds with  $\varepsilon = 2\delta C/(\beta - 1)$ .

*Step 2.* Now a straightforward integration argument (cf. [29, Lemma 7.1]), together with Doob's maximal inequality, yield the  $L^p$  estimate

$$\sup_{t \ge 0} \|N_t\|_p \le \|N^*\|_p \le C_{p,X} \|M^*\|_p \le \frac{pC_{p,X}}{p-1} \sup_{t \ge 0} \|M_t\|_p, \qquad 1$$

for any pair of continuous, orthogonal, differentially subordinated martingales such that  $M_0 = 0$ . Here

$$C_{p,X}^{p} = \frac{\delta^{-p} \beta^{p}}{1 - \beta^{p} \cdot 2\delta C / (\beta - 1)},$$
(6.4.23)

which, if we let  $\beta = 1 + p^{-1}$  and  $\delta = (10Cp)^{-1}$ , depends only on p and the constant in (6.4.17). This in turn yields the corresponding  $L^p$  inequality for the periodic Hilbert transform for functions of integral 0. By Theorem 6.4.3 the assumption on the zero-average can be omitted, and hence X is UMD by [79, Corollary 5.2.11].

Now we will take a closer look at the classical "LlogL" estimates of Zygmund [194]. For a Banach space *X* and a step function  $f : \mathbb{T} \to X$ , we define

$$\|f\|_{L\log L(\mathbb{T};X)} := \int_{\mathbb{T}} (\|f(s)\| + 1) \log(\|f(s)\| + 1) \, \mathrm{d}s$$

and denote

$$\hbar_{L\log L,X} = |\mathscr{H}_X^{\mathbb{T}}|_{L\log L(\mathbb{T};X) \to L^1(\mathbb{T};X)} := \sup_{f:\mathbb{T} \to X \text{ step}} \frac{\|\mathscr{H}_X^{\mathbb{T}} f\|_{L^1(\mathbb{T};X)}}{\|f\|_{L\log L(\mathbb{T};X)}}.$$

*Remark* 6.4.23. In the light of Theorem 6.4.3, we have

$$\begin{split} \hbar_{L\log L,X} &= |\mathscr{H}_X^{\mathbb{T},0}|_{L\log L(\mathbb{T};X) \to L^1(\mathbb{T};X)} = |\mathscr{H}_X^{\mathbb{R}}|_{L\log L(\mathbb{R};X) \to L^1(\mathbb{R};X)} \\ &= |\mathscr{H}_X^{\text{dis}}|_{L\log L(\mathbb{Z};X) \to L^1(\mathbb{Z};X)} \end{split}$$

for any Banach space *X*.

We will establish the following statement.

**Theorem 6.4.24.** Let X be a Banach space. Then X has the UMD property if and only if  $\hbar_{L\log L,X} < \infty$ .

For the proof we will need the following lemma.

**Lemma 6.4.25.** Let X be a UMD Banach space. Then there exists a constant  $C_X$  depending only on X such that  $\hbar_{p,X} \leq C_X \frac{p}{p-1}$  for all 1 .

*Proof.* Let  $M, N : \mathbb{R}_+ \times \Omega \to X$  be continuous orthogonal martingales such that  $N \ll^w$  *M* and  $N_0 = 0$ . As we have already seen above,

$$\sup_{t\geq 0} (\mathbb{E}||N_t||^p)^{\frac{1}{p}} \leq \frac{p}{p-1} C_{p,X} \sup_{t\geq 0} (\mathbb{E}||M_t||^p)^{\frac{1}{p}},$$

where  $C_{p,X} \le 10Cpe(1-e/5)^{-1/p}$  (see (6.4.23) and the discussion following it). Therefore, if 1 , we may assume that this constant depends only on*C*(whichessentially depends only on*X*). The claim follows from the sharpness part of Theorem 6.3.1.

*Proof of Theorem* 6.4.24. The inequality  $\hbar_{L\log L,X} < \infty$  implies UMD by Theorem 6.4.16 applied to  $\Phi(x) = (||x|| + 1) \log(||x|| + 1)$  and  $\Psi(x) = ||x||, x \in X$ . The converse holds true by Lemma 6.4.25 and Yano's extrapolation argument (see e.g. [56, 182]).

## 6.4.4. Weak differential subordination of martingales: sharper L<sup>p</sup>-inequalities

As it was shown in (4.4.1), for a UMD Banach space *X*, any 1 and any*X*-valued local martingales*M*and*N* $such that <math>N \stackrel{w}{\ll} M$ , we have

$$\mathbb{E}\|N_t\|^p \le c_{p,X}^p \mathbb{E}\|M_t\|^p, \quad t \ge 0,$$

with  $c_{p,X} \leq \beta_{p,X}^2(\beta_{p,X}+1)$ . The purpose of this subsection is to show that this upper bound can be substantially improved.

**Theorem 6.4.26.** Let X be a Banach space, let  $1 and assume that M, N are local martingales satisfying <math>N \stackrel{w}{\ll} M$ . Then

$$\mathbb{E}\|N_t\|^p \le (\beta_{p,X} + \hbar_{p,X})^p \mathbb{E}\|M_t\|^p \text{ for any } t \ge 0.$$
(6.4.24)

*Remark* 6.4.27. Note that  $\hbar_{p,X} \leq \beta_{p,X}^2$  (see (2.3.1)), so (6.4.24) gives

$$(\mathbb{E} \| N_t \|^p)^{\frac{1}{p}} \le \beta_{p,X} (\beta_{p,X} + 1) (\mathbb{E} \| M_t \|^p)^{\frac{1}{p}} \quad t \ge 0,$$

which is better than (4.4.1).

For the proof of Theorem 6.4.26 we will need the Burkholder function (see p. 47).

*Remark* 6.4.28. Suppose that the Banach space *X* is finite-dimensional and let  $U : X \times X \to \mathbb{R}$  be a zigzag-concave function (e.g. the Burkholder funciton). Let  $\rho : X \times X \to \mathbb{R}_+$  be a compactly supported nonnegative function of class  $C^{\infty}$ . Then the convolution  $U_{\rho} := U * \rho : X \times X \to \mathbb{R}$  is zigzag-concave and of class  $C^{\infty}$  (see e.g. [13]).

While working with the Burkholder function  $U : X \times X \rightarrow \mathbb{R}$  we will use the following notation: for given vectors  $x, y \in X$  instead of writing

$$\frac{\partial^2 U}{\partial (x,0)^2}, \frac{\partial^2 U}{\partial (0,y)^2}, \frac{\partial^2 U}{\partial (x,0)\partial (0,y)}$$

we will write

$$\frac{\partial^2 U}{\partial x^2}, \frac{\partial^2 U}{\partial y^2}, \frac{\partial^2 U}{\partial x \partial y}$$

Therefore for the convenience of the reader throughout this subsection we always assume that the first coordinate of any vector in  $X \times X$  is x (perhaps with a subscript), while the second coordinate is y (perhaps with a subscript). The same holds for partial derivatives.

We also will need the following lemma.

**Lemma 6.4.29.** Let X be a finite-dimensional Banach space, let  $F: X \times X \to \mathbb{R}$  be a zigzagconcave function and let  $(x_0, y_0) \in X \times X$  be such that F is twice Fréchet differentiable at  $(x_0, y_0)$ . Let  $(x, y) \in X \times X$  be such that y = x. Then for each  $\lambda \in [-1, 1]$ ,

$$\frac{\partial^2 F(x_0, y_0)}{\partial x^2} + 2\lambda \frac{\partial^2 F(x_0, y_0)}{\partial x \partial y} + \frac{\partial^2 F(x_0, y_0)}{\partial y^2} \le 0.$$

*Proof.* Since the function

$$\lambda \mapsto \frac{\partial^2 F(x_0, y_0)}{\partial x^2} + 2\lambda \frac{\partial^2 F(x_0, y_0)}{\partial x \partial y} + \frac{\partial^2 F(x_0, y_0)}{\partial y^2}$$

is linear in  $\lambda \in [-1,1]$ , it is sufficient to check the cases  $\lambda = \pm 1$ . To this end notice that

$$\frac{\partial^2 F(x_0, y_0)}{\partial x^2} \pm 2 \frac{\partial^2 F(x_0, y_0)}{\partial x \partial y} + \frac{\partial^2 F(x_0, y_0)}{\partial y^2} = \frac{\partial^2}{\partial t^2} F(x_0 + tx, y_0 \pm tx) \Big|_{t=0} \le 0,$$

where the latter follows from Definition 2.10.1.

*Proof of Theorem 6.4.26.* We begin with similar reductions as in the proof of Theorem 6.3.1. First, we may assume that *X* is a finite-dimensional Banach space. Let  $d \ge 1$  be the dimension of *X*. Let  $M = M^c + M^d$  and  $N = N^c + N^d$  be the Meyer-Yoeurp decompositions (see Subsection 6.2.4). Then by Proposition 6.2.11  $N^c \stackrel{w}{\ll} M^c$  and  $N^d \stackrel{w}{\ll} M^d$ . Let  $\tau = (\tau_s)_{s\ge 0}$  be the time-change constructed in Step 1 of the proof of Theorem 6.3.1 (see also the proof of Proposition 4.4.3). So, there exists a 2*d*-dimensional standard Brownian motion *W* on an extended probability space  $(\widetilde{\Omega}, \widetilde{\mathscr{F}}, \widetilde{\mathbb{P}})$  equipped with an extended filtration  $\widetilde{\mathbb{F}} = (\widetilde{\mathscr{F}}_t)_{t\ge 0}$ , and there exist two progressively measurable processes  $\phi, \psi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(\mathbb{R}^{2d}, X)$  such that  $M^c \circ \tau = \phi \cdot W$  and  $N^c \circ \tau = \psi \cdot W$ . Let us redefine  $M := M \circ \tau$  and  $N := N \circ \tau$  (hence  $M^c := M^c \circ \tau$ ,

 $M^d := M^d \circ \tau$ ,  $N^c := N^c \circ \tau$ , and  $N^d := N^d \circ \tau$  by Theorem 2.4.25). Without loss of generality we may further assume that M and N terminate after some deterministic time:  $M_t = M_{t \wedge T}$  and  $N_t = N_{t \wedge T}$  for some fixed parameter  $T \ge 0$ . Analogously to Proposition 6.3.5 there exists a progressively measurable  $A : \mathbb{R}_+ \times \Omega \to \mathscr{L}(\mathbb{R}^{2d})$  which satisfies  $||A|| \le 1$  on  $\mathbb{R}_+ \times \Omega$  and  $\psi = \phi A$ . Let us define  $A^{\text{sym}} := \frac{A + A^T}{2}$ ,  $A^{\text{asym}} := \frac{A - A^T}{2}$ . If we set

$$N^{\text{sym}} := N^d + (\phi A^{\text{sym}}) \cdot W, \qquad N^{\text{asym}} := (\phi A^{\text{asym}}) \cdot W,$$

then  $N^{\text{sym}} \stackrel{w}{\ll} M$  and  $N^{\text{asym}} \stackrel{w}{\ll} M$ . Indeed, if  $N^{\text{sym}} = N^{\text{sym},c} + N^{\text{sym},d}$  and  $N^{\text{asym}} = N^{\text{asym},c} + N^{\text{asym},d}$  are the corresponding Meyer-Yoeurp decompositions, then  $N^{\text{sym},d} = N^d \stackrel{w}{\ll} M^d$ ,  $N^{\text{asym},d} = 0 \stackrel{w}{\ll} M^d$ , and for any  $x^* \in X^*$  and  $t \ge 0$ , we have

$$[\langle N^{\text{sym},c}, x^* \rangle]_t = \int_0^t \left\| \frac{A(s) + A^T(s)}{2} \phi^*(s) x^* \right\|^2 \mathrm{d}s \le \int_0^t \left\| \frac{A(s) + A^T(s)}{2} \right\|^2 \|\phi^*(s) x^*\|^2 \mathrm{d}s \le \int_0^t \|\phi^*(s) x^*\|^2 \mathrm{d}s = [\langle M^c, x^* \rangle]_t.$$

Here  $\left\|\frac{A(s)+A^{T}(s)}{2}\right\| \le 1$  by the triangle inequality. Therefore  $N^{\text{sym},c} \stackrel{w}{\ll} M^{c}$  and, analogously,  $N^{\text{asym},c} \stackrel{w}{\ll} M^{c}$ , so the weak differential subordination holds by virtue of Proposition 6.2.11.

Let us now show that

$$\mathbb{E}\|N_t^{\operatorname{asym}}\|^p \le \hbar_{p,X}^p \mathbb{E}\|M_t\|^p \quad \text{for } t \ge 0.$$
(6.4.25)

We have  $N_0^{\text{asym}} = 0$  and  $N^{\text{asym}} \stackrel{w}{\ll} M$ ; we will prove in addition that M and  $N^{\text{asym}}$  are orthogonal. For fixed  $x^* \in X^*$  and  $t \ge 0$  we may write

$$\begin{split} [\langle M, x^* \rangle, \langle N^{\operatorname{asym}}, x^* \rangle]_t &= [\langle M^c, x^* \rangle, \langle N^{\operatorname{asym}}, x^* \rangle]_t + [\langle M^d, x^* \rangle, \langle N^{\operatorname{asym}}, x^* \rangle]_t \\ &= [\langle M^c, x^* \rangle, \langle N^{\operatorname{asym}}, x^* \rangle]_t = [\langle \phi \cdot W, x^* \rangle, \langle (\phi A^{\operatorname{asym}}) \cdot W, x^* \rangle]_t \\ &= [\langle \phi, x^* \rangle \cdot W, \langle (\phi A^{\operatorname{asym}}), x^* \rangle \cdot W]_t \\ &= \int_0^t \langle \phi^*(s) x^*, A^{\operatorname{asym}*}(s) \phi^*(s) x^* \rangle \, \mathrm{d}s = 0, \end{split}$$

where the second equality is a consequence of pure discontinuity of  $M^d$  and continuity of  $N^{asym}$ , while the last equality follows from the fact that  $A^{asym}$  is antisymmetric. This gives the orthogonality of the processes and (6.4.25) follows from (6.3.1).

The next step is to show that

$$\mathbb{E} \|N_t^{\text{sym}}\|^p \le \beta_{p,X}^p \mathbb{E} \|M_t\|^p \quad \text{for } t \ge 0.$$
(6.4.26)

Let  $U: X \times X \to \mathbb{R}$  be the Burkholder function guaranteed by Theorem 3.3.7. Using the same argument as in [13], we may assume that U is of class  $C^{\infty}$  (see also Remark 6.4.28). Applying Itô's formula (2.12.1) for a fixed basis  $(x_i)_{i=1}^d$  of X with the

dual basis  $(x_i^*)_{i=1}^d$  of  $X^*$ , we get

$$\mathbb{E}U(M_t, N_t^{\text{sym}}) = \mathbb{E}U(M_0, N_0^{\text{sym}}) + \frac{1}{2}\mathbb{E}I_1 + \mathbb{E}I_2,$$

where

$$I_{1} := \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U(M_{s-}, N_{s-}^{\text{sym}})}{\partial x_{i} \partial x_{j}} d[\langle M, x_{i}^{*} \rangle, \langle M, x_{j}^{*} \rangle]_{s}^{c} + \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U(M_{s-}, N_{s-}^{\text{sym}})}{\partial y_{i} \partial y_{j}} d[\langle N^{\text{sym}}, x_{i}^{*} \rangle, \langle N^{\text{sym}}, x_{j}^{*} \rangle]_{s}^{c}$$
(6.4.27)
$$+ 2 \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U(M_{s-}, N_{s-}^{\text{sym}})}{\partial x_{i} \partial y_{j}} d[\langle M, x_{i}^{*} \rangle, \langle N^{\text{sym}}, x_{j}^{*} \rangle]_{s}^{c}$$

and

$$\begin{split} I_2 := \sum_{0 \le s \le t} (\Delta U(M_s, N_s^{\text{sym}}) - \langle \partial_x U(M_{s-}, N_{s-}^{\text{sym}}), \Delta M_s \rangle \\ - \langle \partial_y U(M_{s-}, N_{s-}^{\text{sym}}), \Delta N_s^{\text{sym}} \rangle). \end{split}$$

Here  $\partial_x U(\cdot), \partial_y U(\cdot) \in X^*$  are the corresponding Fréchet derivatives of *U* in the first and the second *X*-subspace of the product space  $X \times X$ . Let us first show that  $\mathbb{E}I_1 \leq 0$ . Observe that

$$\sum_{i,j=1}^{d} \frac{\partial^2 U(M_{s-}, N_{s-}^{\text{sym}})}{\partial x_i \partial x_j} \langle \phi^* x_i^*, \phi^* x_j^* \rangle$$

$$+ \sum_{i,j=1}^{d} \frac{\partial^2 U(M_{s-}, N_{s-}^{\text{sym}})}{\partial y_i \partial y_j} \langle A^{\text{sym}*} \phi^* x_i^*, A^{\text{sym}*} \phi^* x_j^* \rangle \qquad (6.4.28)$$

$$+ 2 \sum_{i,j=1}^{d} \frac{\partial^2 U(M_{s-}, N_{s-}^{\text{sym}})}{\partial x_i \partial y_j} \langle \phi^* x_i^*, A^{\text{sym}*} \phi^* x_j^* \rangle \leq 0.$$

Note that by Corollary 2.11.3 and convexity of U in the second variable,

$$\sum_{i,j=1}^{d} \frac{\partial^2 U(M_{s-}, N_{s-}^{\text{sym}})}{\partial y_i \partial y_j} \langle A^{\text{sym}*} \phi^* x_i^*, A^{\text{sym}*} \phi^* x_j^* \rangle$$

$$\leq \sum_{i,j=1}^{d} \frac{\partial^2 U(M_{s-}, N_{s-}^{\text{sym}})}{\partial y_i \partial y_j} \langle \phi^* x_i^*, \phi^* x_j^* \rangle.$$
(6.4.29)

The operator  $P_{\text{Ran}(\phi^*)}A^{\text{sym}*}P_{\text{Ran}(\phi^*)}$  is symmetric and

$$\|P_{\operatorname{Ran}(\phi^*)}A^{\operatorname{sym}*}P_{\operatorname{Ran}(\phi^*)}\| \le 1.$$

Therefore by the spectral theorem there exist a [-1,1]-valued sequence  $(\lambda_i)_{i=1}^{2d}$  and an orthonormal basis  $(\tilde{h}_i)_{i=1}^{2d}$  of  $(\mathbb{R}^{2d})^*$  such that  $P_{\text{Ran}(\phi^*)}A^{\text{sym}*}P_{\text{Ran}(\phi^*)}\tilde{h}_i = \lambda_i\tilde{h}_i$ .

Moreover, since  $\operatorname{Ran}(P_{\operatorname{Ran}(\phi^*)}A^{\operatorname{sym}*}P_{\operatorname{Ran}(\phi^*)}) \subset \operatorname{Ran}(\phi^*)$ ,  $\tilde{h}_i \in \operatorname{Ran}(\phi^*)$  if  $\lambda_i \neq 0$ , so we may assume that there exists a basis  $(\tilde{x}_i)_{i=1}^d$  of X with the dual basis  $(\tilde{x}_i^*)_{i=1}^d$  such that  $\phi^* \tilde{x}_i^* = \tilde{h}_i$  for  $1 \leq i \leq m$  and  $\phi^* \tilde{x}_i^* = 0$  for  $m < i \leq d$ , where  $m \in \{0, \ldots, d\}$  is the dimension of  $\phi^*$ . By Lemma 2.11.2 the expression on the left-hand side of (6.4.28) does not depend on the choice of the basis of X and the corresponding dual basis. Therefore, using (6.4.29), it is not bigger than

$$\sum_{i=1}^{m} \frac{\partial^2 U(M_{s-},N_{s-}^{\mathrm{sym}})}{\partial x_i \partial x_i} + \sum_{i=1}^{m} \frac{\partial^2 U(M_{s-},N_{s-}^{\mathrm{sym}})}{\partial y_i \partial y_i} + 2\sum_{i=1}^{m} \lambda_i \frac{\partial^2 U(M_{s-},N_{s-}^{\mathrm{sym}})}{\partial x_i \partial y_i}$$

which is bounded from above by 0 (see Lemma 6.4.29). Thus, (6.4.28) follows. Therefore by (6.4.27) and (6.4.28), we see that

$$\begin{split} I_{1} &= \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U(M_{s-}, N_{s-}^{\text{sym}})}{\partial x_{i} \partial x_{j}} \langle \phi^{*} x_{i}^{*}, \phi^{*} x_{j}^{*} \rangle \, \mathrm{d}s \\ &+ \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U(M_{s-}, N_{s-}^{\text{sym}})}{\partial y_{i} \partial y_{j}} \langle A^{\text{sym}*} \phi^{*} x_{i}^{*}, A^{\text{sym}*} \phi^{*} x_{j}^{*} \rangle \, \mathrm{d}s \\ &+ 2 \int_{0}^{t} \sum_{i,j=1}^{d} \frac{\partial^{2} U(M_{s-}, N_{s-}^{\text{sym}})}{\partial x_{i} \partial y_{j}} \langle \phi^{*} x_{i}^{*}, A^{\text{sym}*} \phi^{*} x_{j}^{*} \rangle \, \mathrm{d}s \leq 0, \end{split}$$

and hence the expectation of  $I_1$  is nonpositive. The inequality  $I_2 \le 0$  can be proved by repeating the arguments from proof of Theorem 3.3.17, while for the estimate  $U(M_0, N_0^{\text{sym}}) \le 0$ , consult Remark 3.3.9. Therefore, we have

$$\mathbb{E}\|N_t^{\text{sym}}\|^p - \beta_{p,X}^p \mathbb{E}\|M_t\|^p \le \mathbb{E}U(M_t, N_t^{\text{sym}}) \le \mathbb{E}U(M_0, N_0^{\text{sym}}) \le 0,$$

so (6.4.26) holds. The general inequality (6.4.24) follows from (6.4.25), (6.4.26), and the triangle inequality.  $\Box$ 

*Remark* 6.4.30. It is an open problem whether there exists a Burkholder function U such that -U is plurisubharmonic (note that  $X \times X \simeq X + iX$ , so the plurisubharmonicity condition is well-defined). If it exists, then  $\hbar_{p,X} \leq \beta_{p,X}$  by Theorem 6.3.2, and so the open problem outlined in Remark 2.3.2 is solved. Unfortunately, plurisubharmonicity of -U is discovered only in the Hilbert space case (see [179] and Remark 3.5.4).

### 6.4.5. Weak differential subordination of harmonic functions

Let *X* be a Banach space, let  $d \ge 1$  be a fixed dimension and let  $\mathcal{O}$  be an open subset of  $\mathbb{R}^d$ . A function  $f : \mathcal{O} \to X$  is called *harmonic* if it takes its values in a finite-dimensional subspace of *X*, is twice-differentiable, and

$$\Delta f(s) := \sum_{i=1}^{d} \partial_i^2 f(s) = 0, \ s \in \mathcal{O}.$$

For each  $s \in \mathcal{O}$ , we define  $\nabla f(s) \in \mathcal{L}(\mathbb{R}^d, X)$  by

$$\nabla f(s)(a_1e_1+\cdots a_de_d)=\sum_{i=1}^d a_i\partial_i f(s), \ a_1,\ldots,a_d\in\mathbb{R},$$

where  $(e_i)_{i=1}^d$  is the basis of  $\mathbb{R}^d$ .

**Definition 6.4.31.** Let *X*, *d*,  $\mathcal{O}$  be as above and assume that  $f, g : \mathcal{O} \to X$  are harmonic functions. Then

1. *g* is said to be *weakly differentially subordinate* to *f* (which will be denoted by  $g \stackrel{w}{\ll} f$ ) if

$$|\langle \nabla g(s), x^* \rangle| \le |\langle \nabla f(s), x^* \rangle|, \quad s \in \mathcal{O}, x^* \in X^*; \tag{6.4.30}$$

2. *f* and *g* are said to be *orthogonal* if

$$\left\langle \langle \nabla f(s), x^* \rangle, \langle \nabla g(s), x^* \rangle \right\rangle = 0, \quad s \in \mathcal{O}, x^* \in X^*.$$
 (6.4.31)

Here  $|\cdot|$  in (6.4.30) is assumed to be the usual Euclidean norm in  $(\mathbb{R}^d)^* \simeq \mathbb{R}^d$ , and  $\langle \cdot, \cdot \rangle$  in (6.4.31) is the usual scalar product in  $(\mathbb{R}^d)^* \simeq \mathbb{R}^d$ .

The notion of weak differential subordination of vector-valued harmonic functions extends the concept originally formulated in the one-dimensional case by Burkholder [37]. As shown in that paper, the differential subordination of harmonic functions lead to the corresponding  $L^p$ -inequalities for 1 . The aimof this subsection is to show the extension of that result to general weakly dif $ferentially subordinated harmonic functions and to show more general <math>\Phi$ ,  $\Psi$ -type estimates under the orthogonality assumption. We start with recalling the definition of a harmonic measure.

**Definition 6.4.32.** Let  $\mathcal{O} \subset \mathbb{R}^d$  be an open set containing the origin and let  $\partial \mathcal{O}$  be the *boundary* of  $\mathcal{O}$ . The probability measure  $\mu$  on  $\partial \mathcal{O}$  is called a *harmonic measure with respect to the origin,* if for any Borel subset  $A \subset \partial \mathcal{O}$  we have

$$\mu(A) := \mathbb{P}\{W_{\tau} \in A\}.$$

Here  $W : \mathbb{R}_+ \times \Omega \to \mathbb{R}^d$  is a standard Brownian motion starting from 0 and  $\tau$  is the exit-time of *W* from  $\mathcal{O}$ .

**Theorem 6.4.33.** Let X be a Banach space, let  $d \ge 1$  be a fixed dimension and let  $\mathcal{O}$  be an open, bounded subset of  $\mathbb{R}^d$  containing the origin. Assume further that  $\Phi, \Psi : X \to \mathbb{R}_+$ are continuous functions such that  $\Psi$  is convex and  $\Psi(0) = 0$ . Then for any continuous functions  $f, g : \overline{\mathcal{O}} \to X$  harmonic and orthogonal on  $\mathcal{O}$  satisfying  $g \ll^w f$  and g(0) = 0 we have

$$\int_{\partial \mathcal{O}} \Psi(g(s)) \, \mathrm{d}\mu(s) \leq C_{\Phi,\Psi,X} \int_{\partial \mathcal{O}} \Phi(f(s)) \, \mathrm{d}\mu(s).$$

*Here*  $\mu$  *is the harmonic measure on*  $\partial \mathcal{O}$  *with respect to the origin and the least admissible*  $C_{\Phi,\Psi,X}$  equals  $|\mathcal{H}_X^T|_{\Phi,\Psi}$ .

*Remark* 6.4.34. We do not assume that  $\Phi$  is convex because both f and g take their values in a finite-dimensional subspace of X, see Remark 6.3.12.

*Proof of Theorem* 6.4.33. Let  $W : \mathbb{R}_+ \times \Omega \to \mathbb{R}^d$  be a standard Brownian motion and let  $\tau := \inf\{t \ge 0 : W_t \notin \mathcal{O}\}$ . Then both  $M := f(W^{\tau})$  and  $N := g(W^{\tau})$  are martingales since both f and g are harmonic on  $\mathcal{O}$  (see e.g. [89, Theorem 18.5]). By Itô's formula and the fact that both f and g are harmonic we have

$$M_t = f(W_t^{\mathsf{T}}) = f(0) + \int_0^t \nabla f(W_s^{\mathsf{T}}) \, \mathrm{d}W_s^{\mathsf{T}}, \quad t \ge 0,$$
$$N_t = g(W_t^{\mathsf{T}}) = \int_0^t \nabla g(W_s^{\mathsf{T}}) \, \mathrm{d}W_s^{\mathsf{T}}, \quad t \ge 0,$$

where in the second line we have used the equality g(0) = 0. Therefore for any  $x^* \in X^*$  and any  $0 \le u \le t$  we have

$$\begin{split} [\langle N, x^* \rangle]_t - [\langle N, x^* \rangle]_u &= \int_u^t \|\langle \nabla g(W_s^\tau), x^* \rangle \|^2 \, \mathrm{d}s \\ &\leq \int_u^t \|\langle \nabla f(W_s^\tau), x^* \rangle \|^2 \, \mathrm{d}s = [\langle M, x^* \rangle]_t - [\langle M, x^* \rangle]_u, \end{split}$$

and

$$[\langle M, x^* \rangle, \langle N, x^* \rangle]_t = \int_0^t \left\langle \langle \nabla g(W_s^{\mathsf{T}}), x^* \rangle, \langle \nabla f(W_s^{\mathsf{T}}), x^* \rangle \right\rangle \mathrm{d}s = 0.$$

Consequently, *M* and *N* are orthogonal and  $N \stackrel{w}{\ll} M$ , so

$$\begin{split} \int_{\partial \mathcal{O}} \Psi(g(s)) \, \mathrm{d}\mu(s) &= \lim_{t \to \infty} \mathbb{E} \Psi(g(W_t^{\mathsf{T}})) \\ &\leq \lim_{t \to \infty} |\mathscr{H}_X^{\mathbb{T}}|_{\Phi, \Psi} \mathbb{E} \Phi(f(W_t^{\mathsf{T}})) = |\mathscr{H}_X^{\mathbb{T}}|_{\Phi, \Psi} \int_{\partial \mathcal{O}} \Phi(f(s)) \, \mathrm{d}\mu(s). \end{split}$$

Here the first and the last equality follow from the dominated convergence theorem and the definition of  $\mu$ , while the middle one is due to Theorem 6.3.1.

The sharpness of the constant  $C_{\Phi,\Psi,X} = |\mathscr{H}_X^T|_{\Phi,\Psi}$  follows from the case d = 2,  $\mathscr{O} \subset \mathbb{R}^2$  being the unit disc, f and g being such that  $g|_{\partial\mathscr{O}} = \mathscr{H}_X^T(f|_{\partial\mathscr{O}})$  (in this case  $\mu$  becomes the probability Lebesgue measure on the unit circle  $\partial\mathscr{O}$ ).

Remark 6.4.35. Sharpness of the estimate

$$\int_{\partial \mathcal{O}} \Psi(g(s)) \, \mathrm{d}\mu(s) \leq |\mathcal{H}_X^{\mathbb{T}}|_{\Phi, \Psi} \int_{\partial \mathcal{O}} \Phi(f(s)) \, \mathrm{d}\mu(s)$$

for a fixed domain  $\mathcal{O}$  remains open. Nevertheless, in the case d = 2 and  $\mathcal{O}$  being bounded with a Jordan boundary (e.g. polygon-shaped) the sharpness follows immediately from the Carathéodory's theorem (see e.g. [63, Subsection I.3 and Appendix F]).
Let us turn to the corresponding result for  $L^p$ -estimates for differentially subordinate harmonic functions (i.e., not necessarily orthogonal).

**Theorem 6.4.36.** Let X, d and  $\mathcal{O}$  be as in the previous statement. Assume further that  $f, g: \overline{\mathcal{O}} \to X$  are continuous functions harmonic on  $\mathcal{O}$  satisfying  $g \overset{w}{\ll} f$  and  $g(0) = a_0 f(0)$  for some  $a_0 \in [-1,1]$ . Then for any 1 we have

$$\left(\int_{\partial \mathscr{O}} \|g(s)\|^p \,\mathrm{d}\mu(s)\right)^{\frac{1}{p}} \le C_{p,X} \left(\int_{\partial \mathscr{O}} \|f(s)\|^p \,\mathrm{d}\mu(s)\right)^{\frac{1}{p}},\tag{6.4.32}$$

where  $\mu$  is the harmonic measure of  $\partial \mathcal{O}$ , and the least admissible constant  $C_{p,X}$  is within the segment  $[\hbar_{p,X}, \beta_{p,X} + \hbar_{p,X}]$ .

*Remark* 6.4.37. In the scalar-valued setting it is known that the optimal  $C_{p,\mathbb{R}}$  is within the range  $[\cot(\frac{\pi}{2p^*}), p^* - 1]$ . The precise identification of  $C_{p,\mathbb{R}}$  is an open problem formulated by Burkholder in [37].

*Proof of Theorem* 6.4.36. This is quite similar to the proof of the latter statement, so we weill be brief and only indicate the necessary changes which need to be implemented. For the lower bound  $C_{p,X} \ge \hbar_{p,X}$ , modify appropriately the last sentence of the proof of Theorem 6.4.33. To show the upper bound for  $C_{p,X}$ , consider the martingales  $M := f(W^{\tau})$  and  $N := g(W^{\tau})$ , where W and  $\tau$  are as previously. Arguing as in the proof of Theorem 6.4.33, we show that  $N \stackrel{w}{\ll} M$  and hence

$$\begin{split} \left( \int_{\partial \mathcal{O}} \|g(s)\|^p \, \mathrm{d}\mu(s) \right)^{\frac{1}{p}} &= \lim_{t \to \infty} (\mathbb{E} \|N_t\|^p)^{\frac{1}{p}} \\ &\leq \limsup_{t \to \infty} (\beta_{p,X} + \hbar_{p,X}) (\mathbb{E} \|M_t\|^p)^{\frac{1}{p}} \\ &\leq \lim_{t \to \infty} (\beta_{p,X} + \hbar_{p,X}) (\mathbb{E} \|M_t\|^p)^{\frac{1}{p}} \\ &= (\beta_{p,X} + \hbar_{p,X}) \Big( \int_{\partial \mathcal{O}} \|f(s)\|^p \, \mathrm{d}\mu(s) \Big)^{\frac{1}{p}}. \end{split}$$

This completes the proof.

*Remark* 6.4.38. Note that any significant improvement for the upper bound of  $C_{p,X}$  in (6.4.32) could automatically solve an open problem. Let us outline two remarkable examples. If one could show that  $C_{p,X} \leq C\beta_{p,X}$  for some universal constant C > 0, then the open problem outlined in Remark 2.3.2 will be solved. On the other hand, if one could show that  $C_{p,X} = \hbar_{p,X}$ , then the question of Burkholder concerning the optimal constant  $C_{p,R}$  in the real-valued case would be answered (see Remark 6.4.37).

## 6.4.6. Inequalities for singular integral operators

Our final application concerns the extension of  $\Phi$ ,  $\Psi$ -estimates from the setting of nonperiodic Hilbert transform to the case of odd-kernel singular integral operators on  $\mathbb{R}^d$ . We start with the notion of a *directional Hilbert transform*: given a unit vector  $\theta \in \mathbb{R}^d$ , we define the operator  $\mathcal{H}_{\theta}$  by

$$\mathcal{H}_{\theta}f(x) = \frac{1}{\pi} \text{p.v.} \int_{\mathbb{R}} f(x-t\theta) \frac{\mathrm{d}t}{t}, \ x \in \mathbb{R}^d,$$

where *f* is a sufficiently regular real-valued function on  $\mathbb{R}^d$ , and call it the Hilbert transform of *f* in the direction  $\theta$ . For example, if  $e_1$  stands for the unit vector  $(1,0,0,\ldots,0) \in \mathbb{R}^d$ , then  $\mathscr{H}_{e_1}$  is obtained by applying the Hilbert transform in the first variable followed by the identity operator in the remaining variables. Consequently, by Fubini's theorem, we see that for any functions  $\Phi$ ,  $\Psi : X \to [0,\infty)$  and any step function  $f : \mathbb{R}^d \to X$  (finite linear combination of characteristic functions of rectangles) we have

$$\int_{\mathbb{R}^d} \Psi(\mathcal{H}_{e_1}f) \mathrm{d} x \leq |\mathcal{H}_X^{\mathbb{R}}|_{\Phi,\Psi} \int_{\mathbb{R}^d} \Phi(f) \mathrm{d} x.$$

Now, if *A* is an arbitrary orthogonal matrix, we have

$$\mathcal{H}_{Ae_1}(f)(x) = \mathcal{H}_{e_1}(f \circ A)(A^{-1}x), \ x \in \mathbb{R}^d,$$

so the above inequality holds true for any directional Hilbert transform  $\mathcal{H}_{\theta}$ .

Suppose that  $\Omega: S^{d-1} \to \mathbb{R}$  is an odd function satisfying  $||\Omega||_{L^1(S^{d-1})} = 1$  and define the associated operator

$$T_{\Omega}f(x) = \frac{2}{\pi} \mathbf{p.v.} \int_{\mathbb{R}^d} \frac{\Omega(y/|y|)}{|y|^d} f(x-y) \mathrm{d}y, \ x \in \mathbb{R}^d.$$

Then  $T_{\Omega}$  can be expressed as an average of directional Hilbert transforms:

$$T_{\Omega}f(x) = \int_{\mathbb{S}^{d-1}} \Omega(\theta) \mathcal{H}_{\theta}f(x) d\theta, \ x \in \mathbb{R}^d.$$

(Sometimes this identity is referred to as the method of rotations.) Consequently, if  $\Psi$  is convex and even, we get

$$\begin{split} \int_{\mathbb{R}^d} \Psi(T_\Omega f) \mathrm{d}x &= \int_{\mathbb{R}^d} \Psi\left(\int_{S^{d-1}} \Omega(\theta) \,\mathcal{H}_\theta f(x) \mathrm{d}\theta\right) \mathrm{d}x \\ &\leq \int_{S^{d-1}} |\Omega(\theta)| \int_{\mathbb{R}^d} \Psi(\mathcal{H}_\theta f(x)) \mathrm{d}x \mathrm{d}\theta \leq |\mathcal{H}_{\Phi,\Psi}^{\mathbb{R}}| \int_{\mathbb{R}^d} \Phi(f) \mathrm{d}x. \end{split}$$

In particular, if we fix *d* and  $j \in \{1, 2, ..., d\}$ , then the kernel

$$\Omega_{j,d}(\theta) = \frac{\pi \Gamma\left(\frac{d+1}{2}\right)}{2\pi^{(d+1)/2}} \theta_j, \qquad \theta \in S^{d-1}$$

gives rise to the Riesz transform  $R_j$ . Therefore, we see that any  $\Phi$ ,  $\Psi$ -estimate for the nonperiodic Hilbert transform (where  $\Psi$  is assumed to be a convex and odd function on X) holds true, with an unchanged constant, also in the context of Riesz transforms.

The following theorem connects the  $\Phi$ ,  $\Psi$ -norm of an odd power of a Riesz transform with the  $\Phi$ ,  $\Psi$ -norm of the Hilbert transform.

**Theorem 6.4.39.** Let X be a Banach space,  $d \ge 1$ ,  $j \in \{1, ..., d\}$ ,  $m \ge 1$  be odd. Let  $R_{j,X}$  be the corresponding Riesz transform acting on X-valued step functions,  $\Phi, \Psi : X \to \mathbb{R}_+$  be convex continuous such that  $\Psi$  is even. Then

$$|R_{j,X}^{m}|_{\Phi,\Psi} \leq \Big| \frac{2\Gamma(\frac{m+d}{2})}{\Gamma(\frac{d}{2})\Gamma(\frac{m}{2})} \mathscr{H}_{X}^{\mathbb{R}} \Big|_{\Phi,\Psi}.$$

*Proof.* The proof follows from the discussion above, the fact that  $R_{j,X}^m$  is a singular integral of the following form (see e.g. [83, p. 33]):

$$R_{j,X}^m f(x) = \frac{\Gamma(\frac{m+d}{2})}{\pi^{\frac{d}{2}} \Gamma(\frac{m}{2})} \int_{\mathbb{R}^d} \frac{f(x-y)y_j^m}{|y|^{m+d}} \,\mathrm{d}y, \ x \in \mathbb{R}^d,$$

where  $f : \mathbb{R}^d \to X$  is a step function, and the fact that the volume of  $S^{d-1}$  equals  $2\pi^{\frac{d}{2}}/\Gamma(\frac{d}{2})$ .

Notice that if *d* is fixed, then  $\frac{2\Gamma(\frac{m+d}{2})}{\Gamma(\frac{d}{2})\Gamma(\frac{m}{2})}$  is of the order  $m^{d/2}$ , so in particular we have that for all 1

$$\|R_{j,X}^m\|_{L^p(\mathbb{R}^d;X)\to L^p(\mathbb{R}^d;X)} \lesssim_d m^{d/2} \|\mathscr{H}_X^\mathbb{R}\|_{L^p(\mathbb{R};X)\to L^p(\mathbb{R};X)}$$

## 6.4.7. Hilbert operators

Let *X* be a Banach space, let *d* be a positive integer and pick  $j \in \{1, ..., d\}$ . Let  $f : \mathbb{R}_{j+}^d \to X$  be locally integrable function, where  $\mathbb{R}_{j+}^d = \{x \in \mathbb{R}^d : x_j > 0\}$ . We define  $T_j f : \mathbb{R}_{j+}^d \to X$  by the formula

$$T_j f(x) := \frac{\Gamma(\frac{d+1}{2})}{\pi^{(d+1)/2}} \int_{\mathbb{R}^d_{j_+}} \frac{f(y)(x_j + y_j)}{|x + y|^{d+1}} \, \mathrm{d}y, \ x \in \mathbb{R}^d_{j_+}.$$

This type of operators resembles Riesz transforms, but due to the domain restrictions the use of principal value is not necessary. Note that if d = 1, then  $T_j$  is the *Hilbert operator* T given by

$$Tf(x) := \frac{1}{\pi} \int_{\mathbb{R}_+} \frac{f(y)}{x+y} \, \mathrm{d}y, \ x \in \mathbb{R}_+.$$

We have the following statement.

**Theorem 6.4.40.** Let X be a Banach space,  $\Phi, \Psi : X \to \mathbb{R}_+$  be convex continuous such that  $\Psi$  is even,  $d \ge 1$ ,  $j \in \{1, ..., d\}$ , 1 . Then

$$|T_{i}|_{\Phi,\Psi} \le |\mathscr{H}_{X}^{\mathbb{R}}|_{\Phi,\Psi}. \tag{6.4.33}$$

*Proof.* By the discussion in Subsection 6.4.6 it is sufficient to show that

$$|T_j|_{\Phi,\Psi} \le |R_{j,X}|_{\Phi,\Psi}.$$

Fix a step function  $f : \mathbb{R}_{j+}^d \to X$ . Let  $\tilde{f} : \mathbb{R}^d \to X$  be such that  $\tilde{f}(x_1, \dots, x_d) = 0$  if  $x_j < 0$ and  $\tilde{f}|_{\mathbb{R}_{j+}^d} = f$ . Then  $T_j f(x) = R_{j,X} \tilde{f}(-x)$  for any  $x \in \mathbb{R}_{j+}^d$ , and therefore

$$\begin{split} \int_{\mathbb{R}_{j+}^{d}} \Psi(T_{j}f(x)) \, \mathrm{d}x &= \int_{\mathbb{R}^{d}} \Psi(R_{j,X}\tilde{f}(-x)) \mathbf{1}_{x_{j}>0} \, \mathrm{d}x \leq \int_{\mathbb{R}^{d}} \Psi(R_{j,X}\tilde{f}(-x)) \, \mathrm{d}x \\ &= \int_{\mathbb{R}^{d}} \Psi(R_{j,X}\tilde{f}(x)) \, \mathrm{d}x \leq |R_{j,X}|_{\Phi,\Psi} \int_{\mathbb{R}^{d}} \Phi(\tilde{f}(x)) \, \mathrm{d}x \\ &= |R_{j,X}|_{\Phi,\Psi} \int_{\mathbb{R}_{j+}^{d}} \Phi(f(x)) \, \mathrm{d}x. \end{split}$$

*Remark* 6.4.41. Notice that if  $\Phi$  and  $\Psi$  are of the form  $\Phi(x) = \phi(||x||)$ ,  $\Psi(x) = \psi(||x||)$  for some convex symmetric functions  $\phi, \psi : \mathbb{R} \to \mathbb{R}_+$ , then one can improve (6.4.33). Indeed, one can show that  $|T_j|_{\Phi,\Psi} = |T_j|_{\phi,\psi}$ , which does not depend on the Banach space *X*: for any step function  $f : \mathbb{R}_{j+}^d \to X$  one has that

$$\begin{split} \int_{\mathbb{R}_{j+}^{d}} \Psi(T_{j}f(x)) \, \mathrm{d}x &= \int_{\mathbb{R}_{j+}^{d}} \psi(\|T_{j}f(x)\|) \, \mathrm{d}x \\ &= \int_{\mathbb{R}_{j+}^{d}} \psi\Big( \left\|\frac{\Gamma(\frac{d+1}{2})}{\pi^{(d+1)/2}} \int_{\mathbb{R}_{j+}^{d}} \frac{f(y)(x_{j}+y_{j})}{|x+y|^{d+1}} \, \mathrm{d}y \right\| \Big) \, \mathrm{d}x \\ &\leq \int_{\mathbb{R}_{j+}^{d}} \psi\Big(\frac{\Gamma(\frac{d+1}{2})}{\pi^{(d+1)/2}} \int_{\mathbb{R}_{j+}^{d}} \frac{g(y)(x_{j}+y_{j})}{|x+y|^{d+1}} \, \mathrm{d}y \Big) \, \mathrm{d}x \\ &= \int_{\mathbb{R}_{j+}^{d}} \psi(T_{j}g(x)) \, \mathrm{d}x \leq |T_{j}|_{\phi,\psi} \int_{\mathbb{R}_{j+}^{d}} \phi(g(x)) \, \mathrm{d}x \\ &= |T_{j}|_{\phi,\psi} \int_{\mathbb{R}_{j+}^{d}} \Phi(f(x)) \, \mathrm{d}x, \end{split}$$

where  $g : \mathbb{R}_{j+}^d \to \mathbb{R}_+$  is a step function such that  $g(\cdot) = ||f(\cdot)||$ . In particular, if  $\Phi(x) = \Psi(x) = ||x||^p$  for some 1 , then by [145, Theorem 1.1]

$$\|T_{j}\|_{L^{p}(\mathbb{R}^{d}_{j_{+}};X)\to L^{p}(\mathbb{R}^{d}_{j_{+}};X)} = \sin^{-1}(\pi/p).$$

# III

# STOCHASTIC INTEGRATION AND BURKHOLDER–DAVIS–GUNDY INEQUALITIES

# **7** *L<sup>q</sup>*-valued Burkholder-Rosenthal inequalities and sharp estimates for stochastic integrals

This chapter is based on the paper  $L^q$ -valued Burkholder-Rosenthal inequalities and sharp estimates for stochastic integrals by Sjoerd Dirksen and Ivan Yaroslavtsev, see [54].

We prove sharp maximal inequalities for  $L^q$ -valued stochastic integrals with respect to any Hilbert space-valued local martingale. Our proof relies on new Burkholder-Rosenthal type inequalities for martingales taking values in an  $L^q$ -space.

2010 Mathematics Subject Classification. Primary 60G44, 60H05 Secondary: 60G42 Key words and phrases. Martingale inequalities, vector-valued stochastic integration, Burkholder-Rosenthal inequalities, decoupling, random measures, martingale decompositions.

# 7.1. INTRODUCTION

This work is motivated by the semigroup approach to stochastic partial differential equations. In this approach one first reformulates an SPDE as a stochastic ordinary differential equation in a suitable infinite-dimensional state space X and then establishes existence, uniqueness and regularity properties of a mild solution via a fixed point argument. An important ingredient for this argument is a maximal inequality for the X-valued stochastic convolution associated with the semigroup generated by the operator in the stochastic evolution equation. The semigroup approach for equations driven by Gaussian noise in Hilbert spaces is well-established and can be found in [47]. This theory has more recently been developed in two directions. Firstly, the theory for equations driven by Gaussian noise has been extended to the context of UMD Banach spaces, see e.g. [126, 127, 128]. In particular, the latter results cover  $L^q$ -spaces and Sobolev spaces and, as a consequence, allow to achieve better regularity results than the Hilbert space theory. Secondly, there has been increased interested in equations driven by discontinuous noise, e.g. Poisson- and Lévy-type noise [26, 58, 97, 111, 112, 113, 148]. The latter results are mostly restricted to the Hilbert space setting. The development of this theory in a non-Hilbertian setting is hindered by the fact that maximal inequalities for vector-valued stochastic convolutions with respect to discontinuous noise are not yet well-understood. In general, only some non-sharp maximal estimates based on geometric assumptions on the Banach space are available [52, 193]. In fact, even the theory for 'vanilla' stochastic integrals (corresponding to the trivial semigroup) is incomplete. Sharp maximal inequalities for  $L^q$ -valued stochastic integrals with respect to Poisson random measures were obtained only recently [51].

The main purpose of the present chapter is to contribute to the foundation of the semigroup approach by proving sharp estimates for  $L^q$ -valued stochastic integrals with respect to general Hilbert-space valued local martingales. In our main result, Theorem 7.5.30, we identify a suitable norm  $\|\cdot\|_{M,p,q}$  so that, for any elementary predictable processes  $\Phi$  with values in the bounded operators from H into  $L^q(S)$ ,

$$c_{p,q} \|\!\| \Phi \|\!\|_{M,p,q} \le \left( \mathbb{E} \sup_{0 \le s \le t} \left\| \int_0^s \Phi \ dM \right\|_{L^q(S)}^p \right)^{\frac{1}{p}} \le C_{p,q} \|\!\| \Phi \|\!\|_{M,p,q}, \tag{7.1.1}$$

with universal constants  $c_{p,q}$ ,  $C_{p,q}$  depending only on p and q. Let us emphasize two important points. Firstly, the norm  $\|\cdot\|_{M,p,q}$  can be computed in terms of *predictable* quantities, which is important for applications. Secondly, we call the estimates in (7.1.1) 'sharp' as these inequalities are two-sided and therefore identify the largest possible class of  $L^p$ -stochastically integrable processes. We do not require the constants  $c_{p,q}$  and  $C_{p,q}$  to be sharp or even to depend optimally on pand q. For applications to stochastic evolution equations, the precise constants in fact do not play a role. In forthcoming work together with Marinelli [53], we show that the upper bound (7.1.1) can be transferred to a large class of stochastic convolutions and apply these new estimates to obtain improved well-posedness and regularity results for the associated stochastic evolution equations in  $L^q$ -spaces.

Let us roughly sketch our approach to (7.1.1). As a starting point, we use a classical result due to Meyer [122] and Yoeurp [190] to decompose the integrator as a sum of three local martingales  $M = M^c + M^q + M^a$ , where  $M^c$  is continuous,  $M^q$  is purely discontinuous and quasi-left continuous, and  $M^a$  is purely discontinuous with accessible jumps. Sharp bounds for stochastic integrals with respect to continuous local martingales were already obtained in a more general setting [177].

To estimate the integral with respect to  $M^a$  we prove, more generally, sharp bounds for an arbitrary purely discontinuous  $L^q$ -valued local martingale with accessible jumps in Theorem 7.5.8. To establish this result we first show that such a process can be represented as an essentially discrete object, namely a sum of jumps occurring at predictable times. Using an approximation argument, the problem can then be further reduced to proving *Burkholder-Rosenthal type inequalities* for  $L^q$ valued discrete-time martingales. In general, if  $1 \le p < \infty$  and X is a Banach space, we understand under Burkholder-Rosenthal inequalities estimates for X-valued martingale difference sequences  $(d_i)$  of the form

$$c_{p,X} \parallel (d_i) \parallel_{p,X} \le \left( \mathbb{E} \left\| \sum_i d_i \right\|_X^p \right)^{\frac{1}{p}} \le C_{p,X} \parallel (d_i) \parallel_{p,X},$$
(7.1.2)

where  $||\cdot|||_{p,X}$  is a suitable norm on  $(d_i)$  which can be computed explicitly in terms of the *predictable moments* of the individual differences  $d_i$ . In the scalar-valued case, these type of inequalities were proven by Burkholder [29], following work of Rosenthal [161] in the independent case: for  $2 \le p < \infty$ 

$$\left(\mathbb{E}\left|\sum_{i=1}^{n} d_{i}\right|^{p}\right)^{\frac{1}{p}} \approx_{p} \max\left\{\left(\sum_{i=1}^{n} \mathbb{E}|d_{i}|^{p}\right)^{\frac{1}{p}}, \left(\mathbb{E}\left(\sum_{i=1}^{n} \mathbb{E}_{i-1}|d_{i}|^{2}\right)^{\frac{p}{2}}\right)^{\frac{1}{p}}\right\}.$$
(7.1.3)

Here we write  $A \leq_{\alpha} B$  if there is a constant  $c_{\alpha} > 0$  depending only on  $\alpha$  such that  $A \leq c_{\alpha}B$  and write  $A \equiv_{\alpha} B$  if both  $A \leq_{\alpha} B$  and  $B \leq_{\alpha} A$  hold. To state our  $L^q$ -valued extension, we fix a filtration  $\mathbb{F} = (\mathscr{F}_i)_{i\geq 0}$ , denote by  $(\mathbb{E}_i)_{i\geq 0}$  the associated sequence of conditional expectations and set  $\mathbb{E}_{-1} := \mathbb{E}$ . Let  $(S, \Sigma, \rho)$  be any measure space. Let us introduce the following norms on the linear space of all finite sequences  $(f_i)$  of random variables in  $L^{\infty}(\Omega; L^q(S))$ . Firstly, for  $1 \leq p, q < \infty$  we set

$$\|(f_i)\|_{S^p_q} = \left(\mathbb{E}\left\|\left(\sum_{i} \mathbb{E}_{i-1}|f_i|^2\right)^{\frac{1}{2}}\right\|_{L^q(S)}^p\right)^{\frac{1}{p}},\tag{7.1.4}$$

From the work of Junge on conditional sequence spaces [87] one can deduce that this expression is a norm. We let  $S_q^p$  denote the completion with respect to this

norm. Furthermore, we define

$$\|(f_{i})\|_{D_{q,q}^{p}} = \left( \mathbb{E} \left( \sum_{i} \mathbb{E}_{i-1} \|f_{i}\|_{L^{q}(S)}^{q} \right)^{\frac{L}{q}} \right)^{\frac{L}{p}},$$

$$\|(f_{i})\|_{D_{p,q}^{p}} = \left( \sum_{i} \mathbb{E} \|f_{i}\|_{L^{q}(S)}^{p} \right)^{\frac{1}{p}}.$$

$$(7.1.5)$$

Clearly these expressions define two norms and we let  $D_{p,q}^p$  and  $D_{q,q}^p$  denote the completions in these norms. Although these spaces depend on the filtration  $\mathbb{F}$ , we will suppress this from the notation. We let  $\hat{S}_{q}^p$ ,  $\hat{D}_{q,q}^p$  and  $\hat{D}_{p,q}^p$  denote the closed subspaces spanned by all martingale difference sequences in the above spaces.

**Theorem 7.1.1.** Let  $1 < p, q < \infty$  and let S be any measure space. If  $(d_i)$  is an  $L^q(S)$ -valued martingale difference sequence, then

$$\left(\mathbb{E}\left\|\sum_{i} d_{i}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \approx_{p,q} \|(d_{i})\|_{\hat{s}_{p,q}},\tag{7.1.6}$$

where  $\hat{s}_{p,q}$  is given by

$$\begin{split} \hat{S}^{p}_{q} \cap \hat{D}^{p}_{q,q} \cap \hat{D}^{p}_{p,q} & \text{if } 2 \leq q \leq p < \infty; \\ \hat{S}^{p}_{q} \cap (\hat{D}^{p}_{q,q} + \hat{D}^{p}_{p,q}) & \text{if } 2 \leq p \leq q < \infty; \\ (\hat{S}^{p}_{q} \cap \hat{D}^{p}_{q,q}) + \hat{D}^{p}_{p,q} & \text{if } 1 < p < 2 \leq q < \infty; \\ (\hat{S}^{p}_{q} + \hat{D}^{p}_{q,q}) \cap \hat{D}^{p}_{p,q} & \text{if } 1 < q < 2 \leq p < \infty; \\ \hat{S}^{p}_{q} + (\hat{D}^{p}_{q,q} \cap \hat{D}^{p}_{p,q}) & \text{if } 1 < q \leq p \leq 2; \\ \hat{S}^{p}_{q} + \hat{D}^{p}_{q,q} + \hat{D}^{p}_{p,q} & \text{if } 1 < p \leq q \leq 2. \end{split}$$

Consequently, if  $\mathscr{F} = \sigma(\bigcup_{i \ge 0} \mathscr{F}_i)$ , then the map  $f \mapsto (\mathbb{E}_i f - \mathbb{E}_{i-1} f)_{i \ge 0}$  induces an isomorphism between  $L_0^p(\Omega; L^q(S))$ , the subspace of mean-zero random variables in  $L^p(\Omega; L^q(S))$ , and  $s_{p,q}$ .

Let us say a few words about the proof of Theorem 7.1.1. We derive the upper bound in (7.1.6) from the known special case that the  $d_i$  are independent [51] by applying powerful decoupling techniques due to Kwapień and Woyczyński [102]. In the scalar-valued case this route was already traveled by Hitczenko [75] to deduce the optimal order of the constant in the classical Burkholder-Rosenthal inequalities (7.1.3) from the one already known for martingales with independent increments. The lower bound in (7.1.6) is derived by using a duality argument. For this purpose, we show that for  $1 < p, q < \infty$  the spaces  $s_{p,q}$  satisfy the duality relation

$$(s_{p,q})^* = s_{p',q'}, \ \frac{1}{p} + \frac{1}{p'} = 1, \ \frac{1}{q} + \frac{1}{q'} = 1.$$

The only non-trivial step in proving this duality is to show that  $(D_{q,q}^p)^* = D_{q',q'}^{p'}$ . In Section 7.4 we prove a more general result: we show that if *X* is a reflexive separable Banach space, then for the space  $H_p^{s_q}(X)$  of all adapted *X*-valued sequences  $(f_i)$ 

such that

$$\|(f_i)\|_{H^{s_q}_p(X)} = \left(\mathbb{E}\Big(\sum_i \mathbb{E}_{i-1} \|f_i\|_X^q\Big)^{\frac{p}{q}}\Big)^{\frac{1}{p}} < \infty,$$

the identity  $(H_p^{s_q}(X))^* = H_{p'}^{s_{q'}}(X^*)$  holds isomorphically with constants depending only on *p* and *q*. Somewhat surprisingly, this result only seems to be known in the literature if  $X = \mathbb{R}$  and either  $1 or <math>2 \le q \le p < \infty$  (see [181]).

Let us now discuss our approach to the integral of  $\Phi$  with respect to  $M^q$ , the purely discontinuous quasi-left continuous part of M. We first show that this integral can be represented as an integral with respect to  $\overline{\mu}^{M^q}$ , the compensated version of the random measure  $\mu^{M^q}$  that counts the jumps of  $M^q$ . In Theorem 7.5.22 we then prove the following sharp estimates for integrals with respect to  $\overline{\mu} = \mu - v$ , where  $\mu$  is any integer-valued random measure that has a compensator v that is non-atomic in time. This result covers  $\mu^{M^q}$  as a special case. To formulate our result, let  $(J, \mathscr{J})$  be a measurable space and  $\widetilde{\mathscr{P}}$  be the predictable  $\sigma$ -algebra on  $\mathbb{R}_+ \times \Omega \times J$ . For  $1 < p, q < \infty$  we define the spaces  $\hat{\mathscr{L}}_q^p$ ,  $\hat{\mathscr{D}}_{q,q}^p$  and  $\hat{\mathscr{D}}_{p,q}^p$  as the Banach spaces of all  $\widetilde{\mathscr{P}}$ -measurable functions  $F : \mathbb{R}_+ \times \Omega \times J \to L^q(S)$  for which the corresponding norms

$$\begin{split} \|F\|_{\hat{\mathscr{D}}_{q}^{p}} &:= \left(\mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J} |F|^{2} \,\mathrm{d}v\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \\ \|F\|_{\hat{\mathscr{D}}_{q,q}^{p}} &:= \left(\mathbb{E}\left(\int_{\mathbb{R}_{+}\times J} \|F\|_{L^{q}(S)}^{q} \,\mathrm{d}v\right)^{\frac{p}{q}}\right)^{\frac{1}{p}}, \\ \|F\|_{\hat{\mathscr{D}}_{p,q}^{p}} &:= \left(\mathbb{E}\int_{\mathbb{R}_{+}\times J} \|F\|_{L^{q}(S)}^{p} \,\mathrm{d}v\right)^{\frac{1}{p}} \end{split}$$

are finite.

**Theorem 7.1.2.** Fix  $1 < p, q < \infty$ . Let  $\mu$  be an optional  $\widetilde{\mathscr{P}}$ - $\sigma$ -finite random measure on  $\mathbb{R}_+ \times J$  and suppose that its compensator  $\nu$  is non-atomic in time. Then for any  $\widetilde{\mathscr{P}}$ measurable  $F : \mathbb{R}_+ \times \Omega \times J \to L^q(S)$ ,

$$\left(\mathbb{E}\sup_{0\leq s\leq t}\left\|\int_{[0,s]\times J}F(u,x)\bar{\mu}(\mathrm{d} u,\mathrm{d} x)\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \approx_{p,q} \|F\mathbf{1}_{[0,t]}\|_{\mathscr{I}_{p,q}},$$

where  $\mathcal{I}_{p,q}$  is given by

$$\begin{split} \hat{\mathcal{S}}_{q}^{p} \cap \hat{\mathcal{D}}_{q,q}^{p} \cap \hat{\mathcal{D}}_{p,q}^{p} \ \text{if} \ 2 \leq q \leq p < \infty, \\ \hat{\mathcal{S}}_{q}^{p} \cap (\hat{\mathcal{D}}_{q,q}^{p} + \hat{\mathcal{D}}_{p,q}^{p}) \ \text{if} \ 2 \leq p \leq q < \infty, \\ (\hat{\mathcal{S}}_{q}^{p} \cap \hat{\mathcal{D}}_{q,q}^{p}) + \hat{\mathcal{D}}_{p,q}^{p} \ \text{if} \ 1 < p < 2 \leq q < \infty, \\ (\hat{\mathcal{S}}_{q}^{p} + \hat{\mathcal{D}}_{q,q}^{p}) \cap \hat{\mathcal{D}}_{p,q}^{p} \ \text{if} \ 1 < q < 2 \leq p < \infty, \\ \hat{\mathcal{S}}_{q}^{p} + (\hat{\mathcal{D}}_{q,q}^{p} \cap \hat{\mathcal{D}}_{p,q}^{p}) \ \text{if} \ 1 < q \leq p \leq 2, \\ \hat{\mathcal{S}}_{q}^{p} + \hat{\mathcal{D}}_{q,q}^{p} + \hat{\mathcal{D}}_{p,q}^{p} \ \text{if} \ 1 < p \leq q \leq 2. \end{split}$$

In the scalar-valued case this result is due to A.A. Novikov [131]. In the special case that  $\mu$  is a Poisson random measure, Theorem 7.5.22 was obtained in [51]. A very different proof of the upper bounds in Theorem 7.5.22, based on tools from stochastic analysis, was discovered independently of our work in [110].

The proof of the upper bounds in Theorem 7.1.2 relies on the Burkholder-Rosenthal inequalities in Theorem 7.1.1, a Banach space-valued extension of Novikov's inequality in the special case that  $v(\mathbb{R}_+ \times J) \leq 1$  a.s. (Proposition 7.5.15), and a timechange argument. For the lower bounds, the non-trivial work is to show that

$$(\hat{\mathscr{S}}_{q}^{p})^{*} = \hat{\mathscr{S}}_{q'}^{p'}, \qquad (\hat{\mathscr{D}}_{q,q}^{p})^{*} = \hat{\mathscr{D}}_{q',q}^{p'}$$

hold isomorphically with constants depending only on p and q. These duality statements are derived in Appendix 7.A.

The chapter is structured as follows. In Section 7.3 we prove Theorem 7.1.1. Section 7.4 contains the proof of the duality for the space  $H_p^{s_q}(X)$ . In Section 7.5 we prove the sharp bounds (7.1.1). In particular, Subsections 7.5.2, 7.5.3 and 7.5.5 are dedicated to integration with respect to local martingales with accessible jumps, purely discontinuous quasi-left continuous local martingales and continuous local martingales, respectively. These three parts can be read independently of each other.

# 7.2. PRELIMINARIES

Throughout,  $(\Omega, \mathscr{F}, \mathbb{P})$  denotes a complete probability space. If *X* and *Y* are Banach spaces, then  $\mathscr{L}(X, Y)$  denotes the Banach space of bounded linear operators from *X* into *Y*.

In the following, we will frequently use duality arguments for sums and intersections of Banach spaces. Let us recall some basic facts in this direction. If (X, Y) is a compatible couple of Banach spaces, i.e., X, Y are continuously embedded in a Hausdorff topological vector space, then their intersection  $X \cap Y$  and sum X + Yare Banach spaces under the norms

$$||z||_{X \cap Y} = \max\{||z||_X, ||z||_Y\}$$

and

$$||z||_{X+Y} = \inf\{||x||_X + ||y||_Y : z = x + y, x \in X, y \in Y\}.$$

If  $X \cap Y$  is dense in both X and Y, then

$$(X \cap Y)^* = X^* + Y^*, \qquad (X + Y)^* = X^* \cap Y^*$$
(7.2.1)

hold isometrically. The duality brackets under these identifications are given by

$$\langle x^*, x \rangle = \langle x^* |_{X \cap Y}, x \rangle$$
  $(x^* \in X^* + Y^*)$ 

and

$$\langle x^*, x \rangle = \langle x^*, y \rangle + \langle x^*, z \rangle \qquad (x^* \in X^* \cap Y^*, \ x = y + z \in X + Y), \tag{7.2.2}$$

respectively, see e.g. [98, Theorem I.3.1].

The following observation facilitates a duality argument that we will use repeatly below. We provide a proof for the convenience of the reader.

**Lemma 7.2.1.** Let X and Y be Banach spaces, X be reflexive, U be a dense linear subspace of X, and let V be a dense linear subspace of  $X^*$ . Consider  $j_0 \in \mathcal{L}(U, Y)$  and  $k_0 \in \mathcal{L}(V, Y^*)$  so that ran  $j_0$  is dense in Y and  $\langle x^*, x \rangle = \langle k_0(x^*), j_0(x) \rangle$  for each  $x \in U, x^* \in V$ . Then

- (i) there exists  $j \in \mathcal{L}(X, Y)$ ,  $k \in \mathcal{L}(X^*, Y^*)$  such that  $j|_U = j_0$ ,  $k|_V = k_0$ ,
- (*ii*) ran j = Y, ran  $k = Y^*$ , in particular k and j are invertible, and
- (iii) for each  $x \in X$  and  $x^* \in X^*$

$$\frac{1}{\|k\|} \|x\| \le \|j(x)\| \le \|j\| \|x\|,$$

$$\frac{1}{\|j\|} \|x^*\| \le \|k(x^*)\| \le \|k\| \|x^*\|.$$
(7.2.3)

*Proof.* (i) holds due to the continuity of  $j_0$  and  $k_0$ , and as a consequence

$$\langle x^*, x \rangle = \langle k(x^*), j(x) \rangle, \ x \in X, x^* \in X^*$$

Notice that *j* and *k* are embeddings. Indeed,  $\langle x^*, x \rangle = \langle k(x^*), j(x) \rangle$  for each  $x \in X$ ,  $x^* \in X^*$ , so for each nonzero  $x \in X$ ,  $x^* \in X^*$  both j(x) and  $k(x^*)$  define nonzero linear functionals on  $Y^*$  and *Y* respectively, hence they are nonzero.

For (ii), fix any  $y^* \in Y^*$ . Since  $j \in \mathcal{L}(X, Y)$ , we can define  $x^* \in X^*$  by

$$\langle x^*, x \rangle := \langle y^*, j(x) \rangle, \qquad x \in X.$$

Since  $\langle x^*, x \rangle = \langle k(x^*), j(x) \rangle$  and hence  $\langle y^* - k(x^*), j(x) \rangle = 0$  for any  $x \in X$ , we conclude by density of ran *j* that  $y^* = k(x^*)$ . Thus ran  $k = Y^*$  and *k* is invertible by the bounded inverse theorem. Using reflexivity of *X* one can similarly prove the statement for *j*. To prove (iii), we note that for each  $x \in X$ 

$$\| j(x) \| = \sup_{x^* \in X^*, \| k(x^*) \| = 1} \langle k(x^*), j(x) \rangle = \sup_{x^* \in X^*, \| k(x^*) \| = 1} \langle x^*, x \rangle$$
  
$$\geq \sup_{x^* \in X^*, \| x^* \| = \frac{1}{\| k \|}} \langle x^*, x \rangle = \frac{1}{\| k \|} \| x \|,$$

and obviously  $||j(x)|| \le ||j|| ||x||$ . The estimates for *k* are derived similarly.

# 7.3. L<sup>q</sup>-VALUED BURKHOLDER-ROSENTHAL INEQUALITIES

In this section we prove Theorem 7.1.1. Our starting point is the following  $L^q$ -valued version of the classical Rosenthal inequalities [161]. For  $1 \le p, q < \infty$  let  $S_q$  and  $D_{p,q}$  be the spaces of all sequences of  $L^q(S)$ -valued random variables such the respective norms

$$\|(f_i)\|_{S_q} = \left\| \left( \sum_i \mathbb{E} |f_i|^2 \right)^{\frac{1}{2}} \right\|_{L^q(S)},$$

$$\|(f_i)\|_{D_{p,q}} = \left( \sum_i \mathbb{E} \|f_i\|_{L^q(S)}^p \right)^{\frac{1}{p}}$$
(7.3.1)

are finite. Note that the following result corresponds to a special case of Theorem 7.1.1, in which the martingale differences  $d_i$  are independent.

**Theorem 7.3.1.** [51] Let  $1 < p, q < \infty$  and let  $(S, \Sigma, \sigma)$  be a measure space. If  $(\xi_i)$  is a sequence of independent, mean-zero random variables taking values in  $L^q(S)$ , then

$$\left(\mathbb{E}\left\|\sum_{i}\xi_{i}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \approx_{p,q} \|(\xi_{i})\|_{s_{p,q}},\tag{7.3.2}$$

where  $s_{p,q}$  is given by

$$\begin{split} S_q \cap D_{q,q} \cap D_{p,q} & \text{if } 2 \leq q \leq p < \infty; \\ S_q \cap (D_{q,q} + D_{p,q}) & \text{if } 2 \leq p \leq q < \infty; \\ (S_q \cap D_{q,q}) + D_{p,q} & \text{if } 1 < p < 2 \leq q < \infty; \\ (S_q + D_{q,q}) \cap D_{p,q} & \text{if } 1 < q < 2 \leq p < \infty; \\ S_q + (D_{q,q} \cap D_{p,q}) & \text{if } 1 < q \leq p \leq 2; \\ S_q + D_{q,q} + D_{p,q} & \text{if } 1 < p \leq q \leq 2. \end{split}$$

Moreover, the estimate  $\leq_{p,q}$  in (7.3.2) remains valid if p = 1, q = 1 or both.

To derive the upper bound in Theorem 7.1.1 we use the following decoupling techniques from [102]. Let  $(\Omega, \mathscr{F}, \mathbb{P})$  be a complete probability space, let  $(\mathscr{F}_i)_{i\geq 0}$  be a filtration and let *X* be a (quasi-)Banach space. Two  $(\mathscr{F}_i)_{i\geq 1}$ -adapted sequences  $(d_i)_{i\geq 1}$  and  $(e_i)_{i\geq 1}$  of *X*-valued random variables are called *tangent* if for every  $i \geq 1$  and  $A \in \mathscr{B}(X)$ 

$$\mathbb{P}(d_i \in A | \mathscr{F}_{i-1}) = \mathbb{P}(e_i \in A | \mathscr{F}_{i-1}).$$
(7.3.3)

An  $(\mathscr{F}_i)_{i\geq 1}$ -adapted sequence  $(e_i)_{i\geq 1}$  of *X*-valued random variables is said to satisfy *condition* (*CI*) if, firstly, there is a sub- $\sigma$ -algebra  $\mathscr{G} \subset \mathscr{F}_{\infty} = \sigma(\cup_{i\geq 0}\mathscr{F}_i)$  such that for every  $i \geq 1$  and  $A \in \mathscr{B}(X)$ ,

$$\mathbb{P}(e_i \in A | \mathscr{F}_{i-1}) = \mathbb{P}(e_i \in A | \mathscr{G}) \tag{7.3.4}$$

and, secondly,  $(e_i)_{i\geq 1}$  consists of  $\mathscr{G}$ -independent random variables, i.e. for all  $n \geq 1$ and  $A_1, \ldots, A_n \in \mathscr{B}(X)$ ,

$$\mathbb{E}(\mathbf{1}_{e_1 \in A_1} \cdot \ldots \cdot \mathbf{1}_{e_n \in A_n} | \mathscr{G}) = \mathbb{E}(\mathbf{1}_{e_1 \in A_1} | \mathscr{G}) \cdot \ldots \cdot \mathbb{E}(\mathbf{1}_{e_n \in A_n} | \mathscr{G}).$$

It is shown in [102] that for every  $(\mathscr{F}_i)_{i\geq 1}$ -adapted sequence  $(d_i)_{i\geq 1}$  there exists an  $(\mathscr{F}_i)_{i\geq 1}$ -adapted sequence  $(e_i)_{i\geq 1}$  on a possibly enlarged probability space which is tangent to  $(d_i)_{i\geq 1}$  and satisfies condition (CI). This sequence is called a *decoupled tangent sequence* for  $(d_i)_{i\geq 1}$  and is unique in law.

To derive the upper bound in Theorem 7.1.1 for a given martingale difference sequence  $(d_i)_{i\geq 1}$  we apply Theorem 7.3.1 conditionally to its decoupled tangent sequence  $(e_i)_{i\geq 1}$ . For this approach to work, we will need to relate various norms on  $(d_i)_{i\geq 1}$  and  $(e_i)_{i\geq 1}$ . One of these estimates can be formulated as a Banach space property. Following [45], we say that a (quasi-)Banach space *X* satisfies *the p*-*decoupling property* if for some  $0 there is a constant <math>C_{p,X}$  such that for any complete probability space  $(\Omega, \mathcal{F}, \mathbb{P})$ , any filtration  $(\mathcal{F}_i)_{i\geq 0}$ , and any  $(\mathcal{F}_i)_{i\geq 1}$ -adapted sequence  $(d_i)_{i\geq 1}$  in  $L^p(\Omega, X)$ ,

$$\left(\mathbb{E}\left\|\sum_{i=1}^{n} d_{i}\right\|_{X}^{p}\right)^{\frac{1}{p}} \leq C_{p,X} \left(\mathbb{E}\left\|\sum_{i=1}^{n} e_{i}\right\|_{X}^{p}\right)^{\frac{1}{p}},\tag{7.3.5}$$

for all  $n \ge 1$ , where  $(e_i)_{i\ge 1}$  is the decoupled tangent sequence of  $(d_i)_{i\ge 1}$ . It is shown in [45, Theorem 4.1] that this property is independent of p, so we may simply say that X satisfies the *decoupling property* if it satisfies the p-decoupling property for some (then all) 0 . Known examples of spaces satisfying the decoupling $property are the <math>L^q(S)$ -spaces for any  $0 < q < \infty$  and UMD Banach spaces. If X is a UMD Banach space, then one can also *recouple*, meaning that for all 1 there $is a constant <math>c_{p,X}$  such that for any martingale difference sequence  $(d_i)_{i\ge 1}$  and any associated decoupled tangent sequence  $(e_i)_{i\ge 1}$ ,

$$\left(\mathbb{E} \left\| \sum_{i=1}^{n} e_{i} \right\|_{X}^{p} \right)^{\frac{1}{p}} \le c_{p,X} \left(\mathbb{E} \left\| \sum_{i=1}^{n} d_{i} \right\|_{X}^{p} \right)^{\frac{1}{p}}.$$
(7.3.6)

Conversely, if both (7.3.5) and (7.3.6) hold for some (then all) 1 , then*X*must be a UMD space. This equivalence is independently due to McConnell [119] and Hitczenko [74].

To further relate a sequence with its decoupled tangent sequence we use the following technical observation, which is a special case of [45, Lemma 2.7].

**Lemma 7.3.2.** Let X be a (quasi-)Banach space and for every  $i \ge 1$  let  $h_i : X \to X$  be a Borel measurable function. Let  $(d_i)_{i\ge 1}$  be an  $(\mathscr{F}_i)_{i\ge 1}$ -adapted sequence and  $(e_i)_{i\ge 1}$  a decoupled tangent sequence. Then  $(h_i(e_i))_{i\ge 1}$  is a decoupled tangent sequence for  $(h_i(d_i))_{i\ge 1}$ .

We are now ready to prove the announced result.

*Proof.* (*Of Theorem* 7.1.1) *Step* 1: *upper bounds.* We will only give a proof in the case  $1 \le q \le 2 \le p < \infty$ . The other cases are proved analogously. Let us write  $\mathbb{E}_{\mathscr{G}} = \mathbb{E}(\cdot|\mathscr{G})$  for brevity. By density we may assume that the  $d_i$  take values in  $L^q(S) \cap L^\infty(S)$ . Fix an arbitrary decomposition  $d_i = d_{i,1} + d_{i,2}$ , where  $d_{i,1}, d_{i,2}$  are  $L^q(S) \cap L^\infty(S)$ -valued martingale difference sequences. Let  $e_i = (e_{i,1}, e_{i,2})$  be the decoupled tangent sequence for the martingale difference sequence  $(d_{i,1}, d_{i,2})$  which takes values in  $(L^q(S) \cap L^\infty(S)) \times (L^q(S) \cap L^\infty(S))$ . Lemma 7.3.2 implies that  $d_{i,\alpha}$  is the decoupled tangent sequence for  $e_{i,\alpha}$ ,  $\alpha = 1, 2$ , and  $e_{i,1} + e_{i,2}$  is the decoupled tangent sequence for  $d_i$ . By the decoupling property for  $L^q(S)$ ,

$$\left(\mathbb{E}\left\|\sum_{i}d_{i}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \lesssim_{p,q} \left(\mathbb{E}\left\|\sum_{i}e_{i,1}+e_{i,2}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}}$$

Since the summands  $e_{i,1} + e_{i,2}$  are *G*-conditionally independent and *G*-mean zero, we can apply Theorem 7.3.1 conditionally to find, a.s.,

$$\begin{split} \left( \mathbb{E}_{\mathscr{G}} \left\| \sum_{i} e_{i,1} + e_{i,2} \right\|_{L^{q}(S)}^{p} \right)^{\frac{1}{p}} \\ \lesssim_{p,q} \max \left\{ \left\| \left( \sum_{i} \mathbb{E}_{\mathscr{G}} |e_{i,1}|^{2} \right)^{\frac{1}{2}} \right\|_{L^{q}(S)} \\ + \left( \sum_{i} \mathbb{E}_{\mathscr{G}} \|e_{i,2}\|_{L^{q}(S)}^{q} \right)^{\frac{1}{q}}, \left( \sum_{i} \mathbb{E}_{\mathscr{G}} \|e_{i,1} + e_{i,2}\|_{L^{q}(S)}^{p} \right)^{\frac{1}{p}} \right\}. \end{split}$$

Now we take L<sup>p</sup>-norms on both sides and apply the triangle inequality to obtain

$$\begin{split} \left(\mathbb{E}\left\|\sum_{i}d_{i}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \\ \lesssim_{p,q}\max\left\{\left(\mathbb{E}\left\|\left(\sum_{i}\mathbb{E}_{\mathscr{G}}|e_{i,1}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \\ +\left(\mathbb{E}\left(\sum_{i}\mathbb{E}_{\mathscr{G}}\|e_{i,2}\|_{L^{q}(S)}^{q}\right)^{\frac{p}{q}}\right)^{\frac{1}{p}}, \left(\sum_{i}\mathbb{E}\|e_{i,1}+e_{i,2}\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}}\right\} \end{split}$$

By the properties (7.3.4) and (7.3.3) of a decoupled tangent sequence,

$$\mathbb{E}_{\mathscr{G}}|e_{i,1}|^2 = \mathbb{E}_{i-1}|e_{i,1}|^2 = \mathbb{E}_{i-1}|d_{i,1}|^2,$$

and therefore

$$\left(\sum_{i} \mathbb{E}_{\mathscr{G}} |e_{i,1}|^2\right)^{\frac{1}{2}} = \left(\sum_{i} \mathbb{E}_{i-1} |d_{i,1}|^2\right)^{\frac{1}{2}}.$$

Similarly,

$$\mathbb{E}_{\mathscr{G}} \| e_{i,2} \|_{L^{q}(S)}^{q} = \mathbb{E}_{i-1} \| d_{i,2} \|_{L^{q}(S)}^{q}$$

We conclude that

$$\left(\mathbb{E}\left\|\sum_{i}d_{i}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}}$$

$$\lesssim_{p,q} \max\left\{ \left( \mathbb{E} \left\| \left( \sum_{i} \mathbb{E}_{i-1} | d_{i,1} |^2 \right)^{\frac{1}{2}} \right\|_{L^q(S)}^p \right)^{\frac{1}{p}} + \left( \mathbb{E} \left( \sum_{i} \mathbb{E}_{i-1} \| d_{i,2} \|_{L^q(S)}^q \right)^{\frac{p}{q}} \right)^{\frac{1}{p}}, \left( \sum_{i} \mathbb{E} \| d_i \|_{L^q(S)}^p \right)^{\frac{1}{p}} \right\}$$

Taking the infimum over all decompositions as above yields the inequality ' $\leq_{p,q}$ ' in (7.1.6).

*Step 2: lower bounds.* We deduce the lower bounds by duality. Since  $(S_q^p)^* = S_{q'}^{p'}$  (by [87]),  $(D_{p,q}^p)^* = D_{p',q'}^{p'}$ , and  $(D_{q,q}^p)^* = D_{q',q'}^{p'}$  (by Theorem 7.4.1 below) hold isomorphically with constants depending only on *p* and *q*, it follows from (7.2.1) that  $s_{p,q}^* = s_{p',q'}$  with duality bracket

$$\langle (f_i), (g_i) \rangle = \sum_i \mathbb{E} \langle f_i, g_i \rangle \qquad ((f_i) \in s_{p,q}, \ (g_i) \in s_{p',q'}).$$

Let  $\hat{x}^* \in (\hat{s}_{p,q})^*$ . Define the map  $P: s_{p,q} \to \hat{s}_{p,q}$  by

$$P((f_i)) = (\Delta_i f_i),$$

where  $\Delta_i := \mathbb{E}_i - \mathbb{E}_{i-1}$ . By the triangle inequality and Jensen's inequality one readily sees that *P* is a bounded projection. As a consequence, we can define  $x^* \in s_{p,q}^*$  by  $x^* = \hat{x}^* \circ P$ . Let  $(g_i) \in s_{p',q'}$  be such that

$$x^*((f_i)) = \sum_i \mathbb{E}\langle f_i, g_i \rangle \qquad ((f_i) \in s_{p,q}).$$

Then, for any  $(f_i) \in \hat{s}_{p,q}$ ,

$$\hat{x}^*((f_i)) = \sum_i \mathbb{E}\langle f_i, g_i \rangle = \sum_i \mathbb{E}\langle f_i, \Delta_i g_i \rangle = \langle (f_i), P(g_i) \rangle.$$

This shows that  $(\hat{s}_{p,q})^* = \hat{s}_{p',q'}$  isomorphically. Let *U* and *V* be the dense linear subspaces spanned by all finite martingale difference sequences in  $\hat{s}_{p,q}$  and  $\hat{s}_{p',q'}$ , respectively. Define

$$Y = \overline{\operatorname{span}}\left\{\sum_{i} d_{i} : (d_{i}) \in U\right\} \subset L^{p}(\Omega; L^{q}(S)).$$

By Step 1, we can define two maps  $j_0 \in \mathcal{L}(U, Y)$ ,  $k_0 \in \mathcal{L}(V, Y^*)$  by

$$j_0((d_i)) = \sum_i d_i, \qquad k_0((\tilde{d}_i)) = \sum_i \tilde{d}_i.$$

By the martingale difference property,

$$\langle j_0((d_i)), k_0((\tilde{d}_i)) \rangle = \mathbb{E} \left\langle \sum_i d_i, \sum_i \tilde{d}_i \right\rangle = \sum_i \mathbb{E} \langle d_i, \tilde{d}_i \rangle = \langle (d_i), (\tilde{d}_i) \rangle.$$
(7.3.7)

The lower bounds now follow immediately from Lemma 7.2.1.

For the final assertion of the theorem, suppose that  $\mathscr{F} = \sigma(\cup_{i\geq 0}\mathscr{F}_i)$ . Let  $f \in L_0^p(\Omega; L^q(S))$  and define  $f_n = \mathbb{E}_n f$ . Then  $\lim_{n\to\infty} f_n = f$  (see e.g. [79, Theorem 3.3.2]). Conversely, let  $(f_n)_{n\geq 1}$  be a martingale with  $\sup_{n\geq 1} ||f_n||_{L^p(\Omega; L^q(S))} < \infty$ . By reflexivity of  $L^q(S)$  we have  $L^p(\Omega; L^q(S)) = (L^{p'}(\Omega; L^{q'}(S)))^*$  and hence its unit ball is weak\*-compact. Let f be the weak\*-limit of  $(f_n)$ . It is easy to check that  $f_n = \mathbb{E}_n f$ . In conclusion, any martingale difference sequence  $(d_i)_{i\geq 0}$  of a bounded martingale in  $L^p(\Omega; L^q(S))$  corresponds uniquely to an  $f \in L^p(\Omega; L^q(S))$  such that

$$f - \mathbb{E}f = \sum_{i} d_{i}, \qquad d_{i} = \mathbb{E}_{i}f - \mathbb{E}_{i-1}f.$$

The two-sided inequality (7.1.6) now implies that the map  $f \mapsto (\mathbb{E}_i f - \mathbb{E}_{i-1} f)_{i \ge 0}$  is a linear isomorphism between  $L_0^p(\Omega; L^q(S))$  and  $\hat{s}_{p,q}$ , with constants depending only on p and q.

*Remark* 7.3.3. Let  $1 < p, q < \infty$ . Define  $\hat{S}_q^{p,odd}$ ,  $\hat{D}_{q,q}^{p,odd}$  and  $\hat{D}_{p,q}^{p,odd}$  as the closed subspaces of  $\hat{S}_q^p$ ,  $\hat{D}_{q,q}^p$  and  $\hat{D}_{p,q}^p$ , respectively, spanned by all  $L^q$ -valued martingale difference sequences  $(d_i)_{i\geq 0}$  such that  $d_{2i} = 0$  for each  $i \geq 0$ . By the proof of Theorem 7.1.1, any  $L^q$ -valued martingale difference sequence  $(d_i)_{i\geq 0}$  such that  $d_{2i} = 0$  for each  $i \geq 0$  satisfies

$$\left(\mathbb{E}\left\|\sum_{i}d_{i}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \approx_{p,q} \|(d_{i})\|_{\hat{s}_{p,q}^{odd}},$$

where  $\hat{s}_{p,q}^{odd}$  is given by

$$\begin{split} \hat{S}_{q}^{p,odd} \cap \hat{D}_{q,q}^{p,odd} \cap \hat{D}_{p,q}^{p,odd} & \text{if } 2 \leq q \leq p < \infty; \\ \hat{S}_{q}^{p,odd} \cap (\hat{D}_{q,q}^{p,odd} + \hat{D}_{p,q}^{p,odd}) & \text{if } 2 \leq p \leq q < \infty; \\ (\hat{S}_{q}^{p,odd} \cap \hat{D}_{q,q}^{p,odd}) + \hat{D}_{p,q}^{p,odd} & \text{if } 1 < p < 2 \leq q < \infty; \\ (\hat{S}_{q}^{p,odd} + \hat{D}_{q,q}^{p,odd}) \cap \hat{D}_{p,q}^{p,odd} & \text{if } 1 < q < 2 \leq p < \infty; \\ \hat{S}_{q}^{p,odd} + (\hat{D}_{q,q}^{p,odd} \cap \hat{D}_{p,q}^{p,odd}) & \text{if } 1 < q \leq p \leq 2; \\ \hat{S}_{q}^{p,odd} + \hat{D}_{q,q}^{p,odd} + \hat{D}_{p,q}^{p,odd} & \text{if } 1 < p \leq q \leq 2. \end{split}$$

This fact will be used in the proof of Theorem 7.5.5.

*Remark* 7.3.4. Let us compare our result to the literature. As was mentioned in the introduction, the scalar-valued version of Theorem 7.1.1 is due to Burkholder [29], following work of Rosenthal [161]. A version for noncommutative martingales, as well as a version of (7.1.3) for  $1 , was obtained by Junge and Xu [88]. Various upper bounds for the moments of a martingale with values in a uniformly 2-smooth (or equivalently, cf. [152], martingale type 2) Banach space were obtained by Pinelis [150], with constants of optimal order. For instance, if <math>2 \le p < \infty$  then ([150], Theorem 4.1)

$$\left(\mathbb{E}\left\|\sum_{i} d_{i}\right\|_{X}^{p}\right)^{\frac{1}{p}} \lesssim p(\mathbb{E}\sup_{i} \|d_{i}\|_{X}^{p})^{\frac{1}{p}} + \sqrt{p}\tau_{2}(X) \left(\mathbb{E}\left(\sum_{i} \mathbb{E}_{i-1} \|d_{i}\|_{X}^{2}\right)^{\frac{p}{2}}\right)^{\frac{1}{p}},$$
(7.3.8)

where  $\tau_2(X)$  is the 2-smoothness constant of *X*. As already remarked in [150], due to the presence of the second term on the right hand side this type of inequality cannot hold in a Banach space which is not 2-uniformly smooth (or equivalently, has martingale type 2). On the other hand, one can show that the reverse inequality holds (with different constants) if and only if *X* is 2-uniformly convex (or equivalently, has martingale cotype 2). Thus, a two-sided inequality involving the norm on the right hand side of (7.3.8) can only hold in a space with both martingale type and cotype equal to 2. Such a space is necessarily isomorphic to a Hilbert space by a well-known result of Kwapień (see e.g. [2], Theorem 7.4.1).

It should be mentioned that the dependence of the implicit constants on p and q in (7.1.6) is not optimal. We leave it as an interesting open problem to determine the optimal dependence on the constants.

# **7.4.** The dual of $H_p^{s_q}(X)$

In the proof of Theorem 7.1.1 we used the fact that  $(D_{q,q}^p)^* = D_{q',q'}^{p'}$  holds isomorphically (with constants depending only on *p* and *q*) for all  $1 < p, q < \infty$ . In this section we will prove a more general statement.

Let  $(\Omega, \mathscr{F}, \mathbb{P})$  be a probability space with a filtration  $\mathbb{F} = (\mathscr{F}_k)_{k \ge 0}$ , *X* be a Banach space, and let  $1 < p, q < \infty$ . For an adapted sequence  $f = (f_k)_{k \ge 0}$  of *X*-valued random variables we define

$$s_q^n(f) := \left(\sum_{k=0}^n \mathbb{E}_{k-1} \|f_k\|^q\right)^{1/q}, \ s_q(f) := \left(\sum_{k=0}^\infty \mathbb{E}_{k-1} \|f_k\|^q\right)^{1/q},$$

where  $\mathbb{E}_k = \mathbb{E}(\cdot|\mathscr{F}_k)$ ,  $\mathbb{E}_{-1} = \mathbb{E}$ . We let  $H_p^{s_q}(X)$  be the space of all adapted sequences  $f = (f_k)_{k \ge 0}$  satisfying

$$\|f\|_{H_p^{s_q}(X)} := (\mathbb{E}s_q(f)^p)^{1/p} < \infty.$$

Similarly we define  $H_p^{s_q^n}(X)$ . We will prove the following result, which was only known before if  $X = \mathbb{R}$  and either  $1 or <math>2 \le q \le p < \infty$  (see [181, Theorem 15] and the remark following it).

**Theorem 7.4.1.** Let X be a reflexive separable Banach space,  $1 < p, q < \infty$ . Then  $(H_p^{s_q}(X))^* = H_{p'}^{s_{q'}}(X^*)$  isomorphically. The isomorphism is given by

$$g \mapsto F_g, \quad F_g(f) = \mathbb{E}\left(\sum_{k=0}^{\infty} \langle f_k, g_k \rangle\right) \quad \left(f \in H_p^{s_q}(X), g \in H_{p'}^{s_{q'}}(X^*)\right), \tag{7.4.1}$$

and

$$\min\left\{\frac{q}{p}, \frac{q'}{p'}\right\} \|g\|_{H^{s_{q'}}_{p'}(X^*)} \le \|F_g\|_{(H^{s_q}_p(X))^*} \le \|g\|_{H^{s_{q'}}_{p'}(X^*)}.$$
(7.4.2)

In particular,  $H_p^{s_q}(X)$  is a reflexive Banach space.

To prove this result, we will first extend an argument of Csörgő [46] to show that  $(H_p^{s_q^n}(X))^*$  and  $H_{p'}^{s_{q'}^n}(X^*)$  are isomorphic if  $1 < p, q < \infty$ , with isomorphism constants depending on p, q and n. In particular, this shows that  $H_p^{s_q^n}(X)$  is reflexive. In a second step, we exploit this reflexivity to show that the isomorphism constants do not depend on n. The proof of this result, Theorem 7.4.5, relies on an argument of Weisz [181]. After this step, it is straightforward to deduce Theorem 7.4.1.

We start by introducing an operator that serves as a replacement for the signfunction in a vector-valued context.

**Lemma 7.4.2.** Let X be a Banach space with a separable dual. Fix  $\varepsilon > 0$ . Then there exists a discrete-valued Borel-measurable function  $P_{\varepsilon} : X^* \to X$  such that  $||P_{\varepsilon}(x^*)|| = 1$  and

$$(1-\varepsilon)\|x^*\| \le \langle P_{\varepsilon}x^*, x^* \rangle \le \|x^*\| \tag{7.4.3}$$

for each  $x^* \in X^*$ .

*Proof.* Let  $(x_n^*)_{n\geq 1}$  be a dense subset of the unit sphere of  $X^*$ . For each  $n\geq 1$  define  $U_n = U \cap B(x_n^*, \frac{\varepsilon}{2})$ , where  $B(y^*, r)$  denotes the ball in  $X^*$  with radius r and center  $y^*$ . Define  $V_1 = U_1$  and

$$V_n = U_n \setminus \left(\bigcup_{k=1}^{n-1} V_k\right), \qquad n \ge 2$$

For each  $n \ge 1$  one can find an  $x_n \in X$  such that  $||x_n|| = 1$  and  $\langle x_n, x_n^* \rangle \ge 1 - \frac{\varepsilon}{2}$ . Now define

$$P_{\varepsilon}(x^*) := \sum_{n=1}^{\infty} \mathbf{1}_{V_n} \Big( \frac{x^*}{\|x^*\|} \Big) x_n, \ x^* \in X^*.$$

This function is Borel since the  $V_n$  are Borel sets. As the  $V_n$  form a disjoint cover of the unit sphere, for every  $x^* \in X^*$  there exists a unique  $n = n(x^*)$  so that  $x^*/||x^*|| \in V_n$ . Hence,  $||P_{\varepsilon}(x^*)|| = 1$  and

$$\langle P_{\varepsilon}(x^*), x^* \rangle = \|x^*\| \left\langle x_n, \frac{x^*}{\|x^*\|} \right\rangle \ge \|x^*\| \langle x_n, x_n^* \rangle - \frac{\varepsilon}{2} \|x^*\| \ge (1-\varepsilon) \|x^*\|,$$

so (7.4.3) follows.

**Theorem 7.4.3.** Let X be a reflexive separable Banach space,  $1 < p, q < \infty$ ,  $n \ge 0$ . Then  $(H_p^{s_q^n}(X))^* = H_{p'}^{s_{q'}^n}(X^*)$  isomorphically (with constants depending on p, q and n). The isomorphism is given by

$$g \mapsto F_g, \quad F_g(f) = \mathbb{E}\left(\sum_{k=0}^n \langle f_k, g_k \rangle\right) \quad \left(f \in H_p^{s_q^n}(X), g \in H_{p'}^{s_{q'}^n}(X^*)\right). \tag{7.4.4}$$

In particular,  $H_p^{s_q^n}(X)$  is a reflexive Banach space.

*Proof.* The main argument is inspired by the proof of [46, Theorem 1]. By the conditional Hölder inequality and the usual version of Hölder's inequality,

$$|F_{g}(f)| \leq \mathbb{E} \Big( \sum_{k=0}^{n} \mathbb{E}_{k-1} (\|f_{k}\| \|g_{k}\|) \Big)$$
  
$$\leq \mathbb{E} \Big( \sum_{k=0}^{n} (\mathbb{E}_{k-1} \|f_{k}\|^{q})^{1/q} (\mathbb{E}_{k-1} \|g_{k}\|^{q'})^{1/q'} \Big)$$
  
$$\leq \|f\|_{H^{s^{n}}_{p}(X)} \|g\|_{H^{s^{n}}_{q'}(X^{*})}.$$
  
(7.4.5)

Hence, the functional  $F_g$  is bounded and  $||F_g|| \le ||g||_{S^q_{p'}(X^*)}$ .

To prove that  $||F_g|| \gtrsim_{p,q,n} ||g||_{H^{q'}_{p'}(X^*)}$  we need to construct an appropriate  $f \in \mathcal{F}_{p'}$ 

 $H_p^{s_q^n}(X)$  with

$$\|f\|_{H^{s^n}_p(X)} \lesssim_{p,q,n} 1, \qquad \langle F_g, f \rangle \gtrsim_{p,q,n} \|g\|_{H^{s^n}_{p'}(X^*)}$$

Fix  $0 < \varepsilon < 1$ . We define *f* by setting

$$f_{k} := P_{\varepsilon}g_{k} \frac{\|g_{k}\|^{q'-1}}{\|g\|_{B_{r'}^{p'-1}(X^{*})}^{p'-1}} (\mathbb{E}_{k-1}\|g_{k}\|^{q'})^{\frac{p'-q'}{q'}}, \qquad 0 \le k \le n$$

where  $P_{\varepsilon}$  is as in Lemma 7.4.2. Using pp' = p + p' and qq' = q + q' we find

$$\begin{split} \|f\|_{H_{p'}^{s^{n}_{q}}(X)}^{p} &= \mathbb{E}\Big(\sum_{k=0}^{n} \mathbb{E}_{k-1} \|f_{k}\|^{q}\Big)^{p'q} = \frac{1}{\|g\|_{H_{p'}^{s^{n}_{q'}}(X^{*})}^{p(p'-1)}} \mathbb{E}\Big(\sum_{k=0}^{n} (\mathbb{E}_{k-1} \|g_{k}\|^{q'})^{\frac{(p'-1)q}{q'}}\Big)^{\frac{p}{q}} \\ &= n, p, q \; \frac{1}{\|g\|_{H_{p'}^{s^{n}_{q'}}(X^{*})}^{p'}} \mathbb{E}\Big(\sum_{k=0}^{n} (\mathbb{E}_{k-1} \|g_{k}\|^{q'})\Big)^{\frac{p'}{q'}} = 1, \end{split}$$

so  $f \in H_p^{s_q^n}(X)$ . Moreover,

$$\begin{split} \langle F_{g}, f \rangle &\geq (1 - \varepsilon) \frac{1}{\|g\|_{p'^{-1}}^{p'-1}} \mathbb{E} \sum_{k=0}^{n} \|g_{k}\|^{q'} (\mathbb{E}_{k-1} \|g_{k}\|^{q'})^{\frac{p'-q'}{q'}} \\ &= (1 - \varepsilon) \frac{1}{\|g\|_{p'^{-1}}^{p'-1}} \mathbb{E} \sum_{k=0}^{n} (\mathbb{E}_{k-1} \|g_{k}\|^{q'})^{\frac{p'}{q'}} \\ &= (1 - \varepsilon) \frac{1}{\|g\|_{p'^{-1}}^{p'-1}} \mathbb{E} \left( \sum_{k=0}^{n} \mathbb{E}_{k-1} \|g_{k}\|^{q'} \right)^{\frac{p'}{q'}} \\ &= \log_{p,q,n} (1 - \varepsilon) \frac{1}{\|g\|_{p'^{-1}}^{p'-1}} \mathbb{E} \left( \sum_{k=0}^{n} \mathbb{E}_{k-1} \|g_{k}\|^{q'} \right)^{\frac{p'}{q'}} = \|g\|_{H_{p'}^{s,n}}^{s,n}, \end{split}$$

as desired, since  $\varepsilon$  was arbitrary and can be chosen, say,  $\frac{1}{2}$ .

Now we will show that every  $F \in (H_p^{\mathbb{S}_q^n}(X))^*$  is equal to  $F_g$  for a suitable  $g \in H_{p'}^{\mathbb{S}_q^n}(X^*)$ . For this purpose we consider the disjoint direct sum of  $(\Omega, \mathscr{F}_k, \mathbb{P})$ , k = 0, ..., n. Formally, we set  $\Omega_k = \Omega \times \{k\}$ ,  $\widetilde{\mathscr{F}}_k = \mathscr{F}_k \times \{k\}$  and define a probability measure  $\mathbb{P}_k$  on  $\widetilde{\mathscr{F}}_k$  by  $\mathbb{P}_k(A \times \{k\}) = \mathbb{P}(A)$ . Now the disjoint direct sum  $(\Omega^n, \mathscr{F}^n, \mathbb{P}^n)$  is defined by

$$\Omega^n = \bigcup_{k=0}^n \Omega_k, \qquad \mathscr{F}^n = \{A \in \Omega^n : A \cap \Omega_k \in \widetilde{\mathscr{F}}_k, \text{ for all } 1 \le k \le n\}$$

and

$$\mathbb{P}^{n}(A) = \sum_{k=0}^{N} \mathbb{P}_{k}(A \cap \Omega_{k}), \qquad A \in \mathscr{F}^{n}.$$

Let  $P_k : (\Omega, \mathscr{F}_k) \to (\Omega^n, \mathscr{F}^n)$ ,  $P_k(\omega) = (\omega, k)$ , be the measurable bijection between  $(\Omega, \mathscr{F}_k)$  and its disjoint copy. We can now define an  $X^*$ -valued set function  $\mu$  by

$$\langle \mu(A), x \rangle := F \big( (x \cdot \mathbf{1}_{P_k^{-1}(A \cap \Omega_k)})_{k=0}^n \big), \qquad A \in \mathcal{F}^n, \; x \in X.$$

We will show that  $\mu$  is  $\sigma$ -additive, absolutely continuous with respect to  $\mathbb{P}^n$  and of finite variation. Let us first show that  $\mu$  is of finite variation. Let  $(A_m)_{m=1}^M \subset \mathscr{F}^n$  be disjoint such that  $\cup_m A_m = \Omega^n$ . Then

$$\begin{split} \sum_{m=1}^{M} \|\mu(A_m)\| &= \sum_{m=1}^{M} \sup_{x_m \in X: \|x_m\| = 1} F\left((x_m \cdot \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)})_{k=0}^n\right) \\ &= \sup_{(x_m)_{m=1}^M \subset X: \|x_m\| = 1} \sum_{m=1}^M F\left((x_m \cdot \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)})_{k=0}^n\right) \\ &= \sup_{(x_m)_{m=1}^M \subset X: \|x_m\| = 1} F\left(\left(\sum_{m=1}^M x_m \cdot \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)}\right)_{k=0}^n\right) \\ &\leq \|F\| \sup_{(x_m)_{m=1}^M \subset X: \|x_m\| = 1} \left\|\left(\sum_{m=1}^M x_m \cdot \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)}\right)_{k=0}^n\right\|_{H_p^{s_n^n}(X)} \\ &= \|F\| \sup_{(x_m)_{m=1}^M \subset X: \|x_m\| = 1} \left(\mathbb{E}\left(\sum_{k=0}^n \mathbb{E}_{k-1}\right) \left\|\sum_{m=1}^M x_m \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)}\right\|^q\right)^{\frac{p}{q}}\right)^{\frac{1}{p}} \\ &\leq \|F\| \left[\mathbb{E}\left(\sum_{k=0}^n \mathbb{E}_{k-1}\left(\sum_{m=1}^M \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)}\right)^q\right)^{\frac{p}{q}}\right]^{\frac{1}{p}} \\ &= \|F\| \left(\mathbb{E}\left(\sum_{k=0}^n \mathbb{E}_{k-1}\mathbf{1}_\Omega\right)^{\frac{p}{q}}\right)^{\frac{1}{p}} = \|F\|(n+1)^{\frac{1}{q}} \end{split}$$

Now let us prove the  $\sigma$ -additivity. Obviously  $\mu$  is additive. Let  $(A_m)_{m\geq 0} \subset \mathscr{F}_n$  be such that  $A_m \searrow \emptyset$ . Then

$$\|\mu(A_m)\| = \sup_{x \in X: \|x\| = 1} |F((x \cdot \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)})_{k=0}^n)|$$

$$\leq \|F\| \sup_{x \in X: \|x\| = 1} \| (x \cdot \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)})_{k=0}^n \|_{H_p^{s_q^n}(X)}$$
  
=  $\|F\| \left( \mathbb{E} \left( \sum_{k=0}^n \mathbb{E}_{k-1} \mathbf{1}_{P_k^{-1}(A_m \cap \Omega_k)} \right)^{\frac{p}{q}} \right)^{\frac{1}{p}} \to 0 \text{ as } m \to \infty,$ 

by the monotone convergence theorem. This computation also shows that  $\mu$  is absolutely continuous with respect to  $\mathbb{P}^n$ .

Since *X* is reflexive,  $X^*$  has the Radon-Nikodym property (see e.g. [79, Theorem 1.3.21]). Thus, there exists a  $g \in L^1(\Omega^n; X^*)$  so that

$$\mu(A) = \int_A g \, \mathrm{d}\mathbb{P}^n = \sum_{k=0}^n \int_{A \cap \Omega_k} g \, \mathrm{d}\mathbb{P}_k.$$

If we now define  $g_k := g \circ P_k$  then  $g_k$  is  $\mathscr{F}_k$ -measurable and

$$\mu(A) = \sum_{k=0}^{n} \int_{P_k^{-1}(A \cap \Omega_k)} g_k \, \mathrm{d}\mathbb{P}.$$

It now follows for  $f = (f_k)_{k=0}^n \in H_p^{s_q^n}(X)$  with  $f_k$  bounded for all k = 0, ..., n that

$$F(f) = F_g(f) = \mathbb{E} \sum_{k=0}^n \langle f_k, g_k \rangle.$$
(7.4.6)

Now fix a general  $f \in H_p^{s_q^n}(X)$ . Fix  $0 < \varepsilon < 1$  and let  $h := (h_k)_{k=0}^n = (||f_k|| P_{\varepsilon} g_k)_{k=0}^n$ . Define  $h^m := (h_k^m)_{k=0}^n = (h_k \mathbf{1}_{||h_k|| \le m})_{k=0}^n$  for each  $m \ge 1$ . Then formula (7.4.6) holds for  $h^m$ . But  $F(h^m) \to F(h)$  as m goes to infinity, so by the monotone convergence theorem  $F(h) = \mathbb{E}\sum_{k=0}^n \langle h_k, g_k \rangle$ . This shows that

$$\mathbb{E}\sum_{k=0}^{n} |\langle f_k, g_k \rangle| \le \mathbb{E}\sum_{k=0}^{n} ||f_k|| ||g_k|| \le (1-\varepsilon)^{-1} \mathbb{E}\sum_{k=0}^{n} \langle h_k, g_k \rangle < \infty.$$
(7.4.7)

Now consider  $f^m := (f_k^m)_{k=0}^n = (f_k \mathbf{1}_{||f_k|| \le m})_{k=0}^n$ . Since (7.4.6) holds for  $f^m$  and  $F(f^m) \to F(f)$ , we can use (7.4.7) and the dominated convergence theorem to conclude that f satisfies (7.4.6).

It remains to prove that  $g \in H_{p'}^{s_{q'}^n}(X^*)$ . For each  $m \ge 1$  we consider the approximation  $g^m := (g_k \mathbf{1}_{\|g_k\| \le m})_{k=0}^n$ . Then  $\|g^m\|_{S_{p'}^n(X^*)} \lesssim_{p,q,n} \|F_{g^m}\| \le \|F\|$ . Therefore by the monotone convergence theorem  $\|g\|_{H_{p'}^{s_{q'}}(X^*)} \lesssim_{p,q,n} \|F\|$ .

One can easily show the following simple lemma.

**Lemma 7.4.4.** Let X and Y be reflexive Banach spaces such that  $X^*$  is isomorphic to Y and

$$a \|x^*\|_Y \le \|x^*\|_{X^*} \le b \|x^*\|_Y, \ x^* \in X^*.$$

Then  $Y^*$  is isomorphic to  $X^{**} = X$  and

$$a \|x\|_X \le \|x\|_{Y^*} \le b \|x\|_X, \ x \in X.$$

**Theorem 7.4.5.** Let *X* be a reflexive separable Banach space,  $1 < p, q < \infty, n \ge 0$ . Then

$$\min\left\{\frac{q}{p}, \frac{q'}{p'}\right\} \|g\|_{B^{s^{n}}_{p'}(X^{*})} \leq \|F_{g}\|_{\left(H^{s^{n}}_{p}(X)\right)^{*}} \leq \|g\|_{B^{s^{n}}_{p'}(X^{*})}.$$
(7.4.8)

*Proof.* We already proved in Theorem 7.4.3 that  $H_p^{s_q^n}(X)$  is reflexive, so by Lemma 7.4.4 it is enough to show (7.4.8) for  $p \le q$ . It was already noted in (7.4.5) that  $||F_g|| \le ||g||_{B^{q'}_{p'}(X^*)}$ . It is sufficient to show (7.4.8) for a bounded g. The following

construction is in essence the same as in [181, Theorem 15]. Set

$$(\nu_k)_{k=0}^n = \left(\frac{(s_{q'}^k(g))^{p'-q'}}{\|g\|_{s_{q'}^{p'-1}}^{p'-1}}\right)_{k=0}^n.$$

Fix  $0 < \varepsilon < 1$ . Let us define  $h \in H_p^{s_q^n}(X)$  by setting

$$h_k = v_k \|g_k\|^{q'-1} P_{\varepsilon} g_k$$

where  $P_{\varepsilon}: X^* \to X$  is as given in Lemma 7.4.2. Then

$$(s_q^n(h))^q \le \sum_{k=0}^n \frac{(s_{q'}^{k}(g))^{qp'-qq'}}{\|g\|_{s_q^{n'}(X^*)}^{qp'-q}} \mathbb{E}_{k-1} \|g_k\|^{q'} \le \frac{(s_{q'}^n(g))^{qp'-(q-1)q'}}{\|g\|_{s_q^{n'}(X^*)}^{qp'-q}}.$$

and therefore

$$\mathbb{E}(s_q^n(h))^p \leq \frac{\mathbb{E}(s_{q'}^n(g))^{(qp'-(q-1)q')\frac{p}{q}}}{\|g\|_{s_{q'}^{pp'-p}}^{pp'-p}} = 1.$$

As a consequence,

$$\begin{aligned} \|F_{g}\| &\geq |\langle F_{g}, h\rangle| \\ &\geq (1-\varepsilon) \frac{1}{\|g\|_{p'^{-1}}^{p'-1}} \mathbb{E} \sum_{k=0}^{n} (s_{q'}^{k}(g))^{p'-q'} \mathbb{E}_{k-1} \|g_{k}\|^{q'} \\ &H_{p'}^{q'}(X^{*}) \end{aligned}$$

$$= (1-\varepsilon) \frac{1}{\|g\|_{p'^{-1}}^{p'-1}} \mathbb{E} \sum_{k=0}^{n} (s_{q'}^{k}(g))^{p'-q'} ((s_{q'}^{k}(g))^{q'} - (s_{q'}^{k-1}(g))^{q'}).$$

$$(7.4.9)$$

By the mean value theorem,

$$x^{\alpha} - 1 \le \alpha (x - 1) x^{\alpha - 1}, \quad x, \alpha \ge 1.$$
 (7.4.10)

. .

Applying this for  $x = \frac{(s_{q'}^k(g))^{q'}}{(s_{q'}^{k-1}(g))^{q'}} \ge 1$  and  $\alpha = \frac{p'}{q'} \ge 1$  we find

$$\frac{q}{p'}((s_{q'}^k(g))^{p'} - (s_{q'}^{k-1}(g))^{p'}) \le ((s_{q'}^k(g))^{q'} - (s_{q'}^{k-1}(g))^{q'})(s_{q'}^k(g))^{p'-q'}.$$

Combining this with (7.4.9) and letting  $\varepsilon \rightarrow 0$ ,

$$\|F_{g}\| \geq \frac{q'}{p'\|g\|_{s^{n}_{p'}(X^{*})}^{p'-1}} \mathbb{E}(s^{n}_{q'}(g))^{p'} = \frac{q'}{p'}\|g\|_{H^{s^{n}}_{p'}(X^{*})}.$$

We can now deduce the main result of this section.

*Proof of Theorem* 7.4.1. Let  $F \in (H_p^{s_q}(X))^*$ . For every  $n \ge 0$  there exists an  $F_n \in (H_p^{s_q^n}(X))^*$  such that  $\langle F, f \rangle = \langle F_n, (f_k)_{k=0}^n \rangle$  for each  $f \in H_p^{s_q}(X)$  satisfying  $f_m = 0$  for all m > n. Thanks to Theorem 7.4.3, for each  $n \ge 0$  there exists a  $g^n = (g_k^n)_{k=0}^n \in H_{p'}^{s_{q'}^n}(X)$  such that  $F_n = F_{g^n}$ . Obviously  $g_k^m = g_k^n$  for each  $m, n \ge k$ , so there exists a unique  $g = (g_k)_{k=0}^\infty$  such that  $g^n = (g_k)_{k=0}^n$ . Moreover, Theorem 7.4.5 implies

$$\min\left\{\frac{q}{p}, \frac{q'}{p'}\right\} \|g^n\|_{H^{s^n}_{p'}(X)} \le \|F_n\|_{\left(H^{s^n}_p(X)\right)^*} \le \|F\|_{\left(H^{s^n}_p(X)\right)^*},$$

so  $g \in H_{p'}^{s_{q'}}(X)$  and

$$\min\left\{\frac{q}{p},\frac{q'}{p'}\right\}\|g\|_{H^{s_{q'}}_{p'}(X)} \le \|F\|_{(H^{s_q}_p(X))^*}.$$

Now obviously  $F = F_g$ , as these two functionals coincide on the dense subspace of all finitely non-zero sequences in  $H_p^{s_q}(X)$ , and (7.4.1) and (7.4.2) hold.

# 7.5. Sharp bounds for $L^q$ -valued stochastic integrals

We now turn to proving sharp bounds for stochastic integrals.

## 7.5.1. Decomposition of stochastic integrals

To prove sharp bounds for the stochastic integral, we will decompose it by decomposing the integrator *M* into three parts. By Proposition 2.5.1, if  $M = M^c + M^q + M^a$  is the canonical decomposition of *M*, then the canonical decomposition of  $\Phi \cdot M$  is given by

$$\Phi \cdot M = \Phi \cdot M^c + \Phi \cdot M^q + \Phi \cdot M^a. \tag{7.5.1}$$

The following four subsections are dedicated to sharp estimates of the respective parts on the right hand side. In Subsection 7.5.6 we combine our work to estimate  $\Phi \cdot M$ .

# 7.5.2. Purely discontinuous martingales with accessible jumps

In this section we prove Burkholder-Rosenthal type inequalities for purely discontinuous martingales with accessible jumps. As an immediate consequence, we find sharp bounds for the accessible jump part in (7.5.1).

As a first step, we will show that we can represent a purely discontinuous martingales with accessible jumps as a sum of jumps occurring at predictable times.

The following lemma follows from Theorem 9.7.12.

**Lemma 7.5.1.** Let  $1 < p, q < \infty$ ,  $M : \mathbb{R}_+ \times \Omega \to L^q(S)$  be a purely discontinuous  $L^p$ -martingale with accessible jumps. Let  $\mathcal{T} = (\tau_n)_{n=0}^{\infty}$  be any sequence of predictable stopping times with disjoint graphs that exhausts the jumps of M. Then for each  $n \ge 0$ 

$$M_t^n = \sum_{k=0}^n \Delta M_{\tau_k} \mathbf{1}_{[0,t]}(\tau_k)$$
(7.5.2)

defines an  $L^p$ -martingale. Moreover, for any  $t \ge 0$ ,  $\|M_t - M_t^n\|_{L^p(\Omega; L^q(S))} \to 0$  as  $n \to \infty$ . If  $\sup_{t\ge 0} \mathbb{E} \|M_t\|^p < \infty$ , then  $\|M_{\infty} - M_{\infty}^n\|_{L^p(\Omega; L^q(S))} \to 0$  for  $n \to \infty$ .

**Definition 7.5.2.** For  $1 < p, q < \infty$  we define  $\mathcal{M}_{p,q}^{\text{acc}}$  as the linear space of all  $L^q(S)$ -valued purely discontinuous  $L^p$ -martingales with accessible jumps, endowed with the norm  $||M||_{\mathcal{M}_{p,q}^{\text{acc}}} := ||M_{\infty}||_{L^p(\Omega; L^q(S))}$ .

The following proposition follows from Proposition 2.4.18.

**Proposition 7.5.3.** For any  $1 < p, q < \infty$  the space  $\mathcal{M}_{p,q}^{\text{acc}}$  is a Banach space.

We now turn to the Burkholder-Rosenthal inequalities. Let  $1 < p, q < \infty$  and let  $M : \mathbb{R}_+ \times \Omega \to L^q(S)$  be a purely discontinuous martingale with accessible jumps. Let  $\mathcal{T} = (\tau_n)_{n \ge 0}$  be a sequence of predictable stopping times with disjoint graphs that exhausts the jumps of M. We define three expressions

$$\|M\|_{\widetilde{S}_{q}^{p}} = \left(\mathbb{E}\left\|\left(\sum_{n\geq0}\mathbb{E}_{\mathscr{F}_{n}}|(\Delta M(\omega)(s))_{\tau_{n}}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}}, \\\|M\|_{\widetilde{D}_{q,q}^{p}} = \left(\mathbb{E}\left(\sum_{n\geq0}\mathbb{E}_{\mathscr{F}_{n}}||\Delta M(\omega)_{\tau_{n}}||_{L^{q}(S)}^{q}\right)^{\frac{p}{q}}\right)^{\frac{1}{p}},$$
(7.5.3)  
$$\|M\|_{\widetilde{D}_{p,q}^{p}} = \left(\mathbb{E}\sum_{t\geq0}\|\Delta M_{t}\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}}.$$

**Proposition 7.5.4.** *The expressions in* (7.5.3) *do not depend on the choice of the family*  $\mathcal{T}$ *.* 

*Proof.* Assume that  $\mathcal{T}' = (\tau'_m)_{m \ge 0}$  is another family of predictable stopping times with disjoint graphs that exhausts the jumps of *M*. Notice that due to [85, Proposition I.2.11],

$$\mathscr{F}_{\tau-} \cap \{\tau = \sigma\} = \mathscr{F}_{\sigma-} \cap \{\tau = \sigma\}$$

for any pair of predictable stopping times  $\tau$  and  $\sigma$ . Therefore for a.e.  $s \in S$  a.s.

$$\begin{split} \sum_{n\geq 0} \mathbb{E}_{\mathscr{F}_{\tau_{n-}}} |(\Delta M(\omega)(s))_{\tau_{n}}|^{2} \stackrel{(*)}{=} \sum_{n\geq 0} \mathbb{E}_{\mathscr{F}_{\tau_{n-}}} \Big( \sum_{m\geq 0} |(\Delta M(\omega)(s))_{\tau_{n}}|^{2} \mathbf{1}_{\tau_{n}=\tau'_{m}} \Big) \\ &= \sum_{n\geq 0} \sum_{m\geq 0} \mathbb{E}_{\mathscr{F}_{\tau_{n-}}} \Big( |(\Delta M(\omega)(s))_{\tau_{n}}|^{2} \mathbf{1}_{\tau_{n}=\tau'_{m}} \Big) \\ &= \sum_{n\geq 0} \sum_{m\geq 0} \mathbb{E}_{\mathscr{F}_{\tau'_{m-}}} \Big( |(\Delta M(\omega)(s))_{\tau_{n}}|^{2} \mathbf{1}_{\tau_{n}=\tau'_{m}} \Big) \\ &= \sum_{m\geq 0} \mathbb{E}_{\mathscr{F}_{\tau'_{m-}}} \Big( \sum_{n\geq 0} |(\Delta M(\omega)(s))_{\tau'_{m}}|^{2} \mathbf{1}_{\tau_{n}=\tau'_{m}} \Big) \\ &\stackrel{(*)}{=} \sum_{n\geq 0} \mathbb{E}_{\mathscr{F}_{\tau'_{m-}}} |(\Delta M(\omega)(s))_{\tau'_{m}}|^{2}, \end{split}$$

where (\*) holds since

$$\mathbb{P}\{\exists u \ge 0 : \Delta M_u \neq 0, u \notin \{\tau_0, \tau_1, \ldots\}\} \\ = \mathbb{P}\{\exists u \ge 0 : \Delta M_u \neq 0, u \notin \{\tau'_0, \tau'_1, \ldots\}\} = 0.$$

Therefore we can conclude that  $\|M\|_{\widetilde{S}^p_q}$  does not depend on the choice of the exhausting family. The same holds for  $\|M\|_{\widetilde{D}^p_{a,q}}$  by an analogous argument.

We let  $\tilde{S}_{q}^{p}$ ,  $\tilde{D}_{q,q}^{p}$  and  $\tilde{D}_{p,q}^{p}$  denote the sets of all purely discontinuous martingales with accessible jumps for which the respective expressions in (7.5.3) are finite. We will prove shortly that the expressions in (7.5.3) are norms. For a fixed family  $\mathcal{T} = (\tau_{n})_{n\geq 0}$  of predictable stopping times with disjoint graphs we let  $\tilde{S}_{q}^{p,\mathcal{T}}$ ,  $\tilde{D}_{q,q}^{p,\mathcal{T}}$ and  $\tilde{D}_{p,q}^{p,\mathcal{T}}$  be the subsets of  $\tilde{S}_{q}^{p}$ ,  $\tilde{D}_{q,q}^{p}$  and  $\tilde{D}_{p,q}^{p}$  consisting of martingales M with  $\{t \in \mathbb{R}_{+} : \Delta M_{t} \neq 0\} \subset \{\tau_{0}, \tau_{1}, \ldots\}$  a.s.

We start by proving a version of the main theorem of this subsection (Theorem 7.5.8 below) for a martingales with finitely many jumps.

**Theorem 7.5.5.** Let  $1 < p, q < \infty$ ,  $N \ge 1$ ,  $\mathcal{T} = (\tau_n)_{n=0}^N$  be a finite family of predictable stopping times with disjoint graphs. Then  $\widetilde{S}_q^{p,\mathcal{T}}$ ,  $\widetilde{D}_{q,q}^{p,\mathcal{T}}$  and  $\widetilde{D}_{p,q}^{p,\mathcal{T}}$  are Banach spaces under the norms in (7.5.3). As a consequence,  $A_{p,q}^{\mathcal{T}}$  given by

$$\begin{split} \widetilde{S}_{q}^{p,\mathcal{F}} \cap \widetilde{D}_{q,q}^{p,\mathcal{F}} \cap \widetilde{D}_{p,q}^{p,\mathcal{F}} & \text{if } 2 \leq q \leq p < \infty, \\ \widetilde{S}_{q}^{p,\mathcal{F}} \cap (\widetilde{D}_{q,q}^{p,\mathcal{F}} + \widetilde{D}_{p,q}^{p,\mathcal{F}}) & \text{if } 2 \leq p \leq q < \infty, \\ (\widetilde{S}_{q}^{p,\mathcal{F}} \cap \widetilde{D}_{q,q}^{p,\mathcal{F}}) + \widetilde{D}_{p,q}^{p,\mathcal{F}} & \text{if } 1 < p < 2 \leq q < \infty, \\ (\widetilde{S}_{q}^{p,\mathcal{F}} + \widetilde{D}_{q,q}^{p,\mathcal{F}}) \cap \widetilde{D}_{p,q}^{p,\mathcal{F}} & \text{if } 1 < q < 2 \leq p < \infty, \\ \widetilde{S}_{q}^{p,\mathcal{F}} + (\widetilde{D}_{q,q}^{p,\mathcal{F}} \cap \widetilde{D}_{p,q}^{p,\mathcal{F}}) & \text{if } 1 < q \leq p \leq 2, \\ \widetilde{S}_{q}^{p,\mathcal{F}} + \widetilde{D}_{q,q}^{p,\mathcal{F}} + \widetilde{D}_{p,q}^{p,\mathcal{F}} & \text{if } 1 < p \leq q \leq 2. \end{split}$$

$$(7.5.4)$$

is a well-defined Banach space. Moreover,  $(\mathscr{A}_{p,q}^{\mathscr{T}})^* = \mathscr{A}_{p',q'}^{\mathscr{T}}$  with isomorphism given by

$$g \mapsto F_{g}, \quad F_{g}(f) = \mathbb{E} \sum_{t \in \mathcal{T}} \langle \Delta g_{t}, \Delta f_{t} \rangle \stackrel{(*)}{=} \mathbb{E} \langle g_{\infty}, f_{\infty} \rangle,$$
$$\|F_{g}\|_{(\mathscr{A}_{p,q}^{\mathcal{T}})^{*}} \approx_{p,q} \|g\|_{\mathscr{A}_{p',q'}^{\mathcal{T}}}.$$

$$(7.5.5)$$

Finally, for any purely discontinuous  $L^p$ -martingale  $M : \mathbb{R}_+ \times \Omega \to L^q(S)$  with accessible jumps such that  $\{t \in \mathbb{R}_+ : \Delta M_t \neq 0\} \subset \{\tau_0, \dots, \tau_N\}$  a.s., and for all  $t \ge 0$ ,

$$\left(\mathbb{E}\sup_{0\leq s\leq t}\|M_t\|_{L^q(S)}^p\right)^{\frac{1}{p}} \approx_{p,q} \|M\mathbf{1}_{[0,t]}\|_{\mathscr{A}_{p,q}^{\mathscr{F}}}.$$
(7.5.6)

*Proof.* The idea of the proof is to identify  $\tilde{S}_{q}^{p,\mathcal{T}}$ ,  $\tilde{D}_{q,q}^{p,\mathcal{T}}$  and  $\tilde{D}_{p,q}^{p,\mathcal{T}}$  with discrete martingale spaces  $S_{q}^{p}$ ,  $D_{q,q}^{p}$  and  $D_{p,q}^{p}$  for an appropriate filtration. Since the  $\tau_{i}$  have disjoint graphs, we can find predictable stopping times  $\tau'_{0}, \ldots, \tau'_{N}$  such that

$$\{\tau_0(\omega),\ldots,\tau_N(\omega)\}=\{\tau'_0(\omega),\ldots,\tau'_N(\omega)\}$$

and  $\tau'_0(\omega) < \ldots < \tau'_N(\omega)$  for a.e.  $\omega \in \Omega$ . Indeed, we can set  $\tau'_0 = \min\{\tau_0, \ldots, \tau_N\}$  and

$$\tau'_{i} = \min(\{\tau_{0}, \dots, \tau_{N}\} \setminus \{\tau'_{0}, \dots, \tau'_{i}\}), \qquad 0 \le i \le N - 1.$$

Fix the sequence of  $\sigma$ -algebras  $\mathbb{G} = (\mathscr{G}_k)_{k=0}^{2N+1} = (\mathscr{F}_{\tau'_0-}, \mathscr{F}_{\tau'_0}, \dots, \mathscr{F}_{\tau'_N-}, \mathscr{F}_{\tau'_N})$ . Using [89, Lemma 25.2] and the fact that  $(\tau'_n)_{n=0}^N$  is a.s. a strictly increasing sequence one can show that  $\mathbb{G}$  is a filtration.

Consider Banach spaces  $\hat{S}_q^{p,odd}$ ,  $\hat{D}_{q,q}^{p,odd}$  and  $\hat{D}_{p,q}^{p,odd}$  with respect to the filtration  $\mathbb{G}$  that were defined in Remark 7.3.3. For any purely discontinuous  $L^q$ -valued martingale M with accessible jumps in  $\mathcal{T}$  we can construct a  $\mathbb{G}$ -martingale difference sequence  $(d_k)_{n=0}^{2N+1}$  by setting  $d_{2n} = 0$ ,  $d_{2n+1} = \Delta M_{\tau_n}$  for n = 0, ..., N. Indeed, by [89, Lemma 26.18] (see also [85, Lemma 2.27]) for each n = 0, ..., N

$$\mathbb{E}(d_{2n+1}|\mathscr{G}_{2n}) = \mathbb{E}(\Delta M_{\tau'_n}|\mathscr{F}_{\tau'_n-}) = 0.$$

By Lemma 2.4.6,

$$\|M\|_{\tilde{S}^{p,\mathcal{F}}_q} = \|(d_n)\|_{S^{p,odd}_q}, \ \|M\|_{\tilde{D}^{p,\mathcal{F}}_{q,q}} = \|(d_n)\|_{D^{p,odd}_{q,q}}, \ \|M\|_{\tilde{D}^{p,\mathcal{F}}_{p,q}} = \|(d_n)\|_{D^{p,odd}_{p,q}}.$$

Moreover, by Corollary 2.4.7 any element  $(d_k)_{k=0}^{2N+1}$  of  $\hat{S}_q^{p,odd}$ ,  $D_{q,q}^{p,odd}$ , or  $D_{p,q}^{p,odd}$  (so in particular,  $d_{2n} = 0$  for each n = 0, ..., N) can be converted back to an element M of  $\tilde{S}_q^{p,\mathcal{T}}$ ,  $\tilde{D}_{q,q}^{p,\mathcal{T}}$ , or  $\tilde{D}_{p,q}^{p,\mathcal{T}}$ , respectively, by defining

$$M_t = \sum_{n=0}^N d_{2n+1} \mathbf{1}_{[0,t]}(\tau'_n), \quad t \ge 0.$$

Using this identification, we find that  $\tilde{S}_q^{p,\mathcal{T}}$ ,  $\tilde{D}_{q,q}^{p,\mathcal{T}}$ , and  $\tilde{D}_{p,q}^{p,\mathcal{T}}$  are Banach spaces. As a consequence,  $\mathcal{A}_{p,q}^{\mathcal{T}}$  is a well-defined Banach space that is isometrically isomorphic

to  $\hat{s}_{p,q}^{odd}$ . The duality statement now follows from the duality for  $s_{p,q}^{odd}$  and (\*) in (7.5.5) follows from (7.3.7).

Now let us show (7.5.6). By Doob's maximal inequality,

$$\left(\mathbb{E}\sup_{0\leq s\leq t}\|M_t\|_{L^q(S)}^p\right)^{\frac{1}{p}} \sim_p \left(\mathbb{E}\|M_t\|_{L^q(S)}^p\right)^{\frac{1}{p}}.$$

Again define a G-martingale difference sequence  $(d_n)_{n=0}^{2N+1}$  by setting  $d_{2n} = 0$ ,  $d_{2n+1} = \Delta M_{\tau_n}$ , where n = 0, ..., N. Then by Remark 7.3.3

$$\|M_{\infty}\|_{L^{p}(\Omega;X)} = \left\|\sum_{n=0}^{2N+1} d_{k}\right\|_{L^{p}(\Omega;X)} \approx_{p,q} \|(d_{n})_{n=0}^{2N+1}\|_{s_{p,q}^{odd}} = \|M\|_{\mathscr{A}_{p,q}^{\mathcal{T}}}.$$

To treat the general case we use an approximation argument based on the following observation.

**Lemma 7.5.6.** Let  $1 < p, q < \infty$ . Let M be in  $\tilde{S}_q^p$ ,  $\tilde{D}_{q,q}^p$ , or  $\tilde{D}_{p,q}^p$  and let  $\mathcal{T} = (\tau_n)_{n\geq 0}$  be any sequence of predictable stopping times with disjoint graphs that exhausts the jumps of M. Consider the process  $M^n = M_{\mathcal{T}}^n$  defined in (7.5.2). Then  $M^n \to M$  in  $\tilde{S}_q^p$ ,  $\tilde{D}_{q,q}^p$ , or  $\tilde{D}_{p,q}^p$ , respectively. As a consequence,  $\tilde{S}_q^p$ ,  $\tilde{D}_{q,q}^p$ , and  $\tilde{D}_{p,q}^p$  are normed linear spaces and  $\mathscr{A}_{p,q}$  given by

$$\begin{split} \widetilde{S}_{q}^{p} \cap \widetilde{D}_{q,q}^{p} \cap \widetilde{D}_{p,q}^{p} & \text{if } 2 \leq q \leq p < \infty, \\ \widetilde{S}_{q}^{p} \cap (\widetilde{D}_{q,q}^{p} + \widetilde{D}_{p,q}^{p}) & \text{if } 2 \leq p \leq q < \infty, \\ (\widetilde{S}_{q}^{p} \cap \widetilde{D}_{q,q}^{p}) + \widetilde{D}_{p,q}^{p} & \text{if } 1 < p < 2 \leq q < \infty, \\ (\widetilde{S}_{q}^{p} + \widetilde{D}_{q,q}^{p}) \cap \widetilde{D}_{p,q}^{p} & \text{if } 1 < q < 2 \leq p < \infty, \\ \widetilde{S}_{q}^{p} + (\widetilde{D}_{q,q}^{p} \cap \widetilde{D}_{p,q}^{p}) & \text{if } 1 < q \leq p \leq 2, \\ \widetilde{S}_{q}^{p} + \widetilde{D}_{q,q}^{p} + \widetilde{D}_{p,q}^{p} & \text{if } 1 < p \leq q \leq 2. \end{split}$$
(7.5.7)

is a well-defined normed linear space. If  $M \in \mathcal{A}_{p,q}$ , then there exists a sequence of predictable stopping times  $\mathcal{T}$  with disjoint graphs that exhausts the jumps of M so that  $M^n_{\mathcal{T}} \to M$  in  $\mathcal{A}_{p,q}$ .

*Proof.* We prove the two first statements only for  $\tilde{S}_q^p$ . By the dominated convergence theorem, we obtain  $M^n \to M$  in  $\tilde{S}_q^p$  as well as  $||M^n||_{\tilde{S}_q^p} \nearrow ||M||_{\tilde{S}_q^p}$ . Suppose now that  $M, N \in \tilde{S}_q^p$ . By [85, Lemma I.2.23], there exists a sequence  $\mathcal{T} = \{\tau_n\}_{n \ge 0}$  of predictable stopping times with disjoint graphs that exhausts the jumps of both M and N. Now clearly,  $(M + N)^n = M^n + N^n$  and so

$$\|M+N\|_{\widetilde{S}^p_q} = \lim_{n \to \infty} \|M^n + N^n\|_{\widetilde{S}^p_q}$$
$$= \lim_{n \to \infty} \|M^n + N^n\|_{\widetilde{S}^{p,\mathcal{F}}_q}$$

$$\leq \lim_{n \to \infty} \|M^n\|_{\widetilde{S}^{p,\mathcal{T}}_a} + \|N^n\|_{\widetilde{S}^{p,\mathcal{T}}_a} = \|M\|_{\widetilde{S}^p_q} + \|N\|_{\widetilde{S}^p_q}.$$

Let us prove the final statement if  $p \le q \le 2$ , the other cases are similar. Let  $M \in \mathcal{A}_{p,q}$ and let  $M_1 \in \widetilde{S}_q^p$ ,  $M_2 \in \widetilde{D}_{q,q}^p$ ,  $M_3 \in \widetilde{D}_{p,q}^p$  be such that  $M = M_1 + M_2 + M_3$ . Let  $\mathcal{T} = \{\tau_n\}_{n\ge 0}$ be a sequence of predictable stopping times with disjoint graphs that exhausts the jumps of  $M_1$ ,  $M_2$  and  $M_3$ . Then  $M^n = M_1^n + M_2^n + M_3^n$  and by the above,

$$\|M - M^n\|_{\mathscr{A}_{p,q}} \le \|M_1 - M_1^n\|_{\widetilde{S}_q^{p,\mathcal{F}}} + \|M_2 - M_2^n\|_{\widetilde{D}_{q,q}^p} + \|M_3 - M_3^n\|_{\widetilde{D}_{p,q}^p} \to 0$$

as  $n \to \infty$ .

**Lemma 7.5.7.** Let  $1 < p, q < \infty$ ,  $N \ge 1$ ,  $\mathcal{T} = (\tau_n)_{n=0}^N$  be a finite family of predictable stopping times with disjoint graphs. Then  $\mathscr{A}_{p,q}^{\mathcal{T}} \hookrightarrow \mathscr{A}_{p,q}$  isometrically.

*Proof.* We will consider only the case  $p \le q \le 2$ , the other cases can be shown analogously. Let  $M \in \mathscr{A}_{p,q}^{\mathscr{T}}$ . Then automatically  $M \in \mathscr{A}_{p,q}$  and  $||M||_{\mathscr{A}_{p,q}^{\mathscr{T}}} \ge ||M||_{\mathscr{A}_{p,q}}$ . Let us show the reverse inequality. Fix  $\varepsilon > 0$ , and let  $M^1 \in \widetilde{S}_q^p$ ,  $M^2 \in \widetilde{D}_{q,q}^p$  and  $M^3 \in \widetilde{D}_{p,q}^p$  be martingales such that  $M = M^1 + M^2 + M^3$  and

$$\|M\|_{\mathscr{A}_{p,q}} \ge \|M^1\|_{\widetilde{S}_q^p} + \|M^2\|_{\widetilde{D}_{q,q}^p} + \|M^3\|_{\widetilde{D}_{p,q}^p} - \varepsilon.$$

By Lemma 2.4.5 we can define martingales  $\widetilde{M}^1$ ,  $\widetilde{M}^2$  and  $\widetilde{M}^3$  by

$$\widetilde{M}_t^i = \sum_{s \in \mathcal{T} \cap [0,t]} \Delta M_s^i, \quad t \ge 0, \ i = 1, 2, 3.$$

$$(7.5.8)$$

Notice that  $|\Delta \widetilde{M}_t^i(\omega)(s)| \le |\Delta M_t^i(\omega)(s)|$  for each  $t \ge 0$ ,  $\omega \in \Omega$ ,  $s \in S$  and i = 1, 2, 3. Therefore  $\widetilde{M}^1 \in \widetilde{S}_q^p$ ,  $\widetilde{M}^2 \in \widetilde{D}_{q,q}^p$  and  $\widetilde{M}^3 \in \widetilde{D}_{p,q}^p$  and  $\|\widetilde{M}^1\|_{\widetilde{S}_q^p} \le \|M^1\|_{\widetilde{S}_q^p}$ ,  $\|\widetilde{M}^2\|_{\widetilde{D}_{q,q}^p} \le \|M^2\|_{\widetilde{D}_{p,q}^p}$  and  $\|\widetilde{M}^3\|_{\widetilde{D}_{p,q}^p} \le \|M^3\|_{\widetilde{D}_{p,q}^p}$ . Moreover,  $M = \widetilde{M}^1 + \widetilde{M}^2 + \widetilde{M}^3$ . Indeed, since all the martingales here are purely discontinuous with accessible jumps, by (7.5.8) we find for each  $t \ge 0$  a.s.

$$\begin{split} M_t &= \sum_{s \in \mathcal{T} \cap [0,t]} \Delta M_s = \sum_{s \in \mathcal{T} \cap [0,t]} \left( \Delta M_s^1 + \Delta M_s^2 + \Delta M_s^3 \right) \\ &= \sum_{s \in \mathcal{T} \cap [0,t]} \Delta M_s^1 + \sum_{s \in \mathcal{T} \cap [0,t]} \Delta M_s^2 + \sum_{s \in \mathcal{T} \cap [0,t]} \Delta M_s^3 \\ &= \widetilde{M}_t^1 + \widetilde{M}_t^2 + \widetilde{M}_t^3. \end{split}$$

Therefore

$$\begin{split} \|M\|_{\mathscr{A}_{p,q}^{\mathcal{F}}} &\leq \|\widetilde{M}^1\|_{\widetilde{S}_q^p} + \|\widetilde{M}^2\|_{\widetilde{D}_{q,q}^p} + \|\widetilde{M}^3\|_{\widetilde{D}_{p,q}^p} \\ &\leq \|M^1\|_{\widetilde{S}_q^p} + \|M^2\|_{\widetilde{D}_{q,q}^p} + \|M^3\|_{\widetilde{D}_{p,q}^p} \leq \|M\|_{\mathscr{A}_{p,q}} + \varepsilon \end{split}$$

Since  $\varepsilon$  was arbitrary, we conclude that  $||M||_{\mathscr{A}_{p,q}^{\mathcal{T}}} \leq ||M||_{\mathscr{A}_{p,q}}$ , and consequently  $||M||_{\mathscr{A}_{p,q}^{\mathcal{T}}} = ||M||_{\mathscr{A}_{p,q}}$ .

We can now readily deduce the main theorem of this section.

**Theorem 7.5.8.** Let  $1 < p, q < \infty, M : \mathbb{R}_+ \times \Omega \to L^q(S)$  be a purely discontinuous martingale with accessible jumps. Then for all  $t \ge 0$  one has that

$$\left(\mathbb{E}\sup_{0\leq s\leq t}\|M_t\|_{L^q(S)}^p\right)^{\frac{1}{p}} \approx_{p,q} \|M\mathbf{1}_{[0,t]}\|_{\mathcal{A}_{p,q}},\tag{7.5.9}$$

where  $\mathcal{A}_{p,q}$  is as in (7.5.7). In particular,  $\mathcal{A}_{p,q}$  is a Banach space of  $L^p$ -martingales.

*Proof.* Suppose first that  $M \in \mathscr{A}_{p,q}$ . By Lemma 7.5.6 there exists a sequence of predictable stopping times with disjoint graphs that exhausts the jumps of M so that  $M^n_{\mathcal{T}} \to M$  in  $\mathscr{A}_{p,q}$ . In particular,  $(M^n_{\mathcal{T}})_{n\geq 0}$  is Cauchy in  $\mathscr{A}_{p,q}$ . By Lemma 7.5.7 and Theorem 7.5.5 it follows that it is also Cauchy in  $\mathscr{M}^{\text{acc}}_{p,q}$ . By Proposition 7.5.3  $(M^n_{\mathcal{T}})_{n\geq 0}$  converges and clearly the limit is M.

Suppose now that  $M \in \mathcal{M}_{p,q}^{\text{acc}}$ . It suffices to show that  $M \in \mathcal{A}_{p,q}$ . Indeed, Lemma 7.5.6 then shows that there is a sequence of predictable stopping times with disjoint graphs that exhausts the jumps of M so that  $M_{\mathcal{T}}^n \to M$  in  $\mathcal{A}_{p,q}$ . By Lemma 7.5.1 we also have  $M_{\mathcal{T}}^n \to M$  in  $\mathcal{M}_{p,q}^{\text{acc}}$  and so the lower bound in (7.5.9) follows from Lemma 7.5.7 and Theorem 7.5.5. We will show that  $M \in \mathcal{A}_{p,q}$  in the two cases  $2 \le q \le p$  and  $p \le q \le 2$ , the other cases can be treated analogously.

*Case*  $2 \le q \le p$ . We will show that  $||M||_{\widetilde{S}_q^p} \lesssim_{p,q} ||M||_{\mathscr{M}_{p,q}^{\operatorname{acc}}}$ . The analogous statements for  $\widetilde{D}_{a,q}^p$  and  $\widetilde{D}_{p,q}^p$  can be shown in the same way. By Theorem 7.5.5,

$$\|M^n\|_{\widetilde{S}^p_q} \lesssim_{p,q} \|M^n\|_{\mathcal{M}^{\mathrm{acc}}_{p,q}}$$

Also, by Theorem 3.3.17 we have  $||M^n||_{\mathcal{M}_{p,q}^{\mathrm{acc}}} \lesssim_{p,q} ||M||_{\mathcal{M}_{p,q}^{\mathrm{acc}}}$  for all  $n \ge 0$ . Therefore  $||M^n||_{\widetilde{S}_{p}^p} \lesssim_{p,q} ||M||_{\mathcal{M}_{p,q}^{\mathrm{acc}}}$  uniformly in n, so by monotone convergence

$$\begin{split} \|M\|_{\widetilde{S}_{q}^{p}}^{p} &= \mathbb{E}\left\|\left(\sum_{m\geq 0} \mathbb{E}_{\mathscr{F}_{\tau_{m}}} |(\Delta M(\omega)(s))_{\tau_{m}}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p} \\ &= \lim_{n\rightarrow\infty} \mathbb{E}\left\|\left(\sum_{m=0}^{n} \mathbb{E}_{\mathscr{F}_{\tau_{m}}} |(\Delta M(\omega)(s))_{\tau_{m}}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p} \\ &= \lim_{n\rightarrow\infty} \|M^{n}\|_{\widetilde{S}_{q}^{p}}^{p} \lesssim_{p,q} \|M\|_{\mathscr{M}_{p,q}^{\mathrm{acc}}}^{p}. \end{split}$$

*Case*  $p \le q \le 2$ . Observe that  $||M^n||_{\mathscr{A}_{p,q}} = \sum_{p,q} ||M^n||_{\mathscr{M}_{p,q}^{acc}}$  for each  $n \ge 0$  by Theorem 7.5.5 and since  $(M^n)_{n\ge 0}$  is a Cauchy sequence in  $\mathscr{M}_{p,q}^{acc}$  due to Lemma 7.5.1, it follows that  $(M^n)_{n\ge 0}$  is a Cauchy sequence in  $\mathscr{A}_{p,q}$ . Thus there exists a subsequence  $(M^{n_k})_{k\ge 0}$  such that

$$\|M^{n_{k+1}} - M^{n_k}\|_{\mathscr{A}_{p,q}} < \frac{1}{2^{k+1}}, \ k \ge 0.$$

Let  $N^k = M^{n_k} - M^{n_{k-1}}$ ,  $k \ge 1$ ,  $N^0 = M^{n_0}$ . Set  $n_{-1} = -1$ . By Theorem 7.5.5, for each  $k \ge 0$  there exist  $N^{k,1}$ ,  $N^{k,2}$  and  $N^{k,3}$  such that  $N^{k,1} \in \tilde{S}^p_q$ ,  $N^{k,2} \in \tilde{D}^p_{q,q}$ ,  $N^{k,3} \in \tilde{D}^p_{p,q}$ ,  $N^k = N^{k,1} + N^{k,2} + N^{k,3}$ ,

$$\{t: \Delta N_t^{k,i} \neq 0, i = 1, 2, 3\} \subset \{\tau_{n_{k-1}+1}, \dots, \tau_{n_k}\}, \text{ a.s.},$$

and

$$\|N^{k,1}\|_{\tilde{S}^{p}_{q}} + \|N^{k,2}\|_{\tilde{D}^{p}_{q,q}} + \|N^{k,3}\|_{\tilde{D}^{p}_{p,q}} < \frac{1}{2^{k}}, \quad k \ge 1, \\\|N^{0,1}\|_{\tilde{S}^{p}_{q}} + \|N^{0,2}\|_{\tilde{D}^{p}_{q,q}} + \|N^{0,3}\|_{\tilde{D}^{p}_{p,q}} \le 2\|M^{n_{0}}\|_{\mathscr{A}_{p,q}}.$$

$$(7.5.10)$$

Let

$$M^{m,i} := \sum_{k=0}^{m} N^{k,i}, \quad m \ge 0, \quad i = 1, 2, 3.$$

Then by (7.5.10),  $(M^{m,1})_{m\geq 0}$ ,  $(M^{m,2})_{m\geq 0}$  and  $(M^{m,3})_{m\geq 0}$  are Cauchy sequences in  $\widetilde{S}_{q}^{p}$ ,  $\widetilde{D}_{q,q}^{p}$  and  $\widetilde{D}_{p,q}^{p}$  respectively. By construction, each of  $M^{m,i}$ ,  $m \geq 0$ , i = 1,2,3, has finitely many jumps occurring in  $\{\tau_{0}, ..., \tau_{n_{m}}\}$ , so by Theorem 7.5.5 the sequences  $(M^{m,1})_{m\geq 0}$ ,  $(M^{m,2})_{m\geq 0}$  and  $(M^{m,3})_{m\geq 0}$  are Cauchy in  $\mathcal{M}_{p,q}^{\mathrm{acc}}$  as well. Due to Proposition 7.5.3 there exist  $\widetilde{M}^{1}$ ,  $\widetilde{M}^{2}$  and  $\widetilde{M}^{3}$  such that  $M^{m,i} \to \widetilde{M}^{i}$  in  $\mathcal{M}_{p,q}^{\mathrm{acc}}$  as  $m \to \infty$  for each i = 1,2,3. Since  $M^{m,1} + M^{m,2} + M^{m,3} \to M$  in  $\mathcal{M}_{p,q}^{\mathrm{acc}}$  as  $m \to \infty$  by Lemma 7.5.1, it follows that  $M = \widetilde{M}^{1} + \widetilde{M}^{2} + \widetilde{M}^{3}$ .

Let us now show that the jumps of  $\widetilde{M}^1$ ,  $\widetilde{M}^2$  and  $\widetilde{M}^3$  are exhausted by  $\mathcal{T} = (\tau_n)_{n\geq 0}$ . Indeed, assume that for some i = 1, 2, 3 there exists a predictable stopping time  $\tau$  such that  $\mathbb{P}\{\Delta \widetilde{M}^i_{\tau} \neq 0, \tau \notin \{\tau_0, \tau_1, \ldots\}\} > 0$ . Then by separability of  $X = L^q(S)$  there exists an  $x^* \in X^*$  such that

$$\mathbb{P}\{\langle \Delta \widetilde{M}_{\tau}^{i}, x^{*} \rangle \neq 0, \tau \notin \{\tau_{0}, \tau_{1}, \ldots\}\} > 0$$
(7.5.11)

and so, by the Burkholder-Davis-Gundy inequality,

$$\mathbb{E}|\langle (M^{m,i} - \widetilde{M}^{i})_{\infty}, x^{*} \rangle|^{p} \approx_{p} \mathbb{E}[\langle M^{m,i} - \widetilde{M}^{i}, x^{*} \rangle]_{\infty}^{\frac{p}{2}}$$
$$= \mathbb{E}\Big(\sum_{u \ge 0} |\langle \Delta (M^{m,i} - \widetilde{M}^{i})_{u}, x^{*} \rangle|^{2}\Big)^{\frac{p}{2}}$$
$$\geq \mathbb{E}|\langle \Delta \widetilde{M}^{i}_{\tau}, x^{*} \rangle|^{p} \mathbf{1}_{\tau \notin \{\tau_{0}, \tau_{1}, \ldots\}},$$
(7.5.12)

where the final inequality holds as  $\mathbb{P}\{\Delta M_{\tau}^{m,i} \neq 0, \tau \notin \{\tau_0, \tau_1, \ldots\}\} = 0$ . But the last expression in (7.5.12) does not vanish as  $m \to \infty$  because of (7.5.11), which contradicts with the fact that  $M^{m,i} \to \widetilde{M}^i$  in  $\mathcal{M}_{p,q}^{acc}$ .

By monotone convergence,

$$\begin{split} \|\widetilde{M}^{1}\|_{\widetilde{S}^{p}_{q}}^{p} &= \mathbb{E}\left\|\left(\sum_{n\geq 0} \mathbb{E}_{\mathscr{F}_{\tau_{n}-}} |(\Delta\widetilde{M}^{1}(\omega)(s))_{\tau_{n}}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p} \\ &= \lim_{m \to \infty} \mathbb{E}\left\|\left(\sum_{n=0}^{n_{m}} \mathbb{E}_{\mathscr{F}_{\tau_{n}-}} |(\Delta\widetilde{M}^{1}(\omega)(s))_{\tau_{n}}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p} \\ &= \lim_{m \to \infty} \|M^{m,1}\|_{\widetilde{S}^{p}_{q}}^{p}, \end{split}$$

and the last expression is bounded due to the fact that  $M^{m,1}$  is a Cauchy sequence in  $\widetilde{S}_q^p$ . By the same reasoning  $\widetilde{M}^2 \in \widetilde{D}_{q,q}^p$  and  $\widetilde{M}^3 \in \widetilde{D}_{p,q}^p$ , so  $M \in \mathscr{A}_{p,q}$ . This completes the proof.

Theorem 7.5.8 and Lemma 7.5.26 yield the following sharp estimates.

**Corollary 7.5.9.** Let  $1 < p, q < \infty$ ,  $M : \mathbb{R}_+ \times \Omega \to H$  be a purely discontinuous  $L^p$ -martingale with accessible jumps,  $X = L^q(S)$ ,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X)$  be elementary predictable. Then for all  $t \ge 0$  one has that

$$\left(\mathbb{E}\sup_{0\leq s\leq t}\|(\Phi\cdot M)_t\|_{L^q(S)}^p\right)^{\frac{1}{p}} \sim_{p,q} \|(\Phi\mathbf{1}_{[0,t]})\cdot M\|_{\mathscr{A}_{p,q}},$$

where  $\mathcal{A}_{p,q}$  is as given in (7.5.7).

## 7.5.3. Quasi-left continuous purely discontinuous martingales

We now turn to estimates for the stochastic integral  $\Phi \cdot M$  in the case that M is a purely discontinuous quasi-left continuous local martingale. We will first show in Lemma 7.5.11 that one can (essentially) represent  $\Phi \cdot M$  as a stochastic integral  $\Phi_H \star \bar{\mu}^M$ , where  $\bar{\mu}^M$  is the compensated version of the jump measure  $\mu^M$  of M. Afterwards, in Theorem 7.5.22 we prove sharp bounds for stochastic integrals of the form  $f \star \bar{\mu}$ , where  $\mu$  is any integer-valued random measure with a compensator that is non-atomic in time. By combining these two observations, we immediately find sharp bounds for  $\Phi \cdot M$ .

To any purely discontinuous local martingale *M* we can associate an integervalued random measure  $\mu^M$  on  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{B}(H)$  by setting

$$\mu^M(\omega;B\times A):=\sum_{u\in B}\mathbf{1}_{A\backslash\{0\}}(\Delta M_u(\omega)), \ \omega\in\Omega,$$

for each  $B \in \mathscr{B}(\mathbb{R}_+)$ ,  $A \in \mathscr{B}(H)$ . That is,  $\mu^M(\omega; B \times A)$  counts the number of jumps within the time set *B* with size in *A* on the trajectory belonging to the sample point  $\omega$ .

Recall that a process  $M : \mathbb{R}_+ \times \Omega \to H$  is called quasi-left continuous if  $\Delta M_\tau = 0$ a.s. on the set  $\{\tau < \infty\}$  for each predictable stopping time  $\tau$  (see [85, Chapter I.2] for more information). If  $M : \mathbb{R}_+ \times \Omega \to H$  is a quasi-left continuous local martingale, then  $\mu^M$  is  $\widetilde{\mathscr{P}}$ - $\sigma$ -finite and there exists a compensator  $v^M$  (see e.g. [85, Proposition II.1.16] and [89, Theorem 25.22]). If M is, in addition, purely discontinuous, then the following characterization holds thanks to [85, Corollary II.1.19].

**Lemma 7.5.10.** Let *H* be a separable Hilbert space and  $M : \mathbb{R}_+ \times \Omega \to H$  be a purely discontinuous local martingale. Let  $\mu^M$  and  $\nu^M$  be the associated integer-valued random measure and its compensator. Then *M* is quasi-left continuous if and only if  $\nu^M$  is non-atomic in time.

Let us now show that  $\Phi \cdot M$  can(essentially) be represented as a stochastic integral with respect to  $\bar{\mu}_M$ .

**Lemma 7.5.11.** Let X be a Banach space, H be a Hilbert space,  $1 \le p < \infty$ ,  $M : \mathbb{R}_+ \times \Omega \rightarrow H$  be a purely discontinuous quasi-left continuous local martingale, and  $\Phi : \mathbb{R}_+ \times \Omega \rightarrow \mathcal{L}(H, X)$  be elementary predictable. Define  $\Phi_H : \mathbb{R}_+ \times \Omega \times H \rightarrow X$  by

$$\Phi_H(t,\omega,h) := \Phi(t,\omega)h, \quad t \ge 0, \omega \in \Omega, h \in H.$$

Then there exists an increasing sequence  $(A_n)_{n\geq 1} \in \widetilde{\mathscr{P}}$  such that  $\cup_n A_n = \mathbb{R}_+ \times \Omega \times J$ ,  $\Phi_H \mathbf{1}_{A_n}$  is integrable with respect to  $\overline{\mu}^M$  for each  $n \geq 1$ , and

- (i) if  $\Phi \cdot M \in L^p(\Omega; X)$  then  $(\Phi_H \mathbf{1}_{A_n}) \star \overline{\mu}^M \to \Phi \cdot M$  in  $L^p(\Omega; X)$ ;
- (*ii*) if  $\Phi \cdot M \notin L^p(\Omega; X)$  then  $\|(\Phi_H \mathbf{1}_{A_n}) \star \overline{\mu}^M\|_{L^p(\Omega; X)} \to \infty$  for  $n \to \infty$ .

*Proof.* For each  $k, l \ge 1$  we define a stopping time  $\tau_{k,l}$  by

$$\tau_{k,l} = \inf\{t \in \mathbb{R}_+ : \#\{s \in [0, t] : \|\Delta M_s\| \in [1/k, k]\} = l\}.$$

Since *M* has càdlàg trajectories,  $\tau_{k,l}$  is a.s. well-defined and takes its values in  $[0,\infty]$ . Moreover,  $\tau_{k,l} \to \infty$  for each  $k \ge 1$  a.s. as  $l \to \infty$ .

Set  $B_k = \{h \in H : ||h|| \in [1/k, k]\}$ . For each  $k, l \ge 1$  define  $A_{k,l} = \mathbf{1}_{[0,\tau_{k,l}] \times B_k} \subset \widetilde{\mathscr{P}}$ . Then  $\Phi_H \mathbf{1}_{A_{k,l}}$  is integrable with respect to  $\mu^M$ . Indeed, a.s.

$$\left((\Phi_H \mathbf{1}_{A_{k,l}}) \star \mu^M\right)_{\infty} \le \sup \|\Phi\| k \left(\mathbf{1}_{A_{k,l}} \star \mu^M\right)_{\infty} \le \sup \|\Phi\| kl.$$

Since  $\tau_{k,l} \to \infty$  for each  $k \ge 1$  a.s. as  $l \to \infty$ , we can find a subsequence  $(\tau_{k_n,l_n})_{n\ge 1}$  such that  $k_n \ge n$  for each  $n \ge 1$  and  $\inf_{m\ge n} \tau_{k_m,l_m} \to \infty$  a.s. as  $n \to \infty$ . Let  $\tau_n = \inf_{m\ge n} \tau_{k_m,l_m}$  and define  $(A_n)_{n\ge 1} \subset \widetilde{\mathscr{P}}$  by

$$A_n = \mathbf{1}_{[0,\tau_n] \times B_n}.$$

Then  $\cup_n A_n = \mathbb{R}_+ \times \Omega \times J$  and  $\Phi_H \mathbf{1}_{A_n}$  is integrable with respect to  $\overline{\mu}^M$  for all  $n \ge 1$ .

Now prove that  $(\Phi_H \mathbf{1}_{A_n}) \star \overline{\mu}^M \to \Phi \cdot M$  in  $L^p(\Omega; X)$ . Since  $\Phi$  is simple, it takes its values in a finite dimensional subspace of X, so we can endow X with a Euclidean norm  $\|\cdot\|$ . First suppose that  $(\Phi \cdot M)_{\infty} \notin L^p(\Omega; X)$ . By the Burkholder-Davis-Gundy inequality this is equivalent to the fact that  $[\Phi \cdot M]_{\infty}^{\frac{1}{2}} \notin L^p(\Omega; X)$ . Notice that

$$\mathbb{E} \left\| (\Phi_{H} \mathbf{1}_{A_{n}}) \star \bar{\mu}^{M} \right|_{\infty} \right\|^{p} \approx_{p} \mathbb{E} \left[ (\Phi_{H} \mathbf{1}_{A_{n}}) \star \bar{\mu}^{M} \right]_{\infty}^{\frac{p}{2}} \\ = \mathbb{E} \left( \sum_{t \in [0, \tau_{n}]} \left\| \Delta (\Phi \cdot M)_{t} \right\|^{2} \mathbf{1}_{\|\Delta M_{t}\| \in [1/n, n]} \right)^{\frac{p}{2}},$$

and the last expression monotonically goes to infinity since  $\tau_n \rightarrow \infty$  a.s. and

$$\mathbb{E}\left(\sum_{t\geq 0} \|\Delta(\Phi\cdot M)_t\|^2\right)^{\frac{p}{2}} = \mathbb{E}[\Phi\cdot M]_{\infty}^{\frac{p}{2}} = \infty.$$

So if  $(\Phi \cdot M)_{\infty} \notin L^{p}(\Omega; X)$ , then  $\| ((\Phi_{H} \mathbf{1}_{A_{n}}) \star \overline{\mu}^{M})_{\infty} \|_{L^{p}(\Omega; X)} \to \infty$  as  $n \to \infty$ .

Now assume that  $(\Phi \cdot M)_{\infty} \in L^{p}(\Omega; X)$ . Then

$$\mathbb{E} \left\| \left( \Phi \cdot M \right)_{\infty} - \left( \left( \Phi_{H} \mathbf{1}_{A_{n}} \right) \star \bar{\mu}^{M} \right)_{\infty} \right\|^{p} \approx_{p} \mathbb{E} \left[ \Phi \cdot M - \left( \Phi_{H} \mathbf{1}_{A_{n}} \right) \star \bar{\mu}^{M} \right]_{\infty}^{\frac{p}{2}} \\ = \mathbb{E} \left( \sum_{t \in [0, \tau_{n}]} \left\| \Delta (\Phi \cdot M)_{t} \right\|^{2} \mathbf{1}_{\|\Delta M_{t}\| \notin [1/n, n]} \\ + \sum_{t \in (\tau_{n}, \infty)} \left\| \Delta (\Phi \cdot M)_{t} \right\|^{2} \right)^{\frac{p}{2}} \to 0, \ n \to \infty$$

by the dominated convergence theorem.

By Lemmas 7.5.10 and 7.5.11 it now suffices to obtain sharp bounds for the stochastic integral  $(F \star \bar{\mu})_{\infty}$ , where  $\mu$  is any optional integer-valued random measure whose compensator v is non-atomic in time.

#### 7.5.4. Integrals with respect to random measures

Throughout this subsection,  $\mu$  denotes an optional integer-valued random measure whose compensator  $\nu$  is non-atomic in time, i.e.,  $\nu(\{t\} \times J) = 0$  a.s. for all  $t \ge 0$ . The following result was first shown in [131, Theorem 1].

**Lemma 7.5.12** (A.A. Novikov). Let  $f : \mathbb{R}_+ \times \Omega \times J \to \mathbb{R}$  be  $\widetilde{\mathcal{P}}$ -measurable. Then

$$\begin{split} \mathbb{E}|f \star \bar{\mu}|^p &\lesssim_p \mathbb{E}|f|^p \star v \text{ if } 1 \leq p \leq 2, \\ \mathbb{E}|f \star \bar{\mu}|^p &\lesssim_p (\mathbb{E}|f|^2 \star v)^{\frac{p}{2}} + \mathbb{E}|f|^p \star v \text{ if } p \geq 2. \end{split}$$

The following lemma easily follows from [85, Theorem II.1.33] (or from [70, p.98] and [131, (6)] as well).

**Lemma 7.5.13.** Let *H* be a Hilbert space,  $f : \mathbb{R}_+ \times \Omega \times J \to H$  be  $\widetilde{\mathscr{P}}$ -measurable. Then

$$\mathbb{E}\|f \star \bar{\mu}\|^{2} = \mathbb{E}\|f\|^{2} \star v.$$
(7.5.13)

Equivalently, for each  $\widetilde{\mathscr{P}}$ -measurable  $f, g : \mathbb{R}_+ \times \Omega \times J \to H$  such that  $\mathbb{E} ||f||^2 \star v < \infty$  and  $\mathbb{E} ||g||^2 \star v < \infty$ 

$$\mathbb{E}\langle f \star \bar{\mu}, g \star \bar{\mu} \rangle = \mathbb{E}\langle f, g \rangle \star \nu. \tag{7.5.14}$$

*Proof.* The case  $H = \mathbb{R}$  can be deduced from [85, II.1.34] as *v* is assumed to be non-atomic in time. By applying this special case coordinate-wise, we obtain the general case.

**Corollary 7.5.14.** Let X be a Banach space,  $1 , <math>\mu$  be a random measure,  $\nu$  be the corresponding compensator,  $F : \mathbb{R}_+ \times \Omega \times J \to X$  and  $G : \mathbb{R}_+ \times \Omega \times J \to X^*$  be simple  $\widetilde{\mathscr{P}}$ -measurable functions. Then for each  $A \in \widetilde{\mathscr{P}}$  such that  $\mathbb{E}(\mathbf{1}_A \star \mu)_{\infty} < \infty$  the stochastic integrals  $(F\mathbf{1}_A) \star \overline{\mu}$  and  $(G\mathbf{1}_A) \star \overline{\mu}$  are well-defined and

$$\mathbb{E}\langle (F\mathbf{1}_A) \star \bar{\mu}, (G\mathbf{1}_A) \star \bar{\mu} \rangle = \mathbb{E}(\langle F, G \rangle \mathbf{1}_A) \star \nu. \tag{7.5.15}$$

 $\square$
*Proof.* Without loss of generality we can assume that *X* is finite dimensional. By Lemma 2.8.2, we can also redefine  $F := F\mathbf{1}_A$ ,  $G := G\mathbf{1}_A$ . First notice that since  $||F||_{\infty}$ ,  $||G||_{\infty} < \infty$  and  $\mathbb{E}\mu(F \neq 0)$ ,  $\mathbb{E}\mu(G \neq 0) < \infty$ , both integrals  $F \star \bar{\mu}$  and  $G \star \bar{\mu}$  exist. Moreover, every finite dimensional space is isomorphic to a Hilbert space, so by Lemma 7.5.13 both  $F \star \bar{\mu}$  and  $G \star \bar{\mu}$  are  $L^2$ -integrable, and therefore the left-hand side of (7.5.15) is well-defined.

Now let *d* be the dimension of *X*,  $(x_k)_{k=1}^d$  and  $(x_k^*)_{k=1}^d$  be bases of *X* and  $X^*$  respectively. Then there exist simple  $\widetilde{\mathscr{P}}$ -measurable  $F^1, \ldots, F^d, G^1, \ldots, G^d : \mathbb{R}_+ \times \Omega \times J \to \mathbb{R}$  such that  $F = F^1 x_1 + \cdots + F^d x_d$  and  $G = G^1 x_1^* + \cdots + G^d x_d^*$ . Now (7.5.14) implies

$$\mathbb{E}\langle F \star \bar{\mu}, G \star \bar{\mu} \rangle = \sum_{k,l=1}^{d} \langle x_k, x_l^* \rangle \mathbb{E} \left( F^k \star \bar{\mu} \cdot G^l \star \bar{\mu} \right) = \sum_{k,l=1}^{d} \langle x_k, x_l^* \rangle \mathbb{E} (F^k G^l) \star v$$
$$= \mathbb{E} \left( \sum_{k,l=1}^{d} \langle x_k, x_l^* \rangle F^k G^l \right) \star v = \mathbb{E} \langle F, G \rangle \star v.$$

The following proposition extends Novikov's inequalities presented in Lemma 7.5.12 in the case that  $v(\mathbb{R}_+ \times J) \leq 1$  a.s. If  $X = L^q(S)$  this result can be seen as a special case of Theorem 7.5.22 below. In the proof we will use the measure  $\mathbb{P} \times v$  on  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$  that is defined by setting

$$\mathbb{P} \times \nu \Big(\bigcup_{i=1}^n A_i \times B_i\Big) := \sum_{i=1}^n \mathbb{E}(\mathbf{1}_{A_i} \nu(B_i)),$$

for disjoint  $A_i \in \mathscr{F}$  and disjoint  $B_i \in \mathscr{B}(\mathbb{R}_+) \otimes \mathscr{J}$ , and extending  $\mathbb{P} \times v$  to  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$  via the Carathéodory extension theorem.

**Proposition 7.5.15.** Suppose that  $v(\mathbb{R}_+ \times J) \leq 1$  a.s. Let X be a Banach space and  $f : \mathbb{R}_+ \times \Omega \times J \to X$  be simple  $\widetilde{\mathscr{P}}$ -measurable. Then for all 1

$$\mathbb{E} \| F \star \bar{\mu} \|^p \eqsim_p \mathbb{E} \| F \|^p \star \nu.$$

*Proof.* We first prove  $\leq_p$ , and later deduce  $\geq_p$  by a duality argument.

*Step 1: upper bounds.* The case  $X = \mathbb{R}$  follows from Lemma 7.5.12 and the fact that  $\|\cdot\|_{L^2(\mathbb{R}_+ \times \Omega \times J, \mathbb{P} \otimes v)} \leq \|\cdot\|_{L^p(\mathbb{R}_+ \times \Omega \times J, \mathbb{P} \otimes v)}$  for each  $p \geq 2$  since  $\mathbb{P} \otimes v(\mathbb{R}_+ \times \Omega \times J) \leq 1$ . Now let *X* be a general Banach space. Then

$$\mathbb{E}\|F \star \bar{\mu}\|^{p} \overset{(i)}{\lesssim_{p}} \mathbb{E}\|F \star \mu\|^{p} + \mathbb{E}\|F \star \nu\|^{p} \overset{(ii)}{\leq} \mathbb{E}\big|\|F\| \star \mu\big|^{p} + \mathbb{E}\big|\|F\| \star \nu\big|^{p} \overset{(iii)}{\lesssim_{p}} \mathbb{E}\big|\|F\| \star \bar{\mu}\big|^{p} + \mathbb{E}\big|\|F\| \star \nu\big|^{p} \overset{(iv)}{\lesssim_{p}} \mathbb{E}\|F\|^{p} \star \nu,$$

where (*i*) and (*iii*) follow from the fact that  $\bar{\mu} = \mu - v$  and the triangle inequality, (*ii*) follows from [79, Proposition 1.2.2], and (*iv*) follows from the real-valued case and the fact that a.s.

$$\|\cdot\|_{L^1(\mathbb{R}_+\times J;\nu)} \le \|\cdot\|_{L^p(\mathbb{R}_+\times J;\nu)}.$$

*Step 2: lower bounds.* We can assume that *X* is finite dimensional since *F* is simple. Let  $Y = L^{p}(\mathbb{R}_{+} \times \Omega \times J, \mathbb{P} \otimes v; X)$ . Recall that by [79, Proposition 1.3.3]  $Y^{*} = L^{p'}(\mathbb{R}_{+} \times \Omega \times J, \mathbb{P} \otimes v; X^{*})$  and  $(L^{p}(\Omega; X))^{*} = L^{p'}(\Omega; X^{*})$ . Therefore due to the upper bounds from Step 1 and Corollary 7.5.14

$$(\mathbb{E}||F||^{p} \star v)^{\frac{1}{p}} = \sup_{G \in Y^{*}: ||G|| \leq 1} \mathbb{E}\langle F, G \rangle \star v = \sup_{G \in Y^{*}: ||G|| \leq 1} \mathbb{E}\langle F \star \bar{\mu}, G \star \bar{\mu} \rangle$$
$$\lesssim_{p} \sup_{\xi \in L^{p'}(\Omega; X^{*}): ||\xi|| \leq 1} \mathbb{E}\langle F \star \bar{\mu}, \xi \rangle = (\mathbb{E}||F \star \bar{\mu}||^{p})^{\frac{1}{p}}.$$

*Remark* 7.5.16. The condition  $v(\mathbb{R}_+ \times J) \le 1$  a.s. is necessary in general. Indeed, let *N* be a Poisson process with intensity parameter  $\lambda$  and let  $\mu$  be the random measure on  $\mathbb{R}_+ \times \{0\}$  defined by  $\mu([0, t] \times \{0\}) = N_t$ . Then the corresponding compensator *v* satisfies  $v([0, t] \times \{0\}) = \lambda t$ . In particular,

$$\mathbb{E}|\mathbf{1}_{[0,1]} \star \bar{\mu}|^4 = \mathbb{E}|N-\lambda|^4 = \sum_{k=0}^{\infty} \frac{(k-\lambda)^4 \lambda^k e^{-\lambda}}{k!} = \lambda(3\lambda+1),$$

which is not comparable with  $\mathbb{E}|\mathbf{1}_{[0,1]}|^4 \star v = \lambda$  if  $\lambda$  is large.

The condition  $v(\mathbb{R}_+ \times J) \le 1$  a.s. is however not needed for the upper bounds if  $1 \le p \le 2$  and *X* is a Hilbert space. Indeed, for p = 1

$$\mathbb{E}\|F \star \bar{\mu}\| \le \mathbb{E}\|F \star \mu\| + \mathbb{E}\|F \star \nu\| \le \mathbb{E}\|F\| \star \mu + \mathbb{E}\|F\| \star \nu = 2\mathbb{E}\|F\| \star \nu,$$

and for case p = 2 follows immediately from Lemma 7.5.13:

$$\mathbb{E}\|F \star \bar{\mu}\|^2 = \mathbb{E}\|F\|^2 \star \nu.$$

Therefore by the vector-valued Riesz-Thorin theorem [79, Theorem 2.2.1] for each  $1 \le p \le 2$ 

$$\left(\mathbb{E}\|F\star\bar{\mu}\|^{p}\right)^{\frac{1}{p}} \leq 2\left(\mathbb{E}\|F\|^{p}\star\nu\right)^{\frac{1}{p}}.$$

**Corollary 7.5.17.** Suppose that  $v(\mathbb{R}_+ \times J) \leq 1$  a.s. Let X be a Banach space,  $f : \mathbb{R}_+ \times \Omega \times J \rightarrow X$  be simple  $\widetilde{\mathcal{P}}$ -measurable. Then for each  $p \in (1, \infty)$  a.s.

$$(\mathbb{E}\|F \star \bar{\mu}\|^p |\mathscr{F}_0) \eqsim_p (\mathbb{E}\|F\|^p \star \nu |\mathscr{F}_0). \tag{7.5.16}$$

*Proof.* Fix  $A \in \mathcal{F}_0$ . Then by Lemma 2.8.2 and Proposition 7.5.15

$$\mathbb{E}(\|F \star \bar{\mu}\|^p \cdot \mathbf{1}_A) = \mathbb{E}\|(F \cdot \mathbf{1}_A) \star \bar{\mu}\|^p = p \mathbb{E}\|F \cdot \mathbf{1}_A\|^p \star v = \mathbb{E}(\|F\|^p \star v \cdot \mathbf{1}_A).$$

Since A is arbitrary, (7.5.16) holds.

For each  $m \ge 1$  let  $\mathscr{P}_m$  be the  $\sigma$ -field on  $\mathbb{R}_+ \times \Omega$  generated by all  $\mathscr{P}$ -measurable  $f : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  such that  $f|_{(\frac{n}{2m}, \frac{n+1}{2m}] \times \Omega}$  is  $\mathscr{B}((\frac{n}{2^m}, \frac{n+1}{2^m}]) \otimes \mathscr{F}_{\frac{n}{2^m}}$ -measurable for each  $n \ge 0$ . Then the following theorem holds.

 $\square$ 

**Theorem 7.5.18.** Let  $f : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be bounded and  $\mathscr{P}$ -measurable. Then for each  $m \ge 1$ 

$$\mathbb{E}(f|\mathscr{P}_m)(s) = \sum_{n \ge 0} \mathbb{E}\left(f(s) \middle| \mathscr{F}_{\frac{n}{2^m}}\right), \quad s \in \left(\frac{n}{2^m}, \frac{n+1}{2^m}\right], n \ge 0.$$
(7.5.17)

*Moreover,*  $\mathbb{E}(f|\mathscr{P}_m) \to f \text{ a.s. on } \mathbb{R}_+ \times \Omega \text{ as } m \to \infty$ .

*Proof.* Let us first show (7.5.17). Fix  $m \ge 1$ . Fix a simple  $\mathscr{P}_m$ -measurable process  $g : \mathbb{R}_+ \times \Omega \to \mathbb{R}$ . Then for each  $n \ge 0$  and  $s \in (\frac{n}{2^m}, \frac{n+1}{2^m}]$  a random variable g(s) is  $\mathscr{F}_{\frac{n}{2^m}}$ -measurable. Define  $\tilde{f} : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  by

$$\widetilde{f}(s) = \sum_{n \ge 0} \mathbb{E}\left(f(s) \middle| \mathscr{F}_{\frac{n}{2^m}}\right) \mathbf{1}_{s \in \left(\frac{n}{2^m}, \frac{n+1}{2^m}\right]}, \quad s \ge 0.$$

Then for each  $n \ge 0$  and  $s \in (\frac{n}{2^m}, \frac{n+1}{2^m}]$ 

$$\mathbb{E}[(f(s) - \tilde{f}(s))g(s)] = \mathbb{E}\left[\mathbb{E}\left((f(s) - \tilde{f}(s))g(s)\middle|\mathscr{F}_{\frac{n}{2^{m}}}\right)\right]$$
$$= \mathbb{E}\left[\mathbb{E}\left((f(s) - \tilde{f}(s))\middle|\mathscr{F}_{\frac{n}{2^{m}}}\right)g(s)\right] = 0.$$

Therefore

$$\mathbb{E}\int_{\mathbb{R}_{+}} \left(f(s) - \widetilde{f}(s)\right)g(s) \,\mathrm{d}s = \int_{\mathbb{R}_{+}} \mathbb{E}\left[\left(f(s) - \widetilde{f}(s)\right)g(s)\right] \,\mathrm{d}s = 0,$$

and hence (7.5.17) holds. Now notice that  $(\mathscr{P}_m)_{m\geq 1}$  forms a filtration on  $\mathbb{R}_+ \times \Omega$ , and obviously  $\sigma \{ \cup_m \mathscr{P}_m \} = \mathscr{P}$ . Therefore the second part of the theorem follows from the martingale convergence theorem (see e.g. [89, Theorem 7.23]).

**Corollary 7.5.19.** Let  $F : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be an increasing predictable function such that  $F(t) - F(s) \le C(t - s)$  a.s. for all  $0 \le s \le t$  and for some fixed constant  $C \ge 0$  and F(0) = 0 a.s. Then for each fixed  $T \ge 0$ 

$$F(T) = \lim_{m \to \infty} \sum_{n=0}^{\lfloor 2^m T \rfloor - 1} \mathbb{E} \left[ F\left(\frac{n+1}{2^m}\right) - F\left(\frac{n}{2^m}\right) \middle| \mathscr{F}_{\frac{n}{2^m}} \right],$$

where the last limit holds a.s. and in  $L^p(\Omega)$  for all 1 .

For the proof we will need the following lemma.

**Lemma 7.5.20.** Let  $F : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be an increasing predictable function such that  $F(t) - F(s) \le C(t-s)$  a.s. for all  $0 \le s \le t$  and for some fixed constant  $C \ge 0$  and F(0) = 0 a.s. Then there exists a predictable  $f : \mathbb{R}_+ \times \Omega \to [0, C]$  such that  $F(T) = \int_0^T f(s) ds$  for each fixed  $T \ge 0$ .

*Proof. F* is a.s. differentiable in *t* because *F* is Lipschitz, so there exists  $f : \mathbb{R}_+ \times \Omega \rightarrow [0, C]$  such that for a.e.  $\omega \in \Omega$  and  $t \ge 0$ 

$$f(t,\omega) = \lim_{\varepsilon \to 0} \frac{F(t,\omega) - F((t-\varepsilon) \lor 0,\omega)}{\varepsilon}$$

Since *F* is predictable,  $t \mapsto F(t) - F((t-\varepsilon) \lor 0)$  is a predictable process as well for each  $\varepsilon \ge 0$ , so the obtained *f* is predictable.

*Proof of Corollary* 7.5.19. Let  $f : \mathbb{R}_+ \times \Omega \to [0, C]$  be as defined in Lemma 7.5.20. Then by Theorem 7.5.18,  $\mathbb{E}(f | \mathscr{P}_m)$  exists and converges to f a.s. on  $\mathbb{R}_+ \times \Omega$ . Moreover, fis bounded by C, so  $\mathbb{E}(f | \mathscr{P}_m)$  is bounded by C as well. Therefore for each  $m \ge 1$  we find using (7.5.17)

$$\sum_{n=0}^{[2^{m}T]-1} \mathbb{E}\left[F\left(\frac{n+1}{2^{m}}\right) - F\left(\frac{n}{2^{m}}\right)\right|\mathscr{F}_{2^{\overline{m}}}^{n}\right] = \sum_{n=0}^{[2^{m}T]-1} \mathbb{E}\left[\int_{\left(\frac{n}{2^{m}}, \frac{n+1}{2^{m}}\right)} f(s) \, \mathrm{d}s \left|\mathscr{F}_{\frac{n}{2^{m}}}\right]\right]$$
$$= \sum_{n=0}^{[2^{m}T]-1} \int_{\left(\frac{n}{2^{m}}, \frac{n+1}{2^{m}}\right)} E(f(s) \left|\mathscr{F}_{\frac{n}{2^{m}}}\right| \, \mathrm{d}s$$
$$= \int_{\left(0, \frac{[2^{m}T]}{2^{m}}\right]} \mathbb{E}(f|\mathscr{P}_{m})(s) \, \mathrm{d}s,$$

and since  $\frac{[2^m T]}{2^m} \to T$  as  $m \to \infty$ ,

$$\lim_{m \to \infty} \sum_{n=0}^{\lfloor 2^m T \rfloor - 1} \mathbb{E} \left[ F\left(\frac{n+1}{2^m}\right) - F\left(\frac{n}{2^m}\right) \middle| \mathscr{F}_{\frac{n}{2^m}} \right] = \lim_{m \to \infty} \int_{\left(0, \frac{\lfloor 2^m T \rfloor}{2^m}\right]} \mathbb{E}(f|\mathscr{P}_m)(s) \, \mathrm{d}s$$
$$= \int_{(0,T]} f(s) \, \mathrm{d}s = F(T),$$

where the limit holds a.s., and since  $F(T) \le CT$  and all the functions above are bounded by CT as well, by the dominated convergence theorem the limit holds in  $L^{p}(\Omega)$  for each 1 .

In the proof of Theorem 7.5.22 we will use a time-change argument. We recall some necessary definitions and results. A nondecreasing, right-continuous family of stopping times  $\tau = (\tau_s)_{s\geq 0}$  is called a *random time-change*. If  $\mathbb{F}$  is right-continuous, then according to [89, Lemma 7.3] the same holds true for the *induced filtration*  $\mathbb{G} = (\mathscr{G}_s)_{s\geq 0} = (\mathscr{F}_{\tau_s})_{s\geq 0}$ .

For a random time-change  $\tau = (\tau_s)_{s \ge 0}$  and for a random measure  $\mu$  we define  $\mu \circ \tau$  in the following way:

$$\mu \circ \tau((s,t] \times B) = \mu((\tau_s,\tau_t] \times A), \quad t \ge s \ge 0, A \in \mathcal{J}.$$

 $\mu$  is said to be  $\tau$ -continuous if  $\mu((\tau_{s-}, \tau_s] \times J) = 0$  a.s. for each  $s \ge 0$ , where we let  $\tau_{s-} := \lim_{\epsilon \to 0} \tau_{s-\epsilon}, \tau_{0-} := \tau_0$ . Later we will need the following proposition.

**Proposition 7.5.21.** Let  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  be a strictly increasing continuous predictable process such that  $A_0 = 0$  and  $A_t \to \infty$  as  $t \to \infty$  a.s. Then

$$\tau_s = \{t : A_t = s\}, s \ge 0.$$

defines a random time-change  $\tau = (\tau_s)_{s \ge 0}$ . It satisfies  $(A \circ \tau)(t) = (\tau \circ A)(t) = t$  a.s. for each  $t \ge 0$ . Let  $\mathbb{G} = (\mathscr{G}_s)_{s \ge 0} = (\mathscr{F}_{\tau_s})_{s \ge 0}$  be the induced filtration. Then  $(A_t)_{t \ge 0}$  is a random time-change with respect to  $\mathbb{G}$ . Moreover, for any random measure  $\mu$  the following hold:

- (*i*) if  $\mu$  is  $\mathbb{F}$ -optional, then  $\mu \circ \tau$  is  $\mathbb{G}$ -optional,
- (*ii*) *if*  $\mu$  *is*  $\mathbb{F}$ *-predictable, then*  $\mu \circ \tau$  *is*  $\mathbb{G}$ *-predictable,*
- (iii) if  $\mu$  is an  $\mathbb{F}$ -optional random measure with a compensator  $\nu$ , then  $\nu \circ \tau$  is a compensator of  $\mu \circ \tau$ , and for each  $\widetilde{\mathscr{P}}$ -measurable simple  $F : \mathbb{R}_+ \times \Omega \times J \to \mathbb{R}$  such that  $\mathbb{E}(F \star \mu)_{\infty} < \infty$  we have  $\mathbb{E}((F \circ \tau) \star (\mu \circ \tau))_{\infty} < \infty$  and a.s.

$$(F \star \mu)_{\infty} = ((F \circ \tau) \star (\mu \circ \tau))_{\infty},$$
  

$$(F \star \nu)_{\infty} = ((F \circ \tau) \star (\nu \circ \tau))_{\infty},$$
  
(7.5.18)

$$(F \star \bar{\mu})(\tau_s) = ((F \circ \tau) \star (\overline{\mu \circ \tau}))(s), \quad s \ge 0. \tag{7.5.19}$$

*Proof.* First of all notice that since *A* is strictly increasing and continuous a.s.,  $s \mapsto \tau_s$  is an a.s. continuous function, so any random measure  $\mu$  is  $\tau$ -continuous. Therefore (i) and (ii) follow from [84, Theorem 10.27(c,d)]. Let us prove (iii). The fact that  $v \circ \tau$  is a compensator of  $\mu \circ \tau$  holds due to [84, Theorem 10.27(e)], while the rest follows from [84, Theorem 10.28], and in particular (7.5.18) follows from the definition of  $\mu \circ \tau$  and  $v \circ \tau$ .

For more information on time-changes for random measures we refer to [84, Chapter X].

Let  $(S, \Sigma, \rho)$  be a measure space. For  $1 < p, q < \infty$  we define  $\hat{\mathscr{S}}_{q, r}^{p}$ ,  $\hat{\mathscr{D}}_{q, q}^{p}$  and  $\hat{\mathscr{D}}_{p, q}^{p}$  as the Banach spaces of all functions  $F : \mathbb{R}_{+} \times \Omega \times J \to L^{q}(S)$  that are  $\mathcal{P}$ -measurable and for which the corresponding norms are finite:

$$\|F\|_{\hat{\mathscr{D}}_{q}^{p}} := \left(\mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J}|F|^{2} \,\mathrm{d}\nu\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}}, \\\|F\|_{\hat{\mathscr{D}}_{q,q}^{p}} := \left(\mathbb{E}\left(\int_{\mathbb{R}_{+}\times J}\|F\|_{L^{q}(S)}^{q} \,\mathrm{d}\nu\right)^{\frac{p}{q}}\right)^{\frac{1}{p}},$$
(7.5.20)  
$$\|F\|_{\hat{\mathscr{D}}_{p,q}^{p}} := \left(\mathbb{E}\int_{\mathbb{R}_{+}\times J}\|F\|_{L^{q}(S)}^{p} \,\mathrm{d}\nu\right)^{\frac{1}{p}}.$$

We show in Appendix 7.A that

$$(\hat{\mathcal{S}}_{q}^{p})^{*} = \hat{\mathcal{S}}_{q'}^{p'}, \qquad (\hat{\mathcal{D}}_{q,q}^{p})^{*} = \hat{\mathcal{D}}_{q',q'}^{p'}, \qquad (\hat{\mathcal{D}}_{p,q}^{p})^{*} = \hat{\mathcal{D}}_{p',q'}^{p'}$$

hold isomorphically with constants depending only on *p* and *q*.

**Theorem 7.5.22.** Fix  $1 < p, q < \infty$ . Let  $\mu$  be an optional  $\widetilde{\mathcal{P}}$ - $\sigma$ -finite random measure on  $\mathbb{R}_+ \times \mathscr{J}$  and suppose that its compensator  $\nu$  is non-atomic in time. Then for any simple  $\widetilde{\mathcal{P}}$ -measurable  $F : \mathbb{R}_+ \times \Omega \times J \to L^q(S)$  and for any  $A \in \widetilde{\mathcal{P}}$  with  $\mathbb{E}\mathbf{1}_A \star \mu < \infty$ 

$$\left(\mathbb{E}\sup_{0\leq s\leq t}\|((F\mathbf{1}_{A})\star\bar{\mu})_{s}\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \sim_{p,q} \|F\mathbf{1}_{A}\mathbf{1}_{[0,t]}\|_{\mathscr{I}_{p,q}},\tag{7.5.21}$$

where  $\mathcal{I}_{p,q}$  is given by

$$\hat{\mathscr{P}}_{q}^{p} \cap \hat{\mathscr{D}}_{q,q}^{p} \cap \hat{\mathscr{D}}_{p,q}^{p} \text{ if } 2 \leq q \leq p < \infty,   
 \hat{\mathscr{P}}_{q}^{p} \cap (\hat{\mathscr{D}}_{q,q}^{p} + \hat{\mathscr{D}}_{p,q}^{p}) \text{ if } 2 \leq p \leq q < \infty,   
 (\hat{\mathscr{P}}_{q}^{p} \cap \hat{\mathscr{D}}_{q,q}^{p}) + \hat{\mathscr{D}}_{p,q}^{p} \text{ if } 1 (\hat{\mathscr{P}}_{q}^{p} + \hat{\mathscr{D}}_{q,q}^{p}) \cap \hat{\mathscr{D}}_{p,q}^{p} \text{ if } 1 < q < 2 \leq p < \infty,   
 \hat{\mathscr{P}}_{q}^{p} + (\hat{\mathscr{D}}_{q,q}^{p} \cap \hat{\mathscr{D}}_{p,q}^{p}) \text{ if } 1 < q \leq p \leq 2,   
 \hat{\mathscr{P}}_{q}^{p} + \hat{\mathscr{D}}_{q,q}^{p} + \hat{\mathscr{D}}_{p,q}^{p} \text{ if } 1 (7.5.22)$$

*Proof.* By Lemma 2.8.2 we can assume without loss of generality that  $F := F\mathbf{1}_A$ ,  $\mu := \mu \mathbf{1}_A$ , and that there exists a  $T \ge 0$  such that F(t) = 0 for each  $t \ge T$ . Since F is simple, it is uniformly bounded on  $\mathbb{R}_+ \times \Omega \times J$  and, due to the fact that  $\mathbb{E}\mathbf{1}_A \star \mu = \mathbb{E}\mu(\mathbb{R}_+ \times \Omega) < \infty$ , we find  $\mathbb{E}||F \star \mu|| < \infty$ . Consequently  $F \star \mu$  exists and it is a local martingale. Therefore Doob's maximal inequality implies

$$\left(\mathbb{E}\|(F\star\bar{\mu})_{t}\|^{p}\right)^{\frac{1}{p}} \approx_{p} \left(\mathbb{E}\sup_{0\leq s\leq t}\|(F\star\bar{\mu})_{s}\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}}$$

and so it is enough to show that

$$(\mathbb{E}\|(F\star\bar{\mu})_t\|^p)^{\frac{1}{p}} \approx_{p,q} \|F\mathbf{1}_{[0,t]}\|_{\mathscr{I}_{p,q}}.$$
(7.5.23)

The proof consists of two steps. In the first step we assume that v is absolutely continuous with respect to Lebesgue measure. In this case, we can derive the upper bounds in (7.5.23) from the Burkholder-Rosenthal inequalities, Corollary 7.5.17 and Corollary 7.5.19. The lower bounds then follow by duality. In the second step we deduce the general result via a time-change argument based on Proposition 7.5.21.

Step 1:  $v((s, t] \times J) \le (t - s)$  for each  $t \ge s \ge 0$  a.s. We will consider the cases  $2 \le q \le p < \infty$  and 1 , the proofs in the other cases are similar.

*Case*  $2 \le q \le p < \infty$ : Fix  $m \ge 1$ . Let  $F_n := F\mathbf{1}_{\left(\frac{n}{2^m}, \frac{n+1}{2^m}\right)}$  for each  $n \ge 0$ . Then

$$(d_n)_{n\geq 0} := \left( (F_n \star \bar{\mu})_\infty \right)_{n\geq 0}$$

is an  $L^q(S)$ -valued martingale difference sequence with respect to a filtration  $(\mathscr{F}_{\frac{n+1}{2m}})_{n\geq 0}$ . Theorem 7.1.1 implies

$$\mathbb{E} \| (F \star \bar{\mu})_{\infty} \|_{L^{q}(S)}^{p} = \mathbb{E} \left\| \sum_{n \ge 0} (F_{n} \star \bar{\mu})_{\infty} \right\|_{L^{q}(S)}^{p} = \mathbb{E} \left\| \sum_{n \ge 0} d_{n} \right\|_{L^{q}(S)}^{p} \sim_{p,q} \| (d_{n}) \|_{s_{p,q}}^{p} \\ \approx_{p} (\| (d_{n}) \|_{S_{q}^{p}}^{p} + \| (d_{n}) \|_{D_{q,q}^{p}}^{p} + \| (d_{n}) \|_{D_{p,q}^{p}}^{p})^{p}.$$

To bound  $||(d_n)||_{S^p_a}$ , observe that

$$\|(d_n)\|_{S^p_q} = \left(\mathbb{E}\left\|\sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^m}}} |d_n|^2 \right\|_{L^q(S)}^p\right)^{\frac{1}{p}} = \left(\mathbb{E}\left\|\sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^m}}} |(F_n \star \bar{\mu})_{\infty}|^2 \right\|_{L^q(S)}^p\right)^{\frac{1}{p}}$$

$$\stackrel{(*)}{\sim}_{p} \left( \mathbb{E} \left\| \sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^{m}}}} \left( |F_{n}|^{2} \star \nu \right)_{\infty} \right\|_{L^{q}(S)}^{p} \right)^{\frac{1}{p}}$$

$$= \left( \mathbb{E} \left\| \sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^{m}}}} \left( \left( |F|^{2} \star \nu \right)_{\frac{n+1}{2^{m}}} - \left( |F|^{2} \star \nu \right)_{\frac{n}{2^{m}}} \right) \right\|_{L^{q}(S)}^{p} \right)^{\frac{1}{p}},$$

$$(7.5.24)$$

where (\*) holds by Corollary 7.5.17 and the fact that

$$v\left(\left(\frac{n}{2^m}, \frac{n+1}{2^m}\right)\right) \le \frac{n+1}{2^m} - \frac{n}{2^m} = \frac{1}{2^m} \le 1.$$

Notice that for a.e.  $\omega \in \Omega$ , all  $s \in S$ , and each  $t \ge u \ge 0$ 

$$\begin{aligned} (|F|^2 \star v)_t(s,\omega) - (|F|^2 \star v)_u(s,\omega) &\leq \sup |F(s)|^2 (v((u,t] \times J)(\omega)) \\ &\leq \sup |F(s)|^2 (t-u), \end{aligned}$$

so by Corollary 7.5.19

$$\sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^{m}}}}\left( \left(|F|^{2} \star v\right)_{\frac{n+1}{2^{m}}} - \left(|F|^{2} \star v\right)_{\frac{n}{2^{m}}} \right) \to \left(|F|^{2} \star v\right)_{T} = \left(|F|^{2} \star v\right)_{\infty}$$

a.s. as  $m \to \infty$ . Therefore thanks to (7.5.24)

$$\|(d_n)\|_{S^p_q} \approx \left(\mathbb{E}\left\|\sum_n \mathbb{E}_{\mathscr{F}_{\frac{n}{2^m}}}\left((|F|^2 \star \nu)_{\frac{n+1}{2^m}} - (|F|^2 \star \nu)_{\frac{n}{2^m}}\right)\right\|_{L^q(S)}^p\right)^{\frac{1}{p}}$$

$$\xrightarrow{m \to \infty} (\mathbb{E}\|(|F|^2 \star \nu)_{\infty}\|_{L^q(S)}^p)^{\frac{1}{p}} = \|F\|_{\mathscr{S}_q^p}.$$
(7.5.25)

Now let us estimate  $||(d_n)||_{D^p_{a,a}}$ . Analogously to (7.5.24)

$$\begin{split} \|(d_{n})\|_{D_{q,q}^{p}} &= \left(\mathbb{E}\left(\sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^{m}}}} \|d_{n}\|_{L^{q}(S)}^{q}\right)^{\frac{p}{q}}\right)^{\frac{1}{p}} = \left(\mathbb{E}\left(\sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^{m}}}} \|(F_{n} \star \bar{\mu})_{\infty}\|_{L^{q}(S)}^{q}\right)^{\frac{p}{q}}\right)^{\frac{1}{p}} \\ &= \left(\mathbb{E}\left(\sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^{m}}}} \left(\|F_{n}\|_{L^{q}(S)}^{q} \star \nu\right)_{\infty}\right)^{\frac{p}{q}}\right)^{\frac{1}{p}} \\ &= \left(\mathbb{E}\left(\sum_{n} \mathbb{E}_{\mathscr{F}_{\frac{n}{2^{m}}}} \left((\|F\|_{L^{q}(S)}^{q} \star \nu)_{\frac{n+1}{2^{m}}} - (\|F\|_{L^{q}(S)}^{q} \star \nu)_{\frac{n}{2^{m}}}\right)\right)^{\frac{p}{q}}\right)^{\frac{1}{p}}, \end{split}$$
(7.5.26)

and similarly to (7.5.25) the last expression converges to  $||F||_{\mathscr{D}^p_{q,q}}$ . The same can be shown for  $\mathscr{D}^p_{p,q}$ .

*Case*  $1 : Let <math>\mathscr{I}_{\text{elem}}(\widetilde{\mathscr{P}})$  denote the linear space of all simple  $\widetilde{\mathscr{P}}$ -measurable  $L^q(S)$ -valued functions. This linear space is dense in  $\hat{\mathscr{P}}_q^p$ ,  $\hat{\mathscr{D}}_{p,q}^p$  and  $\hat{\mathscr{D}}_{q,q}^p$ . Let  $F \in \mathscr{I}_{\text{elem}}(\widetilde{\mathscr{P}})$ . Fix a decomposition  $F = F_1 + F_2 + F_3$  with  $F_\alpha \in \mathscr{I}_{\text{elem}}(\widetilde{\mathscr{P}})$ .

Fix  $m \ge 1$  and set  $F_{n,\alpha} = F_{\alpha} \mathbf{1}_{\left(\frac{n}{2^m}, \frac{n+1}{2^m}\right)}$ ,  $d_{n,\alpha} = F_{n,\alpha} \star \bar{\mu}$ ,  $\alpha = 1, 2, 3$ , so that

$$(F \star \bar{\mu})_T = (F \star \bar{\mu})_{\infty} = \sum_n d_{n,1} + d_{n,2} + d_{n,3}$$

Then by Theorem 7.1.1, (7.5.24), (7.5.25) and (7.5.26) we conclude that

$$\left(\mathbb{E}\|(F\star\bar{\mu})_{\infty}\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \lesssim_{p,q} \|F_{1}\|_{\mathscr{S}_{q}^{p}} + \|F_{2}\|_{\mathscr{D}_{p,q}^{p}} + \|F_{3}\|_{\mathscr{D}_{q,q}^{p}}$$

Since  $\mathscr{I}_{\text{elem}}(\widetilde{\mathscr{P}})$  is dense in  $\hat{\mathscr{P}}_{q}^{p}$ ,  $\hat{\mathscr{D}}_{p,q}^{p}$  and  $\hat{\mathscr{D}}_{q,q}^{p}$ , we conclude by taking the infimum over  $F_1, F_2, F_3$  as above that

$$\left(\mathbb{E}\|(F\star\bar{\mu})_{\infty}\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} \lesssim_{p,q} \|F\|_{\mathscr{I}_{p,q}}.$$

The duality argument: Fix  $t < \infty$ ,  $1 < p, q < \infty$ . Using the upper bounds in (7.5.23) we can obtain the stochastic integral  $(F \star \bar{\mu})_t$  as an  $L^p$ -limit of the integrals of the corresponding simple approximations of F in  $\mathscr{I}_{p,q}$ . Let Y be the closure of the linear subspace  $\cup_{F \in \mathscr{I}_{p,q}} (F \star \bar{\mu})_t$  in  $L^p(\Omega; L^q(S))$  and let  $X = \mathscr{I}_{p,q}$ . By Corollary 7.A.8,  $X^* = \mathscr{I}_{p',q'}$ . Let U (resp. V) be the dense subspace of X (resp.  $X^*$ ) consisting of all  $\widetilde{\mathscr{P}}$ -measurable simple  $L^q(S)$ -valued (resp.  $L^{q'}(S)$ -valued) functions. Define both  $j_0 : U \to Y$  and  $k_0 : V \to Y^*$  by  $F \mapsto (F \star \bar{\mu})_t$ . Note that  $k_0$  maps into  $Y^*$  since each  $(F \star \bar{\mu})_t$  is in  $L^{p'}(\Omega; L^{q'}(S))$ , so it defines a bounded linear functional on Y. By the upper bounds in (7.5.23),  $j_0$  and  $k_0$  are bounded. Moreover, by the definition of Y, ran  $j_0$  is dense in Y. Finally, by Corollary 7.5.14  $\langle F^*, F \rangle = \langle k_0(F^*), j_0(F) \rangle$  for all  $F \in U$  and  $F^* \in V$ . Now (7.5.21) follows from Lemma 7.2.1.

Step 2: general case. Recall that, due to our assumptions in the beginning of the proof,  $\mathbb{E}\mu(\mathbb{R}_+ \times \Omega) = \mathbb{E}\nu(\mathbb{R}_+ \times \Omega) < \infty$ . Since  $\nu$  is non-atomic in time, we can define a continuous strictly increasing predictable process  $A : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  by

$$A_t = v([0, t] \times J) + t, t \ge 0.$$

Let  $\tau = (\tau_s)_{s\geq 0}$  be the time-change defined in Proposition 7.5.21. Then according to Proposition 7.5.21 the random measure  $\mu_{\tau} := \mu \circ \tau$  is G-optional, where  $G := (\mathscr{G}_s)_{s\geq 0} = (\mathscr{F}_{\tau_s})_{s\geq 0}$ . Moreover,  $v_{\tau} := v \circ \tau$  is G-predictable and a compensator of  $\mu_{\tau}$ . Let  $G := F \circ \tau$ . Notice that for each  $t \geq s \geq 0$  a.s.

$$v_{\tau}((s,t] \times J) = v((\tau_{s},\tau_{t}] \times J) = v((0,\tau_{t}] \times J) - v((0,\tau_{s}] \times J)$$
  

$$\leq v((0,\tau_{t}] \times J) - v((0,\tau_{s}] \times J) + (\tau_{t} - \tau_{s})$$
  

$$= (v((0,\tau_{t}] \times J) + \tau_{t}) - (v((0,\tau_{s}] \times J) + \tau_{s}) = t - s.$$
(7.5.27)

Let  $\mathscr{I}_{p,q}^{\tau}$  be defined as  $\mathscr{I}_{p,q}$  but for the random measure  $v_{\tau}$ . By (7.5.27) Step 1 yields  $\mathbb{E} \| G \star \bar{\mu}_{\tau} \|^p \approx_{p,q} \| G \|_{\mathscr{I}_{p,q}^{\tau}}^p$ . Indeed, by (7.5.19),  $\mathbb{E} \| (G \star \bar{\mu}_{\tau})_{\infty} \|^p = \mathbb{E} \| F \star \bar{\mu} \|^p$ . Moreover, for given  $F_i$  and  $G_i = F_i \circ \tau$ , i = 1, 2, 3, it follows from (7.5.18) that

$$\mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J}|G_{1}|^{2} \,\mathrm{d}\nu_{\tau}\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p} = \mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J}|F_{1}|^{2} \,\mathrm{d}\nu\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p} = \|F_{1}\|_{\mathscr{S}_{p}^{q}}^{p},\\ \mathbb{E}\left(\int_{\mathbb{R}_{+}\times J}\|G_{2}\|_{L^{q}(S)}^{q} \,\mathrm{d}\nu_{\tau}\right)^{\frac{p}{q}} = \mathbb{E}\left(\int_{\mathbb{R}_{+}\times J}\|F_{2}\|_{L^{q}(S)}^{q} \,\mathrm{d}\nu\right)^{\frac{p}{q}} = \|F_{2}\|_{\mathscr{S}_{p,q}^{p}}^{p},\\ \mathbb{E}\int_{\mathbb{R}_{+}\times J}\|G_{3}\|_{L^{q}(S)}^{p} \,\mathrm{d}\nu_{\tau} = \mathbb{E}\int_{\mathbb{R}_{+}\times J}\|F_{3}\|_{L^{q}(S)}^{p} \,\mathrm{d}\nu = \|F_{3}\|_{\mathscr{B}_{p,q}^{p}}^{p}.$$

Consequently,  $||G||_{\mathscr{I}_{p,q}^{\dagger}} = ||F||_{\mathscr{I}_{p,q}}$ . We conclude that  $\mathbb{E}||F \star \overline{\mu}||^p = \sum_{p,q} ||F||_{\mathscr{I}_{p,q}}^p$ .

*Remark* 7.5.23. Let us compare our result to the literature. The upper bounds in Theorem 7.5.22 were discovered in the scalar-valued case by A.A. Novikov in [131, Theorem 1]. By exploiting an orthonormal basis one can easily extend this result to the Hilbert-space valued integrands, see [114, Section 3.3] for details. The paper [114] contains several other proofs of the Hilbert-space valued version of Novikov's inequality. In the context of Poisson random measures, Theorem 7.5.22 was obtained in [51]. Some one-sided extensions of the latter result in the context of general Banach spaces were obtained in [52]. However, these bounds, which are based on the martingale type and cotype of the space, are only matching in the Hilbert-space case and not optimal in general (in particular for  $L^q$ -spaces). A very different proof of the upper bounds in Theorem 7.5.22, which exploits tools from stochastic analysis, was discovered independently of our work by Marinelli in [110].

As a corollary, we obtain the following sharp bounds for stochastic integrals.

**Theorem 7.5.24.** Fix  $1 < p, q < \infty$ . Let H be a Hilbert space,  $(S, \Sigma, \rho)$  be a measure space and let  $M : \mathbb{R}_+ \times \Omega \to H$  be a purely discontinuous quasi-left continuous local martingale. Let  $\Phi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(H, L^q(S))$  be elementary predictable. Then

$$\left(\mathbb{E}\sup_{0\leq s\leq t}\|(\Phi\cdot M)_s\|_{L^q(S)}^p\right)^{\frac{1}{p}} \approx_{p,q} \|\Phi_H \mathbf{1}_{[0,t]}\|_{\mathscr{I}_{p,q}},\tag{7.5.28}$$

where  $\Phi_H : \mathbb{R}_+ \times \Omega \times H \to L^q(S)$  is defined by

$$\Phi_H(t,\omega,h) := \Phi(t,\omega)h, \quad t \ge 0, \omega \in \Omega, h \in H,$$

and  $\mathcal{I}_{p,q}$  is given as in (7.5.22) for  $v = v^M$ .

*Proof.* The result follows from Doob's maximal inequality, Lemma 7.5.11, Theorem 7.5.22, and the fact that  $\|\Phi_H \mathbf{1}_{A_n}\|_{\mathscr{I}_{p,q}} \nearrow \|\Phi_H\|_{\mathscr{I}_{p,q}}$  as  $n \to \infty$  by the monotone convergence theorem.

### 7.5.5. Integration with respect to continuous martingales

Finally, let us recall the known sharp bounds for  $L^q$ -valued stochastic integrals with respect to continuous local martingales. These bounds are a special case of the main result in [177].

For  $F : \mathbb{R}_+ \to \mathbb{R}_+$  nondecreasing, we define a measure  $\rho_F$  on  $\mathscr{B}(\mathbb{R}_+)$  by

$$\rho_F((s, t]) = F(t) - F(s), \ 0 \le s < t < \infty.$$

If *X* is a Banach space and  $1 \le p \le \infty$ , then we write  $L^p(\mathbb{R}_+, F; X)$  for the Banach space  $L^p(\mathbb{R}_+, \rho_F; X)$ .

Let  $M : \mathbb{R}_+ \times \Omega \to H$  be a continuous local martingale. Then by Subsection 2.2.1 one can define a continuous predictable process  $[M] : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  and a strongly

progressively measurable  $q_M : \mathbb{R}_+ \times \Omega \to \mathscr{L}(H)$  such that [M] is a quadratic variation of M and  $\int_0^{\cdot} \langle q_M(s)h,h \rangle d[M]_s$  is a quadratic variation of [Mh] for each  $h \in H$ . The following theorem immediately follows from [177].

**Theorem 7.5.25.** Let *H* be a Hilbert space,  $1 < p, q < \infty$ . Let  $M : \mathbb{R}_+ \times \Omega \to H$  be a continuous local martingale,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, L^q(S))$  be elementary predictable. Then

$$\mathbb{E}\left(\sup_{0\leq s\leq t}\|(\Phi\cdot M)_s\|^p\right) \approx_{p,q} \mathbb{E}\|\Phi q_M^{\frac{1}{2}}\mathbf{1}_{[0,t]}\|_{\gamma(L^2(\mathbb{R}_+,[M^c];H),L^q(S))}^p.$$

### 7.5.6. Integration with respect to general local martingales

We can now combine the sharp estimates obtained for the three special type of stochastic integrals to obtain sharp estimates for  $\Phi \cdot M$ , where *M* is an arbitrary local martingale.

**Lemma 7.5.26.** Let *H* be a Hilbert space, *X* be a Banach space space,  $M : \mathbb{R}_+ \times \Omega \to H$  be a local martingale,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X)$  be elementary predictable,  $F : \mathbb{R}_+ \times \Omega \times H \to X$  be elementary  $\widetilde{\mathcal{P}}$ -measurable. Then

- (*i*) *if M is continuous, then*  $\Phi \cdot M$  *is continuous,*
- (ii) if M is purely discontinuous quasi-left continuous, then  $F \star \overline{\mu}^M$  is purely discontinuous quasi-left continuous,
- (iii) if M is purely discontinuous with accessible jumps, then  $\Phi \cdot M$  is purely discontinuous with accessible jumps.

*Proof.* Since  $\Phi$  is elementary predictable, *X* can be assumed to be finite-dimensional.

(i) holds since if *M* is continuous, then the formula (2.5.2) defines an a.s. continuous process.

To prove pure discontinuity in (ii) one has to endow *X* with a Euclidean norm and notice that if *M* is purely discontinuous quasi-left continuous then by [85, Proposition II.1.28]  $[F \star \bar{\mu}^M]_t = \sum_{0 \le s \le t} ||F(\Delta M)||^2$  a.s. for all  $t \ge 0$  since  $F \star v^M$  is absolutely continuous, so it does not effect on the quadratic variation. Therefore  $[F \star \bar{\mu}^M]$  is purely discontinuous, and so  $F \star \bar{\mu}^M$  is purely discontinuous by [89, Theorem 26.14]. Quasi-left continuity then follows as  $\Delta (F \star \bar{\mu}^M)_{\tau} = F(\Delta M_{\tau}) = 0$  a.s. for any predictable stopping time  $\tau$ .

Pure discontinuity of  $\Phi \cdot M$  in (iii) follows from the same argument as in (ii), and the rest can be proven using the fact that a.s.

$$\{t \in \mathbb{R}_+ : \Delta(\Phi \cdot M)_t \neq 0\} \subset \{t \in \mathbb{R}_+ : \Delta M_t \neq 0\}.$$

The following observation is fundamental for the duality argument used to prove the lower bounds in Theorem 7.5.29.

**Lemma 7.5.27.** Let *H* be a Hilbert space, *X* be a Banach space,  $M^c, M^q : \mathbb{R}_+ \times \Omega \to H$ be continuous and purely discontinuous quasi-left continuous martingales,  $M^{a,1} : \mathbb{R}_+ \times \Omega \to X$ ,  $M^{a,2} : \mathbb{R}_+ \times \Omega \to X^*$  be purely discontinuous martingales with accessible jumps,  $\Phi_1 : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X), \Phi_2 : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X^*)$  be elementary predictable,  $F_1 : \mathbb{R}_+ \times \Omega \times H \to X$ ,  $F_2 : \mathbb{R}_+ \times \Omega \times H \to X^*$  be elementary  $\widetilde{\mathscr{P}}$ -measurable. Assume that  $(\Phi_1 \cdot M^c)_{\infty}$ ,  $(F_1 \star \overline{\mu}^{M^q})_{\infty}, M^{a,1}_{\infty} \in L^p(\Omega; X)$  and  $(\Phi_2 \cdot M^c)_{\infty}, (F_2 \star \overline{\mu}^{M^q})_{\infty}, M^{a,2}_{\infty} \in L^{p'}(\Omega; X^*)$  for some  $1 . Then, for all <math>t \ge 0$ ,

$$\mathbb{E}\langle (\Phi_1 \cdot M^c + F_1 \star \bar{\mu}^{M^q} + M^{a,1})_t, (\Phi_2 \cdot M^c + F_2 \star \bar{\mu}^{M^q} + M^{a,2})_t \rangle = \mathbb{E}\langle (\Phi_1 \cdot M^c)_t, (\Phi_2 \cdot M^c)_t \rangle + \mathbb{E}\langle (F_1 \star \bar{\mu}^{M^q})_t, (F_2 \star \bar{\mu}^{M^q})_t \rangle + \mathbb{E}\langle M_t^{a,1}, M_t^{a,2} \rangle.$$
(7.5.29)

**Lemma 7.5.28.** Let X be a Banach space,  $X_0 \subset X$  be a finite-dimensional subspace,  $1 , <math>M^q : \mathbb{R}_+ \times \Omega \to X_0$  be a purely discontinuous quasi-left continuous  $L^p$ -martingale,  $M_0^q = 0$ ,  $M^a : \mathbb{R}_+ \times \Omega \to X^*$  be a purely discontinuous  $L^{p'}$ -martingale with accessible jumps. Then  $\mathbb{E}\langle M_t^q, M_t^a \rangle = 0$  for each  $t \ge 0$ .

*Proof.* Let *d* be the dimension of  $X_0$ ,  $x_1, ..., x_d$  be a basis of  $X_0$ . Then there exist purely discontinuous quasi-left continuous  $L^p$ -martingales  $M^{q,1}, ..., M^{q,d} : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  such that  $M^q = M^{q,1}x_1 + \cdots + M^{q,d}x_d$ . Thus for any i = 1, ..., d and any purely discontinuous  $L^{p'}$ -martingale  $N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  with accessible jumps  $[M^{q,i}, N] = 0$  a.s. by [89, Corollary 26.16]. Hence [85, Proposition I.4.50(a)] implies that  $M^{q,i}N$  is a local martingale, and due to integrability it is a martingale. Notice also that all  $M^{q,i}$  start at zero, therefore

$$\mathbb{E}\langle M_t^q, M_t^a \rangle = \sum_{i=1}^d \mathbb{E}M_t^{q,i} \langle x_i, M_t^a \rangle = \sum_{i=1}^d \mathbb{E}M_0^{q,i} \langle x_i, M_0^a \rangle = 0.$$

*Proof of Lemma* 7.5.27. Since all the integrands  $\Phi_1$ ,  $\Phi_2$ ,  $F_1$ ,  $F_2$  are elementary, one can suppose that *X* and *X*<sup>\*</sup> are finite dimensional, so we can endow these spaces with Euclidean norms. Since by Lemma 7.5.26  $\Phi_1 \cdot M^c$  and  $\Phi_2 \cdot M^c$  are continuous,  $F_1 \star \bar{\mu}^{M^q}$ ,  $F_1 \star \bar{\mu}^{M^q}$ ,  $M^{a,1}$  and  $M^{a,2}$  are purely discontinuous, then [85, Definition I.4.11] implies that for each  $t \ge 0$ 

$$\begin{split} \mathbb{E} \langle (\Phi_1 \cdot M^c)_t, (F_2 \star \bar{\mu}^{M^q})_t \rangle &= \mathbb{E} [\Phi_1 \cdot M^c, F_2 \star \bar{\mu}^{M^q}]_t = 0, \\ \mathbb{E} \langle (\Phi_2 \cdot M^c)_t, (F_1 \star \bar{\mu}^{M^q})_t \rangle &= \mathbb{E} [\Phi_2 \cdot M^c, F_1 \star \bar{\mu}^{M^q}]_t = 0, \\ \mathbb{E} \langle (\Phi_1 \cdot M^c)_t, M_t^{a,2} \rangle &= \mathbb{E} [\Phi_1 \cdot M^c, M^{a,2}]_t = 0, \\ \mathbb{E} \langle (\Phi_2 \cdot M^c)_t, M_t^{a,1} \rangle &= \mathbb{E} [\Phi_2 \cdot M^c, M^{a,1}]_t = 0. \end{split}$$

Moreover, thanks to Lemma 7.5.26 and Lemma 7.5.28

$$\mathbb{E}\langle M_t^{a,1}, (F_2 \star \bar{\mu}^{M^q})_t \rangle = \mathbb{E}\langle M_t^{a,2}, (F_1 \star \bar{\mu}^{M^q})_t \rangle = 0,$$

so (7.5.29) easily follows.

**Theorem 7.5.29.** Let *H* be a Hilbert space,  $1 < p, q < \infty$ . Let  $M^c$ ,  $M^q : \mathbb{R}_+ \times \Omega \to H$ be continuous and purely discontinuous quasi-left continuous local martingales,  $M^a :$  $\mathbb{R}_+ \times \Omega \to L^q(S)$  be a purely discontinuous  $L^p$ -martingale with accessible jumps,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}^q(S)$  be elementary predictable,  $F : \mathbb{R}_+ \times \Omega \times H \to L^q(S)$  be elementary  $\widetilde{\mathscr{P}}$ measurable. If  $\Phi \cdot M^c$  and  $F \star \mu^{M^q}$  are  $L^p$ -martingales, then

$$\left( \mathbb{E} \sup_{0 \le s \le t} \left\| \left( \Phi \cdot M^{c} + F \star \bar{\mu}^{M^{q}} + M^{a} \right)_{\infty} \right\|_{L^{q}(S)}^{p} \right)^{\frac{1}{p}}$$

$$= \sum_{p,q} \left( \mathbb{E} \| \Phi q_{M^{c}}^{\frac{1}{2}} \|_{\gamma(L^{2}(\mathbb{R}_{+}, [M^{c}]; H), X)}^{p} \right)^{\frac{1}{p}} + \| F \|_{\mathscr{I}_{p,q}} + \| M^{a} \|_{\mathscr{A}_{p,q}},$$

$$(7.5.30)$$

where  $\mathscr{I}_{p,q}$  is given as in (7.5.22) for  $v = v^{M^q}$ ,  $\mathscr{A}_{p,q}$  is given as in (7.5.7).

*Proof.* The estimate  $\leq_{p,q}$  follows from the triangle inequality and Theorems 7.5.25, 7.5.22 and 7.5.8. Let us now prove  $\geq_{p,q}$  via duality. Without loss of generality due to the proof of Theorem 7.5.8 and due to Lemma 7.5.7 we can assume that there exists  $N \geq 1$  and a sequence of predictable stopping times  $\mathcal{T} = (\tau_n)_{n=0}^N$  such that M has a.s. at most N jumps and a.s.  $\{t \in \mathbb{R}_+ : \Delta M_t \neq 0\} \subset \{\tau_0, ..., \tau_N\}$ . Define the Banach space

$$X := L^{p}(\Omega; \gamma(L^{2}(\mathbb{R}_{+}, [M^{c}]; H), L^{q}(S))) \times \mathscr{I}_{p,q} \times \mathscr{A}_{p,q}^{\mathcal{T}}$$

and let *Y* be the closure of the lineare subspace  $\cup_{(\Phi,F,M^a)\in X} (\Phi \cdot M^c + F \star \bar{\mu}^{M^d} + M^a)_{\infty}$  in  $L^p(\Omega; L^q(S))$ . Then by [79, Proposition 1.3.3], the Trace duality (2.9.2), Corollary 7.A.8 and the duality statement in Theorem 7.5.5

$$X^* = L^{p'}(\Omega; \gamma(L^2(\mathbb{R}_+, [M^c]; H), L^{q'}(S))) \times \mathscr{I}_{p',q'} \times \mathscr{A}_{p',q'}^{\mathscr{T}}$$

By the upper bounds in (7.5.30), the maps  $j : X \to Y$  and  $k : X^* \to Y^*$  defined via  $(\Phi, F, M^a) \mapsto (\Phi \cdot M^c + F \star \overline{\mu}^{M^d} + M^a)_{\infty}$  are both continuous linear mappings. Let  $x = (\Phi_1, F_1, M_1^a) \in X$ ,  $x^* = (\Phi_2, F_2, M_2^a) \in X^*$  be such that  $\Phi_1$  and  $\Phi_2$  are elementary predictable, and  $F_1$  and  $F_2$  are elementary  $\widetilde{\mathscr{P}}$ -measurable. Then  $\langle \tilde{x}^*, \tilde{x} \rangle = \langle k(\tilde{x}^*), j(\tilde{x}) \rangle$  by Lemma 7.5.27 and (7.5.5) and so Lemma 7.2.1 yields  $\gtrsim_{p,q}$  in (7.5.30).

**Theorem 7.5.30.** Let H be a Hilbert space,  $1 < p, q < \infty$ . Let  $M : \mathbb{R}_+ \times \Omega \to H$  be a local martingale,  $M^c, M^q, M^d : \mathbb{R}_+ \times \Omega \to H$  be local martingales such that  $M_0^c = M_0^q = 0$ ,  $M^c$  is continuous,  $M^q$  is purely discontinuous quasi-left continuous,  $M^a$  is purely discontinuous with accessible jumps,  $M = M^c + M^q + M^a$ . Let  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, L^q(S))$  be elementary predictable. Then,

$$\left( \mathbb{E} \sup_{0 \le s \le t} \| (\Phi \cdot M)_s \|_{L^q(S)}^p \right)^{\frac{1}{p}}$$
  
$$\approx_{p,q} \left( \mathbb{E} \| \Phi q_{M^c}^{\frac{1}{2}} \mathbf{1}_{[0,t]} \|_{\gamma(L^2(\mathbb{R}_+, [M^c]; H), X)}^p \right)^{\frac{1}{p}}$$
  
$$+ \| \Phi_H \mathbf{1}_{[0,t]} \|_{\mathscr{I}_{p,q}} + \| (\Phi \mathbf{1}_{[0,t]}) \cdot M^a \|_{\mathscr{A}_{p,q}}, \quad (7.5.31)$$

where  $\Phi_H : \mathbb{R}_+ \times \Omega \times H \to L^q(S)$  is defined by

$$\Phi_H(t,\omega,h) := \Phi(t,\omega)h, \quad t \ge 0, \omega \in \Omega, h \in H,$$

 $\mathscr{I}_{p,q}$  is given as in (7.5.22) for  $v = v^{M^q}$ , and  $\mathscr{A}_{p,q}$  is as defined in (7.5.7).

*Proof.* First of all notice that  $\Phi \cdot M$  is an  $L^q(S)$ -valued local martingale, so by Doob's maximal inequality

$$\mathbb{E}\sup_{0\le s\le t} \|(\Phi\cdot M)_s\|_{L^q(S)}^p = \mathbb{E}\|(\Phi\cdot M)_t\|_{L^q(S)}^p.$$
(7.5.32)

Since  $\Phi$  is elementary predictable, we can assume that *X* is finite dimensional. Consequently, (7.5.31) holds by (7.5.32), Lemma 7.5.26 and Theorem 7.5.29.

*Remark* 7.5.31. Let  $M = (M_n)_{n\geq 0}$  be a discrete  $L^q$ -valued martingale. Then due to the *Strong Doob maximal inequality* (also known as the *Fefferman-Stein inequality*), presented e.g. in [79, Theorem 3.2.7] and [3, Theorem 2.6],

$$\left(\mathbb{E}\left(\int_{S} |\sup_{n\geq 0} M_{n}(s)|^{q} \,\mathrm{d}s\right)^{\frac{p}{q}}\right)^{\frac{1}{p}} \sim_{p,q} \left(\mathbb{E}\sup_{n\geq 0} \|M_{n}\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}}.$$

As a consequence, for any continuous time martingale  $M : \mathbb{R}_+ \times \Omega \to L^q(S)$ 

$$\left(\mathbb{E}\|\sup_{t\geq 0} M_t\|_{L^q(S)}^p\right)^{\frac{1}{p}} \sim_{p,q} \left(\mathbb{E}\sup_{t\geq 0} \|M_t\|_{L^q(S)}^p\right)^{\frac{1}{p}}.$$

Indeed, this follows by the existence of a pointwise càdlàg version of M and by approximating M by a discrete-time martingale. Thus, all the sharp bounds for stochastic integrals proved in this section, in particular Theorems 7.5.8, 7.5.22, 7.5.24, 7.5.25, and, finally, Theorems 7.5.29 and 7.5.30, remain valid if we move the supremum over time inside the  $L^q$ -norm.

## **7.A.** DUALS OF $\mathscr{S}_{q}^{p}$ , $\mathscr{D}_{q,q}^{p}$ , $\mathscr{D}_{p,q}^{p}$ , $\hat{\mathscr{S}}_{q}^{p}$ , $\hat{\mathscr{D}}_{q,q}^{p}$ , and $\hat{\mathscr{D}}_{p,q}^{p}$

In this section we will find the duals of  $\mathscr{S}_q^p$ ,  $\mathscr{D}_{q,q}^p$ ,  $\mathscr{D}_{p,q}^p$ ,  $\hat{\mathscr{D}}_q^p$ ,  $\hat{\mathscr{D}}_{q,q}^p$ ,  $\hat{\mathscr{D}}_{q,q}^p$ , and  $\hat{\mathscr{D}}_{p,q}^p$  for all  $1 < p, q < \infty$ . As a consequence, we show the duality for the space  $\mathscr{I}_{p,q}$  that was used to prove the lower bounds in Theorem 7.5.22.

## **7.A.1.** $\mathcal{D}_{q,q}^p$ and $\mathcal{D}_{p,q}^p$ spaces

Let *X* be a Banach space and consider any random measure v on  $\mathbb{R}_+ \times J$ . In sequel we will assume that  $\int_{\mathbb{R}_+ \times J} \mathbf{1}_A dv$  is an  $\mathbb{R}_+$ -valued random variable for each  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{J}$ -measurable  $A \subset \mathbb{R}_+ \times J$ . Notice that this condition always holds for any optional random measure v. Indeed, without loss of generality we may assume that there exist  $A_{\mathbb{R}_+} \in \mathscr{B}(\mathbb{R}_+)$  and  $A_J \in \mathscr{J}$  such that  $A = A_{\mathbb{R}_+} \times A_J$ . Let  $\widetilde{A} = A \times \Omega$ . Then  $\widetilde{A} \in \widetilde{\mathcal{O}}$  (since  $A_{\mathbb{R}_+} \times \Omega \in \mathcal{O}$ ), therefore  $\mathbf{1}_{\widetilde{A}} \star v$  is an optional process, and

$$\int_{\mathbb{R}_+\times J} \mathbf{1}_A \,\mathrm{d}\nu = \lim_{t\to\infty} (\mathbf{1}_{\widetilde{A}} \star \nu)_t$$

is an  $\overline{\mathbb{R}}_+$ -valued  $\mathscr{F}$ -measurable function as a monotone limit of  $\overline{\mathbb{R}}_+$ -valued  $\mathscr{F}$ -measurable functions.

We define  $\mathscr{D}_q^p(X)$  to be the space of all  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$ -strongly measurable functions  $f : \mathbb{R}_+ \times \Omega \times J \to X$  such that

$$\|f\|_{\mathscr{D}^p_q(X)} := \left(\mathbb{E}\left(\int_{\mathbb{R}_+\times J} \|f\|_X^q \,\mathrm{d}v\right)^{\frac{p}{q}}\right)^{\frac{1}{p}} < \infty.$$

Recall that the measure  $\mathbb{P} \otimes v$  on  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$  is defined by setting

$$\mathbb{P} \otimes v \Big( \bigcup_{i=1}^n A_i \times B_i \Big) := \sum_{i=1}^n \mathbb{E}(\mathbf{1}_{A_i} v(B_i)),$$

for disjoint  $A_i \in \mathscr{F}$  and disjoint  $B_i \in \mathscr{B}(\mathbb{R}_+) \otimes \mathscr{J}$  and extending  $\mathbb{P} \times v$  to  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$  via the Carathéodory extension theorem.

The following result is well-known if v is a deterministic measure. The argument for random measures is similar and provided for the reader's convenience.

**Theorem 7.A.1.** Let  $1 < p, q < \infty$ , X be reflexive. Then  $(\mathcal{D}_q^p(X))^* = \mathcal{D}_{q'}^{p'}(X^*)$ . Moreover

$$\|\phi\|_{\mathcal{D}_{q'}^{p'}(X^*)} = \|\phi\|_{(\mathcal{D}_{q}^{p}(X))^*}, \ \phi \in \mathcal{D}_{q'}^{p'}(X^*).$$
(7.A.1)

*Proof.* First we suppose that  $\mathbb{E}v(\mathbb{R}_+ \times J) < \infty$ . By approximation we can assume that  $v(\mathbb{R}_+ \times J) \leq N$  a.s., for some  $N \in \mathbb{N}$ . In this case we can proceed with a standard argument using the Radon-Nikodym property of  $X^*$ . Let  $F \in (\mathcal{D}_q^p(X))^*$ . On  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{F}$  we can define and  $X^*$ -valued measure  $\theta$  by setting

$$\langle \theta(A), x \rangle := F(\mathbf{1}_A \cdot x) \qquad (\mathscr{B}(\mathbb{R}_+) \otimes A \in \mathscr{F} \otimes \mathscr{J}, x \in X).$$

It is straightforward to verify that  $\theta$  is  $\sigma$ -additive and absolutely continuous with respect to  $\mathbb{P} \times v$ . Moreover,  $\theta$  is of finite variation. Indeed, if  $A_1, \ldots, A_n$  is a disjoint partition of  $\mathbb{R}_+ \times \Omega \times J$ , then

$$\sum_{i=1}^{n} \|\theta(A_{i})\| = \sup_{(x_{i})_{i=1}^{n} \subset B_{X}} \sum_{i=1}^{n} F(\mathbf{1}_{A_{i}} x_{i})$$

$$= \sup_{(x_{i})_{i=1}^{n} \subset B_{X}} F\left(\sum_{i=1}^{n} \mathbf{1}_{A_{i}} x_{i}\right)$$

$$\leq \|F\|_{(\mathscr{D}_{q}^{p}(X))^{*}} \sup_{(x_{i})_{i=1}^{n} \subset B_{X}} \left(\mathbb{E}\left(\int_{\mathbb{R}_{+} \times J} \left\|\sum_{i=1}^{n} \mathbf{1}_{A_{i}} x_{i}\right\|_{X}^{q} dv\right)^{p/q}\right)^{1/p}$$

$$= \|F\|_{(\mathscr{D}_{q}^{p}(X))^{*}} \sup_{(x_{i})_{i=1}^{n} \subset B_{X}} \left(\mathbb{E}\left(\int_{\mathbb{R}_{+} \times J} \sum_{i=1}^{n} \mathbf{1}_{A_{i}} \|x_{i}\|_{X}^{q} dv\right)^{p/q}\right)^{1/p}$$

$$\leq \|F\|_{(\mathscr{D}_{q}^{p}(X))^{*}} (\mathbb{E}\nu(\mathbb{R}_{+} \times J)^{p/q})^{1/p}.$$
(7.A.2)

(Here  $B_X$  is a unit ball in X). By the Radon-Nikodym property of  $X^*$ , there exists an  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$ -strongly measurable  $X^*$ -valued function f such that

$$F(g) = F_f(g) = \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle f, g \rangle \, \mathrm{d}\nu$$

for each  $g \in \mathcal{D}_q^p(X)$ . By Hölder's inequality, it is immediate that

$$\|F\|_{(\mathscr{D}^p_q(X))^*} \le \|f\|_{\mathscr{D}^{p'}_{q'}(X^*)}$$

To show the reverse estimate, we may assume that  $f \in \mathscr{D}_{q'}^{p'}(X^*)$  has norm 1 and show that  $||F_f||_{(\mathscr{D}_q^p(X))^*} \ge 1$ . By approximation, we may furthermore assume that f is simple, i.e.,

$$f = \sum_{m,n} \mathbf{1}_{A_n} \mathbf{1}_{B_{nm}} x_{nm}^*,$$

for  $A_n \in \mathscr{F}$  and  $B_{nm} \in \mathscr{B}(\mathbb{R}_+) \otimes \mathscr{J}$  disjoint and  $x_{nm}^* \in X^*$ . Define

$$g = \sum_{m,n} \mathbf{1}_{A_n} \Big( \sum_m \nu(B_{nm}) \| x_{nm}^* \|_{X^*}^{q'} \Big)^{\frac{p'}{q'} - 1} \mathbf{1}_{B_{nm}} x_{nm} \| x_{nm}^* \|_{X^*}^{q' - 1},$$
(7.A.3)

where the  $x_{nm} \in X$  satisfy the condition in Lemma 7.4.2, i.e. for some  $0 < \varepsilon < 1$ 

$$(1-\varepsilon) \|x_{nm}^*\| \le \langle x_{nm}, x_{nm^*} \rangle, \qquad \|x_{nm}\|_X = 1.$$

By assumption,

$$\|f\|_{D_{q'}^{p'}(X^*)}^{p'} = \sum_{n} \mathbb{P}(A_n) \Big( \sum_{m} \nu(B_{nm}) \|x_{nm}^*\|_{X^*}^{q'} \Big)^{p'/q'} = 1.$$

Therefore, also

$$\begin{split} \|g\|_{D_{q}^{p}(X)}^{q} &= \sum_{n} \mathbb{P}(A_{n}) \Big( \sum_{m} \nu(B_{nm}) \|x_{nm}^{*}\|_{X^{*}}^{q'} \Big)^{\left(\frac{p'}{q'}-1\right)p} \Big( \sum_{m} \nu(B_{nm}) \|x_{nm}^{*}\|_{X^{*}}^{(q'-1)q} \Big)^{\frac{p}{q}} \\ &= \sum_{n} \mathbb{P}(A_{n}) \Big( \sum_{m} \nu(B_{nm}) \|x_{nm}^{*}\|^{q'} \Big)^{p'/q'} = 1, \end{split}$$

as

$$qq' = q + q'$$
  $\frac{pp'}{q'} - p + \frac{p}{q} = \frac{p'}{q'}.$  (7.A.4)

Moreover,

$$\begin{split} F_{f}(g) &= \mathbb{E} \int \langle f, g \rangle \ d\nu \\ &= \sum_{n} \mathbb{P}(A_{n}) \sum_{m} \nu(B_{nm}) \langle x_{nm}, x_{nm^{*}} \rangle \|x_{nm}^{*}\|_{X^{*}}^{q'-1} \Big( \sum_{m} \nu(B_{nm}) \|x_{nm}^{*}\|_{X^{*}}^{q'} \Big)^{\frac{p'}{q'}-1} \\ &\geq \sum_{n} \mathbb{P}(A_{n}) \sum_{m} \nu(B_{nm}) (1-\varepsilon) \|x_{nm}^{*}\|_{X^{*}}^{q'} \Big( \sum_{m} \nu(B_{nm}) \|x_{nm}^{*}\|_{X^{*}}^{q'} \Big)^{\frac{p'}{q'}-1} \\ &= (1-\varepsilon) \sum_{n} \mathbb{P}(A_{n}) \Big( \sum_{m} \nu(B_{nm}) \|x_{nm}^{*}\|_{X^{*}}^{q'} \Big)^{\frac{p'}{q'}} = (1-\varepsilon). \end{split}$$

Since  $\varepsilon$  was arbitrary,  $||F_f||_{(\mathcal{D}^p_a(X))^*} \ge 1$ .

Let now  $\mathbb{E}v(\mathbb{R}_+ \times J) = \infty$  and assume that  $\mathbb{P} \times v$  is  $\sigma$ -finite. Then there exists a sequence  $(A_n)_{n\geq 1} \subset \mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$  such that  $A_n \subset A_{n+1}$  for each  $n \geq 1$ ,  $\bigcup_{n\geq 1} A_n = \mathbb{R}_+ \times \Omega \times J$ , and  $\mathbb{P} \otimes v(A_n) < \infty$  for each  $n \geq 1$ . Let  $v_n := v \cdot \mathbf{1}_{A_n}$ . Then each  $F \in (\mathscr{D}_q^p(X))^*$  can be considered as a linear functional on the closed subspace of  $\mathscr{D}_q^p(X)$  consisting of all functions with support in  $A_n$ . By the previous part of the proof, for each  $n \geq 1$  there exists  $f_n \in \mathscr{D}_{q'}^{p'}(X^*)$  with support in  $A_n$  such that

$$F(g \cdot \mathbf{1}_{A_n}) = F_{f_n}(g \cdot \mathbf{1}_{A_n}) = \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle f_n, g \rangle \mathbf{1}_{A_n} \, \mathrm{d}v$$

and

$$\|f_n\|_{\mathcal{D}_{q'}^{p'}(X^*)} \le \|F_{f_n}\|_{(\mathcal{D}_{q}^p(X))^*} \le \|F\|_{(\mathcal{D}_{q}^p(X))^*}.$$

Obviously  $f_{n+1}\mathbf{1}_{A_n} = f_n$  for each  $n \ge 1$ , hence there exists  $f : \Omega \times \mathbb{R}_+ \times J \to X^*$  such that  $f\mathbf{1}_{A_n} = f_n$  for each  $n \ge 1$ . But then Fatou's lemma implies

$$\|f\|_{\mathscr{D}_{q'}^{p'}(X^*)} \le \liminf_{n \to \infty} \|f_n\|_{\mathscr{D}_{q'}^{p'}(X^*)} \le \|F\|_{(\mathscr{D}_{q}^{p}(X))^*},$$

so  $f \in \mathcal{D}_{q'}^{p'}(X^*)$ . On the other hand, by Hölder's inequality

$$\|F\|_{(\mathscr{D}^{p}_{q}(X))^{*}} \leq \|f\|_{\mathscr{D}^{p'}_{q'}(X^{*})}.$$

Since the bounded linear functionals *F* and *F*<sub>f</sub> agree on a dense subset of  $\mathscr{D}_q^p(X)$ , it follows that *F* = *F*<sub>f</sub> and (7.A.1) holds.

Finally, let v be general. Let  $F \in (\mathcal{D}_q^p(X))^*$  be of norm 1. Let  $\varepsilon_n \downarrow 0$  and let  $(g_n)_{n \ge 1}$ be a sequence in the unit sphere of  $\mathcal{D}_q^p(X)$  satisfying  $F(g_n) \ge (1 - \varepsilon_n)$ . By strong measurability of the  $g_n$ , there exists an  $A \in \mathcal{B}(\mathbb{R}_+) \otimes \mathcal{F} \otimes \mathcal{J}$  so that  $\mathbb{P} \times v$  is  $\sigma$ -finite on A and  $g_n = 0$  on  $A^c \mathbb{P} \times v$ -a.e. Let  $\tilde{F} \in (\mathcal{D}_q^p(X))^*$  be defined by  $\tilde{F}(g) = F(g\mathbf{1}_A)$ . The previous part of the proof shows that there exists an  $f \in \mathcal{D}_{q'}^{p'}(X^*)$  so that  $\tilde{F} = F_f$ and  $\|\tilde{F}\|_{(\mathcal{D}_q^p(X))^*} = \|f\|_{\mathcal{D}_{q'}^{p'}(X^*)}$ . It remains to show that  $F = \tilde{F}$ . To prove this, suppose that there exists a  $g_0 \in \mathcal{D}_q^p(X)$  of norm 1 with  $\operatorname{supp}(g_0) \subset A^c$  and  $F(g_0) = \delta > 0$ . Let  $0 < \lambda < 1$ . Then, for any  $n \ge 1$ ,

$$\|(1-\lambda^{p})^{1/p}g_{0}+\lambda g_{n}\|_{\mathscr{D}_{q}^{p}(X)}^{p}=(1-\lambda^{p})\|g_{0}\|_{\mathscr{D}_{q}^{p}(X)}^{p}+\lambda^{p}\|g_{n}\|_{\mathscr{D}_{q}^{p}(X)}^{p}=1$$

and

$$F((1-\lambda^p)^{1/p}g_0+\lambda g_n) \ge (1-\lambda^p)^{1/p}\delta + \lambda(1-\varepsilon_n).$$

As a consequence,

$$||F|| \ge \sup_{0 < \lambda < 1} (1 - \lambda^p)^{1/p} \delta + \lambda.$$

One easily checks that the supremum is attained in

$$\lambda = \left(1 + \delta^{1/(1 - \frac{1}{p})}\right)^{-1/p}$$

and so ||F|| > 1, a contradiction.

We now turn to proving a similar duality statement for  $\hat{\mathscr{D}}_q^p(X)$ , the space of all  $\widetilde{\mathscr{P}}$ -measurable functions in  $\mathscr{D}_q^p(X)$ . In the proof we will use the following 'reverse' version of the dual Doob inequality [52, Lemma 2.10].

**Lemma 7.A.2** (Reverse dual Doob inequality). Fix  $0 . Let <math>\mathbb{F} = (\mathscr{F}_n)_{n\ge 0}$  be a filtration and let  $(\mathbb{E}_n)_{n\ge 0}$  be the associated sequence of conditional expectations. If  $(f_n)_{n\ge 0}$  is a sequence of non-negative random variables in  $L^1(\mathbb{P})$ , then

$$\left(\mathbb{E}\Big|\sum_{n\geq 0}f_n\Big|^p\right)^{\frac{1}{p}} \leq p^{-1}\left(\mathbb{E}\Big|\sum_{n\geq 0}\mathbb{E}_n f_n\Big|^p\right)^{\frac{1}{p}}.$$

**Theorem 7.A.3.** Let X be a reflexive space and let v be a predictable,  $\widetilde{\mathcal{P}}$ - $\sigma$ -finite random measure on  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{J}$  that is non-atomic in time. Then, for  $1 < p, q < \infty$ ,

$$(L^{p}_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^{q}(\nu; X)))^{*} = L^{p'}_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^{q'}(\nu; X^{*}))$$

with isomorphism given by

$$g \mapsto F_g, \qquad F_g(h) = \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle g, h \rangle d\nu \qquad (g \in L^{p'}_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^{q'}(\nu)), h \in L^p_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^q(\nu))).$$

Moreover,

$$\min\left\{\left(\frac{p}{q}\right)^{1/q}\frac{q'}{p'}, \left(\frac{p'}{q'}\right)^{1/q'}\frac{q}{p}\right\}\|g\|_{L^{p'}_{\widetilde{\mathscr{D}}}(\mathbb{P};L^{q'}(\nu;X^*))} \le \|F_g\| \le \|g\|_{L^{p'}_{\widetilde{\mathscr{D}}}(\mathbb{P};L^{q'}(\nu;X^*))}.$$
(7.A.5)

*Proof. Step 1: reduction.* It suffices to prove the result for  $p \le q$ . Indeed, once this is known we can deduce the case  $q \le p$  as follows. Observe that  $L^{p'}_{\widetilde{\mathscr{P}}}(\mathbb{P}; L^{q'}(v; X^*))$  is a closed subspace of  $\mathscr{D}^{p'}_{q'}(X^*) = L^{p'}(\mathbb{P}; L^{q'}(v; X^*))$ . By Theorem 7.A.1,  $\mathscr{D}^{p'}_{q'}(X^*)$  is reflexive and therefore  $L^{p'}_{\widetilde{\mathscr{P}}}(\mathbb{P}; L^{q'}(v; X^*))$  is reflexive as well. Therefore, as  $p' \le q'$ ,

$$(L^{p}_{\widetilde{\mathscr{P}}}(\mathbb{P};L^{q}(\nu;X)))^{*} = L^{p'}_{\widetilde{\mathscr{P}}}(\mathbb{P};L^{q'}(\nu;X^{*}))^{**} = L^{p'}_{\widetilde{\mathscr{P}}}(\mathbb{P};L^{q'}(\nu;X^{*})).$$

Hence, if  $F \in (L^p_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^q(v; X)))^*$ , then there exists an  $f \in L^{p'}_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^{q'}(v; X^*))$  so that for any  $g \in L^p_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^q(v; X))$ 

$$F(g) = F_g(f) = \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle f, g \rangle dv.$$

Moreover, the bounds (7.A.5) follow from Lemma 7.4.4. Thus, for the remainder of the proof, we can assume that  $p \le q$ .

Step 2: norm estimates. Let us now show that (7.A.5) holds. Since the upper bound is immediate from Hölder's inequality, we only need to show that for any  $g \in L^{p'}_{\overline{\alpha}}(\mathbb{P}; L^{q'}(\nu; X^*))$ ,

$$\|F_g\| \ge \left(\frac{p}{q}\right)^{1/q} \frac{q'}{p'} \|g\|_{L^{p'}(\mathbb{P}; L^{q'}(\nu; X^*))}.$$
(7.A.6)

It suffices to show this on a dense subset of  $L^{p'}_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^{q'}(v; X^*))$ . Indeed, suppose that  $g_n \to g$  in  $L^{p'}_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^{q'}(v; X^*))$  and that (7.A.6) holds for  $g_n$ , for all  $n \ge 1$ . Then,

$$\left(\frac{p}{q}\right)^{1/q} \frac{q'}{p'} \|g_n\|_{L^{p'}(\mathbb{P}; L^{q'}(v; X^*))} \le \|F_{g_n}\| \le \|F_g\| + \|g - g_n\|_{L^{p'}(\mathbb{P}; L^{q'}(v; X^*))}$$

and by taking limits on both sides we see that g also satisfies (7.A.6).

Let us first assume that

$$v((s,t] \times J) \le (t-s) \qquad \text{a.s.}, \qquad \text{for all } 0 \le s \le t \qquad (7.A.7)$$

By the previous discussion, we may assume that  $||g||_{L^{p'}(\mathbb{P};L^{q'}(v;X^*))} = 1$  and that g is of the form

$$g = \sum_{n=0}^{N_{m*}} \sum_{\ell=0}^{L} \mathbf{1}_{(n/2^{m*},(n+1)/2^{m*}]} \mathbf{1}_{B_{\ell}} g_{n\ell},$$

where  $N_{m_*} < \infty$ ,  $g_{n\ell}$  is simple and  $\mathscr{F}_{n/2^{m_*}}$ -measurable for all n and  $\ell$ , and the  $B_\ell$  are disjoint sets in  $\mathscr{J}$  of finite  $\mathbb{P} \otimes v$ -measure. For  $m \ge m_*$  define

$$\mathbf{g}^{(m)} = \sum_{n=0}^{N_m} \sum_{\ell=0}^{L} \mathbf{1}_{(n/2^m, (n+1)/2^m]} \mathbf{1}_{B_\ell} \mathbf{g}_{n\ell}^{(m)}$$

so that  $g^{(m)} = g$ . Then clearly,  $g_{n\ell}^{(m)}$  is  $\mathscr{F}_{n/2^m}$ -measurable for all n and  $\ell$ . Let us now fix an  $m \ge m_*$ . We define, for any  $0 \le k \le N_m$ ,

$$\bar{s}_{q'}^k(g) = \left(\sum_{n=0}^k \sum_{\ell=0}^L \|g_{n\ell}^{(m)}\|^{q'} \mathbb{E}_{n/2^m} v((n/2^m, (n+1)/2^m] \times B_\ell)\right)^{1/q}$$

and set

$$\alpha = (\mathbb{E}\bar{s}_{q'}^{N_m}(g^{(m)})^{p'})^{1/p'}$$

Let  $P_{\varepsilon}$  be as in Lemma 7.4.2. We define a  $\widetilde{\mathcal{P}}$ -measurable function *h* by

$$h = \sum_{n=0}^{N_m} \sum_{\ell=0}^{L} \mathbf{1}_{(n/2^m,(n+1)/2^m]} \mathbf{1}_{B_\ell} h_{n\ell}$$

where, for  $0 \le n \le N_m$  and  $0 \le \ell \le L$ ,  $h_{n\ell}$  is the  $\mathscr{F}_{n/2^m}$ -measurable function

$$h_{n\ell} = \frac{1}{\alpha^{p'-1}} (\bar{s}_{q'}^n(g^{(m)}))^{p'-q'} \|g_{n\ell}^{(m)}\|^{q'-1} P_{\varepsilon} g_{n\ell}^{(m)}.$$

Since  $p/q \le 1$ , Lemma 7.A.2 implies

$$\begin{split} \|h\|_{L^{p}(\mathbb{P};L^{q}(\nu))} &= \left(\mathbb{E}\Big(\sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \|h_{n\ell}\|^{q} \nu((n/2^{m},(n+1)/2^{m}] \times B_{\ell})\Big)^{p/q}\Big)^{1/p} \\ &\leq \left(\frac{q}{p}\right)^{1/q} \Big(\mathbb{E}\Big(\sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \|h_{n\ell}\|^{q} \mathbb{E}_{n/2^{m}} \nu((n/2^{m},(n+1)/2^{m}] \times B_{\ell})\Big)^{p/q}\Big)^{1/p} \\ &= \left(\frac{q}{p}\right)^{1/q} (\mathbb{E}\bar{s}_{q}^{N_{m}}(h)^{p})^{1/p}. \end{split}$$

Now observe that

$$\begin{split} \bar{s}_{q}^{N_{m}}(h)^{q} &= \sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \|h_{n\ell}\|^{q} \mathbb{E}_{n/2^{m}} v((n/2^{m}, (n+1)/2^{m}] \times B_{\ell}) \\ &\leq \frac{1}{\alpha^{(p'-1)q}} \sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \|g_{n\ell}^{(m)}\|^{(q'-1)q} \bar{s}_{q'}^{n}(g^{(m)})^{(p'-q')q} \mathbb{E}_{n/2^{m}} v((n/2^{m}, (n+1)/2^{m}] \times B_{\ell}) \\ &\leq \frac{1}{\alpha^{(p'-1)q}} \bar{s}_{q'}^{N_{m}}(g^{(m)})^{(p'-q')q} \sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \|g_{n\ell}^{(m)}\|^{q'} \mathbb{E}_{n/2^{m}} v((n/2^{m}, (n+1)/2^{m}] \times B_{\ell}) \\ &= \frac{1}{\alpha^{(p'-1)q}} \bar{s}_{q'}^{N_{m}}(g^{(m)})^{p'q-q'q+q'}. \end{split}$$

Using (7.A.4) it follows that

$$\begin{split} \|h\|_{L^{p}(\mathbb{P};L^{q}(v))}^{p} &\leq \left(\frac{q}{p}\right)^{p/q} \frac{1}{\alpha^{(p'-1)p}} \bar{s}_{q'}^{N_{m}}(g^{(m)})^{(p'q-q'q+q')p/q} \\ &= \left(\frac{q}{p}\right)^{p/q} \frac{1}{\alpha^{p'}} \mathbb{E} \bar{s}_{q'}^{N_{m}}(g^{(m)})^{p'} = \left(\frac{q}{p}\right)^{p/q}. \end{split}$$

Moreover, by Lemma 7.4.2,

$$\begin{split} F_{g}(h) &= \mathbb{E} \sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \langle g_{n\ell}^{(m)}, h_{n\ell} \rangle v((n/2^{m}, (n+1)/2^{m}] \times B_{\ell}) \\ &= \mathbb{E} \sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \langle g_{n\ell}^{(m)}, h_{n\ell} \rangle \mathbb{E}_{n/2^{m}} v((n/2^{m}, (n+1)/2^{m}] \times B_{\ell}) \\ &\geq (1-\varepsilon) \frac{1}{\alpha^{p'-1}} \mathbb{E} \sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \| g_{n\ell}^{(m)} \|^{q'} \bar{s}_{q'}^{n} (g^{(m)})^{p'-q'} \mathbb{E}_{n/2^{m}} v((n/2^{m}, (n+1)/2^{m}] \times B_{\ell}) \\ &= (1-\varepsilon) \frac{1}{\alpha^{p'-1}} \mathbb{E} \sum_{n=0}^{N_{m}} \bar{s}_{q'}^{n} (g^{(m)})^{p'-q'} (\bar{s}_{q'}^{n} (g^{(m)})^{q'} - \bar{s}_{q'}^{n-1} (g^{(m)})^{q'}). \end{split}$$

Now apply (7.4.10) for  $\alpha = p'/q' \ge 1$  and  $x = \bar{s}_{q'}^n (g^{(m)})^{q'} / \bar{s}_{q'}^{n-1} (g^{(m)})^{q'} \ge 1$  to obtain

$$\begin{split} F_{g}(h) &\geq (1-\varepsilon) \frac{1}{\alpha^{p'-1}} \mathbb{E} \sum_{n=0}^{N_{m}} \frac{q'}{p'} \left( \bar{s}_{q'}^{n}(g^{(m)})^{p'} - \bar{s}_{q'}^{n-1}(g^{(m)})^{p'} \right) \\ &= (1-\varepsilon) \frac{q'}{p'} \frac{1}{\alpha^{p'-1}} \mathbb{E} \bar{s}_{q'}^{N_{m}}(g^{(m)})^{p'} \\ &= (1-\varepsilon) \frac{q'}{p'} \left( \mathbb{E} \bar{s}_{q'}^{N_{m}}(g^{(m)})^{p'} \right)^{1/p'} \\ &= (1-\varepsilon) \frac{q'}{p'} \left( \mathbb{E} \left( \sum_{n=0}^{N_{m}} \sum_{\ell=0}^{L} \|g_{n\ell}^{(m)}\|^{q'} \mathbb{E}_{n/2^{m}} v \left( (n/2^{m}, (n+1)/2^{m}] \times B_{\ell} \right) \right)^{p'/q'} \right)^{1/p'} \\ &= (1-\varepsilon) \frac{q'}{p'} \left( \mathbb{E} \left( \sum_{n=0}^{N_{m}} \mathbb{E}_{n/2^{m}} \left( (\|g\|^{q'} \star v)_{(n+1)/2^{m}} - (\|g\|^{q'} \star v)_{n/2^{m}} \right) \right)^{p'/q'} \right)^{1/p'} \end{split}$$

In conclusion, for any  $m \ge m_*$  we find

$$\|F_g\| \ge \left(\frac{p}{q}\right)^{1/q} \frac{q'}{p'} \left( \mathbb{E}\left(\sum_{n=0}^{N_m} \mathbb{E}_{n/2^m}((\|g\|^{q'} \star v)_{(n+1)/2^m} - (\|g\|^{q'} \star v)_{n/2^m})\right)^{p'/q'}\right)^{1/p'}.$$

Taking  $m \rightarrow \infty$ , we find using Corollary 7.5.19 that

$$||F_g|| \ge \left(\frac{p}{q}\right)^{1/q} \frac{q'}{p'} ||g||_{L^{p'}_{\widetilde{\mathcal{P}}}(\mathbb{P};L^{q'}(v;X^*))}$$

Let us now remove the additional restriction (7.A.7) on v. In this case, we define a strictly increasing, predictable, continuous process

$$A_t := v([0, t] \times J) + t, \qquad t \ge 0$$

and a random time change  $\tau = (\tau_s)_{s \ge 0}$  by

$$\tau_s = \{t : A_t = s\}.$$

By Proposition 7.5.21,  $A \circ \tau(t) = t$  a.s. for any  $t \ge 0$ , and hence by continuity of A and  $\tau$ , a.s.  $A \circ \tau(t) = t$  for all  $t \ge 0$ . As was noted in (7.5.27), we have  $v_{\tau}((s, t] \times J) \le t - s$  a.s. for all  $s \le t$ . By Proposition 7.5.21, we can now write

$$\begin{split} \|F_{g}\| &= \sup_{\|h\|_{L^{p}_{\widetilde{\mathcal{P}}}(\mathbb{P};L^{q}(v;X))} \leq 1} \mathbb{E} \int_{\mathbb{R}_{+} \times J} \langle g,h \rangle dv \\ &\geq \sup_{\|\tilde{h} \circ A\|_{L^{p}_{\widetilde{\mathcal{P}}}(\mathbb{P};L^{q}(v;X))} \leq 1} \mathbb{E} \int_{\mathbb{R}_{+} \times J} \langle g,\tilde{h} \circ A \rangle dv \\ &= \sup_{\|\tilde{h}\|_{L^{p}_{\widetilde{\mathcal{P}}}(\mathbb{P};L^{q}(v;X))} \leq 1} \mathbb{E} \int_{\mathbb{R}_{+} \times J} \langle g \circ \tau,\tilde{h} \rangle dv_{\tau}. \end{split}$$

Applying the previous part of the proof for  $v = v_{\tau}$ , we find

$$\|F_g\| \ge \left(\frac{p}{q}\right)^{1/q} \frac{q'}{p'} \|g \circ \tau\|_{L^{p'}_{\mathscr{D}}(\mathbb{P}; L^{q'}(v_{\tau}; X^*))} = \|g\|_{L^{p'}_{\mathscr{D}}(\mathbb{P}; L^{q'}(v; X^*))}$$

This completes our proof of (7.A.5).

Step 3: representation of linear functionals. It now remains to show that every  $F \in (L^p_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^q(v; X)))^*$  is of the form  $F_g$  for a suitable  $\widetilde{\mathscr{P}}$ -measurable function g. We will first assume that  $\mathbb{E}v(\mathbb{R}_+ \times J) < \infty$ . On  $\widetilde{\mathscr{P}}$  we can define an  $X^*$ -valued measure  $\theta$  by setting

$$\langle \theta(A), x \rangle := F(\mathbf{1}_A \cdot x) \qquad (A \in \widetilde{\mathscr{P}}, x \in X).$$

Then  $\theta$  is  $\sigma$ -additive, absolutely continuous with respect to  $\mathbb{P} \times v$ . Moreover, by the same calculation as in (7.A.2), for any disjoint partition  $A_1, \ldots, A_n \in \widetilde{\mathcal{P}}$  of  $\mathbb{R}_+ \times \Omega \times J$ ,

$$\begin{split} \sum_{i=1}^{n} \|\theta(A_{i})\| &\leq \|F\|_{(L^{p}_{\widetilde{\mathscr{P}}}(\mathbb{P};L^{q}(v;X)))^{*}} (\mathbb{E}\nu(\mathbb{R}_{+} \times J)^{p/q})^{1/p} \\ &\leq \|F\|_{(L^{p}_{\widetilde{\mathscr{P}}}(\mathbb{P};L^{q}(v;X)))^{*}} (\mathbb{E}\nu(\mathbb{R}_{+} \times J))^{1/q}, \end{split}$$

so  $\theta$  is of finite variation. By the Radon-Nikodym property of  $X^*$ , there exists a  $\widetilde{\mathscr{P}}$ -measurable  $X^*$ -valued function g such that

$$F(h) = F_g(h) = \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle g, h \rangle \, \mathrm{d}v$$

for each  $h \in L^p_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^q(v; X))$ . The extension to the general case, where v is  $\widetilde{\mathcal{P}}$ - $\sigma$ -finite, can now be obtained in the same way as in the proof of Theorem 7.A.1.

Remark 7.A.4. The reader may wonder whether the duality

$$(L^{p}_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^{q}(\nu; X)))^{*} = L^{p'}_{\widetilde{\mathcal{P}}}(\mathbb{P}; L^{q'}(\nu; X^{*}))$$

remains valid if v is any random measure and  $\widetilde{\mathscr{P}}$  is replaced by an arbitrary sub- $\sigma$ -algebra of  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$ . It turns out that, surprisingly, one cannot expect such a general result. Indeed, it was pointed out by Pisier [151] that there exist two probability spaces ( $\Omega_1, \mathscr{F}_1, \mathbb{P}_1$ ), ( $\Omega_2, \mathscr{F}_2, \mathbb{P}_2$ ) and a sub- $\sigma$ -algebra  $\mathscr{G}$  of  $\mathscr{F}_1 \otimes \mathscr{F}_2$ , so that the duality

$$(L^p_{\mathcal{Q}}(\mathbb{P}_1; L^q(\mathbb{P}_2)))^* = L^{p'}_{\mathcal{Q}}(\mathbb{P}_1; L^{q'}(\mathbb{P}_2))$$

does not even hold isomorphically. This counterexample in particular shows that the duality results claimed in [109] are not valid without imposing additional assumptions.

### **7.A.2.** $\mathscr{S}_q^p$ and $\hat{\mathscr{S}}_q^p$ spaces

Let *v* be any random measure on  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{J}$ . Recall that  $\mathscr{S}_q^p$  is the space of all  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{F} \otimes \mathscr{J}$ -strongly measurable functions  $f : \mathbb{R}_+ \otimes \Omega \otimes J \to L^q(S)$  satisfying

$$\|f\|_{\mathscr{S}_{q}^{p}} = \left(\mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J} |f|^{2} \,\mathrm{d}\nu\right)^{\frac{1}{2}}\right\|_{L^{q}(S)}^{p}\right)^{\frac{1}{p}} < \infty.$$
(7.A.8)

The proof of the following result is analogous to Theorem 7.A.1. We leave the details to the reader.

**Theorem 7.A.5.** Let  $1 < p, q < \infty$ . Then  $(\mathscr{S}_q^p)^* = \mathscr{S}_{a'}^{p'}$  and

$$\|f\|_{\mathcal{S}_{q'}^{p'}} = \sum_{p,q} \|f\|_{(\mathcal{S}_{q}^{p})^{*}}, \ f \in \mathcal{S}_{q'}^{p'}.$$

Let us now prove the desired duality for  $\hat{\mathscr{S}}_q^p$ , the subspace of all  $\widetilde{\mathscr{P}}$ -strongly measurable functions in  $\mathscr{S}_q^p$ .

**Theorem 7.A.6.** Let  $1 < p, q < \infty$ . Suppose that v is a predictable,  $\widetilde{\mathcal{P}}$ - $\sigma$ -finite random measure on  $\mathscr{B}(\mathbb{R}_+) \otimes \mathscr{J}$  that is non-atomic in time. Then  $(\widehat{\mathscr{F}}_q^p)^* = \widehat{\mathscr{F}}_{a'}^{p'}$  and

$$\|f\|_{\hat{\mathscr{S}}_{q'}^{p'}} \approx_{p,q} \|f\|_{(\hat{\mathscr{S}}_{q}^{p})^{*}}, \ f \in \hat{\mathscr{S}}_{q'}^{p'}.$$
(7.A.9)

For the proof of Theorem 7.A.6 we will the following assertion. Given a filtration  $\mathbb{F} = (\mathscr{F}_n)_{n \ge 0}$  and  $1 < p, q < \infty$ , we define  $Q_q^p$  to be the Banach space of all adapted  $L^q(S)$ -valued sequences  $(f_n)_{n \ge 0}$  satisfying

$$\|(f_n)_{n\geq 0}\|_{Q^p_q} := \left(\mathbb{E}\left\|\left(\sum_{n=0}^{\infty} |f_n|^2\right)^{\frac{1}{2}}\right\|_{L^q(S)}^p\right)^{\frac{1}{p}} < \infty.$$
(7.A.10)

**Proposition 7.A.7.** Let  $1 < p, q < \infty$ . Then  $(Q_q^p)^* = Q_{q'}^{p'}$  isomorphically, with duality bracket given by

$$\langle (f_n)_{n\geq 0}, (g_n)_{n\geq 0} \rangle := \mathbb{E} \sum_{n=0}^{\infty} \langle f_n, g_n \rangle \qquad ((g_n)_{n\geq 0} \in Q_{q'}^{p'}, \ (f_n)_{n\geq 0} \in Q_q^p).$$

Moreover,

$$\|(g_n)_{n\geq 0}\|_{Q_{q'}^{p'}} = p,q \|(g_n)_{n\geq 0}\|_{(Q_q^p)^*}.$$

*Proof.* Consider the filtration  $\mathbb{G} = (\mathcal{G}_n)_{n \ge 0} = (\mathcal{F}_{n+1})_{n \ge 0}$ . Let  $S_q^p$  be the conditional sequence space defined in (7.1.4) for the filtration  $\mathbb{G}$ . First notice that  $Q_q^p$  is a closed subspace and

$$\|(f_n)_{n\geq 0}\|_{Q^p_a} = \|(f_n)_{n\geq 0}\|_{S^p_a}, \quad \text{for all } (f_n)_{n\geq 0} \in Q^p_q.$$

Let *F* be in  $(Q_q^p)^*$ . Then by the Hahn-Banach theorem and [87] there exists  $\tilde{g} = (\tilde{g}_n)_{n\geq 0} \in S_{q'}^{p'}$  such that  $\|\tilde{g}\|_{S_{q'}^{p'}} \approx_{p,q} \|F\|_{(Q_q^p)^*}$  and

$$F(f) = \mathbb{E} \sum_{n=1}^{\infty} \langle f_n, \tilde{g}_n \rangle, \ f = (f_n)_{n \ge 0} \in Q_q^p.$$

Now let  $(g_n)_{n\geq 0}$  be the  $\mathbb{F}$ -adapted  $L^q(S)$ -valued sequence defined by  $g_n = \mathbb{E}_n \tilde{g}_n$  for  $n \geq 0$  (recall that  $\mathbb{E}_n(\cdot) := \mathbb{E}(\cdot | \mathscr{F}_n)$ ). Then, on the one hand, the conditional Jensen inequality yields

$$\begin{aligned} \|(g_{n})_{n\geq 0}\|_{Q_{q'}^{p'}}^{p'} &= \|(g_{n})_{n\geq 0}\|_{S_{q'}^{p'}}^{p'} = \mathbb{E}\left\|\left(\sum_{n=1}^{\infty} |g_{n}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q'}(S)}^{p'} = \mathbb{E}\left\|\left(\sum_{n=1}^{\infty} |\mathbb{E}_{n}\tilde{g}_{n}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q'}(S)}^{p'} \\ &\leq \mathbb{E}\left\|\left(\sum_{n=1}^{\infty} \mathbb{E}_{n}|\tilde{g}_{n}|^{2}\right)^{\frac{1}{2}}\right\|_{L^{q'}(S)}^{p'} = \|(\tilde{g}_{n})_{n\geq 0}\|_{S_{q'}^{p'}}^{p'}, \end{aligned}$$

and, on the other hand, for each  $f = (f_n)_{n \ge 0} \in Q_q^p$  the  $\mathbb{F}$ -adaptedness of  $(f_n)_{n \ge 0}$  implies

$$F(f) = \mathbb{E}\sum_{n=1}^{\infty} \langle f_n, \tilde{g}_n \rangle = \mathbb{E}\sum_{n=1}^{\infty} \mathbb{E}_n \langle f_n, \tilde{g}_n \rangle = \mathbb{E}\sum_{n=1}^{\infty} \langle f_n, \mathbb{E}_n \tilde{g}_n \rangle = \mathbb{E}\sum_{n=1}^{\infty} \langle f_n, g_n \rangle.$$

Therefore, for each  $F \in (Q_q^p)^*$  there exists a  $(g_n)_{n \ge 0} \in Q_{p'}^{q'}$  such that

$$F(f) = \mathbb{E} \sum_{n \ge 0} \langle f_n, g_n \rangle, \quad f = (f_n)_{n \ge 0} \in Q_q^p$$
$$\|(g_n)_{n \ge 0}\|_{Q_{n'}^{q'}} \lesssim_{p,q} \|F\|_{(Q_q^p)^*}.$$

The inequality  $||F||_{(Q_q^p)^*} \le ||(g_n)_{n\ge 0}||_{Q_{p'}^{q'}}$  follows immediately from Hölder's inequality.

*Proof of Theorem 7.A.6.* The proof contains two parts. In the first part, consisting of several steps, we will show that  $||f||_{\hat{\mathscr{P}}_{q}^{p'}} \approx_{p,q} ||f||_{(\hat{\mathscr{P}}_{q}^{p})^{*}}$ . In the second part we show

that  $(\hat{\mathscr{S}}_q^p)^* = \hat{\mathscr{S}}_{q'}^{p'}$ .

Step 1: J is finite, v is Lebesgue. Let  $J = \{j_1, ..., j_K\}$ ,  $v(\omega)$  be the product of Lebesgue measure and the counting measure on  $\mathbb{R}_+ \times J$  for all  $\omega \in \Omega$  (i.e.  $v((s, t] \times j_k) = t - s$  for each k = 1, ..., K and  $t \ge s \ge 0$ ). Fix  $f \in \hat{\mathscr{S}}_{q'}^{p'}$ . Without loss of generality we can assume that f is simple and that there exist  $N, M \ge 1$  and a sequence of random variables  $(f_{k,m})_{k=1,m=0}^{k=K,m=M}$  such that  $f_{k,m}$  is  $\mathscr{F}_{\overline{N}}^m$ -measurable and  $f(t, j_k) = f_{k,m}$  for each k = 1, ..., K, m = 0, ..., M, and  $t \in (\frac{m}{N}, \frac{m+1}{N}]$ . Let  $\mathbb{G} = (\mathscr{G}_{k,m})_{k=1,m=0}^{k=K,m=M} := (\mathscr{F}_{\overline{M}})_{k=1,m=0}^{k=K,m=M}$ . Then  $\mathbb{G}$  forms a filtration with respect to the reverse lexicographic order on the pairs  $(k,m), 1 \le k \le K$  and  $0 \le m \le M$ . Let  $Q_{a'}^{p'}$  be as defined in (7.A.10) for  $\mathbb{G}$ . Then

$$\|f\|_{\hat{\mathscr{S}}_{q'}^{p'}} = \frac{1}{\sqrt{N}} \|(f_{k,m})_{k=1,m=0}^{k=K,m=M}\|_{Q_{q'}^{p'}}.$$
(7.A.11)

By Proposition 7.A.7 there exists a G-adapted  $(g_{k,m})_{k=1,m=0}^{k=K,m=M} \in Q_q^p$  such that

$$\left\| \left(g_{k,m}\right)_{k=1,m=0}^{k=K,m=M} \right\|_{Q_q^p} = 1$$

and

$$\langle (f_{k,m})_{k=1,m=0}^{k=K,m=M}, (g_{k,m})_{k=1,m=0}^{k=K,m=M} \rangle \sim_{p,q} \| (f_{k,m})_{k=1,m=0}^{k=K,m=M} \|_{Q_{q'}^{p'}}$$

Let  $g : \mathbb{R}_+ \times \Omega \times J \to L^q(S)$  be defined by setting  $g(t, j_k) = \sqrt{N}g_{k,m}$  for each k = 1, ..., K, m = 1, ..., M, and  $t \in (\frac{m}{N}, \frac{m+1}{N}]$ . Then  $g \in \hat{\mathcal{S}}_q^p$ , and analogously to (7.A.11)

$$\|g\|_{\hat{\mathscr{S}}_{q}^{p}} = \|(g_{k,m})_{k=1,m=0}^{k=K,m=M}\|_{Q_{q}^{p}} = 1.$$

Moreover,

$$\begin{split} \langle f,g\rangle &= \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle f(t,j),g(t,j)\rangle \,\mathrm{d}t \,\mathrm{d}j = \frac{1}{\sqrt{N}} \mathbb{E} \sum_{k=1,m=0}^{k=K,m=M} \langle f_{k,m},g_{k,m}\rangle \\ &= \sum_{p,q} \frac{1}{\sqrt{N}} \left\| (f_{k,m})_{k=1,m=0}^{k=K,m=M} \right\|_{Q_{q'}^{p'}} = \|f\|_{\hat{\mathcal{S}}_{q'}^{p'}}, \end{split}$$

which finishes the proof.

Step 2: *J* is finite,  $v((s, t] \times J) \leq t - s$  a.s. for each  $t \geq s \geq 0$ . Let  $v_0$  be the product of Lebesgue measure and the counting measure on  $\mathbb{R}_+ \times J$  (see Step 1). Then clearly  $\mathbb{P} \otimes v$  is absolutely continuous with respect to  $\mathbb{P} \otimes v_0$  and by the Radon-Nikodym theorem there exists a  $\widetilde{\mathscr{P}}$ -measurable density  $\phi : \mathbb{R}_+ \times \Omega \times J \to \mathbb{R}_+$  such that  $d(\mathbb{P} \otimes v) = \phi \ d(\mathbb{P} \otimes v_0)$ . Fix  $f \in \hat{\mathscr{P}}_{q'}^{p'}$ . Let  $\hat{\mathscr{P}}_{q'}^{p',v_0}$  be as defined in (7.A.8) for the random measure  $v_0$ . Then  $f_0 := f \cdot \sqrt{\phi} \in \hat{\mathscr{P}}_{q'}^{p',v_0}$ , and  $\|f\|_{\hat{\mathscr{P}}_{q'}^{p'}} = \|f_0\|_{\hat{\mathscr{P}}_{q'}^{p',v_0}}$ . By Step 1 there exists a  $g_0 \in \hat{\mathscr{P}}_{q}^{p,v_0}$  such that  $\|g_0\|_{\hat{\mathscr{P}}_{q}^{p,v_0}} = 1$  and  $\langle f_0, g_0 \rangle \sim_{p,q} \|f_0\|_{\hat{\mathscr{P}}_{q'}^{p',v_0}}$ . Let  $g = g_0 \mathbf{1}_{\phi \neq 0} \frac{1}{\sqrt{\phi}}$ . Then

$$\langle f,g\rangle = \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle f,g\rangle \, \mathrm{d}\nu = \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle f,g\rangle \phi \, \mathrm{d}\nu_0 = \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle f\sqrt{\phi},g\sqrt{\phi}\rangle \, \mathrm{d}\nu_0$$

$$= \mathbb{E} \int_{\mathbb{R}_+ \times J} \langle f_0, g_0 \rangle \, \mathrm{d}\nu_0 = \langle f_0, g_0 \rangle \eqsim_{p,q} \| f_0 \|_{\hat{\mathscr{S}}_{q'}^{p',\nu_0}} = \| f \|_{\hat{\mathscr{S}}_{q'}^{p'}}$$

and

$$\begin{split} \|g\|_{\hat{\mathscr{P}}_{q}^{p}} &= \left(\mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J}|g|^{2}\,\mathrm{d}\nu\right)^{\frac{1}{2}}\right\|^{p}\right)^{\frac{1}{p}} = \left(\mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J}|g_{0}|^{2}\mathbf{1}_{\phi\neq 0}\,\frac{1}{\phi}\,\mathrm{d}\nu\right)^{\frac{1}{2}}\right\|^{p}\right)^{\frac{1}{p}} \\ &= \left(\mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J}|g_{0}|^{2}\mathbf{1}_{\phi\neq 0}\,\mathrm{d}\nu_{0}\right)^{\frac{1}{2}}\right\|^{p}\right)^{\frac{1}{p}} \le \left(\mathbb{E}\left\|\left(\int_{\mathbb{R}_{+}\times J}|g_{0}|^{2}\,\mathrm{d}\nu_{0}\right)^{\frac{1}{2}}\right\|^{p}\right)^{\frac{1}{p}} \\ &= \|g_{0}\|_{\hat{\mathscr{P}}_{q}^{p,v_{0}}} = 1. \end{split}$$

Therefore  $||f||_{\hat{\mathscr{S}}_{q'}^{p'}} \equiv_{p,q} ||f||_{(\hat{\mathscr{S}}_{q}^{p})^{*}}.$ 

Step 3: *J* is finite, *v* is general. Without loss of generality we can assume that  $\mathbb{E}v(\mathbb{R}_+ \times J) < \infty$ . Then by a time-change argument as was used in the proof of Theorem 7.A.3, we can assume that  $v((s, t] \times J) \le t - s$  a.s. for each  $t \ge s \ge 0$ , and apply Step 2.

Step 4: *J* is general, *v* is general. Without loss of generality assume that  $\mathbb{E}v(\mathbb{R}_+ \times J) < \infty$ . Let *f* be simple  $\widetilde{\mathscr{P}}$ -measurable. Then there exists a  $K \ge 1$  and a partition  $J = J_1 \cup \cdots \cup J_K$  of *J* into disjoint sets such that f(i) = f(j) for all  $i, j \in J_k$  and each  $k = 1, \dots, K$ . Fix  $j_k \in J_k$ ,  $k = 1, \dots, K$ , and define  $\widetilde{J} = \{j_1, \dots, j_K\}$ . Let  $\widetilde{v}$  be a new random measure on  $\mathbb{R}_+ \times \Omega \times \widetilde{J}$  defined by

$$\widetilde{\nu}(A \times \{j_k\}) = \nu(A \times J_k), \qquad A \in \mathcal{P}, \ k = 1, \dots, K.$$

Let  $\hat{\mathscr{S}}_{q'}^{p',\tilde{v}}$  be as constructed in (7.A.8) for the measure  $\tilde{v}$ . Let  $\tilde{f} \in \hat{\mathscr{S}}_{q'}^{p',\tilde{v}}$  be such that  $\tilde{f}(j_k) = f(j_k)$  for each k = 1, ..., K. Then  $\|\tilde{f}\|_{\hat{\mathscr{S}}_{q'}^{p',\tilde{v}}} = \|f\|_{\hat{\mathscr{S}}_{q'}^{p'}}$ . By Step 3 there exists a  $\tilde{g} \in \hat{\mathscr{S}}_{q}^{p,\tilde{v}}$  such that  $\|\tilde{g}\|_{\hat{\mathscr{S}}_{q}^{p,\tilde{v}}} = 1$  and  $\langle \tilde{f}, \tilde{g} \rangle \approx_{p,q} \|\tilde{f}\|_{\hat{\mathscr{S}}_{q'}^{p',\tilde{v}}}$ .

Define  $g \in \hat{\mathscr{P}}_q^p$  by setting  $g(j) = \tilde{g}(j_k)$  for each k = 1, ..., K and  $j \in J_k$ . Then  $\|g\|_{\hat{\mathscr{P}}_q^p} = \|\tilde{g}\|_{\hat{\mathscr{P}}_q^{p,\tilde{v}}} = 1$ . Moreover,

$$\begin{split} \langle f,g\rangle &= \mathbb{E} \int_{\mathbb{R}_{+}\times J} \langle f(t,j),g(t,j)\rangle \,\mathrm{d}\nu(t,j) = \mathbb{E} \sum_{k=1}^{K} \int_{\mathbb{R}_{+}\times J_{k}} \langle f(t,j),g(t,j)\rangle \,\mathrm{d}\nu(t,j) \\ &= \mathbb{E} \int_{\mathbb{R}_{+}\times \widetilde{J}} \langle \widetilde{f}(t,j),\widetilde{g}(t,j)\rangle \,\mathrm{d}\widetilde{\nu}(t,j) \approx_{p,q} \|\widetilde{f}\|_{\hat{\mathcal{G}}_{q'}^{p',\widetilde{\nu}}} = \|f\|_{\hat{\mathcal{G}}_{q'}^{p'}}. \end{split}$$

Hence,  $||f||_{\hat{\mathscr{G}}_{q'}^{p'}} = p,q ||f||_{(\hat{\mathscr{G}}_{q}^{p})^{*}}.$ 

Step 5:  $(\hat{\mathscr{P}}_q^p)^* = \hat{\mathscr{P}}_{q'}^{p'}$ . In Step 4 we proved that  $\hat{\mathscr{P}}_{q'}^{p'} \hookrightarrow (\hat{\mathscr{P}}_q^p)^*$  isomorphically, so it remains to show that  $(\hat{\mathscr{P}}_q^p)^* = \hat{\mathscr{P}}_{q'}^{p'}$ . This identity follows from the same Radon-Nikodym argument that was presented in Step 3 of the proof of Theorem 7.A.3.

**Corollary 7.A.8.** Let  $1 < p, q < \infty$ . Then  $\mathscr{I}_{p,q}^* = \mathscr{I}_{p',q'}$ , where  $\mathscr{I}_{p,q}$  is as defined in (7.5.22), and

$$\|f\|_{\mathscr{I}_{p',q'}} = \sum_{p,q} \|f\|_{\mathscr{I}_{p,q}^*}, \ f \in \mathscr{I}_{p',q'}.$$
(7.A.12)

*Proof.* The result follows by combining Theorem 7.A.3 (for  $X = L^{q}(S)$ ), Theorem 7.A.6 and (7.2.2).

# BURKHOLDER–DAVIS–GUNDY INEQUALITIES IN UMD BANACH FUNCTION SPACES

This chapter is based on the paper *Pointwise properties of martingales with values in Banach function spaces* by Mark Veraar and Ivan Yaroslavtsev, see [178].

In this chapter we consider local martingales with values in a UMD Banach function space. We prove that such martingales have a version which is a martingale field. Moreover, a new Burkholder–Davis–Gundy type inequality is obtained.

2010 Mathematics Subject Classification. Primary: 60G44 Secondary: 60B11, 60H05, 60G48. Key words and phrases. Local martingale, quadratic variation, UMD Banach function spaces, Burkholder-Davis-Gundy inequalities, lattice maximal function.

### **8.1.** INTRODUCTION

The discrete Burkholder–Davis–Gundy inequality (see [29, Theorem 3.2]) states that for any  $p \in (1,\infty)$  and martingales difference sequence  $(d_j)_{j=1}^n$  in  $L^p(\Omega)$  one has

$$\left\|\sum_{j=1}^{n} d_{j}\right\|_{L^{p}(\Omega)} \approx_{p} \left\|\left(\sum_{j=1}^{n} |d_{j}|^{2}\right)^{1/2}\right\|_{L^{p}(\Omega)}.$$
(8.1.1)

Moreover, there is the extension to continuous-time local martingales *M* (see [89, Theorem 26.12]) which states that for every  $p \in [1,\infty)$ ,

$$\|\sup_{t\in[0,\infty)} |M_t|\|_{L^p(\Omega)} \approx_p \|[M]_{\infty}^{1/2}\|_{L^p(\Omega)}.$$
(8.1.2)

Here  $t \mapsto [M]_t$  denotes the quadratic variation process of *M*.

In the case *X* is a UMD Banach function space the following variant of (8.1.1) holds (see [164, Theorem 3]): for any  $p \in (1, \infty)$  and martingales difference sequence  $(d_j)_{j=1}^n$  in  $L^p(\Omega; X)$  one has

$$\left\|\sum_{j=1}^{n} d_{j}\right\|_{L^{p}(\Omega;X)} \approx_{p} \left\|\left(\sum_{j=1}^{n} |d_{j}|^{2}\right)^{1/2}\right\|_{L^{p}(\Omega;X)}.$$
(8.1.3)

Moreover, the validity of the estimate also characterizes the UMD property.

It is a natural question whether (8.1.2) has a vector-valued analogue as well. The main result of this chapter states that this is indeed the case:

**Theorem 8.1.1.** Let X be a UMD Banach function space over a  $\sigma$ -finite measure space  $(S, \Sigma, \mu)$ . Assume that  $N : \mathbb{R}_+ \times \Omega \times S \to \mathbb{R}$  is such that  $N|_{[0,t] \times \Omega \times S}$  is  $\mathscr{B}([0,t]) \otimes \mathscr{F}_t \otimes \Sigma$ -measurable for all  $t \ge 0$  and such that for almost all  $s \in S$ ,  $N(\cdot, \cdot, s)$  is a martingale with respect to  $(\mathscr{F}_t)_{t\ge 0}$  and  $N(0, \cdot, s) = 0$ . Then for all  $p \in (1, \infty)$ ,

$$\|\sup_{t\geq 0} |N(t,\cdot,\cdot)|\|_{L^{p}(\Omega;X)} \approx_{p,X} \sup_{t\geq 0} \|N(t,\cdot,\cdot)\|_{L^{p}(\Omega;X)} \approx_{p,X} \|[N]_{\infty}^{1/2}\|_{L^{p}(\Omega;X)}.$$
(8.1.4)

where [N] denotes the quadratic variation process of N.

By standard methods we can extend Theorem 8.1.1 to spaces *X* which are isomorphic to a closed subspace of a Banach function space (e.g. Sobolev and Besov spaces, etc.)

The two-sided estimate (8.1.4) can for instance be used to obtain two-sided estimates for stochastic integrals for processes with values in infinite dimensions (see [126] and [177]). In particular, applying it with  $N(t, \cdot, s) = \int_0^t \Phi(\cdot, s) dW$  implies the following maximal estimate for the stochastic integral

$$\left\| s \mapsto \sup_{t \ge 0} \left| \int_0^t \Phi(\cdot, s) \, \mathrm{d}W \right| \right\|_{L^p(\Omega; X)} = p_{X} \sup_{t \ge 0} \left\| s \mapsto \int_0^t \Phi(\cdot, s) \, \mathrm{d}W \right\|_{L^p(\Omega; X)}$$

$$=_{p,X} \left\| s \mapsto \left( \int_0^\infty \Phi^2(t,s) \, \mathrm{d}t \right)^{1/2} \right\|_{L^p(\Omega;X)}, \qquad (8.1.5)$$

where *W* is a Brownian motion and  $\Phi : \mathbb{R}_+ \times \Omega \times S \to \mathbb{R}$  is a progressively measurable process such that the right-hand side of (8.1.5) is finite. The second norm equivalence was obtained in [126]. The norm equivalence with the left-hand side is new in this generality. The case where *X* is an *L*<sup>*q*</sup>-space was recently obtained in [4] using different methods.

It is worth noticing that the second equivalence of (8.1.4) in the case of  $X = L^q$  was obtained by Marinelli in [110] for some range of  $1 < p, q < \infty$  by using an interpolation method.

The UMD property is necessary in Theorem 8.1.1 by necessity of the UMD property in (8.1.3) and the fact that any discrete martingale can be transformed to a continuous-time one. Also in the case of continuous martingales, the UMD property is necessary in Theorem 8.1.1. Indeed, applying (8.1.5) with *W* replaced by an independent Brownian motion  $\widetilde{W}$  we obtain

$$\left\|\int_0^\infty \Phi \,\mathrm{d} W\right\|_{L^p(\Omega;X)} \approx_{p,X} \left\|\int_0^\infty \Phi \,\mathrm{d} \widetilde{W}\right\|_{L^p(\Omega;X)}$$

for all predictable step processes  $\Phi$ . The latter holds implies that *X* is a UMD Banach space (see [61, Theorem 1]).

In the special case that  $X = \mathbb{R}$  the above reduces to (8.1.2). In the proof of Theorem 8.1.1 the UMD property is applied several times:

- The boundedness of the lattice maximal function (see [24, 60, 164]).
- The *X*-valued Meyer–Yoeurp decomposition of a martingale (see Theorem 4.3.1).
- The square-function estimate (8.1.3) (see [164]).

It remains open whether there exists a predictable expression for the right-hand side of (8.1.4). One would expect that one needs simply to replace [*N*] by its predictable compensator, the *predictable quadratic variation*  $\langle N \rangle$ . Unfortunately, this does not hold true already in the scalar-valued case: if *M* is a real-valued martingale, then

$$\begin{split} & \mathbb{E}|M|_t^p \lesssim_p \mathbb{E}\langle M \rangle_t^{\frac{p}{2}}, \ t \ge 0, \ p < 2, \\ & \mathbb{E}|M|_t^p \gtrsim_p \mathbb{E}\langle M \rangle_t^{\frac{p}{2}}, \ t \ge 0, \ p > 2, \end{split}$$

where both inequalities are known not to be sharp (see [29, p. 40], [114, p. 297], and [140]). The question of finding such a predictable right-hand side in (8.1.4) was answered only in the case  $X = L^q$  for  $1 < q < \infty$  by Dirsken and the author (see

[54]). The key tool exploited there was the so-called *Burkholder-Rosenthal inequalities,* which are of the following form:

$$\mathbb{E}\|M_N\|^p \approx_{p,X} \||(M_n)_{0 \le n \le N}\||_{p,X}^p,$$

where  $(M_n)_{0 \le n \le N}$  is an *X*-valued martingale,  $\|\|\cdot\|\|_{p,X}$  is a certain norm defined on the space of *X*-valued  $L^p$ -martingales which depends only on *predictable moments* of the corresponding martingale. Therefore using approach of [54] one can reduce the problem of continuous-time martingales to discrete-time martingales. However, the Burkholder-Rosenthal inequalities are explored only in the case  $X = L^q$ .

Thanks to (8.1.2) the following natural question arises: can one generalize (8.1.4) to the case p = 1, i.e. whether

$$\|\sup_{t\geq 0} |N(t,\cdot,\cdot)|\|_{L^{1}(\Omega;X)} \approx_{p,X} \|[N]_{\infty}^{1/2}\|_{L^{1}(\Omega;X)}$$
(8.1.6)

holds true? Unfortunately the outlined earlier techniques cannot be applied in the case p = 1. Moreover, the obtained estimates cannot be simply extrapolated to the case p = 1 since those contain the  $UMD_p$  constant, which is known to have infinite limit as  $p \rightarrow 1$ . Therefore (8.1.6) remains an open problem. Note that in the case of a continuous martingale M inequalities (8.1.4) can be extended to the case  $p \in (0, 1]$  due to the classical Lenglart approach (see Corollary 8.4.4).

### **8.2. PRELIMINARIES**

Throughout the chapter any filtration satisfies the *usual conditions* (see [85, Definition 1.1.2 and 1.1.3]), unless the underlying martingale is continuous (then the corresponding filtration can be assumed general).

Recall that for a given measure space  $(S, \Sigma, \mu)$ , the linear space of all real-valued measurable functions is denoted by  $L^0(S)$ .

**Definition 8.2.1.** Let  $(S, \Sigma, \mu)$  be a measure space. Let  $n : L^0(S) \to [0, \infty]$  be a function which satisfies the following properties:

- (i) n(x) = 0 if and only if x = 0,
- (ii) for all  $x, y \in L^0(S)$  and  $\lambda \in \mathbb{R}$ ,  $n(\lambda x) = |\lambda| n(x)$  and  $n(x + y) \le n(x) + n(y)$ ,
- (iii) if  $x \in L^{0}(S)$ ,  $y \in L^{0}(S)$ , and  $|x| \le |y|$ , then  $n(x) \le n(y)$ ,
- (iv) if  $0 \le x_n \uparrow x$  with  $(x_n)_{n=1}^{\infty}$  a sequence in  $L^0(S)$  and  $x \in L^0(S)$ , then  $n(x) = \sup_{n \in \mathbb{N}} n(x_n)$ .

Let *X* denote the space of all  $x \in L^0(S)$  for which  $||x|| := n(x) < \infty$ . Then *X* is called the *normed function space associated to n*. It is called a *Banach function space* when  $(X, || \cdot ||_X)$  is complete.

We refer the reader to [108, 124, 164, 178, 192] for details on Banach function spaces.

*Remark* 8.2.2. Let *X* be a Banach function space over a measure space  $(S, \Sigma, \mu)$ . Then *X* is continuously embedded into  $L^0(S)$  endowed with the topology of convergence in measure on sets of finite measure. Indeed, assume  $x_n \to x$  in *X* and let  $A \in \Sigma$  be of finite measure. We claim that  $\mathbf{1}_A x_n \to \mathbf{1}_A x$  in measure. For this it suffices to show that every subsequence of  $(x_n)_{n\geq 1}$  has a further subsequence which convergences a.e. to *x*. Let  $(x_{n_k})_{k\geq 1}$  be a subsequence. Choose a subsubsequence  $(\mathbf{1}_A x_{n_k})_{\ell\geq 1} =: (y_\ell)_{\ell\geq 1}$  such that  $\sum_{\ell=1}^{\infty} ||y_\ell - x|| < \infty$ . Then by [192, Exercise 64.1]  $\sum_{\ell=1}^{\infty} |y_\ell - x|$  converges in *X*. In particular,  $\sum_{\ell=1}^{\infty} |y_\ell - x| < \infty$  a.e. Therefore,  $y_\ell \to x$  a.e. as desired.

Given a Banach function space *X* over a measure space *S* and Banach space *E*, let *X*(*E*) denote the space of all strongly measurable functions  $f : S \to E$  with  $||f||_E \in X$ . The space *X*(*E*) becomes a Banach space when equipped with the norm  $||f||_{X(E)} = ||s \mapsto ||f(s)||_E ||_X$ .

A Banach function space has the UMD property if and only if (8.1.3) holds for some (or equivalently, for all)  $p \in (1,\infty)$  (see [164]). A broad class of Banach function spaces with UMD is given by the reflexive Lorentz–Zygmund spaces (see [43]) and the reflexive Musielak–Orlicz spaces (see [107]).

**Definition 8.2.3.**  $N : \mathbb{R}_+ \times \Omega \times S \to \mathbb{R}$  is called (continuous) (local) martingale field if  $N|_{[0,t]\times\Omega\times S}$  is  $\mathscr{B}([0,t]) \otimes \mathscr{F}_t \otimes \Sigma$ -measurable for all  $t \ge 0$  and  $N(\cdot, \cdot, s)$  is a (continuous) (local) martingale with respect to  $(\mathscr{F}_t)_{t\ge 0}$  for almost all  $s \in S$ .

Let *X* be a Banach space,  $\tau$  be a stopping time,  $V : \mathbb{R}_+ \times \Omega \to X$  be a càdlàg process. Then we define  $\Delta V_\tau : \Omega \to X$  as follows

$$\Delta V_{\tau} := V_{\tau} - \lim_{\varepsilon \to 0} V_{(\tau - \varepsilon) \vee 0}.$$

### **8.3.** LATTICE DOOB'S MAXIMAL INEQUALITY

Doob's maximal  $L^p$ -inequality immediately implies that for martingale fields

$$\left\| \sup_{t \ge 0} \| N(t, \cdot) \|_X \right\|_{L^p(\Omega)} \le \frac{p}{p-1} \sup_{t \ge 0} \| N(t) \|_{L^p(\Omega; X)}, \ 1$$

In the next lemma we prove a stronger version of Doob's maximal  $L^p$ -inequality. As a consequence in Theorem 8.3.2 we will obtain the same result in a more general setting.

**Lemma 8.3.1.** Let X be a UMD Banach function space and let  $p \in (1,\infty)$ . Let N be a càdlàg martingale field with values in a finite dimensional subspace of X. Then for all T > 0,

$$\left\|\sup_{t\in[0,T]}|N(t,\cdot)|\right\|_{L^p(\Omega;X)}\eqsim_{p,X}\sup_{t\in[0,T]}\|N(t)\|_{L^p(\Omega;X)}$$

whenever one of the expression is finite.

*Proof.* Clearly, the left-hand side dominates the right-hand side. Therefore, we can assume the right-hand side is finite and in this case we have

$$\|N(T)\|_{L^{p}(\Omega;X)} = \sup_{t \in [0,T]} \|N(t)\|_{L^{p}(\Omega;X)} < \infty.$$

Since *N* takes values in a finite dimensional subspace it follows from Doob's  $L^p$ -inequality (applied coordinatewise) that the left-hand side is finite.

Since N is a càdlàg martingale field and by Definition 8.2.1(iv) we have that

$$\lim_{n \to \infty} \left\| \sup_{0 \le j \le n} |N(jT/n, \cdot)| \right\|_{L^p(\Omega; X)} = \left\| \sup_{t \in [0, T]} |N(t, \cdot)| \right\|_{L^p(\Omega; X)}.$$

Set  $M_j = N_{jT/n}$  for  $j \in \{0, ..., n\}$  and  $M_j = M_n$  for j > n. It remains to prove

$$\left\|\sup_{0\leq j\leq n}|M_j(\cdot)|\right\|_{L^p(\Omega;X)}\leq C_{p,X}\|M_n\|_{L^p(\Omega;X)}.$$

If  $(M_j)_{j=0}^n$  is a Paley–Walsh martingale (see [79, Definition 3.1.8 and Proposition 3.1.10]), this estimate follows from the boundedness of the dyadic lattice maximal operator [164, pp. 199–200 and Theorem 3]. In the general case one can replace  $\Omega$  by a divisible probability space and approximate  $(M_j)$  by Paley-Walsh martingales in a similar way as in [79, Corollary 3.6.7].

**Theorem 8.3.2** (Doob's maximal  $L^p$ -inequality). Let X be a UMD Banach function space over a  $\sigma$ -finite measure space and let  $p \in (1, \infty)$ . Let  $M : \mathbb{R}_+ \times \Omega \to X$  be a martingale such that

- 1. for all  $t \ge 0$ ,  $M(t) \in L^p(\Omega; X)$ ;
- 2. for a.a  $\omega \in \Omega$ ,  $M(\cdot, \omega)$  is in  $\mathcal{D}([0, \infty); X)$ .

Then there exists a martingale field  $N \in L^p(\Omega; X(\mathcal{D}_b([0,\infty))))$  such that for a.a.  $\omega \in \Omega$ , all  $t \ge 0$  and a.a.  $s \in S$ ,  $N(t, \omega, \cdot) = M(t, \omega)(s)$  and

$$\|\sup_{t\geq 0} |N(t,\cdot)|\|_{L^{p}(\Omega;X)} \approx_{p,X} \sup_{t\geq 0} \|M(t,\cdot)\|_{L^{p}(\Omega;X)}.$$
(8.3.1)

Moreover, if M is continuous, then N can be chosen to be continuous as well.

*Proof.* We first consider the case where *M* becomes constant after some time T > 0. Then

$$\sup_{t \ge 0} \|M(t, \cdot)\|_{L^p(\Omega; X)} = \|M(T)\|_{L^p(\Omega; X)}.$$

Let  $(\xi_n)_{n\geq 1}$  be simple random variables such that  $\xi_n \to M(T)$  in  $L^p(\Omega; X)$ . Let  $M_n(t) = \mathbb{E}(\xi_n | \mathscr{F}_t)$  for  $t \geq 0$ . Then by Lemma 8.3.1

$$\left\|\sup_{t\geq 0}|N_n(t,\cdot)-N_m(t,\cdot)|\right\|_{L^p(\Omega;X)} \approx_{p,X} \left\||M_n(T,\cdot)-M_m(T,\cdot)|\right\|_{L^p(\Omega;X)} \to 0$$

as  $n, m \to \infty$ . Therefore,  $(N_n)_{n \ge 1}$  is a Cauchy sequence and hence converges to some N from the space  $L^p(\Omega; X(\mathcal{D}_b([0,\infty))))$ . Clearly,  $N(t, \cdot) = M(t)$  and (8.3.1) holds in the special case that M becomes constant after T > 0.

In the case *M* is general, for each T > 0 we can set  $M^T(t) = M(t \wedge T)$ . Then for each T > 0 we obtain a martingale field  $N^T$  as required. Since  $N^{T_1} = N^{T_2}$  on  $[0, T_1 \wedge T_2]$ , we can define a martingale field *N* by setting  $N(t, \cdot) = N^T(t, \cdot)$  on [0, T]. Finally, we note that

$$\lim_{T \to \infty} \sup_{t \ge 0} \|M^T(t)\|_{L^p(\Omega;X)} = \sup_{t \ge 0} \|M(t)\|_{L^p(\Omega;X)}.$$

Moreover, by Definition 8.2.1(iv) we have

$$\lim_{T \to \infty} \left\| \sup_{t \ge 0} |N^T(t, \cdot)| \right\|_{L^p(\Omega; X)} = \left\| \sup_{t \ge 0} |N(t, \cdot)| \right\|_{L^p(\Omega; X)},$$

Therefore the general case of (8.3.1) follows by taking limits.

Now let *M* be continuous, and let  $(M_n)_{n\geq 1}$  be as before. By the same argument as in the first part pf the proof we can assume that there exists T > 0 such that  $M_t = M_{t\wedge T}$  for all  $t \geq 0$ . By Theorem 4.3.1 there exists a unique decomposition  $M_n = M_n^c + M_n^d$  such that  $M_n^d$  is purely discontinuous and starts at zero and  $M_n^c$  has continuous paths a.s. Then by (4.3.1)

$$\|M(T) - M_n^c(T)\|_{L^p(\Omega;X)} \le \beta_{p,X} \|M(T) - M_n(T)\|_{L^p(\Omega;X)} \to 0.$$

Since  $M_n^c$  takes values in a finite dimensional subspace of *X* we can define a martingale field  $N_n$  by  $N_n(t, \omega, s) = M_n^c(t, \omega)(s)$ . Now by Lemma 8.3.1

$$\left\|\sup_{0\leq t\leq T}|N_n(t,\cdot)-N_m(t,\cdot)|\right\|_{L^p(\Omega;X)}\sim_{p,X}\left\||M_n^c(T,\cdot)-M_m^c(T,\cdot)|\right\|_{L^p(\Omega;X)}\to 0.$$

Therefore,  $(N_n)_{n\geq 1}$  is a Cauchy sequence and hence converges to some *N* from the space  $L^p(\Omega; X(C_b([0,\infty))))$ . Analogously to the first part of the proof,  $N(t, \cdot) = M(t)$  for all  $t \geq 0$ .

*Remark* 8.3.3. Note that due to the construction of N we have that  $\Delta M_{\tau}(s) = \Delta N(\cdot, s)_{\tau}$  for any stopping time  $\tau$  and almost any  $s \in S$ . Indeed, let  $(M_n)_{n \ge 1}$  and  $(N_n)_{n \ge 1}$  be as in the proof of Theorem 8.3.2. Then on the one hand

$$\begin{split} \|\Delta M_{\tau} - \Delta (M_n)_{\tau}\|_{L^p(\Omega;X)} &\leq \left\| \sup_{0 \leq t \leq T} \|M(t) - M_n(t)\|_X \right\|_{L^p(\Omega)} \\ &\approx_p \|M(T) - M_n(T)\|_{L^p(\Omega;X)} \to 0, \ n \to \infty. \end{split}$$

On the other hand

$$\begin{split} \|\Delta N_{\tau} - \Delta(N_n)_{\tau}\|_{L^p(\Omega;X)} &\leq \big\| \sup_{0 \leq t \leq T} |N(t) - N_n(t)| \big\|_{L^p(\Omega;X)} \\ &\approx_{p,X} \big\| |N(T) - N_n(T)| \big\|_{L^p(\Omega;X)} \to 0, \quad n \to \infty. \end{split}$$

Since  $||M_n(t) - N_n(t, \cdot)||_{L^p(\Omega;X)} = 0$  for all  $n \ge 0$ , we have that by the limiting argument  $||\Delta M_\tau - \Delta N_\tau(\cdot)||_{L^p(\Omega;X)} = 0$ , so the desired follows from Definition 8.2.1(*i*).

One could hope there is a more elementary approach to derive continuity of N in the case M is continuous: if the filtration  $\tilde{\mathbb{F}} := (\tilde{\mathscr{F}}_t)_{t \ge 0}$  is generated by M, then M(s) is  $\tilde{\mathbb{F}}$ -adapted for a.e.  $s \in S$ , and one might expect that M has a continuous version. Unfortunately, this is not true in general as follows from the next example.

**Example 8.3.4.** There exists a continuous martingale  $M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$ , a filtration  $\widetilde{\mathbb{F}} = (\widetilde{\mathscr{F}}_t)_{t \ge 0}$  generated by M and all  $\mathbb{P}$ -null sets, and a purely discontinuous nonzero  $\widetilde{\mathbb{F}}$ -martingale  $N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$ . Let  $W : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a Brownian motion,  $L : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be a Poisson process such that W and L are independent. Let  $\mathbb{F} = (\mathscr{F}_t)_{t \ge 0}$  be the filtration generated by W and L. Let  $\sigma$  be an  $\mathbb{F}$ -stopping time defined as follows

$$\sigma = \inf\{u \ge 0 : \Delta L_u \neq 0\}.$$

Let us define

$$M := \int \mathbf{1}_{[0,\sigma]} \, \mathrm{d}W = W^{\sigma}.$$

Then *M* is a martingale. Let  $\widetilde{\mathbb{F}} := (\widetilde{\mathscr{F}}_t)_{t \ge 0}$  be generated by *M*. Note that  $\widetilde{\mathscr{F}}_t \subset \mathscr{F}_t$  for any  $t \ge 0$ . Define a random variable

 $\tau = \inf\{t \ge 0 : \exists u \in [0, t] \text{ such that } M \text{ is a constant on } [u, t]\}.$ 

Then  $\tau = \sigma$  a.s. Moreover,  $\tau$  is a  $\tilde{\mathbb{F}}$ -stopping time since for each  $u \ge 0$ 

$$\mathbb{P}\{\tau = u\} = \mathbb{P}\{\sigma = u\} = \mathbb{P}\{\Delta L_u^\sigma \neq 1\} \le \mathbb{P}\{\Delta L_u \neq 1\} = 0,$$

and hence

$$\{\tau \le u\} = \{\tau < u\} \cup \{\tau = u\} \subset \widetilde{\mathscr{F}}_u.$$

Therefore  $N : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  defined by

$$N_t := \mathbf{1}_{[\tau,\infty)}(t) - t \wedge \tau \quad t \ge 0,$$

is an  $\tilde{\mathbb{F}}$ -martingale since it is  $\tilde{\mathbb{F}}$ -measurable and since  $N_t = (L_t - t)^{\sigma}$  a.s. for each  $t \ge 0$ , hence for each  $u \in [0, t]$ 

$$\mathbb{E}(N_t|\widetilde{\mathscr{F}}_u) = \mathbb{E}(\mathbb{E}(N_t|\mathscr{F}_u)|\widetilde{\mathscr{F}}_u) = \mathbb{E}(\mathbb{E}((L_t - t)^{\sigma}|\mathscr{F}_u)|\widetilde{\mathscr{F}}_u) = (L_u - u)^{\sigma} = N_u$$

due to the fact that  $t \mapsto L_t - t$  is an  $\tilde{\mathbb{F}}$ -measurable  $\mathbb{F}$ -martingale (see [93, Problem 1.3.4]). But  $(N_t)_{t\geq 0}$  is not continuous since  $(L_t)_{t\geq 0}$  is not continuous.

### 8.4. MAIN RESULT

Theorem 8.1.1 will be a consequence of the following more general result.

**Theorem 8.4.1.** Let X be a UMD Banach function space over a  $\sigma$ -finite measure space  $(S, \Sigma, \mu)$  and let  $p \in (1, \infty)$ . Let  $M : \mathbb{R}_+ \times \Omega \to X$  be a local  $L^p$ -martingale with respect to  $(\mathscr{F}_t)_{t\geq 0}$  and assume  $M(0, \cdot) = 0$ . Then there exists a mapping  $N : \mathbb{R}_+ \times \Omega \times S \to \mathbb{R}$  such that

- (1) for all  $t \ge 0$  and a.a.  $\omega \in \Omega$ ,  $N(t, \omega, \cdot) = M(t, \omega)$ ,
- (2) N is a local martingale field,
- (3) the following estimate holds

$$\|\sup_{t\geq 0} |N(t,\cdot,\cdot)|\|_{L^{p}(\Omega;X)} \approx_{p,X} \|\sup_{t\geq 0} \|M(t,\cdot)\|_{X}\|_{L^{p}(\Omega)} \approx_{p,X} \|[N]_{\infty}^{1/2}\|_{L^{p}(\Omega;X)}.$$
 (8.4.1)

To prove Theorem 8.4.1 we first prove a completeness result.

**Proposition 8.4.2.** Let X be a Banach function space over a  $\sigma$ -finite measure space S,  $1 \le p < \infty$ . Let

$$\begin{split} \mathrm{MQ}^p(X) &:= \{N: \mathbb{R}_+ \times \Omega \times S \to \mathbb{R}: N \text{ is a martingale field,} \\ N_0(s) &= 0 \; \forall s \in S, \; and \; \|N\|_{\mathrm{MQ}^p(X)} < \infty \}, \end{split}$$

where

$$\|N\|_{\mathrm{MQ}^{p}(X)} := \|[N]_{\infty}^{1/2}\|_{L^{p}(\Omega;X)}.$$
(8.4.2)

Then  $(MQ^p(X), \|\cdot\|_{MQ^p(X)})$  is a Banach space. Moreover, if  $N_n \to N$  in  $MQ^p$ , then there exists a subsequence  $(N_{n_k})_{k\geq 1}$  such that pointwise a.e. in *S*, we have  $N_{n_k} \to N$  in  $L^1(\Omega; \mathcal{D}_b([0,\infty)))$ .

*Proof.* Let us first check that  $MQ^p(X)$  is a normed vector space. For this only the triangle inequality requires some comments. By the well-known estimate for local martingales *M*, *N* (see [89, Theorem 26.6(iii)]) we have that a.s.

$$[M+N]_{t} = [M]_{t} + 2[M,N]_{t} + [N]_{t}$$
  

$$\leq [M]_{t} + 2[M]_{t}^{1/2}[N]_{t}^{1/2} + [N]_{t} = ([M]_{t}^{1/2} + [N]_{t}^{1/2})^{2}, \qquad (8.4.3)$$

Therefore,  $[M+N]_t^{1/2} \le [M]_t^{1/2} + [N]_t^{1/2}$  a.s. for all  $t \in [0,\infty]$ .

Let  $(N_k)_{k\geq 1}$  be such that  $\sum_{k\geq 1} ||N_k||_{MQ^p(X)} < \infty$ . It suffices to show that  $\sum_{k\geq 1} N_k$  converges in  $MQ^p(X)$ . Observe that by monotone convergence in  $\Omega$  and Jensen's inequality applied to  $||\cdot||_X$  for any  $n > m \ge 1$  we have

$$\begin{split} \left\|\sum_{k=m+1}^{n} \mathbb{E}[N_{k}]_{\infty}^{1/2}\right\|_{X} &= \left\|\sum_{k=1}^{n} \mathbb{E}[N_{k}]_{\infty}^{1/2} - \sum_{k=1}^{m} \mathbb{E}[N_{k}]_{\infty}^{1/2}\right\|_{X} \\ &= \left\|\mathbb{E}\sum_{k=m+1}^{n} [N_{k}]_{\infty}^{1/2}\right\|_{X} \le \mathbb{E}\left\|\sum_{k=m+1}^{n} [N_{k}]_{\infty}^{1/2}\right\|_{X} \\ &= \left\|\sum_{k=m+1}^{n} [N_{k}]_{\infty}^{1/2}\right\|_{L^{1}(\Omega;X)} \le \left\|\sum_{k=m+1}^{n} [N_{k}]_{\infty}^{1/2}\right\|_{L^{p}(\Omega;X)} \\ &\le \sum_{k=m+1}^{n} \left\|[N_{k}]_{\infty}^{1/2}\right\|_{L^{p}(\Omega;X)} \to 0, \ m, n \to \infty, \end{split}$$
(8.4.4)
where the latter holds due to the fact that  $\sum_{k\geq 1} \left\| [N_k]_{\infty}^{1/2} \right\|_{L^p(\Omega;X)} < \infty$ . Thus  $\sum_{k=1}^n \mathbb{E}[N_k]_{\infty}^{1/2}$  converges in *X* as  $n \to \infty$ , where the corresponding limit coincides with its pointwise limit  $\sum_{k\geq 1} \mathbb{E}[N_k]_{\infty}^{1/2}$  by Remark 8.2.2. Therefore, since any element of *X* is finite a.s. by Definition 8.2.1, we can find  $S_0 \in \Sigma$  such that  $\mu(S_0^c) = 0$  and pointwise in  $S_0$ , we have  $\sum_{k\geq 1} \mathbb{E}[N_k]_{\infty}^{1/2} < \infty$ . Fix  $s \in S_0$ . In particular, we find that  $\sum_{k\geq 1} [N_k]_{\infty}^{1/2}$  converges in  $L^1(\Omega)$ . Moreover, since by the scalar Burkholder-Davis-Gundy inequalities  $\mathbb{E}\sup_{t\geq 0} |N_k(t,\cdot,s)| = \mathbb{E}[N_k(s)]_{\infty}^{1/2}$ , we also obtain that

$$N(\cdot, s) := \sum_{k \ge 1} N_k(\cdot, s) \text{ converges in } L^1(\Omega; \mathcal{D}_b([0, \infty)).$$
(8.4.5)

Let  $N(\cdot, s) = 0$  for  $s \notin S_0$ . Then *N* defines a martingale field. Moreover, by the scalar Burkholder-Davis-Gundy inequalities

$$\lim_{m \to \infty} \left[ \sum_{k=n}^{m} N_k(\cdot, s) \right]_{\infty}^{1/2} = \left[ \sum_{k=n}^{\infty} N_k(\cdot, s) \right]_{\infty}^{1/2}$$

in  $L^{1}(\Omega)$ . Therefore, by considering an a.s. convergent subsequence and by (8.4.3) we obtain

$$\left[\sum_{k=n}^{\infty} N_k(\cdot, s)\right]_{\infty}^{1/2} \le \sum_{k=n}^{\infty} [N_k(\cdot, s)]_{\infty}^{1/2}.$$
(8.4.6)

It remains to prove that  $N \in MQ^p(X)$  and  $N = \sum_{k\geq 1} N_k$  with convergence in  $MQ^p(X)$ . Let  $\varepsilon > 0$ . Choose  $n \in \mathbb{N}$  such that  $\sum_{k\geq n+1} ||N_k||_{MQ^p(X)} < \varepsilon$ . It follows from (8.4.4) that  $\mathbb{E} ||\sum_{k\geq 1} [N_k]_{\infty}^{1/2} ||_X < \infty$ , so  $\sum_{k\geq 1} [N_k]_{\infty}^{1/2}$  a.s. converges in *X*. Now by (8.4.6), the triangle inequality and Fatou's lemma, we obtain

$$\begin{split} \left\| \left[ \sum_{k \ge n+1} N_k \right]_{\infty}^{1/2} \right\|_{L^p(\Omega;X)} &\leq \left\| \sum_{k=n+1}^{\infty} [N_k]_{\infty}^{1/2} \right\|_{L^p(\Omega;X)} \\ &\leq \sum_{k=n+1}^{\infty} \left\| [N_k]_{\infty}^{1/2} \right\|_{L^p(\Omega;X)} \\ &\leq \liminf_{m \to \infty} \sum_{k=n+1}^{m} \left\| [N_k]_{\infty}^{1/2} \right\|_{L^p(\Omega;X)} < \varepsilon^p \end{split}$$

Therefore,  $N \in MQ^p(X)$  and  $||N - \sum_{k=1}^n N_k||_{MQ^p(X)} < \varepsilon$ .

For the proof of the final assertion assume that  $N_n \to N$  in  $MQ^p(X)$ . Choose a subsequence  $(N_{n_k})_{k\geq 1}$  such that  $||N_{n_k}-N||_{MQ^p(X)} \leq 2^{-k}$ . Then  $\sum_{k\geq 1} ||N_{n_k}-N||_{MQ^p(X)} < \infty$  and hence by (8.4.5) we see that pointwise a.e. in *S*, the series  $\sum_{k\geq 1} (N_{n_k}-N)$  converges in  $L^1(\Omega; \mathcal{D}_b([0,\infty)))$ . Therefore,

$$N_{n_k} \to N$$
 in  $L^1(\Omega; \mathscr{D}_b([0,\infty); X))$ 

as required.

For the proof of Theorem 8.4.1 we will need the following lemma presented in [55, Théorème 2].

**Lemma 8.4.3.** Let  $1 , <math>M : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  be an  $L^p$ -martingales. Let T > 0. For each  $n \ge 1$  define

$$R_n := \sum_{k=1}^n |M_{\frac{Tk}{n}} - M_{\frac{T(k-1)}{n}}|^2.$$

Then  $R_n$  converges to  $[M]_T$  in  $L^{p/2}$ .

*Proof of Theorem 8.4.1.* The existence of the local martingale field *N* together with the first estimate in (8.4.1) follows from Theorem 8.3.2. It remains to prove

$$\|\sup_{t\geq 0} \|M(t,\cdot)\|_X \|_{L^p(\Omega)} \approx_{p,X} \|[N]_{\infty}^{1/2}\|_{L^p(\Omega;X)}.$$
(8.4.7)

Due to Definition 8.2.1(iv) it suffices to prove the above norm equivalence in the case *M* and *N* becomes constant after some fixed time *T*.

Step 1: The finite dimensional case. Assume that *M* takes values in a finite dimensional subspace *Y* of *X* and that the right hand side of (8.4.7) is finite. Then we can write  $N(t,s) = M(t)(s) = \sum_{j=1}^{n} M_j(t)x_j(s)$ , where each  $M_j$  is a scalar-valued martingale with  $M_j(T) \in L^p(\Omega)$  and  $x_1, \ldots, x_n \in X$  form a basis of *Y*. Note that for any  $c_1, \ldots, c_n \in L^p(\Omega)$  we have that

$$\left\|\sum_{j=1}^{n} c_{j} x_{j}\right\|_{L^{p}(\Omega; X)} \approx_{p, Y} \sum_{j=1}^{n} \|c_{j}\|_{L^{p}(\Omega)}.$$
(8.4.8)

Fix  $m \ge 1$ . Then by (8.1.3) and Doob's maximal inequality

$$\|\sup_{t\geq 0} \|M(t,\cdot)\|_{X} \|_{L^{p}(\Omega)} \approx_{p} \|M(T,\cdot)\|_{L^{p}(\Omega;X)}$$

$$= \left\|\sum_{i=1}^{m} M_{\frac{Ti}{m}} - M_{\frac{T(i-1)}{m}} \right\|_{L^{p}(\Omega;X)}$$

$$\approx_{p,X} \left\| \left(\sum_{i=1}^{m} |M_{\frac{Ti}{m}} - M_{\frac{T(i-1)}{m}}|^{2}\right)^{\frac{1}{2}} \right\|_{L^{p}(\Omega;X)},$$
(8.4.9)

and by (8.4.8) and Lemma 8.4.3 the right hand side of (8.4.9) converges to

$$\|[M]_{\infty}^{1/2}\|_{L^{p}(\Omega;X)} = \|[N]_{\infty}^{1/2}\|_{L^{p}(\Omega;X)}.$$

Step 2: Reduction to the case where *M* takes values in a finite dimensional subspace of *X*. Let  $M(T) \in L^p(\Omega; X)$ . Then we can find simple functions  $(\xi_n)_{n\geq 1}$  in  $L^p(\Omega; X)$ such that  $\xi_n \to M(T)$ . Let  $M_n(t) = \mathbb{E}(\xi_n | \mathscr{F}_t)$  for all  $t \geq 0$  and  $n \geq 1$ ,  $(N_n)_{n\geq 1}$  be the corresponding martingale fields. Then each  $M_n$  takes values in a finite dimensional subspace  $X_n \subseteq X$ , and hence by Step 1

$$\|\sup_{t\geq 0} \|M_n(t,\cdot) - M_m(t,\cdot)\|_X \|_{L^p(\Omega)} \approx_{p,X} \|[N_n - N_m]_{\infty}^{1/2}\|_{L^p(\Omega;X)}$$

for any  $m, n \ge 1$ . Therefore since  $(\xi_n)_{n\ge 1}$  is Cauchy in  $L^p(\Omega; X)$ ,  $(N_n)_{n\ge 1}$  converges to some N in  $MQ^p(X)$  by the first part of Proposition 8.4.2.

Let us show that *N* is the desired local martingale field. Fix  $t \ge 0$ . We need to show that  $N(\cdot, t, \cdot) = M_t$  a.s. on  $\Omega$ . First notice that by the second part of Proposition 8.4.2 there exists a subsequence of  $(N_n)_{n\ge 1}$  which we will denote by  $(N_n)_{n\ge 1}$  as well such that  $N_n(\cdot, t, \sigma) \rightarrow N(\cdot, t, \sigma)$  in  $L^1(\Omega)$  for a.e.  $\sigma \in S$ . On the other hand by Jensen's inequality

$$\left\|\mathbb{E}|N_n(\cdot,t,\cdot) - M_t|\right\|_X = \left\|\mathbb{E}|M_n(t) - M(t)|\right\|_X \le \mathbb{E}\|M_n(t) - M(t)\|_X \to 0, \quad n \to \infty.$$

Hence  $N_n(\cdot, t, \cdot) \to M_t$  in  $X(L^1(\Omega))$ , and thus by Remark 8.2.2 in  $L^0(S; L^1(\Omega))$ . Therefore we can find a subsequence of  $(N_n)_{n\geq 1}$  (which we will again denote by  $(N_n)_{n\geq 1}$ ) such that  $N_n(\cdot, t, \sigma) \to M_t(\sigma)$  in  $L^1(\Omega)$  for a.e.  $\sigma \in S$  (here we use that fact that  $\mu$  is  $\sigma$ -finite), so  $N(\cdot, t, \cdot) = M_t$  a.s. on  $\Omega \times S$ , and consequently by Definition 8.2.1(*iii*),  $N(\omega, t, \cdot) = M_t(\omega)$  for a.a.  $\omega \in \Omega$ . Thus (8.4.7) follows by letting  $n \to \infty$ .

Step 3: Reduction to the case where the left-hand side of (8.4.7) is finite. Assume that the left-hand side of (8.4.7) is infinite, but the right-hand side is finite. Since *M* is a local  $L^p$ -martingale we can find a sequence of stopping times  $(\tau_n)_{n\geq 1}$  such that  $\tau_n \uparrow \infty$  and  $\|M_T^{\tau_n}\|_{L^p(\Omega;X)} < \infty$  for each  $n \ge 1$ . By the monotone convergence theorem and Definition 8.2.1(*iv*)

$$\begin{split} \| [N]_{\infty}^{1/2} \|_{L^{p}(\Omega;X)} &= \lim_{n \to \infty} \| [N^{\tau_{n}}]_{\infty}^{1/2} \|_{L^{p}(\Omega;X)} \eqsim_{p,X} \limsup_{n \to \infty} \| M_{T}^{\tau_{n}} \|_{L^{p}(\Omega;X)} \\ &= \lim_{n \to \infty} \| M_{T}^{\tau_{n}} \|_{L^{p}(\Omega;X)} = \lim_{n \to \infty} \left\| \sup_{0 \le t \le T} \| M_{t}^{\tau_{n}} \|_{X} \right\|_{L^{p}(\Omega)} \\ &= \left\| \sup_{0 \le t \le T} \| M_{t} \|_{X} \right\|_{L^{p}(\Omega)} = \infty \end{split}$$

and hence the right-hand side of (8.4.7) is infinite as well.

We use an extrapolation argument to extend part of Theorem 8.4.1 to  $p \in (0, 1]$  in the continuous-path case.

**Corollary 8.4.4.** Let X be a UMD Banach function space over a  $\sigma$ -finite measure space and let  $p \in (0,\infty)$ . Let M be a continuous local martingale  $M : \mathbb{R}_+ \times \Omega \to X$  with  $M(0,\cdot) = 0$ . Then there exists a continuous local martingale field  $N : \mathbb{R}_+ \times \Omega \times S \to \mathbb{R}$  such that for a.a.  $\omega \in \Omega$ , all  $t \ge 0$ , and a.a.  $s \in S$ ,  $N(t, \omega, \cdot) = M(t, \omega)(s)$  and

$$\|\sup_{t\geq 0} \|M(t,\cdot)\|_X \|_{L^p(\Omega)} \approx_{p,X} \|[N]_{\infty}^{1/2}\|_{L^p(\Omega;X)}.$$
(8.4.10)

*Proof.* By a stopping time argument we can reduce to the case where  $||M(t, \omega)||_X$  is uniformly bounded in  $t \in \mathbb{R}_+$  and  $\omega \in \Omega$  and M becomes constant after a fixed time T. Now the existence of N follows from Theorem 8.4.1 and it remains to prove (8.4.10) for  $p \in (0, 1]$ . For this we can use a classical argument due to Lenglart. Indeed, for both estimates we can apply [106] or [156, Proposition IV.4.7] to the continuous increasing processes  $Y, Z : \mathbb{R}_+ \times \Omega \to \mathbb{R}_+$  given by

$$Y_u = \mathbb{E} \sup_{t \in [0,u]} \|M(t, \cdot)\|_X,$$

$$Z_u = \| s \mapsto [N(\cdot, \cdot, s)]_u^{1/2} \|_X,$$

where  $q \in (1,\infty)$  is a fixed number. Then by (8.4.1) for any bounded stopping time  $\tau$ , we have

$$\mathbb{E}Y_{\tau}^{q} = \sup_{t \ge 0} \|M(t \wedge \tau, \cdot)\|_{X}^{q} \approx_{q, X} \mathbb{E}\|s \mapsto [N(\cdot \wedge \tau, \cdot, s)]_{\infty}^{1/2}\|_{X}^{q}$$

$$\stackrel{(*)}{=} \mathbb{E}\|s \mapsto [N(\cdot, \cdot, s)]_{\tau}^{1/2}\|_{X}^{q} = \mathbb{E}Z_{\tau}^{q},$$

where we used [89, Theorem 17.5] in (\*). Now (8.4.10) for  $p \in (0, q)$  follows from [106] or [156, Proposition IV.4.7].

As we saw in Theorem 8.3.2, continuity of *M* implies pointwise continuity of the corresponding martingale filed *N*. The following corollaries of Theorem 8.4.1 are devoted to proving the same type of assertions concerning pure discontinuity, quasi-left continuity, and having accessible jumps.

Let  $\tau$  be a stopping time. Then  $\tau$  is called *predictable* if there exists a sequence of stopping times  $(\tau_n)_{n\geq 1}$  such that  $\tau_n < \tau$  a.s. on  $\{\tau > 0\}$  for each  $n \geq 1$  and  $\tau_n \nearrow \tau$  a.s. A càdlàg process  $V : \mathbb{R}_+ \times \Omega \to X$  is called to have *accessible jumps* if there exists a sequence of predictable stopping times  $(\tau_n)_{n\geq 1}$  such that  $\{t \in \mathbb{R}_+ : \Delta V \neq 0\} \subset \{\tau_1, ..., \tau_n, ...\}$  a.s.

**Corollary 8.4.5.** Let X be a UMD function space over a measure space  $(S, \Sigma, \mu)$ ,  $1 , <math>M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous  $L^p$ -martingale with accessible jumps. Let N be the corresponding martingale field. Then  $N(\cdot, s)$  is a purely discontinuous martingale with accessible jumps for a.e.  $s \in S$ .

*Proof of Corollary* 8.4.5. Without loss of generality we can assume that there exists  $T \ge 0$  such that  $M_t = M_T$  for all  $t \ge T$ , and that  $M_0 = 0$ . Since M has accessible jumps, there exists a sequence of predictable stopping times  $(\tau_n)_{n\ge 1}$  such that a.s.

$$\{t \in \mathbb{R}_+ : \Delta M \neq 0\} \subset \{\tau_1, \dots, \tau_n, \dots\}.$$

For each  $m \ge 1$  define a process  $M^m : \mathbb{R}_+ \times \Omega \to X$  in the following way:

$$M^m(t):=\sum_{n=1}^m \Delta M_{\tau_n} \mathbf{1}_{[0,t]}(\tau_n), \ t\geq 0.$$

Note that  $M^m$  is a purely discontinuous  $L^p$ -martingale with accessible jumps by Lemma 2.4.5. Let  $N^m$  be the corresponding martingale field. Then  $N^m(\cdot, s)$  is a purely discontinuous martingale with accessible jumps for almost any  $s \in S$  due to Remark 8.3.3. Moreover, for any  $m \ge \ell \ge 1$  and any  $t \ge 0$  we have that a.s.  $[N^m(\cdot, s)]_t \ge [N^\ell(\cdot, s)]_t$ . Define  $F : \mathbb{R}_+ \times \Omega \times S \to \mathbb{R}_+ \cup \{+\infty\}$  in the following way:

$$F(t,\cdot,s) := \lim_{m \to \infty} [N^m(\cdot,s)]_t, \quad s \in S, t \ge 0.$$

Note that  $F(\cdot, \cdot, s)$  is a.s. finite for almost any  $s \in S$ . Indeed, by Theorem 8.4.1 and 3.3.17 we have that for any  $m \ge 1$ 

$$\left\| [N^m]^{1/2}_\infty \right\|_{L^p(\Omega;X)} \sim_{p,X} \|M^m(T,\cdot)\|_{L^p(\Omega;X)} \leq \beta_{p,X} \|M(T,\cdot)\|_{L^p(\Omega;X)},$$

so by Definition 8.2.1(*iv*),  $F(\cdot, \cdot, s)$  is a.s. finite for almost any  $s \in S$  and

$$\|F_{\infty}^{1/2}\|_{L^{p}(\Omega;X)} = \|F_{T}^{1/2}\|_{L^{p}(\Omega;X)} = \lim_{m \to \infty} \|[N^{m}]_{T}^{1/2}\|_{L^{p}(\Omega;X)}$$
$$\lesssim_{p,X} \limsup_{m \to \infty} \|M^{m}(T,\cdot)\|_{L^{p}(\Omega;X)} \lesssim_{p,X} \|M(T,\cdot)\|_{L^{p}(\Omega;X)}.$$

Moreover, for almost any  $s \in S$  we have that  $F(\cdot, \cdot, s)$  is pure jump and

$$\{t \in \mathbb{R}_+ : \Delta F \neq 0\} \subset \{\tau_1, \dots, \tau_n, \dots\}.$$

Therefore to this end it suffices to show that F(s) = [N(s)] a.s. on  $\Omega$  for a.e.  $s \in S$ . Note that by Definition 8.2.1(*iv*),

$$\|(F - [N^m])^{1/2}(\infty)\|_{L^p(\Omega;X)} \to 0, \quad m \to \infty$$
 (8.4.11)

so by Theorem 8.4.1  $(M^m(T))_{m\geq 1}$  is a Cauchy sequence in  $L^p(\Omega; X)$ . Let  $\xi$  be its limit,  $M^0: \mathbb{R}_+ \times \Omega \to X$  be a martingale such that  $M^0(t) = \mathbb{E}(\xi | \mathscr{F}_t)$  for all  $t \geq 0$ . Then by Proposition 2.2.16  $M^0$  is purely discontinuous. Moreover, for any stopping time  $\tau$  a.s.

$$\Delta M_{\tau}^{0} = \lim_{m \to \infty} \Delta M_{\tau}^{m} = \lim_{m \to \infty} \Delta M_{\tau} \mathbf{1}_{\{\tau_{1}, \dots, \tau_{m}\}}(\tau) = \Delta M_{\tau},$$

where the latter holds since the set { $\tau_1, ..., \tau_n, ...$ } exhausts the jump times of M. Therefore  $M = M^0$  since both M and  $M^0$  are purely discontinuous with the same jumps, and hence [N] = F (where  $F(s) = [M^0(s)]$  by (8.4.11)). Consequently  $N(\cdot, \cdot, s)$  is purely discontinuous with accessible jumps for almost all  $s \in S$ .

*Remark* 8.4.6. Note that the proof of Corollary 8.4.5 also implies that  $M_t^m \to M_t$  in  $L^p(\Omega; X)$  for each  $t \ge 0$ .

A càdlàg process  $V : \mathbb{R}_+ \times \Omega \to X$  is called *quasi-left continuous* if  $\Delta V_\tau = 0$  a.s. for any predictable stopping time  $\tau$ .

**Corollary 8.4.7.** Let X be a UMD function space over a measure space  $(S, \Sigma, \mu)$ ,  $1 , <math>M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous quasi-left continuous  $L^p$ -martingale. Let N be the corresponding martingale field. Then  $N(\cdot, s)$  is a purely discontinuous quasi-left continuous martingale for a.e.  $s \in S$ .

*Proof.* Without loss of generality we can assume that there exists  $T \ge 0$  such that  $M_t = M_T$  for all  $t \ge T$ , and that  $M_0 = 0$ . Let  $\mu$  be a random measure defined on  $\mathbb{R}_+ \times X$  in the following way

$$\mu(A \times B) = \sum_{t \ge 0} \mathbf{1}_A(t) \mathbf{1}_{B \setminus \{0\}}(\Delta M_t),$$

where  $A \subset \mathbb{R}_+$  is a Borel set, and  $B \subset X$  is a ball. For each  $k, \ell \ge 1$  we define a stopping time  $\tau_{k,\ell}$  as follows

$$\tau_{k,\ell} = \inf\{t \in \mathbb{R}_+ : \#\{u \in [0,t] : \|\Delta M_u\|_X \in [1/k,k]\} = \ell\}.$$

Since *M* has càdlàg trajectories,  $\tau_{k,\ell}$  is a.s. well-defined and takes its values in  $[0,\infty]$ . Moreover,  $\tau_{k,\ell} \to \infty$  for each  $k \ge 1$  a.s. as  $\ell \to \infty$ , so we can find a subsequence  $(\tau_{k_n,\ell_n})_{n\ge 1}$  such that  $k_n \ge n$  for each  $n \ge 1$  and  $\inf_{m\ge n} \tau_{k_m,\ell_m} \to \infty$  a.s. as  $n \to \infty$ . Define  $\tau_n = \inf_{m\ge n} \tau_{k_m,\ell_m}$  and define  $M^n := (\mathbf{1}_{[0,\tau_n]}\mathbf{1}_{B_n}) \star \bar{\mu}$ , where  $\bar{\mu} = \mu - \nu$  is such that  $\nu$  is a compensator of  $\mu$  and  $B_n = \{x \in X : \|x\| \in [1/n, n]\}$ . Then  $M^n$  is a purely discontinuous quasi-left continuous martingale by Lemma 2.8.1. Moreover, a.s.

$$\Delta M_t^n = \Delta M_t \mathbf{1}_{[0,\tau_n]}(t) \mathbf{1}_{[1/n,n]}(\|\Delta M_t\|), \quad t \ge 0.$$

so by Theorem 3.3.17  $M^n$  is an  $L^p$ -bounded martingale.

The rest of the proof is analogous to the proof of Corollary 8.4.5 and uses the fact that  $\tau_n \rightarrow \infty$  monotonically a.s.

**Corollary 8.4.8.** Let X be a UMD Banach function space,  $1 , <math>M : \mathbb{R}_+ \times \Omega \to X$  be an  $L^p$ -martingale. Let N be the corresponding martingale field. Let  $M = M^c + M^q + M^a$ be the canonical decomposition,  $N^c$ ,  $N^q$ , and  $N^a$  be the corresponding martingale fields. Then  $N(s) = N^c(s) + N^q(s) + N^a(s)$  is the canonical decomposition of N(s) for a.e.  $s \in$ S. In particular, if  $M_0 = 0$  a.s., then M is continuous, purely discontinuous quasi-left continuous, or purely discontinuous with accessible jumps if and only if N(s) is so for a.e.  $s \in S$ .

*Proof.* The first part follows from Theorem 8.3.2, Corollary 8.4.5, and Corollary 8.4.7 and the fact that  $N(s) = N^c(s) + N^q(s) + N^a(s)$  is then a canonical decomposition of a local martingale N(s) which is unique due to Remark 2.4.24. Let us show the second part. One direction follows from Theorem 8.3.2, Corollary 8.4.5, and Corollary 8.4.7. For the other direction assume that N(s) is continuous for a.e.  $s \in S$ . Let  $M = M^c + M^q + M^a$  be the canonical decomposition,  $N^c$ ,  $N^q$ , and  $N^a$  be the corresponding martingale fields of  $M^c$ ,  $M^q$ , and  $M^a$ . Then by the first part of the theorem and the uniqueness of the canonical decomposition (see Remark 2.4.24) we have that for a.e.  $s \in S$ ,  $N^q(s) = N^a(s) = 0$ , so  $M^q = M^a = 0$ , and hence M is continuous. The proof for the case of pointwise purely discontinuous quasileft continuous N or pointwise purely discontinuous N with accessible jumps is similar.

# BURKHOLDER–DAVIS–GUNDY INEQUALITIES AND STOCHASTIC INTEGRATION IN GENERAL UMD BANACH SPACES

This chapter is based on the paper *Burkholder–Davis–Gundy inequalities in UMD Banach spaces* by Ivan Yaroslavtsev, see [187].

In this chapter we prove Burkholder–Davis–Gundy inequalities for a general martingale M with values in a UMD Banach space X. Assuming that  $M_0 = 0$ , we show that the following two-sided inequality holds for all  $1 \le p < \infty$ :

$$\mathbb{E} \sup_{0 \le s \le t} \|M_s\|^p \sim_{p,X} \mathbb{E}\gamma([[M]]_t)^p, \quad t \ge 0.$$
(\*)

Here  $\gamma([[M]]_t)$  is the L<sup>2</sup>-norm of the unique Gaussian measure on X having

$$[[M]]_t(x^*, y^*) := [\langle M, x^* \rangle, \langle M, y^* \rangle]_t$$

*as its covariance bilinear form. This extends to general UMD spaces Theorem 8.1.1, where a pointwise version of*  $(\star)$  *was proved for UMD Banach functions spaces X.* 

We show that for continuous martingales, (\*) holds for all 0 , and that for purely discontinuous martingales the right-hand side of (\*) can be expressed more explicitly in terms of the jumps of*M*. For martingales with independent increments, (\*) is shown to hold more generally in reflexive Banach spaces*X*with finite cotype. In the converse direction, we show that the validity of (\*) for arbitrary martingales implies the UMD property for*X*.

As an application we prove various Itô isomorphisms for vector-valued stochastic integrals with respect to general martingales, which extends earlier results by van Neerven, Veraar, and Weis for vector-valued stochastic integrals with respect to a Brownian motion. We also provide Itô isomorphisms for vector-valued stochastic integrals with respect to compensated Poisson and general random measures.

<sup>2010</sup> Mathematics Subject Classification. Primary: 60G44, 60H05, 28C20 Secondary: 60G57, 46B42. Key words and phrases. Burkholder–Davis–Gundy inequalities, UMD Banach spaces, Itô isomorphism, Gaussian measures, random measures, Banach function spaces.

## **9.1.** INTRODUCTION

In the celebrated paper [40], Burkholder, Davis, and Gundy proved that if  $M = (M_t)_{t\geq 0}$  is a real-valued martingale satisfying  $M_0 = 0$ , then for all  $1 \leq p < \infty$  and  $t \geq 0$  one has the two-sided inequality

$$\mathbb{E}\sup_{0\le s\le t}|M_s|^p \approx_p \mathbb{E}[M]_t^{\frac{p}{2}},\tag{9.1.1}$$

where [M] is the quadratic variation of M, i.e.,

$$[M]_t := \mathbb{P} - \lim_{\text{mesh}(\pi) \to 0} \sum_{n=1}^N |M(t_n) - M(t_{n-1})|^2,$$
(9.1.2)

where the limit in probability is taken over partitions  $\pi = \{0 = t_0 < ... < t_N = t\}$ whose mesh approaches 0. Later, Burkholder [36, 38] and Kallenberg and Sztencel [90] extended (9.1.1) to Hilbert space-valued martingales (see also [115]). They showed that if *M* is a martingale with values in a Hilbert space *H* satisfying  $M_0 = 0$ , then for all  $1 \le p < \infty$  and  $t \ge 0$  one has

$$\mathbb{E}\sup_{0\le s\le t} \|M_s\|^p \approx_p \mathbb{E}[M]_t^{\frac{p}{2}},\tag{9.1.3}$$

where the quadratic variation [*M*] is defined as in (9.1.2) with absolute values replaced by norms in *H*. A further result along these lines was obtained in Chapter 8. There it is shown that if *M* is an  $L^p$ -bounded martingale,  $1 , with <math>M_0 = 0$ , that takes values in a UMD Banach function space *X* over a measure space (*S*,  $\Sigma$ ,  $\mu$ ) (see Section 8.2 for the definition), then for all  $t \ge 0$ :

$$\mathbb{E}\sup_{0\leq s\leq t} \|M_s(\sigma)\|^p \approx_{p,X} \mathbb{E}\|[M(\sigma)]_t^{\frac{1}{2}}\|^p, \qquad (9.1.4)$$

where the quadratic variation  $[M(\sigma)]_t$  is considered pointwise in  $\sigma \in S$ . Although this inequality seems to be particularly useful from a practical point of view, it does not give any hint how to work with a general Banach space since not every (UMD) Banach space has a Banach function space structure (e.g. noncommutative  $L^q$ -spaces).

Therefore the following natural question is rising up. *Given a Banach space X. Is there an analogue of* (9.1.3) *for a general X-valued local martingale M and how then should the right-hand side of* (9.1.3) *look like*? In the current article we present the following complete solution to this problem for local martingales M with values in a UMD Banach space X.

**Theorem 9.1.1.** Let X be a UMD Banach space. Then for any local martingale M:  $\mathbb{R}_+ \times \Omega \rightarrow X$  with  $M_0 = 0$  and any  $t \ge 0$  the covariation bilinear form  $[[M]]_t$  is well-defined and bounded almost surely, and for all  $1 \le p < \infty$  we have

$$\mathbb{E}\sup_{0\le s\le t} \|M_s\|^p \approx_{p,X} \mathbb{E}\gamma([[M]]_t)^p.$$
(9.1.5)

Here  $\gamma(V)$ , where  $V : X^* \times X^* \to \mathbb{R}$  is a given nonnegative symmetric bilinear form, is the  $L^2$ -norm of an *X*-valued Gaussian random variable  $\xi$  with

$$\mathbb{E}\langle\xi, x^*\rangle^2 = V(x^*, x^*), \ x^* \in X^*.$$

We call  $\gamma(V)$  the Gaussian characteristic of V (see Section 9.3).

Let us explain briefly the main steps of the proof of Theorem 9.1.1. This discussion will also clarify the meaning of the term on the right-hand side, which is equivalent to the right-hand side of (9.1.3) if X is a Hilbert space, and of (9.1.4) (up to a multiplicative constant) if X is a UMD Banach function space.

In Section 9.2 we start by proving the discrete-time version of Theorem 9.1.1, which takes the following simple form

$$\mathbb{E}\sup_{1\le m\le N}\left\|\sum_{n=1}^{m}d_{n}\right\|^{p} \approx_{p,X} \mathbb{E}\left(\mathbb{E}_{\gamma}\left\|\sum_{n=1}^{N}\gamma_{n}d_{n}\right\|^{2}\right)^{\frac{p}{2}},\tag{9.1.6}$$

where  $(d_n)_{n=1}^N$  is an *X*-valued martingale difference sequence and  $(\gamma_n)_{n=1}^N$  is a sequence of independent standard Gaussian random variables defined on a probability space  $(\Omega_{\gamma}, \mathbb{P}_{\gamma})$ . (9.1.6) follows from a decoupling inequality due to Garling [61] and a martingale transform inequality due to Burkholder [35] (each of which holds if and only if *X* has the UMD property) together with the equivalence of Rademacher and Gaussian random sums with values in spaces with finite cotype due to Maurey and Pisier (see [117]).

Theorem 9.1.1 is derived from (9.1.6) by finite-dimensional approximation and discretization. This is a rather intricate procedure and depends on some elementary, but nevertheless important properties of a Gaussian characteristic  $\gamma(\cdot)$ . In particular in Section 9.3 we show that for a finite dimensional Banach space *X* there exists a proper continuous extension of the Gaussian characteristic to all (not necessarily nonnegative) symmetric bilinear forms  $V : X^* \times X^* \to \mathbb{R}$ , with the bound

$$(\gamma(V))^2 \lesssim_X \sup_{\|x^*\| \le 1} V(x^*, x^*).$$

Next, in Section 9.5, under the assumptions of Theorem 9.1.1 we show that M has a well-defined *covariation bilinear form*, i.e. for each  $t \ge 0$  and for almost all  $\omega \in \Omega$  there exists a symmetric bilinear form  $[[M]]_t(\omega) : X^* \times X^* \to \mathbb{R}$  such that for all  $x^*, y^* \in X^*$  one has

$$[[M]]_t(x^*, y^*) = [\langle M, x^* \rangle, \langle M, y^* \rangle]_t \text{ a.s.}$$

Next we prove that the bilinear form  $[[M]]_t(\omega)$  has a finite Gaussian characteristic  $\gamma([[M]]_t)$  for almost all  $\omega \in \Omega$ . After these preparations we prove Theorem 9.1.1. We also show that the UMD property is necessary for the conclusion of the theorem to hold true (see Subsection 9.7.3).

In Section 9.6 we develop three ramifications of our main result:

- if *M* is continuous, the conclusion of Theorem 9.1.1 holds for all 0 .
- if *M* is purely discontinuous, the theorem can be reformulated in terms of the jumps of *M*.
- if *M* has independent increments, the UMD assumption on *X* can be weakened to reflexivity and finite cotype.

The first two cases are particularly important in view of the fact that any UMD space-valued local martingale has a unique *Meyer-Yoeurp decomposition* into a sum of a continuous local martingale and a purely discontinuous local martingale (see [184, 185]).

A reasonable part of the chapter, namely Section 9.7, is devoted to applications of Theorem 9.1.1 and results related to Theorem 9.1.1. Let us outline some of them. In Subsection 9.7.1 we develop a theory of vector-valued stochastic integration. Our starting point is a result of van Neerven, Veraar, and Weis [126]. They proved that if  $W_H$  is a cylindrical Brownian motion in a Hilbert space H and  $\Phi : \mathbb{R}_+ \times \Omega \rightarrow$  $\mathscr{L}(H, X)$  is an elementary predictable process, then for all  $0 and <math>t \ge 0$  one has the two-sided inequality

$$\mathbb{E}\sup_{0\leq s\leq t}\left\|\int_{0}^{s}\Phi\,\mathrm{d}W_{H}\right\|^{p}\approx_{p,X}\mathbb{E}\|\Phi\|_{\gamma(L^{2}([0,t];H),X)}^{p}.$$
(9.1.7)

Here  $\|\Phi\|_{\gamma(L^2([0,t];H),X)}$  is the  $\gamma$ -radonifying norm of  $\Phi$  as an operator from a Hilbert space  $L^2([0,t];H)$  into X (see (2.9.1) for the definition); this norm coincides with the Hilbert-Schmidt norm given X is a Hilbert space. This result was extended to continuous local martingales in [175, 177].

Theorem 9.1.1 directly implies (9.1.7). More generally, if  $M = \int \Phi d\widetilde{M}$  for some *H*-valued martingale  $\widetilde{M}$  and elementary predictable process  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X)$ , then it follows from Theorem 9.1.1 that for all  $1 \le p < \infty$  and  $t \ge 0$  one has

$$\mathbb{E}\sup_{0\leq s\leq t} \left\| \int_0^s \Phi \,\mathrm{d}\widetilde{M} \right\|^p \approx_{p,X} \mathbb{E} \|\Phi q_{\widetilde{M}}^{1/2}\|_{\gamma(L^2(0,t;[\widetilde{M}]),X)}^p.$$
(9.1.8)

Here  $q_{\widetilde{M}}$  is the quadratic variation derivative of  $\widetilde{M}$  and  $\gamma(L^2(0, t; [\widetilde{M}]), X)$  is a suitable space of  $\gamma$ -radonifying operator associated with  $\widetilde{M}$  (see Subsection 9.7.1 for details). This represents a significant improvement of (9.1.7).

In Subsection 9.7.2 we apply our results to vector-valued stochastic integrals with respect to a compensated Poisson random measure  $\tilde{N}$ . We show that if N is a Poisson random measure on  $\mathbb{R}_+ \times J$  for some measurable space  $(J, \mathscr{J})$ , v is its compensator,  $\tilde{N} := N - v$  is the corresponding compensated Poisson random measure, then for any UMD Banach space X, for any elementary predictable  $F : J \times \mathbb{R}_+ \times \Omega \rightarrow X$ , and for any  $1 \le p < \infty$  one has that

$$\mathbb{E}\sup_{0\leq s\leq t}\left\|\int_{J\times[0,s]}Fd\widetilde{N}\right\|^{p} \approx_{p,X} \mathbb{E}\|F\|_{\gamma(L^{2}(J\times[0,t];N),X)}^{p}, \quad t\geq 0.$$
(9.1.9)

We also show that (9.1.9) holds if one considers a general quasi-left continuous random measure  $\mu$  instead of *N*.

In Subsection 9.7.4 we prove the following *martingale domination inequality*: for all local martingales *M* and *N* with values in a UMD Banach space *X* such that

$$||N_0|| \le ||M_0||$$
 a.s.

and

$$[\langle N, x^* \rangle]_{\infty} \leq [\langle M, x^* \rangle]_{\infty}$$
 almost surely, for all  $x^* \in X^*$ ,

for all  $1 \le p < \infty$  we have that

$$\mathbb{E}\sup_{t\geq 0}\|N_t\|^p \lesssim_{p,X} \mathbb{E}\sup_{t\geq 0}\|M_t\|^p.$$

This extends *weak differential subordination*  $L^p$ -estimates obtained in [184, 189] (which used to be known to hold only for 1 , see [146, 184, 189]).

Finally, in Section 9.8, we prove that for any UMD Banach function space *X* over a measure space  $(S, \Sigma, \mu)$ , that any *X*-valued local martingale *M* has a pointwise local martingale version  $M(\sigma)$ ,  $\sigma \in S$ , such that if  $1 \le p < \infty$ , then for  $\mu$ -almost all  $\sigma \in S$  one has

$$\mathbb{E}\sup_{0\leq s\leq t}\|M_s(\sigma)\|^p \approx_{p,X} \mathbb{E}\left\|[M(\sigma)]_t^{\frac{1}{2}}\right\|^p$$

for all  $t \ge 0$ , which extends (9.1.4) to the case p = 1 and general local martingales.

In conclusion we wish to notice that it remains open whether one can find a *predictable* right-hand side in (9.1.5): so far such a predictable right-hand side was explored only in the real-valued case and in the case  $X = L^q(S)$ ,  $1 < q < \infty$ , see *Burkholder–Novikov–Rosenthal inequalities* in the forthcoming paper [53].

# **9.2.** BURKHOLDER–DAVIS–GUNDY INEQUALITIES: THE DISCRETE TIME CASE

Let us show discrete Burkholder–Davis–Gundy inequalities.

**Theorem 9.2.1.** Let X be a UMD Banach space,  $(d_n)_{n\geq 1}$  be an X-valued martingale difference sequence. Then for any  $1 \leq p < \infty$ 

$$\mathbb{E}\sup_{m\geq 1} \left\| \sum_{n=1}^{m} d_n \right\|^p \approx_{p,X} \mathbb{E} \| (d_n)_{n=1}^{\infty} \|_{\gamma(\ell^2,X)}^p.$$
(9.2.1)

*Proof.* Without loss of generality we may assume that there exists  $N \ge 1$  such that  $d_n = 0$  for all n > N. Let  $(r_n)_{n \ge 1}$  be a sequence of independent Rademacher random variables,  $(\gamma_n)_{n \ge 1}$  be a sequence of independent standard Gaussian random variables. Then

$$\mathbb{E}\sup_{m\geq 1}\left\|\sum_{n=1}^{m}d_{n}\right\|^{p} \stackrel{(i)}{\sim}_{p} \mathbb{E}\mathbb{E}_{r}\sup_{m\geq 1}\left\|\sum_{n=1}^{N}r_{n}d_{n}\right\|^{p} \stackrel{(ii)}{\sim}_{p,X} \mathbb{E}\mathbb{E}_{r}\left\|\sum_{n=1}^{N}r_{n}d_{n}\right\|^{p}$$

$$\stackrel{(iii)}{\approx}_{p,X} \mathbb{E}\mathbb{E}_{\gamma} \left\| \sum_{n=1}^{N} \gamma_n d_n \right\|^p \stackrel{(iv)}{\approx}_p \mathbb{E}\left(\mathbb{E}_{\gamma} \left\| \sum_{n=1}^{N} \gamma_n d_n \right\|^2\right)^{\frac{p}{2}}$$

$$= \mathbb{E}\|(d_n)_{n=1}^{\infty}\|_{\gamma(\ell^2,X)}^p,$$

$$(9.2.2)$$

where (*i*) follows from [35, (8.22)], (*ii*) holds by [80, Proposition 6.1.12], (*iii*) follows from [80, Corollary 7.2.10 and Proposition 7.3.15], and (*iv*) follows from [80, Proposition 6.3.1].

*Remark* 9.2.2. If we collect all the constants in (9.2.2) then one can see that those constants behave well as  $p \rightarrow 1$ , i.e. for any  $1 < r < \infty$  there exist positive  $C_{r,X}$  and  $c_{r,X}$  such that for any  $1 \le p \le r$ 

$$c_{r,X}\mathbb{E}\|(d_n)_{n=1}^{\infty}\|_{\gamma(\ell^2,X)}^p \le \mathbb{E}\sup_{m\ge 1}\left\|\sum_{n=1}^m d_n\right\|^p \le C_{r,X}\mathbb{E}\|(d_n)_{n=1}^{\infty}\|_{\gamma(\ell^2,X)}^p$$

*Remark* 9.2.3. Fix 1 and a UMD Banach space*X*. By Doob's maximal inequality (2.2.1) and Theorem 9.2.1 we have that

$$\mathbb{E}\left\|\sum_{n=1}^{\infty} d_n\right\|^p \approx_p \mathbb{E}\sup_{m\geq 1}\left\|\sum_{n=1}^m d_n\right\|^p \approx_{p,X} \mathbb{E}\left\|(d_n)_{n=1}^{\infty}\right\|_{\gamma(\ell^2,X)}^p.$$

Let us find the constants in the equivalence

$$\mathbb{E}\left\|\sum_{n=1}^{\infty} d_n\right\|^p \approx_{p,X} \mathbb{E}\|(d_n)_{n=1}^{\infty}\|_{\gamma(\ell^2,X)}^p.$$

Since *X* is UMD, it has a finite cotype *q* (see [80, Definition 7.1.1. and Proposition 7.3.15]), and therefore by modifying (9.2.2) (using decoupling inequalities [79, p. 282] instead of [35, (8.22)] and [80, Proposition 6.1.12]) one can show that

$$\begin{split} \frac{1}{\beta_{p,X}c_{p,X}} \Big( \mathbb{E}\|(d_n)_{n=1}^{\infty}\|_{\gamma(\ell^2,X)}^p \Big)^{\frac{1}{p}} &\leq \Big( \mathbb{E}\left\|\sum_{n=1}^m d_n\right\|^p \Big)^{\frac{1}{p}} \\ &\leq 2\beta_{p,X}\kappa_{p,2} \Big( \mathbb{E}\|(d_n)_{n=1}^{\infty}\|_{\gamma(\ell^2,X)}^p \Big)^{\frac{1}{p}}, \end{split}$$

where  $c_{p,X}$  depends on p, the cotype of X, and the Gaussian cotype constant of X (see [80, Proposition 7.3.15]), while  $\kappa_{p,q}$  is the Kahane-Khinchin constant (see [80, Section 6.2]).

In the following theorem we show that *X* having the UMD property is necessary for Theorem 9.2.1 to hold.

**Theorem 9.2.4.** Let X be a Banach space and  $1 \le p < \infty$  be such that (9.2.1) holds for any martingale difference sequence  $(d_n)_{n\ge 1}$ . Then X is UMD.

*Proof.* Note that for any set  $(x_n)_{n=1}^N$  of elements of X and for any [-1,1]-valued sequence  $(\varepsilon_n)_{n=1}^N$  we have that  $\|(\varepsilon_n x_n)_{n=1}^N\|_{\gamma(\ell_N^2, X)} \le \|(x_n)_{n=1}^N\|_{\gamma(\ell_N^2, X)}$  by the ideal property (see [80, Theorem 9.1.10]). Therefore if (9.2.1) holds for any X-valued martingale difference sequence  $(d_n)_{n\geq 1}$ , then we have that for any [-1,1]-valued sequence  $(\varepsilon_n)_{n\geq 1}$ 

$$\mathbb{E}\sup_{m\geq 1}\left\|\sum_{n=1}^{m}\varepsilon_{n}d_{n}\right\|^{p} \lesssim_{p,X} \mathbb{E}\sup_{m\geq 1}\left\|\sum_{n=1}^{m}d_{n}\right\|^{p}.$$
(9.2.3)

If p > 1, then (9.2.3) together with (2.2.1) implies the UMD property. If p = 1, then (9.2.3) for p = 1 implies (9.2.3) for any p > 1 (see [79, Theorem 3.5.4]), and hence it again implies UMD.

Now we turn to the continuous-time case. It turns out that in this case the righthand side of (9.2.1) transforms to a so-called *Gaussian characteristic* of a certain bilinear form generated by a quadratic variation of the corresponding martingale. Therefore before proving our main result (Theorem 9.5.1) we will need to outline some basic properties of a Gaussian characteristic (see Section 9.3). We will also need some preliminaries concerning continuous-time Banach space-valued martingales (see Section 9.4).

## **9.3.** GAUSSIAN CHARACTERISTICS

The current section is devoted to the definition and some basic properties of one of the main object of the chapter – a Gaussian characteristic of a bilinear form. Many of the statements here might seem to be obvious for the reader. Nevertheless we need to show them before reaching our main Theorem 9.5.1.

#### 9.3.1. Basic definitions

Let us first recall some basic facts on Gaussian measures. Let *X* be a Banach space. An *X*-valued random variable  $\xi$  is called *Gaussian* if  $\langle \xi, x^* \rangle$  has a Gaussian distribution for all  $x^* \in X^*$ . Gaussian random variables enjoy a number of useful properties (see [21, 100]). We will need the following *Gaussian covariance domination inequality* (see [21, Corollary 3.3.7] and [80, Theorem 6.1.25] for the case  $\Phi = \|\cdot\|^p$ ).

**Lemma 9.3.1.** Let X be a Banach space,  $\xi, \eta$  be centered X-valued Gaussian random variables. Assume that  $\mathbb{E}\langle \eta, x^* \rangle^2 \leq \mathbb{E}\langle \xi, x^* \rangle^2$  for all  $x^* \in X^*$ . Then  $\mathbb{E}\Phi(\eta) \leq \mathbb{E}\Phi(\xi)$  for any convex symmetric continuous function  $\Phi: X \to \mathbb{R}_+$ .

Let *X* be a Banach space. We denote the linear space of all continuous  $\mathbb{R}$ -valued bilinear forms on *X* × *X* by *X*<sup>\*</sup>  $\otimes$  *X*<sup>\*</sup>. Note that this linear space can be endowed with the following natural norm:

$$\|V\| := \sup_{x \in X, \|x\| \le 1} |V(x, x)|, \tag{9.3.1}$$

where the latter expression is finite due to bilinearity and continuity of *V*. A bilinear form *V* is called *nonnegative* if  $V(x, x) \ge 0$  for all  $x \in X$ , and *V* is called *symmetric* if V(x, y) = V(y, x) for all  $x, y \in X$ .

Let *X* be a Banach space,  $\xi$  be a centered *X*-valued Gaussian random variable. Then  $\xi$  has a *covariance bilinear form*  $V : X^* \times X^* \to \mathbb{R}$  such that

$$V(x^*, y^*) = \mathbb{E}\langle \xi, x^* \rangle \langle \xi, y^* \rangle, \ x^*, y^* \in X.$$

Notice that a covariance bilinear form is always continuous, symmetric, and nonnegative. It is worth noticing that one usually considers a *covariance operator* Q :  $X^* \rightarrow X^{**}$  defined by

$$\langle Qx^*, y^* \rangle = \mathbb{E}\langle \xi, x^* \rangle \langle \xi, y^* \rangle, \ x^*, y^* \in X.$$

But since there exists a simple one-to-one correspondence between bilinear forms and  $\mathscr{L}(X^*, X^{**})$ , we will work with covariance bilinear forms instead. We refer the reader to [21, 48, 68, 173] for details.

Let  $V: X^* \times X^* \to \mathbb{R}$  be a symmetric continuous nonnegative bilinear form. Then V is said to have a finite *Gaussian characteristic*  $\gamma(V)$  if there exists a centered X-valued Gaussian random variable  $\xi$  such that V is the covariance bilinear form of  $\xi$ . Then we set  $\gamma(V) := (\mathbb{E} ||\xi||^2)^{\frac{1}{2}}$  (this value is finite due to the Fernique theorem, see [21, Theorem 2.8.5]). Otherwise we set  $\gamma(V) = \infty$ . Note that then for all  $x^*, y^* \in X^*$  one has the following control of continuity of V:

$$|V(x^*, x^*)^{\frac{1}{2}} - V(y^*, y^*)^{\frac{1}{2}}| = (\mathbb{E}|\langle \xi, x^* \rangle|^2)^{\frac{1}{2}} - (\mathbb{E}|\langle \xi, y^* \rangle|^2)^{\frac{1}{2}}$$
  
$$\leq (\mathbb{E}|\langle \xi, x^* - y^* \rangle|^2)^{\frac{1}{2}} \leq (\mathbb{E}||\xi||^2)^{\frac{1}{2}} ||x^* - y^*|| = ||x^* - y^*||\gamma(V).$$
(9.3.2)

*Remark* 9.3.2. Note that for any *V* with  $\gamma(V) < \infty$  the distribution of the corresponding centered *X*-valued Gaussian random variable  $\xi$  is *uniquely determined* (see [21, Chapter 2]).

*Remark* 9.3.3. Note that if *X* is finite dimensional, then  $\gamma(V) < \infty$  for any nonnegative symmetric bilinear form *V*. Indeed, in this case *X* is isomorphic to a finite dimensional Hilbert space *H*, so there exists an eigenbasis  $(h_n)_{n=1}^d$  making *V* diagonal, and then the corresponding Gaussian random variable will be equal to  $\xi := \sum_{n=1}^{d} V(h_n, h_n) \gamma_n h_n$ , where  $(\gamma_n)_{n=1}^d$  are independent standard Gaussian.

#### **9.3.2.** Basic properties of $\gamma(\cdot)$

Later we will need the following technical lemmas.

**Lemma 9.3.4.** Let X be a reflexive (separable) Banach space,  $V : X^* \times X^* \to \mathbb{R}$  be a symmetric continuous nonnegative bilinear form. Then there exist a (separable) Hilbert space H and  $T \in \mathcal{L}(H, X)$  such that

$$V(x^*, y^*) = \langle T^* x^*, T^* y^* \rangle, x^*, y^* \in X^*.$$

Proof. See [27, pp. 57-58] or [100, p. 154].

The following lemma connects Gaussian characteristics and  $\gamma$ -norms (see (2.9.1)) and it can be found e.g. in [125, Theorem 7.4] or in [27, 129].

**Lemma 9.3.5.** Let X be a separable Banach space, H be a separable Hilbert space,  $T \in \mathcal{L}(H,X), V : X^* \times X^* \to \mathbb{R}$  be a symmetric continuous nonnegative bilinear form such that  $V(x^*, y^*) = \langle T^*x^*, T^*y^* \rangle$  for all  $x^*, y^* \in X^*$ . Then  $\gamma(V) = ||T||_{\gamma(H,X)}$ .

*Remark* 9.3.6. Fix a Hilbert space *H* and a Banach space *X*. Note that even though by the lemma above there exists a natural embedding of  $\gamma$ -radonifying operators from  $\mathcal{L}(H, X)$  to the space of symmetric nonnegative bilinear forms on  $X^* \times X^*$ , this embedding is neither *injective* nor *linear*. This also explains why we need to use bilinear forms with finite Gaussian characteristics instead of  $\gamma$ -radonifying operators: in the proof of our main result – Theorem 9.5.1 – we will need various statements (like triangular inequalities and convergence theorems) for *bilinear forms*, not *operators*.

Now we will prove some statements about approximation of nonnegative symmetric bilinear forms by finite dimensional ones in  $\gamma(\cdot)$ .

**Lemma 9.3.7.** Let X be a reflexive Banach space,  $Y \subset X^*$  be a finite dimensional subspace. Let  $P: Y \hookrightarrow X^*$  be an inclusion operator. Let  $V: X^* \times X^* \to \mathbb{R}$  and  $V_0: Y \times Y \to \mathbb{R}$  be symmetric continuous nonnegative bilinear forms such that  $V_0(x_0^*, y_0^*) = V(Px_0^*, Py_0^*)$  for all  $x_0^*, y_0^* \in Y$ . Then  $\gamma(V_0)$  is well-defined and  $\gamma(V_0) \leq \gamma(V)$ .

*Proof.* First of all notice that  $\gamma(V_0)$  is well-defined since *Y* is finite dimensional, hence reflexive, and thus has a predual space coinciding with its dual. Without loss of generality assume that  $||V||_{\gamma} < \infty$ . Let  $\xi_V$  be a centered *X*-valued Gaussian random variable with *V* as the covariance bilinear form. Define  $\xi_{V_0} := P^* \xi_V$  (note that  $Y^* \hookrightarrow X$  due to the Hahn-Banach theorem). Then for all  $x_0^*, y_0^* \in X_0^*$ 

$$\mathbb{E}\langle\xi_{V_0}, x_0^*\rangle\langle\xi_{V_0}, y_0^*\rangle = \mathbb{E}\langle\xi_V, Px_0^*\rangle\langle\xi_V, Py_0^*\rangle = V(Px_0^*, Py_0^*) = V_0(x_0^*, y_0^*),$$

so  $V_0$  is the covariance bilinear form of  $\xi_{V_0}$  and since  $||P^*|| = ||P|| = 1$ 

$$\gamma(V_0) = (\mathbb{E} \|\xi_{V_0}\|^2)^{\frac{1}{2}} = (\mathbb{E} \|P^*\xi_V\|^2)^{\frac{1}{2}} \le (\mathbb{E} \|\xi_V\|^2)^{\frac{1}{2}} = \gamma(V).$$
(9.3.3)

**Proposition 9.3.8.** Let X be a separable reflexive Banach space,  $V : X^* \times X^* \to \mathbb{R}$  be a symmetric continuous nonnegative bilinear form. Let  $Y_1 \subset Y_2 \subset ... \subset Y_m \subset ...$  be a sequence of finite dimensional subspaces of  $X^*$  with  $\overline{\bigcup_m Y_m} = X^*$ . Then for each  $m \ge 1$  a symmetric continuous nonnegative bilinear form  $V_m = V|_{Y_m \times Y_m}$  is well-defined and  $\gamma(V_m) \to \gamma(V)$  as  $m \to \infty$ .

*Proof.* First of all notice that  $V_m$ 's are well-defined since each of the  $Y_m$  is finite dimensional, hence reflexive, and thus has a predual space coinciding with its dual (which we will call  $X_m$  and which can even be embedded into X due to the Hahn-Banach theorem). Let  $P_m : Y_m \to X^*$  be the inclusion operator (thus is particular  $||P_m|| \leq 1$ ). Let a Hilbert space H and an operator  $T \in \mathcal{L}(H, X)$  be as constructed in Lemma 9.3.4. Let  $(h_n)_{n\geq 1}$  be an orthonormal basis of H, and  $(\gamma_n)_{n\geq 1}$  be a sequence of standard Gaussian random variables. For each  $N \geq 1$  define a centered Gaussian random variable  $\xi_N := \sum_{n=1}^N \gamma_n T h_n$ . Then for each  $m \geq 1$  the centered Gaussian random variable  $\sum_{n=1}^{\infty} \gamma_n P_m^* T h_n$  is well-defined (since  $P_m^* T$  has a finite rank, and every finite rank operator has a finite  $\gamma$ -norm, see [80, Section 9.2]), and for any  $x^* \in Y_m$  we have that

$$V_m(x^*, x^*) = V(x^*, x^*) = ||T^*x^*|| = ||T^*P_mx^*|| = \mathbb{E}\Big\langle \sum_{n=1}^{\infty} \gamma_n P_m^* Th_n, x^* \Big\rangle^2,$$

so  $V_m$  is the covariance bilinear form of  $\sum_{n=1}^{\infty} \gamma_n P_m^* T h_n$ , and

$$\gamma(V_m) = \left(\mathbb{E}\left\|\sum_{n=1}^{\infty} \gamma_n P_m^* T h_n\right\|^2\right)^{\frac{1}{2}} = \left(\mathbb{E}\left\|P_m^* \sum_{n=1}^{\infty} \gamma_n T h_n\right\|^2\right)^{\frac{1}{2}}.$$

The latter expression converges to  $\gamma(V)$  by Lemma 9.3.5 and due to the fact that  $\|P_m^* x\| \to \|x\|$  monotonically for each  $x \in X$  as  $m \to \infty$ .

The next lemma provides the Gaussian characteristic with the triangular inequality.

**Lemma 9.3.9.** Let X be a reflexive Banach space,  $V, W : X^* \times X^*$  be symmetric continuous nonnegative bilinear forms. Then  $\gamma(V + W) \leq \gamma(V) + \gamma(W)$ .

*Proof.* If  $\max\{\gamma(V), \gamma(W)\} = \infty$  then the lemma is obvious. Let  $\gamma(V), \gamma(W) < \infty$ . Let  $\xi_V$  and  $\xi_W$  be *X*-valued centered Gaussian random variables corresponding to *V* and *W* respectively. Without loss of generality we can set  $\xi_V$  and  $\xi_W$  independent. Let  $\xi_{V+W} = \xi_V + \xi_W$ . Then  $\xi_{V+W}$  is an *X*-valued centered Gaussian random variable (see [21]) and for any  $x^* \in X^*$  due to the independence of  $\xi_V$  and  $\xi_W$ 

$$\mathbb{E}\langle \xi_{V+W}, x^* \rangle^2 = \mathbb{E}\langle \xi_V + \xi_W, x^* \rangle^2 = \mathbb{E}\langle \xi_V, x^* \rangle^2 + \mathbb{E}\langle \xi_W, x^* \rangle^2 = (V+W)(x^*, x^*).$$

So  $\xi_{V+W}$  has V + W as the covariation bilinear form, and therefore

$$\gamma(V+W) = (\mathbb{E}\|\xi_{V+W}\|^2)^{\frac{1}{2}} \le (\mathbb{E}\|\xi_V\|^2)^{\frac{1}{2}} + (\mathbb{E}\|\xi_W\|^2)^{\frac{1}{2}} = \gamma(V) + \gamma(W)$$

Now we discuss such important properties of  $\gamma(\cdot)$  as monotonicity and monotone continuity.

**Lemma 9.3.10.** Let X be a separable Banach space,  $V, W : X^* \times X^* \to \mathbb{R}$  be symmetric continuous nonnegative bilinear forms such that  $W(x^*, x^*) \leq V(x^*, x^*)$  for all  $x^* \in X^*$ . Then  $\gamma(W) \leq \gamma(V)$ .

*Proof.* The lemma follows from Lemma 9.3.5 and [80, Theorem 9.4.1].

**Lemma 9.3.11.** Let X be a separable reflexive Banach space,  $(V_n)_{n\geq 1}$  be symmetric continuous nonnegative bilinear forms on  $X^* \times X^*$  such that  $V_n(x^*, x^*) \to 0$  for any  $x^* \in X^*$ monotonically as  $n \to \infty$ . Assume additionally that  $\gamma(V_n) < \infty$  for some  $n \ge 1$ . Then  $\gamma(V_n) \to 0$  monotonically as  $n \to \infty$ .

*Proof.* Without loss of generality assume that  $\gamma(V_1) < \infty$ . First notice that by Lemma 9.3.10 the sequence  $(\gamma(V_n))_{n \ge 1}$  is monotone and bounded by  $\gamma(V_1)$ .

By Lemma 9.3.4 we may assume that there exists a separable Hilbert space H and a sequence of operators  $(T_n)_{n\geq 1}$  from H to X such that  $V_n(x^*, x^*) = ||T_n^*x^*||^2$  for all  $x^* \in X^*$  (note that we are working with one Hilbert space since all the separable Hilbert spaces are isometrically isomorphic). Let  $T \in \mathcal{L}(H, X)$  be the zero operator. Then  $T_n^*x^* \to T^*x^* = 0$  as  $n \to \infty$  for all  $x^* \in X^*$ , and hence by [80, Theorem 9.4.2], Lemma 9.3.5, and the fact that  $||T_nx^*|| \le ||T_1x^*||$  for all  $x^* \in X^*$ 

$$\lim_{n \to \infty} \gamma(V_n) = \lim_{n \to \infty} \|T_n\|_{\gamma(H,X)} = \|T\|_{\gamma(H,X)} = 0.$$

The following lemma follows for Lemma 9.3.9 and 9.3.11.

**Lemma 9.3.12.** Let X be a separable reflexive Banach space, V,  $(V_n)_{n\geq 1}$  be symmetric continuous nonnegative bilinear forms on  $X^* \times X^*$  such that  $V_n(x^*, x^*) \nearrow V(x^*, x^*)$  for any  $x^* \in X^*$  monotonically as  $n \to \infty$ . Then  $\gamma(V_n) \nearrow \gamma(V)$  monotonically as  $n \to \infty$ .

*Remark* 9.3.13. Notice that  $\gamma(\cdot)$  is not a norm. Indeed, it is easy to see that  $\gamma(\alpha V) = \sqrt{\alpha}\gamma(V)$  for any  $\alpha \ge 0$  and any nonnegative symmetric bilinear form *V*: if we fix any *X*-valued Gaussian random variable  $\xi$  having *V* as its covariance bilinear form, then  $\sqrt{\alpha}\xi$  has  $\alpha\gamma(V)$  as its covariance bilinear form. Therefore it is a natural question whether  $\gamma(\cdot)^2$  satisfies the triangle inequality and hence has the norm properties. It is easy to check the triangle inequality if *X* is Hilbert: indeed, for any *V* and *W* 

$$\gamma(V+W)^2 = \mathbb{E}\|\xi_{V+W}\|^2 = \mathbb{E}\|\xi_V\|^2 + \mathbb{E}\|\xi_W\|^2 + 2\mathbb{E}\langle\xi_V,\xi_W\rangle = \gamma(V)^2 + \gamma(W)^2,$$

where  $\xi_V$ ,  $\xi_W$ , and  $\xi_{V+W}$  are as in the latter proof.

It turns out that if such a triangular inequality holds for some Banach space *X*, then this Banach space must have a *Gaussian type* 2 (see [80, Subsection 7.1.d]). Indeed, let *X* be such that for all nonnegative symmetric bilinear forms *V* and *W* on  $X^* \times X^*$ 

$$\gamma (V+W)^2 \le \gamma (V)^2 + \gamma (W)^2.$$
 (9.3.4)

Fix  $(x_i)_{i=1}^n \subset X$  and a sequence of independent standard Gaussian random variables  $(\xi_i)_{i=1}^n$ . For each i = 1, ..., n define a symmetric bilinear form  $V_i : X^* \times X^* \to \mathbb{R}$  as  $V_i(x^*, y^*) := \langle x_i, x^* \rangle \cdot \langle x_i, y^* \rangle$ . Let  $V = V_1 + \cdots + V_n$ . Then by (9.3.4) and the induction argument

$$\mathbb{E}\left\|\sum_{i=1}^{n}\xi_{i}x_{i}\right\|^{2} \stackrel{(*)}{=} \gamma(V)^{2} \leq \sum_{i=1}^{n}\gamma(V_{i})^{2} \stackrel{(**)}{=} \sum_{i=1}^{n}\mathbb{E}\|\xi_{i}x_{i}\|^{2} = \sum_{i=1}^{n}\|x_{i}\|^{2},$$

where (\*) follows from the fact that  $\sum_{i=1}^{n} \xi_i x_i$  is a centered Gaussian random variable the fact that for all  $x^*, y^* \in X^*$ 

$$\mathbb{E}\Big\langle\sum_{i=1}^n\xi_ix_i,x^*\Big\rangle\cdot\Big\langle\sum_{i=1}^n\xi_ix_i,y^*\Big\rangle=\sum_{i=1}^n\langle x_i,x^*\rangle\cdot\langle x_i,y^*\rangle=V(x^*,y^*),$$

while (\*\*) follows analogously by exploiting the fact that  $\xi_i x_i$  is a centered Gaussian random variable with the covariance bilinear form  $V_i$ . Therefore by [80, Definition 7.1.17], *X* has a Gaussian type 2 with the corresponding Gaussian type constant  $\tau_{2,X}^{\gamma} = 1$ .

#### 9.3.3. Finite dimensional case

Even though a Gaussian characteristic is well-defined only for some nonnegative symmetric forms, it can be extended in a proper continuous way to all the symmetric forms given *X* is finite dimensional. Let *X* be a finite dimensional Banach space. Notice that in this case  $\gamma(V) < \infty$  for any nonnegative symmetric bilinear form *V* (see Remark 9.3.3). Let us define  $\gamma(V)$  for a general symmetric  $V \in X^{**} \otimes X^{**} = X \otimes X$  in the following way:

$$\gamma(V) := \inf\{\gamma(V^+) + \gamma(V^-) : V^+, V^- \text{ are nonnegative and } V = V^+ - V^-\}.$$
 (9.3.5)

Notice that  $\gamma(V)$  is well-defined and finite for any symmetric *V*. Indeed, by a well known linear algebra fact (see e.g. [166, Theorem 6.6 and 6.10]) any symmetric bilinear form *V* has an eigenbasis  $(x_n^*)_{n=1}^d$  of  $X^*$  that *diagonalizes V*, i.e. there exists  $(\lambda_n)_{n=1}^d \in \mathbb{R}$  such that for all  $(a_n)_{n=1}^d, (b_n)_{n=1}^d \in \mathbb{R}$  we have that for  $x^* = \sum_{n=1}^d a_n x_n^*$  and  $y^* = \sum_{n=1}^d b_n x_n^*$ 

$$V(x^*, y^*) = \sum_{n=1}^d \sum_{m=1}^d a_n b_m V(x_n^*, x_m^*) = \sum_{n=1}^d \lambda_n a_n b_n$$

Therefore it is sufficient to define

$$V^{+}(x^{*}, y^{*}) := \sum_{n=1}^{d} \mathbf{1}_{\lambda_{n} \ge 0} \lambda_{n} a_{n} b_{n}, \quad V^{-}(x^{*}, y^{*}) := \sum_{n=1}^{d} \mathbf{1}_{\lambda_{n} < 0}(-\lambda_{n}) a_{n} b_{n}$$

and then  $\gamma(V) \leq \gamma(V^+) + \gamma(V^-) < \infty$  due to the fact that  $V^+$  and  $V^-$  are nonnegative and by Remark 9.3.3. (In fact, one can check that  $\gamma(V) = \gamma(V^+) + \gamma(V^-)$ , but we will not need this later, so we leave this fact without a proof).

Now we will develop some basic and *elementary* (but nonetheless important) properties of such a general  $\gamma(\cdot)$ .

**Lemma 9.3.14.** Let  $V : X^* \times X^* \to \mathbb{R}$  be a nonnegative symmetric bilinear form. Then  $\gamma(V)$  defined by (9.3.5) coincides with  $\gamma(V)$  defined in Subsection 9.3.1. In other words, these definitions agree given V is nonnegative.

*Proof.* Fix nonnegative  $V^+$  and  $V^-$  such that  $V = V^+ - V^-$ . Then  $\gamma(V^+) + \gamma(V^-) = \gamma(V + V^-) + \gamma(V^-) \ge \gamma(V) + \gamma(V^-) \ge \gamma(V)$  by Lemma 9.3.10, so  $\gamma(V)$  does not change.

**Lemma 9.3.15.** Let  $V, W : X^* \times X^* \to \mathbb{R}$  be symmetric bilinear forms. Then  $\gamma(V) - \gamma(W) \le \gamma(V - W)$ .

*Proof.* Denote V - W by U. Fix  $\varepsilon > 0$ . Then there exist symmetric nonnegative bilinear forms  $W^+, W^-, U^+, U^-$  such that  $W = W^+ - W^-, U = U^+ - U^-$ , and

$$\gamma(W) \ge \gamma(W^+) + \gamma(W^-) - \varepsilon,$$

 $\gamma(U) \geq \gamma(U^+) + \gamma(U^-) - \varepsilon.$ 

Then since V = U + W by (9.3.5) and Lemma 9.3.9

$$\begin{split} \gamma(V) - \gamma(W) &= \gamma((W^+ + U^+) - (W^- + U^-)) - \gamma(W^+ - W^-) \\ &\leq \gamma(W^+ + U^+) + \gamma(W^- + U^-) - \gamma(W^+) - \gamma(W^-) + \varepsilon \\ &\leq \gamma(U^+) + \gamma(U^-) + \varepsilon \leq \gamma(U) + 2\varepsilon, \end{split}$$

and by sending  $\varepsilon \rightarrow 0$  we conclude the desired.

**Lemma 9.3.16.** Let  $V : X^* \times X^* \to \mathbb{R}$  be a symmetric bilinear form. Then  $\gamma(V) = \gamma(-V)$  and  $\gamma(\alpha V) = \sqrt{\alpha}\gamma(V)$  for any  $\alpha \ge 0$ .

*Proof.* The first part follows directly from (9.3.5). For the second part we have that due to (9.3.5) it is enough to justify  $\gamma(\alpha V) = \sqrt{\alpha}\gamma(V)$  only for nonnegative *V*, which was done in Remark 9.3.13.

**Proposition 9.3.17.** The function  $\gamma(\cdot)$  defined by (9.3.5) is continuous on the linear space of all symmetric bilinear forms endowed with  $\|\cdot\|$  defined by (9.3.1). Moreover,  $\gamma(V)^2 \leq_X \|V\|$  for any symmetric bilinear form  $V : X^* \times X^* \to \mathbb{R}$ .

*Proof.* Due to Lemma 9.3.15 and 9.3.16 it is sufficient to show that  $\gamma(\cdot)$  is bounded on the unit ball with respect to the norm  $\|\cdot\|$  in order to prove the first part of the proposition. Let us show this boundedness. Let *U* be a fixed symmetric nonnegative element of  $X \otimes X$  such that U + V is nonnegative and such that  $U(x^*, x^*) \ge$  $V(x^*, x^*)$  for any symmetric *V* with  $\|V\| \le 1$  (since *X* is finite dimensional, one can take  $U(x^*) := c \|\|x^*\|^2$  for some Euclidean norm  $\|\|\cdot\|\|$  on  $X^*$  and some big enough

constant c > 0). Fix a symmetric  $V : X^* \times X^* \rightarrow \mathbb{R}$  with  $||V|| \le 1$ . Then V = (U + V) - U, and by (9.3.5)

$$\gamma(V) \leq \gamma(U+V) + \gamma(U) = \gamma(2U) + \gamma(U),$$

which does not depend on *V*.

Let us show the second part. Due to the latter consideration there exists a constant  $C_X$  depending only on X such that  $\gamma(V) \leq C_X$  if  $||V|| \leq 1$ . Therefore by Lemma 9.3.16 we have that for a general symmetric V

$$\gamma(V)^{2} = \|V\|\gamma(V/\|V\|)^{2} \le C_{X}^{2}\|V\|.$$

Later we will also need the following elementary lemma.

**Lemma 9.3.18.** There exists vectors  $(x_i^*)_{i=1}^n$  in  $X^*$  such that

$$|||V||| := \sum_{i=1}^{n} |V(x_i^*, x_i^*)|$$
(9.3.6)

*defines a norm on the space of all symmetric bilinear forms on*  $X^* \times X^*$ *. In particular we have that*  $||V|| = \sum_X ||V||$  *for any symmetric bilinear form*  $V : X^* \times X^* \to \mathbb{R}$ *.* 

We will demonstrate here the proof for the convenience of the reader.

*Proof.* First notice that  $\|\cdot\|$  clearly satisfies the triangular inequality. Let us show that there exists a set  $(x_i^*)_{i=1}^n$  such that  $\|\|V\|\| = 0$  implies V = 0. Let  $(y_i^*)_{i=1}^d$  be a basis of  $X^*$ . Then there exist  $i, j \in \{1, ..., d\}$  such that

$$0 \neq V(y_i^*, y_j^*) = (V(y_i^* + y_j^*, y_i^* + y_j^*) - V(y_i^* - y_j^*, y_i^* - y_j^*))/4$$

(otherwise V = 0). This means that for these *i* and *j* 

$$|V(y_i^* + y_j^*, y_i^* + y_j^*)| + |V(y_i^* - y_j^*, y_i^* - y_j^*)| \neq 0,$$

so in particular

$$\sum_{i=1}^{d} \sum_{j=1}^{d} |V(y_{i}^{*} + y_{j}^{*}, y_{i}^{*} + y_{j}^{*})| + |V(y_{i}^{*} - y_{j}^{*}, y_{i}^{*} - y_{j}^{*})| \neq 0.$$

It remains to notice that the latter sum has the form (9.3.6) for a proper choice of  $(x_i^*)_{i=1}^n$  independent of *V*.

In order to show the last part of the lemma we need to notice that the space of symmetric bilinear forms is finite dimensional if *X* is so, so all the norms on the linear space of symmetric bilinear forms are equivalent, and therefore  $||V|| = \sum_X |||V||$  for any symmetric bilinear form  $V: X^* \times X^* \to \mathbb{R}$ .

# 9.4. COVARIATION BILINEAR FORMS

We continue with the definition of a covariation bilinear form and its basic properties.

Let *X* be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a local martingale. Fix  $t \ge 0$ . Then *M* is said to have a *covariation bilinear from*  $[[M]]_t$  at  $t \ge 0$  if there exists a continuous bilinear form-valued random variable  $[[M]]_t : X^* \times X^* \times \Omega \to \mathbb{R}$  such that for any fixed  $x^*, y^* \in X^*$  a.s.  $[[M]]_t(x^*, y^*) = [\langle M, x^* \rangle, \langle M, y^* \rangle]_t$ .

*Remark* 9.4.1. Let us outline some basic properties of the covariation bilinear forms, which follow directly from [89, Theorem 26.6] (here we presume the existence of  $[[M]]_t$  and  $[[N]]_t$  for all  $t \ge 0$ )

- (i)  $t \mapsto [[M]]_t$  is nondecreasing, i.e.  $[[M]]_t(x^*, x^*) \ge [[M]]_s(x^*, x^*)$  a.s. for all  $0 \le s \le t$ and  $x^* \in X^*$ ,
- (ii)  $[[M]]^{\tau} = [[M^{\tau}]]$  a.s. for any stopping time  $\tau$ ,
- (iii)  $\Delta[[M]]_{\tau}(x^*, x^*) = |\langle \Delta M_{\tau}, x^* \rangle|^2$  a.s. for any stopping time  $\tau$ .

*Remark* 9.4.2. If *X* is finite dimensional, then it is isomorphic to a Hilbert space, and hence existence of  $[[M]]_t$  follows from existence of  $[M]_t$  with the following estimate a.s.

$$\|[[M]]_t\| = \sup_{x^* \in X^*, \|x^*\| \le 1} [[M]]_t(x^*, x^*) = \sup_{x^* \in X^*, \|x^*\| \le 1} [\langle M, x^* \rangle, \langle M, x^* \rangle]_t \lesssim_X [M]_t.$$

For a general infinite dimensional Banach space the existence of  $[[M]]_t$  remains an open problem. In Theorem 9.5.1 we show that if *X* has the UMD property, then existence of  $[[M]]_t$  follows automatically; moreover, in this case  $\gamma([[M]]_t) < \infty$  a.s. (see Section 9.3 and Theorem 9.5.1, which is way stronger than continuity.

# 9.5. BURKHOLDER–DAVIS–GUNDY INEQUALITIES: THE CONTINU-OUS-TIME CASE

The following theorem is the main theorem of the chapter.

**Theorem 9.5.1.** Let X be a UMD Banach space. Then for any local martingale M:  $\mathbb{R}_+ \times \Omega \rightarrow X$  with  $M_0 = 0$  and any  $t \ge 0$  the covariation bilinear form  $[[M]]_t$  is well-defined and bounded almost surely, and for all  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{0\le s\le t} \|M_s\|^p \approx_{p,X} \mathbb{E}\gamma([[M]]_t)^p.$$
(9.5.1)

For the proof we will need the following technical lemma which follows from [180, Theorem 6].

**Lemma 9.5.2.** Let X be a finite dimensional Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a martingale such that  $\mathbb{E}\sup_{t\geq 0} \|M_t\| < \infty$ . Then there exists a sequence  $(M^n)_{n\geq 1}$  of X-valued uniformly bounded martingales such that  $\mathbb{E}\sup_{t\geq 0} \|M_t - M_t^n\| \to 0$  as  $n \to \infty$ .

*Proof of Theorem* 9.5.1. *Step* 1: *finite dimensional case.* First note that in this case  $[[M]]_t$  exists and bounded a.s. due to Remark 9.4.2. Fix  $1 \le p < \infty$ . We will prove separately the cases p > 1 and p = 1.

*Case* p > 1. For each  $N \ge 1$  fix a partition  $0 = t_1^N < ... < t_{n_N}^N = t$  with the mesh not exceeding 1/N. For each  $\omega \in \Omega$  and  $N \ge 1$  define a bilinear form  $V_N : X^* \times X^* \to \mathbb{R}$  as follows:

$$V_N(x^*, x^*) := \sum_{i=1}^{n_N} \langle M_{t_i^N} - M_{t_{i-1}^N}, x^* \rangle^2, \ x^* \in X^*.$$
(9.5.2)

Note that  $(M_{t_i^N} - M_{t_{i-1}^N})_{i=1}^{n_N}$  is a martingale difference sequence with respect to the filtration  $(\mathscr{F}_{t_i^N})_{i=1}^{n_N}$ , so by Theorem 9.2.1 and (2.2.1)

$$\mathbb{E} \|M_t\|^p = \mathbb{E} \left\| \sum_{i=1}^{n_N} M_{t_i^N} - M_{t_{i-1}^N} \right\|^p \approx_{p,X} \mathbb{E} \left( \mathbb{E}_{\gamma} \left\| \sum_{i=1}^{n_N} \gamma_i (M_{t_i^N} - M_{t_{i-1}^N}) \right\|^2 \right)^{\frac{p}{2}} = \mathbb{E} \gamma(V_N)^p,$$
(9.5.3)

where  $(\gamma_i)_{i=1}^{n_N}$  is a sequence of independent Gaussian standard random variables, and the latter equality holds due to the fact that for any fixed  $\omega \in \Omega$  the random variable  $\sum_{i=1}^{n_N} \gamma_i (M_{t_i^N} - M_{t_{i-1}^N})(\omega)$  is Gaussian and by (9.5.2)

$$V_N(x^*,x^*) = \mathbb{E}_{\gamma} \left\langle \sum_{i=1}^{n_N} \gamma_i (M_{t_i^N} - M_{t_{i-1}^N})(\omega), x^* \right\rangle^2, \ x^* \in X^*.$$

Therefore it is sufficient to show that  $\gamma(V_N - [[M]]_t) \to 0$  in  $L^p(\Omega)$  as  $N \to \infty$ . Indeed, if this is the case, then by (9.5.3) and by Lemma 9.3.15

$$\mathbb{E}\gamma(\llbracket M \rrbracket_t)^p = \lim_{N \to \infty} \mathbb{E}\gamma(V_N)^p \eqsim_{p,X} \mathbb{E} \|M_t\|^p.$$

Let us show this convergence. Note that by Proposition 9.3.17 and Lemma 9.3.18 a.s.

$$\gamma (V_N - [[M]]_t)^2 \lesssim_X \|V_N - [[M]]_t\| \lesssim_X \|V_N - [[M]]_t\|$$

(where  $\|\|\cdot\|\|$  is as in (9.3.6)) Therefore we need to show that  $\|\|V_N - [[M]]_t\|| \to 0$  in  $L^{\frac{p}{2}}(\Omega)$ , which follows from the fact that for any  $x_i^*$  from Lemma 9.3.18, i = 1, ..., n, we have that

$$V_{N}(x_{i}^{*}, x_{i}^{*}) = \sum_{i=1}^{n_{N}} \langle M_{t_{i}^{N}} - M_{t_{i-1}^{N}}, x_{i}^{*} \rangle^{2} \rightarrow [\langle M, x_{i}^{*} \rangle]_{t}$$

in  $L^{\frac{p}{2}}$ -sense by [55, Théorème 2].

*Case* p = 1. First assume that *M* is an  $L^2$ -bounded martingale. Then due to Remark 9.2.2 and Case p > 1 we can show that

$$c_X \mathbb{E}\gamma(\llbracket M \rrbracket_t)^p \le \mathbb{E} \sup_{0 \le s \le t} \|M_s\|^p \le C_X \mathbb{E}\gamma(\llbracket M \rrbracket_t)^p$$
(9.5.4)

for any  $1 for some universal positive constants <math>c_X$  and  $C_X$ . (9.5.1) then follows as p in (9.5.4) approaches to 1 by the dominated convergence theorem.

Now let *M* be a general martingale. Then by Lemma 9.5.2 there exists a sequence of *X*-valued  $L^2$ -bounded martingales  $(M^m)_{m\geq 1}$  such that  $\mathbb{E}\sup_{0\leq s\leq t} ||M_s - M_s^m|| \to 0$  as  $m \to \infty$ . In particular then we have that

$$\mathbb{E}\sup_{0\le s\le t} \|M_s^m\| \to \mathbb{E}\sup_{0\le s\le t} \|M_s\|, \ m \to \infty,$$
(9.5.5)

since  $M \mapsto \mathbb{E} \sup_{0 \le s \le t} ||M_s||$  defines a norm. Notice that due to the real-valued Burkholder– Davis–Gundy inequality for any fixed  $x^* \in X^*$ 

$$\mathbb{E}[\langle M_t - M_t^m, x^* \rangle]^{\frac{1}{2}} \sim \mathbb{E}\sup_{0 \le s \le t} |\langle M_s - M_s^m, x^* \rangle| \to 0, \quad m \to \infty,$$

so  $[\langle M_t^m - M_t, x^* \rangle]^{\frac{1}{2}} \to 0$  in  $L^1$ -sense. Therefore by Lemma 9.3.15, 9.3.18, and Proposition 9.3.17

$$|\gamma([[M^m]]_t) - \gamma([[M]]_t)| \le \gamma([[M^m]]_t - [[M]]_t) \lesssim_X \sum_{i=1}^n [\langle M^m - M, x_i^* \rangle]_t \to 0$$

in  $L^{\frac{1}{2}}$ , where  $(x_i^*)_{i=1}^n \subset X^*$  is as in Lemma 9.3.18. Hence we have that

$$\mathbb{E}\gamma[[M]]_t^{\frac{1}{2}} = \lim_{n \to \infty} \mathbb{E}\gamma[[M^m]]_t^{\frac{1}{2}} \stackrel{(*)}{\sim}_X \lim_{n \to \infty} \mathbb{E} \sup_{0 \le s \le t} \|M_s^m\| \stackrel{(**)}{=} \mathbb{E} \sup_{0 \le s \le t} \|M_s\|,$$

where (\*) follows from the first part of this case, and (\*\*) holds by (9.5.5).

Step 2: infinite dimensional case. First assume that M is an  $L^p$ -bounded martingale. Without loss of generality we can assume X to be separable. Since X is UMD, X is reflexive, so  $X^*$  is separable as well. Let  $Y_1 \subset Y_2 \subset ... \subset Y_n \subset ...$  be a family of finite dimensional subspaces of  $X^*$  such that  $\overline{\bigcup_n Y_n} = X^*$ . For each  $n \ge 1$  let  $P_n : Y_n \to X^*$  be the inclusion operator. Then  $||P_n^*|| \le 1$  and  $P_n^*M$  is a well-defined  $Y_n^*$ -valued  $L^p$ -bounded martingale. By Step 1 this martingale a.s. has a covariation bilinear form  $[|P_n^*M]|_t$  acting on  $Y_n \times Y_n$  and

$$\mathbb{E}\gamma([[P_n^*M]]_t)^p \stackrel{(*)}{\sim}_{p,X} \mathbb{E} \sup_{0 \le s \le t} \|P_n^*M_s\|^p \le \mathbb{E} \sup_{0 \le s \le t} \|M_s\|^p,$$
(9.5.6)

where (\*) is independent of *n* due to [79, Proposition 4.2.17]. Note that a.s.  $[[P_n^*M]]_t$  and  $[[P_m^*M]]_t$  agree for all  $m \ge n \ge 1$ , i.e. a.s.

$$[[P_m^*M]]_t(x^*, y^*) = [[P_n^*M]]_t(x^*, y^*) = [\langle M, x^* \rangle, \langle M, y^* \rangle]_t, \ x^*, y^* \in Y_n.$$
(9.5.7)

Let  $\Omega_0 \subset \Omega$  be a subset of measure 1 such that (9.5.7) holds for all  $m \ge n \ge 1$ . Fix  $\omega \in \Omega_0$ . Then by (9.5.7) we can define a bilinear form (not necessarily continuous!) V on  $Y \times Y$  (where  $Y := \bigcup_n Y_n \subset X^*$ ) such that  $V(x^*, y^*) = [[P_n^*M]]_t(x^*, y^*)$  for all  $x^*, y^* \in Y_n$  and  $n \ge 1$ .

Let us show that *V* is continuous (and hence has a continuous extension to  $X^* \times X^*$ ) and  $\gamma(V) < \infty$  a.s. on  $\Omega_0$ . Notice that by Lemma 9.3.7 the sequence  $(\gamma([[P_n^*M]]_t))_{n\geq 1}$  is increasing a.s. on  $\Omega_0$ . Moreover, by the monotone convergence theorem and (9.5.6)  $(\gamma([[P_n^*M]]_t))_{n\geq 1}$  has a limit a.s. on  $\Omega_0$ . Let  $\Omega_1 \subset \Omega_0$  be a subset of full measure such that  $(\gamma([[P_n^*M]]_t))_{n\geq 1}$  has a limit on  $\Omega_1$ . Then by (9.3.2) *V* is continuous on  $\Omega_1$  and hence has a continuous extension to  $X^* \times X^*$  (which we will denote by *V* as well for simplicity). Then by Proposition 9.3.8  $\gamma(V) = \lim_{n\to\infty} \gamma([[P_n^*M]]_t)$  monotonically on  $\Omega_1$  and hence by monotone convergence theorem ans the fact that  $\|P_n^*x\| \to \|x\|$  as  $n \to \infty$  monotonically for all  $x \in X$ 

$$\mathbb{E}\sup_{0\leq s\leq t} \|M_s\|^p = \lim_{n\to\infty} \mathbb{E}\sup_{0\leq s\leq t} \|P_n^*M_s\|^p = \sum_{p,X} \lim_{n\to\infty} \mathbb{E}\gamma([[P_n^*M]]_t)^p = \mathbb{E}(\gamma(V))^p.$$

It remains to show that  $V = [[M]]_t$  a.s., i.e.  $V(x^*, x^*) = [\langle M, x^* \rangle]_t$  a.s. for any  $x^* \in X^*$ . If  $x^* \in Y$ , then the desired follows from the construction of *V*. Fix  $x^* \in X^* \setminus Y$ . Since *Y* is dense in  $X^*$ , there exists a Cauchy sequence  $(x_n^*)_{n \ge 1}$  in *Y* converging to  $x^*$ . Then since  $V(x_n^*, x_n^*) = [\langle M, x_n^* \rangle]_t$  a.s. for all  $n \ge 1$ ,

$$\lim_{n \to \infty} |V(x_n^*, x_n^*) - [\langle M, x^* \rangle]_t|^{\frac{p}{2}} \lesssim_p \lim_{n \to \infty} [\langle M, x^* - x_n^* \rangle]_t^{\frac{p}{2}} \approx_p \lim_{n \to \infty} \mathbb{E} |\langle M, x^* - x_n^* \rangle|^p$$
  
$$\leq \lim_{n \to \infty} \mathbb{E} ||M||^p ||x^* - x_n^*||^p = 0,$$

so due to a.s. continuity of V,  $V(x^*, x^*)$  and  $[\langle M, x^* \rangle]_t$  coincide a.s.

Now let *M* be a general local martingale. By a stopping time argument we can assume that *M* is an  $L^1$ -bounded martingale, and then the existence of  $[[M]]_t$  follows from the case p = 1.

Let us now show (9.5.1). If the left-hand side is finite then M is an  $L^p$ -bounded martingale and the desired follows from the previous part of the proof. Let the left-hand side be infinite. Then it is sufficient to notice that by Step 1

$$\mathbb{E} \sup_{0\leq s\leq t} \|P_n^*M_s\|^p \approx_{p,X} \mathbb{E}\gamma([[P_n^*M]]_t)^p,$$

for any (finite or infinite) left-hand side, and the desired will follow as  $n \to \infty$  by the fact that  $||P_n^*M_s|| \to ||M_s||$  and  $\gamma([|P_n^*M]]_t) \to \gamma([|M]]_t)$  monotonically a.s. as  $n \to \infty$ , and the monotone convergence theorem.

*Remark* 9.5.3. Note that *X* being a UMD Banach space is necessary in Theorem 9.5.1 (see Theorem 9.2.4 and [126]).

*Remark* 9.5.4. Because of Lemma 9.3.5 the reader may suggest that if *X* is a UMD Banach space, then for any *X*-valued local martingale *M*, for any  $t \ge 0$ , and for a.a.  $\omega \in \Omega$  there exist a natural choice of a Hilbert space  $H(\omega)$  and a natural choice of an operator  $T(\omega) \in \mathcal{L}(H(\omega), X)$  such that for all  $x^*, y^* \in X^*$  a.s.

$$[[M]]_{t}(x^{*}, y^{*}) = \langle T^{*}x^{*}, T^{*}y^{*} \rangle.$$

If this is the case, then by Lemma 9.3.5 and Theorem 9.5.1

$$\mathbb{E}\sup_{0\leq s\leq t}\|M_s\|^p \approx_{p,X} \mathbb{E}\|T\|_{\gamma(H,X)}^p.$$

Such a natural pair of  $H(\omega)$  and  $T(\omega)$ ,  $\omega \in \Omega$ , is known for purely discontinuous local martingales (see Theorem 9.6.5) and for stochastic integrals (see Subsection 9.7.1 and 9.7.2). Unfortunately, it remains open how such *H* and *T* should look like for a general local martingale *M*.

#### 9.6. RAMIFICATIONS OF THEOREM 9.5.1

Let us outline some ramifications of Theorem 9.5.1.

#### 9.6.1. Continuous and purely discontinuous martingales

In the following theorems we will consider separately the cases of continuous and purely discontinuous martingales. First we show that if *M* is continuous, then Theorem 9.5.1 holds for the whole range 0 .

**Theorem 9.6.1.** Let X be a UMD Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a continuous local martingale. Then we have that for any 0

$$\mathbb{E}\sup_{0\le s\le t} \|M_s\|^p \approx_{p,X} \mathbb{E}\gamma([[M]]_t)^p, \quad t\ge 0.$$
(9.6.1)

For the proof we will need the following technical lemma.

**Lemma 9.6.2.** Let X be a UMD Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a continuous local martingale. Then the function  $t \mapsto \gamma([[M]]_t), t \ge 0$ , is continuous a.s.

*Proof.* By Pettis measurability theorem [79, Theorem 1.1.20] *X* can be assumed to be separable, and since *X* is UMD, it is reflexive, so  $X^*$  is separable as well. Therefore there exists a linearly independent set  $(x_n^*)_{n\geq 1} \subset X^*$  such that  $\overline{\text{span}(x_n^*)_{n\geq 1}} = X^*$ . By [89, Theorem 26.6(iv)] there exists a set  $\Omega_0 \in \Omega$  of full measure such that for any  $m, n \geq 1$  the function  $t \mapsto [\langle M, x_n^* \rangle, \langle M, x_m^* \rangle]_t$ ,  $t \geq 0$ , is continuous on  $\Omega_0$ , and such that  $[\langle M, x_n^* \rangle, \langle M, x_m^* \rangle]_t = [[M]]_t (x_n^*, x_m^*)$  on  $\Omega_0$  for all  $t \geq 0$  (see the construction of [[M]] in the proof of Theorem 9.5.1).

Fix  $y^* \in X^*$ . Let us show that  $t \mapsto [[M]]_t(y^*, y^*)$ ,  $t \ge 0$ , is continuous on  $\Omega_0$ . Let  $(y_k^*)_{k\ge 1} \in \text{span}(x_n^*)_{n\ge 1}$  be such that  $y_k^* \to y^*$  as  $k \to \infty$ . Fix  $\omega \in \Omega_0$  and  $t \ge 0$ . We know that  $[[M]]_s$ ,  $0 \le s \le t$ , defines a bounded symmetric bilinear form such that  $[[M]]_s(x^*, x^*) \ge 0$  for all  $x^* \in X^*$ . Therefore  $x^* \mapsto \sqrt{[[M]]_s(x^*, x^*)}$  defines a Euclidean norm, and hence since  $[[M]]_t(x^*, x^*) \ge [[M]]_s(x^*, x^*)$  for all  $x^* \in X^*$  and  $0 \le s \le t$ 

$$\sup_{0 \le s \le t} |[[M]]_{s}(y^{*}, y^{*}) - [[M]]_{s}(y^{*}_{k}, y^{*}_{k})| \lesssim \sup_{0 \le s \le t} |[[M]]_{s}(y^{*} - y^{*}_{k}, y^{*} - y^{*}_{k})| = [[M]]_{t}(y^{*} - y^{*}_{k}, y^{*} - y^{*}_{k}),$$
(9.6.2)

where the latter vanishes as  $k \to \infty$ . Notice that  $s \mapsto [[M]]_s(y_k^*, y_k^*), 0 \le s \le t$ , is continuous since  $[[M]]_s(y_k^*, y_k^*)$  is a finite linear combination of  $[[M]]_s(x_n^*, x_m^*)$  by bilinearity and because  $(y_k^*)_{k\ge 1} \in \text{span}(x_n^*)_{n\ge 1}$ . Consequently  $s \mapsto [[M]]_s(y^*, y^*), 0 \le s \le t$ , is continuous as well by (9.6.2).

Therefore we have that  $t \mapsto [[M]]_t(y^*, y^*), t \ge 0$ , is continuous for all  $y^* \in X^*$  on  $\Omega_0$ , and then continuity of  $t \mapsto \gamma([[M]]_t)$  on  $\Omega_0$  follows from Lemma 9.3.12.

*Proof of Theorem* 9.6.1. The case  $p \ge 1$  follows from Theorem 9.5.1. Let us treat the case  $0 . First we show that <math>(\gamma([[M]]_t))_{t\ge 0}$  is a predictable process:  $(\gamma([[M]]_t))_{t\ge 0}$  is a monotone limit of processes  $(\gamma([[P_n^*M]]_t))_{t\ge 0}$  (where  $P_n$ 's are as in the proof of Theorem 9.5.1), which are predictable due to the fact that  $([[P_n^*M]]_t)_{t\ge 0}$  is a  $Y_n^* \otimes Y_n^*$ -valued predictable process and  $\gamma : Y_n^* \otimes Y_n^* \to \mathbb{R}_+$  is a fixed measurable function. Moreover, by Lemma 9.6.2  $(\gamma([[M]]_t))_{t\ge 0}$  is continuous a.s., and by Remark 9.4.1 and Lemma 9.3.10  $(\gamma([[M]]_t))_{t\ge 0}$  is increasing a.s.

Now since  $(\gamma([[M]]_t))_{t\geq 0}$  is continuous predictable increasing, (9.6.1) follows from the case  $p \geq 1$  and Lenglart's inequality (see [106] and [156, Proposition IV.4.7]).

**Theorem 9.6.3.** Let X be a UMD Banach space,  $(M^n)_{n\geq 1}$  be a sequence of X-valued continuous local martingales such that  $M_0^n = 0$  for all  $n \geq 1$ . Then  $\sup_{t\geq 0} ||M_t^n|| \to 0$  in probability as  $n \to \infty$  if and only if  $\gamma([[M^n]]_{\infty}) \to 0$  in probability as  $n \to \infty$ .

*Proof.* The proof follows from the classical argument due do Lenglart (see [106]), but we will recall this argument for the convenience of the reader. We will show only one direction, the other direction follows analogously. Fix  $\varepsilon, \delta > 0$ . For each  $n \ge 1$  define a stopping time  $\tau_n$  in the following way:

$$\tau_n := \inf\{t \ge 0 : M_t^n > \varepsilon\}.$$

Then by (9.5.1) and Chebyshev's inequality

$$\begin{split} \mathbb{P}(\gamma(\llbracket M^n \rrbracket]_{\infty}) > \delta) &\leq \mathbb{P}(\tau_n < \infty) + \mathbb{P}(\gamma(\llbracket M^n \rrbracket]_{\tau_n}) > \delta) \\ &\leq \mathbb{P}(\sup_{t \ge 0} \|M_t^n\| > \varepsilon) + \delta^{-\frac{1}{2}} \mathbb{E}\gamma(\llbracket M^n \rrbracket]_{\tau_n})^{\frac{1}{2}} \\ &\lesssim_X \mathbb{P}(\sup_{t \ge 0} \|M_t^n\| > \varepsilon) + \delta^{-\frac{1}{2}} \mathbb{E}\|M_{\tau_n}^n\| \\ &\leq \mathbb{P}(\sup_{t \ge 0} \|M_t^n\| > \varepsilon) + \delta^{-\frac{1}{2}} \varepsilon, \end{split}$$

and the latter vanishes for any fixed  $\delta > 0$  as  $\varepsilon \to 0$  and  $n \to \infty$ .

*Remark* 9.6.4. Note that Theorem 9.6.3 does not hold for general martingales even in the real-valued case, see [89, Exercise 26.5].

For the next theorem recall that  $\ell^2([0, t])$  is the *nonseparable* Hilbert space consisting of all functions  $f : [0, t] \to \mathbb{R}$  which support  $\{s \in [0, t] : f(s) \neq 0\}$  is countable and  $||f||_{\ell^2([0,t])} := \sum_{0 \le s \le t} |f(s)|^2 < \infty$ .

**Theorem 9.6.5.** Let X be a UMD Banach space,  $1 \le p < \infty$ ,  $M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous martingale. Then for any  $t \ge 0$ 

$$\mathbb{E}\sup_{0\leq s\leq t}\|M_s\|^p \approx_{p,X} \mathbb{E}\|(\Delta M_s)_{0\leq s\leq t}\|_{\gamma(\ell^2([0,t]),X)}^p$$

*Proof.* It is sufficient to notice that for any  $x^* \in X^*$  a.s.

$$[\langle M, x^* \rangle]_t = \sum_{0 \le s \le t} |\langle \Delta M_s, x^* \rangle|^2,$$

and apply Theorem 9.5.1 and Lemma 9.3.5.

*Remark* 9.6.6. Note that martingales in Theorem 9.6.1 and 9.6.5 cover all the martingales if *X* is UMD. More specifically, if *X* has the UMD property, then any *X*-valued local martingale *M* has a unique decomposition  $M = M^c + M^d$  into a sum of a continuous local martingale  $M^c$  and a purely discontinuous local martingale  $M^d$  (see Chapter 4 and 5).

#### 9.6.2. Martingales with independent increments

Here we show that both Theorem 9.2.1 and 9.5.1 hold in much more general Banach spaces given the corresponding martingale has independent increments.

**Proposition 9.6.7.** Let X be a Banach space,  $(d_n)_{n\geq 1}$  be an X-valued martingale difference sequence with independent increments. Then for any 1

$$\mathbb{E}\sup_{m\geq 1}\left\|\sum_{n=1}^{m}d_{n}\right\|^{p}\lesssim_{p}\mathbb{E}\left\|(d_{n})_{n=1}^{\infty}\right\|_{\gamma(\ell^{2},X)}^{p}.$$

*Moreover, if X has a finite cotype, then* 

$$\mathbb{E}\sup_{m\geq 1}\left\|\sum_{n=1}^{m}d_{n}\right\|^{p} \approx_{p,X} \mathbb{E}\left\|(d_{n})_{n=1}^{\infty}\right\|_{\gamma(\ell^{2},X)}^{p}.$$

*Proof.* Let  $(r_n)_{n\geq 1}$  be a sequence of independent Rademacher random variables,  $(\gamma_n)_{n\geq 1}$  be a sequence of independent standard Gaussian random variables. Then

$$\mathbb{E}\sup_{m\geq 1}\left\|\sum_{n=1}^{m}d_{n}\right\|^{p} \stackrel{(i)}{\approx}_{p} \mathbb{E}\left\|\sum_{n=1}^{\infty}d_{n}\right\|^{p} \stackrel{(ii)}{\approx}_{p} \mathbb{E}\mathbb{E}_{r}\left\|\sum_{n=1}^{N}r_{n}d_{n}\right\|^{p} \stackrel{(iii)}{\approx}_{p} \mathbb{E}\mathbb{E}_{\gamma}\left\|\sum_{n=1}^{N}\gamma_{n}d_{n}\right\|^{p} \stackrel{(iv)}{\approx}_{p} \mathbb{E}\left(\mathbb{E}_{\gamma}\left\|\sum_{n=1}^{N}\gamma_{n}d_{n}\right\|^{2}\right)^{\frac{p}{2}} = \mathbb{E}\|(d_{n})_{n=1}^{\infty}\|_{\gamma(\ell^{2},\chi)}^{p},$$

where (*i*) follows from (2.2.1), (*ii*) follows from [104, Lemma 6.3], (*iii*) holds by [80, Proposition 6.3.2], and finally (*iv*) follows from [80, Proposition 6.3.1].

If *X* has a finite cotype, then one has  $\eqsim_{p,X}$  instead of  $\leq_p$  in (*iii*) (see [80, Corollary 7.2.10]), and the second part of the proposition follows.

Based on Proposition 9.6.7 and the proof of Theorem 9.5.1 one can show the following assertion. Notice that we presume the reflexivity of *X* since it was assumed in the whole Section 9.3.

**Proposition 9.6.8.** Let X be a reflexive Banach space,  $1 \le p < \infty$ ,  $M : \mathbb{R}_+ \times \Omega \to X$  be an  $L^p$ -bounded martingale with independent increments such that  $M_0 = 0$ . Let  $t \ge 0$ . If M has a covariation bilinear form  $[[M]]_t$  at t, then

$$\mathbb{E}\sup_{0\leq s\leq t}\|M_s\|^p\lesssim_{p,X}\mathbb{E}\gamma([[M]]_t)^p.$$

Moreover, if X has a finite cotype, then the existence of  $[[M]]_t$  is guaranteed, and

$$\mathbb{E}\sup_{0\leq s\leq t}\|M_s\|^p \approx_{p,X} \mathbb{E}\gamma([[M]]_t)^p.$$

# 9.7. APPLICATIONS AND MISCELLANEA

Here we provide further applications of Theorem 9.5.1.

#### 9.7.1. Itô isomorphism: general martingales

Let *H* be a Hilbert space, *X* be a Banach space. For each  $x \in X$  and  $h \in H$  we denote the linear operator  $g \mapsto \langle g, h \rangle x$ ,  $g \in H$ , by  $h \otimes x$ . The process  $\Phi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(H, X)$  is called *elementary predictable* with respect to the filtration  $\mathbb{F} = (\mathscr{F}_t)_{t \ge 0}$  if it is of the form

$$\Phi(t,\omega) = \sum_{k=1}^{K} \sum_{m=1}^{M} \mathbf{1}_{(t_{k-1},t_k] \times B_{mk}}(t,\omega) \sum_{n=1}^{N} h_n \otimes x_{kmn}, \quad t \ge 0, \ \omega \in \Omega,$$

where  $0 = t_0 < ... < t_K < \infty$ , for each k = 1, ..., K the sets  $B_{1k}, ..., B_{Mk}$  are in  $\mathscr{F}_{t_{k-1}}$  and the vectors  $h_1, ..., h_N$  are in H. Let  $\widetilde{M} : \mathbb{R}_+ \times \Omega \to H$  be a local martingale. Then we define the *stochastic integral*  $\Phi \cdot \widetilde{M} : \mathbb{R}_+ \times \Omega \to X$  of  $\Phi$  with respect to  $\widetilde{M}$  as follows:

$$(\Phi \cdot \widetilde{M})_t := \sum_{k=1}^K \sum_{m=1}^M \mathbf{1}_{B_{mk}} \sum_{n=1}^N \langle (\widetilde{M}(t_k \wedge t) - \widetilde{M}(t_{k-1} \wedge t)), h_n \rangle x_{kmn}, \ t \ge 0.$$
(9.7.1)

Notice that for any  $t \ge 0$  the stochastic integral  $\Phi \cdot \widetilde{M}$  obtains a covariation bilinear form  $[[\Phi \cdot \widetilde{M}]]_t$  which is a.s. continuous on  $X^* \times X^*$  and which has the following form due to (2.2.6) and (9.7.1)

$$[ [\Phi \cdot \widetilde{M} ] ]_t(x^*, x^*) = \left[ \left\langle \int_0^{\cdot} \Phi \, \mathrm{d}\widetilde{M}, x^* \right\rangle \right]_t = \left[ \int_0^{\cdot} (\Phi^* x^*)^* \, \mathrm{d}\widetilde{M} \right]_t$$

$$= \int_0^t \| q_{\widetilde{M}}^{1/2}(s) \Phi^*(s) x^* \|^2 \, \mathrm{d}[\widetilde{M}]_s, \quad t \ge 0.$$

$$(9.7.2)$$

*Remark* 9.7.1. If  $X = \mathbb{R}$ , then by the real-valued Burkholder–Davis–Gundy inequality and the fact that for any elementary predictable  $\Phi$ 

$$\left[\int_0^{\cdot} \Phi \,\mathrm{d}\widetilde{M}\right]_t = \int_0^t \|q_{\widetilde{M}}^{1/2}(s)\Phi^*(s)\|^2 \,\mathrm{d}[\widetilde{M}]_s, \ t \ge 0,$$

one has an isomorphism

$$\mathbb{E}\sup_{t\geq 0} |(\Phi\cdot\widetilde{M})_t| \sim \mathbb{E}\left(\int_0^\infty \|q_{\widetilde{M}}^{1/2}(s)\Phi(s)\|^2 d[\widetilde{M}]_s\right)^{\frac{1}{2}},$$

so one can extend the definition of a stochastic integral to *all* predictable  $\Phi$  :  $\mathbb{R}_+ \times \Omega \rightarrow H$  with

$$\mathbb{E}\left(\int_0^\infty \|q_{\widetilde{M}}^{1/2}(s)\Phi(s)\|^2 d[\widetilde{M}]_s\right)^{\frac{1}{2}} < \infty,$$
(9.7.3)

by extending the stochastic integral operator from a dense subspace of all elementary predictable processes satisfying (9.7.3). We refer the reader to [89, 121, 123] for details.

*Remark* 9.7.2. Let  $X = \mathbb{R}^d$  for some  $d \ge 1$ . Then analogously to Remark 9.7.1 one can extend the definition of a stochastic integral to all predictable processes  $\Phi$  :  $\mathbb{R}_+ \times \Omega \rightarrow \mathcal{L}(H, \mathbb{R}^d)$  with

$$\mathbb{E}\left(\sum_{n=1}^{d}\int_{0}^{\infty}\|q_{\widetilde{M}}^{1/2}(s)\Phi^{*}(s)e_{n}\|^{2}d[\widetilde{M}]_{s}\right)^{\frac{1}{2}} = \mathbb{E}\|q_{\widetilde{M}}^{1/2}\Phi^{*}\|_{HS(\mathbb{R}^{d},L^{2}(\mathbb{R}_{+};[\widetilde{M}]))}$$
$$= \mathbb{E}\|\Phi q_{\widetilde{M}}^{1/2}\|_{HS(L^{2}(\mathbb{R}_{+};[\widetilde{M}]),\mathbb{R}^{d})} < \infty.$$

where  $(e_n)_{n=1}^d$  is a basis of  $\mathbb{R}^d$ ,  $||T||_{HS(H_1,H_2)}$  is the Hilbert-Schmidt norm of an operator *T* acting form a Hilbert space  $H_1$  to a Hilbert space  $H_2$ , and  $L^2(\mathbb{R}_+; A)$  for a given increasing  $A : \mathbb{R}_+ \to \mathbb{R}$  is a Hilbert space of all functions  $f : \mathbb{R}_+ \to \mathbb{R}$  such that  $\int_{\mathbb{R}_+} ||f(s)||^2 dA(s) < \infty$ .

Now we present the Itô isomorphism for vector-valued stochastic integrals with respect to general martingales, which extends [126, 175, 177].

**Theorem 9.7.3.** Let H be a Hilbert space, X be a UMD Banach space,  $\widetilde{M} : \mathbb{R}_+ \times \Omega \to H$  be a local martingale,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathcal{L}(H, X)$  be elementary predictable. Then for all  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{0\leq s\leq t}\left\|\int_{0}^{s}\Phi\,\mathrm{d}\widetilde{M}\right\|^{p} \approx_{p,X} \mathbb{E}\|\Phi q_{\widetilde{M}}^{1/2}\|_{\gamma(L^{2}([0,t],[\widetilde{M}];H),X)}^{p}, \quad t\geq 0,$$

where  $[\widetilde{M}]$  is the quadratic variation of  $\widetilde{M}$ ,  $q_{\widetilde{M}}$  is the quadratic variation derivative (see Section 2.2.1), and  $\|\Phi q_{\widetilde{M}}^{1/2}\|_{\gamma(L^2([0,t],[\widetilde{M}];H),X)}^p$  is the  $\gamma$ -norm (see (2.9.1)).

*Proof.* Fix  $t \ge 0$ . Then the theorem holds by Theorem 9.5.1, Lemma 9.3.5, and the fact that by (9.7.2) for any fixed  $x^* \in X^*$  a.s.

$$\begin{split} \left[\left\langle \int_0^{\cdot} \Phi \, \mathrm{d}\widetilde{M}, x^* \right\rangle \right]_t &= \left[\int_0^{\cdot} \langle \Phi, x^* \rangle \, \mathrm{d}\widetilde{M}\right]_t = \int_0^t \|q_M^{\frac{1}{2}} \Phi^* x^*\|^2 \, \mathrm{d}[\widetilde{M}]_s \\ &= \|q_M^{\frac{1}{2}} \Phi^* x^*\|_{L^2([0,t],[\widetilde{M}];H)}^2. \end{split}$$

265

Theorem 9.7.3 allows us to provide the following general stochastic integration result. Recall that a predictable process  $\Phi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(H, X)$  is called *strongly predictable* if there exists a sequence  $(\Phi_n)_{n\geq 1}$  of elementary predictable  $\mathscr{L}(H, X)$ -valued processes such that  $\Phi$  is a pointwise limit of  $(\Phi_n)_{n\geq 1}$ .

**Corollary 9.7.4.** Let H be a Hilbert space, X be a UMD Banach space,  $\widetilde{M} : \mathbb{R}_+ \times \Omega \to H$  be a local martingale,  $\Phi : \mathbb{R}_+ \times \Omega \to \mathscr{L}(H, X)$  be strongly predictable such that  $\mathbb{E} \| \Phi q_{\widetilde{M}}^{1/2} \|_{\gamma(L^2(\mathbb{R}_+, [\widetilde{M}]; H), X)} < \infty$ . Then there exists a martingale  $\Phi \cdot \widetilde{M}$  which coincides with the stochastic integral given  $\Phi$  is elementary predictable such that

$$\langle \Phi \cdot \widetilde{M}, x^* \rangle = (\Phi^* x^*) \cdot \widetilde{M}, \quad x^* \in X^*, \tag{9.7.4}$$

*where the latter integral is defined as in Remark* 9.7.1*. Moreover, then we have that for any*  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{t\geq 0} \|(\Phi\cdot\widetilde{M})_t\|^p \approx_{p,X} \mathbb{E}\|\Phi q_{\widetilde{M}}^{1/2}\|_{\gamma(L^2(\mathbb{R}_+,[\widetilde{M}];H),X)}^p.$$
(9.7.5)

For the proof we will need the following technical lemma.

**Lemma 9.7.5.** Let X be a reflexive separable Banach space,  $Y_1 \subset Y_2 \subset ... \subset Y_n \subset ... \subset X^*$ be finite dimensional subspaces such that  $\overline{\bigcup_n Y_n} = X^*$ . Let  $P_n : Y_n \hookrightarrow X^*$ ,  $n \ge 1$ , and  $P_{n,m} : Y_n \hookrightarrow Y_m$ ,  $m \ge n \ge 1$ , be the inclusion operators. For each  $n \ge 1$  let  $x_n \in Y_n^*$  be such that  $P_{n,m}^* x_m = x_n$  for all  $m \ge n \ge 1$ . Assume also that  $\sup_n ||x_n|| < \infty$ . Then there exists  $x \in X$  such that  $P_n^* x = x_n$  for all  $n \ge 1$  and  $||x|| = \lim_{n \to \infty} ||x_n||$  monotonically.

*Proof.* Set  $C = \sup_n ||x_n||$ . First notice that  $(x_n)_{n\geq 1}$  defines a bounded linear functional on  $Y = \bigcup_n Y_n$ . Indeed, fix  $y \in Y_n$  for some fixed  $n \geq 1$  (then automatically  $y \in Y_m$  for any  $m \geq n$ ). Define  $\ell(y) = \langle x_n, y \rangle$ . Then this definition of  $\ell$  agrees for different *n*'s since for any  $m \geq n$  we have that

$$\langle x_m, y_n \rangle = \langle x_m, P_{n,m} y_n \rangle = \langle P_{n,m}^* x_m, y_n \rangle = \langle x_n, y_n \rangle.$$

Moreover, this linear functional is bounded since  $|\langle x_n, y_n \rangle| \le ||x_n|| ||y_n|| \le C ||y_n||$ . So, it can be continuously extended to the whole space  $X^*$ . Since X is reflexive, there exists  $x \in X$  such that  $\ell(x^*) = \langle x^*, x \rangle$  for any  $x^* \in X^*$ . Then for any fixed  $n \ge 1$  and for any  $y \in Y_n$  we have that

$$\langle x_n, y \rangle = \ell(y) = \langle x, y \rangle = \langle x, P_n y \rangle = \langle P_n^* x, y \rangle,$$

so  $P_n^* x = x_n$ . The latter follows from the fact that  $||P_n^* x|| \to ||x||$  monotonically as  $n \to \infty$  for any  $x \in X$ .

*Proof of Corollary* 9.7.4. We will first consider the finite dimensional case and then deduce the infinite dimensional case.

*Finite dimensional case.* Since *X* is finite dimensional, it is isomorphic to a finite dimensional Euclidean space, and so the  $\gamma$ -norm is equivalent to the Hilbert-Schmidt norm (see e.g. [80, Proposition 9.1.9]). Then  $\Phi$  is stochastically integrable

with respect to  $\widetilde{M}$  due to Remark 9.7.2, so (9.7.4) clearly holds and we have that for any  $x^* \in X^*$  a.s.

$$[\langle \Phi \cdot \widetilde{M}, x^* \rangle]_t = [(\Phi^* x^*) \cdot \widetilde{M}]_t = \int_0^t \|q_{\widetilde{M}}^{1/2}(s) \Phi^*(s) x^*\|^2 d[\widetilde{M}]_s, \quad t \ge 0,$$

thus (9.7.5) follows from Theorem 9.5.1 and Lemma 9.3.5.

Infinite dimensional case. Let now *X* be general. Since  $\Phi$  is strongly predictable, it takes values in a separable subspace of *X*, so we may assume that *X* is separable. Since *X* is UMD, it is reflexive, so *X*<sup>\*</sup> is separable as well, and there exists a sequence  $Y_1 \subset Y_2 \subset ... \subset Y_n \subset ... \subset X^*$  of finite dimensional subsets of *X*<sup>\*</sup> such that  $\overline{\bigcup_n Y_n} = X^*$ . For each  $m \ge n \ge 1$  define inclusion operators  $P_n : Y_n \hookrightarrow$  $X^*$  and  $P_{n,m} : Y_n \hookrightarrow Y_m$ . Notice that by the ideal property [80, Theorem 9.1.10]  $\mathbb{E} \| P_n^* \Phi q_{\widetilde{M}}^{1/2} \|_{\gamma(L^2(\mathbb{R}_+, [\widetilde{M}]; H), Y_n^*)} < \infty$  for any  $n \ge 1$ , so since  $Y_n^*$  is finite dimensional, the stochastic integral  $(P_n^* \Phi) \cdot \widetilde{M}$  is well-defined by the case above and

$$\mathbb{E}\sup_{t\geq 0} \left\| \left( (P_n^*\Phi) \cdot \widetilde{M} \right)_t \right\| \approx_X \mathbb{E} \| P_n^*\Phi q_{\widetilde{M}}^{1/2} \|_{\gamma(L^2(\mathbb{R}_+, [\widetilde{M}]; H), Y_n^*)},$$
(9.7.6)

where the equivalence is independent of *n* since  $Y_n \subset X^*$  for all  $n \ge 1$  and due to [79, Proposition 4.2.17] and Theorem 9.5.1. Denote the stochastic integral  $(P_n^*\Phi) \cdot \widetilde{M}$  by  $Z^n$ . Note that  $Z^n$  is  $Y_n^*$ -valued, and since  $P_{n,m}^*P_m^*\Phi = P_n^*\Phi$  for all  $m \ge n \ge 1$ ,  $P_{m,n}^*Z_t^m = Z_t^n$  a.s. for any  $t \ge 0$ . Therefore by Lemma 9.7.5 there exists a process  $Z : \mathbb{R}_+ \times \Omega \to X$  such that  $P_n^*Z = Z^n$  for all  $n \ge 1$ . Let us show that Z is integrable. Fix  $t \ge 1$ . Notice that by Lemma 9.7.5 the limit  $||Z_t|| = \lim_{n\to\infty} ||P_n^*Z_t|| = \lim_{n\to\infty} ||Z_t^n||$  is monotone, so by the monotone convergence theorem, (9.7.6), and the ideal property [80, Theorem 9.1.10]

$$\mathbb{E}\|Z_t\| = \lim_{n \to \infty} \mathbb{E}\|Z_t^n\| \lesssim_X \limsup_{n \to \infty} \mathbb{E}\|P_n^* \Phi q_{\widetilde{M}}^{1/2}\|_{\gamma(L^2(\mathbb{R}_+, [\widetilde{M}]; H), Y_n^*)} \le \mathbb{E}\|\Phi q_{\widetilde{M}}^{1/2}\|_{\gamma(L^2(\mathbb{R}_+, [\widetilde{M}]; H), X)}.$$

Now let us show that *Z* is a martingale. Since *Z* is integrable, due to [79, Section 2.6] it is sufficient to show that  $\mathbb{E}(\langle Z_t, x^* \rangle | \mathscr{F}_s) = \langle Z_s, x^* \rangle$  for all  $0 \le s \le t$  for all  $x^*$  from some dense subspace *Y* of  $X^*$ . Set  $Y = \bigcup_n Y_n$  and  $x^* \in Y_n$  for some  $n \ge 1$ . Then for all  $0 \le s \le t$ 

$$\mathbb{E}(\langle Z_t, x^* \rangle | \mathscr{F}_s) = \mathbb{E}(\langle Z_t, P_n x^* \rangle | \mathscr{F}_s) = \mathbb{E}(\langle P_n^* Z_t, x^* \rangle | \mathscr{F}_s)$$
$$= \mathbb{E}(\langle Z_t^n, x^* \rangle | \mathscr{F}_s) = \langle Z_s^n, x^* \rangle = \langle Z_s, x^* \rangle,$$

so *Z* is a martingale. Finally, let us show (9.7.5). First notice that for any  $n \ge 1$  and  $x^* \in Y_n \subset X^*$  a.s.

$$[\langle Z, x^* \rangle]_t = [\langle Z^n, x^* \rangle]_t = \int_0^t \|q_{\widetilde{M}}^{1/2}(s)\Phi^*(s)x^*\|^2 d[\widetilde{M}]_s, \quad t \ge 0;$$

the same holds for a general  $x^* \in X^*$  by a density argument. Then (9.7.5) follows from Theorem 9.5.1 and Lemma 9.3.5.

*Remark* 9.7.6. As the reader can judge, the basic assumptions on  $\Phi$  in Corollary 9.7.4 can be weakened by a stopping time argument. Namely, one can assume that  $\Phi q_{\widetilde{M}}^{1/2}$  is *locally* in  $L^1(\Omega, \gamma(L^2(\mathbb{R}_+, [\widetilde{M}]; H), X))$  (i.e. there exists an increasing sequence  $(\tau_n)_{n\geq 1}$  of stopping times such that  $\tau_n \to \infty$  a.s. as  $n \to \infty$  and  $\Phi q_{\widetilde{M}}^{1/2} \mathbf{1}_{[0,\tau_n]}$  is in  $L^1(\Omega, \gamma(L^2(\mathbb{R}_+, [\widetilde{M}]; H), X))$  for all  $n \geq 1$ ). Notice that such an assumption is a natural generalization of classical assumptions for stochastic integration in the real-valued case (see e.g. [89, p. 526]).

#### 9.7.2. Itô isomorphism: Poisson and general random measures

Let  $(J, \mathscr{J})$  be a measurable space, N be a Poisson random measure on  $J \times \mathbb{R}_+$ ,  $\tilde{N}$  be the corresponding compensated Poisson random measure (see e.g. [51, 89, 95, 165] and Section 2.8 for details). Then by Theorem 9.6.5 for any UMD Banach space X, for any  $1 \le p < \infty$ , and for any elementary predictable  $F : J \times \mathbb{R}_+ \times \Omega \to X$  we have that

$$\mathbb{E}\sup_{0\leq s\leq t}\left\|\int_{J\times[0,s]}F\,\mathrm{d}\widetilde{N}\right\|^p \approx_{p,X} \mathbb{E}\|F\|_{\gamma(L^2(J\times[0,t];N),X)}^p, \quad t\geq 0.$$
(9.7.7)

The same holds for a general quasi-left continuous random measure (see Section 2.8 for the definition and the details): if  $\mu$  is a general quasi-left continuous random measure on  $J \times \mathbb{R}_+$ , v is its compensator, and  $\bar{\mu} := \mu - v$ , then for any  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{0\le s\le t} \left\| \int_{J\times[0,t]} F \,\mathrm{d}\bar{\mu} \right\|^p \approx_{p,X} \mathbb{E}\|F\|_{\gamma(L^2(J\times[0,t];\mu),X)}^p, \quad t\ge 0.$$
(9.7.8)

The disadvantage of right-hand sides of (9.7.7) and (9.7.8) is that both of them are not predictable and do not depend continuously on time a.s. on  $\Omega$  (therefore they seem not to be useful from the SPDE's point of view since one may not produce a fixed point argument). For example, if  $X = L^q$  for some  $1 < q < \infty$ , then such predictable a.s. continuous in time right-hand sides do exist (see [51] and Chapter 7).

# 9.7.3. Necessity of the UMD property

As it follows from Remark 9.5.3, Theorem 9.5.1 holds only in the UMD setting. The natural question is whether there exists an appropriate right-hand side of (9.5.1) in terms of  $([\langle M, x^* \rangle, \langle M, y^* \rangle])_{x^*,y^* \in X^*}$  for some non-UMD Banach space *X* and some  $1 \le p < \infty$ . Here we show that this is impossible.

Assume that for some Banach space *X* and some  $1 \le p < \infty$  there exists a function *G* acting on families of stochastic processes parametrized by  $X^* \times X^*$  (i.e. each family has the form  $V = (V_{X^*,y^*})_{x^*,y^* \in X^*}$ ) taking values in  $\mathbb{R}$  such that for any *X*-valued local martingale *M* starting in zero we have that

$$\mathbb{E}\sup_{t\geq 0} \|M_t\|^p \approx_{p,X} G([[M]]), \tag{9.7.9}$$

where we denote  $[[M]] = ([\langle M, x^* \rangle, \langle M, y^* \rangle])_{x^*, y^* \in X^*}$  for simplicity (note that the latter might not have a proper bilinear structure). Let us show that then *X* must have the UMD property.

Fix any *X*-valued  $L^p$ -bounded martingale difference sequence  $(d_n)_{n=1}^N$  and any  $\{-1, 1\}$ -valued sequence  $(\varepsilon_n)_{n=1}^N$ . Let  $e_n := \varepsilon_n d_n$  for all n = 1, ..., N. For every  $x^*, y^* \in X^*$  define a stochastic process  $V_{x^*, y^*} : \mathbb{R}_+ \times \Omega \to \mathbb{R}$  as

$$V_{x^*}(t) = \sum_{n=1}^{N \wedge [t]} \langle d_n, x^* \rangle \cdot \langle d_n, y^* \rangle = \sum_{n=1}^{N \wedge [t]} \langle e_n, x^* \rangle \cdot \langle e_n, y^* \rangle, \quad t \ge 0$$

(recall that [t] is the integer part of t). Let  $V := (V_{x^*,y^*})_{x^*,y^* \in X^*}$ . Then by (9.7.9)

$$\mathbb{E}\sup_{k\geq 0} \left\| \sum_{n=1}^{k} d_n \right\|^p \approx_{p,X} G(V) \approx_{p,X} \mathbb{E}\sup_{k\geq 0} \left\| \sum_{n=1}^{k} e_n \right\|^p.$$
(9.7.10)

Since *N*,  $(d_n)_{n=1}^N$ , and  $(\varepsilon_n)_{n=1}^N$  are general, (9.7.10) implies that *X* is a UMD Banach space (see the proof of Theorem 9.2.4).

#### 9.7.4. Martingale domination

The next theorem shows that under some natural domination assumptions on martingales one gets  $L^p$ -estimates.

**Theorem 9.7.7.** Let X be a UMD Banach space,  $M, N : \mathbb{R}_+ \times \Omega \to X$  be local martingales such that  $||N_0|| \le ||M_0||$  a.s. and  $[\langle N, x^* \rangle]_{\infty} \le [\langle M, x^* \rangle]_{\infty}$  a.s. for all  $x^* \in X^*$ . Then for all  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{t\geq 0} \|N_t\|^p \lesssim_{p,X} \mathbb{E}\sup_{t\geq 0} \|M_t\|^p.$$
(9.7.11)

Note that the assumptions in Theorem 9.7.7 are a way more general than the weak differential subordination assumptions, so Theorem 9.7.7 significantly improves Theorem 4.4.1 and 6.4.26, and extends these results to the case p = 1 as well.

*Proof of Theorem* 9.7.7. First notice that by a triangular inequality

$$\mathbb{E}\sup_{t\geq 0} \|M_t\|^p \approx_p \mathbb{E}\|M_0\|^p + \mathbb{E}\sup_{t\geq 0} \|M_t - M_0\|^p,$$
  
$$\mathbb{E}\sup_{t\geq 0} \|N_t\|^p \approx_p \mathbb{E}\|N_0\|^p + \mathbb{E}\sup_{t\geq 0} \|N_t - N_0\|^p.$$

Consequently we can reduce the statement to the case  $M_0 = N_0 = 0$  a.s. (by setting  $M := M - M_0$ ,  $N := N - N_0$ ), and then the proof follows directly from Theorem 9.5.1 and Lemma 9.3.10.

*Remark* 9.7.8. It is not known what the sharp constant is in (9.7.11). Nevertheless, sharp inequalities of such type have been discovered in the scalar case by

Osękowski in [137]. It was shown there that if M and N are real-valued  $L^p$ -bounded martingales such that a.s.

$$[N]_t \le [M]_t, t \ge 0, \text{ if } 1$$

$$[N]_{\infty} - [N]_{t-} \le [M]_{\infty} - [M]_{t-}, t \ge 0, \text{ if } 2 \le p < \infty,$$

then

$$(\mathbb{E}|N_{\infty}|^{p})^{\frac{1}{p}} \le (p^{*}-1)(\mathbb{E}|M_{\infty}|^{p})^{\frac{1}{p}}, \ 1$$

where  $p^* := \max\{p, \frac{p}{p-1}\}.$ 

#### 9.7.5. Martingale approximations

The current subsection is devoted to approximation of martingales. Namely, we will extend Lemma 9.5.2 by Weisz (see [180, Theorem 6]) to general UMD Banach space-valued martingales. Here is the main theorem of the current subsection.

**Theorem 9.7.9.** Let X be a UMD Banach space,  $1 \le p < \infty$ ,  $M : \mathbb{R}_+ \times \Omega \to X$  be a martingale such that  $\mathbb{E}\sup_{t\ge 0} \|M_t\|^p < \infty$ . Then there exists a sequence  $(M^n)_{n\ge 1}$  of X-valued  $L^{\infty}$ -bounded martingales such that  $\mathbb{E}\sup_{t\ge 0} \|M_t - M_t^n\|^p \to 0$  as  $n \to \infty$ .

In order to prove Theorem 9.7.9 we will need to show similar approximation results for quasi-left continuous purely discontinuous martingales and purely discontinuous martingales with accessible jumps. Both cases will be considered separately.

#### QUASI-LEFT CONTINUOUS PURELY DISCONTINUOUS MARTINGALES

Before stating the corresponding approximation theorem let us show the following proposition.

**Proposition 9.7.10.** Let X be a Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous quasi-left continuous martingale. Then there exist sequences of positive numbers  $(a_n)_{n\geq 1}$ ,  $(b_n)_{n\geq 1}$ , and a sequence of X-valued purely discontinuous quasi-left continuous martingales  $(M^n)_{n\geq 1}$  such that

$$\sup_{t} \|\Delta M_{t}^{n}\| \le a_{n}, \quad \#\{t \ge 0 : \Delta M_{t}^{n} \ne 0\} \le b_{n} \quad a.s. \quad \forall n \ge 1,$$
$$\{t \ge 0 : \Delta M_{t}^{n} \ne 0\} \subset \{t \ge 0 : \Delta M_{t}^{m} \ne 0\} \quad a.s. \quad \forall m \ge n \ge 1,$$
(9.7.12)

$$\Delta M_t^n = \Delta M_t \quad \forall t \ge 0 \ s.t. \ \Delta M_t^n \neq 0 \ a.s. \ \forall n \ge 1, \tag{9.7.13}$$

and

$$\cup_{n \ge 1} \{t \ge 0 : \Delta M_t^n \neq 0\} = \{t \ge 0 : \Delta M_t \neq 0\} \ a.s.$$
(9.7.14)

*Sketch of the proof.* Let  $\mu^M$  be a random measure defined on  $(\mathbb{R}_+ \times X, \mathscr{B}(\mathbb{R}_+) \otimes \mathscr{B}(X))$  by

$$\mu^{M}(A \times B) = \sum_{t \in A} \mathbf{1}_{\Delta M_{t} \in B \setminus \{0\}}, \ A \in \mathscr{B}(\mathbb{R}_{+}), B \in \mathscr{B}(X)$$

Let  $v^M$  be the corresponding compensator,  $\bar{\mu}^M := \mu^M - v^M$ . Due to the proof of Lemma 7.5.11 there exists an a.s. increasing sequence  $(\tau_n)_{n\geq 1}$  of stopping times such  $\tau_n \to \infty$  a.s. as  $n \to \infty$ , and such = that there exist positive sequences  $(a_n)_{n\geq 1}$ ,  $(b_n)_{n\geq 1}$  with  $(a_n)_{n\geq 1}$  being increasing natural and with

$$\#\{t \ge 0 : \|\Delta M_t^{\tau_n}\| \in [1/a_n, a_n]\} \le b_n.$$

Define a predictable set  $A_n := [0, \tau_n] \times B_n \subset \mathbb{R}_+ \times X$ , where  $B_n := \{x \in X : ||x|| \in [1/a_n, a_n]\}$ . Then the desired  $M^n$  equals the stochastic integral

$$M_t^n := \int_{[0,t]\times X} \mathbf{1}_{A_n}(s,x) x \,\mathrm{d}\bar{\mu}^M(\,\mathrm{d} s,\,\mathrm{d} x), \quad t \ge 0,$$

where the latter is a well-defined martingale since by Section 2.8 it is sufficient to check that for any  $t \ge 0$ 

$$\begin{aligned} \int_{[0,t]\times X} \|\mathbf{1}_{A_n}(s,x)x\| \,\mathrm{d}\mu^M(\,\mathrm{d} s,\,\mathrm{d} x) &= \int_{A_n\cap[0,t]\times X} \|x\| \,\mathrm{d}\mu^M(\,\mathrm{d} s,\,\mathrm{d} x) \\ &= \sum_{t\in[0,\tau_n\wedge t]} \|\Delta M_t^{\tau_n}\|\mathbf{1}_{\Delta M_t^{\tau_n}\in[1/a_n,a_n]} \le a_n b_n < \infty. \end{aligned}$$

All the properties of the sequence  $(M^n)_{n\geq 1}$  then follow from the construction, namely from the fact that  $A_n$  are a.s. increasing with  $\cup_n A_n = \mathbb{R}_+ \times X \setminus \{0\}$  a.s., and the fact that  $v^M$  is non-atomic in time since M is quasi-left continuous (see Section 2.8 and Subsection 7.5.4).

In the next theorem we show that the martingales obtained in Proposition 9.7.10 approximate M in the strong  $L^p$ -sense.

**Theorem 9.7.11.** Let X be a UMD Banach space, M be an X-valued martingale,  $(M^n)_{n\geq 1}$  be a sequence of X-valued martingales constructed in Proposition 9.7.10. Assume that for some fixed  $1 \leq p < \infty$ ,  $\mathbb{E}\sup_{t\geq 0} \|M_t\|^p < \infty$ . Then  $\mathbb{E}\sup_{t\geq 0} \|M_t^n\|^p < \infty$  for all  $n \geq 1$  and

$$\mathbb{E}\sup_{t\geq 0}\|M_t - M_t^n\|^p \to 0, \quad n \to \infty.$$

*Proof.* First of all notice that by Theorem 9.6.5, (9.7.13), and [80, Proposition 6.1.5] for any  $n \ge 1$ 

$$\mathbb{E}\sup_{t\geq 0} \|M_t^n\|^p \approx_{p,X} \mathbb{E}\left(\mathbb{E}_{\gamma} \left\|\sum_{t\geq 0} \gamma_s \Delta M_s^n\right\|^2\right)^{\frac{p}{2}} \leq \mathbb{E}\left(\mathbb{E}_{\gamma} \left\|\sum_{t\geq 0} \gamma_s \Delta M_s\right\|^2\right)^{\frac{p}{2}} \approx_{p,X} \mathbb{E}\sup_{t\geq 0} \|M_t\|^p.$$
Let us show the second part of the theorem. Note that by (9.7.13) a.s. for all  $x^* \in X^*$ 

$$[[M-M^n]]_{\infty}(x^*,x^*) = \sum_{t\geq 0} \langle \Delta M_t, x^* \rangle^2 \mathbf{1}_{\Delta M_t \neq \Delta M_t^n},$$

which monotonically vanishes as  $n \to \infty$  by (9.7.12) and (9.7.14). Consequently, the desired follows form Theorem 9.5.1, Lemma 9.3.11, and the monotone convenience theorem.

#### PURELY DISCONTINUOUS MARTINGALES WITH ACCESSIBLE JUMPS

Now let us turn to purely discontinuous martingales with accessible jumps.

Let *X* be a Banach space  $1 , <math>M : \mathbb{R}_+ \times \Omega \to X$  be a purely discontinuous  $L^p$ bounded martingale with accessible jumps,  $(\tau_n)_{n\geq 0}$  be a set of predictable stopping times with disjoint graphs such that (2.4.4) holds. Thanks to Lemma 2.4.5 for each  $n \geq 1$  we can define a martingale

$$M_t^n = \sum_{i=1}^n \Delta M_{\tau_i} \mathbf{1}_{[0,t]}(\tau_i), \quad t \ge 0.$$
(9.7.15)

*Does*  $(M^n)_{n\geq 1}$  *converge to M in strong*  $L^p$ *-sense?* The following theorem answers this question in the UMD case.

**Theorem 9.7.12.** Let X be a UMD Banach space,  $M : \mathbb{R}_+ \times \Omega \to X$  be a martingale with accessible jumps,  $(M^n)_{n\geq 1}$  be as in (9.7.15). Assume that  $\mathbb{E}\sup_{t\geq 0} \|M_t\|^p < \infty$  for some fixed  $1 \leq p < \infty$ . Then  $\mathbb{E}\sup_{t\geq 0} \|M_t^n\|^p < \infty$  for all  $n \geq 1$  and

$$\mathbb{E}\sup_{t\geq 0}\|M_t-M_t^n\|^p\to 0, \ n\to\infty.$$

*Proof.* The proof is fully analogous to the proof of Theorem 9.7.11.

**PROOF OF THEOREM 9.7.9** 

Let us now prove Theorem 9.7.9. Since *X* is a UMD Banach space, *M* has the canonical decomposition, i.e. there exist an *X*-valued continuous local martingale  $M^c$ , an *X*-valued purely discontinuous quasi-left continuous local martingale  $M^q$ , and an *X*-valued purely discontinuous local martingale  $M^a$  with accessible jumps such that  $M_0^c = M_0^q = 0$  and  $M = M^c + M^q + M^a$  (see Chapter 4 and 5 for details). Moreover, by (9.7.17) and a triangle inequality

$$\mathbb{E}\sup_{t\geq 0} \left( \|M_t^c\|^p + \|M_t^q\|^p + \|M_t^a\|^p \right) \approx_{p,X} \mathbb{E}\sup_{t\geq 0} \|M_t\|^p,$$

so it is sufficient to show Theorem 9.7.9 for each of these three cases separately. By [79, Theorem 1.3.2 and 3.3.16] *M* converges a.s., so we can assume that there exists T > 0 such that  $M_t = M_T$  a.s. for all  $t \ge T$ .

*Case 1: M is continuous.* The theorem follows from the fact that every continuous martingale is locally bounded and the fact that  $M_t = M_T$  for all  $t \ge T$ .

*Case 2: M is purely discontinuous quasi-left continuous.* By Theorem 9.7.11 one can assume that *M* has uniformly bounded jumps. Then the theorem follows from the fact that any adapted càdlàg process with uniformly bounded jumps is local uniformly bounded and the fact that  $M_t = M_T$  for all  $t \ge T$ .

*Case 3: M is purely discontinuous with accessible jumps.* By Theorem 9.7.12 we can assume that there exist predictable stopping times  $(\tau_n)_{n=1}^N$  with disjoint graphs such that

$$M_t = \sum_{n=1}^{N} \Delta M_{\tau_n} \mathbf{1}_{[0,t]}(\tau_n), \ t \ge 0.$$

Fix  $\varepsilon > 0$ . Without loss of generality we may assume that the stopping times  $(\tau_n)_{n=1}^N$  are bounded a.s. Due to the proof of Theorem 7.5.5 we may additionally assume that  $(\tau_n)_{n=1}^N$  is a.s. increasing. Then by the proof of Theorem 7.5.5 (or [89, Lemma 26.18] in the real-valued case) the sequence  $(0, \Delta M_{\tau_1}, 0, \Delta M_{\tau_2}, ..., 0, \Delta M_{\tau_N})$  is a martingale difference sequence with respect to the filtration

$$\mathbb{G} := (\mathscr{F}_{\tau_1-}, \mathscr{F}_{\tau_1}, \mathscr{F}_{\tau_2-}, \mathscr{F}_{\tau_2}, \dots, \mathscr{F}_{\tau_N-}, \mathscr{F}_{\tau_N})$$

(see [89, Lemma 25.2] for the definition of  $\mathscr{F}_{\tau-}$ ). As any discrete  $L^p$ -bounded martingale difference sequence,  $(0, \Delta M_{\tau_1}, 0, \Delta M_{\tau_2}, ..., 0, \Delta M_{\tau_N})$  can be approximated in a strong  $L^p$ -sense by a uniformly bounded *X*-valued  $\mathbb{G}$ -martingale difference sequence  $(0, d_1^{\varepsilon}, 0, d_2^{\varepsilon}, ..., 0, d_N^{\varepsilon})$  such that

$$\mathbb{E}\sup_{n=1}^{N} \left\|\sum_{i=1}^{n} \Delta M_{\tau_{i}} - d_{i}^{\varepsilon}\right\|^{p} < \varepsilon.$$

The martingale difference sequence  $(0, d_1^{\varepsilon}, 0, d_2^{\varepsilon}, ..., 0, d_N^{\varepsilon})$  can be translated back to a martingale on  $\mathbb{R}_+$  in the same way as it was shown in the proof of Theorem 7.5.5, i.e. one can define a process  $N^{\varepsilon} : \mathbb{R}_+ \times \Omega \to X$  such that

$$N_t^{\varepsilon} := \sum_{n=1}^N d_n \mathbf{1}_{[0,t]}(\tau_n), \ t \ge 0,$$

which is a martingale by Lemma 2.4.5 (or see [89, Lemma 26.18] for the real valued version) with

$$\mathbb{E}\sup_{t\geq 0}\|M_t - N_t^{\varepsilon}\|^p = \mathbb{E}\sup_{t\geq 0}\left\|\sum_{0\leq s\leq t}\Delta M_s - \Delta N_s^{\varepsilon}\right\|^p = \mathbb{E}\sup_{n=1}^N\left\|\sum_{i=1}^n\Delta M_{\tau_i} - d_i\right\|^p < \varepsilon,$$

which terminates the proof.

*Remark* 9.7.13. Clearly Theorem 9.7.9 holds true if *X* has a Schauder basis. Therefore it remain open for whether Theorem 9.7.9 holds true for a general Banach space.

#### 9.7.6. The canonical decomposition

As it was shown in Chapter 4 and 5, the canonical decomposition of a UMD Banach space-valued martingale is unique, and by Section 4.3 together with (2.2.1) we have that for any 1 and for any <math>i = c, q, a

$$\mathbb{E}\sup_{t\geq 0} \|M_t^i\|^p \lesssim_{p,X} \mathbb{E}\sup_{t\geq 0} \|M_t\|^p.$$
(9.7.16)

Theorem 9.7.7 allows us to extend (9.7.16) to the case p = 1. Indeed, due to Subsection 2.4.3 we have that for any  $x^* \in X^*$  a.s.

$$[\langle M, x^* \rangle]_t = [\langle M^c, x^* \rangle]_t + [\langle M^q, x^* \rangle]_t + [\langle M^a, x^* \rangle]_t, \quad t \ge 0,$$

so by Theorem 9.7.7

$$\mathbb{E}\sup_{t\geq 0} \|M_t^i\|^p \lesssim_{p,X} \mathbb{E}\sup_{t\geq 0} \|M_t\|^p,$$
(9.7.17)

for all  $1 \le p < \infty$  and any i = c, q, a.

#### 9.7.7. Covariation bilinear forms for pairs of martingales

Let *X* be a UMD Banach space,  $M, N : \mathbb{R}_+ \times \Omega \to X$  be local martingales. Then for any fixed  $t \ge 0$  and any  $x^*, y^* \in X^*$  we have that by [89, Theorem 26.6(iii)] a.s.

$$[\langle M, x^* \rangle, \langle N, y^* \rangle]_t \le [[M]]_t (x^*, x^*) [[N]]_t (y^*, y^*).$$

Thus analogously the proof of Theorem 9.5.1 (by exploiting a subspace *Y* of *X*<sup>\*</sup> that is a linear span of a countable subset of *X*<sup>\*</sup>) there exists a bounded bilinear form-valued random variable  $[[M, N]]_t : \Omega \to X \otimes X$  such that  $[\langle M, x^* \rangle, \langle N, y^* \rangle]_t = [[M, N]]_t(x^*, y^*)$  for any  $x^*, y^* \in X^*$  a.s.

Now let *X* and *Y* be UMD Banach spaces (perhaps different),  $M : \mathbb{R}_+ \times \Omega \to X$ ,  $N : \mathbb{R}_+ \times \Omega \to Y$  be local martingales. Then we can show that for any  $t \ge 0$  there exists a bilinear form-valued process  $[[M, N]]_t : \Omega \to X \otimes Y$  such that  $[[M, N]]_t = [\langle M, x^* \rangle, \langle N, y^* \rangle]_t$  a.s. for any  $x^* \in X^*$  and  $y^* \in Y^*$ . Indeed, one can presume the Banach space to be  $X \times Y$  and extend both *M* and *N* to take values in this Banach space. Then by the first part of the present subsection there exists a bilinear form  $[[M, N]]_t$  acting on  $(X \times Y)^* \times (X \times Y)^*$  such that for any  $x^* \in X^*$  and  $y^* \in Y^*$  a.s.

$$[[M, N]]_t ((x^*, y^*), (x^*, y^*)) = [\langle M, (x^*, y^*) \rangle, \langle N, (x^*, y^*) \rangle]_t$$
  
= [\langle M, x^\* \rangle, \langle N, y^\* \rangle]\_t. (9.7.18)

It remains to restrict  $[[M, N]]_t$  back to  $X \otimes Y$  from  $(X \times Y) \otimes (X \times Y)$  which is possible by (9.7.18).

Interesting things happen given  $Y = \mathbb{R}$ . In this case  $[[M, N]]_t$  takes values in  $X \otimes \mathbb{R} \simeq X$ , so  $[[M, N]]_t$  is simply *X*-valued, and it is easy to see that

$$[[M,N]]_t = \mathbb{P} - \lim_{\text{mesh}\to 0} \sum_{i=1}^n (M(t_n) - M(t_{n-1}))(N(t_n) - N(t_{n-1})),$$
(9.7.19)

where the limit in probability is taken over partitions  $0 = t_0 < ... < t_n = t$ , and it is taken in a *weak* sense (i.e. (9.7.19) holds under action of any linear functional  $x^* \in X^*$ ). It remains open whether (9.7.19) holds in a strong sense.

### 9.8. UMD BANACH FUNCTION SPACES

The goal of the present section is to show that a weaker version of (8.4.1) holds for p = 1.

**Theorem 9.8.1.** Let X be a UMD Banach function space over a  $\sigma$ -finite measure space  $(S, \Sigma, \mu), M : \mathbb{R}_+ \times \Omega \to X$  be a local martingale. Then there exists a local martingale field  $N : \mathbb{R}_+ \times \Omega \times S \to \mathbb{R}$  such that  $N(\omega, t, \cdot) = M_t(\omega)$  for all  $t \ge 0$  for a.a.  $\omega \in \Omega$ , and for all  $1 \le p < \infty$ 

$$\mathbb{E}\sup_{t\geq 0} \|M_t\|^p \approx_{p,X} \mathbb{E}\|[N]_{\infty}^{1/2}\|^p,$$
(9.8.1)

Let us first show the discrete version of Theorem 9.8.1, which was shown in [164, Theorem 3] for the case  $p \in (1, \infty)$ .

**Proposition 9.8.2.** Let X be a UMD Banach function space over a measure space  $(S, \Sigma, \mu)$ ,  $(d_n)_{n\geq 1}$  be an X-valued martingale difference sequence. Then for all  $1 \leq p < \infty$ 

$$\mathbb{E}\sup_{N\geq 1} \left\| \sum_{n=1}^{N} d_{n} \right\|^{p} \approx_{p,X} \mathbb{E} \left\| \left( \sum_{n=1}^{\infty} |d_{n}|^{2} \right)^{\frac{1}{2}} \right\|^{p}.$$

*Proof.* The proof follows from Theorem 9.2.1 and the equivalence [80, (9.26)] between the  $\gamma$ -norm and the square function.

*Remark* 9.8.3. By Remark 9.2.2 and [80, (9.26)] one has that for any  $r \in (1, \infty)$  there exist positive  $C_{r,X}$  and  $c_{r,X}$  such that for any  $1 \le p \le r$ 

$$c_{r,X}\mathbb{E}\left\|\left(\sum_{n=1}^{\infty}|d_{n}|^{2}\right)^{\frac{1}{2}}\right\|^{p} \leq \mathbb{E}\sup_{N\geq 1}\left\|\sum_{n=1}^{N}d_{n}\right\|^{p} \leq C_{r,X}\mathbb{E}\left\|\left(\sum_{n=1}^{\infty}|d_{n}|^{2}\right)^{\frac{1}{2}}\right\|^{p}.$$

*Proof of Theorem* 9.8.1. We will consider separately the cases p > 1 and p = 1.

*Case* p > 1. This case was covered in Theorem 8.4.1. Nevertheless, we wish to notice that by modifying the proof from Theorem 8.4.1 by using Proposition 9.8.2 one can obtain better behavior of the equivalence constants in (9.8.1). Namely, by exploiting the same proof together with Proposition 9.8.2 and Remark 9.8.3 one obtains that for any  $p' \in (1,\infty)$  there exist positive  $C_{p',X}$  and  $c_{p',X}$  (the same as in Remark 9.8.3) such that for any 1

$$c_{p',X} \mathbb{E} \| [N]_{\infty}^{1/2} \|^{p} \le \mathbb{E} \sup_{t \ge 0} \| M_{t} \|^{p} \le C_{p',X} \mathbb{E} \| [N]_{\infty}^{1/2} \|^{p}.$$
(9.8.2)

*Case* p = 1. By Theorem 9.7.9 there exists a sequence  $(M^n)_{n\geq 1}$  of uniformly bounded *X*-valued martingales such that

$$\mathbb{E}\sup_{t\geq 0}\|M_t - M_t^n\| \to 0, \quad n \to \infty.$$
(9.8.3)

Since  $M^n$  is uniformly bounded for any  $n \ge 1$ ,  $\mathbb{E}\sup_{t\ge 0} \|M_t^n\|^2 < \infty$ , so by Case p > 1 there exists a local martingale field  $N^n$  such that  $N^n(\omega, t, \cdot) = M_t^n(\omega)$  for all  $t \ge 0$  for a.a.  $\omega \in \Omega$ . By (9.8.2) one has that there exist positive constants  $C_X$  and  $c_X$  such that for all  $m, n \ge 1$ 

$$c_X \mathbb{E} \| [N^n - N^m]_{\infty}^{1/2} \| \le \mathbb{E} \sup_{t \ge 0} \| M_t^n - M_t^m \| \le C_X \mathbb{E} \| [N^n - N^m]_{\infty}^{1/2} \|,$$

hence due to (9.8.3)  $(N^n)_{n\geq 1}$  is a Cauchy sequence in MQ<sup>1</sup>(*X*). Since by Proposition 8.4.2 the linear space MQ<sup>1</sup>(*X*) endowed with the norm (8.4.2) is Banach, there exists a limit *N* of  $(N^n)_{n\geq 1}$  in MQ<sup>1</sup>(*X*).

Let us show that *N* is the desired local martingale field. Fix  $t \ge 0$ . We need to who that  $N(\cdot, t, \cdot) = M_t$  a.s. on  $\Omega$ . First notice that by the last part of Proposition 8.4.2 there exists a subsequence of  $(N^n)_{n\ge 1}$  which we will denote by  $(N^n)_{n\ge 1}$  as well such that  $N^n(\cdot, t, \sigma) \rightarrow N(\cdot, t, \sigma)$  in  $L^1(\Omega)$  for a.e.  $\sigma \in S$ . On the other hand by Jensen's inequality

$$\left\|\mathbb{E}|N^{n}(\cdot,t,\cdot)-M_{t}|\right\| = \left\|\mathbb{E}|M_{t}^{n}-M_{t}|\right\| \leq \mathbb{E}\|M_{t}^{n}-M_{t}\| \to 0, \quad n \to \infty.$$

Hence  $N^n(\cdot, t, \cdot) \to M_t$  in  $X(L^1(\Omega))$ , and thus by Remark 8.2.2 in  $L^0(S; L^1(\Omega))$ . Therefore we can find a subsequence of  $(N^n)_{n\geq 1}$  (which we will again denote by  $(N^n)_{n\geq 1}$ ) such that  $N^n(\cdot, t, \sigma) \to M_t(\sigma)$  in  $L^1(\Omega)$  for a.e.  $\sigma \in S$  (here we use that fact that  $\mu$  is  $\sigma$ -finite), so  $N(\cdot, t, \cdot) = M_t$  a.s. on  $\Omega \times S$ , and consequently by Definition 8.2.1(*iii*),  $N(\omega, t, \cdot) = M_t(\omega)$  for a.a.  $\omega \in \Omega$ .

Let us finally show (9.8.1). Since  $N^n \rightarrow N$  in MQ<sup>1</sup>(*X*) and by (9.8.3)

$$\mathbb{E}\|[N]_{\infty}^{1/2}\| = \lim_{n \to \infty} \mathbb{E}\|[N^n]_{\infty}^{1/2}\| = \sum_{x} \lim_{n \to \infty} \mathbb{E}\sup_{t \ge 0} \|M_t^n\| = \mathbb{E}\sup_{t \ge 0} \|M_t\|,$$

which terminates the proof.

*Remark* 9.8.4. It was shown in Theorem 8.4.1 that in the case p > 1 the equivalence (9.8.1) can be strengthen. Namely, in this case one can show that

$$\mathbb{E} \| \sup_{t \ge 0} |M_t| \|^p \approx_{p, X} \mathbb{E} \| [N]_{\infty}^{1/2} \|^p,$$
(9.8.4)

i.e. one has the same equivalence with a pointwise supremum in *S*. The techniques that provide such an improvement were discovered by Rubio de Francia in [164]. Unfortunately, it remains open whether (9.8.4) holds for p = 1. Surprisingly, (9.8.4) holds for p = 1 and for  $X = L^1(S)$  by a simple Fubini-type argument, so it might be that (9.8.4) holds for p = 1 even for other nonreflexive Banach spaces.

## INDEX

 $F \star \bar{\mu}, 34$ *M*<sup>\*</sup>, 19 *M<sup>a</sup>*, 28 M<sup>c</sup>, 21, 28  $M^{d}, 21$  $M^{q}, 28$ M<sup>τ-</sup>, 25  $N \ll M, 5, 53$  $N \stackrel{w}{\ll} M, 6, 53$ X + Y, 176  $X \cap Y$ , 176 [*M*, *N*], 20, 21 [*M*], 20  $[M]^{a}, 28$  $[M]^{c}, 21, 28$  $[M]^d, 21$  $[M]^{q}, 28$  $\Delta M_{\tau}$ , 24  $\Delta V_{\tau}$ , 96  $\gamma(\cdot), 249$  $\hbar_{p,X}$ , 131  $\mathbb{R}_{+}, 17$ T, 126  $\mathcal{F}_{\tau-}, 24$  $\mathcal{F}_{\tau}, 24$  $\mathscr{H}_{X}^{\mathbb{T}}$ , 126  $\mathscr{H}_X^{\mathbb{R}}$ , 143  $\mathcal{H}_X^{\text{dis}}$ , 143  $\mathcal{M}_{x}^{\hat{p}}, 19$  $\mathcal{M}^p_{\mathbf{v}}(\mathbb{F}), 19$  $\mathcal{M}_{Y}^{\widehat{1},\infty},96$  $\mathcal{M}_{v}^{\hat{p},q}$ , 29  $\mathcal{M}_X^{\hat{p},c}, 21$  $\mathcal{M}^{\hat{p},c}_{x}(\mathbb{F}), 21$ 

 $\mathcal{M}_X^{p,d}$ , 22  $\mathcal{M}_{X}^{p,d}(\mathbb{F}), 22$  $\mu^{M}$ , 98, 197  $\overline{\mathbb{R}}_+, 17$ Ω̃, 33  $\tilde{\mathcal{O}}$ , 33 *9*,33  $a \equiv_A b$ , 17  $a \lor b, 17$  $a \wedge b$ , 17  $a \geq_{A} b, 17$  $a \lesssim_A b$ , 17  $df_n$ , 20  $g \ll f, 45$  $g \stackrel{w}{\ll} f, 45$  $h \otimes x$ , 32 p', 17p\*, 4, 17, 23  $q_{M}, 20$ Beurling-Ahlfors transform, 60 Brownian representation, 33 canonical decomposition of martingales, 7, 28, 30, 83, 101, 241, 274 compensator, 34, 97 corresponding dual basis, 36 covariation, 20, 21 covariation bilinear form, 257 decomposition canonical, 7, 28, 30, 83, 101, 241, 274 Gundy's, 97 Meyer-Yoeurp, 21, 23, 73, 108, 130 Yoeurp, 8, 28, 30, 81, 110 decoupling constant, 151

differential subordination, 4, 5, 45, 53 double Riesz transform, 61 filtration usual conditions, 18 Fourier multiplier, 56 Fourier transform, 56 function biconcave, 35 Burkholder, 48 concave, 35 convex, 35 plurisubharmonic, 128 step, 143 subharmonic, 128 zigzag-concave, 35  $\gamma$ -radonifying operator, 35 Gaussian characteristic, 249 Gundy's decomposition, 97 Hilbert transform, 61, 126, 143 induced filtration, 31 inequality Burkholder–Davis–Gundy, 21, 234, 243,275 Burkholder-Rosenthal, 178 decoupling, 151 Doob's, 19 lattice Doob's, 231 Novikov's, 199, 200 Itô's formula, 37 Lebesgue measure, 49, 72 martingale, 18  $L^p$ -bounded, 19 approximation, 270 continuous, 21, 261 domination, 269 local, 18 orthogonal, 127

Paley-Walsh, 19 purely discontinuous, 21, 22, 261 quasi-left continuous, 27  $\tau$ -continuous, 31 with accessible jumps, 27 with independent increments, 263 measure Gaussian, 249 Lebesgue, 49, 72 random, 33, 199 Meyer-Yoeurp decomposition, 21, 73, 108, 130 operator Beurling-Ahlfors transform, 60 double Riesz transform, 61 Fourier multiplier, 56 Fourier transform, 56  $\gamma$ -radonifying, 35 Hilbert, 166 Hilbert transform, 61, 126, 143 Riesz transform, 166 singular integral, 165 orthogonal martingales, 127 process adapted, 18 càdlàg, càglàd, càd, càg, làd, làg, 17 elementary predictable, 32, 264 optional, 33 predictable, 18 pure jump, 21 quasi-left continuous, 27 Wiener, 33 Wiener multidimensional, 33 with accessible jumps, 27 quadratic variation, 5, 20 quadratic variation derivative, 20 Rademacher distribution, 19 random measure, 33, 199 compensator, 34

integer-valued, 33 non-atomic in time, 33 optional, 33 predictable, 33  $\sigma$ -algebra optional, 18 predictable, 18 Skorohod space, 18 space Banach function, 275 Skorohod, 18 UMD, 23 step function, 143 stochastic integral, 32, 189, 209, 264 w.r.t. random measure, 33, 199, 268 stopping time, 24 graph of, 27 optional, 24 predictable, 25 totally inaccessible, 25 time-change, 31 totally inaccessible, 25 ucp topology, 96 UMD Banach space, 23 usual conditions, 18 variation, 97 weak differential subordination, 4, 6, 45, 53, 84, 102, 157 of harmonic functions, 161 transform, 102 Wiener process multidimensional, 33 Yoeurp decomposition, 8, 28, 30, 81, 110

## REFERENCES

- S. Albeverio and B. Rüdiger. Stochastic integrals and the Lévy-Ito decomposition theorem on separable Banach spaces. *Stoch. Anal. Appl.*, 23(2):217–253, 2005.
- [2] F. Albiac and N. Kalton. *Topics in Banach space theory*. Springer, New York, 2006.
- [3] M. Antoni. *Stochastic Integration in L<sup>p</sup> Spaces*. Diploma Thesis, Karlsruhe Institute of Technology, 2012.
- [4] M. Antoni. *Regular Random Field Solutions for Stochastic Evolution Equations*. PhD thesis, Karlsruhe Institute of Technology, 2017.
- [5] D. Applebaum. Lévy processes and stochastic integrals in Banach spaces. *Probab. Math. Statist.*, 27(1):75–88, 2007.
- [6] L.M. Arutyunyan and I.S. Yaroslavtsev. Measurable polynomials on infinitedimensional spaces. *Dokl. Akad. Nauk*, 449(6):627–631, 2013.
- [7] R. Bañuelos and A. Lindeman, II. A martingale study of the Beurling-Ahlfors transform in R<sup>n</sup>. J. Funct. Anal., 145(1):224–265, 1997.
- [8] R. Bañuelos. The foundational inequalities of D. L. Burkholder and some of their ramifications. *Illinois J. Math.*, 54(3):789–868, 2010.
- [9] R. Bañuelos, A. Bielaszewski, and K. Bogdan. Fourier multipliers for nonsymmetric Lévy processes. In *Marcinkiewicz centenary volume*, volume 95 of *Banach Center Publ.*, pages 9–25. Polish Acad. Sci. Inst. Math., Warsaw, 2011.
- [10] R. Bañuelos and K. Bogdan. Lévy processes and Fourier multipliers. J. Funct. Anal., 250(1):197–213, 2007.
- [11] R. Bañuelos and P. Janakiraman. L<sup>p</sup>-bounds for the Beurling-Ahlfors transform. *Trans. Amer. Math. Soc.*, 360(7):3603–3612, 2008.
- [12] R. Bañuelos and M. Kwaśnicki. On the *l<sup>p</sup>*-norm of the Discrete Hilbert transform. *arXiv*:1709.07427, 2017.
- [13] R. Bañuelos and A. Osekowski. Martingales and sharp bounds for Fourier multipliers. Ann. Acad. Sci. Fenn. Math., 37(1):251–263, 2012.

- [14] R. Bañuelos and A. Osekowski. Sharp inequalities for the Beurling-Ahlfors transform on radial functions. *Duke Math. J.*, 162(2):417–434, 2013.
- [15] R. Bañuelos and G. Wang. Sharp inequalities for martingales with applications to the Beurling-Ahlfors and Riesz transforms. *Duke Math. J.*, 80(3):575– 600, 1995.
- [16] R. Bañuelos and G. Wang. Orthogonal martingales under differential subordination and applications to Riesz transforms. *Illinois J. Math.*, 40(4):678–691, 1996.
- [17] F. Baumgartner. Lévy processes with values in locally convex suslin spaces. arXiv:1510.00538, 2015.
- [18] Y.I. Belopol'skaya and Y.L. Dalecky. Stochastic equations and differential geometry, volume 30 of Mathematics and its Applications (Soviet Series). Kluwer Academic Publishers Group, Dordrecht, 1990.
- [19] P. Billingsley. Probability and measure. Wiley Series in Probability and Mathematical Statistics. John Wiley & Sons, Inc., New York, third edition, 1995. A Wiley-Interscience Publication.
- [20] S. Bochner and A.E. Taylor. Linear functionals on certain spaces of abstractly-valued functions. Ann. of Math. (2), 39(4):913–944, 1938.
- [21] V.I. Bogachev. Gaussian measures, volume 62 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 1998.
- [22] V.I. Bogachev. Measure theory. Vol. I, II. Springer-Verlag, Berlin, 2007.
- [23] J. Bourgain. Some remarks on Banach spaces in which martingale difference sequences are unconditional. *Ark. Mat.*, 21(2):163–168, 1983.
- [24] J. Bourgain. Extension of a result of Benedek, Calderón and Panzone. Ark. Mat., 22(1):91–95, 1984.
- [25] Z. Brzeźniak. Stochastic partial differential equations in M-type 2 Banach spaces. *Potential Anal.*, 4(1):1–45, 1995.
- [26] Z. Brzeźniak and E. Hausenblas. Maximal regularity for stochastic convolutions driven by Lévy processes. *Probab. Theory Related Fields*, 145(3-4):615– 637, 2009.
- [27] Z. Brzeźniak and J.M.A.M. van Neerven. Stochastic convolution in separable Banach spaces and the stochastic linear Cauchy problem. *Studia Math.*, 143(1):43–74, 2000.

- [28] D.L. Burkholder. Martingale transforms. Ann. Math. Statist., 37:1494–1504, 1966.
- [29] D.L. Burkholder. Distribution function inequalities for martingales. Ann. Probability, 1:19–42, 1973.
- [30] D.L. Burkholder. A geometrical characterization of Banach spaces in which martingale difference sequences are unconditional. *Ann. Probab.*, 9(6):997– 1011, 1981.
- [31] D.L. Burkholder. Martingale transforms and the geometry of Banach spaces. In *Probability in Banach spaces*, III (Medford, Mass., 1980), volume 860 of Lecture Notes in Math., pages 35–50. Springer, Berlin-New York, 1981.
- [32] D.L. Burkholder. A geometric condition that implies the existence of certain singular integrals of Banach-space-valued functions. In *Conference on harmonic analysis in honor of Antoni Zygmund, Vol. I, II (Chicago, Ill., 1981),* Wadsworth Math. Ser., pages 270–286. Wadsworth, Belmont, CA, 1983.
- [33] D.L. Burkholder. Boundary value problems and sharp inequalities for martingale transforms. Ann. Probab., 12(3):647–702, 1984.
- [34] D.L. Burkholder. An elementary proof of an inequality of R.E.A.C. Paley. Bull. London Math. Soc., 17(5):474–478, 1985.
- [35] D.L. Burkholder. Martingales and Fourier analysis in Banach spaces. In Probability and analysis (Varenna, 1985), volume 1206 of Lecture Notes in Math., pages 61–108. Springer, Berlin, 1986.
- [36] D.L. Burkholder. Sharp inequalities for martingales and stochastic integrals. *Astérisque*, (157-158):75–94, 1988. Colloque Paul Lévy sur les Processus Stochastiques (Palaiseau, 1987).
- [37] D.L. Burkholder. Differential subordination of harmonic functions and martingales. In *Harmonic analysis and partial differential equations (El Escorial,* 1987), volume 1384 of *Lecture Notes in Math.*, pages 1–23. Springer, Berlin, 1989.
- [38] D.L. Burkholder. Explorations in martingale theory and its applications. In École d'Été de Probabilités de Saint-Flour XIX—1989, volume 1464 of Lecture Notes in Math., pages 1–66. Springer, Berlin, 1991.
- [39] D.L. Burkholder. Martingales and singular integrals in Banach spaces. In *Handbook of the geometry of Banach spaces, Vol. I*, pages 233–269. North-Holland, Amsterdam, 2001.

- [40] D.L. Burkholder, B.J. Davis, and R.F. Gundy. Integral inequalities for convex functions of operators on martingales. In *Proceedings of the Sixth Berkeley Symposium on Mathematical Statistics and Probability (Univ. California, Berkeley, Calif., 1970/1971), Vol. II: Probability theory*, pages 223–240. Univ. California Press, Berkeley, Calif., 1972.
- [41] D.L. Burkholder and R.F. Gundy. Extrapolation and interpolation of quasilinear operators on martingales. *Acta Math.*, 124:249–304, 1970.
- [42] K.P. Choi. A sharp inequality for martingale transforms and the unconditional basis constant of a monotone basis in L<sup>p</sup>(0,1). Trans. Amer. Math. Soc., 330(2):509–529, 1992.
- [43] F. Cobos. Some spaces in which martingale difference sequences are unconditional. Bull. Polish Acad. Sci. Math., 34(11-12):695–703 (1987), 1986.
- [44] S.G. Cox and S. Geiss. On decoupling in Banach spaces. arXiv:1805.12377, 2018.
- [45] S.G. Cox and M.C. Veraar. Vector-valued decoupling and the Burkholder-Davis-Gundy inequality. *Illinois J. Math.*, 55(1):343–375 (2012), 2011.
- [46] I. Csörgő. On a type of mixed norm spaces. Acta Math. Hungar., 78(1-2):79– 91, 1998.
- [47] G. Da Prato and J. Zabczyk. Stochastic equations in infinite dimensions. Cambridge University Press, Cambridge, 1992.
- [48] G. Da Prato and J. Zabczyk. Stochastic equations in infinite dimensions, volume 152 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, second edition, 2014.
- [49] C. Dellacherie and P.-A. Meyer. Probabilities and potential. B, volume 72 of North-Holland Mathematics Studies. North-Holland Publishing Co., Amsterdam, 1982. Theory of martingales, Translated from the French by J. P. Wilson.
- [50] E. Dettweiler. Banach space valued processes with independent increments and stochastic integration. In *Probability in Banach spaces, IV (Oberwolfach,* 1982), volume 990 of *Lecture Notes in Math.*, pages 54–83. Springer, Berlin-New York, 1983.
- [51] S. Dirksen. Itô isomorphisms for L<sup>p</sup>-valued Poisson stochastic integrals. Ann. Probab., 42(6):2595–2643, 2014.
- [52] S. Dirksen, J. Maas, and J.M.A.M. van Neerven. Poisson stochastic integration in Banach spaces. *Electron. J. Probab.*, 18:No. 100, 28, 2013.

- [53] S. Dirksen, C. Marinelli, and I.S. Yaroslavtsev. Stochastic evolution equations in *L<sup>p</sup>*-spaces driven by jump noise. *In preparation*.
- [54] S. Dirksen and I.S. Yaroslavtsev. L<sup>q</sup>-valued Burkholder-Rosenthal inequalities and sharp estimates for stochastic integrals. arXiv:1707.00109, 2017.
- [55] C. Doléans. Variation quadratique des martingales continues à droite. Ann. Math. Statist, 40:284–289, 1969.
- [56] D.E. Edmunds and M. Krbec. Variations on Yano's extrapolation theorem. *Rev. Mat. Complut.*, 18(1):111–118, 2005.
- [57] L.C. Evans and R.F. Gariepy. *Measure theory and fine properties of functions*. Studies in Advanced Mathematics. CRC Press, Boca Raton, FL, 1992.
- [58] D. Filipović, S. Tappe, and J. Teichmann. Jump-diffusions in Hilbert spaces: existence, stability and numerics. *Stochastics*, 82(5):475–520, 2010.
- [59] G.B. Folland. A course in abstract harmonic analysis. Studies in Advanced Mathematics. CRC Press, Boca Raton, FL, 1995.
- [60] J. García-Cuerva, R. Macías, and J. L. Torrea. The Hardy-Littlewood property of Banach lattices. *Israel J. Math.*, 83(1-2):177–201, 1993.
- [61] D.J.H. Garling. Brownian motion and UMD-spaces. In *Probability and Banach spaces (Zaragoza, 1985)*, volume 1221 of *Lecture Notes in Math.*, pages 36–49. Springer, Berlin, 1986.
- [62] D.J.H. Garling. Random martingale transform inequalities. In *Probability in Banach spaces 6 (Sandbjerg, 1986)*, volume 20 of *Progr. Probab.*, pages 101–119. Birkhäuser Boston, Boston, MA, 1990.
- [63] J.B. Garnett and D.E. Marshall. *Harmonic measure*, volume 2 of *New Mathe-matical Monographs*. Cambridge University Press, Cambridge, 2008. Reprint of the 2005 original.
- [64] P.M. Gauthier. *Lectures on several complex variables*. Birkhäuser/Springer, 2014.
- [65] S. Geiss. A counterexample concerning the relation between decoupling constants and UMD-constants. *Trans. Amer. Math. Soc.*, 351(4):1355–1375, 1999.
- [66] S. Geiss, S. Montgomery-Smith, and E. Saksman. On singular integral and martingale transforms. *Trans. Amer. Math. Soc.*, 362(2):553–575, 2010.
- [67] M. Girardi and L.W. Weis. Operator-valued martingale transforms and Rboundedness. *Illinois J. Math.*, 49(2):487–516, 2005.

- [68] B. Goldys and J.M.A.M. van Neerven. Transition semigroups of Banach space-valued Ornstein-Uhlenbeck processes. *Acta Appl. Math.*, 76(3):283– 330, 2003.
- [69] L. Grafakos. Classical Fourier analysis, volume 249 of Graduate Texts in Mathematics. Springer, New York, third edition, 2014.
- [70] B. Grigelionis. The representation of integer-valued random measures as stochastic integrals over the Poisson measure. *Litovsk. Mat. Sb.*, 11:93–108, 1971.
- [71] R.F. Gundy. A decomposition for L<sup>1</sup>-bounded martingales. *Ann. Math. Statist.*, 39:134–138, 1968.
- [72] I. Gyöngy. On stochastic equations with respect to semimartingales. III. Stochastics, 7(4):231–254, 1982.
- [73] I. Gyöngy and N.V. Krylov. On stochastic equations with respect to semimartingales. I. *Stochastics*, 4(1):1–21, 1980/81.
- [74] P. Hitczenko. On tangent sequences of UMD-space valued random vectors. Unpublished manuscript.
- [75] P. Hitczenko. Best constants in martingale version of Rosenthal's inequality. Ann. Probab., 18(4):1656–1668, 1990.
- [76] J. Hoffmann-Jørgensen and G. Pisier. The law of large numbers and the central limit theorem in Banach spaces. *Ann. Probability*, 4(4):587–599, 1976.
- [77] B. Hollenbeck, N.J. Kalton, and I.E. Verbitsky. Best constants for some operators associated with the Fourier and Hilbert transforms. *Studia Math.*, 157(3):237–278, 2003.
- [78] L. Hörmander. *Notions of convexity*, volume 127 of *Progress in Mathematics*. Birkhäuser Boston, Inc., Boston, MA, 1994.
- [79] T.P. Hytönen, J.M.A.M. van Neerven, M.C. Veraar, and L.W. Weis. Analysis in Banach spaces. Vol. I. Martingales and Littlewood-Paley theory, volume 63 of Ergebnisse der Mathematik und ihrer Grenzgebiete. Springer, 2016.
- [80] T.P. Hytönen, J.M.A.M. van Neerven, M.C. Veraar, and L.W. Weis. Analysis in Banach spaces. Vol. II. Probabilistic methods and operator theory, volume 67 of Ergebnisse der Mathematik und ihrer Grenzgebiete. Springer, 2017.
- [81] K. Itô. On stochastic processes. I. (Infinitely divisible laws of probability). *Jap. J. Math.*, 18:261–301, 1942.

- [82] T. Iwaniec. Extremal inequalities in Sobolev spaces and quasiconformal mappings. Z. Anal. Anwendungen, 1(6):1–16, 1982.
- [83] T. Iwaniec and G. Martin. Riesz transforms and related singular integrals. J. Reine Angew. Math., 473:25–57, 1996.
- [84] J. Jacod. Calcul stochastique et problèmes de martingales, volume 714 of Lecture Notes in Mathematics. Springer, Berlin, 1979.
- [85] J. Jacod and A.N. Shiryaev. *Limit theorems for stochastic processes*, volume 288 of *Grundlehren der Mathematischen Wissenschaften*. Springer-Verlag, Berlin, second edition, 2003.
- [86] M. Jouak and L. Thibault. Directional derivatives and almost everywhere differentiability of biconvex and concave-convex operators. *Math. Scand.*, 57(1):215–224, 1985.
- [87] M. Junge. Doob's inequality for non-commutative martingales. J. Reine Angew. Math., 549:149–190, 2002.
- [88] M. Junge and Q. Xu. Noncommutative Burkholder/Rosenthal inequalities. Ann. Probab., 31(2):948–995, 2003.
- [89] O. Kallenberg. Foundations of modern probability. Probability and its Applications (New York). Springer-Verlag, New York, second edition, 2002.
- [90] O. Kallenberg and R. Sztencel. Some dimension-free features of vectorvalued martingales. *Probab. Theory Related Fields*, 88(2):215–247, 1991.
- [91] N. Kalton, E. Lorist, and L. Weis. Euclidean structures. In preparation.
- [92] D. Kannan and V. Lakshmikantham, editors. Handbook of stochastic analysis and applications, volume 163 of Statistics: Textbooks and Monographs. Marcel Dekker, Inc., New York, 2002.
- [93] I. Karatzas and S.E. Shreve. Brownian motion and stochastic calculus, volume 113 of Graduate Texts in Mathematics. Springer-Verlag, New York, second edition, 1991.
- [94] N. Kazamaki. Changes of time, stochastic integrals, and weak martingales. *Z. Wahrscheinlichkeitstheorie und Verw. Gebiete*, 22:25–32, 1972.
- [95] J.F.C. Kingman. Poisson processes, volume 3 of Oxford Studies in Probability. The Clarendon Press, Oxford University Press, New York, 1993. Oxford Science Publications.

- [96] A.N. Kolmogorov and S.V. Fomin. *Elements of the theory of functions and functional analysis. Vol. 1. Metric and normed spaces.* Graylock Press, Rochester, N. Y., 1957. Translated from the first Russian edition by Leo F. Boron.
- [97] M. Kovács and E. Hausenblas. Global solutions to the stochastic Volterra equation driven by Lévy noise. arXiv:1612.09457, 2016.
- [98] S. Kreĭn, Y. Petunīn, and E. Semënov. *Interpolation of linear operators*. American Mathematical Society, Providence, R.I., 1982.
- [99] V.M. Kruglov. On natural and predictable processes. *Sankhya A*, 78(1):43–51, 2016.
- [100] H.H. Kuo. Gaussian measures in Banach spaces. Lecture Notes in Mathematics, Vol. 463. Springer-Verlag, Berlin-New York, 1975.
- [101] S. Kwapień. Isomorphic characterizations of inner product spaces by orthogonal series with vector valued coefficients. *Studia Math.*, 44:583–595, 1972. Collection of articles honoring the completion by Antoni Zygmund of 50 years of scientific activity, VI.
- [102] S. Kwapień and W. Woyczyński. *Random series and stochastic integrals: single and multiple*. Probability and its Applications. Birkhäuser Boston Inc., Boston, MA, 1992.
- [103] E. Laeng. Remarks on the Hilbert transform and on some families of multiplier operators related to it. *Collect. Math.*, 58(1):25–44, 2007.
- [104] M. Ledoux and M. Talagrand. Probability in Banach spaces, volume 23 of Ergebnisse der Mathematik und ihrer Grenzgebiete (3). Springer-Verlag, Berlin, 1991. Isoperimetry and processes.
- [105] P. Lelong and L. Gruman. Entire functions of several complex variables, volume 282 of Grundlehren der Mathematischen Wissenschaften. Springer-Verlag, Berlin, 1986.
- [106] E. Lenglart. Relation de domination entre deux processus. Ann. Inst. H. Poincaré Sect. B (N.S.), 13(2):171–179, 1977.
- [107] N. Lindemulder, M.C. Veraar, and I.S. Yaroslavtsev. The UMD property for Musielak–Orlicz spaces. In *Positivity and Noncommutative Analysis. Festschrift in honour of Ben de Pagter on the occasion of his 65th birthday*, Trends in Mathematics. Birkhäuser, 2019. arXiv:1810.13362, to appear.
- [108] J. Lindenstrauss and L. Tzafriri. Classical Banach spaces. II, volume 97 of Ergebnisse der Mathematik und ihrer Grenzgebiete. Springer-Verlag, Berlin-New York, 1979. Function spaces.

- [109] Q. Lü, J. Yong, and X. Zhang. Representation of Itô integrals by Lebesgue/Bochner integrals. J. Eur. Math. Soc. (JEMS), 14(6):1795–1823, 2012.
- [110] C. Marinelli. On maximal inequalities for purely discontinuous L<sub>q</sub>-valued martingales. arXiv:1311.7120, 2013.
- [111] C. Marinelli, L. Di Persio, and G. Ziglio. Approximation and convergence of solutions to semilinear stochastic evolution equations with jumps. *J. Funct. Anal.*, 264(12):2784–2816, 2013.
- [112] C. Marinelli, C. Prévôt, and M. Röckner. Regular dependence on initial data for stochastic evolution equations with multiplicative Poisson noise. J. Funct. Anal., 258(2):616–649, 2010.
- [113] C. Marinelli and M. Röckner. Well-posedness and asymptotic behavior for stochastic reaction-diffusion equations with multiplicative Poisson noise. *Electron. J. Probab.*, 15:no. 49, 1528–1555, 2010.
- [114] C. Marinelli and M. Röckner. On maximal inequalities for purely discontinuous martingales in infinite dimensions. In *Séminaire de Probabilités XLVI*, volume 2123 of *Lecture Notes in Math.*, pages 293–315. Springer, 2014.
- [115] C. Marinelli and M. Röckner. On the maximal inequalities of Burkholder, Davis and Gundy. *Expo. Math.*, 34(1):1–26, 2016.
- [116] T. Martínez and J.L. Torrea. Operator-valued martingale transforms. *Tohoku Math. J.* (2), 52(3):449–474, 2000.
- [117] B. Maurey and G. Pisier. Séries de variables aléatoires vectorielles indépendantes et propriétés géométriques des espaces de Banach. *Studia Math.*, 58(1):45–90, 1976.
- [118] T.R. McConnell. On Fourier multiplier transformations of Banach-valued functions. *Trans. Amer. Math. Soc.*, 285(2):739–757, 1984.
- [119] T.R. McConnell. Decoupling and stochastic integration in UMD Banach spaces. *Probab. Math. Statist.*, 10(2):283–295, 1989.
- [120] M. Métivier. Semimartingales, volume 2 of de Gruyter Studies in Mathematics. Walter de Gruyter & Co., Berlin-New York, 1982. A course on stochastic processes.
- [121] M. Métivier and J. Pellaumail. Stochastic integration. Academic Press [Harcourt Brace Jovanovich, Publishers], New York-London-Toronto, Ont., 1980. Probability and Mathematical Statistics.
- [122] P.-A. Meyer. Un cours sur les intégrales stochastiques. Springer, Berlin, 1976.

- [123] P.A. Meyer. Notes sur les intégrales stochastiques. I. Intégrales hilbertiennes. pages 446–462. Lecture Notes in Math., Vol. 581, 1977.
- [124] P. Meyer-Nieberg. Banach lattices. Universitext. Springer-Verlag, Berlin, 1991.
- [125] J.M.A.M. van Neerven. γ-radonifying operators—a survey. In *The AMSI-ANU Workshop on Spectral Theory and Harmonic Analysis*, volume 44 of *Proc. Centre Math. Appl. Austral. Nat. Univ.*, pages 1–61. Austral. Nat. Univ., Canberra, 2010.
- [126] J.M.A.M. van Neerven, M.C. Veraar, and L.W. Weis. Stochastic integration in UMD Banach spaces. Ann. Probab., 35(4):1438–1478, 2007.
- [127] J.M.A.M. van Neerven, M.C. Veraar, and L.W. Weis. Maximal L<sup>p</sup>-regularity for stochastic evolution equations. *SIAM J. Math. Anal.*, 44(3):1372–1414, 2012.
- [128] J.M.A.M. van Neerven, M.C. Veraar, and L.W. Weis. Stochastic maximal L<sup>p</sup>regularity. Ann. Probab., 40(2):788–812, 2012.
- [129] J.M.A.M. van Neerven and L.W. Weis. Stochastic integration of functions with values in a Banach space. *Studia Math.*, 166(2):131–170, 2005.
- [130] A.L. Neidhardt. Stochastic Integrals in 2-Uniformly Smooth Banach Spaces. PhD thesis, University of Wisconsin, 1978.
- [131] A.A. Novikov. Discontinuous martingales. Teor. Verojatnost. i Primemen., 20:13–28, 1975.
- [132] M. Ondreját. Uniqueness for stochastic evolution equations in Banach spaces. Dissertationes Math. (Rozprawy Mat.), 426:63, 2004.
- [133] A. Osękowski. Inequalities for dominated martingales. *Bernoulli*, 13(1):54– 79, 2007.
- [134] A. Osękowski. Sharp LlogL inequalities for differentially subordinated martingales and harmonic functions. *Illinois J. Math.*, 52(3):745–756, 2008.
- [135] A. Osękowski. Sharp maximal inequality for stochastic integrals. Proc. Amer. Math. Soc., 136(8):2951–2958, 2008.
- [136] A. Osękowski. Strong differential subordination and sharp inequalities for orthogonal processes. J. Theoret. Probab., 22(4):837–855, 2009.
- [137] A. Osękowski. On relaxing the assumption of differential subordination in some martingale inequalities. *Electron. Commun. Probab.*, 16:9–21, 2011.

- [138] A. Osękowski. Sharp maximal inequalities for the moments of martingales and non-negative submartingales. *Bernoulli*, 17(4):1327–1343, 2011.
- [139] A. Osękowski. Sharp inequalities for differentially subordinate harmonic functions and martingales. *Canad. Math. Bull.*, 55(3):597–610, 2012.
- [140] A. Osękowski. Sharp martingale and semimartingale inequalities, volume 72 of Instytut Matematyczny Polskiej Akademii Nauk. Monografie Matematyczne (New Series). Birkhäuser/Springer Basel AG, Basel, 2012.
- [141] A. Osękowski. Weak norm inequalities for martingales and geometry of Banach spaces. *Statist. Probab. Lett.*, 82(3):411–418, 2012.
- [142] A. Osękowski. Weak Φ-inequalities for the Haar system and differentially subordinated martingales. *Proc. Japan Acad. Ser. A Math. Sci.*, 88(9):139–144, 2012.
- [143] A. Osękowski. Sharp inequalities for martingales with values in  $\ell_{\infty}^{N}$ . *Electron. J. Probab.*, 18:no. 73, 19, 2013.
- [144] A. Osękowski. Sharp inequalities for Hilbert transform in a vector-valued setting. *Math. Inequal. Appl.*, 18(4):1561–1573, 2015.
- [145] A. Osękowski. Inequalities for Hilbert operator and its extensions: the probabilistic approach. Ann. Probab., 45(1):535–563, 2017.
- [146] A. Osękowski and I.S. Yaroslavtsev. The Hilbert transform and orthogonal martingales in Banach spaces. arXiv:1805.03948, 2018.
- [147] J. Parcet and N. Randrianantoanina. Gundy's decomposition for noncommutative martingales and applications. *Proc. London Math. Soc.* (3), 93(1):227–252, 2006.
- [148] S. Peszat and J. Zabczyk. Stochastic partial differential equations with Lévy noise. An evolution equation approach, volume 113 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, Cambridge, 2007.
- [149] S.K. Pichorides. On the best values of the constants in the theorems of M. Riesz, Zygmund and Kolmogorov. *Studia Math.*, 44:165–179. (errata insert), 1972. Collection of articles honoring the completion by Antoni Zygmund of 50 years of scientific activity, II.
- [150] I. Pinelis. Optimum bounds for the distributions of martingales in Banach spaces. Ann. Probab., 22(4):1679–1706, 1994.
- [151] G. Pisier. Personal communication.

- [152] G. Pisier. Martingales with values in uniformly convex spaces. Israel J. Math., 20(3-4):326–350, 1975.
- [153] G. Pisier. Martingales in Banach spaces, volume 155. Cambridge University Press, 2016.
- [154] D. Pollard. Convergence of stochastic processes. Springer Series in Statistics. Springer-Verlag, New York, 1984.
- [155] P.E. Protter. Stochastic integration and differential equations, volume 21 of Stochastic Modelling and Applied Probability. Springer-Verlag, Berlin, 2005. Second edition. Version 2.1, Corrected third printing.
- [156] D. Revuz and M. Yor. Continuous martingales and Brownian motion, volume 293 of Grundlehren der Mathematischen Wissenschaften. Springer-Verlag, Berlin, third edition, 1999.
- [157] M. Riedle and O. van Gaans. Stochastic integration for Lévy processes with values in Banach spaces. *Stochastic Process. Appl.*, 119(6):1952–1974, 2009.
- [158] M. Riesz. Sur les fonctions conjuguées. Math. Z., 27(1):218–244, 1928.
- [159] L.I. Ronkin. Vvedenie v teoriyu tselykh funktsiĭ mnogikh peremennykh. Izdat. "Nauka", Moscow, 1971.
- [160] L.I. Ronkin. Introduction to the theory of entire functions of several variables. American Mathematical Society, Providence, R.I., 1974. Translated from the Russian by Israel Program for Scientific Translations, Translations of Mathematical Monographs, Vol. 44.
- [161] H.P. Rosenthal. On the subspaces of  $L^p$  (p > 2) spanned by sequences of independent random variables. *Israel J. Math.*, 8:273–303, 1970.
- [162] J. Rosiński and Z. Suchanecki. On the space of vector-valued functions integrable with respect to the white noise. *Collog. Math.*, 43(1):183–201 (1981), 1980.
- [163] B.L. Rozovskii. Stochastic evolution systems, volume 35 of Mathematics and its Applications (Soviet Series). Kluwer Academic Publishers Group, Dordrecht, 1990. Linear theory and applications to nonlinear filtering, Translated from the Russian by A. Yarkho.
- [164] J.L. Rubio de Francia. Martingale and integral transforms of Banach space valued functions. In *Probability and Banach spaces (Zaragoza, 1985)*, volume 1221 of *Lecture Notes in Math.*, pages 195–222. Springer, Berlin, 1986.

- [165] K.-i. Sato. Lévy processes and infinitely divisible distributions, volume 68 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 2013. Translated from the 1990 Japanese original, Revised edition of the 1999 English translation.
- [166] I.R. Shafarevich and A.O. Remizov. *Linear algebra and geometry*. Springer, Heidelberg, 2013. Translated from the 2009 Russian original by David Kramer and Lena Nekludova.
- [167] A.V. Skorohod. Limit theorems for stochastic processes. Teor. Veroyatnost. i Primenen., 1:289–319, 1956.
- [168] E.M. Stein. Singular integrals and differentiability properties of functions. Princeton Mathematical Series, No. 30. Princeton University Press, Princeton, N.J., 1970.
- [169] E.M. Stein and G. Weiss. Introduction to Fourier analysis on Euclidean spaces. Princeton University Press, Princeton, N.J., 1971. Princeton Mathematical Series, No. 32.
- [170] D.W. Stroock and S.R.S. Varadhan. *Multidimensional diffusion processes*. Classics in Mathematics. Springer-Verlag, Berlin, 2006. Reprint of the 1997 edition.
- [171] L. Thibault. Continuity of measurable convex and biconvex operators. Proc. Amer. Math. Soc., 90(2):281–284, 1984.
- [172] E.C. Titchmarsh. Reciprocal formulae involving series and integrals. *Math. Z.*, 25(1):321–347, 1926.
- [173] J.M.A.M. van Neerven. Nonsymmetric Ornstein-Uhlenbeck semigroups in Banach spaces. J. Funct. Anal., 155(2):495–535, 1998.
- [174] M.C. Veraar. Stochastic integration in Banach spaces and applications to parabolic evolution equations. PhD thesis, TU Delft, Delft University of Technology, 2006.
- [175] M.C. Veraar. Continuous local martingales and stochastic integration in UMD Banach spaces. *Stochastics*, 79(6):601–618, 2007.
- [176] M.C. Veraar. Randomized UMD Banach spaces and decoupling inequalities for stochastic integrals. *Proc. Amer. Math. Soc.*, 135(5):1477–1486, 2007.
- [177] M.C. Veraar and I.S. Yaroslavtsev. Cylindrical continuous martingales and stochastic integration in infinite dimensions. *Electron. J. Probab.*, 21:Paper No. 59, 53, 2016.

- [178] M.C. Veraar and I.S. Yaroslavtsev. Pointwise properties of martingales with values in Banach function spaces. arXiv:1803.11063, to appear in High Dimensional Probability VIII, 2018.
- [179] G. Wang. Differential subordination and strong differential subordination for continuous-time martingales and related sharp inequalities. *Ann. Probab.*, 23(2):522–551, 1995.
- [180] F. Weisz. Martingale Hardy spaces with continuous time. In *Probability theory and applications*, volume 80 of *Math. Appl.*, pages 47–75. Kluwer Acad. Publ., Dordrecht, 1992.
- [181] F. Weisz. Martingale operators and Hardy spaces generated by them. *Studia Math.*, 114(1):39–70, 1995.
- [182] S. Yano. Notes on Fourier analysis. XXIX. An extrapolation theorem. J. Math. Soc. Japan, 3:296–305, 1951.
- [183] I.S. Yaroslavtsev. Weak L<sup>1</sup>-estimates for weakly differentially subordinated martingales. *In preparation*.
- [184] I.S. Yaroslavtsev. Martingale decompositions and weak differential subordination in UMD Banach spaces. arXiv:1706.01731, to appear in Bernoulli, 2017.
- [185] I.S. Yaroslavtsev. On the martingale decompositions of Gundy, Meyer, and Yoeurp in infinite dimensions. arXiv:1712.00401, to appear in Ann. Inst. Henri Poincaré Probab. Stat., 2017.
- [186] I.S. Yaroslavtsev. Brownian representations of cylindrical continuous local martingales. *Infin. Dimens. Anal. Quantum Probab. Relat. Top.*, 21(2):1850013, 25, 2018.
- [187] I.S. Yaroslavtsev. Burkholder–Davis–Gundy inequalities in UMD banach spaces. arXiv:1807.05573, 2018.
- [188] I.S. Yaroslavtsev. Even Fourier multipliers and martingale transforms in infinite dimensions. *Indag. Math.* (N.S.), 29(5):1290–1309, 2018.
- [189] I.S. Yaroslavtsev. Fourier multipliers and weak differential subordination of martingales in UMD Banach spaces. *Studia Math.*, 243(3):269–301, 2018.
- [190] Ch. Yoeurp. Décompositions des martingales locales et formules exponentielles. In Séminaire de Probabilités, X (Seconde partie: Théorie des intégrales stochastiques, Univ. Strasbourg, Strasbourg, année universitaire 1974/1975), pages 432–480. Lecture Notes in Math., Vol. 511. Springer, Berlin, 1976.

- [191] D.C. Youla. A normal form for a matrix under the unitary congruence group. *Canad. J. Math.*, 13:694–704, 1961.
- [192] A.C. Zaanen. Integration. North-Holland Publishing Co., Amsterdam; Interscience Publishers John Wiley & Sons, Inc., New York, 1967. Completely revised edition of An introduction to the theory of integration.
- [193] J. Zhu, Z. Brzeźniak, and E. Hausenblas. Maximal inequalities for stochastic convolutions driven by compensated Poisson random measures in Banach spaces. Ann. Inst. Henri Poincaré Probab. Stat., 53(2):937–956, 2017.
- [194] A. Zygmund. *Trigonometric series. Vol. I, II.* Cambridge Mathematical Library. Cambridge University Press, Cambridge, third edition, 2002. With a foreword by Robert A. Fefferman.

### **ACKNOWLEDGMENTS**

First of all, I want to thank my supervisor, Mark Veraar. Mark, thank you very much for all your help during the PhD time, for providing me with highly interesting research topics, for hours of spontaneous and inspiring discussions, for being extremely friendly, open, and supportive, and for being patient with my Russian style. I also thank Jan van Neerven for always being able to give an advice in any situation, and for our numerous thought-provoking conversations, both mathematical and philosophical.

I am very grateful to Sjoerd Dirksen, Carlo Marinelli, Adam Osekowski, and Stefan Geiss for their help in deeper understanding of different phenomena in stochastic and harmonic analysis. First, I would like to thank Sjoerd for our discussions concerning Burkholder-Rosenthal inequalities and the canonical decomposition. I thank Sjoerd and Carlo for a detailed explanation of random measure theory. I am very appreciative to Adam for explaining me a broad class of martingale techniques and for showing me various possible directions to think about. Finally, I thank Stefan very much for our fruitful conversations concerning the Hilbert transform and shift operators and for his invitation and being the host while my three months long stay at University of Jyväskylä which I enjoyed a lot and which was an unforgettable experience for myself. In addition I thank Anni, Antti, Christel, Diu, Miia, Thuan, and Vesa from the Jyväskylä team for their geniality.

I would like to thank Frank Redig for our discussions on martingale theory while I was a teaching assistant for his course "Martingales, Brownian motion and stochastic processes" and while we happened to take the same train from Delft to Leiden.

I also thank Beniamin Goldys, Tuomas Hytönen, Ildoo Kim, Martin Ondreját, Pierre Portal, and Fedor Sukochev for short conversations and for providing me with an opportunity to give talks at their home universities.

Of course nothing would happen without such a friendly and motivating surrounding at the university. I am grateful to all my colleagues for creating this surrounding. Many thanks to our professors Ben and Robbert, (former) associate and assistant professors, Bas, Dion, Dorothee, François, Jeroen, K.P., Marcel, Markus, Martijn, and Wolter, to postdocs Alex and Moritz, and to (former) PhD students Bart, Bas, Chiara, David, Elek, Emiel, Eni, Federico, Jan, Jonas, Konstantin, Lukas, Mario, Nick, Nikita, Richard, and Rik for having a lot of fun together and for hours spent on discussing mathematics and life. Bas, Chiel, Elek, Emiel, Jan, Lukas, Mario, Nick, Nikita, you are great officemates! Elek and Nikita, thanks for discussing the Russian literature. Alex, Bas, Dorothee, Emiel, and Nick, thanks for your detailed descriptions of harmonic and functional analysis to a probabilist. Chiara and Richard, I enjoyed bouldering with you guys very much! Alex, it was extremely pleasant to be an assistant for your course "Fourier analysis", and of course I thank students Arthur, Guillermo, Jan, and Riki for being enthusiastic during the course and for their questions sometimes very far from the topic of the course. Cindy, Dori, Kees, and Xiwei: thank you very much for your help in making our work process relaxed and easy-going.

I also want to thank my scientific aunt, Sonja, for helping me in finding the position in Delft and for providing support especially during the beginning of my PhD.

Now I need to switch to Russian.

Я бы хотел обратиться к моим маме и сёстрам. Вы с детства поддерживали меня во всех моих начинаниях, помогали в трудностях и с пониманием относились к моим экспериментам, хотя порой вам приходилось очень трудно. Спасибо вам за это большое.

Я благодарю свою жену Аню за тепло и заботу. Спасибо за то, что ты рядом. Ну и конечно спасибо моим школьным и университетским товарищам за наши частые встречи и за поддержание духа авантюризма.

I wish everybody the best. Have fun, party like a Russian, and do not be afraid of something new!

Delft, February 2019

Ivan S. Yaroslavtsev

### Ivan YAROSLAVTSEV

Ivan Sergeevich Yaroslavtsev was born on January 1, 1991, in Neftekamsk, the Soviet Union. He completed Kolmogorov school in Moscow in 2009. In the same year he began his studies in Mathematics at Lomonosov Moscow State University. He obtained his Specialist degree (which is equivalent to MSc) in 2014. His diploma thesis "Polylinear functions on infinite-dimensional spaces with measures" was done under the supervision of Professor Vradimir Bogachev and was devoted to solving an open problem concerning polynomials and multilinear forms on linear spaces with Gaussian measures posed by Heinrich von Weizsäcker in the late 1980's (see [21, p. 275] for the problem and [6] for its solution).

In March 2015 he started his PhD research under the supervision of Professor Mark Veraar and Professor Jan van Neerven at the Delft University of Technology. Part of this research was carried out during his stays at University College London, RWTH Aachen University, University of Warsaw, Australian National University, and a three month stay at University of Jyväskylä, Finland.

Ivan Yaroslavtsev is a referee for *Advances in Operator Theory, Electronic Communications in Probability, Indagationes Mathematicae, SIAM Journal on Mathematical Analysis,* and *Zeitschrift für angewandte Mathematik und Physik,* as well as a reviewer for *MathSciNet*.

# LIST OF PUBLICATIONS

- L.M. Arutyunyan and I.S. Yaroslavtsev. On measurable polynomials on infinite-dimensional spaces. *Dokl. Ross. Akad. Nauk* 449, no. 6, 627–631 (in Russian); English transl.: *Dokl. Math.* 87, no. 2, 214–217, 2013.
- (2) L.M. Arutyunyan, E.D. Kosov, and I.S. Yaroslavtsev. On some properties of polynomials measurable with respect to a Gaussian measure. *Dokl. Ross. Akad. Nauk* 457, no. 2, 131–135. (in Russian); English transl.: *Dokl. Math.* 90, no. 1, 419–423, 2014.
- (3) I.S. Yaroslavtsev. On the asymmetry of the past and the future of the ergodic Z-action. *Mat. Zametki* 95, no. 3, 479–480 (in Russian); English transl.: *Math. Notes* 95, no. 3-4, 438–440, 2014.
- (4) L.M. Arutyunyan, E.D. Kosov, and I.S. Yaroslavtsev. On convex compact sets of positive measure in linear spaces. *Math. Notes* 96, no. 3-4, pp 448–450, 2014.
- (5) M.C. Veraar and I.S. Yaroslavtsev. Cylindrical continuous martingales and stochastic integration in infinite dimensions. *Electron. J. Probab.* 21:53 pp., 2016.
- (6) I.S. Yaroslavtsev. Brownian representations of cylindrical continuous local martingales. *Infin. Dimens. Anal. Quantum Probab. Relat. Top.* 21, no. 2, 1850013, 25 pp., 2018.
- (7) I.S. Yaroslavtsev. Fourier multipliers and weak differential subordination in UMD Banach spaces. *Studia Math.* 243, no. 3, 269–301, 2018.
- (8) I.S. Yaroslavtsev. Martingale decompositions and weak differential subordination in UMD Banach spaces. *arXiv:1706.01731, to appear in Bernoulli,* 2017.
- (9) S. Dirksen and I.S. Yaroslavtsev. *L<sup>q</sup>*-valued Burkholder-Rosenthal inequalities and sharp estimates for stochastic integrals. *arXiv*:1707.00109, 2017.
- (10) I.S. Yaroslavtsev. Even Fourier multipliers and martingale transforms in infinite dimensions. *arXiv*:1710.04958, to appear in Indag. Math. (N.S.), 2017.
- (11) I.S. Yaroslavtsev. On the martingale decompositions of Gundy, Meyer, and Yoeurp in infinite dimensions. *arXiv:1712.00401, to appear in Ann. Inst. Henri Poincaré Probab. Stat.*, 2017.

- (12) M.C. Veraar and I.S. Yaroslavtsev. Pointwise properties of martingales with values in Banach function spaces. *arXiv:1803.11063, to appear in High Dimensional Probability VIII,* 2018.
- (13) A. Osękowski and I.S. Yaroslavtsev. The Hilbert transform and orthogonal martingales in Banach spaces. arXiv:1805.03948, 2018.
- (14) I.S. Yaroslavtsev. Burkholder–Davis–Gundy inequalities in UMD Banach spaces. arXiv:1807.05573, 2018.
- (15) N. Lindemulder, M.C. Veraar, and I.S. Yaroslavtsev. The UMD property for Musielak–Orlicz spaces. arXiv:1810.13362, to appear in Positivity and Noncommutative Analysis. Festschrift in honour of Ben de Pagter on the occasion of his 65th birthday, Trends in Mathematics, 2019.
- (16) I.S. Yaroslavtsev. On strongly orthogonal martingales in UMD Banach spaces. arXiv:1812.08049, 2018.
- (17) S. Dirksen, C. Marinelli, and I.S. Yaroslavtsev. Stochastic evolution equations in *L<sup>p</sup>*-spaces driven by jump noise. *In preparation*.
- (18) S. Geiss and I.S. Yaroslavtsev. Dyadic and stochastic shifts and Volterra-type operators. *In preparation*.

For notes